ANL-8144

NDF

239

ned Plu

+ ENDY

ANL-8144 (ENDF 239)

MC²-2: A Code to Calculate Fast Neutron Spectra and Multigroup Cross Sections

by

H. Henryson II, B. J. Toppel, and C. G. Stenberg

BASE TECHNOLOGY





ARGONNE NATIONAL LABORATORY, ARGONNE, ILLINOIS Prepared for the U. S. ENERGY RESEARCH AND DEVELOPMENT ADMINISTRATION under Contract W-31-109-Eng-38

DISCLAIMER

This report was prepared as an account of work sponsored by an agency of the United States Government. Neither the United States Government nor any agency Thereof, nor any of their employees, makes any warranty, express or implied, or assumes any legal liability or responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately owned rights. Reference herein to any specific commercial product, process, or service by trade name, trademark, manufacturer, or otherwise does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or any agency thereof. The views and opinions of authors expressed herein do not necessarily state or reflect those of the United States Government or any agency thereof.

DISCLAIMER

Portions of this document may be illegible in electronic image products. Images are produced from the best available original document.

The facilities of Argonne National Laboratory are owned by the United States Government. Under the terms of a contract (W-31-109-Eng-38) between the U. S. Energy Research and Development Administration, Argonne Universities Association and The University of Chicago, the University employs the staff and operates the Laboratory in accordance with policies and programs formulated, approved and reviewed by the Association.

MEMBERS OF ARGONNE UNIVERSITIES ASSOCIATION

The University of Arizona Carnegie-Mellon University Case Western Reserve University Loyola University The University of Chicago University of Cincinnati Illinois Institute of Technology University of Illinois Indiana University Iowa State University The University of Iowa

Kansas State University The University of Kansas Marquette University Michigan State University The University of Michigan University of Minnesota University of Missouri Northwestern University University of Notre Dame

The Ohio State University Ohio University The Pennsylvania State University Purdue University Saint Louis University Southern Illinois University The University of Texas at Austin Washington University Wayne State University The University of Wisconsin

NOTICE-

This report was prepared as an account of work sponsored by the United States Government. Neither the United States nor the United States Energy Research and Development Administration, nor any of their employees, nor any of their contractors, subcontractors, or their employees, makes any warranty, express or implied, or assumes any legal liability or responsibility for the accuracy, completeness or usefulness of any information, apparatus, product or process disclosed, or represents that its use would not infringe privately-owned rights. Mention of commercial products, their manufacturers, or their suppliers in this publication does not imply or connote approval or disapproval of the product by Argonne National Laboratory or the U. S. Energy Research and Development Administration.

> Printed in the United States of America Available from National Technical Information Service U. S. Department of Commerce 5285 Port Royal Road Springfield, Virginia 22161 Price: Printed Copy \$12.00; Microfiche \$2.25

ANL-8144 (ENDF 239) LMFBR Physics (UC-79d)

ARGONNE NATIONAL LABORATORY 9700 South Cass Avenue Argonne, Illinois 60439

MC²-2: A Code to Calculate Fast Neutron Spectra and Multigroup Cross Sections

by

H. Henryson II, B. J. Toppel, and C. G. Stenberg

Applied Physics Division

This report was prepared as an account of work sponsored by the United States Government. Neither the United States nor the United States Energy Research and Development Administration, nor any of their employees, nor any of their contractors, subcontractors, or their employees, makes any warranty, express or implied, or assumes any legal liability or responsibility for the accuracy, completeness or usefulness of any information, apparetus, product or process disclosed, or represents that it's use would not infninge privately owned rights.

-NOTICE

June 1976



THIS PAGE WAS INTENTIONALLY LEFT BLANK

TABLE OF CONTENTS

			-													Page
CODE	ABSTRA	ст	• • • .•	• • •	• • •	•••	• •	••	•	•••	•	•	• .	•••	•	xi
ABST	RACT .		••••	• •	•••	•••	• •	••	•	••	•	•	•	••	•	xiv
LIST	OF TABI	LES	• • • •	•••		••	• •	••	•	• •	•	•	•	•	•	
LIST	OF FIG	JRES .		• • •	• • •	•••	•••	•••	•		•	•	• •	•	•	
I.	INTRODU	JCTION			• • •			••	•	••	•	•	• •	•	•	1
II.	FUNDAME	ENTAL MO	DDE SPE	CTRUM	CALCU	ULATI	ON	• •		• •	•			•	•	4
	II.A.	P ₁ and	B ₁ Ext	ended	Trans	sport	E Equ	uati	ons			•	• •	•	•	4
	II.B.	Source	Term .		• • •	• •	••	••	•	- •	•	•	• •	•	•	7
	II.C.	Contin	uous Sl	owing	Down	Theo	ory.	• •	•	•. •	•	•	• •	•	•	8
-	II.D.	Continu	uous Sl	owing	Down	Mode	erati	ing	Par	ame	ter	s	• •	•	•	13
	. 11	.D.1.	Goertz	el-Gre	uling	g Mod	lerat	ting	Pa	ram	ete	ers	• •	•	•	14
	-11	L.D.2.	Improv	ed Goe ters	ertze.	l−Gre	eulir	ng M	ode	rat:	íng	5		_	•	15
	IT.E.	Narrow	Resona	nce At	•••	•••	•••	• •	•	••	•	•	•••	•	•	21
	IT.F.	Ultra-1	Fine-Gr		natio	 			-						•	23
	 TT	.F.1.	Multio	roup E	lauati	ions			•		•	•			•	25
	II	.F.2.	Contin	uous S	lowir	ng Do	wn E	Iqua	tio:	ns.	•	•	•••	•	•	27
	II.G.	Elastic	c Scatt	ering	Trans	sfer	Matr	ix.	•	•••	•	•	• •	•	•	32
	II	.G.1.	Light 1	Elemer	nts.		• •	• •	•		٠	•	• •	•	. •	32
	II TT		Heavy Hydrog	Elemer en Fla	its .	 Scat	 tori	• •	•••	••	•	•	•••	•	•	36 38
		Inelast	tic and	(n 2r		ottor	ino		•				• •	•	•	39
	тт	и 1	Tabula	tod Fu	motic		1116	•••	•	•••	•	•	•••	•	•	30
	,II	.H.2.	Evapora	ation	Spect	rum	•••	•••	•		•		•••	•	•	40
	11	.н.з.	Discre	te Lev	vels.	• •	•••	• •	•	• •	٠	•	••	÷	•	40
		II.H	1.3.a.	Appro Energ	oximat v-Ang	te Tr gle C	eatn	ent	Ne; ion	gled	ti	ng		•		41
,		II.H	I.3.b.	Rigor	ous 1	reat	ment	: As	sum	ing	· .					
		•		Isotr	opic	Scat	teri	ng.	•	• •	•	•	• •	•.	•	42
	II.I.	Ultra-H	fine-Gro	oup Ma	crose	copic	Dat	:a .	•		•	•	••	•	•	44
	II	.1.1.	Fissio	n Sour	ce Da	ita.	• •	•••	•	• •	•	•	•••	•	•	44
	11	1.2.	Cross	sine-6 Sectio	n	Exte	naec	• •	ansj	port	•	•		•	•	45
	11	.1.3.	Improv	ed Goe	ertzel	L−Gre	ulir	ng M	ode	rati	ing					10
			rarame	cers .					•							40

	•	50
II.J.1. Ultra-Fine-Group Spectrum Solution	•	50
II.J.2. Eigenvalue Solution	•	53
II.J.3. Inhomogeneous Source Solution	•	54 55
II.J.4. Buckling Search to Cittical	•	56
	. '	50
II.K.1. Broad-Group Inelastic and (n,2n)		56
11.K.2. Broad-Group Absorption Cross Sections	•	57
II.K.3. Elastic Scattering Cross Sections	•	58
II.K.4. Total Cross Section	•	60
II.K.5. Transport Cross Section	•	62
II.K.5.a. Inconsistent P_1 or B_1 Spectrum	•	62
II.K.5.b. Consistent P_1 or B_1 Spectrum	•	63
II.K.6. Average Number of Neutrons per Fission	•	63
II.K.7. Fission Spectrum Distribution	•	64
II.K.8. XS.ISO Cross Sections	•	64
II.L. Broad-Group Spectrum	.• .	64
III. CALCULATION OF RESOLVED RESONANCE INTEGRALS	•	67
III.A. General Formulation	•	67
III.B. Calculation of Isolated Resonance Integrals	•	78
III.C. Calculations of Overlap Integrals	•	79
III.C.1. Selection of Overlapping Resonances	•	79
III.C.2. Asymptotic Algorithms	•	80
III.C.3. Gauss-Jacobi Quadrature Algorithms	•	89
IV. CALCULATION OF UNRESOLVED RESONANCE INTEGRALS	•	94
IV.A. General Formulation	•	94
IV.B. Evaluation of Self-Overlap Term	•	99
IV.B.1. Evaluation of L_1	•	100
IV.B.1.a. Asymptotic Algorithms for L ₁	•	102
-	•	103
IV.B.1.b. Non-Asymptotic Algorithms for L ₁		
IV.B.1.b. Non-Asymptotic Algorithms for L_1 . IV.B.2. Evaluation of L_2	•	105
IV.B.1.b. Non-Asymptotic Algorithms for L ₁ IV.B.2. Evaluation of L ₂	•	105 106

Page Evaluation of Unresolved Resonance Cross IV.C. 107 Unresolved Resonance Integrals and Resolved-IV.D. Unresolved Resonance Interaction 108 Ultra-Fine-Group Unresolved Resonance Cross IV.E. Sections 109 . . V. RABANL, HYPER-FINE-GROUP INTEGRAL TRANSPORT THEORY NEUTRON 112 V.A. 112 Calculation of Elastic Slowing-Down Sources. . . . V.B. 112 118 V.C. V.C.1. One Region (Homogeneous) Problems 118 V.C.2. Multi-Region (Heterogeneous) Problems . . . 122 122 V.C.2.a. Cylindrical Geometry V.C.2.b. 130 Calculation of Hyper-Fine-Group Cross Sections . . . 132 V.D. 133 V.D.1. V.D.2. Resolved Resonance Algorithms 133 133 V.D.2.a. Single Level Breit Wigner. V.D.2.b. Multilevel Breit Wigner. 135 V.D.2.c. Multilevel Adler-Adler 137 V.E. 138 VI. 141 141 VI.A. VI.B. Job Control Language Procedure ARCSP015. 142 155 VI.C. VI.D. 161 161 VI.D.1. Homogeneous Reference Problem VI.D.2. Hyper-Fine-Group Homogeneous Integral 161 Hyper-Fine-Group Heterogeneous Integral VI.D.3. 162 163 VI.E.

v

		Page	•
VII.	PROGRAMMI	NG INFORMATION	
	VII.A. Pr	ogram Structure	
	VII.B. AR	C System Routines	•
•	VII.C. CC	CC Standard Subroutines	I
	VII. VII. VII.	C.1. SEEK	
	VII.D. Ma	chine Language Routines	6
	VII. VII. VII. VII. VII. VII. VII. VII.	D.1. ALLUC. 193 D.2. LOCF .*. 193 D.3. GOWEST 194 D.4. FILEID 194 D.5. CVD. 194 D.6. TIME 194 D.7. CLOCK. 194 D.8. DATE 194 D.9. SECOND 194 D.10. ABEND. 194 D.11. JOBID. 195 D.12. FXP. 195 D.13. DISPOS 195 D.14. TRACER 195 rogram Implementation 195	
	VII. VII.	.E.1. IBM Code Implementation) L
APPEI	NDICES)
Α.	Special H	Function Evaluation)
	A.I.	Legendre Functions of the Second Kind	1
	A.II. A.III.	A Fast Exponential Subroutine-FXP 230 Calculation of Doppler-Broadened Line Shape Functions 237	,
	A.IV.	Numerical Methods Used in Calculating the Isolated Resonance Integral)
	A.V.	Quadratures for Statistical Integration 244	ŧ
	A.VI.	Numerical Methods Used in Calculating Slab Collision Rates	5
		A.VI.a. Exponential Integrals	5 7 7
	A.VII.	Cylindrical Transmission Probabilities 251	L

			Page
в.	MC ² -2 BCD	Input Files	253
	B.I.	A.MCC2	254
	B.II.	A.NIP	276
	B.III.	A.STP015	280
	B.IV.	ACSE12	283
· .	B.V.	ACS009	285
C.	MC ² -2 Bin	ary Interface Files	287
	C.I.	ISOTXS	288
	C.II.	MCC2F1	294
	C.III.	MCC2F2	298
• •	C.IV.	MCC2F3	302
,	C.V.	MCC2F4	305
	c.vi.	MCC2F5	. 315
	C.VII.	MCC2F6	318
	C.VIII.	MCC2F7	324
	C.IX.	MCC2F8	326
	с.х.	xs.iso	329
D.	MC ² -2 Bina	ary Files	336
	D.I.	ATNUAT	337
	D.II.	BC	342
	D.III.	BIGXS(I)	343
	D.IV.	GEOM1	345
	D.V.	IRESCS	348
	D.VI.	LORENZ	, 352
	D.VII.	MACTOT	355
	D.VIII.	MICTOT	359
	D.IX.	OLDSGS	360
	D.X.	OPTICL	361
	D.XI.	PRBCHI	363
	D.XII.	PRBSPC	366
	D.XIII.	RESDAT	379

Page

	D.XIV.	RESINT	•	•	•	•	•	•	•	•	•	•	•	•	•	• ·	•	•	•	•	• :	•	•	•	381
	D.XV.	SCR001	•	•	•	•	•	•	•	•.	•	•	•	•	• ,	•	•	•	•	•	•	•	•	•	385
	D.XVI.	SCR002	•	•	•	•	•	•	٠	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	390
	D.XVII.	SCR003	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	٠	•	•	•	•	•	. •	395
	D.XVIII.	SCR004	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	ŗ	•	٠	•	•	•	407
	D.XVIX.	SCR005	•	•	•	•	•	•	•	•	•	•	•	•	÷	•	•	•	•	•	•	•	•	•	409
	D. XX.	SIGMAP	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•		•	•	•	•	•	•	411
-	D.XXI.	SMSIGS	•	•	•	•	.•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	413
	D.XXII.	SPECTR	•	•	•	•	.•	•	•	•		•	•	•	•	•		•		•	•	•	•	•	416
	D.XXIII.	SPECXS	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	418
	D.XXIV.	SRATES	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	421
	D.XXV.	UNREG.	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	424
	D.XXVI.	UNRES.	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	427-
Ε.	BPOINTER,	A Dyna	ni	c	St	ora	ag	e /	41	100	cat	tic	m	Pı	rog	gra	am	•	.•	•	•	•		•	431
	E.I.	Descri	pt	io	n (of	S	ubj	pro	ogi	rai	n E	a	źka	age	≥.	•	•	•	•	•	•	•	•	432
	. •	E.I.a. E.I.b.		IB CD(A1 A1) 100 100	cat cat	tio tio	on on	•	•	•	•.	•	•	•	•	•	•	•	•	•	•	433 ⁻ 433
F.	MC ² -2 Lib:	rary Ge	ne	rat	ti	on	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	438
ACKN	OWLEDGMENT	s	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•		•	442
REFE	RENCES .				_			:									_				-				- 443

LIST OF FIGURES

Title No. Page 1 MC^2-2 Program Flow 3 2 24 Elastic Scattering Group Structure 34 3 Gauss-Jacobi Quadrature Point Selection. . . 4 90 . . 5 Unresolved Resonance Energy Grid 110 6 Linearization of Ultra-Fine-Group Data 121 7 Slab Geometry Coordinates. 123 8 Slab Geometry Optical Distance 126 . . 9 MC²-2 Module Flow Diagram. 143 MC²-2 Path Driver Listing. 144 10 148 11 MC^2-2 JCL Procedure. 164 12 Minimum JCL Execution Deck 164 13 Input for Sample Problem 1 165 14 Input for Sample Problem 2 Input for Sample Problem 3 166 15 . 16 IBM Overlay Control. 183 185 CDC Segmentation Loader Directives 17 199 18. Preallocation of Object Library. 19 Compilation of FORTRAN Subprograms . . 199 20 199 21 Copy Object Module From File 3 of IBM Code Center 199 Copy of MC^2 -2 IBM Binary Library Files 22 200 23 Compilation and Assembly of CDC Source Code to Generate Object Code 227 Build Segments from Object Code Using Segmentation 24 228 . . . CDC Execution of MC^2-2 Problem 25 229 26 Gauss-Jacobi Quadrature Selection. 241 27 435 . -Generation of MC²-2 Library From Two X-Mode Binary 28 440 Generation of MC^2-2 Library From Three BCD Data Tapes. 29 441

ix

LIST OF TABLES

No.	Title	Page
I	Coefficients K_{nn} , (α)	. 20
II	Heavy Mass Matrix Elements A_n^{ℓ} (j+k)	. 37
III	BCD Input Data Sets	. 156
IV	Prestored Group Structures	. 157
v	MC ² -2 Error Messages	167
VI	MC ² -2 File Information	. 187
VII	IBM Code Center Tape Description	. 198
VIII	Function of MC^2-2 Subprograms	. 201
IX	MC ² -2 ENDF/B-IV Library Files	. 217
Х	CDC Code Center Tape Description	. 224
XI	Function of MC^2-2 Subprograms for CDC Code	. 225
XII	Legendre Function Evaluation	. 235
XIII	Ten Point Quadrature Weights and Abscissae for Statistical Integration	. 245
XIV	BPOINTER Subprogram Descriptions	• 436 ·

٧.

х

MC²-2: A CODE TO CALCULATE FAST NEUTRON SPECTRA AND MULTIGROUP CROSS SECTIONS

CODE ABSTRACT

1. Program Identification: MC^2-2

approximation is not valid.

- Computer for which program is designed and others on which it is operable: IBM 370/195, any IBM OS system with 600K core, CDC 7600.
- 3. Description of Function: MC²-2 solves the neutron slowing down equations to determine spectra for use in generating multigroup neutron cross sections.
- 4. Method of Solution: The extended transport P_1 , B_1 ,

consistent P₁, and consistent B₁ fundamental mode ultra-fine-group equations are solved using continuous slowing down theory and multigroup methods. Fast and accurate resonance integral methods are used in the narrow resonance resolved and unresolved resonance treatments. Multigroup neutron cross sections are generated for arbitrary group structures. A hyper-fine-group integral transport slowing down calculation is available on option to treat the low energy spectrum where the narrow resonance

- 5. Restrictions: Variable dimensioning is used throughout the program so that computer core requirements depend on a variety of problem parameters. Space requirements range from 400K bytes to 800K bytes on IBM equipment depending on the complexity of the problem.
- Running Time: An 1740 group consistent P₁ homogeneous twelve isotope problem with 27 broad groups requires about 4.2 min. CPU and 6 min. PP time on an IBM 370/195. The same problem requires approximately 30% less CPU time on the CDC 7600.
- 7. Unusual Features of the Program: Extreme flexibility is provided in specifying the rigor of a calculation including a choice of four distinct slowing down treatments: multigroup, improved and standard Greuling-Goertzel continuous slowing down, and integral transport theory. All binary data transfers are localized in standard subroutines REED/RITE. Broad group cross section files may be generated in the ARC System XS.ISO⁽¹⁾ and/or CCCC ISOTXS⁽²⁾ formats.

xi

- 8. Related and Auxiliary Programs: Input data files required by MC²-2 may be generated from ENDF/B data⁽³⁾ by the code ETOE-2⁽⁴⁾. The hyper-fine-group integral transport theory module of MC²-2, RABANL, is an improved version of the RABBLE⁽⁵⁾/RABID⁽⁶⁾ codes. Many of the MC²-2 modules are used in the SDX⁽⁷⁾ code.
- 9. Status:
- 10. References: H. Henryson II, B. J. Toppel, C. G. Stenberg, "MC²-2: A Code to Calculate Fast Neutron Spectra and Multigroup Cross Sections," Argonne National Laboratory, ANL-8144 (1976).
- 11. Machine Requirements: A large amount of fast peripheral storage is required. Core requirements depend on problem complexity but virtually any reasonable problem may be executed on IBM equipment with 800K bytes or CDC equipment with 50,000 words of SCM and 100,000 words of directly addressable LCM.
- 12. Programming Language Used: FORTRAN IV. Both IBM and CDC versions of MC^2-2 contain a few Assembler language routines.
- 13. Operating System or Monitor under which Program is Executed: The IBM version of MC²-2 may be executed under OS or VS operating systems and compiled using the Fortran H or program product compilers with the highest level of optimization. The CDC 7600 version of MC²-2 has been implemented on both the LRL Berkeley and Brookhaven National Laboratory computers with their special COKE/SCOPE operating systems. The SEGMENTATION LOADER is required and directly addressable LCM is used. The code was compiled using the FORTRAN Extended Compiler under OPT=1 optimization.

14. Any Other Programming or Operating Information or Restrictions:

15. Name and Establishment of Authors:

H. Henryson II B. J. Toppel C. G. Stenberg Applied Physics Division Argonne National Laboratory Argonne, Illinois 60439

- 16. Material Available: Separate tapes are available for the IBM and CDC versions of MC²-2. The MC²-2 package includes:
 - i) Source decks
 - ii) Test problem input decks
 - iii) Test problem output
 - iv) Eight binary library files processed from ENDF/B-IV
 - v) Source code and BCD library files to generate binary libraries
 - vi) Reference report

17. Category: B

Keywords: cross sections, group constants, spectra, multigroup, resonance, B_{I} method, P_{I} method, slowing

down, infinite media, homogeneous, heterogeneous, ENDF/B, cell calculation

xiii

MC²-2: A Code to Calculate Fast Neutron Spectra and Multigroup Cross Sections

H. Henryson II, B. J. Toppel, and C. G. Stenberg

Applied Physics Division Argonne National Laboratory Argonne, Illinois 60439

ABSTRACT

 $MC^{2}-2$ is a program to solve the neutron slowing down problem using basic neutron data derived from the ENDF/B data files. The spectrum calculated by MC^2-2 is used to collapse the basic data to multigroup cross sections for use in standard reactor neutronics codes. Four different slowing down formulations are used by MC²-2: multigroup, continuous slowing down using the Goertzel-Greuling or Improved Goertzel-Greuling moderating parameters, and a hyper-fine-group integral transport calculation. Resolved and unresolved resonance cross sections are calculated accounting for self-shielding, broadening and overlap effects. This document provides a description of the MC²-2 program. The physics and mathematics of the neutron slowing down problem are derived and detailed information is provided to aid the MC^2-2 user in preparing input for the program and implementation of the program on IBM 370 or CDC 7600 computers.

I. INTRODUCTION

 MC^2-2 is a program for solving the neutron slowing down problem to determine a detailed spectrum for use in deriving multigroup cross sections. The code has been developed to satisfy the need for a rigorous and computationally efficient capability which can serve as a standard for fast reactor calculations. Recent advances in neutron slowing down theory, resonance theory, and numerical methods have been incorporated into the MC^2-2 calculation. A large number of options are available which permit great flexibility in specifying the rigor of a calculation.

The MC^2-2 code was developed at Argonne National Laboratory in the ARC System⁽¹⁾ environment. The programming was performed with strict adherence to the standards established by the Committee on Computer Code Coordination⁽²⁾. As a consequence the modifications required to develop standalone versions of the code for alternative computer configurations are minor. Both IBM and CDC standalone versions of the code have been implemented. The program is structured in eight principal modules (overlays):

- 1) Input Processor (CSI010)
- 2) Unresolved Resonance Calculation (CSC004)
- 3) Resolved Resonance Calculation (CSC005)
- 4) Resolved-Unresolved Resonance Interaction (CSC006)
- 5) Macroscopic Data Processing (CSC007)
- 6) Ultra-Fine-Group Spectrum Calculation and Broad-Group Cross Section Collapse (CSC008)
- 7) Hyper-Fine-Group Integral Transport Spectrum Calculation - RABANL (CSC009)
- 8) Broad-Group Spectrum Calculation (CSC010)

Figure 1 indicates the program flow through these modules. The basic data required by $MC^{2}-2$ are structured in eight data files. The file formats are provided in Appendix C. These data files may be created from the ENDF/B data⁽³⁾ by the program ETOE-2⁽⁴⁾. Multigroup cross sections generated by $MC^{2}-2$ are processed in the ARC System XS.ISO⁽¹⁾ and/or the CCCC ISOTXS⁽²⁾ formats. These structures are included in Appendix C for the sake of completeness.

The major features of MC^2-2 include:

- 1. the ability to specify the rigor of a calculation through input options,
- 2. P₁, B₁, consistent P₁ and consistent B₁ extended transport theory algorithms.

- 3. multigroup, improved or standard Greuling-Goertzel continuous slowing down theory solutions,
- 4. hyper-fine-group integral transport theory solutions,
- 5. flexible broad group structures in a choice of formats,
- 6. consistent treatment of all ENDF/B Version III and Version IV data formats,
- 7. fast and accurate resonance integral treatment of Breit-Wigner and Adler-Adler resolved resonances,
- 8. fast and accurate treatment of unresolved resonances including interference scattering and same-sequence overlap effects,
- 9. isotope dependent fission spectra,
- 10. inhomogeneous source calculation,
- 11. buckling search or group dependent buckling calculations,
- 12. flexible output options.

This report is organized into seven chapters which are intended to provide the physics, mathematics and code user documentation for the program MC^2-2 . Chapter II contains the theoretical development of all the methods and approximations used in the ultra-fine-group fundamental mode spectrum calculation of MC^2-2 . Chapter III contains the information relevant to the calculation of resonance integrals from resolved resonance parameters and Chapter IV is concerned with the calculation of cross sections and resonance integrals from average parameters in the unresolved resonance energy regions. Chapter V describes the RABANL module of the code MC^2-2 . The RABANL calculation is a hyper-fine-group integral transport slowing down calculation which treats resolved resonance absorption rigorously. The

module is modeled after the resonance absorption codes RABBLE⁽⁵⁾ and RABID.⁽⁶⁾ Chapter VI provides a guide for user application and Chapter VII contains information relevant to the programming of MC²-2.



Fig. 1. MC²-2 Program Flow

II. FUNDAMENTAL MODE SPECTRUM CALCULATION

A. P_1 and B_1 Extended Transport Equations

The time independent transport equation is written

$$\nabla \cdot \underline{\Omega} \psi + \Sigma_{\underline{t}} \psi(\underline{r}, u, \underline{\Omega}) = \iiint du' d\Omega' \psi(\underline{r}, u', \underline{\Omega}') \cdot \Sigma_{\underline{c}} (u' \neq u, \underline{\Omega} \cdot \underline{\Omega}') + S(\underline{r}, u)/4\pi \qquad (II.1)*$$

where ψ is the flux defined such that $\psi dV dudN$ is the flux in the volume dV about <u>r</u>, in the element of solid angle $d\Omega$ about Ω , in the lethargy range du about u. The lethargy u is defined as $ln(E_O/E)$. Similarly S is the isotropic source density in the same element of phase space and includes contributions due to fission and/or sources independent of the flux ψ . The macroscopic total cross section is denoted by Σ_t and the macroscopic scattering transfer cross section by Σ_s . If one assumes a homogeneous mixture so that the above quantities are not space dependent and makes use of the fundamental mode ansatz

$$\psi(\underline{\mathbf{r}}, \mathbf{u}, \underline{\Omega}) \equiv \psi(\mathbf{u}, \underline{\Omega}) e^{i\underline{\mathbf{B}}\cdot\underline{\mathbf{r}}}; \quad S(\underline{\mathbf{r}}, \mathbf{u}) \equiv S(\mathbf{u}) e^{i\underline{\mathbf{B}}\cdot\underline{\mathbf{r}}}$$
(II.2)

then Eq. II.1 may be written,

$$(\Sigma_{t} + i\underline{B} \cdot \underline{\Omega}) \ \psi(u, \underline{\Omega}) = \iiint du^{t} \ d\Omega' \ \psi(u', \underline{\Omega}^{t}) \ \Sigma_{s}(u^{t} \rightarrow u, \mu_{o})$$

+ $S(u)/4\pi$ (II.3)

where $\mu_0 \equiv \Omega \cdot \Omega'$. In order to simplify the notation, consider only plane geometry. Expanding the flux and scattering transfer cross section in spherical harmonics,

. . .

$$2\pi \ \psi(\mathbf{u}, \ \underline{\Omega}) = \psi(\mathbf{u}, \ \mu) = \sum_{\ell=0}^{\infty} \frac{2\ell+1}{2} \ \phi_{\ell}(\mathbf{u}) \ P_{\ell}(\mu)$$

$$\Sigma_{s}(\mathbf{u}' \rightarrow \mathbf{u}, \ \mu_{o}) = \sum_{\ell=0}^{\infty} \frac{2\ell+1}{2} \ \Sigma_{s}^{\ell}(\mathbf{u}' \rightarrow \mathbf{u}) \ P_{\ell}(\mu_{o}) \qquad (II.4)$$

where $P_{\ell}(\mu)$ is the ℓ^{th} order Legendre polynomial, substituting into Eq.II.3 and making use of the addition theorem for Legendre polynomials yields

*Equation numbering convention of the form N.n is used where N is the chapter number and n is the equation number in Chapter N.

-4-

$$iB\mu \ \psi(u, \ \mu) + \Sigma_{t} \ \psi(u, \ \mu) = \sum_{\ell=0}^{\infty} \frac{2\ell + 1}{2} \iint du' \ d\mu \ \Sigma_{s}^{\ell}(u' \rightarrow u) \ \phi_{\ell}(u') \ P_{\ell}(\mu) + \frac{S(u)}{2}$$
(II.5)

Multiplying by $P_{\ell}(\mu)$, integrating over the range of μ , and using the recursion relation gives

$$\frac{\ell+1}{2\ell+1} iB\phi_{\ell+1}(u) + \frac{\ell}{2\ell+1} iB\phi_{\ell-1}(u) + \Sigma_{t}\phi_{\ell}(u) = S(u) \delta_{\ell}^{\circ}$$
$$+ \int dE' \Sigma_{s}^{\ell}(u' \rightarrow u) \phi_{\ell}(u') \qquad \ell = 0, 1, \dots, \infty$$
$$\phi_{-1}(u) = 0 \quad . \tag{II.6}$$

Two well known approximations to this infinite set of coupled equations are the P_{N} and B_{N} methods⁽⁸⁾. The P_{N} approximation assumes

$$\phi_{N+1} = 0,$$
 (II.7a)

and the B_{N} approximation assumes

$$\int du' \Sigma_{s}^{\ell}(u' \rightarrow u) \phi_{\ell}(u') = 0 \qquad \ell > N$$

which is equivalent to taking

$$\phi_{N+1} = \phi_N \frac{Q_{N+1} (-\Sigma_t/iB)}{Q_N (-\Sigma_t/iB)}$$
(II.7b)

where $Q_{\ell}(\mathbf{x})$ is a Legendre function of the second kind. Using Eq.II.7a or II.7b, Eqs. II.6 may be written

$$\frac{\ell+1}{2\ell+1} iB\phi_{\ell+1} + \frac{\ell}{2\ell+1} iB\phi_{\ell-1} + \Sigma_{t}\phi_{\ell} = S(u) \delta_{\ell}^{0}$$

$$+ \int du' \Sigma_{s}^{\ell}(u' \rightarrow u) \phi_{\ell}(u') \qquad \ell = 0, 1, \dots, N-1$$

$$\frac{N}{2N+1} iB\phi_{N-1} + \gamma\Sigma_{t}\phi_{N} = \int du' \Sigma_{s}^{N}(u' \rightarrow u) \phi_{N}(u')$$

$$\phi_{-1} = 0$$

$$\gamma = \begin{cases} 1 & P_{N} \\ 1 + \frac{N+1}{2N+1} & \frac{iB}{\Sigma_{t}} & \frac{Q_{N+1} (-\Sigma_{t}/iB)}{Q_{N} (-\Sigma_{t}/iB)} \\ 0 & 0 & 0 \end{cases}, B_{N}$$
(II.8)

-5-

The set of Eqs. II.8 may be further simplified by use of the extended transport approximation $^{(9)}$ which assumes that anisotropic scattering takes place without a change in lethargy so that

$$\int du' \Sigma_{s}^{\ell}(u' \rightarrow u) \phi_{\ell}(u') = \Sigma_{s}^{\ell}(u) \phi_{\ell}(u) \qquad \ell = 2, ..., N \qquad (II.9)$$
$$\Sigma_{s}^{\ell}(u) \equiv \int du' \Sigma_{s}^{\ell}(u \rightarrow u').$$

where

Using Eq.II.9 in Eqs.II.8 one obtains the consistent P_1 and B_1 order N extended transport equations.

$$iB\phi_{1}(u) + \Sigma_{t}\phi_{0}(u) - \varepsilon(u) + \int du' \Sigma_{s}^{h}(u' \rightarrow u) \psi_{0}(u')$$

$$\frac{iB}{3}\phi_{0}(u) + A_{1}(B, u, N) \phi_{1}(u) = \int du' \Sigma_{s}^{1}(u' \rightarrow u) \phi_{1}(u') \quad (II.10)$$

$$\phi_{\ell}(u) = -\frac{\ell}{2\ell + 1} \quad iB \phi_{\ell-1}(u)/A_{\ell}(B, u, N) \quad \ell = 2, ..., N$$

$$A_{\ell}(B, u, N) = b_{\ell-1} + \frac{a_{\ell}}{b_{\ell}} + \frac{a_{\ell+1}}{b_{\ell+1}} - - \frac{a_{N-1}}{b_{N-1}}$$

$$a_{\ell} = \frac{\ell + 1}{2\ell + 1} \frac{\ell + 1}{2(\ell + 1) + 1} \quad B^{2} \quad 1 \leq \ell \leq N - 1 \quad (II.11)$$

$$b_{0} = \begin{cases} \Sigma_{t}(u) , & P_{1} \\ \Sigma_{t}(u) , & B_{1}, N > 1 \\ \gamma \Sigma_{t}(u) = \frac{Btan^{-1} B/\Sigma}{3(1 - \frac{\Sigma}{B} tan^{-1} \frac{B}{\Sigma})} , B_{1}, N = 1 \\ \\ S_{t}(u) - \Sigma_{s}^{\ell+1} , P_{1} \\ \\ \ell = 1, 2, ..., N - 1 \\ \\ \Sigma_{t}(u) - \Sigma_{s}^{\ell+1}(u) + \frac{N+1}{2N+1} iB \frac{Q_{N+1} (-\Sigma_{t}/iB)}{Q_{N} (-\Sigma_{t}/iB)} \delta_{\ell}^{N-1} , B_{1} \end{cases}$$

The consistent P_1 and B_1 Eqs.II.10 with the continued fraction definitions of Eqs.II.11 reduce to the standard forms programmed in the MC²⁽¹⁰⁾ and GAM⁽¹¹⁾ co if one takes the order of the extended transport approximation, N, as unity.

The inconsistent P_1 and B_1 equations are obtained by assuming that Eq.II.9 is applicable for $\ell = 1$. In this case the P_1 integral of Eqs.II.10 disappears and the coefficient b_0 of the continued fraction is modified,

$$b_{\ell} = \begin{cases} \Sigma_{t}(u) - \Sigma_{s}^{\ell+1}(u), & P_{1} \\ & \ell = 0, 1, \dots, N-1 \\ \Sigma_{t}(u) - \Sigma_{s}^{\ell+1}(u) + \frac{N+1}{2N+1} iB & \frac{Q_{N+1}(-\Sigma_{t}/iB)}{Q_{N}(-\Sigma_{t}/iB)} & \delta_{\ell}^{N-1}, B_{1} \end{cases}$$
(II.12)

The sets of Eqs.II.10, II.11 and II.12 define the four spectrum options, consistent and inconsistent P_1 and B_1 extended transport approximations, which are available in the ultra-fine-group spectrum calculations of the MC²-2 code. The following sub-sections will discuss the methods by which these equations are solved.

B. <u>Source Term</u>

The source term of Eqs.II.10, S(u), describes neutron sources due to fission and inhomogeneous sources independent of the flux. The source is assumed to be isotropic in the laboratory system. In MC²-2 scattering sources due to inelastic and (n,2n) scattering are also assumed isotropic and it is convenient to include these scattering sources as components of S(u). With this convention the equations of Section II.A above are applicable if $\Sigma_{s}^{l}(u' \rightarrow u)$ is taken to be the elastic scattering transfer cross section and S(u) is defined

$$S(u) = \frac{1}{k} S_{f}(u) + S_{ne}(u) + S_{fix}(u) \qquad (II.13)$$

$$S_{f}(u) \equiv \text{ fission source}$$

$$S_{ne}(u) \equiv \text{ non-elastic scattering source}$$

$$S_{fix}(u) \equiv \text{ inhomogeneous source}$$

The fission source has the form

$$S_{f}(u) = \sum_{i} \chi_{i}(u) N_{i} \int du' v_{i}(u') \sigma_{f_{i}}(u') \phi(u') \qquad (II.14)$$

where the sum is over all isotopes in the mixture, N_i is the atom density of isotope i, $\chi_i(u)$ is the fraction of fission neutrons emitted in the lethargy range du about u for isotope i, $v_i(u)$ is the number of neutrons emitted per fission for isotope i, $\sigma_{f_i}(u)$ is the microscopic fission cross section for isotope i. The fission spectrum distribution $\chi_i(u)$ may actually depend upon the fission lethargy, but such a possibility is not permitted by the MC²-2 code. An option is available in the code to assign library fission spectra to problem isotopes thus permitting a single fission spectrum distribution for all fissionable isotopes. The scalar k of Eq.II.13 has the value unity if $S_{fix}(u) \neq 0$. If $S_{fix}(u) = 0$, then Eqs.II.10 have a solution only if k is the eigenvalue of the system of equations.

C. Continuous Slowing Down Theory

Two sets of algorithms for the solution of Eqs.II.10 are available in the $MC^{2}-2$ code, multigroup and continuous slowing down theory. In this section the form of the continuous slowing down equations treated by $MC^{2}-2$ is derived. The algorithms are based on the work of Stacey⁽¹²⁾ and are included in this report to provide a self-contained presentation for the convenience of the reader. A general review of continuous slowing down theory has been prepared by Stacey⁽¹³⁾.

The ℓ -th order angular component of the elastic slowing down density is defined

$$\bar{q}_{\ell}(u) \equiv \sum_{i} \int_{u-\ell n^{1}/\alpha_{i}}^{u} du' \int_{u}^{u'+\ell n^{1}/\alpha_{i}} du'' \Sigma_{s_{i}}^{\ell}(u' \rightarrow u'') \phi_{\ell}(u') \quad (II.15)$$

where

 $\alpha_{i} = \left(\frac{A_{i} - 1}{A_{i} + 1}\right)^{2}$

The scattering transfer cross sections of Eqs.II.4 and II.10 are given by

$$\Sigma_{s}^{\ell}(\mathbf{u}' \rightarrow \mathbf{u}) = \sum_{i} \Sigma_{s}^{\ell}(\mathbf{u}' \rightarrow \mathbf{u})$$

as a consequence of the conventions assumed in Section II.B. It is convenient to define an elastic transfer kernel $P_i^{\ell}(u' \rightarrow u)$ and a slowing down kernel $K_i^{\ell}(u' \rightarrow u)$ in the following manner,

$$\Sigma_{s_{i}}^{\ell}(u' \rightarrow u) = \Sigma_{s_{i}}(u') P_{i}^{\ell}(u' \rightarrow u)$$
 (II.16)

$$K_{i}^{\ell}(u' \to u) = \int_{u}^{u' + \ell n^{1}/\alpha} du'' P_{i}^{\ell}(u' \to u''). \qquad (II.17)$$

Differentiation of Eq. II.15 gives

$$\frac{d\bar{q}_{\ell}(u)}{du} = \Sigma_{s}^{\ell}(u) \phi_{\ell}(u) - \sum_{i} \int_{u-\ell n^{1}/\alpha_{i}}^{u} du' \Sigma_{s}(u') P_{i}^{\ell}(u' \rightarrow u) \phi_{\ell}(u'). \quad (II.18,$$

A continuous slowing down approximation results from assuming

$$F_{i}^{\ell}(u') \equiv g_{i}(u') \phi_{\ell}(u') = F_{i}^{\ell}(u) + (u' - u) \frac{d}{du} F_{i}^{\ell}(u)$$
 (II.19)

$$u > u' > u - ln(1/\alpha)$$

where $g_i(u')$ is taken to be either the isotopic scattering cross section, $\sum_{i=1}^{\infty} (u')$, or the macroscopic total cross section $\sum_{i=1}^{\infty} (u')$. The validity of

either approximation depends upon the mixture and the lethargy range in question. Since the total collision density is likely to be a more slowly varying quantity than each of the isotopic scattering collision densities, the latter choice of $g_i(u')$ has a greater range of validity. As the scattering band for Hydrogen ($A_H \stackrel{\sim}{\sim} 1$) is infinite, the two term expansion of Eq.II.19 is not valid in this case for either choice of $g_i(u)$. Hydrogen is therefore treated separately and the slowing down density is represented as

$$q_{\rho}(u) = q_{\rho}(u) + n_{\rho}(u)$$

where $n_{\ell}(u)$ is the Hydrogen slowing down density and $q_{\ell}(u)$ is given by Eq.II.15 with the sum excluding Hydrogen. Substitution of Eq.II.19 into Eqs.II.15 and II.18 gives

$$a_{\ell}(u) = \sum_{i \neq H} [\xi_{i}^{\ell}(u) F_{i}^{\ell}(u) + a_{i}^{\ell}(u) \frac{d}{du} F_{i}^{\ell}(u)]$$
(II.20)

$$\frac{\mathrm{dq}_{\ell}(\mathbf{u})}{\mathrm{du}} = \sum_{i \neq H} \left[c_{i}^{\ell} F_{i}^{\ell}(\mathbf{u}) - e_{i}^{\ell}(\mathbf{u}) \frac{\mathrm{d}}{\mathrm{du}} F_{i}^{\ell}(\mathbf{u}) \right]$$
(II.21)

with the moderating parameters given by

$$\xi_{i}^{\ell}(u) = \int_{u-\ell n^{1}/\alpha_{i}}^{u} du' \frac{\sum_{i}^{s}(u')}{g_{i}(u')} K_{i}^{\ell}(u' \to u)$$

$$a_{i}^{\ell}(u) = \int_{u-\ell n^{1}/\alpha_{i}}^{u} du' (u' - u) \frac{\sum_{i}^{s}(u')}{g_{i}(u')} K_{i}^{\ell}(u' \to u) \quad (II.22)$$

$$c_{i}^{\ell}(u) = \frac{\sum_{i}^{\ell}(u)}{g_{i}(u)} - \int_{u-\ell n^{1}/\alpha_{i}}^{u} du' \frac{\sum_{i}^{s}(u')}{g_{i}(u')} P_{i}^{\ell}(u' \to u)$$

$$e_{i}^{\ell}(u) = \int_{u-\ell n^{1}/\alpha_{i}}^{u} du' (u' - u) \frac{\sum_{s_{i}}^{u}(u')}{g_{i}(u')} P_{i}^{\ell}(u' \rightarrow u)$$

If consideration is limited to the case, $g_i(u') = \Sigma_t(u)$, then composite moderating parameters may be defined

$$\xi_{\ell}(\mathbf{u}) = \sum_{i \neq H} \xi_{i}^{\ell}(\mathbf{u}), \quad a_{\ell}(\mathbf{u}) = \sum_{i \neq H} a_{i}^{\ell}(\mathbf{u}),$$

$$c_{\ell}(u) = \sum_{i \neq H} c_{i}^{\ell}(u), \quad e_{\ell}(u) = \sum_{i \neq H} e_{i}^{\ell}(u) \quad (II.23)$$

and Eqs.II.20 and II.21 may be combined to give

$$\frac{dq_{\ell}(u)}{du} = \varepsilon_{\ell}(u) \hat{\xi}_{\ell}(u) F_{\ell}(u) - \varepsilon_{\ell}(u) q_{\ell}(u)$$
(II.24)

where

$$\epsilon_{\ell}(u) = \frac{1}{\gamma_{\ell}(u)} = \frac{e_{\ell}(u)}{a_{\ell}(u)}$$
(II.25)
$$\hat{\epsilon}_{\ell}(u) = \epsilon_{\ell}(u) + \gamma_{\ell}(u) c_{\ell}(u)$$
(II.26)

and

$$F_{\ell}(u) = \Sigma_{t}(u) \phi_{\ell}(u).$$

If one considers the case $g_i(u') = \sum_{s} (u')$, and defines the composite moderating parameters

$$\xi_{\ell}(u) = -e_{\ell}(u) = \sum_{i \neq H}^{\Sigma} \frac{\sum_{i \neq H}^{S_{i}} \frac{\left(u\right) \xi_{i}^{L}}{\Sigma_{t}(u)}}{\sum_{i \neq H}^{\Sigma} \frac{\left(u\right) - a_{i}^{\ell}}{\sum_{t}(u)}}$$
(II.27)

 $c_{\ell}(u) = 0$

then combination of Eqs.II.20 and II.21 gives Eq.II.24 with the parameter ξ_{ρ} defined

$$\hat{\xi}_{\ell}(u) = \xi_{\ell}(u) \left[1 - \frac{d\gamma_{\ell}(u)}{du} \right] . \qquad (II.28)$$

A result formally similar to Eq.II.24 was obtained by Goertzel and Greuling⁽¹⁴⁾ They obtained their results by replacing the slowing down kernel K₁(u' \rightarrow u) by a synthetic kernel and preserving moments. The standard Goertzel-Greuling approximation corresponds to Eqs.II.24, II.25, II.27 and II.28 with $\frac{d\gamma_{\ell}}{du} = 0$.

-10-

The set of Eqs. II.23, II.24, II.25 and II.26 defines the "Improved Goertzel-Greuling Approximation" derived by Stacey. (12) The microscopic moderating parameters for both approximations are given by Eqs.II.22 with

 $g_{i}(u) = \begin{cases} \Sigma_{i}(u) & Goertzel-Greuling \\ \Sigma_{i}(u) & Improved Goertzel-Greuling \end{cases}$

In Section II.D of this report explicit definitions of the moderating parameters will be given.

The slowing down Eq.II.24 may be combined with the P_1/B_1 balance Eqs.II.10 to obtain solutions for the flux and slowing down density. For the inconsistent P_1/B_1 approximations, one sets

$$\frac{dq_1(u)}{du} = 0$$

which is equivalent to Eq.II.9 for l = 1. Equation II.10 may then be written

$$\Sigma_{ne}(u) \phi_{o}(u) = S(u) - \frac{d\bar{q}_{o}}{du}$$
 (II.29)

where the non-elastic cross section has been defined

$$\Sigma_{ne}(u) \equiv \Sigma_{t}(u) + \frac{B^{2}}{3A_{1}(B, u, N)} - \Sigma_{s}(u). \qquad (II.30)$$

Equations II.24 and II.29 are combined to obtain:

Inconsistent P_1/B_1 Approximation

$$\frac{dq_{o}(u)}{du} = \frac{\left[-(\Sigma_{ne}(u) + \Sigma_{sH}(u)) q_{o}(u) + \hat{\xi}_{o}(u) \Sigma_{t}(u) (S(u) + \eta_{o}(u))\right]}{M(u)}$$
(II.31)

$$\phi_{o}(u) = \frac{q_{o}(u) + \gamma_{o}(u) [S(u) + n_{o}(u)]}{M(u)}$$
(II.32)

$$M(u) = \hat{\xi}_{0}(u) \Sigma_{t}(u) + \gamma_{0}(u) [\Sigma_{ne}(u) + \Sigma_{sH}(u)] . \qquad (II.33)$$

Equation II.31 may be integrated directly

$$q_{o}(u) = \exp\left[-\int_{0}^{u} du' \frac{\left[\sum_{ne}(u') + \sum_{s_{H}}(u')\right]}{M(u')}\right] \cdot \left\{q_{o}(0) + (II.34)\right\}$$
$$\int_{0}^{u} du' \frac{\hat{\xi}_{o}(u') \sum_{t}(u') (S(u') + \eta_{o}(u'))}{M(u')} \exp\left[\int_{0}^{u'} du'' \frac{\left[\sum_{ne}(u'') + \sum_{s_{H}}(u'')\right]}{M(u'')}\right]\right\}$$

Equations II.31-II.34 have been written for the general situation which includes Hydrogen in the mixture. If there were no Hydrogen, the above equations are applicable with $\Sigma_{H}(u) = 0$, $\eta_{O}(u) = 0$.

If one retains both the $\ell = 0$, 1 terms of Eqs.II.10 and II.24, the consistent P_1/B_1 coupled equations are obtained,

$$\frac{\text{Consistent P}_{1}/B_{1} \text{ Approximation}}{\frac{1}{\Delta(u)}} \left\{ \begin{bmatrix} \Sigma_{t} - \Sigma_{s} + \Sigma_{s_{H}} \end{bmatrix} (A_{1} - \Sigma_{s}^{1} + \frac{2}{3} \Sigma_{s_{H}} + \varepsilon_{1} \hat{\varepsilon}_{1} \Sigma_{t}) + \frac{1}{3} B^{2} \end{bmatrix}, \\ \begin{bmatrix} \varepsilon_{0}q_{0} + S + n_{0} \end{bmatrix} + iB\varepsilon_{0}\hat{\varepsilon}_{0}\Sigma_{t} [\varepsilon_{1}q_{1} + \frac{3}{2} n_{1}] \right\} \\ + \frac{dq_{0}}{du} = S(u) + n_{0}(u) \\ \frac{1}{\Delta(u)} \left\{ \begin{bmatrix} [(A_{1} - \Sigma_{s}^{1} + \frac{2}{3} \Sigma_{s_{H}})(\Sigma_{t} - \Sigma_{s} + \Sigma_{s_{H}} + \varepsilon_{0}\hat{\varepsilon}_{0}\Sigma_{t}) + \frac{B^{2}}{3}], \\ [\varepsilon_{1}q_{1} + \frac{3}{2} n_{1}] + \frac{iB}{3} \varepsilon_{1}\hat{\varepsilon}_{1}\Sigma_{t} [\varepsilon_{0}q_{0} + S + n_{0}] \right\} \\ + \frac{dq_{1}}{du} = \frac{3}{2} n_{1}(u) \\ \Delta(u) = (A_{1} - \Sigma_{s}^{1} + \frac{2}{3} \Sigma_{s_{H}} + \varepsilon_{1}\hat{\varepsilon}_{1}\Sigma_{t})(\Sigma_{t} - \Sigma_{s} + \Sigma_{s_{H}} + \varepsilon_{0}\hat{\varepsilon}_{0}\Sigma_{t}) + \frac{1}{3} B^{2} \end{bmatrix}$$

and

$$\begin{split} \phi_{o}(u) &= \frac{1}{\Delta(u)} \left[(A_{1} - \Sigma_{s}^{1} + \frac{2}{3} \Sigma_{s_{H}} + \varepsilon_{1} \hat{\xi}_{1} \Sigma_{t}) (\varepsilon_{o} q_{o} + S_{o} + \eta_{o}) \right. \\ &- iB \left(\varepsilon_{1} q_{1} + \frac{3}{2} \eta_{1} \right) \right] \end{split} \tag{II.36} \\ \phi_{1}(u) &= \frac{1}{\Delta(u)} \left[(\Sigma - \Sigma_{s} + \Sigma_{s_{H}} + \varepsilon_{o} \hat{\xi}_{o} \Sigma_{t}) (\varepsilon_{1} q_{1} + \frac{3}{2} \eta_{1}) \right. \\ &- \frac{iB}{3} \left(\varepsilon_{o} q_{o} + S + \eta_{o} \right) \right] . \end{split}$$

In writing Eqs.II.35 and II.35 Hydrogen elastic scattering has been assumed isotropic in the center of mass system so that

$$\Sigma_{S_{H}}^{1}(u) = \frac{2}{3} \Sigma_{S_{H}}^{u}(u)$$

$$\eta_{o}(u) = \int_{o}^{u} du' \Sigma_{S_{H}}^{u}(u') \phi_{o}(u') e^{-(u-u')}$$
(II.37)
$$\eta_{1}(u) = \frac{2}{3} \int_{o}^{u} du' \Sigma_{S_{H}}^{u}(u') \phi_{1}(u') e^{-\frac{3}{2}(u-u')}.$$

The difference equations used by MC^2-2 in the solution of Eqs.II.32-II.36 will be discussed in Section F.

D. Continuous Slowing Down Moderating Parameters

It is well known that the microscopic Legendre moments of the elastic scattering transfer cross section defined in Eq.II.14 may be represented in (15) the form,

$$\sigma_{s}^{\ell}(u' \rightarrow u) = \frac{\sigma_{s}(u') P_{\ell}[\mu_{o}(u' \rightarrow u)] e^{-(u-u')}}{1-\alpha} \sum_{n=0}^{N} (2n+1) f_{n}(u') \cdot P_{n}[\mu_{c}(u' \rightarrow u)] \qquad (II.38)$$

where

μ_c

ц

μ

 $f_n(u')$ = nth Legendre expansion coefficient at lethargy u' in the center of mass system. $f_n(u')$ = 1

The MC²-2 code permits all N \leq 20.

= cosine of scattering angle in the center of of mass system.

$$\mu_{c} = \frac{(A+1)^{2} e^{-(u-u')} - (A^{2}+1)}{2A} = \frac{1}{1-\alpha} [2e^{-(u-u')} - (1+\alpha)]$$
(II.39)

• = cosine of scattering angle in the laboratory system
$$(\underline{\Omega} \cdot \underline{\Omega}')$$

$$= \frac{(A + 1) e^{-\frac{(u-u')}{2} - (A - 1) e^{\frac{(u-u')}{2}}}}{2}$$
(II.40)

and all other terms have been defined previously. To simplify the notation, the isotope index i has been dropped. Equation II.38 along with Eqs.II.16, II.17 may be used to obtain explicit forms for the isotopic moderating parameters of Equations II.22.

1. Goertzel-Greuling Moderating Parameters

Setting $g_i(u') = \sum_{i=1}^{\infty} (u')$ and dropping the isotope index i, Eqs.II.22 may be written,

$$\xi^{\ell} = -e^{\ell} = \frac{1}{1-\alpha} \sum_{n=0}^{N} (2n+1) \int_{u-\ell n^{1}/\alpha}^{u} du' \int_{u}^{u'+\ell n^{1}/\alpha} du'' f_{n}(u') e^{-(u''-u')}.$$

 $P_{\ell} \left[\mu_{o}(u' \twoheadrightarrow u'') \right] P_{n} \left[\mu_{c}(u' \rightarrow u'') \right]$

$$a^{\ell} = \frac{1}{1-\alpha} \sum_{n=0}^{N} (2n+1) \int_{u-\ell n^{1}/\alpha}^{u} du' (u'-u) \int_{u}^{u'+\ell n^{1}/\alpha} du'' f_{n}(u') e^{-(u''-u')}$$

$$P_{\ell} (\mu_{o}) P_{n} (\mu_{c})$$

$$c^{\ell} = 0.$$

Transforming variables to U = u'' - u' and switching orders of integration gives

$$\xi^{\ell} = \sum_{n=0}^{N} \frac{(2n+1)}{2} \int_{0}^{\ell n^{1}/\alpha} dU P_{n}[\mu_{c}(U)] P_{\ell}[\mu_{0}(U)] \left(-\frac{d\mu_{c}}{dU}\right) \int_{u-U}^{u} du' f_{n}(u')$$

$$a^{\ell} = \sum_{n=0}^{N} \frac{(2n+1)}{2} \int_{0}^{\ell n^{1}/\alpha} dU P_{n}(\mu_{c}) P_{\ell}(\mu_{0}) \left(-\frac{d\mu_{c}}{dU}\right) \int_{u-U}^{u} du' (u'-u) f_{n}(u')$$

If the scattering coefficients $f_n(u)$ are assumed constant in the scattering band $[u - ln(1/\alpha), u]$, the Goertzel-Greuling moderating parameters may be written

$$\xi_{i}^{\ell}(u) = -e_{i}^{\ell}(u) = -\sum_{n=0}^{N} f_{n}(u) T_{\ell n}^{1}(\alpha_{i})$$

$$a_{i}^{\ell}(u) = -\sum_{n=0}^{N} f_{n}(u) T_{\ell n}^{2}(\alpha_{i})$$

$$c_{i}^{\ell}(u) = 0$$
(II.41)

where

$$T_{\ell n}^{m}(\alpha_{i}) \equiv \frac{(-)^{m}}{m!} \frac{(2n+1)}{2} \int_{0}^{\ell n^{1}/\alpha_{i}} dU U^{m} P_{\ell}(\mu_{0}(U)) P_{n}(\mu_{c}(U)).$$

$$\left(-\frac{d\mu_{c}(U)}{dU}\right) \cdot \qquad (II.4)$$

The quantities $T_{\ell n}^{m}$ relate Legendre moments of the scattering cross sections in the center of mass and laboratory systems and have been studied in detail.⁽¹⁶⁻¹⁸⁾ Precalculated values of $T_{\ell n}^{m}$ are provided in the MC²-2 library (MCC2F8 of Appendix C) for each isotope. Numerical experience has shown that six terms in the sums of Eqs.II.41 suffice to calculate the parameters accurately. This is a consequence of the fact that $T_{\ell n}^{m}(\alpha)$ approaches zero at least as fast as the faster of A^{-m} and $A^{-|\ell-n|}$.

2. Improved Goertzel-Greuling Moderating Parameters

Setting $g_{i}(u') = \Sigma_{t}(u')$ Eqs. II.22 may be written $\xi_{i}^{\ell}(u) = \frac{1}{1-\alpha_{i}} \sum_{n=0}^{N} (2n+1) \int_{u-\ell n^{1}/\alpha_{i}}^{u} du' \frac{\Sigma_{s_{i}}(u')}{\Sigma_{t}(u')} f_{n_{i}}(u').$ $\int_{u}^{u'+\ell n^{1}/\alpha_{i}} du'' e^{-(u''-u')} P_{\ell}(\mu_{o}) P_{n}(\mu_{c})$ $a_{i}^{\ell}(u) = \frac{1}{1-\alpha_{i}} \sum_{n=0}^{N} (2n+1) \int_{u-\ell n^{1}/\alpha_{i}}^{u} du' \frac{\Sigma_{s_{i}}(u')}{\Sigma_{t}(u')} (u'-u) f_{n_{i}}(u').$ $\int_{u}^{u'+\ell n^{1}/\alpha_{i}} du'' e^{-(u''-u')} P_{\ell}(\mu_{o}) P_{n}(\mu_{c})$ (II.43)

$$e_{i}^{\ell}(u) = \frac{1}{1-\alpha_{i}} \sum_{n=0}^{N} (2n+1) \int_{u-\ell n^{1}/\alpha_{i}}^{u} du' \frac{\sum_{i}^{\Sigma}(u')}{\sum_{i}(u')} (u'-u) f_{n_{i}}(u').$$

$$e^{-(u-u')} P_{\ell}(\mu_{o}) P_{n}(\mu_{c})$$

$$= \frac{\Sigma_{s_{i}}^{\ell}(u)}{\Sigma_{t}(u)} - \frac{1}{1-\alpha_{i}} \sum_{n=0}^{N} (2n+1) \int_{u-\ell n^{1}/\alpha_{i}}^{u} du' \frac{\Sigma_{s_{i}}(u')}{\Sigma_{t}(u')} f_{n_{i}}(u').$$

$$= e^{-(u-u')} P_{\ell}(\mu_{o}) P_{n}(\mu_{c}).$$

For heavy isotopes which have a small scattering band width, it is reasonable to assume

$$\frac{\sum_{i} (u')}{\sum_{t} (u')} f_{n_{i}}(u') \sqrt[n]{} \left\langle \frac{\sum_{i} f_{n_{i}}}{\sum_{t} u} \right\rangle_{u} \qquad u - \ln^{1}/\alpha_{i} \leq u' \leq u$$

where $\langle \rangle_u$ denotes an average over the lethargy band. With this approximation the same operations which led to Eqs.II.41 give

$$\xi_{i}^{\ell}(u) = -e_{i}^{\ell}(u) = -\sum_{n=0}^{N} \left\langle \frac{\Sigma_{s_{i}}}{\Sigma_{t}} f_{n_{i}} \right\rangle_{u} T_{\ell n}^{1}(\alpha_{i})$$

$$a_{i}^{\ell}(u) = -\sum_{n=0}^{N} \left\langle \frac{z_{i}}{\Sigma_{t}} f_{n} \right\rangle_{u} T_{\ell n}^{2} (\alpha_{i})$$

$$c_{i}^{\ell}(u) = \sum_{n=0}^{N} \left[\frac{\Sigma_{s}(u)}{\Sigma_{t}(u)} f_{n}(u) - \left\langle \frac{\Sigma_{s}}{\Sigma_{t}} f_{n} \right\rangle_{u} \right] T_{\ell n}^{0} (\alpha_{i}) \sim 0$$
(II.44)
$$T_{\ell n}^{0} (\alpha_{i}) \sim 0$$

where the definition

$$\Sigma_{s_{i}}^{\ell}(u) = \sum_{n=0}^{N} \Sigma_{s_{i}}(u) f_{n_{i}}(u) T_{\ell n}^{0}(\alpha_{i})$$
(11.45)

has been used. The setting of $c_{i}^{\ell}(u)$ to zero is an approximation required in $MC^{2}-2$ to avoid numerical difficulties. In the $MC^{2}-2$ code Eqs. II.44 are used for isotopes of mass $A_{i} \geq 200$ ($\ell n^{1}/\alpha_{i} < .02$) so that the averaging assumption is expected to be valid. Numerical experience has shown that four terms in the sums of Eqs. II.44, N = 3, suffice to calculate the parameters accurately for the heavy isotopes.

For lighter materials ($A_i < 200$) the Improved Goertzel-Greuling moderating parameters are derived in the following manner. Define the quantity

$$\mathcal{D}_{n}^{i}, (u) \equiv \sum_{n=n'-1}^{N} \frac{2n+1}{2} f_{n_{i}}(u) K_{nn'}^{i}, (\alpha_{i})$$
 (II.46)

17

where

$$K_{nn}^{i}(\alpha_{i}) \equiv \frac{1}{2^{n}(n'-1)!} \left(\frac{2}{1+\alpha_{i}}\right)^{n'} \sum_{k=0}^{k} \frac{(-)^{n+1-k-n'}(2n-2k)!}{k!(n-k)!(n+1-2k-n')!} \cdot \left(\frac{1+\alpha_{i}}{1-\alpha_{i}}\right)^{n+1-2k} K_{max} \equiv \min\left\{ [n/2], \left[\frac{n+1-n'}{2}\right] \right\} \cdot (11.47)$$

Noting that the Legendre Polynomials $P_n(\mu)$ may be expanded

$$P_{n}(\mu) = \sum_{k=0}^{\lfloor n/2 \rfloor} \frac{(-) (2n - 2k)!}{2^{n} k! (n-k)! (n-2k)!} \mu^{n-2k}$$

and using Eqs.II.39 and II.40 it can be shown that the l = 0, 1 Eqs. II.43 may be written in the form

$$\begin{split} \xi_{\underline{1}}^{0}(\mathbf{u}) &= \prod_{n=1}^{N+1} \frac{1}{n!} \left\{ \int_{u-\ell n^{1}/\alpha_{\underline{1}}}^{u} d\mathbf{u}' \frac{\Sigma_{\underline{3}}(\mathbf{u}')}{\Sigma_{\underline{1}}(\mathbf{u}')} p_{\underline{1}}^{1}(\mathbf{u}') e^{-n'(\mathbf{u}-\mathbf{u}')} \\ &= \alpha_{\underline{1}}^{n'} \int_{u-\ell n^{1}/\alpha_{\underline{1}}}^{u} d\mathbf{u}' \frac{\Sigma_{\underline{3}}(\mathbf{u}')}{\Sigma_{\underline{1}}(\mathbf{u}')} p_{\underline{1}}^{1}(\mathbf{u}') \right\} \\ s_{\underline{1}}^{0}(\mathbf{u}) &= -\prod_{n=1}^{N+1} \frac{1}{n!} \left\{ \int_{u-\ell n^{1}/\alpha_{\underline{1}}}^{u} d\mathbf{u}' \frac{\Sigma_{\underline{3}}(\mathbf{u}')}{\Sigma_{\underline{1}}(\mathbf{u}')} p_{\underline{1}}^{1}(\mathbf{u}') (\mathbf{u} - \mathbf{u}') e^{-n'(\mathbf{u}-\mathbf{u}')} \right. \\ &= \alpha_{\underline{1}}^{n'} \int_{u-\ell n^{1}/\alpha_{\underline{1}}}^{u} d\mathbf{u}' \frac{\Sigma_{\underline{3}}(\mathbf{u}')}{\Sigma_{\underline{1}}(\mathbf{u}')} (\mathbf{u} - \mathbf{u}') p_{\underline{1}}^{1}(\mathbf{u}') (\mathbf{u} - \mathbf{u}') e^{-n'(\mathbf{u}-\mathbf{u}')} \\ &= \alpha_{\underline{1}}^{n'} \int_{u-\ell n^{1}/\alpha_{\underline{1}}}^{u} d\mathbf{u}' \frac{\Sigma_{\underline{3}}(\mathbf{u}')}{\Sigma_{\underline{1}}(\mathbf{u}')} (\mathbf{u} - \mathbf{u}') p_{\underline{1}}^{1}(\mathbf{u}') \right\} \\ c_{\underline{0}}^{0}(\mathbf{u}) &= \prod_{n=1}^{N+1} \left\{ \frac{\Sigma_{\underline{3}}(\mathbf{u})}{\Sigma_{\underline{1}}(\mathbf{u})} - \frac{(1 - \alpha_{\underline{1}}^{n'})}{n'} p_{\underline{1}}^{1}(\mathbf{u}) - \int_{u-\ell n^{1}/\alpha_{\underline{1}}}^{u} d\mathbf{u}' \frac{\Sigma_{\underline{3}}(\mathbf{u}')}{\Sigma_{\underline{1}}(\mathbf{u}')} p_{\underline{1}}^{1}(\mathbf{u}') e^{-n'(\mathbf{u}-\mathbf{u}')} \right\} \\ s_{\underline{1}}^{0}(\mathbf{u}) &= \prod_{n=1}^{N+1} \left\{ \frac{\Sigma_{\underline{3}}(\mathbf{u})}{2\underline{1}(\mathbf{u})} \frac{1 - \alpha_{\underline{1}}^{n'}}{2\underline{1}(\mathbf{u}')} p_{\underline{1}}^{1}(\mathbf{u}') (\mathbf{u} - \mathbf{u}') e^{-n'(\mathbf{u}-\mathbf{u}')} (\mathbf{II} \cdot \mathbf{48}) \\ s_{\underline{1}}^{1}(\mathbf{u}) &= \prod_{n=1}^{N+1} \left\{ \frac{(A_{\underline{1}} + 1)}{2n' - 1} \left[\int_{u-\ell n^{1}/\alpha_{\underline{1}}}^{u} d\mathbf{u}' \frac{\Sigma_{\underline{3}}(\mathbf{u}')}{2\underline{1}(\mathbf{u}')} p_{\underline{1}}^{1}(\mathbf{u}') e^{-(n'+1/2)(\mathbf{u}-\mathbf{u}')} \right. \\ &= \alpha_{\underline{1}}^{n'+1/2} \int_{u-\ell n^{1}/\alpha_{\underline{1}}}^{u} d\mathbf{u}' \frac{\Sigma_{\underline{3}}(\mathbf{u}')}{2\underline{1}(\mathbf{u}')} p_{\underline{1}}^{1}(\mathbf{u}') e^{-(n'+1/2)(\mathbf{u}-\mathbf{u}')} \\ &= \alpha_{\underline{1}}^{n'+1/2} \int_{u-\ell n^{1}/\alpha_{\underline{1}}}^{u} d\mathbf{u}' \frac{\Sigma_{\underline{3}}(\mathbf{u}')}{2\underline{1}(\mathbf{u}')} p_{\underline{1}}^{1}(\mathbf{u}') e^{-(n'-1/2)(\mathbf{u}-\mathbf{u}')} \\ &= \alpha_{\underline{1}}^{n'-1/2} \int_{u-\ell n^{1}/\alpha_{\underline{1}}}^{u} d\mathbf{u}' \frac{\Sigma_{\underline{3}}(\mathbf{u}')}{2\underline{1}(\mathbf{u}')} p_{\underline{1}}^{1}(\mathbf{u}') e^{-(n'-1/2)(\mathbf{u}-\mathbf{u}')} \\ &= \alpha_{\underline{1}}^{n'-1/2} \int_{u-\ell n^{1}/\alpha_{\underline{1}}}^{u} d\mathbf{u}' \frac{\Sigma_{\underline{3}}(\mathbf{u}')}{2\underline{1}(\mathbf{u}')} p_{\underline{1}}^{1}(\mathbf{u}') e^{-(n'-1/2)(\mathbf{u}-\mathbf{u}')} \\ &= \alpha_{\underline{1}}^{n'-1/2} \int_{u-\ell n^{1}/\alpha_{\underline{1}}}^{u} d\mathbf{u}' \frac{\Sigma_{\underline{3}}(\mathbf{u}')}{2\underline{1}(\mathbf{u}')} p_{\underline{1}}^{1}(\mathbf{u}') e^{-(n'-1/2)(\mathbf{u}-\mathbf{u}')} \\ &= \alpha_{\underline{1}}^{n'-1/2} \int$$

-17-

 $- (A_{i} - 1) \int_{u-ln^{1}/\alpha_{i}}^{u} du' \frac{\sum_{i}^{u}(u')}{\sum_{i}^{u}(u')} D_{n}^{i}(u') e^{-(n'-1/2)} (u-u') \right\}$
$$= \frac{\sum_{i=1}^{n} \sum_{i=1}^{n} \sum_{i=1}^{n}$$

Numerical experience has shown that only four terms need be retained in the sums of Eqs. II.48, N = 3. The mass dependent coefficients K_{IIII} (α) defined by Eq. II.47 are given in Table I. These coefficients are provided in the MC²-2 library (MCC2F8 of Appendix C).

TABLE I. Coefficients K_{nn} , (a)

5 3 2 n/n' 1 $\frac{2}{1-\alpha}$ 0 $\frac{-2}{(1-\alpha)}\left(\frac{1+\alpha}{1-\alpha}\right)$ 1 $(1-\alpha)^2$ $\frac{1}{1-\alpha} \left[3 \left(\frac{1+\alpha}{1-\alpha} \right)^2 - 1 \right]$ 2 $-12 \frac{(1+\alpha)}{(1-\alpha)^3}$ $\frac{12}{(1-\alpha)^3}$ -20- $\frac{(1 \ \alpha)}{(1-\alpha)^2} \left[-5 \left(\frac{1+\alpha}{1-\alpha} \right)^2 + 3 \right]$ 3 $\frac{6}{(1-\alpha)^2} \left[5 \left(\frac{1+\alpha}{1-\alpha} \right)^2 - 1 \right]$ $-60 \frac{(1+\alpha)}{(1-\alpha)^4}$ $\frac{40}{(1-\alpha)^{4}}$ $\frac{1}{4(1-\alpha)} \left[35\left(\frac{1+\alpha}{1-\alpha}\right)^4 - 30\left(\frac{1+\alpha}{1-\alpha}\right)^2 + 3 \right] \quad 10 \quad \frac{(1+\alpha)}{(1-\alpha)^3} \left[-7\left(\frac{1+\alpha}{1-\alpha}\right)^2 + 3 \right] \quad \frac{30}{(1-\alpha)^3} \left[7\left(\frac{1-\alpha}{1-\alpha}\right)^2 - 1 \right] \quad -280 \quad \frac{(1+\alpha)}{(1-\alpha)^5} \quad \frac{140}{(1-\alpha)^5} = \frac{140}{(1-\alpha)^$ 4

E. Narrow Resonance Attenuation

The formalism developed in the previous sections is directly applicable to narrow resonances provided that effective resonance cross sections could be calculated. Such a calculation is quite time consuming as evidenced by the fact that the resonance calculation of the code MC^{2} is about one-half of the total computing time for typical problem executions. It is, therefore, advantageous to ignore these resonances in defining the moderating parameters of Section D and then superimpose their effect by attenuating the slowing down density. This is the strategy employed by the code MC^2-2 for the treatment of narrow resolved resonances. In particular, the wide resonances of light materials are treated by the code which prepares the library for MC^2-2 from basic nuclear data, e.g. $ETOE-2^{(4)}$. MC^2-2 treats these resonance cross sections as effective smooth cross sections and includes their contribution in the calculation of moderating parameters. Similarly, effective smooth cross sections for heavy isotopes are calculated by MC²-2 from unresolved resonance parameters according to the algorithms specified in Chapter IV and included in the calculation of the moderating parameters. An asymptotic slowing down density, $q_{\ell}^{asy}(u)$, is calculated using these parameters in Eqs. II.34 or II.35 and the effects of the narrow resolved resonances are superimposed on $q_{\rho}^{asy}(u)$ by use of attenuation factors. The form of the narrow resonance attenuation factors is derived below.

For the sake of simplicity consider the P_1/B_1 equations for a single material and assume zero buckling,

$$F(u) = \int_{0}^{u} du' P^{0} (u - u') h(u') F(u') + S(u)$$
 (II.49)

where

$$F(u) = \Sigma_{t}(u) \phi_{0}(u)$$
$$h(u) = \Sigma_{t}(u) / \Sigma_{t}(u)$$

and all other terms have been defined previously. If one neglects the narrow resolved resonance cross sections, Eq. II.49 has the form

$$F_{as}(u) = \int_{0}^{u} du' P^{0}(u - u') h_{as}(u') F_{as}(u') + S(u)$$
(II.50)

where the asymptotic cross sections are defined

$$\Sigma_{t}^{as}(u) = \Sigma_{t}(u) - \Sigma_{t}^{r}(u), \text{ etc.}$$

and r represents the narrow resolved resonance cross sections. Subtracting Eq. II.50 from Eq.II.49,

$$F(u) = F_{as}(u) + \int_{0}^{u} du' P^{0}(u - u') [h(u') F(u') - h_{as}(u') F_{as}(u')].$$

Following the methods used by Corngold, (19) define the Laplace transform as

$$\hat{f}(s) \equiv L[f(u)] \equiv \int_{0}^{\infty} du e^{-su} f(u)$$

so that

$$\tilde{F}(s) = \tilde{F}_{as}(s) + \tilde{P}^{o}(s) [L(hf) - L(h_{as}F_{as})].$$

Define

$$h(u) = 1 - g(u)$$

 $h_{as}(u) = 1 - g_{as}(u)$

where

$$g(u) = \frac{\sum_{n=1}^{as} + \sum_{n=1}^{r}}{\sum_{t=1}^{as} + \sum_{t=1}^{r}}$$

$$g_{as}(u) = \frac{\sum_{ne}^{as}}{\sum_{t}^{as}}$$

and

$$\tilde{F}(s) = \tilde{F}_{as}(s) - \frac{\tilde{P}^{o}(s) [L(gF) - L(g_{as}F_{as})]}{1 - \tilde{P}^{o}(s)}$$
 (II.51)

Defining

$$\hat{\psi}(s) = \frac{\hat{P}^{o}(s)}{1 - \hat{P}^{o}(s)}$$

Equation II.51 may be inverted to obtain

$$F(u) = F_{as}(u) - \int_{0}^{u} du' \psi(u - u') [g(u') F(u') - g_{as}(u') F_{as}(u')]$$
(II.52)

where the kernel ψ satisfies

$$\psi(u) = \int_{0}^{u} du' P^{0}(u - u') \psi(u') + P^{0}(u) . \qquad (II.53)$$

It is obvious from Eq. II.53 that $\psi(u)$ is the solution of Eq.II.49, for a delta function source in the absence of absorption. It is well known that asymptotically

$$\psi(u) \rightarrow \frac{1}{\xi} \simeq \frac{\sum_{n=0}^{as} \phi^{as}}{q^{as}}$$
(II.54)

Using Eq.II.54 in Eq.II.52 an expression for the attenuation in the asymptotic (non-resonance) collision density is obtained,

$$F(u) \approx F_{as}(u) \prod_{r} (1 - p_{r})$$
(II.55)
$$u_{r} < u$$

where the product is taken over all resonances in the interval $0 \le u_{1} \le u$ and

$$\mathbf{p}_{\mathbf{r}} \equiv \left\{ \frac{\Gamma_{\mathbf{a}}^{\mathbf{r}}}{E_{\mathbf{r}}} \quad \mathbf{J}_{\mathbf{a}}^{\star} - \frac{\Sigma_{\mathbf{ne}}^{\mathbf{as}}}{\Sigma_{\mathbf{t}}^{\mathbf{as}}} \quad \frac{\Gamma_{\mathbf{t}}^{\mathbf{r}}}{E_{\mathbf{r}}} \quad \mathbf{J}_{\mathbf{t}}^{\star} \\ \mathbf{J}_{\mathbf{t}}^{\star} \quad \mathbf{J}_{\mathbf{t}}^{\star} \quad \mathbf{J}_{\mathbf{q}}^{\star} \right\} \quad \frac{\Sigma_{\mathbf{ne}}^{\mathbf{as}} \quad \phi^{\mathbf{as}}}{q^{\mathbf{as}}}$$
(II.56)

$$J_{\mathbf{x}}^{*} = \frac{\mathbf{F}}{\Gamma_{\mathbf{x}}^{\mathbf{r}}} \int_{-\infty}^{\infty} \frac{\Sigma_{\mathbf{x}}^{\mathbf{1}}}{\Sigma_{\mathbf{t}}^{\mathbf{as}} + \Sigma_{\mathbf{t}}^{\mathbf{r}}} d\mathbf{u} \quad .$$
(11.57)

The method for calculating the J* function will be discussed in Chapter III. $\binom{(12)}{}$ derived an attenuation factor formalism by considering the resonance as a point absorber. This leads to Eq.II.56 without the second term in the brackets, thus neglecting the competition between scattering and absorption of the resonance.

F. Ultra-Fine-Group Equations

The lethargy (energy) domain of interest is assumed to be partitioned into equal lethargy intervals of width Δu . The corresponding energy widths are denoted ΔE . By convention, increasing g represents increasing lethargy (decreasing energy),

$$u_{g+1} = u_g + \Delta u$$
$$E_{g+1} = E_g e^{-\Delta u}.$$

The lethargy group width is assumed to be small and, following the MC²⁽¹⁰⁾ conventions, calculations performed in this structure are called ultra-finegroup calculations. The group structure is fixed by the MC²-2 library (Appendix C). It is usually set at $\Delta u \approx 1/120$, corresponding to the MC² structure, giving 2040 groups from 10 MeV to 0.414 eV.

Figure ² gives a schematic representation of the $MC^{2}-2$ group structure. From the maximum problem energy, E_{max} , to a user specified energy, E_{CSD} , the code uses a multigroup form of Eqs.II.10-II.12 in the spectrum calculation. From E_{CSD} to E_{min} the equations of Section II.C, Continuous Slowing Down Theory, are applicable. The user specified energy E_{CSD} must satisfy the relationship





-24-

$$E_{R} < E_{CSD} \leq E_{max}$$

where

E_R = energy of highest energy resolved resonance in the problem mixture

This restriction is a consequence of the use of the attenuation factor treatment of resolved resonances (c.f. Section II.E) in the continuous slowing down energy domain.

The remainder of this section presents the ultra-fine-group equations for the multigroup and continuous slowing down formulations.

1. Multigroup Equations

Integrating Eqs.II.10 and II.11 over a group one obtains

$$iB\phi_{1}^{g} + \Sigma_{t}^{g}\phi_{0}^{g} = S^{g} + \sum_{g' \leq g} \Sigma_{s}^{o} (g' \neq g) \phi_{0}^{g'}$$

$$i \frac{B}{3} \phi_{0}^{g} + A_{1}^{g}\phi_{1}^{g} = \sum_{g' \leq g} \Sigma_{s}^{1}(g' \neq g) \phi_{1}^{g'}$$
(II.59)

$$\phi_{\ell}^{g} = -\frac{\ell}{2\ell+1} iB\phi_{\ell-1}^{g}/A_{\ell}^{g} \qquad \ell = 2, \ldots, N$$

where

 $\phi_{\ell}^{g} = \int_{u_{g-1}}^{u_{g}} \phi_{\ell}(u) \, du \qquad (II.60)$

and

 $\Sigma_{t}^{g} = \frac{1}{\phi_{o}^{g}} \int_{u_{g-1}}^{u_{g}} \Sigma_{t}(u) \phi_{o}(u) du$

$$\Sigma_{g}^{\ell}(g' \rightarrow g) = \frac{1}{\phi_{\ell}^{g'}} \int_{u_{g-1}}^{u_{g}} du \int_{u_{g'-1}}^{u_{g'}} du' \Sigma_{g}^{\ell}(u' \rightarrow u) \phi_{\ell}(u') \quad (II.61)$$
$$A_{\ell}^{g} = \int_{u_{g-1}}^{u_{g}} A_{\ell}(B, u, N) du .$$

Since the flux moments $\phi_{\ell}(u)$ are unknown a spectrum must be assumed in order to derive the ultra-fine-group cross sections defined by Eqs.II.61. This averaging is performed by the processing code which prepares the library files for MC²-2. These ultra-fine-group data are the basic input to MC²-2 and later sections of this report will describe how these data are processed to permit a solution of Eqs. II.59.

As noted earlier in this report, an option of the code permits group dependent buckling. In this case one is solving Eqs.II.59 with B replaced by B_g . To

simplify notation, only the group independent buckling equations will be developed.

In writing Eqs.II.59 it has been assumed that a neutron cannot gain energy as a result of scattering thus precluding the use of MC^2-2 in the thermal energy domain.

Equations II.59 may be written:

$$\frac{\text{Consistent P}_{1}/B_{1} \text{ Approximation}}{\sum_{g' < g} [\Sigma_{g}^{\circ}(g' \rightarrow g) \phi^{g'} - \frac{B}{\Sigma_{r_{1}}^{g}} \Sigma_{g}^{1}(g' \rightarrow g) J^{g'}]}$$

$$\phi^{g} = \frac{\sum_{g' < g} F_{g' < g}^{g} + \frac{B^{2}}{3\Sigma_{r_{1}}^{g}}}{\sum_{r_{1}}^{g} F_{g' < g} \Sigma_{r_{1}}^{1}(g' \rightarrow g) J^{g'}}$$

$$J^{g} = \frac{\frac{B}{3} \phi^{g} + \sum_{g' < g} \Sigma_{g}^{1}(g' \rightarrow g) J^{g'}}{\Sigma_{r_{1}}^{g}}$$

$$\phi^{g}_{\ell} = -\frac{\ell}{2\ell + 1} \text{ iB } \phi^{g}_{\ell - 1}/A^{g}_{\ell}$$

$$\phi^{g} = \phi^{g}_{g}$$

$$J^{g} = i\phi^{g}_{1}$$

$$\Sigma_{r_{2}}^{g} = \Sigma_{t}^{g} - \Sigma_{s}^{\circ}(g \rightarrow g) - \Sigma_{ine1}(g \rightarrow g) - 2\Sigma_{n,2n}(g \rightarrow g)$$

$$(II.62)$$

$$\Sigma_{r_{1}}^{g} \equiv A_{1}^{g} - \Sigma_{s}^{1}(g \neq g)$$

$$S^{g} \equiv \frac{1}{k} S_{f}^{g} + S_{fix}^{g} + \sum_{\sigma' \leq \sigma} \left[\Sigma_{inel}(g' \neq g) + 2\Sigma_{n,2n}(g' \neq g) \right] \phi^{g'}$$
(II.63)

-26-

and

In Eqs. II.63 explicit account has been taken of the elastic (Σ_s^{ℓ}) , inelastic (Σ_{inel}) , and (n,2n), $(\Sigma_{n,2n})$ scattering sources and the source term S^g has been redefined to exclude non-elastic in-group scattering which has been incorporated in the removal cross section Σ_r^g . Equations II.62 and II.63 define the multigroup consistent P_1/B_1 approximation. The inconsistent P_1/B_1

Inconsistent
$$P_1/B_1$$
 Approximation

$$\phi^{g} = \frac{S^{g} + \sum_{\substack{g' < g}} \Sigma_{s}^{o}(g' \rightarrow g) \phi^{g}}{\Sigma_{r_{o}}^{g} + \frac{B^{2}}{3A_{1}^{g}}}$$
$$J^{g} = \frac{B}{3A^{g}} \phi^{g}$$

If one assumes

$$\int_{u_{g-1}}^{u_{g}} f(u) \, du \simeq \left[\theta f_{+} + (1 - \theta) f_{-}\right] \Delta u \qquad (II.65)$$

(II.64)

where

 $f_{+} \equiv f(u_{g}) = f(u) \qquad u_{g} > u > u_{g} - \theta \Delta u$ $f_{-} \equiv f(u_{g-1}) = f(u) \qquad u_{g-1} < u < u_{g} - \theta \Delta u$ $\theta \equiv \text{integration factor} \qquad 0 \le \theta \le 1$ $\Delta u \equiv u_{g} - u_{g-1}$

then the inconsistent P_1/B_1 slowing down Eq.II.34 may be written

 $\frac{\text{Inconsistent P}_{1}/B_{1} \text{ Approximation}}{q_{o}(u_{g}) = q_{o}(u_{g-1}) \exp \left\{ - (\Sigma_{ne}^{g} + \Sigma_{s_{H}}^{g}) \Delta u \left[\frac{\theta}{M_{+}^{g}} + \frac{(1 - \theta)}{M_{-}^{g}} \right] \right\} + \frac{(S^{g} + n_{o}^{g})}{\Delta u} \frac{\Sigma_{t}^{g}}{\Sigma_{ne}^{g} + \Sigma_{s_{H}}^{g}} \left\{ \hat{\xi}_{o}(u_{g-1}) \exp \left[- \frac{(\Sigma_{ne}^{g} + \Sigma_{s_{H}}^{g})}{M_{+}^{g}} \theta \Delta u \right] \right\} - \left[1 - \exp \left(- \frac{(\Sigma_{ne}^{g} + \Sigma_{s_{H}}^{g})}{M_{-}^{g}} (1 - \theta) \Delta u \right) \right]$ (II.66) $+ \hat{\xi}_{o}(u_{g}) \left[1 - \exp \left(- \frac{(\Sigma_{ne}^{g} + \Sigma_{s_{H}}^{g})}{M_{+}^{g}} \theta \Delta u \right) \right] \right\}$

where

$$\Sigma_{ne}^{g} = \Sigma_{t}^{g} + \frac{B^{2}}{3A_{1}^{g}} - \Sigma_{s}^{g}$$

$$A_{+}^{g} = \hat{\xi}_{o}(u_{g}) \Sigma_{t}^{g} + \gamma_{o}(u_{g}) \left[\Sigma_{ne}^{g} + \Sigma_{H}^{g} \right]$$

$$M_{-}^{g} = \hat{\xi}_{o}(u_{g-1}) \Sigma_{t}^{g} + \gamma_{o}(u_{g-1}) \left[\Sigma_{ne}^{g} + \Sigma_{s_{H}}^{g}\right]$$

The group flux ϕ^g may be derived from Eq.II.32

$$\phi^{g} = \left[\frac{\theta q_{o}(u_{g})}{M_{+}^{g}} + \frac{(1-\theta) q_{o}(u_{g-1})}{M_{-}^{g}}\right] \Delta u$$
$$+ (S_{g} + \eta_{o}^{g}) \left[\frac{\theta \gamma_{o}(u_{g})}{M_{+}^{g}} + \frac{(1-\theta) \gamma_{o}(u_{g-1})}{M_{-}^{g}}\right] \qquad (II.68)$$

(II.67)

or from the balance Eq. II.29

$$\phi^{g} = \frac{S^{g} + \eta_{o}^{g} + q_{o}(u_{g-1}) - q_{o}(u_{g})}{\sum_{ne}^{g} + \sum_{h=1}^{g}} .$$
(II.69)

The consistent P_1/B_1 recursion relations are derived by differencing of Eqs. II.35 and II.36 using the approximations

$$\frac{\mathrm{dq}}{\mathrm{du}} \approx \frac{q(u_g) - q(u_{g-1})}{\Delta u}$$

$$f(u)q(u) \approx \theta f_{+}q(u_g) + (1 - \theta) f_{-}q(u_{g-1}).$$

Consistent P_1/B_1 Approximation

$$q_{o}(u_{g}) = \frac{1}{N^{g}} \left\{ \left(\left[\theta Z_{+}^{g} + \frac{1}{\Delta u} \right] \left[(1 - \theta) W_{-}^{g} - \frac{1}{\Delta u} \right] - \theta X_{+}^{g} (1 - \theta) Y_{-}^{g} \right) q_{o}(u_{g-1}) + \left(\left[\theta Z_{+}^{g} + \frac{1}{\Delta u} \right] \left[(1 - \theta) X_{-}^{g} \right] - \theta X_{+}^{g} \left[(1 - \theta) Z_{-}^{g} - \frac{1}{\Delta u} \right] \right) q_{1}(u_{g-1}) - \left(\left[\theta Z_{+}^{g} + \frac{1}{\Delta u} \right] \left[(1 - \gamma_{o}(u_{g}) \theta W_{+}^{g} - (1 - \theta) \gamma_{o}(u_{g-1}) W_{-}^{g} \right] \right) (II.70)$$

$$+ \theta X_{+}^{g} [\gamma_{0}(u_{g}) \theta Y_{+}^{g} + \gamma_{0}(u_{g-1})(1 - \theta) Y_{-}^{g}] \left(\frac{S^{g} + \eta_{0}^{g}}{\Delta u} \right) \\ + \left([\gamma_{1}(u_{g}) \theta X_{+}^{g} + \gamma_{1}(u_{g-1})(1 - \theta) X_{-}^{g}] [\theta Z_{+}^{g} + \frac{1}{\Delta u}] \right) \\ + \theta X_{+}^{g} [1 - \gamma_{1}(u_{g}) \theta Z_{+}^{g} - \gamma_{1}(u_{g-1})(1 - \theta) Z_{-}^{g}] \frac{3}{2} \frac{\eta_{1}^{g}}{\Delta u} \right) \\ + \theta X_{+}^{g} [1 - \gamma_{1}(u_{g}) \theta Z_{+}^{g} - \gamma_{1}(u_{g-1})(1 - \theta) Z_{-}^{g}] \frac{3}{2} \frac{\eta_{1}^{g}}{\Delta u} \right)$$

$$q_{1}(u_{g}) = \frac{-1}{(\theta Z_{+}^{g} + \frac{1}{\Delta u})} \left\{ [(1 - \theta) Z_{-}^{g} - \frac{1}{\Delta u}] q_{1}(u_{g-1}) \right\}$$

+
$$\theta Y_{+}^{g} q_{o}(u_{g}) + (1 - \theta) Y_{-}^{g} q_{o}(u_{g-1})$$
 (11.71)
+ $[\gamma_{o}(u_{g}) \ \theta Y_{+}^{g} + \gamma_{o}(u_{g-1})(1 - \theta) Y_{-}^{g}] \left(\frac{S^{g} + \eta_{o}^{g}}{\Delta u}\right)$

+
$$[\gamma_{1}(u_{g}) \ \theta Z_{+}^{g} + \gamma_{1}(u_{g-1})(1 - \theta) \ Z_{-}^{g}] \frac{3}{2} \frac{\eta_{1}^{g}}{\Delta u} \right\}$$

where

$$N^{g} \equiv (\Theta Y_{+}^{g}) (\Theta X_{+}^{g}) - (\Theta Z_{+}^{g} + \frac{1}{\Delta u}) (\Theta W_{+}^{g} + \frac{1}{\Delta u})$$

$$W^{g} \equiv \frac{1}{\gamma_{o}} \left[(\Sigma_{t}^{g} - \Sigma_{s}^{g} + \Sigma_{s}^{g}) (A_{1}^{g} - \Sigma_{s}^{1}, g + \frac{2}{3} \Sigma_{s}^{g} + \frac{\hat{\xi}_{1}}{\gamma_{1}} \Sigma_{t}^{g}) + \frac{B^{2}}{3} \right] / \Delta^{g}$$

$$X^{g} \equiv \frac{1}{\gamma_{o}\gamma_{1}} \hat{\xi}_{o} \Sigma_{t}^{g} B / \Delta^{g}$$

$$Y^{g} \equiv -\frac{B}{3} \frac{1}{\gamma_{o}\gamma_{1}} \hat{\xi}_{1} \Sigma_{t}^{g} / \Delta^{g}$$

$$Z^{g} \equiv \frac{1}{\gamma} \left[(A_{1}^{g} - \Sigma_{s}^{1}, g + \frac{2}{3} \Sigma_{s}^{g}) (\Sigma_{t}^{g} - \Sigma_{s}^{g} + \Sigma_{s}^{g} + \frac{\hat{\xi}_{o}}{\gamma_{o}} \Sigma_{t}^{g}) + \frac{B^{2}}{3} \right] / \Delta^{g}$$

$$\Delta^{g} \equiv (A_{1}^{g} - \Sigma_{s}^{1}, g + \frac{2}{3} \Sigma_{s}^{g}) (\Sigma_{t}^{g} - \Sigma_{s}^{g} + \Sigma_{s}^{g} + \frac{\hat{\xi}_{o}}{\gamma_{o}} \Sigma_{t}^{g}) + \frac{B^{2}}{3} \right] / \Delta^{g}$$

$$\Delta^{g} \equiv (A_{1}^{g} - \Sigma_{s}^{1}, g + \frac{2}{3} \Sigma_{s}^{g}) + \frac{\hat{\xi}_{1}}{\gamma_{1}} \Sigma_{t}^{g}) (\Sigma_{t}^{g} - \Sigma_{s}^{g} + \Sigma_{s}^{g} + \frac{\hat{\xi}_{o}}{\gamma_{o}} \Sigma_{t}^{g}) + \frac{B^{2}}{3} \right]$$

Expressions for the flux and current follow directly from Eqs. 36,

$$\begin{split} \phi^{g} &= \left[\theta R_{+}^{g} q_{0}(u_{g}) + (1 - \theta) R_{-}^{g} q_{0}(u_{g-1}) - \theta T_{+}^{g} q_{1}(u_{g}) \right. \\ &- (1 - \theta) T_{-}^{g} q_{1}(u_{g-1}) \right] \Delta u + (S^{g} + \eta_{0}^{g}) \left[\gamma_{0}(u_{g}) \theta R_{+}^{g} + \gamma_{0}(u_{g-1}) (1 - \theta) R_{-}^{g} \right] \\ &- \frac{3}{2} \eta_{1}^{g} \left[\gamma_{1}(u_{g}) \theta T_{+}^{g} + \gamma_{1}(u_{g-1}) (1 - \theta) T_{-}^{g} \right] \\ J^{g} &= \left[\theta V_{+}^{g} q_{0}(u_{g}) + (1 - \theta) V_{-}^{g} q_{0}(u_{g-1}) + \theta U_{+}^{g} q_{1}(u_{g}) \right] \end{split}$$

$$+ (1 - \theta) U_{-}^{g} q_{1}(u_{g-1}) \right] \Delta u + (S^{g} + \eta_{o}^{g}) \left[\gamma_{o}(u_{g}) \theta V_{+}^{g} + \gamma_{o}(u_{g-1}) (1 - \theta) V_{-}^{g} \right]$$

$$+ \frac{3}{2} \eta_{1}^{g} \left[\gamma_{1}(u_{g}) \theta U_{+}^{g} + \gamma_{1}(u_{g-1}) (1 - \theta) U_{-}^{g} \right]$$

$$(II.74)$$

where

$$R^{g} \equiv \frac{1}{\gamma_{o}} \left[A_{1}^{g} - \Sigma_{s}^{1} \cdot g + \frac{2}{3} \Sigma_{s_{H}}^{g} + \frac{\xi_{1}}{\gamma_{1}} \Sigma_{t}^{g}\right] / \Delta^{g}$$

$$T^{g} \equiv \frac{B}{\gamma_{1}} \frac{1}{\Delta^{g}}$$

$$U^{g} \equiv \frac{1}{\gamma_{1}} \left[\Sigma_{t}^{g} - \Sigma_{s}^{g} + \Sigma_{s_{H}}^{g} + \frac{\hat{\xi}_{o}}{\gamma_{o}} \Sigma_{t}^{g}\right] / \Delta^{g}$$

$$V^{g} \equiv \frac{B}{3\gamma_{o}} \frac{1}{\Delta^{g}}$$

The calculation of the moderating parameters required in Eqs.II.66-II.75 will be described in Section I. The Hydrogen slowing down density n_{ℓ}^{g} may be calculated recursively. From Eqs. II.37 it follows that

$$\eta_{\ell}(u_{g}) \sim \eta_{\ell}(u_{g-1}) e^{-\frac{(2\ell+1)}{(\ell+1)} \Delta u} + \frac{(\ell+1)}{(2\ell+1)} \frac{\Sigma_{s_{H}}^{g\ell} \phi_{\ell}^{g}}{\Delta u} .$$

$$\left[1 - e^{-\frac{(2\ell+1)}{(\ell+1)} \Delta u}\right] \ell = 0, 1 \qquad (II.76)$$

(11.75)

$$\eta_{\ell}^{g} = \int_{u_{g-1}}^{u_{g}} \eta_{\ell}(u) \, du \, \mathcal{H} \, \eta_{\ell}(u_{g-1}) \, \Delta u \, . \qquad (II.77)$$

The initial conditions for the continuous slowing down calculation are given by

 $\begin{aligned} \eta_{\ell}(0) &= 0 \\ q_{\ell}(0) &= 0 \\ q_{\ell}(u_{CSD}) &= \sum_{i \neq H} \sum_{g=1}^{G} \sum_{g'=G+1}^{NG+1} \Sigma_{s_{i}}^{\ell} (g \neq g') \phi_{\ell}^{g} \quad G > 1 \end{aligned} (II.78) \\ u_{CSD} &\equiv u_{G} = \ell n \left(\frac{E_{max}}{E_{CSD}} \right) \\ G &\equiv \frac{u_{G}}{\Delta u} , NG = \frac{1}{\Delta u} \ell n \left(\frac{E_{max}}{E_{min}} \right) \end{aligned}$

where

G. Elastic Scattering Transfer Matrix

The multigroup spectrum calculation of MC^2-2 requires ultra-fine-group elastic scattering transfer matrices. Using Eqs. II.38 and II.61, the transfer matrices are defined

$$\sigma_{s}^{\ell}(g' \rightarrow g) = \frac{1}{\phi_{\ell}^{g'}} \int_{u_{g-1}}^{u_{g}^{*}} du \int_{u_{g'-1}}^{u_{g'}} du' \frac{\sigma_{s}(u')}{1 - \alpha} P_{\ell}[\mu_{o}(u' \rightarrow u)] e^{-(u-u')} \phi_{\ell}(u')$$

$$\cdot \sum_{n=0}^{N} (2n + 1) f_{n}(u') P_{n}[\mu_{c}(u' \rightarrow u)] \cdot (II.79)$$

Noting that $\sigma_{s}(u')$ does not include the heavy element resolved resonance structure so that in the ultra-fine-group g', $\sigma_{s}(u') \approx \sigma_{s}^{g'}$ and assuming a constant weighting function, Eq. II.79 may be written

$$\sigma_{s}^{\ell}(g' \rightarrow g) = \frac{\sigma_{s}^{g'}}{(1 - \alpha) \Delta u} \sum_{n=0}^{N} (2n + 1) \int_{u_{g-1}}^{u_{g}^{*}} du \int_{u_{g'-1}}^{u_{g'}} du' f_{n}(u') .$$

$$P_{\ell}(\mu_{o}) P_{n}(\mu_{c}) e^{-(u-u')}. \quad (II.80)$$

The code evaluates $\sigma_{g}(g' \rightarrow g)$ in three different options depending upon the mass A of the scattering material.

1. Light Elements

For light materials other than Hydrogen which scatter more than three ultra-fine-groups

$$1 < A < \frac{e^{1.5\Delta u} + 1}{e^{1.5\Delta u} - 1}$$

an algorithm similar to that of the MC^2 code is used to evaluate the transfer matrix. Writing Eq. II.80

$$\sigma_{s}^{\ell}(g' \rightarrow g) = \frac{\sigma_{s}^{g}}{\Delta u} \int_{\substack{u^{*} \\ g'=1}}^{u} du' P_{\ell}(u' \rightarrow g)$$
(II.81)

$$P_{\ell}(u' \rightarrow g) \equiv \sum_{n=0}^{N} \frac{2n+1}{1-\alpha} \int_{u_{g-1}}^{u_{g}^{*}} du f_{n}(u') P_{\ell}(\mu_{o}) P_{n}(\mu_{c}) e^{-(u-u')}$$
(II.82)

$$r \equiv \frac{1 - e^{-(u-u')}}{1 - \alpha} \leq 1$$

so that

$$P_{o}(u' \neq g) = \sum_{n=0}^{N} (2n + 1) \int_{r_{g-1}}^{r_{g}} f_{n}(u') P_{n}(1 - 2r) dr$$

$$= \sum_{n=0}^{N} (2n + 1) \int_{r_{g-1}}^{r_{g}} f_{n}(u') \sum_{m=0}^{n} (-)^{m} \frac{(n + m)!}{(n - m)!} \frac{r^{m}}{m! m!} dr$$

$$= \sum_{m=0}^{N} A_{m} (r_{g}^{m+1} - r_{g-1}^{m+1}) \qquad (II.83)$$

where

$$A_{m} \equiv \sum_{n=m}^{N} f_{n}(u') \qquad \frac{(-)^{m} (2n + 1) (n + m)!}{m! (m + 1)! (n - m)!}$$

Using the analytical expression, Eq.II.83, the transfer matrix of Eq.II.81 is evaluated in the following manner. Each ultra-fine-group, Δu , is subdivided into M hyper-fine-groups, as shown in Fig. 3, such that

$$M\delta u = \Delta u$$
$$u_{g_0} = u_{g}$$
$$u_{g_{M+1}} = u_{g+1}$$
$$u_{g_n} = u_{g} + n\delta u$$

The number of hyper-fine-groups per ultra-fine-group, M, is chosen so that (i) M does not exceed a user input number; (ii) M = 1 if scattering is isotropic or linearly anisotropic in the center of mass system (N \leq 1); (iii) M = 1 if isotope scatters at least twenty ultra-fine-groups, (iv) M is calculated to ensure that isotope scatters at least twenty hyper-finegroups, M = -20 $\Delta u \ ln\alpha$ the scattering from lethargy u g, to hyper-finegroup g, P_o(u_g, \rightarrow g_n) is calculated using Eq. II.83 with the factorial

coefficients for A_m precalculated. The transfer matrix is then obtained using a trapezoidal rule integration,



Fig. 3. Elastic Scattering Group Structure

$$P_{o}(u' \rightarrow g) = \sum_{\substack{g_{n} \in g \\ g_{n} \in g}} P_{o}(u' \rightarrow g_{n})$$

$$\sigma_{s}^{o}(g' \rightarrow g) = \frac{\sigma_{s}^{g'}}{2M} \left[P_{o}(u_{g'-1} \rightarrow g) + P_{o}(u_{g'} \rightarrow g) + 2 \sum_{m=1}^{M-1} P_{o}(u_{g'-1} + m\delta u \rightarrow g) \right].$$
(II.84)

For the last group which can be reached from g', the matrix element is calculated by balance,

$$\sigma_{s}^{o}(g' \rightarrow g^{*}) = \sigma_{s}^{g'} - \sum_{g=g'}^{g^{*}-1} \sigma_{s}^{o}(g' \rightarrow g)$$
(II.85)

where

$$u_{g^{*-1}} < u_{g^{*}} + \ln \frac{1}{\alpha} \le u_{g^{*}}$$

For the consistent P_1/B_1 options of MC²-2, the matrix elements of Equations II.81-II.82 for l=1 are required. Although it is possible to derive $P_1(u' \rightarrow g)$ analytically as with Eq. II.83, this expression is time consuming and difficult to evaluate numerically. It is furthermore possible to obtain a fast and accurate evaluation of the integral by taking

$$P_1(u' \rightarrow g_n) \approx \mu_o(u' \rightarrow \overline{g}_n) P_o(u' \rightarrow g_n).$$
 (II.86)

The small group size, $\delta u \leq \Delta u \approx .008$ makes this a good approximation. The code uses the energy midpoint of the hyper-fine sink group in this calculation. From Eq. II.40, μ_0 is given by

$$\mu_{o}(u' \rightarrow \bar{g}_{n}) \equiv \frac{1}{2} \bar{x}_{n}^{\frac{1}{2}} (1 + A - \frac{A - 1}{\bar{x}_{n}})$$
(II.87)
$$\bar{x}_{n} \equiv \frac{1}{2} e^{-(n-1)\delta u} [1 + e^{-\delta u}] .$$

Equations II.86 and II.87 are used to calculate the $\rm P_1$ matrix in a manner analagous to that of the P_ matrix,

$$P_{1}(u' \rightarrow g) = \sum_{\substack{g_{n} \in g \\ g_{n} \in g}} \mu_{o}(u' \rightarrow \overline{g}_{n}) P_{o}(u' \rightarrow g_{n})$$

$$\sigma_{s}^{1}(g' \rightarrow g) = \frac{\sigma_{s_{1}}^{g'}}{2M} [P_{1}(u_{g'-1} \rightarrow g) + P_{1}(u_{g'} \rightarrow g)$$

$$+ 2 \sum_{m=1}^{M-1} P_{1}(u_{g'-1} + m\delta u \rightarrow g)] \qquad (II.88)$$

where $\sigma_s^{g'}$ is calculated from Eq. II.45,

$$\sigma_{s_1}^g = \sum_{n=0}^N \sigma_s^g f_n^g T_{\ell n}^o(\alpha)$$
(II.89)

and scattering into the last group is calculated by balance as in Eq.II.85

$$\sigma_{s}^{1}(g' \to g^{*}) = \sigma_{s}^{g'} - \sum_{1}^{g^{*}-1} \sigma_{s}(g' \to g) .$$
(II.90)

The accuracy of these methods has been documented by Henryson, et al. (20)

2. Heavy Elements

For materials which scatter less than four ultra-fine-groups (A > 160 for $\Delta u = 1/120$) the method reported by Henryson⁽²¹⁾ is used. Equation II.79 is written

$$\sigma_{s}^{\ell}(g' \rightarrow g) = \frac{\sum_{n=0}^{N} \left\langle \sigma_{n}(u') \phi_{\ell}(u') \right\rangle_{g}, \quad A_{n}^{\ell}(g' \rightarrow g)}{\Delta u \left\langle \phi_{\ell}(u') \right\rangle_{g'}}$$
(II.91)

where $\langle \rangle$ designates a suitable average over the group, and

$$A_{n}^{\ell}(g' \rightarrow g) \equiv \frac{(2n+1)}{1-\alpha} \int_{g'} du' \int_{g} du P_{\ell}(\mu_{o}) P_{n}(\mu_{c}) e^{-(u-u')} . \quad (II.92)$$

As with the light elements, one takes the group width to be small enough to permit a constant weight function so that

$$\left\langle \sigma_{n}(u') \phi_{\ell}(u') \right\rangle_{g'} \stackrel{\sim}{\sim} 1/2 \sigma_{s}^{g'} [f_{n}(u_{g'-1}) + f_{n}(u_{g'})].$$

As shown in Ref. 21, it is possible to change variables and switch the order to integration so as to write Eq. II.92 in terms of a function

$$\overline{T}_{\ell n}^{m}(\alpha,\beta) \equiv \frac{(-)^{m}}{m!} \frac{(2n+1)}{2} \int_{0}^{\beta} dU \ U^{m} P_{\ell}(\mu_{o}) P_{n}(\mu_{c}) \ (-\frac{d\mu_{c}}{dU}) \ .$$
 (II.93)

It is not difficult to show that on integrating Eq.II.92 over all sink groups,

$$A_n^{\ell}(g') = \Delta u T_{\ell n}^{O}(\alpha) = \Delta u \overline{T}_{\ell n}^{O}(\alpha, \ell n \frac{1}{\alpha})$$

where the T function was defined by Equation II.42. The matrix elements $A_n^{\ell}(g' \rightarrow g)$ are given in Table II. These precalculated elements are given on File 8 of the MC²-2 library (Appendix C) for each of the heavy isotopes and are used by MC²-2 in the calculation of Equation II.91.

-36-

TABLE II. Heavy Mass Matrix Elements $A_n^{\ell}(j \rightarrow k)$

k	Group Width/Mass (q $\equiv ln 1/\alpha$)	$A_n^{\ell}(j \neq k)$
	1/3 q <u><</u> ∆u < 1/2 q	
j		$\Delta u \bar{T}_{\ell n}^{0}(\alpha, \Delta u) + \bar{T}_{\ell n}^{1}(\alpha, \Delta u)$
j + 1		$2\Delta u[\bar{T}^{0}_{\ell n}(\alpha, 2\Delta u) - \bar{T}^{0}_{\ell n}(\alpha, \Delta u)] + \bar{T}^{1}_{\ell n}(\alpha, 2\Delta u) - 2\bar{T}^{1}_{\ell n}(\alpha, \Delta u)$
j + 2		$3\Delta u T_{\ell n}^{0}(\alpha) + T_{\ell n}^{1}(\alpha) - 2\overline{T}_{\ell n}^{1}(\alpha, 2\Delta u) + \overline{T}_{\ell n}^{1}(\alpha, \Delta u) - \Delta u [4\overline{T}_{\ell n}^{0}(\alpha, 2\Delta u) - \overline{T}_{\ell n}^{0}(\alpha, \Delta u)]$
j + 3		$-2\Delta u T_{\ell n}^{0}(\alpha) - T_{\ell n}^{1}(\alpha) + 2\Delta u \overline{T}_{\ell n}^{0}(\alpha, 2\Delta u) + \overline{T}_{\ell n}^{1}(\alpha, 2\Delta u)$
$1/2 q \leq \Delta u < q$		
j		$\Delta u \overline{T}_{\ell n}^{0}(\alpha, \Delta u) + \overline{T}_{\ell n}^{1}(\alpha, \Delta u)$
j + 1		$2\Delta u T_{\ell n}^{0}(\alpha) + T_{\ell n}^{1}(\alpha) - 2\Delta u \overline{T}_{\ell n}^{0}(\alpha, \Delta u) - 2\overline{T}_{\ell n}^{1}(\alpha, \Delta u)$
j + 2		$-T^{1}_{\ell n}(\alpha) - \Delta u T^{0}_{\ell n}(\alpha) + \Delta u \overline{T}^{0}_{\ell n}(\alpha, \Delta u) + \overline{T}^{1}_{\ell n}(\alpha, \Delta u)$
$\Delta u \geq q$		
j		$\Delta u T_{\ell n}^{0}(\alpha) + T_{\ell n}^{1}(\alpha)$
j + 1		$-T_{\ell n}^{1}(\alpha)$

-37-

3. Hydrogen Elastic Scattering

It is impossible to treat Hydrogen elastic scattering in the manner outlined for the light elements above because of the computer storage problems associated with a full ultra-fine-group scattering band. Simple recursive relationships exist which are used by MC^2-2 in lieu of storing the Hydrogen matrix. These relationships are given below.

Let $S^g_{\ell_H}$ be the P_{ℓ} elastic scattering source into group g from all

groups above g due to scattering from Hydrogen. Assuming the scattering to be isotropic in the center of mass system, and taking A=1, Eqs. II.40 and II.79 are used to give

$$S_{\ell_{\rm H}}^{g} \equiv \int_{u_{g-1}}^{u_{g}} du \int_{0}^{u_{g-1}} du' \Sigma_{s_{\rm H}}(u') \phi_{\ell}(u') e^{-(1 - \frac{\ell}{2})(u-u')}, \qquad (II.94)$$

$$\ell = 0, 1.$$

Performing the integration over group g it is simple to show

$$S_{\ell_{H}}^{g} = \eta_{\ell}(u_{g-1}) \left[1 - e^{-(1 + \frac{\ell}{2})\Delta u}\right], \quad \ell = 0, 1$$
 (II.95)

where the Hydrogen slowing down density $\eta_{\boldsymbol{\ell}}(u)$ is given by Eqs.II.37 and II.76.

H. Inelastic and (n,2n) Scattering

As noted in Section B above, non-elastic scattering (inelastic, (n,2n)) is treated as a source term in the ultra-fine-group spectrum calculation. The development of the continuous slowing down equations made use of this separation explicitely. In the multigroup formulation this separation is implicit since the code does not prepare a non-elastic ultra-fine-group scattering matrix because of the excessive storage requirements for such a matrix. Rather, the non-elastic scattering source is calculated directly on each pass through the ultra-fine-groups. This is the most time consuming part of the ultra-fine-group spectrum calculation in MC^2-2 .

For both inelastic and (n,2n) scattering, the MC²-2 code permits three descriptions of the secondary energy distributions: (i) tabulated function; (ii) evaporation spectrum; (iii) discrete levels. The remainder of this section will be concerned with the methods used to calculate the ultra-fine-group non-elastic scattering source for each of these descriptions.

1. Tabulated Function

 $P_x(g \neq E_{tab})$

Etab

KΤ

σ^g

^xtab

The following data are provided on the MC^2-2 library file MCC2F6 (c.f. Appendix C),

Probability that a neutron is scattered by process x (inelastic or (n,2n)) from group g to energy point E tab

An array of "sink" energy points

An interpolation law on the sink energies E_{tab}

Cross section in group g for process x multiplied by the fraction of scattering events described by the P_x law

From these definitions it is obvious that the non-elastic scattering source into group g described by the tabulated law is given by

$$S_{ne_{tab}}^{g} = \sum_{g'} \phi^{g'} \left\{ \sigma_{inel_{tab}}^{g'} \int_{E_{g-1}}^{E_{g}} dE P_{inel}(g' \neq E) + 2 \sigma_{n,2n_{tab}}^{g'} \int_{E_{g-1}}^{E_{g}} dE P_{n,2n}(g' \neq E) \right\}.$$
(II.96)

The integrals of Eq. II.96 are evaluated analytically using the interpolation law defined by the KT data.

2. Evaporation Spectrum

For both inelastic and (n,2n) scattering, the MC²-2 library provides evaporation temperatures and fractional probabilities such that

$$S_{ne_{evap}}^{g} = S_{evap_{inel}}^{g} + 2 S_{evap_{n,2n}}^{g}$$

$$S_{evap_{x}}^{g} = \sum_{g'} \sigma_{x}^{g'} \phi^{g'} \sum_{n} \frac{W_{n,x}^{g'}}{I_{n,x}^{g'}} \int_{E_{g-1}}^{E_{g}} dE P_{ev,x,n}(g' \rightarrow E) \qquad (II.97)$$

$$P_{ev,x,n}(g' \rightarrow E) = \begin{cases} 0 & E \geq E_{g'-1} - U_{x,n} \\ E \exp \left(-\frac{E}{\theta_{x,n}^{g'}}\right) & E \leq E_{g'-1} - U_{x,n} \end{cases}$$

where the subscript n is used to indicate a sum over all evaporation spectra for process x and $I_{n,x}^{g'}$ is a normalization factor,

$$I_{n,x}^{g'} = \int_{0}^{E_{g'-1}} \int_{0}^{-U_{x,n}} dE P_{ev}(g' \neq E)$$

For (n,2n) events the constant U is assumed to be zero.

The evaluation of Eq.II.97 requires an exponential for each ultrafine sink group, and it is this calculation which controls the computing time of an MC^2-2 spectrum calculation. A fast exponential function ⁽²²⁾ (Appendix A) is used by MC^2-2 to evaluate the required exponentials. Furthermore, the calculation over sink groups g is terminated once the criterion

$$\frac{\int_{E_{g-1}}^{E_{g}} dE P_{ev,x,n}(g' \rightarrow E)}{\int_{E_{g-1}}^{E_{g'-1}} dE P_{ev,x,n}(g' \rightarrow E)} \leq 10^{-4}$$

is met. The integrals of Eq.II.97 are calculated analytically.

3. <u>Discrete</u> Levels

The MC^2-2 library file MCC2F6 provides Q values and ultra-finegroup level cross sections for all inelastic and/or (n,2n) discrete scattering levels. In addition the library provides the average cosine of the scattering angle in the center of mass system for discrete inelastic scattering by level and group. These data are processed in the manner indicated below. а.

Approximate Treatment Neglecting Energy-Angle Correlation

It is well known that energy and momentum conservation give

$$E = \frac{1 + A^{2}}{(1 + A)^{2}} E' - \frac{A}{A + 1} Q_{\lambda} + \frac{2A}{(1 + A)^{2}} \mu E' \left[1 - \frac{A + 1}{A} \frac{Q_{\lambda}}{E'}\right]^{\frac{1}{2}}$$
(II.98)

where

E = lab energy after collision E' = lab energy before collision μ = cosine of c.m. angle of scatter Q_{λ} = -Q value of reaction for level λ A = mass of scattering isotope.

The threshold energy of the reaction is

$$E_{\lambda} = \frac{A+1}{A} Q_{\lambda} , \qquad (II.99)$$

A rigorous evaluation of the group to group transfer probability accounting for the energy-angle correlation of Eq.II.98 is quite complicated. ⁽²³⁾ Two assumptions serve to simplify the situation without much loss in accuracy:

(i) angle of scattering is fixed;

(ii) $\left(1-\frac{E_{\lambda}}{E'}\right)^{\frac{1}{2}} \approx 1-\frac{1}{2} \frac{E_{\lambda}}{E'}$.

With these assumptions one can write

$$E = E' \left[\frac{1 + A^2 + 2A < \mu}{(1 + A)^2} \right] - \frac{(A + < \mu)}{A + 1} Q_{\lambda}$$
(II.100)
$$E' = \frac{(1 + A)^2}{1 + A^2 + 2A < \mu} E + \frac{(A + < \mu)(A + 1) Q_{\lambda}}{1 + A^2 + 2A < \mu} .$$
(II.101)

It is clear from Eq. II.100 that the probability of scattering from group j to group k for discrete level λ is identically zero unless group k lies between the energy boundaries

$$E_{j-1} \left[\frac{1 + A^2 + 2A < \mu}{(1 + A)^2} \right] - \frac{(A + \langle \mu \rangle) Q_{\lambda}}{1 + A}$$

$$\max(E_{j}, E_{\lambda}) \left[\frac{1 + A^{2} + 2A < \mu}{(1 + A^{2})} \right] - \frac{(A + \langle \mu \rangle)}{1 + A} Q_{\lambda}$$
(II.102)

and it has been assumed that at least a part of group j lay above the threshold,

 $E_{j-1} > E_{\lambda}$.

For those groups, k, which fall partially or totally within this range, the probability of scattering from group j to group k is given by the fractional part of group j which scatters into group k. This can be derived directly from Eq. II.101,

$$P_{\lambda}(j \rightarrow k) \equiv \frac{E_{j-1}^{*} - E_{j}^{*}}{E_{j-1} - E_{j}}$$
 (II.103)

where

$$E_{j-1}^{*} = \min \left[E_{j-1}^{*}, \frac{(1 + A^{2}) E_{k-1}^{*} + (A + 1) (A + \langle \mu \rangle) Q_{\lambda}}{1 + \Lambda^{2} + 2\Lambda \langle \mu \rangle} \right]$$

$$E_{j}^{*} = \max \left[E_{j}^{*}, E_{\lambda}^{*}, \frac{(1 + A^{2}) E_{k}^{*} + (A + 1) (A + \langle \mu \rangle) Q_{\lambda}}{1 + A^{2} + 2\Lambda \langle \mu \rangle} \right]. \quad (II.104)$$

The Eqs.II.102-II.104 are used by $MC^{2}-2$ to calculate the discrete scattering source,

$$S_{ne,d}^{g} = \sum_{g'} \phi^{g'} \left\{ \sum_{\lambda} \sigma_{inel_{\lambda}}^{g'} P_{\lambda}(g' \neq g) + 2 \sum_{\lambda'} \sigma_{n,2n_{\lambda'}}^{g'} P_{\lambda'}(g' \neq g) \right\}$$

The average cosine of the scattering angle is taken to be zero (isotropic in center of mass) for scattering from all (n,2n) levels. The λ,λ' sums represent sums over all discrete levels.

The complete microscopic non-elastic scattering source is given by

 $S_{ne}^{g} = S_{ne_{tab}}^{g} + S_{ne_{cvp}}^{g} + S_{ne_{d}}^{g}$.

The macroscopic sources are defined by summing over all materials using the appropriate atom densities.

b. Rigorous Treatment Assuming Isotropic Scattering

The approximations made above are quite good for source energies far above threshold where the level cross sections σ_{λ}^{g} are large. The simplicity of the formulation along with its accuracy suggest that Eqs. II.102-II.104 provide the recommended procedure for evaluating the discrete inelastic source. On option the user may, however, specify a more rigorous, and correspondingly more time consuming, treatment of discrete level scattering. The option fol-

lows the work of Segev, ⁽²³⁾ which accounts

explicitely for the fact that a neutron scattered at energy E' scatters into a band of energies defined from Eq. II.98 as

$$\left[\frac{1 \pm A \left(1 - \frac{E_{\lambda}}{E'}\right)^{1/2}}{1 + A}\right]^{2} E' . \qquad (II.105)$$

This band of energies must be used instead of Eq. II.100 to define the possible sink groups. Similarly, a neutron scattered into energy E' may be scattered from a range of source energies defined by

$$\begin{bmatrix} \frac{1+A\left(1+\frac{A-1}{A+1}\frac{E_{\lambda}}{E}\right)^{1/2}}{1-A} \end{bmatrix}^{2} E$$
(II.106)
$$\left(\int_{a} \int_{a}$$

and

$$\max \left\{ E_{\lambda}^{(c)}, \left[\frac{1 - A \left(1 + \frac{A - 1}{A + 1} \frac{E_{\lambda}}{E} \right)^{1/2}}{1 - A} \right] \right\}$$
$$E_{\lambda}^{(c)} = \frac{A^{2}}{A^{2} - 1} E_{\lambda}$$

where

max

$$P_{\lambda}(j \rightarrow k) = \frac{1}{E_{j-1} - E_{j}} \int_{\text{group } j} dE' \int_{\text{group } k} dE$$

•
$$\sum_{n} \frac{(2n+1)}{(1-\alpha)E'} \frac{f_n(E')}{\left[1 - \frac{E_{\lambda}}{E'}\right]^{\frac{1}{2}}} P_n(\mu)$$
 (II.107)

Ε

In the more rigorous option of MC^2-2 , Eq. II.107 is evaluated analytically taking proper account of the four domains of integration as discussed in Ref. 23. Because of the time consuming nature of the calculation, Eq. II.107 is solved assuming isotropic scattering in the center-ofmass. For standard fast reactor configurations, it has been found that the discrete inelastic calculation using the approximate method defined by Eqs. II. 103-II. 104 is quite accurate and significantly faster than the algorithms discussed above which account explicitely for the energyangle correlation of discrete level scattering as defined by Eq. II.98.

I. Ultra-Fine-Group Macroscopic Data

The macroscopic data required for solution of the spectrum equations discussed in Section F may in almost all cases be derived from the simple expression

$$\Sigma_{\mathbf{x}}^{\mathbf{g}} = \sum_{\mathbf{i}} N_{\mathbf{i}} \sigma_{\mathbf{x}_{\mathbf{i}}}^{\mathbf{g}}$$

where N_i is the atom density of isotope i in the homogeneous mix and σ_x^g

is the ultra-fine-group cross section for isotope i and process x. This expression is clearly applicable to the scattering matrix data discussed in Sections G and H above. The ultra-fine-group microscopic reaction cross sections are available on the MC^2-2 library file MCC2F5 (c.f. Appendix C). These data are generally processed directly from the ENDF/B data files by the code ETOE-2 assuming a constant weighting.

The unresolved resonance cross sections are treated in the same manner as the above data. The calculation of the ultra-fine-group unresolved cross sections is discussed in Chapter IV. The remainder of this section is concerned with the calculation of macroscopic data which require more processing than that given by Eq. II.108.

1. Fission Source Data

The fission source of Eq. II.14 may be written in group form as

$$S_{f}^{g} = \sum_{i} \chi_{i}^{g} N_{i} \sum_{g'} \upsilon_{i}^{g'} \sigma_{f_{i}}^{g'} \phi^{g'} . \qquad (II.109)$$

The number of neutrons per fission, v_i^g , is derived from the expression

$$v_{i}^{g} = \frac{1}{E_{g-1} - E_{g}} \int_{E_{g}}^{E_{g-1}} v_{i}(E) dE$$

$$v_{i}(E) = A_{o}^{i} + A_{1}^{i}E + A_{2}^{i}E^{2} + A_{3}^{i}E^{3}.$$
(II.110)

The fission spectrum distribution is given by

$$\chi_{i}^{g} = \frac{\int_{E_{g}}^{E_{g-1}} dE \chi_{i}(E)}{\int_{E_{min}}^{E_{max}} dE \chi_{i}(E)} = \frac{\hat{\chi}_{i}^{g}}{\sum_{g}, \hat{\chi}_{i}^{g'}}$$
(II.111)

with

$$\chi_{i}(E) = \alpha_{i} \frac{E}{\tau_{i}^{2}} e^{-E/\tau_{i}} + (1 - \alpha_{i}) \sqrt{\frac{4E}{\pi\beta_{i}^{3}}} e^{-E/\beta_{i}}$$

(II.108)

This combination of a Maxwellian plus an evaporation spectrum suffices to describe all but one of the fission spectra for the most recent release of the ENDF/B data files. Integration of Eq. II.111 gives

$$\begin{aligned} \lambda_{i}^{g} &= \alpha_{i} \left[(1 + \frac{E_{g}}{\tau_{i}}) e^{-E_{g}/\tau_{i}} - (1 + \frac{E_{g-1}}{\tau_{i}}) e^{-E_{g-1}/\tau_{i}} \right] \\ &+ (1 - \alpha_{i}) \left[erf \left[(E_{g-1}/\beta_{i})^{\frac{1}{2}} \right] - erf \left[(E_{g}/\beta_{i})^{\frac{1}{2}} \right] \\ &- \left(\frac{4E_{g-1}}{\pi\beta_{i}} \right)^{\frac{1}{2}} e^{-E_{g-1}/\beta_{i}} + \left(\frac{4E_{g}}{\pi\beta_{i}} \right)^{\frac{1}{2}} e^{-E_{g}/\beta_{i}} \right] . \end{aligned}$$
(II.112)

Equation II.112 is used by $MC^{2}-2$ to evaluate the fission spectrum distribution. It should be noted that the "temperatures" τ_{i} and β_{i} are not energy dependent in the $MC^{2}-2$ formulation although such a description is not rigorous. In practice the temperatures in the $MC^{2}-2$ library are those characteristic of the average fission energy of a typical fast reactor assembly. It is possible on user option to input the fission spectrum temperature β_{i} by isotope which the code will use in lieu of the library data taking α_{i} to be zero.

2. Ultra-Fine-Group Extended Transport Cross Section

The transport-like cross sections A^g_{ℓ} defined as

$$A_{\ell}^{g} = \int_{u_{g-1}}^{u_{g}} A_{\ell}(B, u, N) du$$

are required in solution of the ultra-fine-group spectrum equations. $A_{\ell}(B, u, N)$ is defined by the continued fraction expansion of Eqs.II.11 and II.12. The MC²-2 code does not actually perform the required integration. Rather the parameters b_{ℓ} are simply defined in terms of the group cross sections, so that, for example,

$$A_{1}^{g} = \frac{B \tan^{-1} B/\Sigma_{t}^{g}}{3 (1 - \frac{\Sigma_{t}^{g}}{B} \tan^{-1} \frac{B}{\Sigma_{t}^{g}})}$$

in the consistent B_1 approximation. The cross sections $A_{\mathcal{L}}^g$ are clearly dependent upon the buckling B^2 which may change during the coarse of a calculation if the user specifies a buckling search option. The code does not recalculate $A_{\mathcal{L}}^g$ in that case. Rather, a special user input buckling, κ^2 is used in the calculation of $A_{\mathcal{L}}^g$. This same value of κ^2 is also used

in the leakage correction to the background cross section in the resonance calculation of $MC^{2}-2$, as described in Chapters III and IV. The value of κ^{2} is set by the user and may or may not equal the B^{2} used in the ultra-fine group spectrum calculation.

The continued fraction coefficients depend upon the ratio of Legendre functions of the second kind which can not be calculated for all arguments using the standard forward recurrence relation without a significant loss of accuracy. The algorithms used in MC^2-2 to calculate this ratio are summarized in Appendix A.

3. Improved Goertzel-Greuling Moderating Parameters

In Section D.2 integral expressions were derived defining the isotopic moderating parameters ξ_i^l , a_i^l , c_i^l and e_i^l . The numerical evaluation of these expressions is described below.

The heavy isotope moderating parameters are given by Eq.II.44. The effective ratios are calculated by integrating over the scattering bandwidth,

$$\left\langle \frac{\Sigma_{s_{i}}}{\Sigma_{t}} f_{n_{i}} \right\rangle_{u} \stackrel{\sim}{\sim} \frac{1}{\ell_{n}} \frac{1}{\alpha_{i}} \left\{ \Delta u \sum_{n=1}^{N_{i}} \frac{\Sigma_{s_{i}} (u - (n - \frac{1}{2}) \Delta u)}{\Sigma_{t} (u - (n - \frac{1}{2}) \Delta u)} f_{n_{i}} (u - (n - 1) \Delta u) \right.$$
(II.113)

+
$$(\ln \frac{1}{\alpha_{i}} - N_{i} \Delta u) \frac{\sum_{i} (u - (N_{i} + \frac{1}{2}) \Delta u)}{\sum_{i} (u - (N_{i} + \frac{1}{2}) \Delta u)} f_{n_{i}} (u - N_{i} \Delta u)$$

where N_i is the nearest whole integer to the ratio $-\frac{\ln\alpha_i}{\Delta u} \left(= \left[-\frac{\ln\alpha_i}{\Delta u} \right] \right)$. The ratio is evaluated at ultra-fine-group boundaries and group cross sections are used so that

$$\frac{\sum_{i} (u_{g} - \frac{1}{2} \Delta u)}{\sum_{t} (u_{g} - \frac{1}{2} \Delta u)} \equiv \frac{\sum_{i}^{g}}{\sum_{t}^{g}}$$

The evaluation of the light element moderating parameters as defined by Eqs. II.48 is somewhat more complex. The first step in the development is the definition of integrating factors which permit an accurate numerical evaluation of the integrals:

$$\int_{u-\ell n^{1}/\alpha_{i}}^{u} du' \frac{\Sigma_{s}(u')}{\Sigma_{t}(u')} \mathcal{D}_{m}^{i}(u') \approx W_{l} \sum_{n=1}^{N_{i}} \frac{\Sigma_{s}(u-(n-\frac{1}{2}) \Delta u)}{\Sigma_{t}(u-(n-\frac{1}{2}) \Delta u)} \mathcal{D}_{m}^{i}(u-(n-\frac{1}{2}) \Delta u)$$

$$\int_{u-\ell n^{1}/\alpha_{1}}^{u} du' \frac{\sum_{i}^{(u')}}{\sum_{t}^{(u')}} \mathcal{D}_{m}^{i}(u') (u - u') \approx W_{2} \sum_{i n=1}^{N_{i}} \frac{\sum_{i}^{(u - (n - \frac{1}{2}) \Delta u}}{\sum_{t}^{(u - (n - \frac{1}{2}) \Delta u)}}$$
$$\mathcal{D}_{m}^{i}(u - (n - \frac{1}{2}) \Delta u) (n - \frac{1}{2})$$

-47-

$$\int_{u-\ln^{1}/\alpha_{i}}^{u} du' \frac{\sum_{i}^{s} (u')}{\sum_{t}^{(u')}} \mathcal{D}_{m}^{i}(u') e^{-x(u-u')} \mathcal{X}$$
(II.114)

$$W_{3_{i}n=1}^{x} \sum_{t=1}^{N_{i}} \frac{\sum_{t=1}^{N_{i}} (u - (n - \frac{1}{2}) \Delta u)}{\sum_{t=1}^{N_{i}} \sum_{t=1}^{N_{i}} (u - (n - \frac{1}{2}) \Delta u)} \mathcal{D}_{m}^{i}(u - (n - \frac{1}{2}) \Delta u) e^{-x(n - \frac{1}{2}) \Delta u}$$

$$\int_{u-\ln^{1}/\alpha_{i}}^{u} du' \frac{\sum_{i}^{u}(u')}{\sum_{i}^{u}(u')} \mathcal{D}_{m}^{i}(u') (u - u') e^{-x(u-u')} \mathcal{X}$$

$$N_{i} \sum_{\substack{s_{i} \\ i \\ n=1}}^{N_{i}} \frac{\sum_{\substack{s_{i} \\ \overline{\Sigma}_{t}(u - (n - \frac{1}{2}) \Delta u)}}{\sum_{i} (u - (n - \frac{1}{2}) \Delta u)} \mathcal{D}_{m}^{i}(u - (n - \frac{1}{2}) \Delta u) (n - \frac{1}{2}) e^{-x(n - \frac{1}{2})\Delta u}$$

It is clear from the development of the improved Goertzel-Greuling slowing down theory that if

$$\frac{\sum_{\substack{\mathbf{s}\\\underline{\Sigma}\\\mathbf{t}}}^{\Sigma} f_{n}^{i}$$

is constant over a scattering band, then the IGG moderating parameters should reduce to the standard Goertzel-Greuling form. If this characteristic of the moderating parameters is retained, then it is possible to define integration factors in the following manner:

$$W_{1_{i}} \equiv \ell n^{1} / \alpha_{i} / N_{i}$$

$$W_{2_{i}} \equiv (\ell n^{1} / \alpha_{i} / N_{i})^{2}$$

$$W_{3_{i}}^{x} \equiv \frac{1}{x} (1 - \alpha_{i}^{x}) / \sum_{n=1}^{N_{i}} e^{-x(n-\frac{1}{2})\Delta u}$$

$$W_{4_{i}}^{x} \equiv \frac{1}{x^{2}} (1 - \alpha_{i}^{x}(1 + x \ell n^{1} / \alpha_{i})) / \sum_{n=1}^{N_{i}} (n - \frac{1}{2}) e^{-x(n-\frac{1}{2})\Delta u}.$$
(II.115)

The improved Goertzel-Greuling moderating parameters are then calculated by replacing the integrals of Eqs. II.48 by the numerical approximations

$$\begin{split} \xi_{\underline{i}}^{0}(u) &= \sum_{n'=1}^{N+1} \frac{1}{n'} \left[w_{\underline{i}}^{n'} + w_{\underline{i}}^{n'} (u) - \alpha_{\underline{i}}^{n'} w_{\underline{i}} + \alpha_{\underline{i}}^{n'} (u) \right] \\ a_{\underline{i}}^{0}(u) &= -\sum_{n'=1}^{N+1} \frac{1}{n'} \left[w_{\underline{i}}^{n'} + f_{\underline{i}}^{n',n'} - \alpha_{\underline{i}}^{n'} w_{\underline{2}} + g_{\underline{i}}^{n'} (u) \right] \\ c_{\underline{i}}^{0}(u) &= \sum_{\underline{i}_{\underline{i}}(u) - \sum_{n'=1}^{N+1} w_{\underline{i}}^{n'} + f_{\underline{i}}^{n',n'} - \alpha_{\underline{i}}^{n'} w_{\underline{2}} + g_{\underline{i}}^{n'} (u) \right] \\ c_{\underline{i}}^{0}(u) &= \sum_{n'=1}^{N+1} w_{\underline{i}}^{n'} + f_{\underline{i}}^{n',n'} \\ (II.116) \\ c_{\underline{i}}^{1}(u) &= \sum_{n'=1}^{N+1} \left\{ \frac{(A_{\underline{i}}+1)}{2n'+1} - \left[w_{\underline{3}_{\underline{i}}}^{n'+\underline{i}_{\underline{i}},n'} (u) - \alpha_{\underline{i}}^{n'+\underline{i}_{\underline{i}}} w_{\underline{i}_{\underline{i}}} + d_{\underline{i}}^{n'} (u) \right] \right\} \\ &= \sum_{n'=1}^{(A_{\underline{i}}-1)} \left[w_{\underline{3}_{\underline{i}}}^{n'+\underline{i}_{\underline{i}}} b_{\underline{i}}^{n'+\underline{i}_{\underline{i}},n'} (u) - \alpha_{\underline{i}}^{n'+\underline{i}_{\underline{i}}} w_{\underline{i}_{\underline{i}}} d_{\underline{i}}^{n'} (u) \right] \\ &= \frac{(A_{\underline{i}}-1)}{2n'+1} \left[w_{\underline{3}_{\underline{i}}}^{n'+\underline{i}_{\underline{i}}} b_{\underline{i}}^{n'+\underline{i}_{\underline{i}},n'} (u) - \alpha_{\underline{i}}^{n'+\underline{i}_{\underline{i}}} w_{\underline{i}_{\underline{i}}} d_{\underline{i}}^{n'} (u) \right] \\ &= \frac{(A_{\underline{i}-1})}{2n'+1} \left[w_{\underline{3}_{\underline{i}}}^{n'+\underline{i}_{\underline{i}}} b_{\underline{i}}^{n'+\underline{i}_{\underline{i}},n'} (u) - \alpha_{\underline{i}_{\underline{i}}}^{n'+\underline{i}_{\underline{i}}} w_{\underline{i}_{\underline{i}}} g_{\underline{i}}^{n'} (u) \right] \\ &= \frac{(A_{\underline{i}-1)}}{2n'+1} \left[w_{\underline{i}_{\underline{i}}}^{n'+\underline{i}_{\underline{i}}} f_{\underline{i}}^{n'+\underline{i}_{\underline{i}},n'} (u) - \alpha_{\underline{i}_{\underline{i}}}^{n'+\underline{i}_{\underline{i}}} w_{\underline{i}_{\underline{i}}} g_{\underline{i}}^{n'} (u) \right] \\ &= \frac{(A_{\underline{i}-1)}}{2n'+1} \left[w_{\underline{i}_{\underline{i}}}^{n'+\underline{i}_{\underline{i}}} f_{\underline{i}}^{n'+\underline{i}_{\underline{i}},n'} (u) - \alpha_{\underline{i}_{\underline{i}}}^{n'+\underline{i}_{\underline{i}}} w_{\underline{i}_{\underline{i}}} g_{\underline{i}}^{n'} (u) \right] \\ &= \frac{(A_{\underline{i}-1)}}{2n'+1} \left[w_{\underline{i}_{\underline{i}}}^{n'+\underline{i}_{\underline{i}}} f_{\underline{i}}^{n'+\underline{i}_{\underline{i}},n'} (u) - \alpha_{\underline{i}_{\underline{i}}}^{n'+\underline{i}_{\underline{i}}} w_{\underline{i}_{\underline{i}}} w_{\underline{i}_{\underline{i}}} w_{\underline{i}} w_{\underline{i}$$

-48-

$$f_{i}^{x,m}(u) = \sum_{n=1}^{N_{i}} \frac{\sum_{i}^{x} (u - (n - \frac{1}{2}) \Delta u)}{\sum_{i} (u - (n - \frac{1}{2}) \Delta u)} D_{m}^{i}(u - (n - \frac{1}{2}) \Delta u) (n - \frac{1}{2}) e^{-x(n - \frac{1}{2})\Delta u}$$

$$g_{i}^{m}(u) = \sum_{n=1}^{N} \frac{\sum_{i}^{\sum} (u - (n - \frac{1}{2}) \Delta u)}{\sum_{i}^{\sum} (u - (n - \frac{1}{2}) \Delta u)} \mathcal{D}_{m}^{i}(u - (n - \frac{1}{2}) \Delta u) (n - \frac{1}{2})$$

The Eqs. II.117 are used to calculate the coefficients at the lethargy U_{CSD} between the multigroup and continuous slowing down regions of the calculation. For the lethargy boundaries $u_g > U_{CSD}$ recursion relations are used:

$$b_{i}^{x,m}(u + \Delta u) = b_{i}^{x,m}(u) e^{-x\Delta u} + e^{-x} \frac{\Delta u}{2} \left\{ \frac{\sum_{i} (u + \frac{\Delta u}{2})}{\sum_{t} (u + \frac{\Delta u}{2})} \mathcal{D}_{m}^{i}(u + \frac{\Delta u}{2}) - \frac{\sum_{i} (u - (N_{i} - \frac{1}{2}) \Delta u)}{\sum_{t} (u - (N_{i} - \frac{1}{2}) \Delta u)} \mathcal{D}_{m}^{i}(u - (N_{i} - \frac{1}{2}) \Delta u) e^{-xN_{i}\Delta u} \right\}$$

$$d_{i}^{m}(u + \Delta u) = d_{i}^{m}(u) + \left\{ \frac{\sum_{i} (u + \frac{\Delta u}{2})}{\sum_{t} (u + \frac{\Delta u}{2})} \mathcal{D}_{m}^{i}(u + \frac{\Delta u}{2}) - \frac{\sum_{i} (u - (N_{i} - \frac{1}{2}) \Delta u)}{\sum_{t} (u - (N_{i} - \frac{1}{2}) \Delta u)} \mathcal{D}_{m}^{i}(u - (N_{i} - \frac{1}{2}) \Delta u) \right\}$$

$$f_{i}^{x,m}(u + \Delta u) = e^{-x\Delta u} f_{i}^{x,m}(u) + b_{i}^{x,m}(u + \Delta u)$$
(II.118)

$$-e^{-x\Delta u/2} \begin{cases} \frac{1}{2} \frac{\sum_{s} (u + \frac{\Delta u}{2})}{\sum_{t} (u + \frac{\Delta u}{2})} D_{m}^{i}(u + \frac{\Delta u}{2}) \end{cases}$$

+
$$(N_{i} - \frac{1}{2}) \frac{\sum_{i} (u - (N_{i} - \frac{1}{2}) \Delta u)}{\sum_{i} (u - (N_{i} - \frac{1}{2}) \Delta u)} \mathcal{D}_{m}^{i}(u - (N_{i} - \frac{1}{2}) \Delta u) e^{-xN_{i}\Delta u}$$

-49-

$$g_{1}^{m}(u + \Delta u) = g_{1}^{m}(u) + d_{1}^{m}(u + \Delta u)$$

$$- \frac{1}{2} \frac{\sum_{i} (u + \frac{\Delta u}{2})}{\sum_{i} (u + \frac{\Delta u}{2})} \mathcal{D}_{m}^{i}(u + \frac{\Delta u}{2})$$

$$- (N_{1} - \frac{1}{2}) \frac{\sum_{i} (u - (N_{1} - \frac{1}{2}) \Delta u)}{\sum_{i} (u' - (N_{1} - \frac{1}{2}) \Delta u)} \mathcal{D}_{m}^{i}(u - (N_{1} - \frac{1}{2}) \Delta u) .$$

As noted in Section D.2, the sums over Legendre coefficients of Eqs. II.116 are limited to four terms (N = 3) within the MC^2-2 code.

J. Ultra-Fine-Group Solution Strategies

Once the resonance, transfer matrix, moderating parameter and cross section data have been processed on an ultra-fine-group mesh the solution of the spectrum equations presented in Section F must be performed. In this section the strategies involved in treating the resonance attenuation, buckling search, flux iterations and fixed source or eigenvalue problems are discussed.

1. Ultra-Fine-Group Spectrum Solution

Given an initial fission source, S_f^g , and a buckling, B^2 , it is possible to solve both the multigroup equations, II.62 or II.64 and the continuous slowing down equations II.66 or II.70 and II.71 by sweeping down the energy mesh from group 1 to the last ultra-fine-group in the problem. Both the sources due to inelastic and/or (n,2n) scattering and the Hydrogen slowing down density are updated after calculating the flux (and current) for a given ultra-fine-group. In the continuous slowing down region, the equations II.66 or II.70 are used to obtain $q_0^{asy}(u_g)$ from the values $q_l(u_{g-1})$. The $q_0^{asy}(u_g)$ so obtained does not feel any effects of narrow resonances in the group g. The attenuation of the asymptotic slowing down density is performed in the manner outlined in Section E. One defines

$$\dot{q}_{o}(u_{g}) = Q_{g}q_{o}^{asy}(u_{g})$$

where the attenuation factor Q_{σ} is defined

$$Q_{g} = \prod_{\substack{r \in g \\ r \in g}} (1 - p_{r})$$

$$p_{r} = \left\{ \frac{\Gamma_{a}^{r}}{\sum_{r}^{g}} J_{a}^{*} - \frac{\Sigma_{ne}^{g}}{\sum_{t}^{g}} \frac{\Gamma_{t}^{r}}{\sum_{r}^{g}} J_{t}^{*} \right\} (\Sigma_{t}^{g} + \frac{B^{2}}{3A_{1}^{g}}) (C_{0} + C_{1})$$

and

$$C_{0} = \frac{\theta}{M_{+}^{g}} + \frac{(1 - \theta)}{M_{-}^{g}}$$

$$C_{1} = 0$$

$$C_{0} = \theta R_{+}^{g} + (1 - \theta) R_{-}^{g}$$

$$C_{1} = -\left[(\theta T_{+}^{g} + (1 - \theta) T_{-}^{g} \frac{q_{1}(u_{r})}{q_{0}(u_{r})}\right]$$

$$C_{1} = -\left[(\theta T_{+}^{g} + (1 - \theta) T_{-}^{g} \frac{q_{1}(u_{r})}{q_{0}(u_{r})}\right]$$

$$C_{1} = -\left[(\theta T_{+}^{g} + (1 - \theta) T_{-}^{g} \frac{q_{1}(u_{r})}{q_{0}(u_{r})}\right]$$

In the absence of resonances the slowing down density $q_{l}(u)$ is assumed to vary linearly between group boundaries,

$$q_{\ell}^{as}(u) = q_{\ell}^{as}(u_{g-1}) + \left[q_{\ell}^{as}(u_{g}) - q_{\ell}^{as}(u_{g-1})\right] \frac{(u - u_{g-1})}{\Delta u}$$

The presence of a resonance r introduces a discontinuity such that

$$q_0(u_{r+}) = (1 - p_r)q_0(u_{r-})$$
.

With these conventions and the Eqs.II.32 and II.36 expressions may be derived for the ultra-fine-group flux and current as well as resonance reaction rates.

If A_x^r is the reaction rate for process x (capture, fission, total) of resonance r, then

 $A_{x}^{r} = \int \Sigma_{x} \phi \, du$

which using the narrow resonance approximation can be written

$$A_{x}^{r} \sim \frac{T_{x}}{E_{r}} J_{x}^{*} (\Sigma_{t}^{g} + \frac{B^{2}}{3A_{1}^{g}}) \phi(u_{r}) . \qquad (II.119)$$

The flux of Eq. II.119 is obtained from Eq.II.32 or II.36 using the attenuated slowing down density $q_o(u_r)$,

$$\begin{split} \tilde{\phi}(u_{r}) &= \begin{cases} \left[\frac{\theta}{M_{+}^{g}} + \frac{(1-\theta)}{M_{-}^{g}}\right] q_{0}(u_{r}) + \frac{(S_{g} + \eta_{0}^{g})}{\Delta u} & \frac{\theta \gamma_{0}(u_{g})}{M_{+}^{g}} + \frac{(1-\theta)\gamma_{0}(u_{g-1})}{M_{-}^{g}} & , \\ & \\ Inconsistent P_{1}/B_{1} & (II.120) \\ \left[\theta R_{+}^{g} + (1-\theta) R_{-}^{g}\right] q_{0}(u_{r}) - \left[\theta T_{+}^{g} + (1-\theta) T_{-}^{g}\right] q_{1}(u_{r}) \\ & + \frac{(S_{g} + \eta_{0}^{g})}{\Delta u} & \left[\theta R_{+}^{g} \gamma_{0}(u_{g}) + (1-\theta) R_{-}^{g} \gamma_{0}(u_{g-1})\right] \\ & - \frac{1.5 \eta_{1}^{g}}{\Delta u} & \left[\theta T_{+}^{g} \gamma_{1}(u_{g}) - (1-\theta) T_{-}^{g} \gamma_{1}(u_{g-1})\right] , \\ & \\ Consistent P_{1}/B_{1} & . \end{cases} \end{split}$$

-51-

The group flux ϕ^g and current J^g are obtained in a similar manner,

$$\phi^{g} = \begin{cases} \left[\frac{\theta}{M_{+}^{g}} + \frac{(1-\theta)}{M_{-}^{g}} \right] & (1-f) \int_{g} q_{0}(u) \ du \\ + (S_{g} + \eta_{0}^{g}) \left[\frac{\theta \gamma_{0}(u_{g})}{M_{+}^{g}} + \frac{(1-\theta) \gamma_{0}(u_{g-1})}{M_{-}^{g}} \right], & \text{Inconsistent } P_{1}/B_{1} \\ \left[\theta R_{+}^{g} + (1-\theta) R_{-}^{g} \right] (1-f) \int_{g} q_{0}(u) \ du \\ & (\text{II.121}) \\ - \left[\theta T_{+}^{g} + (1-\theta) T_{-}^{g} \right] \int_{g} q_{1}(u) \ du \\ & + (S_{g} + \eta_{0}^{g}) \left[\theta R_{+}^{g} \gamma_{0}(u_{g}) + (1-\theta) R_{-}^{g} \gamma_{0}(u_{g-1}) \right] \\ - 1.5 \eta_{1}^{g} \left[\theta T_{+}^{g} \gamma_{1}(u_{g}) + (1-\theta) T_{-}^{g} \gamma_{1}(u_{g-1}) \right], \\ & \text{Consistent } P_{1}/B_{1} \end{cases}$$

$$J^{g} = \begin{bmatrix} \theta V_{+}^{g} + (1 - \theta) & V_{-}^{g} \end{bmatrix} \int_{g} q_{0}(u) du + \begin{bmatrix} \theta U_{+}^{g} + (1 - \theta) & U_{-}^{g} \end{bmatrix} \int_{g} q_{1}(u) du$$
(II.122)
+ $(S^{g} + \eta_{0}^{g}) \begin{bmatrix} \gamma_{0}(u_{g}) & \theta V_{+}^{g} + \gamma_{0}(u_{g-1}) & (1 - \theta) & V_{-}^{g} \end{bmatrix} + 1.5 \eta_{1}^{g} \begin{bmatrix} \gamma_{1}(u_{g}) & \theta U_{+}^{g} + \gamma_{1}(u_{g-1}) & (1 - \theta) & U_{-}^{g} \end{bmatrix}.$

The flux depression factor (1 - f) is defined by use of the narrow resonance approximation,

$$f = \frac{1}{\int_{g} q_{o}(u) \ du} \sum_{r \in g} \frac{\Gamma_{t}}{E} J_{t}^{*} q_{o}(u_{r}) . \qquad (II.123)$$

The thermal flux, $\boldsymbol{\varphi}^{\texttt{th}}$ is given by

(II.124)

$$\phi^{\text{th}} = \frac{(s^{\text{th}} + \eta_o(u_{\text{NG}}) + q_o(u_{\text{NG}}))}{\sum_{a}^{\text{th}} \frac{B^2}{3\Sigma_t^{\text{th}}}}$$

The thermal cross sections are either provided by the user or given by the cross sections of the last ultra-fine-group of the problem (g = NG).

-53-

2. Eigenvalue Solution

By decomposing the source term S^g of the ultra-fine-group spectrum equations, it is possible to write the equations in matrix form as

$$H\Phi = \lambda M\Phi + S_{fir}. \qquad (II.125)$$

If S_{fix} , the inhomogeneous source vector, is zero Eq.II.125 represents an eigenvalue problem with

$$\lambda = \frac{1}{k_{eff}}$$
(II.126)

From Eq.II.14 it is clear that the non-zero elements of the vector $M\Phi$ have the form

$$S_{f}^{g} = \sum_{i} \chi_{i}^{g} \sum_{g'} (\nu \Sigma_{f}^{g'})_{i} \phi^{g'}$$
(II.127)

If the fission spectrum distribution is isotope independent, then one can take

$$S_{f}^{g} = \chi^{g}$$
(II.128)

and

 $k_{eff} = \sum_{g'} \nu \Sigma_{f}^{g'} \phi^{g'}$ (II.129)

so that no flux iterations are required in the spectrum solution. If the fission spectrum distribution is isotope dependent then one assumes a flux, calculates S_f^g from Eq. II.127, solves the spectrum equations as detailed in the previous section, and calculates a new fission source and eigenvalue k_{eff} . This process is repeated until the eigenvalue has converged. Convergence is assumed if on the first iteration

$$\max \left| \frac{s_{f}^{g(1)} - s_{f}^{g(0)}}{s_{f}^{g(0)}} \right| < .001$$
 (II.130)

or on following iterations

$$\frac{k^{(n+1)}-k^{(n)}}{k^n} < .0001 \quad n \ge 1.$$
 (II.131)

3. Inhomogeneous Source Solution

If there are no fissionable isotopes in the problem mixture then Eq. II.125 becomes

$$H\Phi = S_{fix}$$

which may be solved directly by sweeping through the ultra-fine-group equations once in precisely the same manner as that used to solve the eigenvalue equation with isotope independent fission spectra. If both fission and inhomogeneous sources are present a different solution strategy is required. Assuming an isotope independent fission spectrum distribution for the sake of simplicity, Eq. II.25 may be written

$$(I - H^{-1} M)\Phi = H^{-1} S_{fix}$$
 (11.132)

where

$$M = \chi f^{T} = \begin{pmatrix} \chi^{1} \\ \chi^{2} \\ \vdots \end{pmatrix} \quad (\nu \Sigma_{f}^{1}, \nu \Sigma_{f}^{2}, \ldots) \quad .$$
(II.133)

Substituting Eq. II.133 into Eq.II.132 and multiplying through by f^{T} it is easy to obtain

$$f^{T} \Phi = \frac{f^{T} H^{-1} S_{fix}}{I - f^{T} H^{-1} \chi}$$
(II.134)

but

$$\Phi = H^{-1} S_{fix} + H^{-1} \chi(f^{T} \Phi)$$
 (II.135)

or from Eq. II.134

Φ

$$\Phi = H^{-1} S_{fix} + H^{-1} \chi \frac{f^{T} H^{-1} S_{fix}}{I - f^{T} H^{-1} \chi}$$
(II.136)

If the eigenvalue equation is defined

 $\Phi_1 + \frac{\lambda_1 \Phi_0}{1 - \lambda}$

$$H\Phi_{o} = \chi$$

$$f^{T}\Phi_{o} = \lambda_{o}$$
(II.137)

then

=
$$H^{-1} S_{fix} + \frac{\Phi_o}{1 - \lambda_o} f^T H^{-1} S_{fix}$$

(II.138)

or

where

$$H\Phi_{1} = S_{fix}$$

$$\lambda_{1} = f^{T}\Phi_{1}$$
(II.)
Based on these equations a three-step strategy is involved in the solution of the inhomogeneous equations:

- (a) Solve the eigenvalue problem of Eqs. II.137 for Φ_0 and λ_0 as outlined in the previous section iterating on the flux if isotope fission spectra are in problem mixture;
- (b) Solve the inhomogeneous problem of Eqs. II.139 for Φ_1 and λ_1 ;
- (c) Construct the fission source from Eq. II.137 as $\chi f^T \Phi$ and solve Eq.II.135 for the flux Φ .

Step (c) is not strictly necessary since the flux is given by Eq.II.138 but it is required to calculate the resonance reaction rates A_x^r of Eq. II.119 which are used in the broad-group cross-section calculation.

4. Buckling Search to Critical

The spectrum and eigenvalue are clearly dependent upon the buckling B^2 . For eigenvalue problems ($S_{fix} = 0$) with group independent buckling, the code MC²-2 provides an option to search on B^2 to $k_{eff} = 1$. In order to invoke this option of the code, the user provides two starting guesses, B_1^2 , B_2^2 and a convergence criterion ε_{B^2} . The code solves the slowing down broblem and calculates an eigenvalue for each of these values of B^2 . Later estimates of B^2 are based upon linear or parabolic fits to the best available k_{eff} vs. B^2 data. The search procedure ends if



For problems with isotope dependent fission spectra data, the flux iteration discussed above is not converged until the buckling search has been completed. The algorithms invoked in choosing the points for the linear or parabolic fit are based upon those used in the ARC System neutronics search modules ⁽⁴⁾, and depend upon both the sign and magnitude of $1 - k_{eff}$. Several checks are made to insure that a search is not permitted to continue if a pathological situation arises, e.g. search to critical for a highly subcritical (blanket) composition.

K. Group Condensation

The individual material microscopic broad-group cross sections are obtained by averaging the ultra-fine-group data over the flux and current spectra obtained by solving the multigroup and continuous slowing down equations. The broad groups are user specified but must be defined such that broad-group energy boundaries fall on ultra-fine-group boundaries.

The broad group data are written in the ISOTXS⁽²⁾ and/or XS.ISO⁽¹⁾ formats as defined in Appendix C. On user option it is possible to neglect the contributions of resolved and unresolved resonances in the edited cross sections thus making it possible to use MC^2-2 for the generation of finegroup SDX libraries.⁽⁷⁾ The broad-group averaging algorithms are defined below.

1. Broad-Group Inelastic and (n,2n) Cross Sections

The broad-group inelastic and (n,2n) transfer matrices are calculated using the algorithms defined in Section H above. In particular

$$\sigma_{\mathbf{x}_{G' \to G}}^{\mathbf{m}} = \frac{\sum_{\substack{g \in G \ g' \in G'}} \left[s_{\mathbf{x}_{d}}^{g' \to g} + s_{\mathbf{x}_{d}}^{g' \to g} + s_{\mathbf{x}_{d}}^{g' \to g} \right]}{\varphi_{G'}}$$
(II.140)

where

 Ξ inelastic or (n,2n)

∑ g€G

х

sum over ultra-fine-groups g contained in broadgroup G

 $S_{tab}^{g \to g} \equiv tabulated law contribution as given inside g' sum of Eq. II.96$

S^{g ~g} = evaporation law contribution as given inside g' sum ^xevap of Eq. II.97

 $s_{d}^{g' \rightarrow g} \equiv$ discrete law contribution as given inside g' sum of equation following Eq. II.104 using either rigorous or approximate P₁

and the broad-group flux is given by

$$\phi^{\mathbf{G}'} \equiv \sum_{\mathbf{g}' \in \mathbf{G}'} \phi^{\mathbf{g}'} . \tag{II.141}$$

The total inelastic and (n,2n) cross sections are obtained by summing over all sink groups

$$\sigma_{\mathbf{x}_{\mathbf{G}'}}^{\mathbf{m}} = \sum_{\mathbf{G}} \sigma_{\mathbf{x}}^{\mathbf{m}}(\mathbf{G}' \to \mathbf{G}) \quad . \tag{II-142}$$

Since the microscopic ultra-fine-group data required by Eq.II.140 are not saved during the course of the MC^2-2 spectrum calculation, the data required for the broad-group condensation are calculated specifically for the cross section edits. It is this calculation which controls both the timing and space requirements of the MC^2-2 condensation calculation. As the number of broad-groups increases beyond approximately 100, I/O times become quite large because of the data management requirements for calculation of Eq.II.140.

2. Broad-Group Absorption Cross Sections

The MC^2-2 library file MCC2F5 (Appendix C) contains ultra-finegroup cross section data which are averaged over the flux spectrum as

$$\sigma_{\mathbf{x}_{G}}^{\mathbf{m}} = \sum_{\mathbf{g}\in G} \sigma_{\mathbf{x}_{g}}^{\mathbf{m}} \phi^{\mathbf{g}} / \phi^{\mathbf{G}}$$
(II.143)

where x includes the six reactions (n,H^1) , (n,H^2) , (n,H^3) , (n,f), (n,γ) and (n,α) . The (n,α) cross section calculated from Eq.II.143 is actually a sum of (n,He^4) and (n,He^3) cross sections. This summing was performed since the ISOTXS format does not allow editing of both the partials. Since the σ_x^m data from the file MCC2F5 do not include either resolved or unresolved g resonance contributions, the fission and radiative capture cross sections of Eq.II.143 are not complete. The contributions from unresolved resonances are obtained using Eq.II.143 with the ultra-fine-group (n,γ) and (n,f) cross sections given by Eq.IV.68. The contribution of resolved resonances to the broad-group radiative capture and fission cross sections is derived from the resonance reaction rates, A_x^r , defined by Eq.II.119. Using this equation it

is clear that the resolved resonance contribution to the broad group cross section is given by

$$\frac{\text{Resolved Resonance Reaction Rate in Broad-Group G}}{\text{Group G Flux}} = \frac{1}{\underset{\substack{N \ \phi}{}^{G}}{\underset{m}{}^{F \in G}}} \sum_{\substack{r \in G \\ r \in M}} A_{x}^{r} \qquad (\text{II.144})$$

where the sum is over all resolved resonances of material m which fall into group G,

 $E_g < E_r < E_{g-1}$.

Thus the broad-group (n, γ) and (n, f) cross sections for a material with both resolved and unresolved data would be given by

$$\sigma_{x_{G}}^{m} = \left\{ \sum_{g \in G} (\sigma_{x_{g}}^{m} + \overline{\sigma}_{x_{g}}^{m}) \phi^{g} + \frac{1}{N_{m}} \sum_{r} A_{x}^{r} \right\} / \phi^{G}$$
(II.145)

where

= floor cross sections from MCC2F5

Ξ

σ^m x

unresolved cross sections from Eq.IV.68

resolved resonance reaction rate from Eq.II.119.

3. Elastic Scattering Cross Sections

In Section G of this chapter the calculation of ultra-tine-group elastic transfer matrices was discussed. The ultra-fine-group cross sections derived from that calculation $\sigma_s^f(g' \rightarrow g)$ include contributions from the ultra-fine-group library (MCC2F5) and the unresolved resonances from Chapter IV. The group condensation of these data are accomplished as

$$\sigma_{e\ell}^{o_{m}^{m}}(G' \rightarrow G) = \frac{\sum_{g' \in G'} \sum_{g \in G} \sigma_{e\ell}^{o_{m}^{m}}(g' \rightarrow g)\phi^{g'}}{\phi^{G'}}$$
(II.146)

$$\sigma_{e\ell}^{l^{m}}(G' \rightarrow G) = \frac{\sum_{g' \in G'} \sum_{g \in G} \sigma_{e\ell}^{l^{m}}(g' \rightarrow g) J^{g'}}{T_{G'}^{G'}} . \qquad (II.147)$$

The P_1 calculation is performed only for the consistent P_N or consistent B_{N} spectrum options in which case the ultra-fine-group current J^{g} is calculated directly using Eqs.II.62 and II.122 and the broad-group current is given by

> $J^{G} = \sum_{g \in G} J^{g}$. (II.148)

The contributions of resolved resonances to the elastic transfer matrix cannot be calculated by use of Eqs.II. 146-II. 147 since ultra-finegroup resolved resonance scattering cross sections are not available. approximate method adapted from the work of $Stacey^{(39)}$ which accounts for the position of the resonance in the group and the shape of the resonance is used in lieu of requiring ultra-fine-group resonance cross sections. The method adopted assumes that resolved resonance materials can scatter only one broad-group. Since the current MC^2-2 library described in Table VIII has resolved data only for materials of mass greater than one hundred, this assumption fixes the smallest broad-group lethargy width at .04 (0.5)ultra-fine-groups). Following the methods used in calculating the resolved absorption cross sections we take

$$\sigma_{\text{res}}^{o^{\text{m}}}(G \rightarrow G') = \frac{1}{N_{\text{m}}\phi^{\text{G}}} \sum_{\substack{\mathbf{r} \in G \\ \mathbf{r} \in \mathbf{m}}} A_{\text{s}}^{\mathbf{r}} P(\mathbf{r} \rightarrow G') \qquad (\text{II.149})$$

where

Ar

resolved resonance r elastic reaction rate

$P(r \rightarrow G')$ probability of scattering a neutron by resonance r into Group G'.

The resonance reaction rate is calculated directly from the resonance integrals discussed in Chapter III,

$$A_{s}^{r} = \frac{1}{E_{r}} \left\{ \Gamma_{t} J_{t}^{*} - \Gamma_{\gamma} J_{\gamma}^{*} - \Gamma_{f} J_{f}^{*} \right\} \left(\Sigma_{t}^{g} + \frac{B^{2}}{3A_{1}^{g}} \right) \stackrel{\sim}{\phi} \left(u_{r} \right)$$
(II.150)

which is the scattering equivalent of Eq.II.119. The resonance integrals of Eq.II.150 take account of overlap, self-shielding, and Doppler broadening effects as described in Chapter III. The removal probability is taken as

$$P(r \rightarrow G + 1) \sim \frac{\int_{\varepsilon}^{u} \int_{\varepsilon}^{u} du K(u, u_{G}) \frac{\sigma_{s}^{r}(u)}{\Sigma_{t}(u)}}{\int_{u}^{u} \int_{\varepsilon}^{u} \frac{du \frac{\sigma_{s}^{r}(u)}{\Sigma_{t}(u)}}{\sum_{t}^{u} \int_{\varepsilon}^{u} \frac{\sigma_{s}^{r}(u)}{\Sigma_{t}(u)}}$$

where

$$K(u,u_G) \equiv \frac{e^{-(u_G^{-u})} - \alpha}{1 - \alpha}$$

and ε is the larger of u_{G-1} and $u_G - \ln \frac{1}{\alpha}$.

It is further assumed that the resonance is isolated and that natural line shapes $^{(26)}$ are applicable so that

$$\frac{\sigma^{r}}{\sum_{t}} \sim \frac{\psi + a\chi}{\beta + \psi + a\chi}$$

 $\Delta = \frac{2E_{r}}{\Gamma}$

$$\psi(\mathbf{u}) \stackrel{\sim}{\sim} \frac{1}{1 + \Delta^2 [e^{\mathbf{u}\mathbf{r}-\mathbf{u}} - 1]^2}$$

$$\chi(\mathbf{u}) \stackrel{\sim}{\sim} \frac{\Delta[\mathrm{e}^{\mathbf{u}\mathbf{r}-\mathbf{u}}-1]}{1+\Delta^2[\mathrm{e}^{\mathbf{u}\mathbf{r}-\mathbf{u}}-1]^2}$$

(II.152)

(11.151)

The terms are defined in detail in Chapter III. The above assumptions make it possible to integrate Eq.II.151 analytically. The integrals are reported in Eqs. 12, 13, 19, 20, 23 of Reference 39. It is important to note that the above approximations are used only to distribute the resolved elastic reaction rate between in-group and removal and have no impact on the ultra-finegroup spectrum or the far more general resonance integral calculations of Chapters III and IV. The in-group probability is given by

$$P(r \rightarrow G) = 1 - P(r \rightarrow G + 1)$$

so that the elastic transfer matrix may be calculated from Eqs.II.149-II.151. The P_1 transfer matrix is calculated in the consistent spectrum options as

$$\sigma_{res}^{l^{m}}(G \rightarrow G') = \frac{2}{3A^{m}N_{m}J^{G}} \sum_{r} \frac{J(u_{r})}{\sqrt[n]{u_{r}}} A_{s}^{r} P(r \rightarrow G') \qquad (II.153)$$

where $J(u_r)$ is the current at lethargy u_r in direct analogy with Eq.II.120. The total elastic scattering matrix is given by the sums of II.146 and II.149 or for the P₁ matrix, II.147 and II.153. The elastic cross section is obtained by summing over sink groups according to Eq.II.142.

The Hydrogen elastic scattering cross sections are collapsed by assuming that scattering is isotropic in the center of mass system. With this assumption it is not difficult to show

$$\sigma_{el}^{\circ}(G \rightarrow G') = \frac{(E_{G'-1} - E_{G'})}{\phi^{G}} \frac{(e^{\Delta u} - 1)}{\Delta u} \sum_{g \in G} \frac{\sigma_{s}^{g} \phi^{g}}{E_{g-1}}$$
(II.154)

$$\sigma_{el}^{1H}(G \to G') = \frac{4}{9} \frac{(E_{G'-1}^{3/2} - E_{G'}^{3/2})}{J^{G}} \frac{(e^{3/2\Delta u} - 1)}{\Delta u} \sum_{g \in G} \frac{\sigma_{s_{H}}^{g}}{E_{g-1}^{3/2}} .$$
 (II.155)

Eq.II.155 is used only for the consistent spectrum options. The corrections required to Eqs.II.154-II.155 to treat in-group scattering are straightforward.

4. Total Cross Section

The flux weighted total cross section is obtained by summing all of the partials discussed above,

$$\sigma_{t_{G}}^{o^{m}} = \sigma_{inel_{G}}^{m} + \sigma_{n,2n_{G}}^{m} + \sigma_{el_{G}}^{o^{m}} + \sigma_{f_{G}}^{m} + \sigma_{nY_{G}}^{m} + \sigma_{nH^{1}_{G}}^{m} + \sigma_{nH^{2}_{G}}^{m} + \sigma_{nH^{3}_{G}}^{m} + \sigma_{nA_{G}}^{m} + \sigma_{nA_{G}}^{m}$$
(II.156)

and the broad-group partial cross sections include floor, unresolved and resolved resonance contributions as discussed above. A current weighted total cross section is calculated as

$$\sigma_{t_{G}}^{1^{m}} = \frac{1}{J^{G}} \sum_{g \in G} \gamma_{g} \sigma_{t_{g}}^{m} J^{g} + \frac{1}{N_{m} \phi^{G}} \sum_{\substack{r \in G \\ r \in m}} \frac{\Gamma_{t}}{E_{r}} J_{t}^{*} (\Sigma_{t}^{g} + \frac{B^{2}}{3A_{1}^{g}}) \phi^{(u_{r})}$$
(II.157)

where

Υg

- Jg = ultra-fine-group current calculated from Eqs. II.62 and II.122 for consistent options and Eq.II.64 for inconsistent options.
- JG. = broad group current from Eq.II.148.
- σ^{m} Ξ ultra-fine-group total floor cross section from library tg file MCC2F5 plus ultra-fine-group total unresolved cross sections from Eq. IV.68.

Ξ spectrum coefficient

$$\left\{\begin{array}{cccc}
\frac{B}{\Sigma_{t}^{g}} \tan^{-1} \frac{B}{\Sigma_{t}^{g}} & & \\
\frac{\Sigma_{t}^{g}}{\Sigma_{t}^{g}} & & \\
\frac{B}{\Sigma_{t}^{g}} & & \\
\frac{\Sigma_{t}^{g}}{\Sigma_{t}^{g}} & & \\
\frac{B}{\Sigma_{t}^{g}} & & \\
\frac{B}{\Sigma_{t}^{g}$$

and the resolved resonance contribution is simply the reaction rate integral used with the flux weighted cross section of Eq.II.156.

For spectrum calculations using the extended transport approximation, higher order flux moments ϕ_0^g are calculated recursively from Eq.II.62. Total cross section moments equivalent to Eq.II.157 are then calculated,

$$\sigma_{t_{G}}^{\ell^{m}} = \frac{1}{\phi_{\ell}^{G}} \sum_{g \in G} \gamma_{g} \sigma_{t_{g}}^{m} \phi_{\ell}^{g} + \sigma_{t_{res_{G}}}^{o^{m}} \qquad \ell = 2, 3, \dots, N \quad (II.158)$$

where

$$\gamma_{g} = \begin{cases} \frac{N+1}{2N+1} & \frac{iB}{\Sigma_{t}^{g}} & \frac{Q_{N+1}(-\Sigma_{t}/iB)}{Q_{N}(-\Sigma_{t}/iB)} & \delta_{\ell}^{N} & \text{for } B_{N} \\ 1 & & & & \\ 1 & & & & \\ 1 & & & & \\ \end{cases}$$

and all other terms have been defined.

Ν

5. Transport Cross Section

In analogy with the total cross section, the MC^2-2 code calculates moments of the transport cross section if the order of the extended transport approximation is greater than unity. The transport cross section algorithms are spectrum option dependent. In all cases, however, the algorithms are defined to conserve the relation

$$\Sigma_{\text{tr}_{G}}^{\ell} = \frac{-\ell}{2\ell+1} \text{ iB } \frac{\phi_{\ell-1}^{G}}{\phi_{0}^{G}} \qquad \ell = 1, 2, \dots, N \qquad (\text{II.159})$$

or for the special case of $\ell = 1$

$$J = -\frac{B}{3\Sigma_{tr}} \phi . \qquad (II.159')$$

The algorithms further conserve the relation .

$$\Sigma_{tr} = \sum_{i}^{N} N_{i} \sigma_{tr_{i}}$$

a. Inconsistent P₁ or B₁ Spectrum

$$\sigma_{tr_{g}}^{\ell^{m}} = \sigma_{t_{G}}^{\ell^{m}} - \sigma_{e\ell_{G}}^{\ell^{m}} \qquad \ell = 1, 2, \dots, N \qquad (II.160)$$

where

 $\sigma_{t_{C}}^{\ell^{m}} = P_{\ell}$ moment of total cross section given by Eq.II.157 or II.158.

 $\sigma_{el_{c}}^{l} = P_{l}$ moment of elastic scattering cross section.

The elastic scattering cross section moments are calculated from Eq.II.45 and II.150 and it is assumed that resolved resonances do not contribute to the cross sections for moments greater than unity,

$$\sigma_{e\ell_{G}}^{\ell_{m}} = \left\{ \sum_{g \in G} \sum_{n} \sigma_{e\ell}^{g} f_{n}^{g} T_{\ell n}^{o}(\alpha_{m}) \phi_{\ell} + \frac{2\delta_{\ell}^{1}}{3A^{m}N_{m}} \sum_{\substack{r \in G \\ r \in m}} A_{s}^{r} \frac{J(u_{r})}{\phi(u_{r})} \right\} / \phi_{\ell}^{G}, \qquad \ell = 1, 2, ..., N$$
(II.161)

b. Consistent P₁ or B₁ Spectrum

The Eqs.II.160 and II.161 are used in the consistent spectrum options as well as the inconsistent options for moments greater than unity (l > 1). The P₁ moment is however calculated directly from the elastic transfer matrices in direct analogy with Eqs.II.147 and II.153. In particular

$$\sigma_{tr_{G}}^{1^{m}} = \sigma_{t_{G}}^{1^{m}} - \frac{1}{J^{G}} \sum_{g \in G} \sum_{g' \leq g} \sigma_{e\ell}^{1^{m}} (g' \rightarrow g) J^{g'}$$
$$- \frac{1}{J^{G}} \frac{2}{3A^{m}N_{m}} \sum_{r} \frac{J(u_{r})}{\partial(u_{r})} A_{s}^{r} P(r \rightarrow G) \quad . \tag{II.162}$$

6. Average Number of Neutrons per Fission

The average number of neutrons per fission is given by

$$v_{G}^{m} = \frac{\left[\sum_{g \in G} v_{m}^{g} \sigma_{f}^{m} \phi^{g} + \frac{1}{N_{m}} \sum_{r \in G} v_{m}^{g} \sigma_{f}^{r}\right]}{\sigma_{f_{G}}^{m} \phi^{G}}$$

(II.163)

where

νg

m

= number of neutrons per fission for material m in ultrafine-group g from Eq.II.110.

 $\sigma_{f_g}^{m} \equiv$ floor plus unresolved ultra-fine-group fission cross g section.

 $\sigma_{f_G}^m \equiv broad-group fission cross section.$

 A_{f}^{r} = resolved resonance fission reaction rate from Eq.II.119.

7. Fission Spectrum Distribution

The isotopic fission spectrum distributions are derived by summing the ultra-fine-group vectors of Eq.II.112,

$$\chi_{\rm m}^{\rm G} = \sum_{\rm g \in G} \hat{\chi}_{\rm m}^{\rm g} \quad . \tag{II.164}$$

The MC^2-2 code edits a set-wide fission spectrum vector as well as the isotopic vectors of Eq.II.164. The set-wide vector is defined

$$\chi_{\text{set}}^{\text{G}} = \frac{\sum_{\substack{g \in G \ m}} \hat{\chi}_{m}^{g} s_{f_{m}}}{\sum_{\substack{m \ m}} s_{f_{m}}}$$

where $\hat{\chi}_{m}^{g}$ are the ultra-fine-group vectors of Eq.II.112 and S_f are the fission sources for material m,

$$S_{f_m} = \sum_{\sigma} N_m v_m^g \sigma_f^m \phi^g + \sum_{r \in m} v_m^{g_r} A_f^r . \qquad (II.166)$$

8. XS.ISO Cross Sections

The ARC System⁽¹⁾ cross section file XS.ISO (c.f. Appendix B) is less general than the CCCC file ISOTXS and contains derived cross sections in addition to the standard cross sections given above. The following conversions for XS.ISO cross sections are consistent with the MC^2 derived cross sections of Reference 10:

$$\begin{bmatrix} \sigma_{tr_{G}}^{m} \end{bmatrix}_{ARC} = \sigma_{tr_{G}}^{1^{m}}$$
(II.167)

$$\left[\sigma_{el}^{m} (G \rightarrow G) \right]_{ARC} = \sigma_{tr_{G}}^{1m} - \sigma_{t_{G}}^{0m} + \sigma_{el}^{0m} (G \rightarrow G)$$
 (III.168)

$$\left[\sigma_{el}^{anis}(G \to G)\right]_{ARC} = \sigma_{t_{G}}^{l_{m}} - \sigma_{t_{G}}^{0_{m}} + \sigma_{el}^{0_{m}}(G \to G) \quad . \tag{III.169}$$

Equations II.168 and II.169 are consistent with Equations 116-118 of Ref. 10.

L. Broad Group Spectrum

The microscopic broad-group cross section data calculated according to the algorithms of Section K are appropriate for use in general multigroup neutronics calculations. One such calculation, extended transport P_1 real and adjoint fundamental mode spectrum, is available in MC²-2 at user option. The module (overlay) CSCO10 solves the following broad-group flux equations

(11.165)

$$\begin{split} \Sigma_{\mathbf{r}_{0}}^{G} \phi_{0}^{G} &= B\phi_{1}^{G} = Q^{G} + \sum_{G' < G} \Sigma_{s}^{0} (G' \rightarrow G) \phi_{0}^{G'} \\ &\frac{B}{3} \phi_{0}^{G} + \Sigma_{\mathbf{r}_{1}}^{G} \phi_{1}^{G} + \frac{2}{3} B\phi_{2}^{G} = \sum_{G' < G} \Sigma_{s}^{1} (G' \rightarrow G) \phi_{1}^{G'} \\ &- (\mathbf{1})^{2\ell} \frac{\ell}{2\ell + 1} B\phi_{\ell-1}^{G} + \Sigma_{t\mathbf{r}_{G}}^{\ell} \phi_{\ell}^{G} - (\mathbf{1})^{2\ell} \frac{\ell + 1}{2\ell + 1} B\phi_{\ell+1}^{G} = 0 \qquad \ell = 2, \dots, N \\ &\phi_{N+1}^{G} = 0 \end{split}$$

where

$$\begin{split} \Sigma_{\mathbf{s}}^{\mathbf{o}}(\mathbf{G}' \rightarrow \mathbf{G}) &= \sum_{\mathbf{m}} N_{\mathbf{m}} \left[\sigma_{\mathbf{e}\ell}^{\mathbf{o}\mathbf{m}}(\mathbf{G}' \rightarrow \mathbf{G}) + \sigma_{\mathbf{i}\mathbf{n}\mathbf{e}\ell}^{\mathbf{m}}(\mathbf{G}' \rightarrow \mathbf{G}) + 2 \sigma_{\mathbf{n},2\mathbf{n}}^{\mathbf{m}}(\mathbf{G}' \rightarrow \mathbf{G}) \right] \\ \Sigma_{\mathbf{s}}^{1}(\mathbf{G}' \rightarrow \mathbf{G}) &= \sum_{\mathbf{m}} N_{\mathbf{m}} \sigma_{\mathbf{e}\ell}^{1\mathbf{m}}(\mathbf{G}' \rightarrow \mathbf{G}) \\ \Sigma_{\mathbf{tr}_{\mathbf{G}}}^{\ell} &= \sum_{\mathbf{m}} N_{\mathbf{m}} \sigma_{\mathbf{tr}_{\mathbf{G}}}^{\ell} \\ \Sigma_{\mathbf{r}_{\mathbf{G}}}^{\mathbf{G}} &= \sum_{\mathbf{m}} N_{\mathbf{m}} \left[\sigma_{\mathbf{t}_{\mathbf{G}}}^{\mathbf{o}\mathbf{m}} - \sigma_{\mathbf{e}\ell}^{\mathbf{o}\mathbf{m}}(\mathbf{G} \rightarrow \mathbf{G}) - \sigma_{\mathbf{i}\mathbf{n}\mathbf{e}\ell}^{\mathbf{m}}(\mathbf{G} \rightarrow \mathbf{G}) - 2\sigma_{\mathbf{n},2\mathbf{n}}^{\mathbf{m}}(\mathbf{G} \rightarrow \mathbf{G}) \right] \\ \Sigma_{\mathbf{r}_{\mathbf{1}}}^{\mathbf{G}} &= \sum_{\mathbf{m}} N_{\mathbf{m}} \left[\sigma_{\mathbf{t}_{\mathbf{G}}}^{1\mathbf{m}} - \sigma_{\mathbf{e}\ell}^{1\mathbf{m}}(\mathbf{G} \rightarrow \mathbf{G}) \right] \\ Q^{\mathbf{G}} &= \frac{1}{K} \chi_{\mathbf{set}}^{\mathbf{G}} \sum_{\mathbf{m}} N_{\mathbf{m}} \sum_{\mathbf{G}'} v_{\mathbf{G}'}^{\mathbf{m}} \sigma_{\mathbf{f}_{\mathbf{G}}}^{\mathbf{m}} \phi_{\mathbf{o}}^{\mathbf{G}'} + \sum_{\mathbf{g}\in\mathbf{G}} s_{\mathbf{f}\mathbf{i}\mathbf{x}}^{\mathbf{g}} . \end{split}$$

The inconsistent spectrum option is defined by setting $\Sigma_s^1(G' \rightarrow G)$ to zero and taking

$$\Sigma_{r_1}^G = \Sigma_{tr_G}^1$$

If there are no inhomogeneous sources, $S_{fix} = 0$, then Equations II.170 specify an eigenvalue problem. In this case the code will search on B^2 to K = 1 according to the procedures noted in Section J.4 and reference 4. The first buckling used will be the critical B^2 from the ultra-fine-group calculation and the second guess increases (decreases) B^2 by 1% if the eigenvalue of the ultra-fine-group calculation is greater (less) than 1.0. The buckling search is stopped when

$$|K - 1| \leq 10^{-4}$$
.

The adjoint eigenvalue problem is also solved by MC^2-2 ,

$$\Sigma_{r_{o}}^{G} \phi_{o}^{G^{*}} - B \phi_{1}^{G^{*}} = Q^{G^{*}} + \sum_{G'>G} \Sigma_{s}^{o} (G \rightarrow G') \phi_{o}^{G'^{*}}$$

$$\frac{B}{3} \phi_{o}^{G^{*}} + \Sigma_{r_{1}}^{G} \phi_{1}^{G^{*}} + \frac{2}{3} B \phi_{2}^{G^{*}} = \sum_{G'>G} \Sigma_{s}^{1} (G \rightarrow G') \phi_{1}^{G'^{*}}$$

$$-(1)^{2\ell} \frac{\ell}{2\ell+1} B \phi_{\ell-1}^{G^{*}} + \Sigma_{tr_{G}}^{\ell} \phi_{\ell}^{G^{*}} - (1)^{2\ell} \frac{\ell+1}{2\ell+1} B \phi_{\ell+1}^{G^{*}} = 0 \quad \ell = 2, ..., N$$

$$\phi_{N+1}^{G^{n}} = 0$$

with

$$q^{G^*} = \frac{1}{K} (v \Sigma_f)_G \sum_{G'} \chi_{set}^{G'} \phi_{o}^{G'^*}$$

The solution of the tridiagonal equations II.170 and II.171 is performed using the forward elimination-backward substitution technique⁽⁴¹⁾, a specialization of the standard Gauss reduction. Since the flux equations II.170 are obtained by summing the ultra-fine-group equations II.59 and using the flux weighted broad-group cross section definitions of Section K, the fluxes, currents and eigenvalue calculated by the module CSCO10 should agree with those obtained from the ultra-fine-group-spectrum solution. In practice the differences in resonance treatments, slowing down formulations and computational precision limit this agreement. Eigenvalue agreement tends always to be better than 0.1% and spectrum agreement is excellent down to the resolved resonance energy domain where the ultra-fine-group spectrum tends to attenuate faster than the broad group spectrum. Because flux weighting is used to collapse the cross section data, it is clear that the broad-group adjoint spectrum of Eq.II.171 is not equivalent to the adjoint of the ultra-fine-group equations.

III. CALCULATION OF RESOLVED RESONANCE INTEGRALS

A. General Formulation

The algorithms involved in the resolved resonance integral calculation developed by R. Hwang⁽²⁵⁾ assume the narrow resonance approximation and account for interference scattering and the effects of overlap with neighboring resolved resonances. Both single level Breit-Wigner and multilevel Breit-Wigner or Adler-Adler representations are accommodated including interaction of single and multilevel resonances.

The effective macroscopic capture cross section over the lethargy range u_1 , u_2 of width Δu due to resonances i for an isotope having atom density N_i can be written as

 $\bar{\Sigma}_{c} = N_{i}\bar{\sigma}_{c} = \frac{\sum_{i}^{u_{2}} \frac{N_{i}\sigma_{c_{i}}(u) F(u) du}{\sum_{t}(u)}}{\int_{u_{1}}^{u_{2}} \frac{F(u) du}{\sum_{t}(u)}}$

(III.1)[.]

where F(u) is the collision density, σ_{i} is the microscopic capture cross section due to resonance i, and Σ_{t} is the total macroscopic cross section for the mixture. If we consider the case of Breit-Wigner single level resonances, extend the range of integration to cover all lethargies, replace du by - dE/E, assume E remains constant at the resonance energy $E_{o_{i}}$, and

assume a constant collision density over each resonance, we may rewrite Eq.III.1 as



$$= \sum_{i} F_{i} \frac{\sum_{p} \Gamma_{\gamma_{i}}}{\Delta u E_{o_{i}} f} \cdot \frac{1}{2} \int_{-\infty}^{\infty} \frac{\psi_{i} dx_{i}}{\beta_{i} + \psi_{i} + a_{i} \chi_{i} + \sum_{i' \neq i} (A_{i'} \psi_{i'} + B_{i'} \chi_{i'})}{\beta_{i} + \psi_{i} + a_{i} \chi_{i} + \sum_{i' \neq i} (A_{i'} \psi_{i'} + B_{i'} \chi_{i'})}$$

$$= \sum_{i} F_{i} \frac{\sum_{o_{i}} \Gamma_{\gamma_{i}}}{\Delta u E_{o_{i}} f} J_{x_{i}}^{*\delta \ell}.$$
(III.2)

where $\Sigma_{t}(\mathbf{u}) = \Sigma_{p} + N_{i}\sigma_{o}\psi(\theta_{i},\mathbf{x}_{i}) + N_{i}\sigma_{o}a_{i}\chi(\theta_{i},\mathbf{x}_{i}) + \sum_{i'\neq i}N_{i'}\sigma_{o}\psi(\theta_{i'},\mathbf{x}_{i'})$

 F_i is the constant collision density for resonance i and f is the so-called flux correction factor which can be written as

$$\mathbf{f} = \sum_{i} \mathbf{F}_{i} - \frac{1}{\Delta u} \sum_{j} \mathbf{F}_{j} \frac{\mathbf{T}_{i}}{\mathbf{E}_{o_{j}}} \mathbf{J}_{i}^{* \delta \ell}$$

The sum in f extends over all resonances in the mixture, and $J_{t_j}^{*s\ell}$ is the total resonance integral defined later in Eq.III.24. The $s\ell$ superscript refers to single level and Eq.III.2 defines the resonance integral $J_{x_i}^{*s\ell}$. Eq.III.2 is the starting point in defining the ultra-fine-group reaction rates and fluxes as given in Chapter II Equations II.119, II.121-II.123. The weighting functions F, are calculated from the slowing down density attenuated by higher energy resonances.

In deriving Eq.III.2, we have used the approximation that $E = E_{o}$ so that

$$\sigma_{c_{i}} = \sigma_{o_{i}} \frac{\Gamma_{i}}{\Gamma_{i}} \psi(\theta_{i}, x_{i}) \equiv \sigma_{o_{i}} \frac{\Gamma_{i}}{\Gamma_{i}} \psi_{i}$$
(III.3)

with a similar expression for σ_{f_i} where Γ_{f_i} replaces Γ_{γ_i} , and

$$\sigma_{t_{i}} = \sigma_{p_{i}} + \sigma_{o_{i}} \psi_{i} + \sigma_{o_{i}} a_{i} \chi(\theta_{i}, x_{i}) = \sigma_{p_{i}} + \sigma_{o_{i}} \psi_{i} + \sigma_{o_{i}} a_{i} \chi_{i}$$
(III.4)

where

$$\sigma_{\mathbf{o}_{\mathbf{i}}} = \text{the peak cross section of the resonance}$$
$$= \frac{2.6039953 \times 10^{6}}{\left| \frac{\mathsf{E}_{\mathbf{o}_{\mathbf{i}}}}{| \mathbf{e}_{\mathbf{o}_{\mathbf{i}}}} \right|} \left[\frac{\mathsf{A}_{\mathbf{i}} + 1}{\mathsf{A}_{\mathbf{i}}} \right]^{2} \frac{\Gamma_{\mathbf{n}}}{\mathsf{g}_{\mathbf{j}_{\mathbf{i}}} \frac{\Gamma_{\mathbf{i}}}{\Gamma_{\mathbf{i}}}}$$

A₁ = the mass number of the nucleus having resonance i

-68-

 $=\frac{2J+1}{2(2I+1)}$ where J is the total spin of the compound ^gji nucleus and I is the spin of the target nucleus Γ_{Υι} = the radiative capture line width $\Gamma_{\mathbf{i}}$ = the total line width Γ_{no}i = the neutron width evaluated at the resonance energy $\psi(\theta_i, x_i) \equiv \psi_i$ = the Doppler broadened symmetric line shape function given in Eq.A.1 of Appendix A and calculated as described there θ = the ratio of natural width to Doppler width Γ_i / Δ_i evaluated at the resonance energy $\Delta_{i} = \begin{bmatrix} \frac{4kT_{i}E_{o_{i}}}{A_{i}} \end{bmatrix}^{\frac{1}{2}}$ with Boltzmann's constant = 8.6168 x 10^{-5} eV/degree Kelvin k ' = temperature of isotope having resonance i in T, degrees Kelvin $= 2(E - E_{o_{i}})/\Gamma_{i}$ x_i $\mathbf{a}_{\mathbf{i}} = \begin{bmatrix} \mathbf{g}_{\mathbf{j}} & \mathbf{\sigma}_{\mathbf{j}} \\ \mathbf{i} & \mathbf{\sigma}_{\mathbf{j}} \\ \mathbf{i} \\ \mathbf{r}_{\mathbf{j}} \\ \mathbf{r}_{\mathbf{j}} \\ \mathbf{\sigma}_{\mathbf{j}} \end{bmatrix}^{\frac{1}{2}}$ σ_{p</sup>ai} = the atom potential scattering cross section $\chi(\theta_i, x_i) \equiv \chi_i$ = the Doppler broadened antisymmetric line shape function given in Eq. A.2 of Appendix A and calculated as described there .

-69-

The summation in the denominator of Eq. III.2 is over all other resonances $i' \neq i$ in the mixture and



(111.5)

The code computes resonance integrals for a homogeneous mixture, and for heterogeneous slab geometries or cylindrical geometries. Whether the system is homogeneous or heterogeneous, and whether slab or cylinder determines the value used for Σ_{p} and hence for β by use of an equivalence principle.

For homogeneous mixtures, Σ_n is given by

$$\Sigma_{p} = \sum_{m} N_{m} \sigma_{t} + \frac{\kappa^{2}}{3 \sum_{m} N_{m} \sigma_{t}}$$
(III.6)

where the summation is over all isotopes in the homogeneous mixture, N_m is the atom density of isotope m, σ_t is the smooth total cross section library m_m

data for the ultra fine group into which the resonance in question falls, and κ^2 is the user supplied extended-transport approximation buckling as used in the continuous slowing down calculation.

For slab geometries, for each slab k and for the material i whose resonance integral is being calculated

$$\Sigma_{p_{\mathbf{i}}}^{\mathbf{k}} = \sum_{\mathbf{m} \in \mathbf{k}} N_{\mathbf{m}}^{\mathbf{k}} \sigma_{\mathbf{t}_{\mathbf{m}}} + \frac{\kappa^{2}}{3 \sum_{\mathbf{m}} N_{\mathbf{m}} \sigma_{\mathbf{t}_{\mathbf{m}}}} + \Sigma_{e_{\mathbf{i}}}^{*\mathbf{k}}$$

where the first summation is over all isotopes m contained in slab region k and the second summation as in Eq. III.6 is over all isotopes in the equivalent homogeneous mixture obtained by homogenizing all slab regions. The so-called "escape" cross section (for the material i whose resonance integral is being calculated) intended to account for the heterogeneity is calculated using

$$\Sigma_{e_{i}}^{*k} = \Sigma_{e}^{k} \frac{a_{1}(1 - C_{i}^{k})}{1 + (a_{2} - 1)C_{i}^{k}}$$
(III.8)

(III.7)

where

$$\Sigma_{e}^{k} = \frac{s^{k}}{4v^{k}}$$
(III.9)

with S^k and V^k the surface area and volume of region k, a_1 and a_2 are user supplied with default values of 1.09, and the Dancoff factor C_i^k is given by

$$C_{i}^{k} = E_{3}(\tau_{L_{i}}^{k}) + E_{3}(\tau_{R_{i}}^{k}).$$
(III.10)

In Eq.III.10, E_3 is the exponential integral of order three defined in Eq.A.37 of Appendix A. The optical thicknesses to the left and right of slab region k for material i denoted by, respectively, $\tau_{L_i}^k$ and $\tau_{R_i}^k$, are computed as follows,

$$\tau_{L_{i}}^{k} = \sum_{\substack{j \neq i \ r \ to \ the}} \sum_{\substack{n_{j} \neq i \ r \ to \ the}} N_{j}^{r} \Delta x_{r}^{\sigma} t_{j}$$
(III.11)
$$\tau_{R_{i}}^{k} = \sum_{\substack{j \neq i \ r \ to \ the}} \sum_{\substack{n_{j} \neq i \ r \ to \ the}} N_{j}^{r} \Delta x_{r}^{\sigma} t_{j}$$
(III.12)

where $\Delta {\bf x}_{\bf r}$ is the thickness of region r. The summation over r continues until a region s is found such that

$$\frac{\frac{N_{i}^{S}\Delta x_{s}}{N_{i}^{k}\Delta x_{k}} > \varepsilon_{het}.$$

The criterion ε_{het} is a user input parameter and has the default value of 0.1.

For cylindrical geometry, the resonance integrals are evaluated only for the central region. All outer regions are homogenized into a single outer region. Denoting the central region with superscript 0 and the homogenized outer region with superscript 1, we have

$$\Sigma_{p}^{0} = \sum_{m \in 0} N_{m}^{0} \tau_{m} + \frac{\kappa^{2}}{3 \sum_{m} N_{m}^{\sigma} \tau_{m}} + \Sigma_{e}^{*0}$$
(III.14)

where the first summation extends over all isotopes contained in the central region and the second summation is as in Eq. III.7 over all isotopes in the equivalent homogeneous mixture of all cylindrical regions.

The escape cross section for the central region Σ_{a}^{*0} is calculated as

$$\Sigma_{e}^{*0} = \Sigma_{e}^{0} \frac{a_{1}(1 - c^{0})}{1 + (a_{2} - 1)c^{0}}$$
(III.15)
$$\Sigma_{e}^{0} = \frac{s^{0}}{4v^{0}}$$
(III.16)

where

with S⁰ and V⁰ the surface area and volume of the central region, a_1 and a_2 are user supplied with default values of 1.35, and the Dancoff factor C⁰ is given by

$$C^{0} = 1 - \gamma - \gamma^{4}(1 - \gamma).$$
 (III.17)

In Eq. III.17

 $\gamma = \frac{\Sigma_{t}^{1}}{\Sigma_{t}^{1} + \Sigma_{e}^{1}}$

(III.13)

with

$$\Sigma_{t}^{1} = \sum_{m \in 1} N_{m}^{2} \sigma_{t}$$
$$\Sigma_{e}^{1} = \frac{s^{0}}{4v^{1}}$$

and V^1 is the volume of the homogenized outer regions. The summation in Eq.III.19 extends over all isotopes contained in the outer regions, including the isotope i if it should be present.

The x on $J_{i}^{*s\ell}$ in Eq.III.2 refers to either capture or fission. Thus the

effective macroscopic fission cross section for the interval u_1 , u_2 can similarly be written as in Eq.III.2

$$\overline{\Sigma}_{\mathbf{f}_{i}} = \frac{\sum_{\mathbf{p}_{i}}^{\mathbf{p}_{i}} \mathbf{f}_{i}}{\Delta u \mathbf{E}_{o_{i}} \mathbf{f}} \mathbf{J}_{\mathbf{x}_{i}}^{* \delta \ell}$$
(III.21)

where Γ_{f_i} is the fission width for resonance i.

In the case of the total cross section, we have

$$\overline{\Sigma}_{t_{i}} = \frac{\sum_{i} \sum_{j=1}^{r} \sum_{i=1}^{r} \frac{1}{2} \int_{-\infty}^{\infty} \frac{(\psi_{i} + a_{i}\chi_{i}) dx_{i}}{\beta_{i} + \psi_{i} + a_{i}\chi_{i} + \sum_{i' \neq i} (A_{i}, \psi_{i'}, + B_{i'}, \chi_{i'})}$$
$$= \frac{\sum_{i} \sum_{j=1}^{r} \sum_{i=1}^{r} J_{t_{i}}^{*_{i}\ell}}{\Delta u E_{o_{i}} \int_{t_{i}}^{*_{i}\ell} J_{t_{i}}^{*_{i}\ell}}.$$
(III.22)

Now the resonance integrals can be factored $\binom{(25)}{\text{to display a lead term}}$ corresponding to the isolated resonance integral (the usual resonance integral as discussed, e.g. by Dresner $\binom{(26)}{}$) and a term due to the overlap of other resonances in the mixture. In particular, $J_{x_i}^{* \& \ell}$ can be rewritten as

$$J_{x_{i}}^{*_{\delta}\ell} = \frac{1}{2} \int_{-\infty}^{\infty} dx_{i} \frac{\psi_{i}}{\beta_{i} + \psi_{i} + a_{i}\chi_{i}} - \frac{1}{2} \int_{-\infty}^{\infty} dx_{i} \frac{\psi_{i}}{\beta_{i} + \psi_{i} + a_{i}\chi_{i}}$$
$$\cdot \frac{\sum_{i'\neq i}^{\sum} (A_{i}, \psi_{i'} + B_{i'}\chi_{i'})}{\beta_{i} + \psi_{i} + a_{i}\chi_{i} + \sum_{i'\neq i}^{\sum} (A_{i}, \psi_{i'} + B_{i'}\chi_{i'})} = J_{i}^{x,\delta\ell} - O_{i,i',o}^{x,\delta\ell} (III.23)$$

where the first integral denoted by $J_{i,i,0}^{x,\delta\ell}$ is the usual isolated resonance integral and the second denoted by $0_{i,i',0}^{x,\delta\ell}$ is the overlap integral due to other single level resonances i'. The zero indicates that resonances are all of one type, that is, all single level.

(III.19)

(III.20)

Eq.III.22 may be similarly factored with $\psi_i + a_i \chi_i$ replacing ψ_i in the numerators of each of the integrals in Eq.III.23. Therefore

$$J_{t_{i}}^{*_{\delta}\ell} = \frac{1}{2} \int_{-\infty}^{\infty} dx_{i} \frac{\psi_{i} + a_{i}\chi_{i}}{\beta_{i} + \psi_{i} + a_{i}\chi_{i}} - \frac{1}{2} \int_{-\infty}^{\infty} dx_{i} \frac{\psi_{i} + a_{i}\chi_{i}}{\beta_{i} + \psi_{i} + a_{i}\chi_{i}} \\ \cdot \frac{\sum_{i'\neq i} (A_{i}, \psi_{i'}, + B_{i'}, \chi_{i'})}{\beta_{i} + \psi_{i} + a_{i}\chi_{i} + \sum_{i'\neq i} (A_{i}, \psi_{i'}, + B_{i'}, \chi_{i'})} = J_{i}^{t,\delta\ell} - O_{i,i',0}^{t,\delta\ell} . \quad (III.24)$$

Now in the case of multilevel resonances, instead of the expressions given in Eqs. III.3 and III.4 we have

$$\sigma_{\mathbf{c}_{\mathbf{k}}} = \sigma_{\mathbf{o}_{\mathbf{k}}} \frac{\varphi_{\mathbf{k}}}{|\mathsf{G}_{\mathbf{t}_{\mathbf{k}}}|} \left[\psi_{\mathbf{k}} + b_{\gamma_{\mathbf{k}}} \chi_{\mathbf{k}} \right]$$
(III.25)
$$\sigma_{\mathbf{f}_{\mathbf{k}}} = \sigma_{\mathbf{o}_{\mathbf{k}}} \frac{G_{\mathbf{f}_{\mathbf{k}}}}{|\mathsf{G}_{\mathbf{t}_{\mathbf{k}}}|} \left[\psi_{\mathbf{k}} + b_{\mathbf{f}_{\mathbf{k}}} \chi_{\mathbf{k}} \right]$$
(III.26)
$$\sigma_{\mathbf{t}_{\mathbf{k}}} = \sigma_{\mathbf{o}_{\mathbf{k}}} \left[\frac{G_{\mathbf{t}_{\mathbf{k}}}}{|\mathsf{G}_{\mathbf{t}_{\mathbf{k}}}|} \psi_{\mathbf{k}} + a_{\mathbf{k}} \chi_{\mathbf{k}} \right] .$$
(III.27)

In Eqs. III.25-III.27 we again set E = E and use the shorthand notation $\psi_k = \psi(\theta_k, \mathbf{x}_k)$ and $\chi_k = \chi(\theta_k, \mathbf{x}_k)$.

The definition of the various coefficients depends upon whether the resonance k is represented by the Adler-Adler or Breit-Wigner forms. In the following, we will assume the flux correction factor f equals 1.

Multilevel Breit-Wigner

- $\sigma_{o_k} = (\text{single level Breit-Wigner } \sigma_o) \cdot |G_{t_k}| \text{ where the single} \\ \text{level Breit-Wigner } \sigma_o \text{ is defined above}$

 $G_{\gamma k}$ = Breit-Wigner multilevel parameter for the radiative γ_k capture reaction

 $= \Gamma_{\gamma_k} / \Gamma_{t_k}$

 Γ_{t_1} = Breit-Wigner multilevel total line width

-73-

 $b_{\gamma_k} = 0$ ^Gf_k = Breit-Wigner multilevel parameter for the fission reaction $= \Gamma_{f_k} / \Gamma_{t_k}$ $b_{f_k} = 0$ a k = (single level Breit-Wigner a) plus the antisymmetric level-level interference contribution, all divided · by G_t. $\theta_{\mathbf{k}} = \Gamma_{\mathbf{t}_{\mathbf{k}}} / \Delta_{\mathbf{k}}$ $x_k = 2(E - E_{o_k})/\Gamma_{t_k}$ Multilevel Adler-Adler $\sigma_{o_{k}} = \frac{2.6039953 \times 10^{6}}{\left| E_{o_{1}} \right|} \left[\frac{A_{k} + 1}{A_{k}} \right]^{2} \frac{\left| G_{t_{k}} \right|}{2\Gamma_{k}^{(s)}}$ $G_{t_1} = \sqrt{E_{o_1}} \left[G_k^T \cos 2\phi_{\ell_1} + H_k^T \sin 2\phi_{\ell_1} \right]$ G_k^T = Adler-Adler symmetric capture cross section parameter from ENDF/B н^Т = Adler-Adler antisymmetric capture cross section parameter from ENDF/B [¢]لړ = phase shift $\Gamma_{k}^{(6)} = S$ -matrix total line width for the Adler-Adler formulation $G_{\gamma_{k}} = \sqrt{E_{o_{k}}} \left[G_{k}^{c} \cos 2\phi_{\ell_{k}} + H_{k}^{c} \sin 2\phi_{\ell_{k}} \right]$ G_k^c = Adler-Adler symmetric capture cross section parameter from ENDF/B H_k^c = Adler-Adler antisymmetric capture cross section parameter from ENDF/B $\begin{bmatrix} H_{k}^{c} \cos 2\phi_{\ell_{k}} - G_{k}^{c} \sin 2\phi_{\ell_{k}} \\ \hline G_{v} \end{bmatrix} \sqrt{E_{o_{k}}}$

$$G_{f_{k}} = \sqrt{\left| E_{o_{k}} \right|} \left[G_{k}^{F} \cos 2\phi_{\ell_{k}} + H_{k}^{F} \cos 2\phi_{\ell_{k}} \right]$$

 G_k^{F} = Adler-Adler symmetric fission cross section parameter from ENDF/B

 H_k^F = Adler-Adler antisymmetric fission cross section parameter from ENDF/B

$$b_{f_{k}} = -0.5 \left[\frac{H_{k}^{F} \cos 2\phi_{\ell_{k}} - G_{k}^{F} \sin 2\phi_{\ell_{k}}}{G_{f_{k}}} \right] \sqrt{|E_{o_{k}}|}$$
$$a_{k} = -0.5 \left[\frac{H_{k}^{T} \cos 2\phi_{\ell_{k}} - G_{k}^{T} \sin 2\phi_{\ell_{k}}}{|G_{t_{k}}|} \right] \sqrt{|E_{o_{k}}|}$$

Using Eqs. III.25-III.27 in Eq. III-1, we have for the case of all Adler-Adler multilevel resonances in the mixture, for example

$$\overline{\Sigma}_{c_{i}} = \frac{\sum_{p} G_{\gamma_{k}} \Gamma_{k}^{(s)}}{\Delta u E_{o_{k}} |G_{t_{k}}|} \frac{1}{2} \int_{-\infty}^{\infty} dx_{k} \frac{\psi_{k} + b_{\gamma_{k}} \chi_{k}}{\beta_{k} + S_{k} \psi_{k} + a_{k} \chi_{k} + \sum_{k' \neq k} (S_{k}, A_{k}, \psi_{k'} + B_{k'} \chi_{k'})}$$
$$= \frac{\sum_{p} \Gamma_{k}^{(s)}}{\Delta u E_{o_{k}}} \frac{G_{t_{k}}}{|G_{t_{k}}|} J_{\gamma_{k}}^{*m\ell} = \frac{\sum_{p} \Gamma_{k}^{(s)}}{\Delta u E_{o_{k}}} S_{k} J_{\gamma_{k}}^{*m\ell}$$
(III.28)

Equation III.28 may be compared with Eq.III.2 and $S_k = G_{t_k} / |G_{t_k}|$. The *ml* superscript refers to multilevel. In the case of multilevel Breit-Wigner resonances, $\Gamma_k^{(s)}$ is replaced by the multilevel total line width and b_{γ_k} is zero. A_k , B_k , and β_k are defined as in Eq.III.5 with σ_{o_k} and a_k defined as appropriate to the multilevel representation involved.

In the case of fission, Eq.III.28 becomes

$$\overline{\Sigma}_{f} = \frac{\Sigma_{p} G_{f_{k}} \Gamma_{k}^{(s)}}{\Delta u E_{o_{k}} |G_{t_{k}}|} J_{f_{k}}^{*m\ell}$$
(III.29)

where $J_{f_k}^{*m\ell}$ is given by the integral of Eq. III.28 but with b_{f_k} replacing γ_k .

For the total cross section

$$\overline{\Sigma}_{t_{k}} = \frac{\Sigma_{p}\Gamma_{k}^{(s)}}{\Delta uE_{o_{k}}} \frac{1}{2} \int_{-\infty}^{\infty} dx_{k} \frac{S_{k}\psi_{k} + a_{k}\chi_{k}}{\beta_{k} + S_{k}\psi_{k} + a_{k}\chi_{k} + \sum_{k'\neq k}(S_{k'}A_{k'}\psi_{k'} + B_{k'}\chi_{k'})}$$
$$= \frac{\Sigma_{p}\Gamma_{k}^{(s)}}{\Delta uE_{o_{k}}} J_{t_{k}}^{*m\ell}. \qquad (III.30)$$

As in the case of the single level resonances, the multilevel resonance integrals may be factored into a lead term and a term due to the overlap of other resonances in the mixture. Thus the integral in Eq.III.28 can be written as

$$J_{\gamma_{k}}^{*m\ell} = \frac{1}{2} \int_{-\infty}^{\omega} dx_{k} \frac{\psi_{k} + b_{\gamma_{k}} \chi_{k}}{\beta_{k} + s_{k} \psi_{k} + a_{k} \chi_{k}} - \frac{1}{2} \int_{-\infty}^{\omega} dx_{k} \frac{\psi_{k} + b_{\gamma_{k}} \chi_{k}}{\beta_{k} + s_{k} \psi_{k} + a_{k} \chi_{k}}$$
$$\cdot \frac{\sum_{k' \neq k} (s_{k}, A_{k}, \psi_{k'}, + B_{k'}, \chi_{k'})}{\beta_{k} + s_{k} \psi_{k} + a_{k} \chi_{k} + \sum_{k' \neq k} (s_{k}, A_{k'}, \psi_{k'}, + B_{k'}, \chi_{k'})}$$
$$= J_{k}^{\gamma, m\ell} - o_{k, k', 0}^{\gamma, m\ell} . \qquad (III.31)$$

The factored form of Eq. III.29 can, by comparison with Eq. III.31 be written

$$J_{f_{k}}^{*m\ell} = J_{k}^{f,m\ell} - O_{k,k',0}^{f,m\ell} .$$
(III.32)

Also, Eq. III. 30 may be factored to yield

$$J_{t_{k}}^{*m\ell} = \frac{1}{2} \int_{-\infty}^{\infty} dx_{k} \frac{S_{k}\psi_{k} + a_{k}\chi_{k}}{\beta_{k} + S_{k}\psi_{k} + a_{k}\chi_{k}} - \frac{1}{2} \int_{-\infty}^{\infty} dx_{k} \frac{S_{k}\psi_{k} + a_{k}\chi_{k}}{\beta_{k} + S_{k}\psi_{k} + a_{k}\chi_{k}}$$
$$\cdot \frac{\sum_{k' \neq k} (S_{k'}A_{k'}\psi_{k'} + B_{k'}\chi_{k'})}{\beta_{k} + S_{k}\psi_{k} + a_{k}\chi_{k} + \sum_{k' \neq k} (S_{k'}A_{k'}\psi_{k'} + B_{k'}\chi_{k'})} = J_{k}^{t,m\ell} - O_{k,k',0}^{t,m\ell}$$
(III.33)

So far we have considered the case of a mixture of either all single level (Eqs.III.23-III.24) or all multilevel (Eqs.III.31-III.33) resonances. Two other situations may also arise, namely a single level resonance in the presence of other single and multilevel level resonances, or a multilevel resonance in the presence of other single and multilevel resonances. Each of these cases may be written in factored form to yield a lead integral for the isolated single level (multilevel) resonance, minus integrals resulting from overlap with other single level (multilevel) and other multilevel (single level) resonances.

-76-

Thus, for a single level resonance i we have

$$J_{x_{i}}^{*,s\ell} = J_{i}^{x,s\ell} - 0_{i,i',o}^{x,s\ell} - 0_{i,i',k}^{x,s\ell}$$
(III.34)

and

$$J_{t,i}^{*s\ell} = J_{i}^{t,s\ell} - 0_{i,i',o}^{t,s\ell} - 0_{i,i',k}^{t,s\ell} .$$
 (III.35)

 $J_{i}^{x,\delta\ell}$ and $0_{i,i',o}^{x,\delta\ell}$ are given in Eq.III.23 and $J_{i}^{t,\delta\ell}$ and $0_{i,i',o}^{t,\delta\ell}$ are given in Eq.III.29. The second overlap terms which depend upon the single level resonances i and i' and upon the multilevel resonances k can be written explicitly as

$$D_{i,i',k}^{x,\delta\ell} = \frac{1}{2} \int_{-\infty}^{\infty} dx_{i} \frac{\psi_{i}}{\beta_{i} + \psi_{i} + a_{i}\chi_{i} + \sum_{i'\neq i}^{\gamma} (A_{i},\psi_{i'} + B_{i'}\chi_{i'})} \cdot \int_{-\infty}^{\infty} \frac{\sum_{k}^{\gamma} (S_{k}A_{ki}\psi_{k} + B_{ki}\chi_{k})}{\beta_{i} + \psi_{i} + a_{i}\chi_{i} + \sum_{i'\neq i}^{\gamma} (A_{i'},\psi_{i'} + B_{i'}\chi_{i'}) + \sum_{k}^{\gamma} (S_{k}A_{ki}\psi_{k} + B_{ki}\chi_{k})} (111.36)$$

and

$$\int_{-\infty}^{\infty} \frac{\psi_{i} + a_{i}\chi_{i}}{\beta_{i} + \psi_{i} + a_{i}\chi_{i} + \sum_{i' \neq i} (A_{i}, \psi_{i'} + B_{i'}, \chi_{i'})} \cdot \int_{-\infty}^{\infty} \frac{\sum_{k} (S_{k}A_{ki}\psi_{k} + B_{ki}\chi_{k})}{\beta_{i} + \psi_{i} + a_{i}\chi_{i} + \sum_{i' \neq i} (A_{i}, \psi_{i'} + B_{ki}\chi_{k})} \cdot (111.37)$$

In Eqs. III.36 and III.37, A_i , and B_i , are defined in Eq. III.5 and

$$A_{ki} = \frac{N_{o_{k}\sigma_{o_{k}}}}{N_{o_{i}\sigma_{i}}}$$
(III.38)

Similarly, for a multilevel resonance k we have

$$J_{\gamma_{k}}^{*ml} = J_{k}^{\gamma,ml} - O_{k,k',0}^{\gamma,ml} - O_{k,k',i}^{\gamma,ml}$$
(III.39)

$$J_{f_{k}}^{*m\ell} = J_{k}^{f,m\ell} - O_{k,k',0}^{f,m\ell} - O_{k,k',i}^{f,m\ell}$$
(III.40)

$$J_{t_{k}}^{*ml} = J_{k}^{t,ml} - O_{k,k',0}^{t,ml} - O_{k,k',i}^{t,ml} .$$
(III.41)

In Eqs.III.37-III.41 the indices k and k' refer to multilevel resonances and ind i refers to single level resonances in the mixture.

 $J_k^{\gamma,m\ell}$ and $0_{k,k',o}^{\gamma,m\ell}$ are given in Eq.III.31. $J_k^{f,m\ell}$ and $0_{k,k',o}^{f,m\ell}$ are the above expressions with b_{f_k} replacing b_{γ_k} . $J_k^{t,m\ell}$ and $0_{k,k',o}^{t,m\ell}$ are given in Eq.III.33. The second overlap terms can be written explicitly as

$$O_{k,k',i}^{\gamma,m\ell} = \frac{1}{2} \int_{-\infty}^{\infty} dx_k \frac{\psi_k + b_{\gamma_k} x_k}{\beta_k + s_k \psi_k + a_k x_k + \sum_{k' \neq k} (s_k, A_k, \psi_k, + b_k, x_{k'})}$$

$$\int_{-\infty}^{\infty} \frac{\sum_{i} (A_{ik}\psi_{i} + B_{ik}\chi_{i})}{\beta_{k} + S_{k}\psi_{k} + a_{k}\chi_{k} + \sum_{k'\neq k} (S_{k'}A_{k'}\psi_{k'} + B_{k'}\chi_{k'}) + \sum_{i} (A_{ik}\psi_{i} + B_{ik}\chi_{i})}$$
(III.42)

with $0_{k,k',i}^{f,m\ell}$ as in Eq.III.42 with b_{f_k} replacing b_{γ_k} , and

$$O_{k,k',i}^{t,m\ell} = \frac{1}{2} \int_{-\infty}^{\infty} dx_{k} \frac{S_{k}\psi_{k} + a_{k}\chi_{k}}{\beta_{k} + S_{k}\psi_{k} + a_{k}\chi_{k} + \sum_{k'\neq k}(S_{k},A_{k},\psi_{k'}, + B_{k},\chi_{k'})} \cdot \int_{-\infty}^{\infty} \frac{\sum_{i}(A_{ik}\psi_{i} + B_{ik}\chi_{i})}{\beta_{k} + S_{k}\psi_{k} + a_{k}\chi_{k} + \sum_{k'\neq k}(S_{k},A_{k},\psi_{k'}, + B_{k'},\chi_{k'})} \cdot (III.43)$$

B. Calculation of Isolated Resonance Integrals

Each of the isolated resonance integrals $J_{i}^{x,s\ell}$, $J_{i}^{t,s\ell}$, $J_{k}^{\gamma,m\ell}$, $J_{k}^{f,m\ell}$, and $J_{k}^{t,m\ell}$ as given in Eqs.III.23, III.24, III.31, III.32 and III.33 represents a special case of the general resonance integral representation

$$J(\beta,\theta,a,b) = \frac{1}{2} \int_{-\infty}^{\infty} dx \frac{T\psi + b\chi}{\beta + S\psi + a\chi}$$
(III.44)

In particular for single level resonances T = S = 1, and b = 0 for capture or fission, and equals a for total. Similarly, for multilevel resonances $b = b_{\gamma}$, b_{f} , or a for capture, fission or total, T = 1 for capture and fission, and T = S for total.

$$J(\beta,\theta,a,b) = J(\beta,\theta,0,0) + I(\beta,\theta,a) - bM(\beta,\theta,a)$$
(III.45)

where

$$J(\beta,\theta,0,0) = T \int_0^\infty dx \, \frac{\psi}{\beta + S\psi}$$
(III.46)

$$I(\beta,\theta,a) = Ta^{2} \int_{0}^{\infty} dx \frac{\psi}{\beta + S\psi} \cdot \frac{\chi^{2}}{(\beta + S\psi)^{2} - a^{2}\chi^{2}}$$
(III.47)

$$M(\beta,\theta,a) = a \int_{0}^{\infty} dx \frac{\chi^{2}}{(\beta + S\psi)^{2} - a^{2}\chi^{2}}.$$
 (III.48)

The quantities $J(\beta, \theta, 0, 0)$, $I(\beta, \theta, a)$ and $M(\beta, \theta, a)$ are calculated using the numerical methods described in Section IV of Appendix A involving fixed point Gauss-Jacobi quadrature or asymptotic expressions depending upon the size of β .

The user may specify that all resonances for a particular isotope be assigned the infinitely dilute resonance integral limiting value $\pi/(2\beta)$. In this case, none of the resonances of this isotope will be involved in the calculation of the overlap integrals for any other resonance in the mixture.

C. Calculation of Overlap Integrals

 $W_{i}^{2} = \frac{\Gamma_{i}}{2} \sqrt{\frac{\beta_{i}+1}{\beta_{i}}}$

1. Selection of Overlapping Resonances

In the calculation of the various overlap integrals, only those neighboring resonances are included which satisfy the following criterion.

Two "widths" are computed for each resonance i

$$W_{i}^{1} = \frac{\Gamma_{i}}{\theta_{i}} \sqrt{\ln \left[2 + \frac{\psi(\theta_{i}, 0)}{\beta_{i}}\right]}$$
(III.49)

(III.50)

(III.51)

and

where Γ_i is the total line width (or S-matrix total line width for Adler-Adler multilevel resonances) and $\psi(\theta_i, 0)$ is calculated as in Eq.A.15 of

Appendix A. Denoting by L_i the larger of W_i^1 and W_i^2 $L_i = \max \left[W_i^1, W_i^2 \right]$,

we include resonance i' in the overlap integral for resonance i if

$$\begin{vmatrix} E_{o_i} - E_{o_i} \end{vmatrix} < 10 L_i.$$
(III.52)

Otherwise resonance i' is ignored.

The code defaults to four neighbors on each side of each resonance as candidates for inclusion in the overlap integrals, but the number of overlap candidates can be specified by the user.

2. Asymptotic Algorithms

ß

The code first evaluates the asymptotic or "large β " approximations⁽²⁵⁾ for the various overlap integrals. The integrands of these integrals (Eqs.III.23, III.24, III.31, III.32, III.33, III.36, III.37, III.42 and III.43) consist of products of ψ and χ functions times a factor of the form

$$\frac{1}{\beta^2 + f}$$
 (III.53)

where f represents the energy dependent part of the denominator. If we add and subtract a quantity σ^* , to be defined below, we can write

$$\frac{1}{2 + f} = \frac{1}{\beta^2 + \sigma^* - (\sigma^* - f)}$$

$$= \frac{1}{\beta^2 + \sigma^*} \left[1 - \frac{1}{\frac{\sigma^* - f}{\beta^2 + \sigma^*}} \right]$$

$$\approx \frac{1}{\beta^2 + \sigma^*} \left[1 + \frac{\sigma^* - f}{\beta^2 + \sigma^*} + \dots \right]$$

$$\approx \frac{1}{\beta^2 + \sigma^*} + \frac{\sigma^* - f}{(\beta^2 + \sigma^*)^2} \quad . \quad (III.54)$$

As a specific example, consider $0^{x,\delta\ell}_{i,i',o}$ as given in Eq.III.23. Using Eq.III.54 we can write

$$0_{i,i',o}^{x,\delta\ell} \approx \frac{1}{2} \int_{-\infty}^{\infty} dx_{i} \frac{\psi_{i} \sum_{i'\neq i}^{\gamma} (A_{i'}\psi_{i'} + B_{i'}\chi_{i'})}{\beta_{i}^{2} + \sigma_{i,i',0}^{*}} + \frac{1}{2} \int_{-\infty}^{\infty} dx_{i} \frac{\left[\sigma_{i,i',0}^{*} - f_{i,i',0}\right] \psi_{i} \sum_{i'\neq i}^{\gamma} (A_{i'}\psi_{i'} + B_{i'}\chi_{i'})}{\left[\beta_{i}^{2} + \sigma_{i,i',0}^{*}\right]^{2}}$$
(III.55)

where

$$f_{i,i',0} = 2\beta_{i}(\psi_{i} + a_{i}\chi_{i}) + (\psi_{i} + a_{i}\chi_{i})^{2} + \beta_{i} \sum_{i'\neq i} (A_{i},\psi_{i}, + B_{i'}\chi_{i'}) + (\psi_{i} + a_{i}\chi_{i}) \sum_{i'\neq i} (A_{i'}\psi_{i'}, + B_{i'}\chi_{i'})$$

Now if we define

-m

$$\sigma_{i,i',0}^{\delta\ell*} = \frac{\int_{-\infty}^{\infty} dx_{i} f_{i,i',0} \psi_{i,i'\neq i} (A_{i'} \psi_{i'} + B_{i'} \chi_{i'})}{\int_{-\infty}^{\infty} dx_{i} \psi_{i} \sum_{i'\neq i}^{(A_{i'} \psi_{i'} + B_{i'} \chi_{i'})} (III.57)}$$

with $f_{i,i',0}$ given by Eq.III.56, the second order term of Eq.III.55 will vanish and we are left with the approximation

$$O_{\mathbf{i},\mathbf{i}',\mathbf{0}}^{\mathbf{x},\delta\ell} \approx \frac{1}{2} \int_{-\infty}^{\infty} d\mathbf{x}_{\mathbf{i}} \frac{\psi_{\mathbf{i}} \sum_{\mathbf{i}'\neq\mathbf{i}}^{(\mathbf{A}_{\mathbf{i}},\psi_{\mathbf{i}'}+\mathbf{B}_{\mathbf{i}'}\chi_{\mathbf{i}'})}{\beta_{\mathbf{i}}^{2} + \sigma_{\mathbf{i},\mathbf{i}',\mathbf{0}}^{\delta\ell*}} \qquad (III.58)$$

Similarly, the total overlap integral as given in Eq.III.24 can be written as the sum of two terms

$$o_{i,i',o}^{t,s\ell} = ov_{1,i',o}^{x,s\ell} + ov_{2,i',o}^{t,s\ell}$$
 (III.59)

where $0V_{i,i',o}^{x,\delta\ell}$ is $0_{i,i',o}^{x,\delta\ell}$ as given in Eq.III.23 and approximated in Eq.III.58

$$0V_{2_{i,i',0}}^{t,\delta\ell} = \frac{1}{2} \int_{-\infty}^{\infty} dx_{i} \frac{a_{i}\chi_{i} \sum_{i'\neq i}^{(A_{i'}\psi_{i'}+B_{i'}\chi_{i'})}}{f_{i,i',0}}$$
(III.60)

where f i,i',o is again given by Eq. III.56.

Using the same technique as above, we may approximate Eq. III.60 as

$$OV_{2_{i,i',0}}^{t,s\ell} \approx \frac{1}{2} \int_{-\infty}^{\infty} dx_{i} \frac{a_{i}\chi_{i} \sum_{i'\neq i}^{(A_{i},\psi_{i'}+B_{i'},\chi_{i'})}}{\beta_{i}^{2} + \sigma_{i,i',0}^{s\ell**}}$$
(III.61)

with $\sigma_{i,i',o}^{**}$ defined as

$$\sigma_{i,i',o}^{\$\ell**} = \frac{\int_{-\infty}^{\infty} dx_{i}f_{i,i',o}a_{i}\chi_{i}}{\int_{-\infty}^{\infty} dx_{i}a_{i}\chi_{i}} \left[(A_{i},\psi_{i},B_{i'},\chi_{i'}) - \int_{-\infty}^{\infty} dx_{i}a_{i}\chi_{i'} \left[(A_{i'},\psi_{i'},B_{i'},\chi_{i'}) - \int_{-\infty}^{\infty} dx_{i'}a_{i'}\chi_{i'} \left[(A_{i'},\psi_{i'},B_{i'},\chi_{i'}) - \int_{-\infty}^{\infty} dx_{i'}dx_{i'} \left[(A_{i'},\psi_{i'},B_{i'},\chi_{i'}) - \int_{-\infty}^{\infty} dx_{i'}dx_{i'} \left[(A_{i'},\psi_{i'},\chi_{i'}) - \int_{-\infty}^{\infty} dx_{i'}dx_{i'} \left[(A_{i'},\chi_{i'},$$

This same technique is used to obtain asymptotic expressions for the various other overlap integrals $0_{k,k',0}^{\gamma,m\ell}$, $0_{i,i',k}^{\chi,\delta\ell}$, etc. where each requires the definition of a corresponding σ^* and σ^{**} appropriate to the specific form of the integrand involved. Note that each of the multilevel overlap integrals, and the single level overlap integral for the total cross section can be written as in Eq.III.58 as the sum of a symmetric term $0V_1$ and an antisymmetric term $0V_2$.

Once the σ^* and σ^{**} have been evaluated, as described later, the integrals remaining in these asymptotic expressions all involve infinite integrals of products of ψ and χ functions for the resonance being evaluated and for the neighboring resonances included in the overlap calculation. These may be integrated using the following identities.⁽²⁷⁾

$$\int_{-\infty}^{\infty} d\mathbf{x}_{i} \psi(\theta_{i}, \mathbf{x}_{i}) \psi(\theta_{j}, \mathbf{x}_{j}) = \int_{-\infty}^{\infty} d\mathbf{x}_{i} \psi_{i} \psi_{j}$$

$$= \pi \frac{\Gamma_{j}}{\Gamma_{i} + \Gamma_{j}} \psi \left[\frac{\Gamma_{i} + \Gamma_{j}}{\sqrt{\Delta_{i}^{2} + \Delta_{j}^{2}}}, \frac{E_{0} - E_{0}}{\Gamma_{i} + \Gamma_{j}} \right]$$

$$= \pi \frac{\Gamma_{j}}{\Gamma_{i} + \Gamma_{j}} \psi \left[\theta_{ij}, \mathbf{x}_{ij} \right]$$

$$= \pi \frac{\Gamma_{j}}{\Gamma_{i} + \Gamma_{j}} \psi_{ij} \qquad (III.6)$$

$$\int_{-\infty} dx_{i} \chi_{j} = 4\pi \frac{\Gamma_{j}}{\Gamma_{i} + \Gamma_{j}} \psi_{ij} \qquad (III.64)$$

3)

$$\int_{-\infty}^{\infty} dx_{i} \psi_{i} \chi_{j} = \pi \frac{\Gamma_{j}}{\Gamma_{i} + \Gamma_{j}} \chi_{ij}$$
(III.65)

where χ_{ij} implies the same notation for χ as in the case of Eq.III.63 for ψ_{ij}

-82-

đ

$$\int_{-\infty}^{\infty} dx_{i} \chi_{i} \psi_{j} = -\pi \frac{\Gamma_{j}}{\Gamma_{i} + \Gamma_{j}} \chi_{ij} \qquad (III.66)$$

$$\int_{-\infty}^{\infty} dx_{i} \psi_{i}^{2} = \frac{\pi}{2} \psi(\sqrt{2}\theta_{i}, 0) \qquad (III.67)$$

$$\int_{-\infty}^{\infty} dx_{i} \chi_{i}^{2} = 2\pi \psi(\sqrt{2}\theta_{i}, 0) \qquad (III.68)$$

The more compact notation identified in Eq. III.63 is used for Eqs. III.64-III.68.

Making use of Eqs.III.63-III.68 in Eqs.III.58-III.61, and in the analogous integrals for the other cases discussed above, we finally may tabulate the asymptotic overlap integrals as follows.

$$O_{i,i',o}^{\mathbf{x},\delta\ell} \approx \frac{\pi}{2} \frac{\sum_{i'\neq i} \frac{1_{i'}}{\Gamma_i + \Gamma_i'}}{\beta_i^2 + \sigma_{i,i',o}^{\delta\ell*}} = O_{1_{i,i',o}}^{\mathbf{x},\delta\ell}$$
(III.69)

$$o_{i,i',o}^{t,\delta\ell} \approx o_{i,i',o}^{x,\delta\ell} + \frac{\pi}{2} \frac{a_i \sum_{\substack{i'\neq i \\ i'\neq i \\ i'\neq i}}^{i'i'} (-A_i, \chi_{ii'} + 4B_i, \psi_{ii'})}{\beta_i^{2} + \sigma_{i,i',o}^{\delta\ell*}}$$

 $= \operatorname{ov}_{1,i',o}^{x,\delta\ell} + \operatorname{ov}_{2,i',o}^{t,\delta\ell}$ (III.70)

$$O_{k,k',o}^{\gamma,m\ell} \approx \frac{\pi}{2} \frac{\sum_{\substack{k' \neq k}} \frac{\Gamma_{k'}}{\Gamma_{k}+\Gamma_{k'}}}{\beta_{k}^{2}+\sigma_{k,k',o}^{m\ell*}}$$

$$+ \frac{\pi}{2} \frac{{}^{b}\gamma_{k\ k}}{\frac{\gamma_{k\ k}}{\beta_{k}}^{2} + \sigma_{k}^{m\ell * *}} (- S_{k}, A_{k}, \chi_{kk}, +4B_{k}, \psi_{kk},)}{\beta_{k}^{2} + \sigma_{k,k}^{m\ell * *}}$$

 $= 0V_{l_{k,k',0}}^{x,ml} + 0V_{k,k',0}^{\gamma,ml}$

(III.71)

-83-

$$o_{k,k',o}^{f,m\ell} \approx o_{k,k',o}^{x,m\ell} + o_{k,k',o}^{f,m\ell} = o_{k,k',o}^{x,m\ell} + \frac{b_{f_{k}}}{b_{\gamma_{k}}} o_{2,k,k',o}^{\gamma,m\ell}$$
(III.72)

$$o_{k,k',o}^{t,m\ell} \approx s_{k} o_{k,k',o}^{x,m\ell} + o_{2,k,k',o}^{t,m\ell}$$
(III.73)

$$= s_{k} o_{k,k',o}^{x,m\ell} + \frac{a_{k}}{b_{\gamma_{k}}} o_{2,k,k',o}^{\gamma,m\ell}$$
(III.73)

$$o_{i,i',k}^{x,\delta\ell} \approx \frac{\pi}{2} \frac{\sum_{k} \frac{\Gamma_{k}}{\Gamma_{i}+\Gamma_{k}}}{\beta_{i}^{2}+\sigma_{i,i',k}^{\delta\ell*}} = o_{1,i',k}^{x,\delta\ell}$$
(III.74)

$$O_{i,i',k}^{t,s\ell} = OV_{i,i',k}^{x,s\ell} + \frac{\pi}{2} \frac{a_{i_{k}}^{\sum} \frac{\Gamma_{k}}{\Gamma_{i}+\Gamma_{k}} (-S_{k}A_{ki}\chi_{ik}+4B_{ki}\psi_{ik})}{\beta_{i}^{2}+\sigma_{i,i',k}^{\delta\ell**}}$$

$$= OV_{1,i',k}^{x,sl} + OV_{2,i,i',o}^{t,sl}$$

$$o_{k,k',i}^{\gamma,m\ell} \approx \frac{\pi}{2} \frac{\sum \frac{\Gamma_{i}}{\Gamma_{i}+\Gamma_{k}} (A_{ik}\psi_{ki}+B_{ik}\chi_{ki})}{\beta_{k}^{2}+\sigma_{k,k',i}^{m\ell*}}$$

i.

$$+ \frac{\pi}{2} \frac{b_{\gamma_{k} i} \sum_{i} \frac{\Gamma_{i}}{\Gamma_{i} + \Gamma_{k}} (- A_{ik} \chi_{ki} + 4B_{ik} \psi_{ki})}{\beta_{k}^{2} + \sigma_{k,k',i}^{m\ell * *}}$$

$$= 0V_{k,k',i}^{x,m\ell} + 0V_{k,k',o}^{\gamma,m\ell}$$
(III.76)

(III.75)

$$O_{k,k',i}^{f,m\ell} \stackrel{\sim}{\sim} OV_{k,k',i}^{x,m\ell} + OV_{k,k',i}^{f,m\ell} = OV_{k,k',i}^{x,m\ell} + \frac{b_{f_{k}}}{b_{\gamma_{k}}} OV_{k,k',i}^{\gamma,m\ell}$$
(III.77)

$$o_{k,k',i}^{t,m\ell} \stackrel{\sim}{\sim} s_{k} ov_{1_{k,k',i}}^{x,m\ell} + \frac{a_{k}}{b_{\gamma_{k}}} ov_{2_{k,k',i}}^{\gamma,m\ell}$$
(III.78)

In the above, A_i and B_i are defined in Eq.III.5, A_{ki} and B_{ki} are defined in Eq.III.38, and ψ_{ik} and χ_{ik} are defined by Eq.III.63. Also, we use the indices i and i' to denote single level resonances and k and k' to denote multilevel resonance.

The evaluation of the σ^* and σ^{**} which appear in Eqs.III.69-III.78 involve infinite integrals of triple products of ψ and χ functions as can be seen for example by using Eq.III.56 in Eqs.III.57 or III.62. Integrals over triple products of different indices are ignored in this process. That is, for example

$$\int_{-\infty}^{\infty} \psi_{\mathbf{i}} \sum_{\mathbf{i}'\neq\mathbf{i}}^{\mathbf{A}_{\mathbf{i}'}} \psi_{\mathbf{i}'} \sum_{\mathbf{i}'\neq\mathbf{i}}^{\mathbf{B}_{\mathbf{i}'}} \chi_{\mathbf{i}'} d\mathbf{x}_{\mathbf{i}} = \int_{-\infty}^{\infty} \psi_{\mathbf{i}} \sum_{\mathbf{i}'\neq\mathbf{i}}^{\mathbf{A}_{\mathbf{i}'}} \psi_{\mathbf{i}'} \sum_{\mathbf{i}'\neq\mathbf{i}}^{\mathbf{B}_{\mathbf{i}''}} \chi_{\mathbf{i}''} d\mathbf{x}_{\mathbf{i}}$$
$$= \int_{-\infty}^{\infty} \left[\psi_{\mathbf{i}} \sum_{\mathbf{i}'\neq\mathbf{i}}^{\mathbf{A}_{\mathbf{i}'}} \psi_{\mathbf{i}'} B_{\mathbf{i}'} \chi_{\mathbf{i}''} + \psi_{\mathbf{i}} \sum_{\mathbf{i}'\neq\mathbf{i}}^{\mathbf{A}_{\mathbf{i}'}} \psi_{\mathbf{i}''} \sum_{\mathbf{i}''\neq\mathbf{i}}^{\mathbf{B}_{\mathbf{i}''}} \chi_{\mathbf{i}''} \right] d\mathbf{x}_{\mathbf{i}}$$
$$\approx \int_{-\infty}^{\infty} \psi_{\mathbf{i}} \sum_{\mathbf{i}'\neq\mathbf{i}}^{\mathbf{A}_{\mathbf{i}'}} \psi_{\mathbf{i}'} B_{\mathbf{i}'} \chi_{\mathbf{i}'} d\mathbf{x}_{\mathbf{i}} .$$

Similarly, the integral

$$\int \psi_{k} \sum_{k' \neq k} S_{k'} A_{k'} \psi_{k'} \sum_{i} A_{ik} \psi_{i} dx_{k'}$$

will be ignored since it involves the triple index k, k', i.

As a specific example, we may write out the expression for $\sigma_{i,i',o}^{s\ell*}$ in detail, with the above approximation, as follows.

$$\sigma_{\mathbf{i},\mathbf{i}',\mathbf{o}}^{\delta\ell^{\star}} \approx \frac{\beta_{\mathbf{i}} \int_{-\infty}^{\infty} d\mathbf{x}_{\mathbf{i}} \psi_{\mathbf{i}} S}{\int_{-\infty}^{\infty} \psi_{\mathbf{i}} \sum_{\mathbf{i}'\neq\mathbf{i}}^{\Sigma} (A_{\mathbf{i}'},\psi_{\mathbf{i}'}+B_{\mathbf{i}'},\chi_{\mathbf{i}'}) d\mathbf{x}_{\mathbf{i}}}$$
(III.79)

where

$$S = 2\psi_{i} \sum_{i'\neq i}^{A_{i}} \psi_{i'} + 2 \sum_{i'\neq i}^{A_{i'}} \psi_{i'} a_{i}\chi_{i'} + \sum_{i'\neq i}^{A_{i'}} (A_{i'}\psi_{i'})^{2} + 2 \sum_{i'\neq i}^{A_{i'}} \psi_{i'} B_{i'}\chi_{i'} + 2\psi_{i'} \sum_{i'\neq i}^{A_{i'}} B_{i'}\chi_{i'} + 2 \sum_{i'\neq i}^{A_{i'}} A_{i'} \psi_{i'} A_{i'} + \sum_{i'\neq i}^{A_{i'}} (B_{i'}\chi_{i'})^{2}.$$
(III.80)

The denominators in Eq. III.79 and the corresponding expressions for the other σ^* and σ^{**} are evaluated using Eqs. III.63-III.68. The various integrals of triple products in the numerators are evaluated using the following expressions ⁽²⁵⁾ where we use the notation of Eq.III.63. Also, we use the transformation

$$\hat{\chi}_{\mathbf{k}} = \frac{1}{2} \chi_{\mathbf{k}}$$
(III.81)

$$\int_{-\infty}^{\infty} \psi_{k}^{2} \psi_{k} dx_{k} = \sqrt{\frac{\pi}{2}} \frac{\pi}{4} \theta_{k} \frac{2\Gamma_{k}}{\Gamma_{k} + \Gamma_{k}} (I_{1} + I_{2})$$
(III.82)

$$\int_{-\infty}^{\infty} \psi_{k}^{2} \hat{\chi}_{k} \, dx_{k} = \sqrt{\frac{\pi}{2}} \frac{\pi}{4} \, \theta_{k} \, \frac{2\Gamma_{k}}{\Gamma_{k} + \Gamma_{k}} \, (I_{3} + I_{4})$$
(III.83)

$$\int_{-\infty}^{\infty} \hat{\chi}_{k}^{2} \psi_{k} dx_{k} = \sqrt{\frac{\pi}{2}} \frac{\pi}{4} \theta_{k} \frac{2\Gamma_{k}}{\Gamma_{k} + \Gamma_{k}} (I_{1} - I_{2})$$
(III.84)

$$\int_{-\infty}^{\infty} \hat{\chi}_{k}^{2} \hat{\chi}_{k} \, dx_{k} = \sqrt{\frac{\pi}{2}} \frac{\pi}{4} \, \theta_{k} \, \frac{2\Gamma_{k}}{\Gamma_{k} + \Gamma_{k}} \, (I_{3} - I_{4})$$
(III.85)

$$\int_{-\infty}^{\infty} \hat{\chi}_{k} \hat{\chi}_{k}, \psi_{k}, dx_{k} = \sqrt{\frac{\pi}{2}} \frac{\pi}{4} \theta_{k} \frac{2\Gamma_{k}}{\Gamma_{k} + \Gamma_{k}} I_{2}$$
(III.86)

$$\int_{-\infty}^{\infty} \hat{\chi}_{\mathbf{k}} \psi_{\mathbf{k}} \psi_{\mathbf{k}} d\mathbf{x}_{\mathbf{k}} = -\sqrt{\frac{\pi}{2}} \frac{\pi}{4} \theta_{\mathbf{k}} \frac{2\Gamma_{\mathbf{k}'}}{\Gamma_{\mathbf{k}} + \Gamma_{\mathbf{k}'}} \mathbf{I}_{4}.$$
(III.87)

The quantities I_1 through I_4 are given by (25)

$$I_{1} = A_{4} \left[A_{5} \psi_{kk}, -A_{7} \frac{\partial \hat{\chi}_{kk}}{\partial x_{kk}} - A_{8} \frac{\partial^{2} \psi_{kk}}{\partial x_{kk}^{2}} + A_{9} \frac{\partial^{3} \hat{\chi}_{kk}}{\partial x_{kk}} \right]$$
(III.88)

$$I_{2} = \frac{2}{\sqrt{\pi}} A_{3} \left[\frac{\partial \hat{\chi}_{kk'}}{\partial x_{kk'}} - \frac{2}{3} A_{3}^{2} \frac{\partial^{2} \hat{\chi}_{kk'}}{\partial x_{kk'}} \right]$$
(III.

$$I_{3} = A_{4} \left[A_{5} \hat{\chi}_{kk} + A_{7} \frac{\partial \psi_{kk}}{\partial x_{kk}} - A_{8} \frac{\partial^{2} \hat{\chi}_{kk}}{\partial x_{kk}} - A_{9} \frac{\partial^{3} \psi_{kk}}{\partial x_{kk}} \right]$$
(III.90)

$$I_{4} = -\frac{2}{\sqrt{\pi}} A_{3} \left[\frac{\partial \psi_{kk'}}{\partial x_{kk'}} - \frac{2}{3} A_{3}^{2} \frac{\partial^{3} \psi_{kk'}}{\partial x_{kk'}^{3}} \right]$$
(III.91)

where

$$\frac{\partial \psi_{kk'}}{\partial \mathbf{x}_{kk'}} = \frac{\theta^2_{kk'}}{2} \left[\hat{\chi}_{kk'} - \mathbf{x}_{kk'} \psi_{kk'} \right]$$
(III.92)

$$\frac{\partial \chi_{kk'}}{\partial \mathbf{x}_{kk'}} = \frac{\theta^2_{kk'}}{2} \left[1 - \mathbf{x}_{kk'} \hat{\chi}_{kk'} - \psi_{kk'} \right]$$
(III.93)

$$\frac{\partial^{2} \hat{\chi}_{kk'}}{\partial x_{kk'}^{2}} = -\frac{\partial^{2} kk'}{2} \left[\hat{\chi}_{kk'} + x_{kk'} \frac{\partial \hat{\chi}_{kk'}}{\partial x_{kk'}} + \frac{\partial \psi_{kk'}}{\partial x_{kk'}} \right]$$
(III.94)

$$\frac{\partial^2 \psi_{kk'}}{\partial x^2_{kk'}} = \frac{\theta^2_{kk'}}{2} \left[\frac{\partial \hat{\chi}_{kk'}}{\partial x^2_{kk'}} - \psi_{kk'} - x_{kk'} \frac{\partial \psi_{kk'}}{\partial x_{kk'}} \right]$$
(III.95)

$$\frac{\partial^{3} \psi_{kk'}}{\partial x_{kk'}^{3}} = \frac{\partial^{2} kk'}{2} \left[\frac{\partial^{2} \hat{\chi}_{kk'}}{\partial x_{kk'}^{2}} - 2 \frac{\partial \psi_{kk'}}{\partial x_{kk'}} - x_{kk'} \frac{\partial^{2} \psi_{kk'}}{\partial x_{kk'}} \right]$$
(III.96)

$$\frac{\partial^{3}\hat{\chi}_{kk'}}{\partial x_{kk'}^{3}} = -\frac{\partial^{2}_{kk'}}{2} \left[2 \frac{\partial \hat{\chi}_{kk'}}{\partial x_{kk'}} + x_{kk'} \frac{\partial^{2} \hat{\chi}_{kk'}}{\partial x_{kk'}^{2}} + \frac{\partial^{2} \psi_{kk'}}{\partial x_{kk'}^{2}} \right]$$
(III.97)

and the coefficients A_3 through A_9 depend upon the size of θ_k .

For
$$\theta_{\rm k} \leq 5.5 \sqrt{2}$$

use is made of the rational approximation

$$e^{t^2} \operatorname{erfc}(t) \approx \sum_{i=1}^{3} \frac{a_i}{1+pt}$$

with, $a_1 = 0.3480242$; $a_2 = -0.0958798$; $a_3 = 0.7478556$; p = 0.47047 and

$$A_{3} = \frac{\frac{1}{\theta_{k}}}{\sqrt{2} (\Gamma_{k} + \Gamma_{k})}$$

Г

(III.98)

-87-

$$A_{4} = \frac{1}{1 + \frac{p\theta_{k}}{\sqrt{2}}}$$
(III.99)
$$A_{5} = a_{1} + a_{2}A_{4} + a_{3}A_{4}^{2}$$
(III.100)
$$A_{6} = pA_{3}$$

$$A_{7} = A_{4}A_{6}(a_{1} + 2a_{2}A_{4} + 3a_{3}A_{4}^{2})$$
(III.101)
$$A_{8} = (A_{4}A_{6})^{2}(a_{1} + 3a_{2}A_{4} + 6a_{3}A_{4}^{2})$$
(III.102)
$$A_{9} = (A_{4}A_{6})^{3}(a_{1} + 4a_{2}A_{4} + 10a_{3}A_{4}^{2}).$$
(III.103)

For
$$\theta_{k} > 5.5 \sqrt{2}$$

 $A_{3} = \frac{\frac{\Gamma_{k}}{\theta_{k}}}{\sqrt{2} (\Gamma_{k} + \Gamma_{k},)}$, as before
 $A_{4} = \frac{1}{\sqrt{\pi} \theta_{k}} \left\{ 1 - \left[1 - \frac{3\sqrt{2}}{2\theta_{k}} \right] / \theta_{k}^{2} \right\}$ (III.104)
 $A_{5} = 1$ (III.105)
 $A_{6} = A_{3}$ (III.105)
 $A_{7} = \frac{\sqrt{2}}{\theta_{k}} A_{6}$ (III.107)
 $A_{8} = A_{7}^{2}$ (III.108)
 $A_{9} = A_{7}^{3}$. (III.109)

Note that the expression for A_4 when $\theta_k > 5.5 \sqrt{2}$ and the break point 5.5 $\sqrt{2}$ both differ from the values reported in Hwang's original work (c.f. Eq. B.22 of Ref. 25). The values used in the code reflect the results of numerical studies aimed at optimizing execution efficiency and minimizing errors in the calculated results.

Now in the evaluation of the quantities I₁ through I₄, the ψ_{kk} , and $\hat{\chi}_{kk}$, as in other parts of MC²-2 are normally evaluated as described in Section III of Appendix A by a bivariate interpolation in prestored tables. However, if $|\mathbf{x}_{kk}, \theta_{kk}, /2| > 12$ or $\theta_{kk}, /2 > 12$, where \mathbf{x}_{kk} , and θ_{kk} , are defined in Eq.III.63, then the following asymptotic expressions are used for ψ_{kk} , $\hat{\chi}_{kk}$, and their second derivatives.

$$\psi_{kk}, \approx \frac{1}{1+x_{kk}^2} \left[1 + \frac{2}{\theta_{kk}^2}, \frac{3x_{kk}^2, -1}{(1+x_{kk}^2)^2} + \frac{12}{\theta_{kk}^4}, \frac{(1-10x_{kk}^2, +5x_{kk}^4)}{(1+x_{kk}^2)^4} \right] (III,110)$$

$$\hat{\chi}_{kk}, \ \& \frac{x_{kk'}}{1+x_{kk'}^2} \left[1 + \frac{2}{\theta_{kk}^2}, \frac{x_{kk'}^2 - 3}{(1+x_{kk'}^2)^2} + \frac{12}{\theta_{kk'}^4}, \frac{(x_{kk'}^4 - 10x_{kk'}^2 + 5)}{(1+x_{kk'}^2)^4} \right] (III.111)$$

$$\frac{\partial^2 \psi_{kk'}}{\partial x_{kk'}^2} \approx \frac{1}{1 + x_{kk'}^2} \left[2 \frac{3x_{kk'}^2 - 1}{(1 + x_{kk'}^2)^2} + \frac{24}{\theta_{kk'}^2} \frac{(1 - 10x_{kk'}^2 + 5x_{kk'}^4)}{(1 + x_{kk'}^2)^4} \right]$$
(III.112)

$$\frac{\partial^{2} \hat{\chi}_{kk'}}{\partial x_{kk'}^{2}} \approx \frac{x_{kk'}}{1 + x_{kk'}^{2}} \left[2 \frac{x_{kk'}^{2} - 3}{(1 + x_{kk'}^{2})^{2}} + \frac{24}{\theta_{kk'}^{2}} \frac{(x_{kk'}^{4} - 10x_{kk'}^{2} + 5)}{(1 + x_{kk'}^{2})^{4}} \right]. \quad (III.113)$$

After computing the asymptotic approximations to the overlap integrals. Eqs.III.69-III.78, the code tests whether these results are adequate or rather if the calculations should be performed using Gauss-Jacobi quadrature and if so, how many quadrature points should be used. The logic used in these tests is shown in Fig. 4. Note that the selection criteria depend upon the ratios of β^2/σ^* , β^2/σ^{**} , and $|0V_2/0V_1|$ for the particular overlap integral being evaluated.

The Gauss-Jacobi quadrature algorithms are described below.

3. Gauss-Jacobi Quadrature Algorithms

The various overlap integrals given in Eqs.III.23, III.24, III.31, III.32, III.33, III.36, III.37, III.42, and III.43 are evaluated using 13, 33, or 61 fixed point Gauss-Jacobi quadrature determined as indicated in Fig. 4.

Since the variable of integration is x_i , other resonances such as i' have their x_i , transformed as

$$\mathbf{x}_{i}, = \frac{\mathbf{E} - \mathbf{E}_{o_{i}}}{\frac{\Gamma_{i}}{2}} = \frac{\mathbf{E} - \mathbf{E}_{o_{i}} + \mathbf{E}_{o_{i}} - \mathbf{E}_{o_{i}}}{\frac{\Gamma_{i}}{2} \left[\frac{\Gamma_{i}}{\Gamma_{i}}\right]}$$
$$= \frac{\mathbf{x}_{i}}{\frac{\Gamma_{i}}{\Gamma_{i}}} + \frac{\mathbf{E}_{o_{i}} - \mathbf{E}_{o_{i}}}{\frac{\Gamma_{i}}{2}}$$

(III.114)

The variable of integration is transformed as in the case of the Gauss-Jacobi quadrature for the isolated resonance integrals as described in Section IV of Appendix A. Now, however, since the integration ranges from $-\infty$ to ∞ , all N Gauss-Jacobi points are used. Also, as discussed in Ref. 25, after transforming the variable of integration as in Eq.A.17 of Appendix A, a_i/β_i

-89-



Fig. 4. Gauss-Jacobi Quadrature Point Selection

-90-
is subtracted from the result to provide a better distribution of quadrature abscissae over the distribution of the integrands involved, which are peaked to the left of the origin.

After completing the integrations as indicated above, the asymmetric parts of the multilevel resonance overlap integrals, or of the total overlap integral for single level resonances, may be recomputed depending upon the testing described below.

As a specific example, we rewrite the asymmetric part of the total overlap integral for the case of all single level resonances, Eq.III.60, as

$$O_{2_{i,i',0}}^{t,\delta\ell} = \frac{1}{2} \sum_{i'\neq i} \int_{-\infty}^{\infty} dx_{i} \frac{a_{i}\chi_{i}}{\beta_{i}+\psi_{i}+a_{i}\chi_{i}} \cdot \frac{A_{i},\psi_{i'}+B_{i'}\chi_{i'}}{\beta_{i}+\psi_{i}+a_{i}\chi_{i}} + \sum_{i'\neq i} (A_{i},\psi_{i'}+B_{i'}\chi_{i'})$$
(III.115)

The quantity

SEPTST =
$$\frac{\begin{vmatrix} E_{o_i} - E_{o_i} \end{vmatrix}}{L_i + L_i}$$
(III.116)

is computed for each term of the sum in Eq.III.115, where L_i and L_i , are obtained as indicated in Eqs. 49-51. If SEPTST < 2.5, the value obtained for the Gauss-Jacobi integration of Eq.III.115 for that i' is used.

However, if SEPTST > 2.5, a special integration procedure described below is used to re-evaluate that term of the sum.

By algebraic manipulation, each term of Eq.III.115 can be written as the sum of two integrals

$$O_{2,i,i',o}^{t,\delta\ell} = \sum_{i'\neq i} \left[F_{i,i',o}^{t,\delta\ell} - S_{i,i',o}^{t,\delta\ell} \right]$$
(III.117)

where

$$F_{i,i',o}^{t,\delta\ell} = \frac{1}{2} \int_{-\infty}^{\infty} dx_{i} \frac{a_{i}\chi_{i}}{\beta_{i}} \cdot \frac{A_{i}\psi_{i}+B_{i}\chi_{i}}{\beta_{i}+\sum_{i'\neq i}(A_{i},\psi_{i'}+B_{i},\chi_{i'})}$$
(III.118)

and

$$S_{i,i',o}^{t,s\ell} = \frac{1}{2} \int_{-\infty}^{\infty} dx_{i} \frac{a_{i}\chi_{i}(\psi_{i}+a_{i}\chi_{i}) (A_{i},\psi_{i},+B_{i},\chi_{i})}{\beta_{i} \left[\beta_{i}+\psi_{i}+a_{i}\chi_{i} + \sum_{i'\neq i} (A_{i},\psi_{i'},+B_{i'},\chi_{i'})\right]}$$

$$\frac{2 \beta_{i} + \psi_{i} + a_{i} \chi_{i} + \sum_{i' \neq i} (A_{i}, \psi_{i'} + B_{i'} \chi_{i'})}{\left[\beta_{i} + \psi_{i'} + a_{i} \chi_{i'}\right] \left[\beta_{i'} + \sum_{i' \neq i} (A_{i'}, \psi_{i'} + B_{i'} \chi_{i'})\right]}$$
(III.119)

 $s_{i,i',o}^{t,\delta\ell}$ is evaluated using the Gauss-Jacobi quadrature as discussed earlier. For $f_{i,i',o}^{t,\delta\ell}$ an asymptotic algorithm similar to that described in Section C.2 is first tried. That is, we approximate $F_{i,i',o}^{t,\delta\ell}$ as

$$F_{i,i',o}^{t,\delta\ell} \approx \frac{1}{2} \int_{-\infty}^{\infty} dx_{i} \frac{a_{i}\chi_{i} \left[A_{i}, \psi_{i}, +B_{i}, \chi_{i'}\right]}{\beta_{i}^{2} + \sigma_{i,i',o}^{\delta\ell * * *}}$$
(III.120)

where

 $\sigma_{\mathbf{i},\mathbf{i}',\mathbf{o}}^{\boldsymbol{\delta\ell}***} = \frac{\int_{-\infty}^{\mathbf{dx}_{\mathbf{i}}\chi_{\mathbf{i}}} \left[A_{\mathbf{i}},\psi_{\mathbf{i}},+B_{\mathbf{i}'}\chi_{\mathbf{i}'}\right] \left[\beta_{\mathbf{i}} \sum_{\mathbf{i}'\neq\mathbf{i}} (A_{\mathbf{i}},\psi_{\mathbf{i}'},+B_{\mathbf{i}'}\chi_{\mathbf{i}'})\right]}{\int_{-\infty}^{\infty} dx_{\mathbf{i}}\chi_{\mathbf{i}} \left[A_{\mathbf{i}},\psi_{\mathbf{i}'},+B_{\mathbf{i}'}\chi_{\mathbf{i}'}\right]} \quad . \quad (\text{III.121})$

Products of triple indices are ignored in Eq.III.121 and Eq.III.120 is evaluated using Eq.III.121 and the various Eqs.III.63-III.68 and Eqs.III.82-III.108 as for the case of the earlier σ^* and σ^{**} .

A criterion T4 is then evaluated, namely

$$T4 = \frac{\beta_i^2}{\sigma \delta \ell^{***}}.$$
(III.122)

If T4 \geq 0.6 or if T \leq - 3.0 the asymptotic expression above is used for $F_{i,i',o}^{t,s\ell}$. However, if not, the criterion T5 is evaluated as

 $T5 = \frac{L_{i}}{L_{i}}$ (III.123)

If SEPTST ≤ 3.5 and T5 ≥ 1.0 , then the code reverts to the use of the original Gauss-Jacobi quadrature of Eq.III.115. However, if not, then $F_{i,\delta\ell}^{t,\delta\ell}$ is finally approximated by

$$F_{\mathbf{i},\mathbf{i}',\mathbf{o}}^{\mathbf{t},\delta\ell} \approx \frac{1}{2} \frac{a_{\mathbf{i}}}{\beta_{\mathbf{i}}} \times \left[\theta_{\mathbf{i}}, \frac{E_{\mathbf{o}_{\mathbf{i}}} - E_{\mathbf{o}_{\mathbf{i}}}}{\frac{\Gamma_{\mathbf{i}}}{2}} \right] \cdot \int_{-\infty}^{\infty} dx_{\mathbf{i}} \frac{A_{\mathbf{i}} \cdot \Psi_{\mathbf{i}} \cdot + B_{\mathbf{i}} \cdot \chi_{\mathbf{i}'}}{\beta_{\mathbf{i}} + A_{\mathbf{i}} \cdot \Psi_{\mathbf{i}'} + B_{\mathbf{i}'} \cdot \chi_{\mathbf{i}'}}$$
$$= \frac{1}{2} \frac{a_{\mathbf{i}}}{\beta_{\mathbf{i}}} \times \left[\theta_{\mathbf{i}}, \frac{E_{\mathbf{o}_{\mathbf{i}}} - E_{\mathbf{o}_{\mathbf{i}}}}{\frac{\Gamma_{\mathbf{i}}}{2}} \right] \cdot \frac{\Gamma_{\mathbf{i}'}}{\Gamma_{\mathbf{i}}} \int_{-\infty}^{\infty} dx_{\mathbf{i}'}, \frac{\Psi_{\mathbf{i}'} + a_{\mathbf{i}'} \cdot \chi_{\mathbf{i}'}}{\beta_{\mathbf{i}'} + \Psi_{\mathbf{i}'} + a_{\mathbf{i}'} \cdot \chi_{\mathbf{i}'}}$$
$$= \frac{a_{\mathbf{i}}}{\beta_{\mathbf{i}}} \times \left[\theta_{\mathbf{i}}, \frac{E_{\mathbf{o}_{\mathbf{i}'}} - E_{\mathbf{o}_{\mathbf{i}}}}{\frac{\Gamma_{\mathbf{i}}}{2}} \right] \cdot \frac{\Gamma_{\mathbf{i}'}}{\Gamma_{\mathbf{i}}} J(\beta_{\mathbf{i}'}, \theta_{\mathbf{i}'}, a_{\mathbf{i}'}, a_{\mathbf{i}'}). \quad (III.124)$$

-93-

This procedure is similarly used on all the other overlap integrals having an asymmetric part, that is a part proportional to $\chi.$

IV. CALCULATION OF UNRESOLVED RESONANCE INTEGRALS

A. General Formulation

The algorithms involved in the unresolved resonance integral calculation developed by R. Hwang⁽²⁵⁾ assume the narrow resonance approximation and account for interference scattering, the effects of accidental overlap with resonances in other spin sequences, and the effects of self-overlap with resonances of the same spin sequence. The single level Breit-Wigner representation is used for the resonance cross sections.

The effective macroscopic capture cross section for material m within an arbitrary energy interval $E_2 - E_1$ can be written as

$$\overline{\Sigma}_{c}^{m}(E^{*}) = N^{m} \overline{\sigma}_{c}^{m}(E^{*}) = \frac{\frac{1}{E_{2} - E_{1}} \int_{E_{1}}^{E_{2}} \frac{N^{m} \sigma_{c}^{m}(E) dE}{\Sigma_{t}(E)}}{\frac{1}{E_{2} - E_{1}} \int_{E_{1}}^{E_{2}} \frac{dE}{\Sigma_{t}(E)}}$$
(IV.1)

where E* is an energy point within the interval $E_2 - E_1$. Eq. IV.1 is equivalent to Eq. III.1. N^m is the atom density of material m, σ_c^m is the microscopic capture cross section for material m, Σ_t is the total macroscopic cross section for the mixture, and as in the case of the resolved resonance integral calculations we assume that the narrow resonance approximation is valid. In order to satisfy the statistical criteria described below we assume no significant attenuation of flux in the energy interval $E_2^{-E_1}$ and a constant collision density.

 σ_c^m and Σ_t correspond to sums over contributing resonances belonging to various spin sequences, that is, resonances having a particular angular momentum and channel spin. If we separate the total cross section in Eq. IV.1 into a resonant part Σ_r (E) and a remaining non-resonant part Σ_r , we can rewrite Eq. IV.1 as

$$\overline{\Sigma}_{c}^{m}(E^{*}) = \frac{\frac{1}{E_{2} - E_{1}} \int_{E_{1}}^{E_{2}} \frac{N^{m} \sum \sigma^{m}(E) dE}{\sum \sum \Gamma_{r_{i}}^{m}(E) + \Sigma_{p}}}{\frac{1}{E_{2} - E_{1}} \int_{-\infty}^{\infty} \frac{dE}{\sum \sum \Gamma_{r_{i}}^{m}(E) + \Sigma_{p}}}.$$
(IV.2)

In Eq. IV.2, s represents a particular spin sequence and i represents the resonances in that sequence. The sums in the numerator of the upper integral ranges only over those sequences belonging to material m while the other sums are over all materials.

Now the interval $E_2 - E_1$ may be replaced by the average spacing of the resonances of the particular sequence being considered, $\langle D_s \rangle$, times the number of resonances contained within the interval, N_s. Also, on the assumption that a large number of resonances i exist in 'the interval, we may replace $\frac{1}{N_s} \sum_{i}$ by an integration over the chi-squared distribution with μ degrees of freedom as indicated in Eq. A.30 of Appendix A. This latter step corresponds to providing a statistical average over the distribution functions of the resonance parameters. Thus, as an example, we may write

$$\lim_{\Delta E \to \infty} \frac{1}{\Delta E} \sum_{i} f(\Gamma_{n}^{i}) = \frac{1}{\langle D \rangle} \int_{0}^{\infty} P_{\mu}(x) f(x \overline{\Gamma_{n}(E^{*})}) dx = \frac{1}{\langle D \rangle} \langle f \rangle_{E^{*}}$$
(IV.3)

where x corresponds to the ratio of neutron width at energy E to mean neutron width at E*, $\overline{\Gamma_n(E^*)}$ is the mean neutron width at energy E*, and μ is the number of entrance channels for neutrons of the particular spin sequence under consideration. P_µ is the chi-squared distribution of order μ given in Eq. A.31 of Appendix A. The fission width would be similarly averaged with perhaps a different number of exit channels as appropriate to the data involved. The angular brackets $\langle \rangle$ will be used to designate the expectation value due to the statistical integration.

The mean neutron width at E* for neutrons of angular momentum ℓ and total channel spin J is given by

$$\overline{\Gamma_{n}(\mathbf{E}^{\star})}_{\ell,J} = \overline{\Gamma_{n}^{o}(\mathbf{E}^{\star})}_{\ell,J} \sqrt{\mathbf{E}^{\star}} V_{\ell}^{\mu}_{\ell,J}$$
(IV.4)

where $\Gamma_n^o(E^*)_{\ell,J}$ is the reduced neutron width at energy E^* , V_ℓ is the penetration factor for neutrons of angular momentum ℓ , and $\mu_{\ell,J}$ is the number of entrance channels for neutrons of the ℓ, J sequence.

$$V_{0} = 1$$

 $V_{1} = \frac{n^{2}}{1 + n^{2}}$
 $V_{2} = \frac{n^{4}}{9 + 3n^{2} + n^{4}}$

 $\frac{R}{1+x}$

n

(IV.5)

where R is the channel radius and λ^* is the reduced neutron wavelength at energy E*.

Resonances in different spin sequences are assumed to be completely uncorrelated whereas resonances in a given spin sequence are assumed to be distributed according to the Dyson two-level correlation function²⁵

$$\Omega(\mathbf{y}) = 1 - \{\xi(\mathbf{y})\}^2 + \frac{\partial \xi(\mathbf{y})}{\partial \mathbf{y}} \quad \mathrm{si}(\mathbf{y}) \tag{IV.6}$$

where

$$y = \frac{\pi |E_{o_k} - E_{o_k}|}{\langle D \rangle}$$
(IV.7)

$$\xi(y) = \frac{\sin|y|}{y}$$
 (IV.8)

$$si(y) = -\int_{y}^{\infty} \frac{\sin t}{t} dt . \qquad (IV.9)$$

Eqs. III.3 and III.4 express the resolved resonance capture and total cross sections in terms of the symmetric and antisymmetric Doppler broadened line shapes ψ and χ . Eq. III.3 is still appropriate for the unresolved region but the low energy limit form of Eq. III.4 must be rewritten for use here as

$$\sigma_{t_{i}} = \cos 2\delta_{\ell_{i}} \left[\frac{\sigma_{p_{i}}}{\cos 2\delta_{\ell_{i}}} + \sigma_{o_{i}}\psi_{i} + \sigma_{o_{i}}a_{i}\chi_{i} \right]$$
(IV.10)

where δ_{ℓ_i} is the phase angle for angular momentum ℓ given by

$$\delta_{0} = m$$

$$\delta_{1} = m - \arctan m$$

$$\delta_{2} = m - \arctan \left[\frac{3m}{3 - m^{2}}\right]$$

$$m = \frac{R}{\lambda \star} .$$
(IV.11)

R is the effective scattering radius and λ^* is the reduced neutron wavelength at energy E*. ψ_i and χ_i are the usual shorthand notation for $\psi(\theta_i,\chi_i)$ and $\chi(\theta_i,\chi_i)$.

In Eq. IV.10, the interference factor a_i is given by

$$a_{i} = \frac{1}{2} \tan(2\delta_{l_{i}}) \quad .$$

(IV.12)

If we express Eq. IV.2 in terms of the ψ and χ functions, perform the statistical average over the distribution functions for the resonance parameters, and factor the resulting equation in a manner typified by Eq. III.23, we may finally write for the expectation value for a given spin sequence

$$\frac{\cos 2\delta_{\ell}}{\langle \mathbf{D}_{k} \rangle} \left\langle \Gamma_{\gamma_{k}} \mathbf{J}_{k}^{\star} \right\rangle^{\frac{1}{2}} \left\{ \frac{1}{\langle \mathbf{D}_{k} \rangle} \left\langle \Gamma_{\gamma_{s}} \mathbf{J}(\beta_{k}, \theta_{k}, a_{k}, 0) \right\rangle - \mathbf{O}_{\gamma_{k}} \right\} \cdot \mathbf{S} + \sum_{i \neq k} r_{\gamma_{ki}} \cdot (\mathbf{IV}.13)$$

In Eq.IV.13, J is defined in Eqs.III.44-III.48 and β_k is defined as in Eq. III.5 except divided by $\cos 2\delta_{\ell}$. S represents the first-order correction for the accidental overlap with the uncorrelated resonances in spin sequences i \neq k given by²⁵

$$S = 1 - \sum_{i \neq k} \left[\frac{1}{\langle D_i \rangle} \left\langle \Gamma_{t_i} J(\beta_i, \theta_i, a_i, a_i) \right\rangle - O_{t_i} \right]. \quad (IV.14)$$

The r represent the higher order corrections for the accidental overlap γ_{ki}

effect which, to second order is approximately given by²⁵

$$\mathbf{r}_{\gamma_{ki}} \stackrel{\sim}{=} \frac{\langle \tau_{\gamma_{k}} \rangle}{\langle \mathbf{D}_{k} \rangle} \frac{1}{\langle \mathbf{D}_{i} \rangle} \left\langle \Gamma_{\mathbf{t}_{i}}^{\beta_{i}} \frac{\partial J(\beta_{i}, \theta_{i}, 0, 0)}{\partial \beta_{i}} \right\rangle$$
(IV.15)

$$\left\langle \tau_{\gamma_{k}} \right\rangle = \left\langle \frac{\Gamma_{\gamma_{k}}}{2\beta_{k}} \int_{-\infty}^{\infty} \frac{\psi^{2}(\theta_{k}, \mathbf{x}_{k}) d\mathbf{x}_{k}}{\beta_{k} + \psi(\theta_{k}, \mathbf{x}_{k})} \right\rangle .$$
(IV.16)

 MC^2 -2 assumes that r in Eq.IV.13 is negligible.

0 represents the capture self-overlap term for resonances of the same spin sequence k and is approximated by

$$O_{\gamma_{k}} \stackrel{\simeq}{=} \frac{1}{\langle D_{k} \rangle} \left\langle \frac{\Gamma_{\gamma_{k}}}{2} \int_{-\infty}^{\infty} \Omega(\delta) \frac{d\delta}{\langle D_{k} \rangle} \cdot \int_{-\infty}^{\infty} \frac{\psi_{k}}{\beta_{k} + \psi_{k}} \frac{A_{k} \psi_{k} dx_{k}}{\beta_{k} + \psi_{k} + k} \right\rangle_{k \text{ and } k'} \cdot (IV.17)$$

In Eq.IV.17 the resonances k and k' all belong to the same spin sequence k. $\Omega(\delta)$ is the probability of finding a resonance k' at a distance $\delta = E - E - E_{k} - E$ In deriving Eq.IV.13, it has been explicitly assumed that products of more than two ψ or χ functions can be neglected. This is the so-called "nearest neighbor" approximation.²⁸ As a simple example, consider

$$\frac{\sigma_{\mathbf{r}}^{1}}{\sigma_{\mathbf{r}}^{1} + \sigma_{\mathbf{r}}^{2} + \sigma_{\mathbf{r}}^{3} + \sigma_{\mathbf{p}}} = \frac{\sigma_{\mathbf{r}}^{2}}{\sigma_{\mathbf{r}}^{1} + \sigma_{\mathbf{p}}} - \frac{\sigma_{\mathbf{r}}^{1}}{\sigma_{\mathbf{r}}^{1} + \sigma_{\mathbf{p}}} \frac{\sigma_{\mathbf{r}}^{2} + \sigma_{\mathbf{r}}^{3}}{\sigma_{\mathbf{r}}^{1} + \sigma_{\mathbf{r}}^{2} + \sigma_{\mathbf{r}}^{3} + \sigma_{\mathbf{p}}}$$

$$= \frac{\sigma_{\mathbf{r}}^{1}}{\sigma_{\mathbf{r}}^{1} + \sigma_{\mathbf{p}}^{2}} - \frac{\sigma_{\mathbf{r}}^{1}}{\sigma_{\mathbf{r}}^{1} + \sigma_{\mathbf{p}}^{2}} \left\{ \frac{\sigma_{\mathbf{r}}^{2}}{\sigma_{\mathbf{r}}^{2} + \sigma_{\mathbf{p}}^{2}} - \frac{\sigma_{\mathbf{r}}^{2}}{\sigma_{\mathbf{r}}^{2} + \sigma_{\mathbf{p}}^{2}} \frac{\sigma_{\mathbf{r}}^{1} + \sigma_{\mathbf{r}}^{3} + \sigma_{\mathbf{p}}^{3}}{\sigma_{\mathbf{r}}^{1} + \sigma_{\mathbf{r}}^{2} + \sigma_{\mathbf{r}}^{3} + \sigma_{\mathbf{p}}^{3}} \right\}$$

$$+ \frac{\sigma_{\mathbf{r}}^{3}}{\sigma_{\mathbf{r}}^{3} + \sigma_{\mathbf{p}}^{2}} - \frac{\sigma_{\mathbf{r}}^{3}}{\sigma_{\mathbf{r}}^{3} + \sigma_{\mathbf{p}}^{2}} \frac{\sigma_{\mathbf{r}}^{2} + \sigma_{\mathbf{r}}^{3} + \sigma_{\mathbf{p}}^{3}}{\sigma_{\mathbf{r}}^{1} + \sigma_{\mathbf{r}}^{2} + \sigma_{\mathbf{r}}^{3} + \sigma_{\mathbf{p}}^{3}} \right\}$$

$$\sum_{\mathbf{v}} \frac{\sigma_{\mathbf{r}}^{1}}{\sigma_{\mathbf{r}}^{1} + \sigma_{\mathbf{p}}^{2}} \left\{ 1 - \frac{\sigma_{\mathbf{r}}^{2}}{\sigma_{\mathbf{r}}^{2} + \sigma_{\mathbf{p}}^{2}} - \frac{\sigma_{\mathbf{r}}^{3}}{\sigma_{\mathbf{r}}^{3} + \sigma_{\mathbf{p}}^{3}} \right\}$$
(IV.18)

We shall also later use the fact that Eq.IV.18 can be approximated by

$$\frac{\sigma_{\mathbf{r}}^{1}}{\sigma_{\mathbf{r}}^{1} + \sigma_{\mathbf{r}}^{2} + \sigma_{\mathbf{r}}^{3} + \sigma_{\mathbf{p}}} \stackrel{'\nu}{\sim} \frac{\sigma_{\mathbf{r}}^{1}}{\sigma_{\mathbf{r}}^{1} + \sigma_{\mathbf{p}}} \left\{ 1 - \frac{\sigma_{\mathbf{r}}^{2}}{\sigma_{\mathbf{r}}^{2} + \sigma_{\mathbf{p}}} \right\} \left\{ 1 - \frac{\sigma_{\mathbf{r}}^{3}}{\sigma_{\mathbf{r}}^{3} + \sigma_{\mathbf{p}}} \right\}$$
(IV.19)

where again we have neglected triple products. Eq.IV.19 makes use of the general first order approximation

$$1 - \sum_{i} B_{i} \sim \prod_{i} (1 - B_{i}) . \qquad (IV.20)$$

The expression corresponding to Eq.IV.13 for fission simply requires replacing Γ_{γ_k} with Γ_{f_k} so that 0_{γ_k} , $r_{\gamma_{ki}}$, and $\langle \tau_{\gamma_k} \rangle$ become respectively 0_{f_k} , $r_{f_{ki}}$, and $\langle \tau_{f_k} \rangle$. In the case of the total reaction, Γ_{t_k} replaces Γ_{γ_k} and $J(\beta_k, \theta_k, a_k, a_k)$ is used in Eq.IV.13. Also, in this case the factor $\cos 2\delta_{\varrho}$ does not appear in Eq.IV.13.

Since the J integrals and other related integrals can be readily evaluated as described in Section B of Chapter III with the statistical averaging accomplished as specified in Section V Appendix A, the main computational effort is related to the evaluation of the self-overlap integrals. , As in the case of the resolved resonances, Eq.IV.13 is evaluated for a homogeneous mixture, and for heterogeneous slab geometries or cylindrical geometries with the value used for Σ and hence for β determined as specified in Eqs.III.16-III.20.

B. Evaluation of the Self-Overlap Term

In order to evaluate Eq. IV.17, we note that

$$\int_{-\infty}^{\infty} \frac{\psi_{k}}{\beta_{k} + \psi_{k}} \frac{A_{k} \psi_{k}}{\beta_{k} + \psi_{k} + A_{k} \psi_{k}} dx_{k} =$$

$$\int_{-\infty}^{\infty} \frac{\psi_{k}}{\beta_{k} + \psi_{k}} \left\{ \frac{\psi_{k}}{\beta_{k} + \psi_{k}} - \frac{A_{k} \psi_{k} \psi_{k}}{(\beta_{k} + \psi_{k})^{2}} + \ldots \right\}$$
(IV.21)

provided the resulting integrals are uniformly convergent. In Eq.IV.21, $A_k = 1/A_k$. Substituting Eq.IV.21 into Eq.IV.17, Hwang has shown²⁵ that

$$O_{\gamma_k} = K_1 - K_2 + \dots$$
 (IV.22)

where $K_1 >> K_2$ if the self-shielding effect is relatively weak.

$$K_{1} = \frac{1}{\langle D_{k} \rangle^{2}} \left\langle \Gamma_{\gamma_{k}} J_{k} \right\rangle \left\langle \Gamma_{k}, J_{k}, \right\rangle - L_{1}$$
(IV.23)

$$L_{1} = \frac{1}{\langle D_{k} \rangle} \int_{-\infty}^{\infty} W\left(\frac{\delta}{\langle D_{k} \rangle}\right) \frac{d\delta}{\langle D_{k} \rangle}$$

$$\cdot \int_{-\infty}^{\infty} \left\langle \frac{\Gamma_{\gamma_{k}}}{2} \frac{\psi_{k}}{\beta_{k} + \psi_{k}} \right\rangle_{k} \left\langle \frac{\psi_{k}}{\beta_{k} + \psi_{k}} \right\rangle_{k} dx_{k} \qquad (IV.24)$$

$$K_2 = -r_{\gamma_{kk}} - L_2 \qquad (IV.25)$$

where $r_{\gamma_{kk}}$ is given by Eqs.IV.15 and IV.16 and

$$L_{2} = -\frac{1}{\langle D_{k} \rangle} \int_{-\infty}^{\infty} W\left(\frac{\delta}{\langle D_{k} \rangle}\right) \frac{d\delta}{\langle D_{k} \rangle} \int_{-\infty}^{\infty} \left\langle \frac{\Gamma_{\gamma_{k}}}{2} \frac{\psi_{k}^{2}}{\beta_{k}(\beta_{k} + \psi_{k})} \right\rangle_{k}$$
$$\cdot \left\langle \beta_{k}, \frac{\partial}{\partial \beta_{k}}, \frac{\psi_{k}}{\beta_{k}, + \psi_{k}} \right\rangle_{k} dx_{k} . \qquad (IV.26)$$

In Eq.IV.24 and IV.25

$$W\left(\frac{\delta}{\langle D_{k} \rangle}\right) = 1 - \Omega\left(\frac{\delta}{\langle D_{k} \rangle}\right)$$
(IV.27)

and in Eq.IV.26

$$-\frac{\partial}{\partial_{\beta_k}} \left(\frac{\psi_k}{\beta_k + \psi_k} \right) = \frac{\psi_k}{(\beta_k + \psi_k)^2} \quad . \qquad (IV.28)$$

The computational effort now is centered on the evaluation of the term L_1 since L_2 can easily be obtained once L_1 is known.

1. Evaluation of L_1

Using the Fourier transform technique, Hwang has shown that Eq.IV.24 can be written in the form

$$L_{1} = \frac{1}{2\langle D_{k} \rangle^{2}} \int_{-\infty}^{\infty} \sqrt{2\pi} \omega(\xi) \left\langle \Gamma_{\gamma_{k}} P_{k} \left(\frac{\Gamma_{t_{k}}}{2} \xi \right) \right\rangle_{k}$$
$$\cdot \left\langle \frac{\Gamma_{t_{k'}}}{2} P_{k'} \left(\frac{\Gamma_{t_{k'}}}{2} \xi \right) \right\rangle_{k'} d\xi \qquad (IV.29)$$

where P_k is the Fourier transform

ω

$$P_{k}(\xi) = \frac{1}{\sqrt{2\pi}} \int_{-\infty}^{\infty} \frac{\psi_{k} \exp(i\xi x_{k})}{\beta_{k} + \psi_{k}} dx_{k} = F\left\{\frac{\psi_{k}}{\beta_{k} + \psi_{k}}\right\}$$
(IV.30)

and $\omega(\xi)$, the Fourier transform of the Dyson function, is given

$$\begin{aligned} (\xi) &= \frac{\langle \mathbf{D}_{k} \rangle}{\sqrt{2\pi}} \left\{ 1 - \left| \frac{\langle \mathbf{D}_{k} \rangle \xi}{\pi} \right| + \left| \frac{\langle \mathbf{D}_{k} \rangle \xi}{2\pi} \right| \ell_{n} \left[1 + \left| \frac{\langle \mathbf{D}_{k} \rangle \xi}{\pi} \right| \right] \right\}; \\ &\qquad \left| \frac{\langle \mathbf{D}_{k} \rangle \xi}{2\pi} \right| \leq 1 \\ &= \frac{\langle \mathbf{D}_{k} \rangle}{\sqrt{2\pi}} \left\{ -1 + \left| \frac{\langle \mathbf{D}_{k} \rangle \xi}{2\pi} \right| \ell_{n} \left[\frac{\left| \langle \mathbf{D}_{k} \rangle \xi / \pi \right| + 1}{\left| \langle \mathbf{D}_{k} \rangle \xi / \pi \right| - 1} \right] \right\}; \end{aligned}$$
(IV.31)
$$\left| \frac{\langle \mathbf{D}_{k} \rangle \xi}{2\pi} \right| > 1 . \end{aligned}$$

As usual, in Eq.IV.29 the quantities inside the angular brackets are the statistically averaged values over the appropriate chi-squared distribution functions of the resonance parameters, and the resonances k and k' belong to the same spin sequence.

Now we make a change of variable to convert Eq.IV.29 into a form amenable to Gauss-Hermite quadrature. If we let

$$\eta = \alpha \xi$$

$$\alpha = \sqrt{\frac{\langle D_k \rangle^2}{\pi}^2 + \frac{\Delta^2}{2} + \gamma^2}$$
(IV.32)
$$\gamma = \left\langle \Gamma_{t_k} \sqrt{\frac{\beta_k + 1}{\beta_k}} \right\rangle$$

and multiplying the integral in Eq.IV.29 by $exp(-\eta^2) \cdot exp(\eta^2)$, Eq.IV.29 can be converted into the equivalent form

$$L_{1} = \frac{1}{2\langle D_{k} \rangle^{2}} \frac{1}{\alpha} \int_{-\infty}^{\infty} e^{-\eta^{2}} \left\{ \sqrt{2\pi} \omega \left(\frac{\eta}{\alpha} \right) \left\langle \prod_{\gamma k} e^{\frac{\Delta^{2} \eta^{2}}{4\alpha^{2}}} P_{k} \left(\frac{\Gamma_{k}}{2\alpha} \right) \right\rangle_{k} \right\}$$
$$\cdot \left\langle \frac{\Gamma_{k}}{2} e^{\frac{\Delta^{2} \eta^{2}}{4\alpha^{2}}} P_{k} \left(\frac{\Gamma_{k}}{2\alpha} \right) \right\rangle_{k} e^{\left[\frac{\langle D_{k} \rangle^{2}}{\pi^{2} \alpha^{2}} + \frac{\gamma^{2}}{\alpha^{2}} \right]} \eta^{2} \right\} d\eta \quad . \quad (IV.33)$$

Eq.IV.33 is of the form

$$\int_{-\infty}^{\infty} e^{-y^2} f(y) dy = \sum_{i=1}^{N} a_i f(y_i) + R_N$$
 (IV.34)

where a, and y, are the weights and zeros of the Hermite polynomial and Eq.IV.34 corresponds to the usual Gauss-Hermite quadrature with the remainder term R_N . Since P_k and ω are each symmetric, only a sum over the positive zeros y, is required to evaluate Eq.IV.33. The code uses N = 10 for the quadrature. Specifically, with N = 10 we assume

$$L_{1} = \frac{1}{2\langle D_{k} \rangle^{2}} \frac{2}{\alpha} \sum_{i=1}^{N/2} a_{i} \sqrt{2\pi} \omega \left(\frac{\eta_{i}}{\alpha}\right) \left\langle \prod_{\gamma_{k}} e^{\frac{1^{2}\eta_{i}^{2}}{4\alpha^{2}}} P_{k} \left(\frac{\prod_{k}}{2\alpha}\right) \right\rangle_{k}$$

$$\cdot \left\langle \frac{\Gamma_{\mathbf{t}_{\mathbf{k}'}}}{2} e^{\frac{\Delta^2 \eta_{\mathbf{i}}^2}{4\alpha^2}} P_{\mathbf{k}'} \left(\frac{\Gamma_{\mathbf{t}_{\mathbf{k}'}} \eta_{\mathbf{i}}}{2\alpha} \right) \right\rangle_{\mathbf{k}'} e^{\left[\frac{\langle \mathbf{p}_{\mathbf{k}} \rangle^2}{\pi^2 \alpha^2} + \frac{\gamma^2}{\alpha^2} \right] \eta_{\mathbf{i}}^2}$$
(IV.35)

The evaluation of Eq.IV.35 depends upon the availability of the Fourier transforms $P_k(\Gamma_t \xi/2)$. Two algorithms are used depending upon the size of β_k relative to $\psi(\theta_k, 0)$. The large β approximation or asymptotic algorithms provide significant computational economy as compared with the non-asymptotic algorithms. Since β_k is usually large compared to $\psi(\theta_k, 0)$ for many of the unresolved resonances, the large β approximation is frequently invoked for realistic problems with subsequent savings in execution time.

a. Asymptotic Algorithms for L1

If $(\beta_k + \psi(\theta_k, 0))/\psi(\theta_k, 0) \ge 2.5$, P_k may be approximated by²⁵

$$\mathbb{P}_{k}\left(\frac{\Gamma_{\mathbf{t}_{k}}}{2}\right) \stackrel{\sim}{=} \sqrt{\frac{\pi}{2}} \exp\left[-\frac{\Delta^{2}}{4}\xi^{2} - \frac{\Gamma_{\mathbf{t}_{k}}}{2}|\xi|\right] \cdot \mathbb{Q}_{k}(\xi)/(\beta_{k} + \rho) + \dots \quad (IV.36)$$

where Δ is the Doppler line width as defined below Eq.III.4 with E replaced by E*.

$$Q_{k}(\xi) = 1 + \frac{\rho - U}{\beta_{k} + \rho}$$
 (IV.37)

$$U_{k} = \frac{\theta_{k}\sqrt{\pi}}{2\sqrt{2}} \left[\exp\left(\frac{\Delta|\xi|}{2\sqrt{2}} + \frac{\theta_{k}}{\sqrt{2}}\right)^{2} \operatorname{Erfc}\left(\frac{\Delta|\xi|}{2\sqrt{2}} + \frac{\theta_{k}}{\sqrt{2}}\right) + \exp\left(\frac{\Delta^{2}\xi^{2}}{8}\right) \operatorname{E_{1}f}\left(\frac{\Delta|\xi|}{2\sqrt{2}}\right) \right]$$
(IV. 38)
$$\rho = \frac{1}{2} \psi(\sqrt{2}\theta_{k}, 0)$$
(IV. 39)

and Erfc and Erf are respectively the complementary error function and the error function which are evaluated as specified for Eq.A.15 of Appendix A.

The asymptotic evaluation of L_1 is then completed with the variable change of Eq.IV.32 and substitution of Eq.IV.36 into Eq.IV.35.

-102-

b. <u>Non-Asymptotic Algorithms for L</u>1

When $(\beta_k + \psi(\theta_k, 0))/\psi(\theta_k, 0) < 2.5$, the more elaborate algorithms described below must be used to obtain the P_k for use in the solution of Eq.IV.35. $\Gamma_{t} \xi$

 $P_k\left(\frac{\Gamma_k}{2}\right)$ is the unique solution of the integral equation

$$\beta_{k}P_{k}\left(\frac{\Gamma_{t_{k}}\xi}{2}\right) + \frac{1}{4}\Gamma_{t_{k}}\int_{-\infty}^{\infty} e^{-\frac{\Lambda^{2}}{4}(\xi - t)^{2} - \frac{\Gamma_{t_{k}}}{2}|\xi - t|}P_{k}\left(\frac{\Gamma_{t_{k}}t}{2}\right)dt$$
$$= \sqrt{\frac{\pi}{2}}e^{-\frac{\Lambda^{2}\xi^{2}}{4} - \frac{\Gamma_{t_{k}}|\xi|}{2}}.$$
 (IV.40)

Making the substitutions $y = \alpha \xi$ and $x = \alpha t$ and multiplying the integral by $exp(-x^2) \cdot exp(x^2)$, Eq.IV.40 may be cast into the form suitable for Gauss-Hermite quadrature

$$e^{\frac{\Delta^{2} \mathbf{y}^{2}}{4\alpha^{2}}} \beta_{k} P_{k} \left(\frac{\Gamma_{k}}{2\alpha}\right) + \frac{1}{4} \frac{\Gamma_{k}}{\alpha} \int_{-\infty}^{\infty} e^{-\mathbf{x}^{2}} \cdot e^{\frac{\Delta^{2} \mathbf{y} \mathbf{x}}{2\alpha\alpha} - \frac{\Delta^{2}}{4} \frac{\mathbf{x}^{2}}{\alpha^{2}} - \frac{\Gamma_{k}}{2} \left|\frac{\mathbf{y} - \mathbf{x}}{\alpha}\right| + \mathbf{x}^{2}}_{P_{k}} \left(\frac{\Gamma_{k}}{2\alpha}\right)_{d\mathbf{x}}$$
$$= \sqrt{\frac{\pi}{2}} e^{-\frac{\Gamma_{k}}{2} \left|\frac{\mathbf{y}}{\alpha}\right|} \cdot \left(1 \nabla \cdot 41\right)$$

The integral in Eq. IV.41 can again be written as a ten point Gauss-Hermite quadrature, so that at each mesh point y_i we have

$$\frac{\Delta^{2} y_{i}^{2}}{e^{4\alpha^{2}}} \beta_{k} P_{k} \left(\frac{\Gamma_{t_{k}} y_{i}}{2\alpha} \right) + \frac{1}{4} \frac{\Gamma_{t_{k}}}{\alpha} \sum_{i=1}^{10} a_{j} e^{\frac{\Delta^{2}}{2} \frac{y_{i}}{\alpha} \frac{x_{i}}{\alpha}} - \frac{\Delta^{2}}{4} \frac{x_{i}^{2}}{\alpha^{2}} - \frac{\Gamma_{t_{k}}}{2} \left| \frac{y_{i} - x_{i}}{\alpha} \right| + x_{j}^{2} P_{k} \left(\frac{\Gamma_{t_{k}} y_{j}}{2\alpha} \right)$$
$$= \sqrt{\frac{\pi}{2}} e^{-\frac{\Gamma_{t_{k}}}{2} \left| \frac{y_{i}}{\alpha} \right|} . \qquad (IV.42)$$

Equation IV.42 can be looked at as a system of 10 equations in the ten unknowns

$$P_k\left(\frac{\Gamma_{t_k}y_i}{2\alpha}\right).$$

Thus we may write the matrix equation

A P = B

(IV.43)

so that the desired P may be obtained by inversion of the A matrix

$$P = A^{-1}B.$$
 (IV.44)

Eqs.IV.42 are normalized by dividing each row of the A matrix by the respective diagonal element, namely

$$N_{i}^{o} = e^{\frac{\Delta^{2} y_{i}^{2}}{4\alpha^{2}}} \left\{ \beta_{k} + \frac{\Gamma_{t_{k}}}{4\alpha} a_{i} e^{y_{i}^{2}} \right\} . \qquad (IV.45)$$

Therefore

$$B_{i} = \frac{\sqrt{\frac{\pi}{2}} e^{-\frac{C_{k}}{2} \left| \frac{y_{i}}{\alpha} \right|}}{N_{i}^{o}}$$
(IV.46)

$$A_{ii} = 1$$
(IV.47)

$$A_{ij} = \left\{ \frac{\Gamma_{k} a_{j}}{4\alpha} e^{\frac{\Delta^{2} y_{i} x_{j}}{2\alpha^{2}} - \frac{\Gamma_{k} |y_{i} - x_{j}|}{2\alpha}}{e^{\frac{\Delta^{2} x_{j}^{2}}{4\alpha^{2}}} e^{\left[\frac{\Delta b_{k}^{2}}{\pi^{2} \alpha^{2}} + \frac{\gamma^{2}}{\alpha^{2}}\right] x_{j}^{2}} \right\} / N_{i}^{O}$$
(IV.48)

In Eq. IV.48 we have made use of the identity

$$e^{x_{j}^{2}} = e^{\frac{\Delta^{2}x_{j}^{2}}{2\alpha^{2}} + \frac{\langle D_{k}\rangle^{2}x_{j}^{2}}{\pi^{2}\alpha^{2}} + \frac{\gamma^{2}x_{j}^{2}}{\alpha^{2}}}$$
(IV.49)

with α and γ defined as before in Eqs. IV. 32.

The inversion of the 10 x 10 matrix A can be considerably simplified and the execution time thereby significantly reduced by partitioning A into four sub-matrixes and recognizing the fact that A = A as is obvious from Eq. IV.48. If we let D represent the inverse of A, we can write

$$AA^{-1} = \begin{bmatrix} A_{11} & A_{12} \\ A_{21} & A_{22} \end{bmatrix} \begin{bmatrix} D_{11} & D_{12} \\ D_{21} & D_{22} \end{bmatrix} = I$$
(IV.50)

where I is the 10 x 10 identity matrix and each of the submatrices is 5 x 5.

By algebraic manipulation of the equations represented by Eq.IV.50, we can easily show that

```
D_{11} = (A_{11} - A_{12}A_{11}^{-1}A_{12})^{-1}
```

 $D_{12} = -A_{11}^{-1}A_{12}D_{11}$

(IV.51)

 $D_{22} = D_{11}$

 $D_{21} = D_{12}$

where we have made use of the fact that $A_{11} = A_{22}$ and $A_{12} = A_{21}$.

Thus the inversion of the 10 x 10 A matrix can be accomplished by the much faster inversion of two 5 x 5 matrices as shown in Eqs. IV.51.

Finally, taking advantage of the fact that P_k is symmetric, we need be concerned only with the positive Gauss-Hermite quadrature points so that we may obtain the upper half of the P vector from $(D_{11} D_{12})B$.

The non-asymptotic evaluation of L_1 is then completed using the P_k obtained as above in Eq.IV.35.

2. Evaluation of L_2

Using the same technique as was used to convert Eq.IV.24 into the form of Eq.IV.29 for L_1 , Hwang has converted Eq.IV.26 into the form

 $L_{2} = -\frac{1}{2\langle D_{k} \rangle^{2}} \int_{-\infty}^{\infty} \sqrt{2\pi} \omega(\xi) \left\langle \Gamma_{\gamma_{k}} \sqrt{\frac{\pi}{2}} \frac{e^{-\frac{\Delta^{2}\xi^{2}}{4} - \frac{\Gamma_{t_{k}}}{2}}}{\beta_{k}} - P_{k} \left(\frac{\Gamma_{t_{k}}}{2}\right) \right\rangle_{k}$

$$\cdot \left\langle \frac{\Gamma_{\mathbf{t}_{\mathbf{k}}}}{2} \beta_{\mathbf{k}}, \frac{\partial}{\partial \beta_{\mathbf{k}}}, P_{\mathbf{k}}, \left(\frac{\Gamma_{\mathbf{t}_{\mathbf{k}}}, \xi}{2} \right) \right\rangle_{\mathbf{k}} d\xi \quad . \tag{IV.52}$$

Making the variable changes indicated in Eqs.IV.32 and multiplying the integral by $\exp(-\eta^2)\exp(\eta^2)$ as for the case of L₁, Eq.IV.52 can be converted to the equivalent form

$$L_{2} = \frac{1}{2 D_{k}^{2}} \frac{1}{\alpha} \int_{-\infty}^{\infty} e^{-\eta^{2}} \left\{ \sqrt{2\pi} \omega \left(\frac{\eta}{\alpha} \right) \Gamma_{\gamma_{k}} \left[\sqrt{\frac{\pi}{2}} \frac{e^{-\frac{\Gamma_{t_{k}} |\eta|}{2\alpha}}}{\beta_{k}} - e^{\frac{\Delta^{2} \eta^{2}}{4\alpha^{2}}} P_{k} \left(\frac{\Gamma_{t_{k}} \eta}{2\alpha} \right) \right]_{k} \right\}$$

-106-

$$\cdot \left\langle \frac{\Gamma_{\mathbf{t}_{k'}}}{2} e^{\frac{\Delta^2 \eta^2}{4\alpha^2}} R_{k'} \left(\frac{\Gamma_{\mathbf{t}_{k'}}}{2\alpha} \right) \right\rangle_{k'} e^{\left[\frac{\langle \mathbf{D}_{k} \rangle^2}{\pi^2 \alpha^2} + \frac{\gamma^2}{\alpha^2} \right]} d\eta \qquad (IV.53)$$

where

$$R_{k} = -\beta_{k} \frac{\partial}{\partial \beta_{k}} P_{k} \left(\frac{t_{k}}{2}\right) . \qquad (IV.54)$$

Eq.IV.53 as before can be evaluated using Gauss-Hermite quadrature. Note that after obtaining L_1 as described earlier, the only new quantity needed to obtain L_2 is R_L .

As for L₁, two algorithms are used to obtain R depending upon the size of β_k relative to $\psi(\theta_k, 0)$.

a. Asymptotic Algorithms for L₂

As for L₁, a fast, large β approximation is used to obtain R_k if $(\beta_k + \psi(\theta_k, 0))/\psi(\theta_k, 0) \ge 2.5$. Specifically,

$$R_{k}\left(\frac{\Gamma_{t_{k}}\xi}{2}\right) \stackrel{\sim}{=} \sqrt{\frac{\pi}{2}} \beta_{k} \exp\left[-\frac{\Delta^{2}\xi^{2}}{4} - \frac{\Gamma_{t_{k}}|\xi|}{2}\right] \cdot \frac{1 + 2[\rho - U_{k}(\xi)]/(\beta_{k} + \rho)}{(\beta_{k} + \rho)^{2}} \quad (1V.55)$$

where \textbf{U}_k and ρ are given in Eqs.IV.38 and IV.39.

b. Non-Asymptotic Algorithms for L₂

When $(\beta_k + \psi(\theta_k, 0))/\psi(\theta_k, 0) < 2.5$, R_k is obtained in a manner similar to that used for obtaining P_k in the non-asymptotic case.

In order to obtain $R_{k'}$, we differentiate Eq.IV.41 with respect to β_k to obtain a new matrix equation similar to Eq.IV.43, namely

AR = V (IV.56)

where A is the matrix defined in Eqs. IV. 47 and IV. 48 and

 $V_{i} = \frac{\beta_{k} P_{k} \left(\frac{\Gamma_{t_{k}} y_{i}}{2\alpha}\right) e^{\frac{\Delta^{2} y_{i}^{2}}{4\alpha^{2}}}}{N_{i}^{0}}$

(IV.57)

with $N_{,}^{O}$ the previously defined normalization factor given in Eq. IV. 45.

Thus we have

$$R = A^{-1}V.$$

Since A^{-1} has already been obtained for use in computing L₁, L₂ is also available with little additional computational effort.

C. Evaluation of Unresolved Resonance Cross Sections

As stated previously, MC^2-2 neglects the higher order corrections for the accidental overlap so that, for example Eq.IV.13 becomes

1

$$\frac{1}{\langle D_{k} \rangle} \left\langle \Gamma_{\gamma_{k}} J_{k}^{\star} \right\rangle \stackrel{\simeq}{=} \frac{1}{\cos 2\delta_{\ell}} \left\{ \frac{1}{\langle D_{k} \rangle} \left\langle \Gamma_{\gamma_{k}} J(\beta_{k}, \theta_{k}, a_{k}, 0) \right\rangle - O_{\gamma_{k}} \right\} \cdot S \quad (IV, 59)$$

(IV.58)

where S is defined by Eq.IV.14 and the evaluation of the overlap term 0 has been discussed in Section B above. γ_k

If we define the flux correction factor f as

$$f = 1 - \sum_{k} \left[\frac{1}{\langle D_{k} \rangle} \left\langle \Gamma_{t_{k}}^{J(\beta_{k},\theta_{k},a_{k},a_{k})} \right\rangle - O_{t_{k}} \right]$$
(IV.60)

where f is just Eq.IV.14 except that the sum extends over all spin sequences, we may write the effective unresolved resonance capture cross section for a given spin sequence k as

$$\overline{\sigma}_{c_{k}} = \frac{\sigma_{p} \left\langle \Gamma_{\gamma_{k}} J_{k}^{*} \right\rangle}{\left\langle D_{k} \right\rangle f} . \qquad (IV.61)$$

If we accept the approximation given in Eq.IV.20, σ depends to first order only on the resonances of sequence k since all c_k other terms cancel in the ratio S/f. Thus

$$s \sim \prod_{i \neq k} \left\{ 1 - \left[\frac{1}{\langle D_i \rangle} \left\langle \Gamma_{t_i} J(\beta_i, \theta_i, a_i, a_i) \right\rangle - O_{t_i} \right] \right\}$$
(IV.62)

١

and

$$f \approx \prod_{k} \left\{ 1 - \left[\frac{1}{\langle D_{k} \rangle} \left\langle \Gamma_{t_{i}} J(\beta_{i}, \theta_{i}, a_{i}, a_{i}) \right\rangle - O_{t_{i}} \right] \right\}$$
(IV.63)

so that

$$\overline{\sigma}_{c_{k}} \sim \frac{\sigma_{p} \left\{ \frac{1}{\langle \overline{D}_{k} \rangle} \left\langle \Gamma_{\gamma_{k}} J(\beta_{k}, \theta_{k}, a_{k}, 0) \right\rangle - 0_{\gamma_{k}} \right\}}{\cos 2\delta_{\ell} \left\{ 1 - \left[\frac{1}{\langle \overline{D}_{k} \rangle} \left\langle \Gamma_{t_{k}} J(\beta_{k}, \theta_{k}, a_{k}, a_{k}) \right\rangle - 0_{t_{k}} \right] \right\}}$$
(IV.64)

100

with similar expressions for the fission and total cross sections where Γ_{γ_k} is replaced by Γ_{f_k} and Γ_{t_k} respectively,

The unresolved cross sections at each library specified E* energy are finally obtained by summing the partial contributions such as given by Eq. IV.64 över all spin sequences. Thus the unresolved cross section for process x (capture, fission or total), material m at energy point E* is given by

$$\overline{\sigma}_{x}^{m}(E^{*}) = \sum_{k \in m} \overline{\sigma}_{k}(E^{*})$$
(IV.65)

where the sum ranges over all sequences belonging to material m.

D. Unresolved Resonance Integrals and Resolved-Unresolved Resonance Interaction

Since the E* points for which the average unresolved resonance parameters are supplied in the library vary from isotope to isotope of each material, the code generates a fixed energy grid onto which the resonance integrals as given by Eq.IV.59 are linearly interpolated. The user may specify the energy grid, or the code will generate the grid using the following algorithm.

The first grid point corresponds to the energy of the top of the highest energy broad group. For points 2 through 25, the successive mesh point energies are in the ratio exp(-0.25); for points 26 through 125 in the ratio exp(-0.05); for points 126 through 149 in the ratio exp(-0.25); and finally mesh point 150 corresponds to EMIN, the energy at the top of the thermal group. If EMIN is encountered before mesh point 150, the grid terminates with point EMIN and fewer than 150 points are in the fixed energy grid. Finally, the fixed grid is truncated so that only those points remain which fall within the unresolved E* points of any of the isotopes in the problem under investigation.

If there is more than one E* point between any of the fixed energy grid points, the unresolved resonance integrals are averaged over the several E* points so that there is a single average resonance integral at a single average energy E* between any two fixed energy grid points. These average resonance integrals as well as the flux correction factors f are then linearly interpolated onto the fixed grid.

Figure 5 schematically represents the energy structure if we denote the fixed energy grid points by ESR, with corresponding midpoint energies E_i .

In the case where resolved and unresolved resonances are present in the same energy region, the code accounts for interaction effects in the following way. If resolved resonances j have energies lying with energies E and E in Fig 5, then the unresolved resonance integral at mesh point ESF, is modified to

$$\frac{1}{\langle D_{k} \rangle} \left\langle \Gamma_{\gamma_{k}} J_{k}^{*} \right\rangle \left\{ 1 - \frac{1}{E_{i} - E_{i-1}} \sum_{j} J_{t}^{*^{\delta}\ell} \right\}$$
(IV.66)

where the sum in Eq.IV.66 includes all resolved resonances in the energy range $E_i - E_{i-1}$ and J_t^{sl} is given by Eq.III.24. If the resolved resonances are multilevel resonances, the J_t^{sl} is replaced by the appropriate multilevel expression given by Eq.III.33. Similar corrections are made to the unresolved fission and total resonance integrals.

The resolved resonance integrals for resonances i in the energy range $E_i - E_{i-1}$ are similarly modified to

 $\frac{J_{x_{i}}^{*Sl}}{\frac{E_{i}}{E_{i}}} f$

where f is given by Eq.IV.63 and the value corresponding to grid point ESF, $J_{x}^{s\ell}$ is given by Eq.III.23, and E is the energy of resolved resonance i. i Corresponding expressions for multilevel resolved resonance integrals are used where appropriate.

For resolved resonances falling within the end points and first corresponding midpoint energies, such as the range $\text{ESF}_1 - \text{E}_1$ for example, the f is the value corresponding to the last grid point.

E. Ultra-Fine-Group Unresolved Resonance Cross Sections

The ultra-fine-group spectrum calculation described in Chapter II and the hyper-fine-group RABANL calculation described in Chapter V make use of group averaged unresolved resonance cross sections. These cross sections are derived by assuming a constant weighting function in the ultra-finegroup so that

$$\sigma_{\mathbf{x}_{m}}^{g} = \frac{1}{\mathbf{E}_{g-1} - \mathbf{E}_{g}} \int_{\mathbf{E}_{g}}^{\mathbf{E}_{g-1}} \overline{\sigma}_{\mathbf{x}}^{m}(\mathbf{E}) d\mathbf{E}$$

(IV.68)

(IV.67)





and $\overline{\sigma}_{x}^{m}(E)$ is defined by Eq.IV.65. Future releases of the ENDF/B data files will provide information to specify the functional form of $\overline{\sigma}_{x}^{m}(E)$ versus E. Since such information is not available with ENDF/B-IV, the code MC²-2 assumes that $\ln \overline{\sigma}_{x}^{m}(E)$ is linear in $\ln E$,

$$\overline{\sigma}_{\mathbf{x}}^{\mathbf{m}}(\mathbf{E}) = \left(\frac{\mathbf{E}}{\mathbf{E}^{\star}}\right)^{\mathbf{A}_{\mathbf{n}}} \overline{\sigma}_{\mathbf{x}}^{\mathbf{m}}(\mathbf{E}^{\star})$$

 $A_{n} \equiv \frac{\ln\left(\frac{\overline{\sigma}_{x}^{m}(E_{n+1}^{\star})}{\overline{\sigma}_{x}^{m}(E_{n}^{\star})}\right)}{\ln\left(\frac{E_{n+1}^{\star}}{E_{n}^{\star}}\right)}$

The averaging of Eq. IV.68 using the ln - ln interpolation of Eq. IV.69 is performed analytically in the same manner as used in the MC² code.⁽¹⁰⁾

(IV.69)

V. RABANL, HYPER-FINE GROUP INTEGRAL TRANSPORT THEORY NEUTRON SLOWING DOWN

A. Introduction

The RABANL algorithms provide rigorous resolved resonance cross sections and are intended for use in the lower energy ranges for which the narrow resonance approximation of the MC^2-2 module CSC006 is not valid. The methods used in RABANL are based on the earlier work of Kier and Robba in the (5) RABBLE code and the improvements afforded for slab geometry by Olson in the RABID code (6) The present coding represents an improvement over these earlier codes with regard to accuracy and numerical stability, and also provides a fully variably dimensioned and CCCC compatible package which can be used in a standalone mode, (as with RABBLE or RABID), or in conjunction with a full MC^2-2 execution. In the latter case, elastic scattering sources as well as inelastic, (n,2n), and fission sources are supplied by the ultra-fine-group MC^2-2 calculation. In either mode of operation, RABANL uses the standard MC²-2 library as derived from ENDF/B data. Thus, e.g., resonance parameters are library supplied as opposed to being provided by the code user as is the case for the earlier codes. The user may also optionally include unresolved resonance cross sections derived from the MC^2-2 module CSC005 in the RABANL calculation as described below.

RABANL may be used for homogeneous one region calculations or for multi-region cylindrical or slab heterogeneous geometries. In the latter cases, the spatial flux distribution is obtained using integral transport theory methods involving calculation of cylindrical or slab collision probabilities.

Section B discusses the algorithms used in computing the elastic slowing down sources. Section C discusses the calculation of the collision rates and fluxes for both homogeneous one region and for multi-region problems. The algorithms used for the calculation of the cross sections are presented in Section D. Finally, in Section E, the calculation of the edited broad group fluxes and cross sections is described.

B. Calculation of Elastic Slowing-Down Sources

In order to accurately calculate the neutron slowing down in the presence of rapidly varying resonance cross sections, the hyper-fine-group (hfg) width used in the calculation must be extremely narrow. By the same token the hfg width should be small compared with the maximum lethargy gained per collision with the heaviest of the nuclides in the mixture so that the assumption of only one collision per group is valid.

The code determines the hfg width relative to the resolved resonance Doppler width $\boldsymbol{\Delta}$

where

 $\Delta = \left[\frac{4kTE}{A}\right]^{\frac{1}{2}}$

(V,1)

for temperature T, energy E, mass A, and with the Boltzmann constant k equal to 8.61708×10^{-5} eV/degree Kelvin. In particular

$$\Delta u_{\rm J}^{\rm hfg} = \frac{\Delta}{\rm NE_{\rm J}} \tag{V.2}$$

for the hfg width appropriate to broad group J where E_J is the upper energy of broad group J, N is specified by the user on card type 14 of data set A.MCC2 with a default value of 4, and Δ is obtained from Eq.V.1 using T = 293 degrees Kelvin, E = E_J , and A = 250. The hfg width is generally broad group dependent. The user may however specify that a fixed hfg width be used for the entire problem. In that case, E_J corresponds to the upper energy of the highest broad group in the problem.

In any case, the code requires that Eq.V.2 yield a width not greater than 0.001. Δu^{hfg} is then finally adjusted to correspond to an integral sub-multiple of the ultra-fine group (ufg) width. This adjustment may result in a lethargy width somewhat larger than 0.001. Thus, for example, for the current Δu^{ufg} of 1/120, the largest possible Δu^{hfg} is 0.00141667. For a broad group with upper energy of 275.36 eV, for example, Δu^{hfg} is 0.00030864. A nucleus of mass 250 could elastically scatter neutrons a maximum lethargy width of 0.016 which corresponds to about 11.3 times the largest hyper-fine group width. Lighter nuclei of course can scatter down a greater lethargy span and hence would contain more hyper-fine groups in their scattering band. For example, a nucleus of mass 23 contains about 62.8 of the maximum width hyper-fine groups within its scattering lethargy band.

The probability per unit lethargy that a neutron is scattered from lethargy u' to lethargy u by a nucleus of mass ratio A is given by

$$P(u' \rightarrow u) = \begin{cases} \frac{1}{1-\alpha} e^{-(u-u')}, & u \ge u' \ge u - \varepsilon \\ 0 & u' \le u - \varepsilon \end{cases}$$
(V.3)

where

$$\alpha = \left[\frac{A-1}{A+1}\right]^{2}$$
(V.4)
$$\varepsilon = \ln(1/\alpha)$$
(V.5)

and A is the ratio of target nuclide mass to the mass of the neutron. Note that for hydrogen, the code uses the actual A (~ 0.99917) rather than the approximation A = 1.0.

In particular, the probability of scattering down ℓ hfg each of width Δu into the group having lower lethargy u is given by

-113-

$$P_{\ell}\Delta u = \frac{1}{1-\alpha} \int_{u_0}^{u_0+\Delta u} du \int_{u_0-\ell\Delta u}^{u_0-(\ell-1)\Delta u} du' e^{-(u-u')}$$
$$= \frac{(1-e^{-\Delta u})^2}{1-\alpha} e^{-(\ell-1)\Delta u}$$
$$= P_1 \Delta u e^{-(\ell-1)\Delta u} \qquad (V.6)$$

Equation V.6 must be corrected for scattering into the lowest energetically possible hfg as indicated below.

The scattering band ϵ is adjusted by making small changes to the mass A so that

$$L = \epsilon / \Lambda u \tag{V.7}$$

is an integer. Thus a given hfg may scatter down into the next L hfg. The probability of scattering down into the lowest hfg is given by

$$P_{lowest}\Delta u = \frac{1}{1-\alpha} \int_{u_0}^{u_0+\Delta u} du \int_{u-\varepsilon}^{u_0-\varepsilon+\Delta u} du' e^{-(u - u')}$$
$$= \frac{\alpha}{1-\alpha} \left\{ -\Delta u + 1 + e^{\Delta u} \right\}$$
(V.8)

where the limits on the second integral account for the fact that not all parts of the source hfg may energetically be able to scatter into the lowest hfg.

The ingroup scattering is given by

$$P_{s} \Delta u = \frac{1}{1-\alpha} \int_{u_{o}}^{u_{o} + \Delta u} du \int_{u_{o}}^{u} du' e^{-(u - u')}$$

$$=\frac{1}{1-\alpha}\left(\Delta u - 1 + e^{-\Delta u}\right) . \qquad (V.9)$$

Now if ℓ is set equal to L in Eq.V.6, we obtain

$$P_{L}\Delta u = \frac{\alpha}{1-\alpha} \quad (e^{\Delta u} - 2 + e^{-\Delta u})$$
 (V.10)

where we make use of the fact that

$$\alpha = e^{-L\Delta u} \qquad (V.11)$$

By comparing Eqs.V.8, V.9, and V.10, we see that

$$P_{lowest} \Delta u = P_L \Delta u - \alpha P_s \Delta u$$
 (V.12)

If we denote the hfg flux per unit lethargy in hfg k by ϕ_k and the macroscopic scattering cross section for that group by Σ_k , the source of neutrons per unit lethargy into group k due to elastic scattering-in from all energetically possible lower lethargy hfg is given by

$$S_{o_{k}} = \sum_{\ell=1}^{L} \sum_{s_{k-\ell}} \phi_{k-\ell} P_{1} e^{\Delta u} e^{\ell \Delta u} - \alpha \sum_{s_{k-L}} \phi_{k-L} P_{s}$$
(V.13)

where P_{ℓ} and P_{s} are obtained from Eqs.V.6 and V.9, respectively. Similarly, the ingroup or self-scattering source of neutrons per unit lethargy is given by

$$S_{s_{k}} = \Sigma_{s_{k}} \phi_{k} P_{s}$$
 (V.14)

In the case of a mixture of various mass nuclides, the previous equations will have a contribution from each of the constituents of the composition in question. In particular, for a heterogeneous problem these equations will have contributions from the constituents appropriate to the particular spatial mesh involved.

Note that in the previous equations, Δu , P_{ℓ} , and P_s have not contained a subscript relating to hfg number. This is appropriate so long as the hfg width is a constant. Since in the normal problem Δu changes from broad group to broad group, P_{ℓ} and P_s become broad group dependent quantities.

The direct evaluation of S for each hfg using the summation in ${}^{\circ}_{k}$ Eq.V.13 would be too time consuming since hundreds, and for light nuclides

thousands, of lower lethargy hfg contribute to the scattering-in source for each hfg. By noting from Eq.V.6 that

$$P_{\ell} = e^{-\Delta u} P_{\ell-1}$$
 (V.15)

and making use of Eq.V.11, one can show that Eq.V.13 may be rewritten in the form

$$S_{o_{k}} = e^{-\Delta u} S_{o_{k-1}} - (P_{1} - e^{-\Delta u} P_{s}) \alpha (\Sigma_{s} \phi)_{k-1-L} + P_{1} (\Sigma_{s} \phi)_{k-1}$$
(V.16)
$$- \alpha P_{s} (\Sigma_{s} \phi)_{k-L} .$$

Thus, the scattering source into hfg k requires only the source into the previous hfg S , and the scattering rates $\sum_{k=1}^{\infty} \phi$ for hyper-fine groups k-1, k-L, and k-L-1.

When a variable hfg lethargy width is used, the lower lethargy scattering rates required in Eq.V.16 may occur in a broad group having a different hfg width from that of the receptor group k. In that case, the required $\Sigma \phi$ is in general obtained by a three point interpolation formula, namely

$$f(x_{o} + ph) = \frac{p(p-1)}{2} f_{-1} + (1-p^{2}) f_{o} + \frac{p(p+1)}{2} f_{1}$$
(V.17)

where the scattering rates f_{-1} , f_{o} , and f_{1} are assigned to a lethargy at the center of the hfg involved. If the scattering source occurs within the first or last hfg of a broad group having a different hfg width from that of the receptor group, then the two point formula

$$f(x_{+}ph) = (1-p)f_{+}pf_{1}$$
 (V.18)

is used for interpolation.

RABANL has provision for storing all energetically reachable lower lethargy scattering rates if the user specifies sufficient core storage. If this is not the case however, the code will store averages of the scattering rates over several hfg as necessary to accommodate the available storage. Results using averaged scattering rates will of course be less accurate than if all required hfg can be core contained. However, in practical problems, only slight changes have been observed in the resultant broad group cross sections if hfg averaging is used.

If scattering rates required in Eq.V.16 occur at energies above the top of the RABANL energy range, they are obtained in two ways depending upon whether RABANL is being run as a standalone or if the full MC²-2 calculation is being made. In the former case, the code assumes $\phi = 1.0$ for all energies above the top of the RABANL energy range and takes Σ_{c} to be

constant and equal to the homogenized hard sphere potential scattering cross sections for the materials in the composition involved. If the full MC²-2 problem is run, the scattering rates $\Sigma_{s}\phi$ are provided to RABANL for all energetically reachable ultra-fine groups above the top of the RABANL energy range in a data set SRATES. RABANL assumes that $\Sigma_{s}\phi$ is constant within any ufg in obtaining the necessary $\Sigma_{s}\phi$ for use in Eq.V.16. If data set SRATES is saved from a previous MC²-2 calculation, then a standalone RABANL calculation will similarly use these ufg $\Sigma_{s}\phi$ rather than the potential scattering cross sections and flat unit flux above the top of the RABANL calculation.

The source into the first, lowest lethargy hfg also depends upon whether RABANL is being run as a standalone code without access to a previously supplied data set SRATES, or rather if such data are available. If no SRATES data are available, from Eq.V.13 with $\phi=1.0$ and $\sum_{s} \sum_{p}$ for all groups, one can show that

$$S_{o_1} = \Sigma_{p} \left\{ \frac{1 - e^{-\Delta u}}{\Delta u} - \alpha P_{s} \right\}$$
(V.19)

where Δu is the hfg lethargy width of the first hfg in the RABANL energy range. As before, for a mixture of materials, Eq.V.19 receives a contribution from each of the constituents with appropriate Σ_p , α , and P for each.

When data set SRATES is available, the source into hfg number 1 is derived by directly performing the sum in Eq.V.13 and assuming that $\Sigma_{s}\phi$ is constant within each of the ufg above the top of the RABANL energy range.

In order to minimize numerical roundoff problems, Eq.V.13 is used to obtain S for the first hfg of each broad group rather than using the Eq.V.16 recursion relationship throughout all hfg in the problem. This has been found to be essential for heavily absorbing compositions for which exclusive use of Eq.V.16 for all hfg has led to the generation of negative sources.

If ingroup scattering is omitted, on user option, then ${\rm P}_{\rm S}=0$ and Eq.V.16 becomes

$$S_{o_{k}} = e^{-\Delta u} S_{o_{k-1}} - \alpha P_{1}(\Sigma_{s}\phi)_{k-1-L} + P_{1}(\Sigma_{s}\phi)_{k-1} . \qquad (V.20)$$

In this case, the P_1 as defined in Eq.V.16 must be altered in order to conserve neutrons. In particular, one can show that

$$\sum_{\ell=1}^{L} P_{\ell} - \alpha P_{s} + P_{s} = 1.0$$
 (V.21)

using Eqs.V.6, V.9, and V.11. That is the sum of the scattering-in source (corrected for scattering from the highest hfg) plus the ingroup self-scattering is normalized to unity. On the other hand if $P_{s}=0$, then one finds using Eq.V.6 that

$$\sum_{\ell=1}^{L} P_{\ell} \Delta u = 1 - e^{-\Delta u} .$$
 (V.22)

Hence if ingroup scattering is deleted so that Eq.V.20 is pertinent, then to force normalization of all scattering to unity, P_1 as defined in Eq.V.6 must be divided by $1 - e^{-\Delta u}$ so that

-117-

$$P_{\ell} = \frac{P_{1} e^{-(\ell-1)\Delta u}}{1 - e^{-\Delta u}}, \text{ if } P_{s} = 0.$$
 (V.23)

The reason why the user might choose to accept this more approximate treatment is to reduce running time in the case of heterogeneous problems as discussed later. For homogeneous problems, there is no running time advantage in deleting the self-scattering.

C. <u>Calculation of Collision Rates</u>

1. One Region (Homogeneous) Problems

From neutron balance, the collision rate in a particular hfg k, $\ensuremath{\text{CR}}_k$ is given by

$$CR_{k} = P_{n\ell_{k}} \left\{ S_{o_{k}} + S_{s_{k}} \right\}$$
(V.24)

where P_{n_k} is the non-leakage probability, and S and S are respectively k_k k_k the scattering-in and self-scattering sources. Using Eq.V.14, Eq.V.24 can be rewritten as

$$CR_{k} = P_{n\ell_{k}} \left\{ S_{o} + (\Sigma_{s}\phi)_{k} P_{s} \right\}$$
$$= P_{n\ell_{k}} \left\{ S_{o_{k}} + \frac{\Sigma_{s}P_{s}}{\Sigma_{t_{k}}} (\Sigma_{t}\phi)_{k} \right\}$$
$$= P_{n\ell_{k}} \left\{ S_{o_{k}} + R_{k} \cdot CR_{k} \right\}$$

where Σ_{t_k} is the macroscopic total cross section for hfg k,

$$R_{k} = \frac{\sum_{k}^{P} s_{k}}{\sum_{t_{k}}},$$

and from neutron balance, $CR_k = \sum_{t_k} \phi_k$.

Solving then for CR_k, we have

$$CR_{k} = \frac{P_{n\ell_{k}} S_{o_{k}}}{1 - P_{n\ell_{k}} R_{k}}$$

(V.26)

(V.25)

-118-

For the special case of an infinite medium, $P_{nl_k} = 1.0$. As discussed earlier, P_s will depend upon hfg for the case in which the hfg widths are variable. The hfg index has been omitted here for simplicity.

The non-leakage probability $P_{n_{l_{r_{r_{r_{r}}}}}}$ is taken to be

$$P_{n\ell_{k}} = \frac{\sum_{k}^{\Sigma} t_{k}}{\sum_{t_{k}} D_{k} B^{2}}$$
(V.27)

where again Σ_{t_k} is the hfg k total macroscopic cross section for the mixture,

$$D_{k} = \frac{1}{3\Sigma_{tr_{k}}} \qquad (V.28)$$

and $\boldsymbol{\Sigma}_{\substack{ \text{tr}_k}}$ is the macroscopic transport cross section for the mixture where

we assume

$$\Sigma_{tr_{k}} = \sum_{i=1}^{M} n_{i} \frac{\sigma_{t_{k}}^{i}}{\sigma_{s_{k}}^{i}} (\sigma_{t_{k}}^{i} - \overline{\mu}^{i} \sigma_{s_{k}}^{i}). \qquad (V.29)$$

In Eq.V.29 the sum extends over all isotopes in the mixture, each with atom density N_i and microscopic total and scattering cross sections given by $\sigma^i_{t_k}$ and $\sigma^i_{s_k}$ respectively. $\overline{\mu}^i$ is 2/3Aⁱ where Aⁱ is the ratio of nuclide i mass to neutron mass. The buckling, B², is user specified on the A.MCC2 type 09 card.

The calculation of the S $_{\rm O_L}$ has been discussed earlier for the case of

elastic slowing down. In addition to the elastic scattering sources, neutrons can also appear in a hfg due to inelastic and (n,2n) scattering or to fissions at energies above the top of the RABANL energy range, or due to external sources specified by the user on the A.MCC2 type 08 cards. The former are referred to as fixed source neutrons in that they do not depend upon the RABANL fluxes, while the latter are referred to as external sources. The fixed sources are supplied on data set SRATES which is generated by an MC²-2 ufg calculation. These sources, as in the case of the external sources are specified at an ufg level. RABANL creates equivalent hfg values for these ufg data by a linearization algorithm as specified below. Suppose ufg data are available for ufg numbers 1 to NG where ${\tt Y}_{\rm I}$ is the ufg datum for ufg I. Defining

$$\Delta_{1} = Y_{2} - Y_{1}$$

$$\Delta_{I} = \frac{Y_{I+1} - Y_{I}}{2}, I = 2, NG - 1$$
(V.30)
$$\Delta_{NG} = Y_{NG} - Y_{NG - 1},$$

then the value assigned to a hfg j lying within ufg I, assuming there are exactly N hfg per ufg, is given by

$$y_j = (Y_I - \Delta_I/2) + f_j \Delta_I$$

where

$$f_j = \frac{2j-1}{2N}, j = 1, 2, ..., N$$
 (V.31)

An example is shown in Fig. 6 for the case of N = 5 and NG = 4. The dashed line represents the linearized hfg histogram equivalent to the solid line ufg histogram.

The fixed sources as supplied on data set SRATES are given for each isotope in the mixture. RABANL homogenizes these sources according to the atom densities of the materials contained in the mixture. If we denote the homogenized hfg fixed sources as S_{fix} , Eq.V.26 is modified to fix_k

$$CR_{k} = \frac{{}^{r}n\ell_{k}{}^{s}t_{k}}{1 - P_{n\ell_{k}}{}^{R}k}$$
(V.32)

where

$$S_{t_k} = S_{o_k} + S_{fix_k} + S_{ext_k}.$$
(V.33)

From Eq. V.32 and the macroscopic total cross section in hfg k, we have then the hfg k flux per unit lethargy

 $\phi_{\mathbf{k}} = \frac{CR_{\mathbf{k}}}{\Sigma_{\mathbf{t}_{\mathbf{k}}}} . \tag{V.34}$

The calculation of the ϕ_k thus proceeds hfg by hfg: the elastic slowing down source is calculated for the hfg in question using the methods in Section B; this source is augmented by S_{fix} and S_{ext} and the collision rate is calculated as shown above; the hfg flux is then obtained from Eq.V.34 and the scattering rate $(\Sigma_s \phi)_k$ is stored for use in later slowing down source calculations as required in Eq.V.16.



Ufg OR hfg NUMBER



RABANL also provides for calculation of cross sections of foils immersed in the flux of the homogeneous composition. The foil material is assumed to have a negligible influence on the spectrum. The expressions for the collision rate in the foil and hence the flux in the foil are presented at the end of Section C.2.a.

2. Multi-region (Heterogeneous) Problems

The integral transport form of the Boltzmann equation can be written in general terms as

$$\phi(\bar{\mathbf{r}},\mathbf{u}) = \iiint d\bar{\mathbf{r}}' \int_{0}^{\mathbf{u}} d\mathbf{u}' \Sigma_{s}(\mathbf{u}')\phi(\bar{\mathbf{r}}',\mathbf{u}') P(\mathbf{u}' \rightarrow \mathbf{u}) T(\mathbf{u},\bar{\mathbf{r}}' \rightarrow \bar{\mathbf{r}})$$
$$= \iiint d\bar{\mathbf{r}}' S(\mathbf{u},\bar{\mathbf{r}}') T(\mathbf{u},\bar{\mathbf{r}}' \rightarrow \bar{\mathbf{r}}) \qquad (V.35)$$

where $\phi(\bar{r}, u)$ is the flux per unit lethargy at space point \bar{r} , $\Sigma_s(u')$ is the scattering cross section at lethargy u', $P(u' \rightarrow u)$ is the probability of a lethargy change of u' to u due to the scattering event at \bar{r}' , $S(u, \bar{r}')$ is the source of neutrons of lethargy u at space point \bar{r}' , and $T(u, \bar{r}' \rightarrow \bar{r})$ is the transport kernel giving the probability that neutrons of lethargy u at space point \bar{r}' will reach space point \bar{r} without suffering a collision. $P(u' \rightarrow u)$ is defined in Eq.V.3. The transport kernel depends upon the geometry of the problem since it involves the geometric distance $|\bar{r}-\bar{r}'|$, namely

$$T(u, \bar{r}' \rightarrow \bar{r}) = \frac{\exp \left[\sum_{t} (u, \bar{r}') |\bar{r} - \bar{r}'| \right]}{4\pi |\bar{r} - \bar{r}'|^2}.$$
 (V.36)

RABANL provides solutions for multi-region slab or cylindrical geometries as described below.

a. Slab Geometry

In the case of infinite slab geometry, with reference to Fig. 7 , the spatial integral in Eq.V.35 can be expressed as

$$\iiint d\mathbf{\bar{r}}' \Sigma_{s}(\mathbf{u}')\phi(\mathbf{\bar{r}}',\mathbf{u}') \frac{\exp\left[-\Sigma_{t}(\mathbf{u},\mathbf{\bar{r}}')|\mathbf{\bar{r}}-\mathbf{\bar{r}}'|\right]}{4\pi|\mathbf{\bar{r}}-\mathbf{\bar{r}}'|^{2}}$$

$$= \int d\mathbf{x}' \Sigma_{s}(\mathbf{u}')\phi(\mathbf{x}',\mathbf{u}') \int_{0}^{\infty} \frac{2\pi\rho d\rho \exp\left[-\Sigma_{t}(\mathbf{u},\mathbf{x}') \left\{\rho^{2}+|\mathbf{x}-\mathbf{x}'|^{2}\right\}^{\frac{1}{2}}\right]}{4\pi \left\{\rho^{2}+|\mathbf{x}-\mathbf{x}'|^{2}\right\}}$$

$$= \frac{1}{2} \int d\mathbf{x}' \Sigma_{s}(\mathbf{u}')\phi(\mathbf{x}',\mathbf{u}') \int_{1}^{\infty} dt \frac{\exp\left[-\Sigma_{t}(\mathbf{u},\mathbf{x}')|\mathbf{x}-\mathbf{x}'|t\right]}{t}$$

$$= \frac{1}{2} \int d\mathbf{x}' \Sigma_{s}(\mathbf{u}')\phi(\mathbf{x}',\mathbf{u}') E_{1}\left\{\Sigma_{t}(\mathbf{u},\mathbf{x}')|\mathbf{x}-\mathbf{x}'|\right\} \qquad (V.37)$$





1.

where E₁ is the exponential integral defined in Eq.A.37 of Appendix A. The t used in transforming the variable of integration in Eq.V.37 is $1/\cos\theta$ where θ is shown in Fig. 7.

The flux is therefore given by

$$\phi(\mathbf{x},\mathbf{u}) = \frac{1}{2} \int d\mathbf{x}' \int_{0}^{11} d\mathbf{u}' \Sigma_{\mathbf{s}}(\mathbf{u}')\phi(\mathbf{x}',\mathbf{u}')P(\mathbf{u}' \rightarrow \mathbf{u})E_{1} \left\{ \Sigma_{\mathbf{t}}(\mathbf{u},\mathbf{x}') | \mathbf{x} - \mathbf{x}' | \right\}$$

$$(V, 38)$$

Now since the current is φ cos $\theta,$ from the definition of the exponential integrals, the current is given by

$$\vec{J}(x,u) = \frac{1}{2} \int dx' \int_0^u du' \Sigma_s(u') \phi(x',u') P(u' \rightarrow u) E_2 \left\{ \Sigma_t(u,x') | x - x' | \right\} .$$
(V.39)

If now the continuous-lethargy expressions in Eqs.V.38 and V.39 are transformed into the equivalent hyper-fine multigroup form, Eq.V.39 can be written as

$$\overline{J}_{k}(\mathbf{x}) = \frac{1}{2} \int d\mathbf{x}' S_{k}(\mathbf{x}') E_{2} \left\{ \Sigma_{t_{k}}(\mathbf{x}') | \mathbf{x} - \mathbf{x}' | \right\}$$
(V.40)

where $\overline{J}_k(x)$ is the hfg k current at x, $S_k(x')$ is the hfg k source at x', and $\Sigma_{t_k}(x')$ is the total macroscopic cross section for hfg k and spatial point x'. The calculation of the S_k has been described earlier in Section B, and S_k contains the source into hfg k due to slowing down from all other hfg plus the ingroup or self-scatter source for hfg k

$$S_{k}(x') = S_{o_{k}}(x') + S_{s_{k}}(x')$$
 (V.41)

Noting that the optical thickness τ_k is given by $\Sigma_k x$, the current at τ_k mean free paths beyond a slab of optical thickness τ_k , is given by

$$\vec{J}_{k}(\tau_{k},\tau_{k}) = \frac{1}{2} \int_{0}^{t_{0}} dx' \left[S_{o_{k}}(x') + S_{s_{k}}(x') \right] E_{2}(\tau_{k} + \Sigma_{t_{k}}x') \qquad (V.42)$$

where $\boldsymbol{\Sigma}_{\substack{\textbf{t}\\ k}}$ is the total macroscopic cross section in the slab for hfg k.

Assuming that the self-scatter source and the in-scattered source have the same dependence on x' and assuming that the spatial source per unit length in the slab is given by a flat component plus a linearly varying component

-125-

$$S_{k}(x') = \frac{1}{t_{1}} \left[\overline{S}_{k} + \frac{\Delta S_{k}}{t_{1}} (t_{1}/2 - x') \right]$$
 (V.43)

as shown in Fig. 8, Eq.V.42 may be rewritten as

$$\hat{J}_{k}(\tau_{k},\tau_{k_{1}}) = \frac{\bar{S}_{k}}{2t_{1}} \int_{0}^{t_{1}} dx' E_{2}(\tau_{k} + \Sigma_{t_{k}}x') - \frac{\Delta S}{2t_{1}^{2}} \int_{0}^{t_{1}} dx' x' E_{2}(\tau_{k} + \Sigma_{t_{k}}x') + \frac{\Delta S}{4t_{1}} \int_{0}^{t_{1}} dx' E_{2}(\tau_{k} + \Sigma_{t_{k}}x') . \qquad (V.44)$$

The first and last integrals of Eq.V.44 are easily evaluated using Eq.A.60 of Appendix A. The second integral may be evaluated by noting that (29)

$$\int x^{m} E_{n}(ax + b) dx = -\sum_{i=0}^{m} \frac{m!}{(m-i)!} \frac{x^{m-i}}{a^{i+1}} E_{n+i+1}(ax + b). \qquad (V.45)$$

Finally, one may show that the current

$$\vec{J}_{k}(\tau_{k},\tau_{k_{1}}) = \frac{\vec{S}}{2\tau_{1}} \left[E_{3}(\tau_{k}) - E_{3}(\tau_{k}+\tau_{k_{1}}) \right]$$

$$+ \frac{\Delta S}{2\tau_{1}} \left\{ \left[E_{3}(\tau_{k}) + E_{3}(\tau_{k}+\tau_{k_{1}}) \right] \right\} 2 - \frac{1}{\tau_{1}} \left[E_{4}(\tau_{k}) - E_{4}(\tau_{k}+\tau_{k_{1}}) \right] \right\} . \quad (V.46)$$

Now the collision rate in a slab is the difference between the uncollided current into the slab and the uncollided current out of the slab. Thus the collision rate in a slab of optical thickness τ_k due to the source in a slab of optical thickness τ_k , when slabs 1 and 2 are separated by τ_k mean free k_1 paths is, using the notation of Eq.V.46

$$\vec{CR}_{k}(1 \rightarrow 2) = \vec{J}_{k}(\tau_{k}, \tau_{k_{1}}) - \vec{J}_{k}(\tau_{k} + \tau_{k_{2}}, \tau_{k_{1}}).$$
 (V.47)

The arrow above CR in Eq. V.47 implies that slab 2 lies to the right of slab 1.

Now if there exists an infinite array of unit cells each made up of a number of slab mesh intervals, then the collision rate in all type 2 slabs which lie to the right of the type 1 slabs is given by the infinite sum

$$\vec{CR}_{\infty}(1 \rightarrow 2) = \sum_{m=0}^{\infty} \left[\vec{J}_{k}(\tau_{k} + mh, \tau_{k}) - \vec{J}(\tau_{k} + \tau_{k} + mh, \tau_{k}) \right]$$
(V.48)





-126-
where h is the optical thickness of the unit cell.

Since neutrons proceed both to the right and to the left, the total collision rate in slab j due to sources in slab i is given by

$$CR_{\infty}(i \rightarrow j) = C\dot{R}_{\infty}(i \rightarrow j) + C\dot{R}_{\infty}(j \leftarrow i)$$
(V.49)

where the arrows above the CR denote the direction of neutron motion. Appendix A describes the numerical methods used in evaluating the infinite sums of E_n functions involved in obtaining Eq.V.49.

Thus, for the neutron source S in mesh interval i, the collision i_k probability from region i to k and hfg k is given by

$$C_{k_{ij}} = CR_{\infty}(i \neq j)/S_{i_{k}}$$
(V.50)

The unit cell may have either periodic or reflective boundary conditions as specified by the user. Thus if a four mesh interval cell having compositions ordered as ABCD is specified to have periodic boundary conditions, the infinite lattice will have the compositions ... ABCDABCDABCD... If instead reflective boundary conditions are specified, the infinite lattice will correspond to ... DCBAABCDDCBAABCD...

We will now drop the hfg index to simplify the notation. If the unit cell consists of N slab mesh intervals, then the collision probability can be represented by an NxN square matrix <u>C</u>. Similarly, the collision rate, the scattering-in sources, and the self-scattering sources are all N dimensional vectors, and are represented by <u>CR</u>, S_o, and S_s respectively.

Using matrix arithmetic then one has from neutron balance

$$\underline{CR} = \underline{C} \left[\underbrace{S}_{0} + \underbrace{S}_{s} \right]$$
(V.51)

Again, as for the case of the homogeneous one region calculation, the S are augmented by the fixed and external sources as in Eq.V.33 so that we rewrite Eq.V.51 as

$$\underline{CR} = \underline{C} \left[\underline{S}_{\underline{t}} + \underline{S}_{\underline{s}} \right]$$
(V.52)

where $S_{\underline{t}}$ is the N dimensional vector representation of Eq.V.30. Now from Eqs.V.25 and V.34, we can replace the self-scattering source $S_{\underline{s}}$ by

$$\frac{S_s}{s} = \frac{R \cdot CR}{CR} \cdot (V \cdot 53)$$

In matrix notation, <u>R</u> will be a square, NxN diagonal matrix with elements given by Eq.V.25 for each slab mesh interval. Thus we have

$$\underline{CR} = \underline{C} \left[\underline{S}_{\underline{t}} + \underline{R} \ \underline{CR} \right]$$
(V.54)

(V.55)

Solving Eq.V.54 for CR we have

$$[\underline{I} - \underline{C} \ \underline{R}]\underline{CR} = \underline{C} \ \underline{S}_{\underline{t}}$$

and hence

$$\underline{CR} = \left[I - \underline{C} \underline{R}\right]^{-1} \underline{C} \underline{S}_{\underline{t}}$$

where \underline{I} is the unit matrix.

Note that the solution of Eq.V.55 requires the inversion of an NxN matrix. Also, if self-scattering is ignored so that \underline{R} is the null matrix, then Eq.V.54 reduces to

$$\underline{CR} = \underline{C} \underbrace{S_{t}}_{s}, \text{ if } P_{s} = 0.$$
(V.56)

Now to this point, the algorithms for the slab geometry problem have referred to infinite geometry. If however the user choose to specify a buckling and permit leakage from each mesh interval, the code calculates the non-leakage probabilities as in Eq.V.27. Expressing these in the form of a square NxN diagonal matrix <u>PNL</u> where each diagonal element is the nonleakage probability for the particular mesh interval, we may rewrite Eq.V.54 as

$$\underline{CR} = \underline{C} \underline{PNL} \left[\underline{S}_{\underline{t}} + \underline{R} \underline{CR} \right].$$
(V.57)

Again, solving for <u>CR</u>, we have in place of Eq.V.55

$$\underline{CR} = [\underline{I} - \underline{C} \ \underline{PNL} \ \underline{R}]^{-1} \ \underline{C} \ \underline{PNL} \ \underline{S}_{\underline{t}}$$
(V.58)

and if self-scattering is omitted,

$$\underline{CR} = \underline{C} \underline{PNL} \underline{S}_{\underline{t}}, P_{\underline{s}} = 0.$$
 (V.59)

Thus, as noted earlier, the execution time is reduced if the user chooses to omit the self-scattering contribution because the matrix inversion is eliminated.

The calculation of the spatially integrated hfg flux per unit lethargy $\phi(i)$ where i denotes spatial mesh interval proceeds as for the homogeneous case, hfg by hfg: the elastic slowing down sources are calculated for each spatial mesh interval; these sources are augmented by the linearized fixed and external sources; the collision probability is calculated as in Eq.V.50 where however the self-scatter source is not included in S₄; the collision

rate is computed using Eq. V.55 or V.56 as appropriate; the hfg spatially integrated flux for each mesh interval is calculated using Eq.V.34; the scattering rates $\sum_{s} \phi$ are computed and stored for use in calculating lower energy slowing sources.

As mentioned above, in computing the collision probability as in Eq.V.50, the self-scattering source is not included since the hfg k flux is not yet available. Since the self-scattering contribution is always a very small part of the total source in a hfg, the alternative procedure requiring an iteration procedure to compute the true collision probability is not justified.

As in the case of the homogeneous calculation, the slab geometry problem provides for the calculation of foil cross sections. The code calculates the cross section for each foil specified at the interface between each spatial mesh interval and that immediately to its right. A foil is also place at the rightmost boundary of the unit cell.

The same procedure is used for the foil calculations as indicated above for the calculation of the collision rates in the slab mesh intervals. Thus, for example, by analogy with Eq.V.48, the collision rate in a foil f due to the sources in all intervals i in the infinite lattice lying to the left of the foil is given by

$$\vec{CR}_{\omega}(i \rightarrow foil) = \sum_{m=0}^{\infty} \left[\vec{J}(\tau + mh, \tau_i) - \vec{J}(\tau + \tau_f + mh, \tau_i) \right]$$
(V.60)

where the arrows indicate that the current is right directed. The foil with optical thickness $\tau_{\rm f}$ is separated from the interval of optical thickness $\tau_{\rm i}$ by the distance τ +mh where again h is the unit cell optical thickness. Similar contributions to the collision rate are provided by the left directed currents which when add to Eq.V.60 yield the total collision rate CR_m(foil).

Since the collision rate in the foil is given by

$$CR_{\infty}(foil) = \int_{foil} \phi(\mathbf{x}) \Sigma_{t_{f}} d\mathbf{x} = \overline{\phi}_{foil} \tau_{f}$$
(V.61)

where $\Sigma_{t_{f}}$ is the total foil macroscopic cross section, the average flux in the foil can be written as

$$\overline{\phi}_{\text{foil}} = \frac{CR_{\infty}(\text{foil})}{\tau_{\text{f}}} . \qquad (V.62)$$

Now for a homogeneous one region problem, the current into the left surface of a foil due to all sources to the left is given by Eq.V.40 as

$$\vec{J}_{L} = \frac{1}{2} \int_{0}^{\infty} dx' S E_{2}(\Sigma_{t}x')$$
$$= \frac{S}{4\Sigma_{t}}$$
(V.63)

-129-

where S is the space independent source of neutrons in the homogenous mixture and Σ_t is the macroscopic total cross section. Similarly, the current at the right surface of the foil due to the region to the left of the foil is given by

$$\vec{J}_{R} = \frac{1}{2} \int_{0}^{\infty} dx^{*} S E_{2}(\Sigma_{t} x' + \tau_{f})$$
$$= \frac{S}{2\Sigma_{t}} E_{3}(\tau_{f}) \qquad (V.64)$$

Hence the collision rate in the foil due to the incident flux from the left half space is

$$CR_{\infty}(foil) = \vec{J}_{L} - \vec{J}_{R} = \frac{S}{2\Sigma_{L}} \left[0.5 - E_{3}(\tau_{f}) \right] .$$
 (V.65)

Since the same contribution is provided by the incident flux from the right half space, the total collision rate then is given by

$$CR_{\infty}(foil) = \frac{S}{\Sigma_{t}} \left[0.5 - E_{3} (\tau_{f}) \right] . \qquad (V.66)$$

Finally, Eq.V.66 in Eq.V.62 yields the average flux in the foil.

b. Cylindrical Geometry

RABANL makes the same approximation as does the RABBLE⁽⁵⁾ code in the case of cylindrical geometry. Namely, the flux at the surface of any cylindrical mesh interval is assumed to be isotropic so that neutron currents at an interface vary as the cosine of the incident angle (the so called cosine current approximation). In this case, the collision rate in an interval need be related only to the neutron currents impinging on its inner and outer surfaces so that, the collision rate is dependent only upon the adjacent mesh intervals.

Following the notation of Ref. 5, let

- J⁺ = the current impinging on the inner surface of the (i + 1)th cylindrical interval (in the increasing radial direction)
- J = the current impinging on the outer surface of the ith interval (in the decreasing radial direction)
- P⁺ = the probability of escape through the outer surface of interval 1 due to a flat volume source
- P⁻ = the probability of escape through the inner surface of interval i due to a flat volume source

- T^{OI}_i = the transmission probability from the inner to the outer surface of interval i
- T^{IO}_i = the transmission probability from the outer to the inner surface of interval i
- T_{i}^{00} = the transmission probability from the outer to the outer surface of interval i.

If mesh interval 1 is the center interval and if isotropic return is assumed as the outer boundary condition for the cylindrical unit cell (the so called white boundary condition), then by neutron balance we have for an N cylindrical mesh interval unit cell

 $J_{i}^{+} = \begin{cases} T_{i}^{00} J_{i}^{-} + P_{i}^{+} S_{i} & i = 1 \\ T_{i}^{01} J_{i-1}^{+} + T_{i}^{00} J_{i}^{-} + P_{i}^{+} S_{i} & i = 2, N \end{cases}$ $J_{i}^{-} = \begin{cases} T_{i}^{01} J_{i+1}^{-} + P_{i+1}^{-} S_{i+1} & i = 1, N-1 \\ J_{i}^{+} & i = N \end{cases}$ (V.67) I = I, N-1 I =

If we define the column vector \underline{J} having 2N elements J_1^+ , J_1^- , J_2^+ , J_2^- , J_3^+ , ..., J_N^- and the column vector \underline{PS} having 2N elements $\underline{P}_1^+ \underline{S}_1$, $\underline{P}_2^- \underline{S}_2$, $\underline{P}_2^- \underline{S}_2$, $\underline{P}_3^- \underline{S}_3$, $\underline{P}_3^+ \underline{S}_3$, ..., $\underline{P}_N^- \underline{S}_N$, $\underline{P}_N^+ \underline{S}_N$, 0, then the 2N Eqs. V.67 plus V.68 may be represented in matrix notation as

T J ≠ PS

and solving for

(V.69)

where <u>T</u> is a 2N+2N square matrix whose elements involve the T_i^{00} , T_i^{01} , and T_i^{10} .

The set of Eqs. V.69 may be solved for the J_i^+ and J_i^- either by inverting the T matrix, or equivalently by the method of forward elimination, backward substitution. RABANL uses the latter procedure.

Having solved for the J_i^+ and J_i^- , the collision rates for each interval i can be obtained from neutron balance as

$$CR_{i} = \begin{cases} S_{1} + J_{1}^{-} - J_{1}^{+} & i = 1 \\ S_{1} + J_{1}^{-} - J_{1}^{+} + J_{1-1}^{+} - J_{1-1}^{-} & i = z, N \end{cases}$$
(V.70)

The above procedure is used if self-scattering is ignored. If this approximation is not made, the code first must compute the collision probabilities. In order to compute the collision probabilities C_{ij} , RABANL proceeds as indicated above by setting $S_i = 1$, setting all other $S_i = 0$,

$$C_{ij} = CR_{j}$$
 $S_{i=1}, S_{j=0}, j \neq i.$ (V.71)

This procedure is repeated for each interval i to complete the evaluation of the $C_{i,i}$.

Having obtained the C_{ij}, the collision rates are calculated as indicated above for slab geometries using Eq. V.55 which requires a matrix inversion. Note that infinite geometry is assumed for the cylindrical geometry option so that the non-leakage probability is always 1.0.

The T^{OI} and T^{OO} are obtained as described in Section VII of Appendix A. The other quantities needed in the evaluation of the CR can be expressed in terms of these transmission probabilities as shown in Ref.5. In particular, define

$$x_{i} = \frac{r_{i-1}}{r_{i}}$$
(V.72)

where $r_0 = 0$ and

$$z_{i} = \Sigma_{t_{i}}(r_{i} - r_{i-1})$$
 (V.73)

where r_i is the outer radius of mesh interval i and Σ_t is the total macroscopic cross section in interval i. Then

$$T_{i}^{IO} = x_{i}T_{i}^{OI}$$
(V.74)

$$P_{i} = \frac{x_{i}(1 - T_{i}^{0})}{2z_{i}(1 + x_{i})}$$
(V.75)

$$P_{i}^{+} = \frac{1 - T_{i}^{00} - T_{i}^{10}}{2z_{i} (1 + x_{i})} .$$
 (V.76)

The foil capability described earlier for homogeneous or slab geometry problems is not provided in the case of cylindrical cells.

D. Calculation of Hyper-fine Group Cross Sections

The cross sections for a particular material in a given hfg are made up of both smooth and resonance contributions. The smooth data are available at an ufg detail from the library as provided by ETOE-2. At user option, as specified on card type 02 of data set A.STP015, the contribution from ufg unresolved resonance cross sections as prepared by the modules CSC004 and CSC008 will be added to the other library provided smooth data. These ufg smooth data are linearized as in the case of the fixed and external sources using the algorithms given in Eqs.V.30 and V.31 to generate the hfg cross sections to be added to the resolved resonance cross sections. The contributions due to the resolved resonances are computed in RABANL during the sweep down through the hfg. The code will accept Breit-Wigner single level resonance parameters, or multilevel parameters in either the Adler-Adler or Breit-Wigner representations. The resolved resonance cross sections for a given hfg are evaluated at the energy corresponding to the average lethargy of the hfg.

1. <u>Resonance Selection</u>

Due to the large number of hfg, the calculation of the resonance cross sections in each due to all resolved resonance in the problem mixture may be rather time consuming. The user has the option of specifying a resolved resonance selection criterion using the type 14 card of data set A.MCC2. An intermediate group is specified corresponding to an integral number of ufg, with the default number being two. The contribution of any resonance falling within the intermediate group is automatically included in each hfg contained in the intermediate group in question. Those resonances lying outside the intermediate group, whose resonant total cross sections at the intermediate group boundary equal or exceed the criterion specified on the type 14 card, are also included for all hfg contained in the intermediate group. The default for this criterion is 0.05. All other resonances are ignored for that intermediate group. In calculating the total resonant cross sections during the selection process, the resonances are Doppler broadened using the temperature of the hottest composition in the problem.

By specifying a criterion of 0.0 on the type 14 card, the user may include the contributions of all resonances in each of the hfg in the calculation. This option, which is the most rigorous is also the most time consuming.

2. Resolved Resonance Algorithms

a. Single Level Breit-Wigner

Single level Breit-Wigner s-wave resonance scattering, capture, and fission cross sections are respectively evaluated using the expressions

$$\sigma_{s}^{s}(E) = \sigma_{o}^{s} \left\{ \frac{\Gamma_{n}^{s}}{\Gamma_{t}^{s}} \psi(\xi, x) + \left[\frac{\sigma_{p} g_{J}^{s} \Gamma_{n}^{s}}{\sigma_{o}^{s} \Gamma_{t}^{s}} \right]^{\frac{1}{2}} \chi(\xi, x) \right\}$$
$$= \sigma_{o}^{s} \left\{ \frac{\Gamma_{n}^{s}}{\Gamma_{t}^{s}} \psi(\xi, x) + a^{s} \chi(\xi, x) \right\}$$

(V.77)

$$\sigma_{c}^{s}(E) = \sigma_{o}^{s} \left[\frac{|E_{o}|}{E} \right]^{\frac{1}{2}} \frac{\Gamma_{\gamma}^{s}}{\Gamma_{t}^{s}} \psi(\xi, \mathbf{x})$$

$$\sigma_{f}^{s}(E) = \sigma_{v}^{s} \left[\frac{|E_{o}|}{E} \right]^{\frac{1}{2}} \frac{\Gamma_{f}^{s}}{\Gamma_{t}^{s}} \psi(\xi, x)$$

In the above,

 σ^{s} = the peak cross section of the s-wave resonance

$$= \frac{2.6039953 \times 10^{6}}{|E_{0}|} \left(\frac{A+1}{A}\right)^{2} g_{J}^{s} \frac{\Gamma_{n}^{s}}{\Gamma_{t}^{s}}$$

 E_{2} = the resonance energy in eV

A = the mass number of the target nucleus

 $g_J^s = \frac{2J+1}{2(2I+1)}$ where J is the total spin of the compound nucleus and I is the spin of the target nucleus

 Γ_t^s = the s-wave total line width

 $\Gamma_{\gamma}^{\mathbf{S}}$ = the s-wave radiation line width

 Γ_{f}^{s} = the s-wave fission line width

 $\Gamma_n^{\mathbf{s}} = \Gamma_{\mathbf{t}}^{\mathbf{s}} - \Gamma_{\gamma}^{\mathbf{s}} - \Gamma_{\mathbf{f}}^{\mathbf{s}}$

E = the laboratory neutron energy in eV corresponding to the average lethargy for the hfg in question

o p = the target nucleus potential scattering cross section $\psi(\xi, \mathbf{x}) = \text{the Doppler-broadened symmetric line shape function}$ $= \frac{\xi \sqrt{\pi}}{2} \operatorname{ReW}(\frac{\xi \mathbf{x}}{2}, \frac{\xi}{2})$

 $\chi(\xi, \mathbf{x})$ = the Doppler-broadened anti-symmetric line shape function = $\xi \sqrt{\pi} \operatorname{ImW}(\frac{\xi \mathbf{x}}{2}, \frac{\xi}{2})$

W(z) = the error function for complex arguments

= $exp(-z^2)$ erfc(-iz) where z = x + iy

(V.78)

(V.79)

 $\xi = \Gamma_t^S / \Delta$ and Δ is defined in Eq.V.l with E defined above and with the Boltzmann constant equal to 8.6168 x 10^{-5} eV/degree Kelvin

$$x = 2(E - E_0)/\Gamma_t^s$$
.

The ψ and χ functions are obtained from bivariate interpolation in prestored tables as described in Section III of Appendix A.

Single level Breit-Wigner p-wave resonance scattering, capture, and fission cross sections are, respectively evaluated using the expressions

$$\sigma_{s}^{p}(E) = \sigma_{o}^{p} \frac{\Gamma_{n}^{p}}{\Gamma_{t}^{p}} \left\{ \psi(\xi, \mathbf{x}) + \frac{\Gamma_{t}^{p}}{2E_{o}} \chi(\xi, \mathbf{x}) + \left[\frac{\Gamma_{t}^{p}}{2E_{o}} \right]^{2} \left[1 - \psi(\xi, \mathbf{x}) \right] \right\} \quad (V.80)$$

$$\sigma_{c}^{p}(E) = \sigma_{o}^{p} \left[\frac{|E_{o}|}{E} \right]^{\frac{1}{2}} \frac{\Gamma_{\gamma}^{p}}{\Gamma_{t}^{p}} \left[\psi(\xi, \mathbf{x}) + \frac{\Gamma_{t}^{p}}{4E_{o}} \chi(\xi, \mathbf{x}) \right]$$
(V.81)

$$\sigma_{\mathbf{f}}^{\mathbf{p}}(\mathbf{E}) = \sigma_{\mathbf{o}}^{\mathbf{p}} \left[\frac{|\mathbf{E}_{\mathbf{o}}|}{\mathbf{E}} \right]^{\frac{1}{2}} \frac{\Gamma_{\mathbf{f}}^{\mathbf{p}}}{\Gamma_{\mathbf{f}}^{\mathbf{p}}} \left[\psi(\boldsymbol{\xi}, \mathbf{x}) + \frac{\Gamma_{\mathbf{f}}^{\mathbf{p}}}{4\mathbf{E}_{\mathbf{o}}} \chi(\boldsymbol{\xi}, \mathbf{x}) \right] . \tag{V.82}$$

In the above,

 $\sigma_{o}^{p} = \text{the peak cross section of the p-wave resonance evaluated} \\ as before but using the parameters for the p-wave resonance$ $<math display="block">\Gamma_{t}^{p} = \text{the p-wave total line width} \\ \Gamma_{n}^{p} = \Gamma_{t}^{p} - \Gamma_{\gamma}^{p} - \Gamma_{f}^{p} \\ \Gamma_{\gamma}^{p} = \text{the p-wave radiation line width} \\ \Gamma_{f}^{p} = \text{the p-wave fission line width} \\ \xi = \Gamma_{t}^{p} / \Delta \\ x = 2(E - E_{o}) / \Gamma_{t}^{p}.$

b. <u>Multilevel Breit-Wigner</u>

Multilevel Breit-Wigner s-wave resonance scattering, capture, and fission cross sections are, respectively evaluated using the expressions

$$\sigma_{s}^{s,bw}(E) = \frac{\sigma_{o}^{bw}}{|G_{t}^{bw}|} \begin{cases} G_{t}^{bw} \psi(\xi,x) + |G_{t}^{bw}| a^{bw} \chi(\xi,x) - G_{\gamma}^{bw} \psi(\xi,x) - G_{f}^{bw} \psi(\xi,x) \end{cases} (V.83)$$

$$\sigma_{c}^{s,bw}(E) = \sigma_{o}^{bw} \frac{G_{\gamma}^{bw}}{|G_{t}^{bw}|} \left[\frac{|E_{o}|}{E}\right]^{\frac{1}{2}} \psi(\xi,x) \qquad (V.84)$$

$$\sigma_{f}^{s,bw}(E) = \sigma_{o}^{bw} \frac{G_{f}^{bw}}{|G_{t}^{bw}|} \left[\frac{|E_{o}|}{E}\right]^{\frac{1}{2}} \psi(\xi, \mathbf{x}) . \qquad (V.85)$$

In the above

 $\sigma_{o}^{bw} = \sigma_{o}^{s}G_{t}$ where σ_{o}^{s} is defined below Eq.V.79

G^{bw}_t = Breit-Wigner multilevel symmetric parameter for the total reaction which is 1 plus the symmetric level-level interference contribution

 G_{γ}^{bw} = Breit-Wigner multilevel symmetric parameter for the radiative capture reaction

 Γ_t^{bw} = the Breit-Wigner multilevel total line width

G^{bw} = Breit-Wigner multilevel symmetric parameter for the fission reaction

 $x = 2(E - E_0) / \Gamma_t^{bw}$

= $\Gamma_{f}^{s}/\Gamma_{f}^{bw}$

 $= \Gamma_{\gamma}^{s} / \Gamma_{t}^{bw}$

 $\xi = \Gamma_{\pm}^{bw}/\Delta.$

Multilevel Breit-Wigner p-wave resonance scattering, capture, and fission cross sections are, respectively evaluated using the single level expressions given by Eqs. V.80, V.81 and V.82 where σ_o^p is replaced by σ_o^{bw} , Γ_t^p is replaced by Γ_t^{bw} , Γ_γ^p is replaced by G_γ^{bw} , Γ_t^{pw} , Γ_f^p is replaced by G_f^{bw} , Γ_t^{bw} , and Γ_n^p is replaced by Γ_t^{bw} , $(1 - G_\gamma^{bw} - G_f^{bw})$.

c. Multilevel Adler-Adler

Multilevel Adler-Adler resonance scattering, capture, and fission cross sections are, respectively evaluated using the expressions

$$\sigma_{s}^{aa} = \left[\frac{|E_{o}|}{E}\right]^{l_{2}} \frac{\sigma_{o}^{aa}}{|G_{t}^{aa}|} \begin{cases} G_{t}^{aa} \ \psi(\xi,x) + |G_{t}^{aa}| \ a^{aa} \ \chi(\xi,x) \end{cases}$$

$$- G_{\gamma}^{aa} \left[\psi(\xi, \mathbf{x}) + b_{\gamma} \chi(\xi, \mathbf{x}) \right] - \left[G_{f}^{aa} \psi(\xi, \mathbf{x}) + b_{f} \chi(\xi, \mathbf{x}) \right] \right\}^{\sim} (V.86)$$

$$\sigma_{c}^{aa}(E) = \sigma_{o}^{aa} \frac{G_{\gamma}^{aa}}{|G_{+}^{aa}|} \left[\frac{|E_{o}|}{E} \right]^{\frac{3}{2}} \left\{ \psi(\xi, \mathbf{x}) + \mathbf{b}_{\gamma} \chi(\xi, \mathbf{x}) \right\}$$
(V.87)

$$\sigma_{f}^{aa}(E) = \sigma_{o}^{aa} \frac{G_{f}^{aa}}{|G_{t}^{aa}|} \left[\frac{|E_{o}|}{E}\right]^{\frac{1}{2}} \left\{\psi(\xi, \mathbf{x}) + \mathbf{b}_{f} \chi(\xi, \mathbf{x})\right\} . \qquad (V.88)$$

In the above

$$\begin{aligned} \sigma_{o}^{aa} &= \frac{2.6039953 \times 10^{6}}{|E_{o}|} \left(\frac{A+1}{A}\right)^{2} \frac{|G_{t}^{aa}|}{2\Gamma_{t}^{aa}} \\ G_{t}^{aa} &= \sqrt{|E_{o}|} \left[G^{T} \cos 2\phi_{\ell} + H^{T} \sin 2\phi_{\ell}\right] \\ G^{T} &= Adler-Adler \text{ symmetric total cross section parameter} \\ \text{from ENDF/B} \\ H^{T} &= Adler-Adler \text{ antisymmetric total cross section parameter} \\ \phi_{\ell} &= \text{phase shift} \\ \Gamma_{t}^{aa} &= S- \text{ matrix total line width for the Adler-Adler formulation} \\ a^{aa} &= -0.5 \left\{\frac{H^{T} \cos 2\phi_{\ell} - G^{T} \sin 2\phi_{\ell}}{|G_{t}^{aa}|}\right\} \sqrt{|E_{o}|} \\ G_{\gamma}^{aa} &= \sqrt{|E_{o}|} \left[G^{C} \cos 2\phi_{\ell} + H^{C} \sin 2\phi_{\ell}\right] \\ G^{C} &= Adler-Adler \text{ symmetric capture cross section parameter from} \\ ENDF/B \end{aligned}$$

H^C = Adler-Adler antisymmetric capture cross section parameter from ENDF/B

$$-138-$$

$$b_{\gamma} = -0.5 \left\{ \frac{H^{C} \cos 2\phi_{\ell} - G^{C} \sin 2\phi_{\ell}}{G_{\gamma}^{aa}} \right\} \sqrt{|E_{o}|}$$

$$G_{f}^{aa} = \sqrt{|E_{o}|} \left[G^{F} \cos 2\phi_{\ell} + H^{F} \sin 2\phi_{\ell} \right]$$

$$G^{F} = Adler-Adler \text{ symmetric fission cross section parameter from ENDF/B}$$

$$H^{F} = Adler-Adler \text{ antisymmetric fission cross section parameter from ENDF/B}$$

$$b_{f} = -0.5 \left\{ \frac{H^{F} \cos 2\phi_{\ell} - G^{F} \sin 2\phi_{\ell}}{G_{f}^{aa}} \right\} \sqrt{|E_{o}|}$$

$$x = 2(E - E_{o})/\Gamma_{t}^{aa}$$

$$\xi = \Gamma_{t}^{aa}/\Delta$$

The various G and H parameters as obtained from ENDF/B are assumed to include the factor $2g_J/\sqrt{|E_0|}$. Thus the factor g_J does not appear explicitly in the expression for σ_0^{aa} .

E. Broad Group Edits

The hfg spatially integrated fluxes per unit lethargy as calculated from Eq. V.34 are summed over the corresponding ufg to obtain the ufg fluxes and these in turn are summed over the broad group involved to obtain the broad group fluxes. Thus the broad group J flux in mesh interval i is given by

$$\phi_{J}^{bg}(i) = \sum_{j} \phi_{j}(i) \Delta u. \qquad (V.89)$$

where the sum over hfg j extends over all hfg contained in bg J. The code edits the broad group fluxes per unit volume by dividing each of the $\phi_{I}^{bg}(i)$ by the corresponding mesh interval volume.

Each spatial region is made up of one or more spatial mesh intervals each of which contains the same mixture of isotopes. The region R integrated bg fluxes are obtained by summing Eq. V.89 to yield

$$\Phi_{J}^{bg}(R) = \sum_{i} \Phi_{J}^{bg}(i)$$
 (V.90)

where the sum over i extends over all mesh intervals contained in region R.

The broad group, region dependent microscopic capture, fission, and scattering cross sections for each material m are obtained by dividing the reaction rate by the integrated flux. Therefore -13**9**-

 $\sigma_{\mathbf{x}_{\mathbf{J}}}^{\mathbf{m}}(\mathbf{R}) = \frac{\sum_{i} \sigma_{\mathbf{x}_{i}}^{\mathbf{m}}(\mathbf{i}) \phi_{\mathbf{j}}(\mathbf{i}) \Delta \mathbf{u}}{\Phi_{\mathbf{J}}^{\mathbf{bg}}(\mathbf{R})}$

where x corresponds to capture, fission, or scattering. $\sigma_{x_i}^{m}$ (i) is the

microscopic cross section for material m in mesh interval i and hfg j evaluated as described in Section D above. The sums over j and i as above extend over all hfg contained in bg J and all mesh intervals contained in region R.

Now in order to calculate the scattering cross section from broad group J to braod group K, we require the probability of scattering from each hfg j in bg J to bg K.

If we let u_j be the lower lethargy of hfg j in bg J, Δu_j be the lethargy width of hfg j, U_K be the lower lethargy of bg K, and ΔU_K be the lethargy width of bg K, then the probability of scattering by material m from hfg j into bg K, if all parts of bg K can be reached by scattering from hfg j, is given by

$$P^{m}(j \neq K) \Delta u_{j} = \frac{1}{1-\alpha^{m}} \int_{U_{K}}^{U_{K}+\Delta U_{K}} du \int_{u_{j}}^{u_{j}+\Delta u_{j}} du' e^{-(u-u')}$$
$$= \frac{e^{\Delta u_{j}}}{1-\alpha^{m}} \left\{ e^{u_{j}} e^{-U_{K}} (1-e^{-\Delta U_{K}}) \right\} . \qquad (V.92)$$

In the case of the lowest broad group, say bg L which can be reached by scattering from hfg J, since not all parts of bg L may be energetically reachable by scattering from hfg j,

$$P^{\mathbf{m}}(\mathbf{j} \neq \mathbf{L}) = \frac{1}{1-\alpha^{\mathbf{m}}} \int_{u_{\mathbf{j}}}^{u_{\mathbf{j}} \neq \Delta u_{\mathbf{j}}} du' \int_{U_{\mathbf{L}}}^{u' \neq \varepsilon} du e^{-(u-u')}$$
$$= \frac{e^{\Delta u_{\mathbf{j}}}}{1-\alpha^{\mathbf{m}}} e^{u_{\mathbf{j}}} e^{-U_{\mathbf{L}}} - \frac{\alpha^{\mathbf{m}} \Delta u_{\mathbf{j}}}{1-\alpha^{\mathbf{m}}} . \qquad (V.93)$$

In Eqs.V.92 and V.93, contributions are made from each of the materials in the mixture.

Using Eqs. V.92 or V.93 as appropriate, the scattering from broad group J to broad group K for material m in region R is given by

$$\sigma_{\mathbf{J} \to \mathbf{K}}^{\mathbf{m}}(\mathbf{R}) = \frac{\sum \sigma_{\mathbf{s}}^{\mathbf{m}}(\mathbf{i}) \phi_{\mathbf{j}}(\mathbf{i}) \mathbf{P}^{\mathbf{m}}(\mathbf{j} \to \mathbf{K})}{\Phi_{\mathbf{J}}^{\mathbf{bg}}(\mathbf{R})} \qquad (V.94)$$

where the sums over i and j are as above.

(V.91)

For foil materilas, the capture, fission, and scattering cross sections are obtained as in Eq. V.91 where now the fluxes used are the averaged fluxes in each foil as given by Eq. V.62.

In addition to the region dependent cross sections, the code also edits quantities needed to calculate cell averaged macroscopic cross sections.

The all volume-averaged atom densities for each material m are computed as

 $N^{m}(C) = \frac{\sum_{R} n^{m}(R) V(R)}{\sum_{R} V(R)}$ (V.95)

where $n^{\mathbf{m}}(\mathbf{R})$ is the atom density of material m in the composition contained in region R and V(R) is the volume of region R. The sums extend over all regions in the cell.

Spatial self-shielding factors are computed for each broad group ${\sf J}$ and material m as

$$F_{J}^{m} = \frac{\sum n^{m}(R) \Phi_{J}^{bg}(R) V(R)}{N^{m}(C) \sum R \Phi_{J}^{bg}(R) V(R)}$$
(V.96)

The product of $N^{m}(C)$ and F_{J}^{m} represents the effective cell averaged atom density for use in creating effective cell averaged macroscopic cross sections. The consistent cell averaged microscopic cross sections for use with the effective atom densities $N^{m}(C)F_{J}^{m}$ are given by

$$\bar{\sigma}_{x_{J}}^{m}(C) = \frac{\sum_{R} \Phi_{J}^{bg}(R) \sigma_{x_{J}}^{m}(R) n^{m}(R) V(R)}{\sum_{R} \Phi_{J}^{bg}(R) n^{m}(R) V(R)} . \qquad (V.97)$$

Thus the effective cell averaged macroscopic cross sections can be obtained from the product $N^{m}(C) F_{J}^{m} \overline{\sigma}_{x_{J}}^{m}(C)$. Equation V.97 is used for capture, fission, scattering, and for transfer between broad groups.

VI. A GUIDE TO USER APPLICATION

A. Standard Path STP015

STP015 is a path driver of the ARC System which invokes the following modules to provide the MC^2-2 capability.

CSI010	(Area 4)	Processes BCD input from files A.MCC2 and A.NIP
CSC004	(Area 5)	Performs the unresolved resonance calculation
CSC005	(Area 6)	Performs the resolved resonance calculation
CSC006	(Area 65)	Calculates Σ_{p} and performs resolved- unresolved resonance overlap calculations
CSC008	(Area 7)	Calculates homogenized ultra-fine- group (ufg) cross sections, elastic scattering matrices, and moderating parameters
CSC009	(Area 8)	Calculates the ufg real spectrum using multigroup and continuous slowing down (CSD) algorithms, obtains a critical buckling, and prepares broad group (bg) cross sections
CSC010	(Area 9)	Calculates the bg fundamental mode real and adjoint spectra and obtains a critical buckling.
CSC011	(RABANL)	Performs a hyper-fine-group (hfg) integral transport calculation and prepares rigorous resolved resonance cross sections
CSE009		Edits broad group cross sections from data set ISOTXS
CSE007		Prepares a single or double precision ARC System XS.ISO data set from the CCCC data set ISOTXS
CSE012		Edits data set XS.ISO

The flow through the code as shown in the flow diagram, Fig. 9 is controlled by standard path STPO15 in conjunction with the user supplied data in the A.STPO15 BCD data set (Appendix B). As can be seen, the various code areas may be selectively included or excluded on user option. As a specific example, the user may select only CSIO10 and RABANL, thereby effecting the equivalent of a RABBLE⁽⁵⁾ or RABID⁽⁶⁾ standalone calculation. Similarly if only module CSEO12 is selected, a standalone XS.ISO editor capability is effected. Also e.g., if data sets related to the unresolved resonance calculation CSC004 are saved, a subsequent related execution of STPO15 could reduce execution time by eliminating execution of this module.

A listing of the path STP015 is provided in Fig. 10.

B. Job Control Language Procedure ARCSP015

The Job Control Language procedure ARCSP015 is listed in Fig. 11. The procedure is consistent with IBM JCL conventions and with release 3.1 of the Attached Support Processor (ASP). The data set blocking is oriented toward use with 3330 magnetic disks.

The symbolic parameters used for the various datasets, their default values, usage, and corresponding DDNAME are given within the listing of the procedure. In addition, the contents of each of the datasets is given on comment cards in the procedure.

The parameters FULLBLK, HALFBLK, and UNITS which are used in many of the datasets are defined with the other symbolic parameters. The block size defaults of 12280 and 6136 for FULLBLK and HALFBLK are consistent with use of 3330 disks, and with the fact that IBM allocates space for buffers in blocks of 2K contiguous bytes (256 real*8 words). In the case of unformatted Fortran data sets (double buffered by default), the number of bytes of core required for buffers is given by 2*BLKSIZE+8 where BLKSIZE is given on the DD card and is rounded up to the nearest double word boundary. Thus, for example in the case of FULLBLK, 24576 bytes would be allocated for buffer space which corresponds to exactly 12, 2K blocks. This is only 8 bytes more than is required for the two buffers. The use of very closely an integral number of 2K byte blocks avoids the problem of possible core fragmentation when data sets are written, closed (rewound) and then read again several times during a run.

The use of SASCR and BATCHDSK for the UNIT parameter throughout the procedure correspond to ANL conventions for generic names corresponding to the class of units involved. SASCR is used to designate permanently resident disk packs for scratch data. BATCHDSK corresponds to both permanently resident and setup disk packs and is used for those files which may be saved on user option for later use. The default names for parameters MCC2F1, MCC2F2, ..., MCC2F8 correspond to the eight MC²-2 library files currently in use. The PRELIB and POSTLIB parameters used in the STEPLIB DD statement are provided to simplify concatenation of program library data sets. They default to empty partitioned data sets.

Through use of the catalogued procedure, the amount of JCL which must be provided by the user is vastly decreased. In particular, Fig. 12 lists the minimal JCL required to execute a problem using the catalogued procedure ARCSP015. Further illustrations of the use of this procedure and of the symbolic parameters are given in Section D, Sample Problem Input.

1



Fig. 9. MC²-2 Module Flow Diagram

Ċ		
ĉ	STDA 15 TS A STAND	NARD PATH DRIVER OF THE ARC SYSTEM WHICH
	CONTROLS THE MCC.	-2 CROSS SECTION DREPARATION CAPARILITY
ĉ	CONTROLS THE HCC	Z CROSS SECTION I MERKATION ON ADDITION
	MODULEC TNROVED	2 Y CMD(15
C	MODULES INVOKED I	51 512015
Ç	CCT 0 10	NCC-3 TRAMA BRACESOP
C		MCC-2 INPUT PROCESSOR
С	CSC004	MCC-2/SDX UNRESULVED RESUMANCE CALCULATION
С	CSC005	MCC-2/SDX RESOLVED RESONANCE CALCULATION
С	CSC006	RESOLVED-UNRESOLVED RESONANCE OVERLAP
С	CSC008	MCC-2 ULTRA-FINE-GROUP MACROSCOPIC DATA
C		PROCESSOR
C	CSC009	MCC-2 ULTRA-FINE-GROUP SPECTRUM CALCULATION
С		AND BROAD GROUP COLLAPSE
С	CSC010	BROAD-GROUP PL SPECTRUM CALCULATION
С	CSC011	HYPER-FINE-GROUP INTEGRAL TRANSPORT SPECTRUM
c		CALCULATION
Ĉ	CSE009	EDITOR OF FILE ISOTXS
č	CSE007	ISOTXS TO XS.ISO FORMAT CONVERSION
č	CSE012	EDITOR OF FILE XS. ISO
Ċ	NILEOO3	STMPLOTTER LINE PRINTER GRAPHICS
ĉ	N0 80 05	
c		
Č	<u>ከምሮ፣አወም አ፣፣ ፣ፕሞም</u> ነ	NT VARIES TO BE DOUBLE PRECISION
ĉ	DECLARE ALL LILE	WAD VANIADING TO DI DOODDI TABCISION
CTDM		
CIBM	DOUDIR DERCICION	
	DOUBLE PRECISION	$DSNAME_{F}CSTUTU_{F}CSCUU4_{F}CSCUU3_{F}CSCU$
		VOTCO CCC011 UNDER DEEC CCE012
	2	XSISO, CSCUIT, UNREG, PREC, CSEUTZ
CIBW		
	DIMENSION NSTAND	(6), PREC (2), DSNAME (89)
	COMMON /OUTPUT/N	OUT
С		,
С	SET LITERAL VARI	ABLES IN DATA STATEMENTS
С		
	DATA CSI010/6HCS	I010/,CSC004/6HCSC004/,CSC005/6HCSC005/,
	1 CSC006/6HCS	C006/,CSC008/6HCSC008/,CSC009/6HCSC009/,
	2 CSC010/6HCS	C010/,CSE009/6HCSE009/,STP015/6HSTP015/,
	3 NUE003/6HNU	E003/, CSE007/6HCSE007/, APATH/8HA.STP015/,
	4 XSISO/6HXS.	ISO/,CSC011/6HCSC011/,UNREG/6HUNREG /,
	5 CSE012/6HCS	E0 12/
	DATA PREC/6HDOUB	LE,6HSINGLE/
	DATA DSNAME/	• •
	16HA.MCC2.6HA.NIP	,8HA.STP015,6HATNUAT,6HBC ,6HBGRES ,
	26HGEOM1 -6HGRPOR	D. 6HISOTXS. 6HLORENZ, 6HMACTOT. 6HMCC2F1.
	36HMCC2F2.6HMCC2F	3.6HMCC2F4.6HMCC2F5.6HMCC2F6.6HMCC2F7.
	46HMCC2F8_6HMTCTO	T. 6HOLDSGS. 6HOPTICL. 6HPLOTTT. 6HPRBCHI.
	56HDRBSDC 6HRESTN	$T_6HSCR001_6HSCR002_6HSCR003_6HSCR004_$
	66HSCROOS 6HSTGMA	P. 6HSMSTGS, 6HSPECTR, 6HSPECIS, 6HSRATES,
٢	76HINDEC SUINDES	6HYS TSO 6HYSTSO2 6HTRESCS
	CHURCHO CONCERNS	Louvoe zoolouvozooslou zupocol

,

t.

```
86HBIGXS1,6HBIGXS2,6HBIGXS3,6HBIGXS4,6HBIGXS5,6HBIGXS6,
     96HBIGXS7,6HBIGXS8,6HBIGXS9,6HBIGXSA,6HBIGXSB,6HBIGXSC,
     A6HBIGXSD,6HBIGXSE,6HBIGXSF,6HBIGXSG,6HBIGXSH,6HBIGXSI,
     B6HBIGXSJ,6HBIGXSK,6HBIGXSL,6HBIGXSM,6HBIGXSN,6HBIGXSO,
     C6HBIGXSP,6HRANDOM,6HRESDAT,6HACS009,6HACSE12,6H$
      DATA I0/0/,ID04/0/,ID05/0/,ID06/0/,ID07/0/,ID08/0/,ID09/0/,
     1
          ID010/0/, ID011/0/, I1/1/, I2/2/, ID012/0/, NSTAND/2*0/, ID013/0/,
     2
          IUNREG/0/
      NOUT = 6
     CALL SYSTEM (DSNAME)
      CALL BCDDS (STP015,N1)
      IF (N1.LE.-2) GO TO 500
      ESTABLISH LOGICAL UNIT NUMBERS FOR FILES MCC2F1-MCC2F8
     IN ORDER TO AVOID THE NEED FOR THEIR SPECIFICATION IN BLOCK=OLD
     DO 10 I=12,19
     CALL SNIFF (DSNAME(I), IDUM, I1)
   10 CONTINUE
     CALL SNIFF (XSISO, IXSISO, IO)
     CALL SNIFF (APATH, IAPATH, IO)
     IF (IAPATH.LE.O) GO TO 110
     READ DATA SET A.STP015
     READ (IAPATH, 1000) APATH, JJ, KK, (NSTAND(I), I=1, JJ)
     IF (NSTAND(1).LE.0) GO TO 100
     READ (IAPATH, 1100) ID04, ID05, ID06, ID07, ID08, ID09, ID010, ID011,
    1
                        ID012, ID013
  100 CONTINUE
     IF (NSTAND(2).GT.0) READ (IAPATH, 1100) IUNREG
REWIND IAPATH
110 CONTINUE
     CHECK ON PRECISION OF EXISTING XS.ISO FILE
     IF (IXSISO.LE.O.OR. (IDO11.NE.O.AND.IDO11.NE.1)) GO TO 130
     NWDS=6
     NSTAND(6) = 0
     READ RECORD 1 OF DATA SET XS.ISO
     CALL REED (IXSISO, I1, NSTAND, NWDS, IO)
     REWIND DATA SET XS. ISO
     CALL REED (IXSISO, IO, DUM, IO, IO)
```

```
IF (NSTAND (6). EQ. IDO11) GO TO 130
```

С С

С

С

С С

С

С С

С

С С

С

С С

С

Fig. 10. MC²-2 Path Driver Listing (Contd.)

-145-

C	1 30 [°]	WRITE (NOUT, 1200) PREC (IDO11+1), PREC (2-IDO11), PREC (2-IDO11) IDO11=NSTAND(6) CONTINUE
č		INVOKE AREA 4
c c		IF (ID04.EQ.0) CALL LINK (CSI010)
c		INVOKE AREA 5
C C		IF (ID05.E0.0) CALL LINK (CSC004)
C	·	INVOKE AREA 6
С		IF (ID06.EQ.0) CALL LINK (CSC005)
C C		INVOKE AREA 6.5
с		IF (ID05.EQ.0.OR.ID06.EQ.0) CALL LINK(CSC006)
C C		INVOKE AREA 7
С		IF (ID07.EQ.0) CALL LINK (CSC008)
C C		INVOKE AREA 8
С		IF (ID08.EQ.0) CALL LINK (CSC009)
C C		INVOKE AREA 9
c	•	IF (ID09.EQ.0) CALL LINK (CSC010) IF (ID012.NE.1) GO TO 120 IF (IUNREG.LE.0) CALL SNIFF (UNREG, IUNREG, I2)
C		INVOKE AREA 10
c	120	CALL LINK (CSC011) CONTINUE
C		EDIT BROAD GROUP CROSS SECTIONS
c		IF (ID010.EQ.0) CALL LINK (CSE009)
		PREPARE A DOUBLE PRECISION OR SINGLE PRECISION VERSION OF DATA SET XS.ISO ACCORDING TO WHETHER IDO11 IS 0 OR 1, RESPECTIVELY
c		IF (ID011.EQ.0.OR.ID011.EQ.1) CALL LINK (CSE007,ID011)
C		EDIT DATA SET XS. ISO
C		

Fig. 10. MC^2-2 Path Driver Listing (Contd.)

,

•

.

-

IF (ID013.NE.0) CALL LINK (CSE012)

INVOKE THE SIMPLOTTER PRINTER GRAPHICS PACKAGE

С С

С

CALL LINK (NUE003, IO)

500 CONTINUE

```
1000 FORMAT (A8, 215/(1615))
```

1100 FORMAT(6X, 1116)

1200 FORMAT (1H0, 129 (1H*) / 1H , 1H*, 3X, A6, 46H PRECISION CROSS SECTIONS MAY 1 NOT BE ADDED TO ,A6,45H PRECISION XS.ISO. NEW DATA ARE PROCESSED 2IN , A6, 11H PRECISION., 4X, 1H*/1H , 129 (1H*)) RETURN

END

-148-

//ARC	SP015 PRO	C ATNUAT='EATNUA	T', DEST=A,			3	K
11		DMPDEST=F,)	(
11		FULLBLK = 12280, G	RPDSP=" (NH	EW,DELETE) •	,	3	K
11		GRPJCL=',UNIT=(SASCR, SEP=	COUMMY1,DU	MMY2))',	2	(*
11		GRPORD= * & GRPORD	•,HALFBLK=	=6136,		2	K
11		ISOCYL=1, ISODSP	=' (NEW, DEI	.ETE) ',		2	κ.
11		ISOJCL= ,UNIT=S	ASCR', ISOT	XXS= * & ISOTX	s',	2	(
11		MCC2F1='C116.MC	C2F1.MEV14	I.V4°,		3	(
11		MCC2F2='C116.MC	C2Ë2',]	ĸ
11		MCC2F3='C116.MC	C2F3.MEV14	↓ ↓ ↓↓↓		3	ζ.
11		MCC2F4= C116. MC	C2F4.MEV14	L.V4*,		3	(
11	· .	MCC2F5=*C116.MC	C2F5. MEV14	↓.V4 ↓		3	(
11		MCC2F6= C116.MC	C2F6.MEV14	↓.▼4╹,		3	K
11		MCC2F7='C116.MC	C2F7.MEV14	. V4 °,		1	K
11		MCC2F8='C116.MC	C2F8.MEV14	.V4*,		3	(j
11		MCC2VOL=,MICRDS	P= ' (NEW, DI	ELETE) ', MIC	KVOL≑,	3	C
11		MICRXS1=*&MICRX	S1 MICRXS	S2=•6MĪĊRXS	2',	2	K
11		MODLIB1= C116.E	09202.MODI	ΙΒ ' ,		2	K
11		MODLIB2="C116.A	RC. MODLIB			2	(
11		OLDSGS=NULLFILF	,OLDSVOL=	PATH= STP01	5,	1	K
11		POSTLIB= SYS1. I	UMMYLIB.	RELIB='SYS	1. DUMMYLIB',	2	X
11		REGN=700K, RESBI	K=12280, RI	ESDSP= • (NEW	, DELETE) ',	3	C
11		RESINT=*&RESINI	RESJCL=	,UNIT=SASC	R',	3	ĸ
11 .		SCATBLK=12280,S	CATDSP=! (I	NEW, DELETE)	۰,	3	Κ.
11		SCATJCL=',SUBAI	LOC= (CYL,	(20,3),DUMM	¥2)',	3	K
11		SMSIGS='&SMSIGS	SPECDSP=	• (NEW, DELE	TE)',	:	χ.
11		SPECTR=* 6SPECTE	. SPECJCL	- ,UNIT=SAS	CR',	3	K
11		TIMLIM= ' (600,0)	,UNITS=B	ATCHDSK, UNI	TSCR=SASCR,	2	K
11		UNRES= '& UNRES '	XSBLK1=31	56 , XSBLK2=6	447		
//*							
//*	* * * * * *	* * * * * * * * * * * * * * * * * * *	******	*****	*****	** * * * * * * *	
//*	*					*	
//*	*	CATALOGED F	ROCEDURE	$rac{1}{2}$ or $MC \approx 2-2$	PROBLEMS	*	
//*	*					*	
//*	*****	* * * * * * * * * * * * * * * * * * * *	****	*****	****	****	
//*	بىلەرىلەر بىلەرىلەر بىلەرىلەر بىلەرىلەر				********	********	*****
//*	*****	* * * * * * * * * * * * * * * * * * * *	******	• • • • • • • • • • • • • •	******	****	*****
//*	D 1 D 1 M F T T D	0 F F A 11 T	VATUE		USACE		FTNNF001
//*					=====	•	========
//*							
1/*	ратн	STP0 15		p	ROGRAM NAME		EXEC
//*	TTMLTM	(600-0)		ŝ	TED TIME LIM	TT	EXEC
//*	REGN	700K		S	TEP REGION S	IZE	EXEC
1/*	MODITRI	C116_B09202	MODITB	T T	IRST REAL IT	BRARY	STEPLIB
//*	MODI.T B2	C116_ARC_ MC	DLTB		RC SYSTEM LT	BRARY	STEPLIB
//*	PRELTR	SYS1.DUMM VI	IB.	רי ד	IRST STEP LT	BRARY	STEPLIB
//*	POSTLIB	SYS1.DUMMYT	IB	Ĺ	AST STEP LTB	RARY	STEPLIB
//*	DEST	A		0	UTPUT DEST.	(PRINTER)	06
1/*	DMPDEST	F		R	OUTE DUMP TO	FICHE	SYSUDUMP

.

Fig. 11. MC²-2 JCL Procedure

.

.

//*	GRPORD	EGRPORD	DSN FOR FILE GRPORD	18
//*	GRPDSP	(NEW, DELETE)	DISPOSITION OF GRPORD	18
//*	GRPJCL	.UNIT= (SASCR.SEP= (DUMMY1.DUMM)	Y2)) GRPORD UNIT & VOL	18
//*	ISOTXS	EISOTXS	DSN FOR FILE ISOTXS	19
//*	ISODSP	(NEW, DELETE)	DISPOSITION OF ISOTXS	19
//*	ISOJCL	UNIT=SASCR	ISOTXS UNIT AND VOLUME	19
//*	ISOCYL	1	NO. CYL FOR X SECTIONS	19,50
//*	XSBLK2	6447	X SECTION BLOCKING	19,50
//*	MCC2F1	C116.MCC2F1.MEV14.V4	LIBRARY FILE 1 DSN	22
//*	MCC2F2	C116.MCC2F2	LIBRARY FILE 2 DSN	23
//*	MCC2F3	C116.MCC2F3.MEV14.V4	LIBRARY FILE 3 DSN	24
//*	MCC2F4	C116.NCC2F4.MEV14.V4	LIBRARY FILE 4 DSN	25
//*	MCC2F5	C116.MCC2F5.MEV14.V4	LIBRARY FILE 5 DSN	26
//*	MCC2F6	C116.MCC2F6.MEV14.V4	LIBRARY FILE 6 DSN	2 7
//*	MCC2F7	C116.MCC2F7.MEV14.V4	LIBRARY FILE 7 DSN	28
//*	MCC2F8	C116.MCC2F8.MEV14.V4	LIBRARY FILE 8 DSN	29
/1*	MCC2VOL		LIBRARY VOLUME	22-29
//*	OLDSGS	NULLFILE	RESTART ELASTIC MATRICE	S 31
//*	OLDSVOL		VOLUME FOR OLDSGS	31
//*	SMSIGS	ESMSIGS	PROBLEM ELASTIC MATRICE	S 43
//*	SCATBLK	12280	BLKSIZE FOR SMSIGS	43
//*	SCATDSP	(NEW,DELETE)	DISPOSITION OF SMSIGS	43
//*	SCATJCL	, SUBALLOC = (CYL, (20, 3), DUMMY2)	SMSIGS UNIT, SPACE & VOL	43
//*	SPECTR	& S P E C T R	DSN FOR SPECTRUM FILE	44
//*	SPECDSP	(NEW,DELETE)	DISPOSITION OF SPECTRUM	44
//*	SPECJCL	, UNIT=SASCR	SPECTRUM UNIT AND VOLUM	E 44
//*	MICRXS1	&MICRXS1	XS.ISO FILE 1 DSN	49
//*	XSBLK1	3156	BLOCKING FOR EMICRXS1	49
//*	MICRXS2	EMICRXS2	XS.ISO FILE 2 DSN	50
//*	MICRDSP	(NEW, DELETE)	DISPOSITION OF XS.ISO	49-50
//*	MICRVOL	* - * * * *	VOLUME OF FILE XS.ISO	49-50
//*	ATNUAT	SATNUAT	DSN FOR FILE ATNUAT	14
//*	RESINT	ERESINT	DSN FOR FILE RESINT	36
//*	UNRES	EUNRES	DSN FOR FILE UNRES	48
//*				
//*	THE FOLLOWING	G THREE PARAMETERS ARE APPLICAT	BLE TO THE THREE	
//* //*	RESUNANCE FIL	LES ATNUAT, RESINT AND UNRES		•
//*	RESBLK	12280	FILE BLKSTZE	14.36.48
//*	RESDSP	(NEW-DELETE)	FILE DISPOSITION	14.36.48
//*	RESJCL	UNIT=SASCR	FILE UNIT AND VOLUME	14.36.48
//*				
//*	THE FOLLOWING	THREE PARAMETERS DEFINE UNIT	AND BLKSIZE FOR A	
//*	VARIETY OF FI	LES		
//*	FULLBLK	12280	FULL TRACK BLOCKING	
//*	HALFBLK	6136	HALF TRACK BLOCKING	
//*	UNITS	BATCHDSK	GENERIC UNIT NAME	
//*	UNITSCR	SASCR	GENERIC SCRATCH UNIT	
//*				

Fig. 11. MC²-2 JCL Procedure (Contd.)

.

.

`

-

,

.

.

//* //* //STP015 EXEC PGM=&PATH,TIME=&TIMLIM,REGION=®N //STEPLIB DD DSN=&PRELIB, DISP=SHR DD DSN=&MODLIB1, DISP=SHR 11 DD DSN=&MODLIB2,DISP=SHR 11 DD DSN=& POSTLIB, DISP=SHR 11 //DUMMY1 DD DSN=&DUM1,UNIT=&UNITSCR,SPACE= (CYL,27,,CONTIG) //DUMMY2 DD DSN=&DUM2,UNIT=&UNITSCR,SPACE=(CYL,55,,CONTIG) //FT05F001 DD DDNAME=SYSIN //* BCD INPUT //FT06F001 DD SYSOUT=&DEST, DCB= (RECFM=FBA, LRECL=133, BLKSIZE=1596) 11 //* PRINTED OUTPUT //FT07F001 DD SYSOUT=B //* PUNCHED OUTPUT //FT09F001 DD UNIT=&UNITSCR, SPACE=(CYL, (1, 1)), DCB = (RECFM = FBS, LRECL = 80, BLKSIZE = 3040)11 //* ARC SYSTEM SPOOLED INPUT FILE //FT11F001 DD DSN=&AMCC2, UNIT=&UNITSCR, SPACE=(TRK, (1,1)), DCB=(RECFM=PBS,LRECL=80,BLKSIZE=3040) 11 //* MCC-2 ALPHANUMERIC INPUT //FT12F001 DD DSN=&ANIP,UNIT=&UNITSCR,SPACE=(TRK,(1,1)), DCB= (RECFM=FBS, LRECL=80, BLKSIZE=3040) 11 //* GENERAL NEUTRONICS ALPHANUMERIC INPUT //FT13F001 DD DSN=&APATH,UNIT=&UNITSCR,SPACE=(TRK, (1,1)), DCB= (RECFM=PBS, LRECL=80, BLKSIZE=320) 11 //* MCC-2 PATH ALPHANUMERIC INPUT //FT14F001 DD DSN=&ATNUAT, DISP=ERESDSPERESJCL. 11 DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&RESBLK), 11 11 SPACE = (CYL, (1, 1))//* UNRESOLVED RESONANCE ATTENUATION FACTORS //FT15F001 DD DSN=&BC,UNIT=&UNITSCR,SPACE=(TRK,(1,1)), 11 DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&FULLBLK) //* BOUNDARY CONDITIONS //FT16F001 DD DSN=&BGRES,UNIT=&UNITSCR,SPACE=(CYL, (1,1)), 11 DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&FULLBLK) //* BROAD GROUP RESONANCE CROSS SECTIONS //FT17F001 DD DSN=&GEOM1,UNIT=&UNITSCR,SPACE=(TRK, (3,1),RLSE), 11 DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&FULLBLK) //* GEOMETRY DATA WRITTEN BY MCC-2 INPUT PROCESSOR //FT18F001 DD DSN=&GRPORD, 11 DISP=&GRPDSP&GRPJCL, 11 SPACE=(CYL, (50, 5), RLSE),11 DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&FULLBLK,DEN=3) 11* GROUP ORDERED INELASTIC AND (N,2N) DATA //FT19F001 DD DSN=&ISOTXS,

Fig. 11. MC²-2 JCL Procedure (Contd.)

DISP=&ISODSP&ISOJCL,

SPACE= (CYL, (&ISOCYL, 1)),

11

11

//		DCB=(RECFM=VBS,LRECL=X,BLKSIZE=GXSBLK2)
//*		BROAD GROUP CROSS SECTION FILE IN CCCC FORMAT ISOTXS
//FT20F001	DD	DSN=ELORENZ, UNIT=EUNITSCR, SPACE=(CYL, (1, 1)),
//		DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&FULLBLK)
//*		LORENTZIAN J-INTEGRALS FOR WIDE RESONANCES
//FT21F001	DD	DSN=&MACTOT,UNIT=&UNITSCR,SPACE=(CYL,(1,1)),
11		DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&FULLBLK)
//*		MACROSCOPIC TOTAL CROSS SECTION
//FT22F001	DD	$DSN = \mathcal{E}MCC2F1$,
11		UNIT=&UNITS,DISP=SHR,
11		VOL=(, RETAIN, SER=&MCC2VOL)
//*		FILE 1 OF MCC-2 LIBRARY
//FT23F001	DD	DSN=&MCC2F2.
//		UNTT=& UNTTS_DISP=SHR.
11		VOL = (, RETAIN, SER = EMCC2VOL)
1/*		FTLE 2 OF MCC-2 LTBRARY
//FT24F001	חח	DSN#£MCC2F3
//11241001	עט	
		VOI - I DEMAIN GED-SHOODIN
//		$\nabla \mathbf{L}^{-} \left(\left\{ \mathbf{r} \mathbf{E} \mathbf{I} \mathbf{X} \mathbf{I} \right\} \right\} = \mathbf{G} \mathbf{R} \mathbf{C} - \mathbf{C} \mathbf{I} \mathbf{T} \mathbf{D} \mathbf{R} \mathbf{D} \mathbf{V}$
//*	תח	FILE 5 OF HUC-2 LIDRARI
//1251001	עע	
//		UNITECUNITS, DISPESHR,
//		VOL= (, RETAIN, SER= EMCC2VOL)
//*		FILE 4 OF MCC-2 LIBRARY
//FT26F001	DD	DSN=&MCC2F5,
//		UNIT=&UNITS, DISP=SHR,
11		VOL=(,RETAIN,SER=&MCC2VOL)
//*		FILE 5 OF MCC-2 LIBRARY
//FT27F001	DD	DSN= EMCC2F6,
11		UNIT=&UNITS, DISP=SHR,
11		VOL= (,RETAIN,SER=&MCC2VOL)
//*		FILE 6 OF MCC-2 LIBRARY
//FT28F001	DD	DSN=&MCC2F7,
11		UNIT=&UNITS, DISP=SHR,
11		VOL= (, RETAIN, SER=&MCC2VOL)
//*		FILE 7 OF MCC-2 LIBRARY
//FT29F001	DD	DSN=&MCC2F8,
		UNIT=&UNITS.DISP=SHR,
		<pre>vol=(,RETAIN,SER=&MCC2VOL)</pre>
//*		FILE 8 OF MCC-2 LIBRARY
//FT30F001	DD	DSN=EMICTOT.UNIT=EUNITSCR.SPACE=(CYL.(1.1)).
//		DCB = (RECFM = VBS - LRECL = X - BLKSTZE = & FULLBLK)
//*		MICROSCOPIC TOTAL CROSS SECTIONS
//FT31F001	ממ	DSN=&OLDSGS
//	20	UNTT=&UNTTS_DISP= (OLD_KEEP)
		VOI = (.RFTAIN SFR= & OIDSVOI)
//		NTCROSCODIC SCAMPRENCE MAMBICES SAVED FROM DEVITOUS
//*	4	RYECHTION, FOR DECENDER BUDDOCEC
// T	תֹת	EVECOTION LOV VESTAVI LAVADODO
// 1321001	עע	DSN=GUPTICL, UNIT=GUNITSCK, SPACE=(CIL, (1, 1)), DCD=(DBCDH=UDC, LDBCL=V, DIKCTED=CDEV, DIK)
//		DCB= (KECLU=VBS, LKECL=X, BLKS12E=GYULLBLK)

Fig. 11. MC²-2 JCL Procedure (Contd.)

.

/

•

-

//* OPTICAL THICKNESS DATA //FT33F001 DD DSN=&PLOTIT,UNIT=&UNITSCR,SPACE=(TRK,(1,1)), DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&FULLBLK) 11 //* SIMPLOTTER DATA FILE //FT34F001 DD DSN=&PRBCHI, UNIT=&UNITSCR, SPACE=(CYL, (1,1)), DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&FULLBLK) 11 //* FISSION SPECTRUM DATA //FT35F001 DD DSN=&PRBSPC,UNIT=&UNITSCR,SPACE=(TRK,(1,1)), DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&FULLBLK) 11 //* MCC-2 GENERAL PROBLEM SPECIFICATIONS //FT36F001 DD DSN=ERESINT, DISP=&RESDSP&RESJCL, 11 DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&RESBLK), 11 SPACE = (CYL, (1, 1))11 //* RESOLVED RESONANCE J-INTEGRALS //FT37F001 DD DSN=&SCR001,SUBALLOC=(CYL, (1, 1), DUMMY1), DCB=(RECFM=VBS,LRECL=X,BLKSIZE=SFULLBLK) 11 //* SCRATCH DATA //FT38F001 DD DSN=&SCR002,SUBALLOC=(CYL, (1, 1), DUMMY1), DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&FULLBLK) 11 //* SCRATCH DATA //FT39F001 DD DSN=&SCR003,SUBALLOC=(CYL, (23, 1), DUMMY1), DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&FULLBLK) 11 //* SCRATCH DATA //FT40F001 DD DSN=&SCR004,SUBALLOC=(CYL, (1, 1), DUMMY1), DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&FULLBLK) 11 //* SCRATCH DATA //FT41F001 DD DSN=&SCR005,SUBALLOC=(CYL, (1, 1), DUMMY1), DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&FULLBLK) 11 //* SCRATCH DATA //FT42F001 DD DSN=&SIGMAP,UNIT=&UNITSCR,SPACE=(CYL,(1,1)), DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&FULLBLK) 11 //* POTENTIAL SCATTERING CROSS SECTIONS //FT43F001 DD DSN=&SMSIGS, DISP=&SCATDSP&SCATJCL, 11 DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&SCATBLK) 11 //* MICROSCOPIC ELASTIC SCATTERING MATRICES //FT44F001 DD DSN=&SPECTR, DISP=& SPECDSP&SPECJCL, SPACE=(TRK, (1,1)), 11 DCB=(RECFN=VBS,LRECL=X,BLKSIZE=&HALFBLK) 11 //* ULTRA-FINE-GROUP FLUX AND CURRENT SPECTRA //FT45F001 DD DSN=&SPECXS,UNIT=&UNITSCR,SPACE=(CYL,(1,1)), DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&FULLBLK) 11 ULTRA-FINE-GROUP MACROSCOPIC CROSS SECTIONS AND //* MODERATING PARAMETERS //* //FT46F001 DD DSN=&SRAIES, UNIT=&UNITSCR, SPACE=(TRK, (1, 1)), DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&FULLBLK) 11 SCATTERING COLLISION DENSITY AND EXTERNAL SOURCE 11* //* FOR INTEGRAL TRANSPORT CALCULATION //FT47F001 DD DSN=&UNREG,UNIT=&UNITSCR,SPACE=(CYL,(1,1)).

Fig. 11. MC²-2 JCL Procedure (Contd.)

-153-

11 DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&FULLBLK) //* UNRESOLVED MICROSCOPIC ULTRA-FINE-GROUP CROSS SECTIONS //FT48F001 DD DSN=&UNRES, DISP=&RESDSP&RESJCL, 11 11 DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&RESBLK), 11 SPACE = (CYL, (1, 1))//* UNRESOLVED RESONANCE CROSS SECTION-ESTAR PAIRS //FT49F001 DD DSN=&MICRXS1, 11 UNIT=&UNITS, VOL=(, RETAIN, SER=&MICRVOL), 11 DISP=&MICRDSP, SPACE=(TRK, (1, 1)), 11 DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&XSBLK1) //* FILE 1 CF BROAD GROUP XS. ISO CROSS SECTION FILE //FT50F001 DD DSN=&MICRXS2, UNIT=&UNITS, VOL= (, RETAIN, SER=&MICRVOL), 11 11 DISP=&MICRDSP,SPACE=(CYL,(&ISOCYL,1)), 11 DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&XSBLK2) //* FILE 2 OF BROAD GROUP XS.ISO CROSS SECTION FILE //FT51F001 DD DSN=&IRESCS,UNIT=&UNITSCR,SPACE=(TRK, (1,1)), DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&FULLBLK) 11 //* BROAD GROUP INTEGRAL TRANSPORT RESONANCE CROSS SECTIONS //* ************ //* THE FOLLOWING 25 FILES (BIGXS1-BIGXSP) ARE SCRATCH FILES TO //* CONTAIN THE MACROSCOPIC ELASTIC SCATTERING MATRICES //* //FT52F001 DD DSN=&BIGXS1,SUBALLOC=(CYL, (2, 1), DUMMY2), 11 DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&HALFBLK) //FT53F001 DD DSN=&BIGXS2,SUBALLOC=(CYL, (2, 1), DUMMY2), DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&HALFBLK) 11 //FT54F001 DD DSN=&BIGXS3,SUBALLOC= (CYL, (2,1),DUMMY2), DCB= (RECFM=VBS, LRECL=X, BLKSIZE=& HALFBLK) 11 //FT55F001 DD DSN=&RTGXS4,SUBALLOC=(CYL,(2,1),DUMMY2), DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&HALFBLK) 11 //FT56F001 DD DSN=&BIGXS5,SUBALLOC=(CYL, (2, 1), DUMMY2), DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&HALFBLK) 11 //FT57F001 DD DSN=&BIGXS6,SUBALLOC=(CYL, (2, 1), DUMMY2), DCB= (RECFM=VBS, LRECL=X, BLKSIZE=& HALFBLK) 11 //FT58F001 DD DSN=&BIGXS7,SUBALLOC= (CYL, (2,1),DUMMY2), DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&HALFBLK) // //FT59F001 DD DSN=&BIGXS8,SUBALLOC=(CYL, (2,1),DUMMY2), DCB= (RECFM=VBS, LRECL=X, BLKSIZE=& HALFBLK) // //FT60F001 DD DSN=&BIGXS9,SUBALLOC=(CYL, (2, 1), DUMMY2), DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&HALFBLK) 11 //FT61F001 DD DSN=&BIGXSA,SUBALLOC=(CYL, (2, 1), DUMMY2), DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&HALFBLK) // //FT62F001 DD DSN=&BIGXSB,SPACE=(CYL,(0,1)),UNIT=&UNITSCR, DCB= (RECFM=VBS, LRECL=X, BLKSIZE=& HALFBLK) // //FT63F001 DD DSN=&BIGXSC,SPACE=(CYL,(0,1)),UNIT=&UNITSCR, DCB= (RECPM=VBS, LRECL=X, BLKSIZE=&HALFBLK) 17 //FT64F001 DD DSN=&BIGXSD,SPACE=(CYL,(0,1)),UNIT=&UNITSCR, DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&HALFBLK) 11

Fig. 11. MC²-2 JCL Procedure (Contd.)

1	/FT65F001	DD	DSN=&BIGXSE,SPACE= (CYL, (0, 1)),UNIT=&UNITSCR,
1	/		DCB=(RECFM=VBS, LRECL=X, BLKSIZE=&HALFBLK)
1	/FT66F001	DD	DSN=&BIGXSF,SPACE= (CYL, (0, 1)),UNIT=&UNITSCR,
1	/		DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&HALFBLK)
1	/FT67F001	DD	DSN=&BIGXSG,SPACE= (CYL, (0, 1)),UNIT=&UNITSCR,
1	/		DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&HALFBLK)
1	/FT68F001	DD	DSN=&BIGXSH, SPACE= (CYL, (0, 1)), UNIT=&UNITSCR,
1	/		DCB= (RECFM=VBS, LRECL=X, BLKSIZE=& HALFBLK)
1	/FT69F001	DD	DSN=&BIGXSI, SPACE= (CYL, (0, 1)), UNIT=&UNITSCR,
1	/		DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&HALFBLK)
1	/FT70F001	DD	DSN=&BIGXSJ,SPACE=(CYL,(0,1)),UNIT=&UNITSCR,
1	/		DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&HALFBLK)
1	/FT71F001	DD	DSN=EBIGXSK, SPACE= (CYL, (0, 1)), UNIT=EUNITSCR,
1	/		DCB= (RECFM=VBS, LRECL=X, BLKSIZE=& HALFBLK)
1	/FT72F001	DD	DŚN=&BIGXSL, SPACE= (CYL, (0, 1)), UNIT=&UNITSCR,
1	/		DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&HALFBLK)
1	/FT73F001	DD	DSN=&BIGXSM,SPACE=(CYL,(0,1)),UNIT=&UNITSCR,
1	1		DCB= (RECFM=VBS, LRECL=X, BLKSIZE=&HALFBLK)
1	/FT74F001	DD	DSN=&BIGXSN,SPACE=(CYL,(0,1)),UNIT=&UNITSCR,
1	/		DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&HALFBLK)
1	/FT75F001	DD	DSN=&BIGXSO,SPACE=(CYL,(0,1)),UNIT=&UNITSCR,
1	/		DCB=(RECFM=VBS,LRECL=X,BLKSIZE=&HALFBLK)
1	/FT76F001	DD	DSN=&BIGXSP,SPACE=(CYL,(0,1)),UNIT=&UNITSCR,
1	/		DCB= (RECFM=VBS, LRECL=X, BLKSIZE=6 HALFBLK)
1	/FT//FUU1	עע	DSN=CRANDOM, SUBALLOC= (CYL, (15,1), DUMMY2), DCB=RECFM=U
1	/ 17/81001	עע	DSN=CRESDAT, UNIT=CUNITSCR, SPACE=(TRR, (T, T)),
	/		$D \subseteq B = (R \subseteq C = V = V = S \cup LR \subseteq C = X \cup R = S \cup R $
1	/* /Em70E001	DD	RESULVED RESUNANCE PARAMETERS
	/ 1/91001	טע	DSN-GACSUV9, UNII-GUNIISCR, SPACE-(IRR, (I, I)),
	/		$\mathbf{M} = \left(\mathbf{n} = \mathbf{n}$
	/* /〒〒00〒001	nn	ALPHANUMERIC INPUT FOR ISUINS EDITOR MODULE CSEUUS
1	/ 1001001	סט	DCD - (DECEN - EBC IDECI - UA DIKSTZE - 3000)
1	/		NINGRATC INDUM ROP VE ISO ROITOR MODULE CSEC12
1		n n	SYCOLD-SDNDDEST
',	/sisobone /*	00	CORP DUMD FROM ABNORMAL TERMINATION
1	/*		CONS SCHI INCH ABNORMAD IBRUINATION
1	/* *:	* * * *	* * * * * * * * * * * * * * * * * * * *
1	/*		
1	, /*		
1	, /* **	****	*****
1	, /*		
1	/ PEND		

Fig. 11. MC²-2 JCL Procedure (Contd.)

C. Problem Specification

The user input to the MC^2-2 code is supplied on various card types from the BCD data sets A.STP015, A.MCC2, A.NIP, ACS009, and ACSE12. These data sets are described in the listings given in Appendix B. Table III summarizes the use of the various BCD data sets. This first release of MC^2-2 lacks a full heterogeneous treatment in that heterogeneous broad group cross sections are not yet generated by CSC009.

The input data deck for the execution of MC^2-2 follows the general ARC System format as described on pages 42-45 of Reference 1. A data set initialization block (DSIB) given the block name "OLD" is used if any data sets are to be used which were generated in previous jobs. In the case of the eight MC^2-2 library files (MCC2F1-MCC2F8), the path driver STP015 has been written so that these files need not be named in a DSIB. The format of these library files which are normally written by the ETOE-2 code are given in Appendix C. One common use of the DSIB with MC^2-2 is the addition of material cross sections to an existing XS.ISO file by the module CSE007. In this case the card DATASET=XS.ISO appears in the dataset initialization block.

The DSIB, if any, is followed in the input deck by a data block given the name STP015. This block may include the path dependent data set A.STP015 and the data sets A.MCC2, A.NIP, ACS009, and ACSE12. These data sets are discussed below in greater detail and examples of BCD input data are given in Section D, Sample Problems.

The BCD input processor has been written with particular attention to easing the burden on the user when preparing problem input. By using defaults, many of the card types may be omitted for a given problem. Thus for example, as a minimum for homogeneous problems, only type 06 cards of data set A.MCC2 need be supplied if the defaults supplied are acceptable to the user. Similarly, for heterogeneous problems, as a minimum one type 03 card of data set A.MCC2 and the type 06, 14, and 15 cards of data set A.NIP need be supplied.

In order to ease user specification work load, eleven prestored broad group structures are available on option in MC^2 -2. These structures are selected as indicated on card type 03, cols. 31-36. If any type 05 or type 07 cards are supplied, the card type 03 selection, if any, is ignored. Similarly, if an old XS.ISO data set is included in the DSIB as discussed above, the card type 03 selection is ignored. If card type 03 is not supplied or if cols. 31-36 are blank and no type 05 or 07 cards are supplied, a built in default structure will be used.

The group structures for each of the eleven sets are given in Table IV which lists the broad group lethargy widths and upper energy boundaries. The upper energy of all of the sets is 10 MeV except for set USS226 which has an upper energy of 14.190675 MeV. Note that the last group for the BOND26 and the last two groups for the HANS16 sets have been changed relative to the original structures. This was required due to the fact that the current MC^2-2 library has a lower energy of 0.41399 eV. Also, the third from the last group in the sets USS212 and USS226 was altered slightly in order to fall on an integral multiple of an MC^2-2 ultra fine group.

TABLE III. BCD Input Data Sets

Data Set Name

A.STP015

Condition

Needed only if various of the code Areas 4 - 9 are to be omitted, if RABANL is to be executed, if broad group cross sections are not to be edited, if data set XS.ISO is not to be written, or if XS.ISO is to be edited.

Always present unless only data set XS.ISO and/or ISOTXS is to be edited.

Needed only if a heterogeneous problem is involved.

Needed only if data set XS.ISO is to be edited, and then only if a nonstandard edit is desired.

Needed only if data set ISOTXS is to be edited, and then only if a nonstandard edit is desired.

A.MCC2

A.NIP

ACSE12

ACS009

	ANL9		NL11	<u> </u>	NL27	A	NL28	BOND26 Lethargy	
Group	Lethargy Width	Group	Lethargy Width	Group	Lethargy Width	Group	Lethargy Width	Group	Lethargy Width
1	1.5	1	1.0	1	0.5	1	0.5	1	0.425
2	1.0	2	1.0	2	0.5	2	0.5	2	0.483333
3	1.5	3	1.0	3	0.5	3	0.5	3 `	0.475
4	1.5	4	1.0	4	0.5	4	0.5	4	0.575
5	1.5	5.	1.0	5	0.5	5	0.5	5	0.56666
6	1.5	6	1.0	6	0.5	6	0.5	6	0.6916666
7	1.5	7	1.0	7	0.5	7	0.5	7	0.6916666
8	4.5	8	1.0	8	0.5	8	0.5	8.	0.6916666
9	œ	9	2.0	9	0.5	9	0.5	9	0.76666
•		10	7.0	10	0.5	10	0.5	10	0.775
		11	8	11	0.5	11	0.5	11	0.758333
				12	0.5	12	0.5	12	0.76666
				13	0.5	13	0.5	13	0.775
				14	0.5	14	0.5	14	0.76666
				15	0.5	15	0.5	15	0.76666
	•			16	0.5	16	0.5	16	0.76666
				17	0.5	17	0.5	17	0.76666
				18	0.5	18	0.5	18	0.76666
				19	0.5	19	0.5	19	0.775
• •				20	0.5	20	0.5	20	0.758333
				21	0.5	21	0.5	21	0.76666
				22	1.0	22	0.5	22	0.775
				23	1.0	23	0.5	23	0.766666
				24	1.0	24	0.5	24	0.766666
				25	1.0	25	0.5	25	0.116666
				26	1.0	26	0.5	26	œ
				27	œ	27 .	1.5		
	-					28	œ		
									•

1

١

TABLE IV. Prestored Broad Group Structures

TABLE IV . Contd.

FI			ANS16			USS	5212*		
Group	Lethargy Width	Group	Lethargy Width	Group	Lethargy _Width	Group	Lethargy Width	Group	Lethargy Width
1	0.5	1	1.2	1	0.025	34	0.05	67	0.05
2	0.5	2	0.758333	2	0.025	35	0.05	68	0.05
3	0.5	3	0.441666	3	0.025	36	0.05	69	0.025
4	0.5	4	0.81666	4	0.025	37	0.05	. 70	0.025
5	0.5	5	1.38333	5	0.025	38	0.025	71	0.05
6	0.5	6	1.775	6	0:025	39	0.025	72	0.05
7	0.5	7	1.73333	7	0.025	40	0.025	73	0.05
8	0.5	8	1.691666	8	0.025	41	0.025	74	0.025
9	0.5	9	1.708333	9	0.025	42	0.05	75	0.016667
10	0.5	10	1.208333	10	0.025	43	0.05	76	0.008333
11	0.5	11	1.091666	11	0.025	44	0.05	77	0.05
12	0.5	12	1.208333	12	0.025	45	0.05	78	0.05
13	0.5	13	1.1	13	0.025	46	0.016667	79	0.05
14	0.5	14	0.766666	14	0.025	. 47	0.016667	80 ·	0.05
15	0.5	15	0.116666	15	0.025	48	0.008333	81	0.05
16	0.5	16	œ	16	0.025	49	0.008333	82	0.05
17	0.5			17	0.008333	50	0.016667	83	0.05
18	0.5			18	0.008333	· 51	0.016667	84	0.05
19	0.5			· 19	0.008333	52	0.016667	85	0.05
20	0.5			20	0.025	53	0.05	86	0.05
21	0.5			21	0.025	54	0.05	87	0.05
22	0.5		· .	· 22	0.025	55	0.025	88	0.025
23	0.5			23	0.025	56	0.025	89	0.025
24	0.5	,		24	0.025	57	0.025	90	0.025
25	0.5			25	0.05	58	0.025	91	0.025
26	1.0			26	0.05	59	0.05	92	0.05
27	1.0		•	27	0.05	60	0.05	93	0.05
28 [°]	1.0			28	0.025	61	0.025	94	0.05
29	1.0			29	0.025	62	0.025	95	0.05
30	8			30	0.025	63	0.025	96	0.05
				31	0.025	. 64	0.025	97	0.05
		•		32	0.05	65	0.05	98	0.05
				33	0.05	66	0.05	99	0.05

.

TABLE IV. Contd.

				USS212*			
Group	Lethargy Width	Group	Lethargy Width	Group	Lethargy Width	Group	Lethargy Width
100	0.05	133	0.05	166	0.058333	199	0.25
101	0.025	134	0.075	167	0.041667	200	0.25
102	0.025	135	0.125	168	0.05	201	0.25
103	0.025	136	0.125	169	0.05	202	0.25
104	0.025	137	0.125	170	0.1	203	0.25
105	0.025	138	0.025	171	0.1	204	0.25
106 ·	0.025	139	0.025	172	0.1	205	0.25
107	0.05	. 140	0.075	173 ·	0.1	206	0.25
108	0.05	141	0.125	174	0.05	207	0.05
109	0.05	142	0.075	175	0.05	208	0.2
110	0.05	143	0.05	. 176	0.1	209	0.08333
111	0.05	144	0.025	177	0.1 .	210	0.16667
112	0.05	145	0.025	178	0.25	211	0.25
113	0.05	146	0.025	179	0.25	212	œ
114	0.05	147	0.025	180	0.25		
115	0.05	148	0.1	181	0.25		
116	0.025	149	0.125	182	0.25		
117	0.025	150	0.125	183	0.25		
118	0.025	151	0.125	184	0.25		
119	0.025	152	0.125	185	0.25		
120	0.05	153	0.125	186	0.25		
121	0.05	154	0.125	187	0.25		
122	0.05	155	0.125	188	0.25		
123	0.05	156	0.125	189	0.25		
124	0.05	157	0.125	190	0.25		
125	0.05	158	0.125	191	0.25		
126 ·	0.05	159	0.1	192	0.25		-
127	0.125	160	0.1	193	0.25		
128	0.125	161	0.05	194	0.25		
129	0.125	162	0.05	195	0.25		
130	0.125	163	0.1	196	0.25		
131	0.075	164	0.1	197	0.25		
132	0.05	165	0.1	198	0.25		

*USS226 is the same as USS212 except that an additional 14 groups with $\Delta u = 0.025$ are added above group 1 up to an upper energy of 14.190675 MeV.

SDX156		W	IARD9
Group	Lethargy Width	Group	Lethargy Width
· 1	0.1	1	1.5
2	0.1	2	1.0
3	0.1	3	1.5
•	•	4	1.5
۵	•	、 5	1.5
•	•	6	1.5
155	. 0.1	7	1.5
156	œ	· 8	6.5
•		9	co '

TABLE IV. Contd.

• • · · ·

D. Sample Problems

A number of examples of typical MC^2-2 problems will be given in this section along with a listing of the user supplied BCD data.

1. Homogeneous Reference Problem

The user input deck is shown in Fig. 13.

This problem creates the XS.ISO data set having the data set names Cl16.BXXXXX.XSISOF1 and Cl16.BXXXXX.XSISOF2 for files 1 and 2 respectively as specified in the symbolic parameters MICRXS1 and MICRXS2. The data set will be catalogued as specified in the symbolic parameter MICRDSP. No data set initialization block is specified.

The type 01 card of data set A.STP015 is set to execute Areas 4, 5, 6, 6.5, 7, 8, and 9, to edit the broad group cross sections, and to create a double precision XS.ISO data set. This could also have been accomplished by supplying a card with only 01 in columns 1 and 2. Also, these same choices could have been selected by omitting the DATASET=A.STP015 card and the type 01 card which follows it since the standard defaults are used.

The data set A.MCC2 type 01 cards provide title information desired by the user.

The type 02 card specifies a BPOINTER main core container of 25000 real*8 words.

The type 03 card uses defaults for all fields except for columns 7-12 and 31-36 in which a consistent P1 fundamental mode calculation and the standard ANL27 broad group structure are specified.

The type 06 cards specify the five isotopes in the problem along with their atom densities. The problem material names were omitted from columns 19-24 and will default to the names given in columns 13-18. Also, the temperatures all default to 300 degrees K and all cross sections will be added to the output cross section data set.

The type 16 card specifies that the fission spectrum of PU2394 be used for all fissionable nuclides.

The type 22 cards classify the problem isotopes and assign values for MeV/fission for U-2384 and PU2394.

2. Hyper-fine Group Homogeneous Integral Transport Problem

The user input deck is shown in Fig. 14.

This problem corresponds to a standalone RABID⁽⁶⁾ calculation for the homogeneous composition given in the previous example.

The type Ol card of data set A.STPO15 is set to execute only the input processor Area 4 and the RABANL module.

As in example 1, a container of 25000 real*8 words is specified on card type 02 of data set A.MCC2.

The same type 06 cards are used as in example 1.

A fixed buckling of 0.0011466 is specified on the type 09 card.

The type 14 card specifies 3354.4 eV as the upper energy for the calculation with all other fields on that card using the default values specified.

The mixture temperature is specified as 300 degress K on the type 21 card.

3. Hyper-fine Group Heterogeneous Integral Transport Problem

The user input deck is shown in Fig. 15.

This problem corresponds to a standalone RABID⁽⁶⁾ calculation for a six slab region heterogeneous problem.

As in example 2, the data set A.STP015 type 01 card is set to execute only Area 4 and RABANL.

A larger container of 30000 real*8 words is specified on the data set A.MCC2 type 02 card. Since this is the default, the type 02 card could have been omitted.

The type 03 card again specifies the ANL27 broad group structure and 1 in column 42 indicates slab geometry.

No data set A.MCC2 type 06 cards are supplied so that the mixture composition will be derived from the data set A.NIP input.

The upper energy of the problem is specified to be 275.36425 electron volts on the type 14 card.

The type 15 cards define two foils.

The type 21 card assigns a temperature of 293 degrees K to each of the compositions defined in the data set A.NIP type 14 cards.

The data set A.NIP type 04 card indicates that the left and right cell boundaries have reflective boundary conditions.

The A.NIP type 06 cards define the dimensions of the three regions in the problem. Because of the reflective boundary conditions, both regions MATX and CLAD 1 will have double the thicknesses as given on the type 06 cards.

The three compositions MATX, U308, and CLADA are defined on the type 14 cards.
The composition - region assignments are made on the type 15 cards.

E. Error Messages

Error messages issued by various subroutines of the modules of MC^2-2 are listed in Table V. The subroutines are presented alphabetically within each module, and the error messages for each subroutine are listed in numerical order.

The convention used is that fatal errors are negative, while non-fatal errors are positive.

As a general rule, fatal errors will not halt execution until the particular module involved has been completed. This will not be true in those cases in which the error results in some subsequent computer system error such as a divide check, core region violation, etc.

Also included are the error messages issued by the ANL version of the CCCC compatible I/O routine REED/RITE. $^{(2)}$

Figure 12

Minimum JCL Execution Deck

//MINJCL JOB (FXXXXX,20,,MM),'YOUR CHOICE',MSGLEVEL=1,CLASS=A, // REGION=600K ACCOUNTING INFORMATION // EXEC ARCSP015 //SYSIN DD * PROBLEM INPUT DATA /*

Figure 13

Input for Sample Problem 1

//SAME	PLE1 JO	DB (F)	XXXXX,	20,,05),'YOUI	R CHOIC	CE',MSO	GLEVEL	=1,CLAS	SS=A,		
11		TNEO	EGIUN=:							•		
	INIING		RMATIO	N								
// CAL	SC ARCI	, CTOJC IM	, [CDVC]-	-10116	BVVVV	7 VCTC	1511					1
		M	LCUNG J-		BVVVV	V VCTC)FI,					
		M	ICKADZ- ICDDCD-	- CIIO		1.1010(JEZ,					
//	מ מת זא	11. k	LCKD3L-	- (NEW	, CALLG,							
BIOCK=		5										
ΠΔΤΔΩ	$\Delta = \Delta$, 										
01	()	0	0	0	0	0	0	0	0	0	
DATASE	ст=А.М(202	0		Ũ	Ŭ	0	v	Ũ	v	•	
01	*****	*****	*****	*****	*****	*****	*****	*****	*****	*****	*****	**
01	*	ENDF	B-IV	DATA								*
01	*	FIVE	ISOTO	PE HOM	OGENEOI	JS REFI	ERENCE	PROBL	ЕМ			*
01	*	BPOI	NTER CO	ONTAIN	ER IS S	SET TO	25000	WORDS				*
01	*	CONS	ISTENT	P1 FU	NDAMEN'	TAL MOI	DE CAL	CULATI	ON			*
01	*	STANI	DARD AL	NL27 B	ROAD GI	ROUP ST	TRUCTU	RE				*
01	*	DEFAU	ULT TEI	MPERAT	URES OI	F 300 I	DEGREES	SK				*
01	*	ALL]	FISSIO	NABLE	ISOTOPI	ES USE	PU239	FISSI	ON SPE	CTRUM		*
01	*	MEV/I	FISSIO	N SET 3	го 201	.815063	L3 FOR	U238				*
01	*	MEV/I	FISSIO	N SET (TO 215	73334	L4 FOR	PU239				*
01	*****	*****	*****	*****	*****	*****	*****	******	*****	*****	*****	**
02	25000)										
03	-	3			ANL2	7						
06		U-238	84	.006	383						•	
06		PU239	94	.0010	086							
06		NA23	4	.010	41						-	
06		0-16	4.	.014	19							
06		FE	4	.018	14							
16	PU2394	' +										
22		U-238	84	2 201	.81506	13		,				
22		PU239	94	1 215	.73334	14	•					
22		NA23	4	6								
22		0-16	4	0								
22		FE	4	5								
/*												

Figure 14

Input for Sample Problem 2

//SA	MPLE2 .	JOB (FXXXX	K,20,,0	5),'Y(OUR CHO	DICE',M	SGLEV	EL=1,CI	ASS=C,	
11		REGIO	₹550K			ŗ		-	F	
ACC	COUNTIN	G INFORMAT	ION							
// E	XEC AR	CSP015								
//SY	SIN DD	*								
BLOC	K=STPO	15								
DATA	SET=A.S	STP015								
01		0 1	1	1	1	1	1	-1	1	
DATA	SET=A.	MCC2								
01	****	******	******	*****	******	*****	*****	******	*****	***
01	*	ENDF/B-IV	/ DATA							*
01	*	FIVE ISO	ГОРЕ НО	MOGENI	EOUS RE	FERENC	E PRO	BLEM		*
01	*	STANDALON	VE HYPE	R-FINI	E-GROUP	INTEG	RAL T	RANSPOF	RT (RABANL)	*
01	*	CALCULAT	LON			• • •			•	*
01	*	BPOINTER	CONTAI	NER IS	S SET T	0 2500	0 WOR	DS	· ·	*
01	*	STANDARD	ANL27	BROAD	GROUP	STRUCT	URE			*
01	*	FIXED BUG	KLING	IS SET	г то .С	011466			•	*
01	*	TOP ENERG	GY OF P	ROBLEN	M IS SE	т то з	354.4	ELECTE	RON VOLTS	*
01	*	MIXTURE 7	EMPERA	TURE]	IS SET	TO 300	DEGR	EES KEL	VIN	*
01	****	******	******	*****	******	*****	****	******	****	***
02	2500	00								
03				A	NL27					
06		U-2384	.00	6383						
06		PU2394	.00	1086						
06		NA23 4	.01	041						
06	•	0-16 4	.01	419						
06		FE 4	.01	814						
09		.0011466								
14		3354.4								
21		2	300.0							
/*										

:

.

Figure 15

Input for Sample Problem 3

//SAMPLE3 JOB (FXXXXX,20,,08), 'YOUR CHOICE', MSGLEVEL=1, CLASS=C, Π REGION=650K ACCOUNTING INFORMATION // EXEC ARCSP015 //SYSIN DD * BLOCK=STP015 DATASET=A.STP015 1 -1 1 1 1 1 01 0 1 1 DATASET=A.MCC2 01 01 * ENDF/B-IV DATA * 01 SIX REGION HETEROGENEOUS PROBLEM ¥ * STANDALONE HYPER-FINE-GROUP INTEGRAL TRANSPORT (RABANL) × 01 * 01 * CALCULATION BPOINTER CONTAINER IS SET TO 30000 WORDS * 01 * * STANDARD ANL27 BROAD GROUP STRUCTURE 01 * * 01 * TOP ENERGY OF PROBLEM IS SET TO 275.36426 EV * ALL COMPOSITION TEMPERATURES SET TO 293 DEGREES K 01 * ******* 01 02 30000 03 ANL27 1 275.36425 14 15 FOILA U-2354 .00003 .00036 PU2414 .000025 .00044 15 FOILB PU2394 .00015 21 MATX 293. U308 293. CLADA 293. DATASET=A.NIP 04 10 10 1 06 0.0 0.286 MATX 0.921 1 06 **U308** 0.286 06 CLAD 1 0.921 0.959 1 4 0.00005 0-16 4 0.00025 ΓE 4 0.04474 14 MATX MO 4 0.005479 MATX CR 4 0.01257 MN55 4 0.00101 14 NI U-2354 0.0000336 U-2384 0.01572 0-16 4 0.04201 14 U308 14 CLADA NI 4 0.009832 CR 4 0.01939 MN55 4 0.00147 14 CLADA MO 4 0.000075 0 - 16 40.00038 FE 4 0.06811 15 MATX MATX 15 U308 U308 15 CLADA CLAD 1 /*

TABLE V. MC²-2 Error Messages

CSI010 (AREA 4)

٠.,

Subroutine CARD05

,

Fatal Error -10100.	CARD PRESUMED TO BE TYPE 05 NOT FOUND
Fatal Error -10200.	THE FIRST GROUP NUMBER AND UPPER ENERGY OF THAT GROUP MUST BE GIVEN ON THE FIRST TYPE 05 CARD
Fatal Error -10300.	BROAD GROUP ENERGIES MUST ALL BE SPECIFIED
Fatal Error -10400.	THE HIGHEST ENERGY BROAD GROUP IN THE PROBLEM MAY NOT LIE ABOVE THE HIGHEST ENERGY IN THE LIBRARY
Fatal Error -10500.	ALL GROUPS MUST BE SPECIFIED
Fatal Error -10600.	BROAD GROUP ENERGIES MUST ALL BE UNIQUE
Fatal Error -10700.	THE SPECIFIED BROAD GROUP STRUCTURE EXTENDS BELOW THE BOTTOM OF THE LIBRARY GROUP STRUCTURE
Fatal Error -10800.	IF ONLY ONE UPPER ENERGY IS GIVEN, IT MUST CORRESPOND TO GROUP 1
Fatal Error -10900.	THE HIGHEST BORAD GROUP SPECIFIED LIES BELOW THE LOWEST ENERGY IN THE LIBRARY
Fatal Error -11000.	THE ADJUSTED BROAD GROUP ENERGY BOUNDARIES MUST ALL BE UNIQUE.
Subroutine CARI	006
Fatal Error -10100.	CARD PRESUMED TO BE TYPE 06 NOT FOUND
Fatal Error -10200.	EACH MATERIAL MUST CORRESPOND TO SOME LIBRARY MATERIAL
Fatal Error -10400.	MATERIAL NAMES MUST BE NON-BLANK
Subroutine CARI	007
Fatal Error -10100.	CARD PRESUMED TO BE TYPE 07 NOT FOUND
Fatal Error -10200.	LETHARGY WIDTHS SPECIFIED ON TYPE 07 CARDS MUST BE GREATER THAN ZERO
Fatal Error -10300	FINAL BROAD GROUP NUMBERS IN COLS. 31-36 ON TYPE 07 CARDS MUST BE .GE. INITIAL BROAD GROUP NUMBERS GIVEN IN COLS. 25-30

Ŋ

CSI010 (AREA 4) (Contd.)

Subroutine CARD07 (contd.)

Fatal Error -10400. FINAL BROAD GROUP NUMBERS IN COLS. 55-60 ON TYPE 07 CARDS MUST BE .GE. INITIAL BROAD GROUP NUMBERS GIVEN IN COLS. 49-54

Fatal Error -10500. THE FIRST BROAD GROUP NUMBER MUST BE 1

Fatal Error -10600. THE BROAD GROUP NUMBERS MUST ALL BE SPECIFIED

Fatal Error -10700. THERE MAY BE NO MORE BROAD GROUPS THAN THERE ARE ULTRA FINE GROUPS IN THE LIBRARY

Fatal Error -10800. THE PRESTORED BROAD GROUP STRUCTURE SELECTED IS NOT CONSISTENT WITH THE LIBRARY ENERGY STRUCTURE

Fatal Error -10900. THE PRESTORED BROAD GROUP STRUCTURE SELECTED IS NOT CONSISTENT WITH THE LIBRARY ENERGY STRUCTURE

Subroutine CARD08

Fatal Error -10100. CARD PRESUMED TO BE TYPE 08 NOT FOUND

Fatal Error -10200. LOWER ENERGY BROAD GROUP NUMBER IN COLS. 31-36 ON TYPE 08 CARDS MUST BE .GE. HIGHER ENERGY BROAD GROUP NUMBER IN COLS. 25-30

Fatal Error -10300. LOWER ENERGY BROAD GROUP NUMBERS IN COLS. 55-60 ON TYPE 08 CARDS MUST BE .GE. HIGHER ENERGY BROAD GROUP NUMBERS IN COLS. 49-54

Fatal Error -10400. MATERIAL SPECIFIED IN COLS. 25-30 ON TYPE 08 CARD DOES NOT CORRESPOND TO ANY FISSION SPECTRUM MATERIAL NAME ON DATA SET MCC2F7

Fatal Error -10500. ONLY INITIAL ENERGY INDEPENDENT FISSION SPECTRA MAY BE SPECIFIED ON TYPE 08 CARDS

Fatal Error -10600. MATERIAL SPECIFIED IN COLS. 25-30 ON TYPE 08 CARD DOES NOT CORRESPOND TO ANY PROBLEM MATERIAL

Subroutine CARD09

Fatal Error -10100. CARD PRESUMED TO BE TYPE 09 NOT FOUND

Fatal Error -10200. THE ULTRA FINE GROUP NUMBER IN COLS. 55-60 ON TYPE 09 CARDS MUST BE GREATER THAN OR EQUAL TO THE ULTRA FINE GROUP NUMBER IN COLS. 49-54

Fatal Error -10300.

CARD PRESUMED TO BE TYPE 09 NOT FOUND

-169-

TABLE V. Contd.

CSI010 (AREA 4) Contd.

Subroutine CARD09 (contd.)

Fatal Error -10400. THE ULTRA FINE GROUP NUMBERS IN COLS. 49-54 ON TYPE 09 CARDS MUST BE .GT.0

Fatal Error - 10500. THE ULTRA FINE GROUP NUMBER IN COLS. 55-60 ON TYPE 09 CARDS MUST BE GREATER THAN OR EQUAL TO THE ULTRA FINE GROUP NUMBER IN COLS. 49-54

Error 10600. THE CONVERGENCE CRITERION ON THE TYPE 09 CARD MUST BE .GE. 0,

Fatal Error -10700. BUCKLING ITERATION IS NOT POSSIBLE FOR AN INHOMOGENEOUS SOURCE PROBLEM

Fatal Error -10800. BUCKLING ITERATION IS NOT ALLOWED FOR GROUP DEPENDENT BUCKLINGS

Fatal Error -10900. BUCKLING ITERATION IS NOT ALLOWED FOR INHOMOGENEOUS SOURCE PROBLEMS

Subroutine CARD10

Fatal Error -10100. CARD PRESUMED TO BE TYPE 10 NOT FOUND

Fatal Error -10200. NUCLIDE IDENTIFICATION LABELS ON TYPE 10 CARDS MUST CORRESPOND TO SOME PROBLEM MATERIAL

Subroutine CARD11

Fatal Error -10100. CARD PRESUMED TO BE TYPE 11 NOT FOUND

Fatal Error -10200. NUCLIDE IDENTIFICATION LABELS ON TYPE 11 CARDS MUST CORRESPOND TO SOME PROBLEM MATERIAL

Subroutine CARD12

Fatal Error -10100. CARD PRESUMED TO BE TYPE 12 NOT FOUND

Fatal Error -10200. MATERIAL NAMES MUST BE NON-BLANK

Error 10300. MATERIALS SPECIFIED ON TYPE 12 CARDS MUST BE PRESENT IN THE PIN REGION

Error 10400. MATERIALS SPECIFIED ON TYPE 12 CARDS MUST BE RESONANCE MATERIALS

Fatal Error -10500. ALL MATERIALS SPECIFIED ON TYPE 12 CARDS MUST CORRESPOND TO SOME PROBLEM MATERIAL

Fatal Error -10600. CARD PRESUMED TO BE TYPE 12 NOT FOUND

CSI010 (AREA 4) Contd.

Subroutine CARD12 (contd.)

- Fatal Error -10700. MATERIAL NAMES MUST BE NON-BLANK
- Fatal Error -10800. CARD PRESUMED TO BE TYPE 12 NOT FOUND
- Fatal Error -10900. MATERIAL NAMES MUST BE NON-BLANK
- Error 11000. ONLY ONE TYPE 12 CARD MAY BE SUPPLIED FOR A GIVEN MATERIAL WITH COLS. 13-18 BLANK
- Fatal Error -11100. ALL REGIONE REFERRED TO IN COLS. 19-24, 31-36, 43-48, 55-60, AND 67-72 ON TYPE 12 CARDS MUST CORRESPOND TO SOME REGION ON THE DATA SET A.NIP TYPE 06 CARDS
- Fatal Error -11200. ALL MATERIALS REFERRED TO ON TYPE 12 CARDS MUST CORRESPOND TO SOME PROBLEM MATERIAL
- Fatal Error -11300. ALL REGIONS REFERRED TO IN COLS. 13-18, 25-30, 37-42, 49-54, AND 61-66 ON TYPE 12 CARDS MUST CORRESPOND TO SOME REGION DEFINED ON THE DATA SET A.NIP TYPE 06 CARDS
- Fatal Error -11400.
 NO REGION REFERRED TO IN COLS. 19-24, 31-36, 43-48, 55-60, OR 67-72 ON TYPE 12 CARDS MAY ALSO BE REFERRED TO IN COLS. 13-18, 25-30, 37-42, 49-57, OR 61-66

Subroutine CARD14

Fatal Error -10100. CARD PRESUMED TO BE TYPE 14 NOT FOUND

- Fatal Error -10300. THE ENERGY SPECIFIED IN COLS. 13-25 OF CARD TYPE 14 OR THE DEFAULT VALUE OF 300.0 VOLTS FALLS INTO THE THERMAL GROUP
- Fatal Error -10400. THE ENERGY SPECIFIED IN COLS. 13-25 OF CARD TYPE 14 OR THE DEFAULT VALUE OF 300.0 VOLTS FALLS ABOVE THE HIGHEST ENERGY IN THE PROBLEM

Subroutine CARD15

- Fatal Error -10100. CARD PRESUMED TO BE TYPE 15 NOT FOUND
- Fatal Error -10200. FOIL LABELS MUST BE NON-BLANK
- Fatal Error -10300. THE FIRST MATERIAL LABEL ON A TYPE 15 CARD WAS BLANK
- Fatal Error -10400. MATERIALS IN FOILS MUST CORRESPOND TO LIBRARY MATERIAL
- Fatal Error -10500. MATERIALS MAKING UP A GIVEN FOIL MUST BE DISTINCT
- Fatal Error -10600. CARD PRESUMED TO BE TYPE 15 NOT FOUND

-171-TABLE V. Contd.

CSI010 (AREA 4) Contd.

Subroutine CARD15 (contd.)

Fatal E	rror -107(0. FOIL	LABELS	MUST	BE	NON-I	BLANK
---------	------------	---------	--------	------	----	-------	-------

- Fatal Error -10800. THE FIRST MATERIAL LABEL ON A TYPE 15 CARD WAS BLANK
- Fatal Error -10900. POSITIVE FOIL THICKNESSES MUST BE SPECIFIED
- Fatal Error -11000. POSITIVE FOIL THICKNESSES MUST BE SPECIFIED

Subroutine CARD16

- Fatal Error -10100. CARD PRESUMED TO BE TYPE 16 NOT FOUND
- Fatal Error -10200. FISSION SPECTRUM NUCLIDE IDENTIFICATION LABELS ON TYPE 16 CARDS MUST CORRESPOND TO LIBRARY FISSION SPECTRA LABELS
- Fatal Error -10300. ONLY ONE TYPE 16 CARD MAY BE SPECIFIED WITH COLS. 13-18 BLANK
- Fatal Error -10400. FISSIONABLE NUCLIDE IDENTIFICATION LABELS ON TYPE 16 CARDS MUST CORRESPOND TO PROBLEM NUCLIDE IDENTIFICATION LABELS

Subroutine CARD17

Fatal Error -10100. CARD PRESUMED TO BE TYPE 17 NOT FOUND

- Fatal Error -10200. LETHARGY WIDTHS SPECIFIED ON TYPE 17 CARDS MUST BE GREATER THAN ZERO
- Fatal Error -10300. FINAL FIXED MESH POINT NUMBERS IN COLS. 31-36 ON TYPE 17 CARDS MUST BE .GE. INITIAL FIXED . MESH POINT NUMBERS GIVEN IN COLS. 25-30
- Fatal Error -10400. FIXED MESH POINT NUMBERS IN COLS. 55-60 ON TYPE 17 CARDS MUST BE .GE. INITIAL MESH POINT NUMBERS GIVEN IN COLS. 49-54
- Fatal Error -10500. THE FIRST FIXED MESH POINT NUMBER MUST BE 1
- Fatal Error -10600. THE FIXED MESH POINT NUMBERS MUST ALL BE SPECIFIED
- Fatal Error -10700. A MAXIMUM OF 500 FIXED ENERGY MESH POINTS ARE PERMITTED

Fatal Error -10800. THE UNRESOLVED ENERGY REGION FALLS BELOW THE FIXED ENERGY GRID

CSI010 (AREA 4) Contd.

Subroutine CARD18

Fatal	Error	-10100.	CARD	PRESUMED	TО	BE	TYPE	18	NOT	FOUN
ralat	CILUI	-IUIU0.	CARD	TUTOOUTTD	10	נוע		T O	TOT	1.001

Error 10200. ALL MATERIALS NAMED ON TYPE 18 CARDS MUST CORRESPOND TO SOME PROBLEM MATERIAL

Subroutine CARD21

Fatal Error -10100. CARD PRESUMED TO BE TYPE 21 NOT FOUND

Fatal Error -10200. CARD PRESUMED TO BE TYPE 21 NOT FOUND

Fatal Error -10300. EACH COMPOSITION LABEL ON TYPE 21 CARDS MUST CORRESPOND TO SOME COMPOSITION LABEL ON DATA SET A.NIP TYPE 14 CARDS

Fatal Error -10400. THE FIRST COMPOSITION LABEL ON THE TYPE 21 CARDS MUST BE NON-BLANK

Subroutine CARD22

Fatal	Error	-10100.	CARD	PRESUMED	то	ΒE	TYPE	22	NOT	FOUND
-------	-------	---------	------	----------	----	----	------	----	-----	-------

Fatal Error -10200. MATERIAL LABELS ON TYPE 22 CARDS MUST CORRESPOND TO LIBRARY MATERIALS

Subroutine CARD23

Fatal Error -10100. CARD PRESUMED TO BE TYPE 23 NOT FOUND

Fatal Error -10200. MORE THAN NPRMAT+MSORS FISSION SPECTRA TEMPERATURES WERE SPECIFIED ON THE TYPE 23 CARDS

Fatal Error -10300. THE MATERIALS NAMED ON THE TYPE 23 CARDS MUST CORRESPOND TO MATERIALS SPECIFIED ON THE TYPE 06 CARDS AND/OR THE MATERIALS SPECIFIED ON THE DATA SET A.NIP TYPE 14 CARDS, OR THE LABELS FOR THE LIBRARY FISSION SPECTRA DATA

Subroutine CSI010 (MAIN)

Fatal Error -10100.	INPUT DATA SET A.MCC2 NOT FOUND
Fatal Error -10200.	CARD PRESUMED TO BE TYPE 02 NOT FOUND
Fatal Error -10300.	CARD PRESUMED TO BE TYPE 04 NOT FOUND
Fatal Error -10400.	CARD PRESUMED TO BE TYPE 03 NOT FOUND
Fatal Error -10500.	INPUT DATA SET A.NIP NOT FOUND

-172-

-173-

CSI010 (AREA 4) Contd.

CSI010 (MAIN) (contd.)

FATAL ETTOT -10600. CARD PRESUMED TO BE TIPE OF NO	NOT FOUND
--	-----------

Fatal Error -10700. TYPE 06 CARDS MUST BE SUPPLIED FOR HOMOGENEOUS PROBLEMS

Fatal Error -10800. CARD PRESUMED TO BE TYPE 19 NOT FOUND

Fatal Error -10900. EXTENDED TRANSPORT APPROXIMATION ORDER MUST BE 1 FOR INCONSISTENT FUNDAMENTAL MODE OPTION

Fatal Error -11000. LEGENDRE ELASTIC SCATTERING TRANSPORT APPROXIMATION PERMITTED ONLY FOR NON-CONSISTENT FUNDAMENTAL MODE OPTIONS

Fatal Error -11100. CARD PRESUMED TO BE TYPE 20 NOT FOUND

Fatal Error -11200. THE BROAD GROUP STRUCTURE SPECIFIED ON THE TYPE 03 CARD DOES NOT CORRESPOND TO ONE OF THE PRESTORED GROUP STRUCTURES

Subroutine RDANIP

- Fatal Error -10100. CARD PRESUMED TO BE TYPE 04 OF DATA SET A.NIP NOT FOUND
- Error 10200. ONLY REFLECTIVE OR PERIODIC BOUNDARY CONDITIONS ARE ALLOWED AND THE SAME CONDITION APPLIES TO BOTH SIDES OF THE CELL. THE PERIODIC CONDITION IS BEING USED
- Fatal Error -10300. DATA SET A.NIP TYPE 06 CARDS MUST BE SUPPLIED IF COLS. 37-42 ON CARD TYPE 03 OF DATA SET A.MCC2 ARE NON-ZERO
- Fatal Error -10400. DATA SET A.NIP TYPE 14 CARDS MUST BE SUPPLIED IF COLS. 37-42 ON CARD TYPE 03 OF DATA SET A.MCC2 ARE NON-ZERO
- Fatal Error -10500. DATA SET A.NIP TYPE 15 CARDS MUST BE SUPPLIED IF COLS. 37-42 ON TYPE 03 CARD OF DATA SET A.MCC2 ARE NON-ZERO

Error 10600. ONLY THE REFLECTIVE BOUNDARY CONDITION IS ALLOWED AT THE LEFT BOUNDARY OF A CYLINDRICAL CELL

Error 10700. ONLY WHITE BOUNDARY CONDITIONS ARE ALLOWED FOR THE RIGHT BOUNDARY OF A CYLINDRICAL CELL.

CSI010 (AREA 4) Contd.

Subroutine STRTCH

Fatal Error -10100. IF THE ATOMIC DENSITY OF SOME MATERIAL IS NOT SUPPLIED ON THE TYPE 06 CARDS, THE MATERIAL MUST APPEAR IN SOME HETEROGENEOUS REGION

Fatal Error -10200. EACH MATERIAL MUST CORRESPOND TO SOME LIBRARY MATERIAL

Subroutine TESTBG

Fatal Error -10100. THE HIGHEST BROAD GROUP AS READ FROM DATA SET XS.ISO MAY NOT LIE ABOVE THE HIGHEST ENERGY IN THE LIBRARY

Fatal Error -10300. THE HIGHEST BROAD GROUP ENERGY AS READ FROM DATASET XS.ISO DOES NOT FALL ON AN ULTRA-FINE-GROUP BOUNDARY

Fatal Error -10500. THE SPECIFIED BROAD GROUP STRUCTURE AS READ FROM DATASET XS.ISO DOES NOT FALL ON ULTRA-FINE-GROUP BOUNDARIES

Subroutine TYPE06

Fatal Error -10100. CARD PRESUMED TO BE TYPE 06 OF DATA SET A.NIP NOT FOUND

Fatal Error -10200. REGION LABELS MUST BE NON-BLANK

Fatal Error -10300. INVALID REGION BOUNDARY COORDINATES GIVEN ON DATA SET A.NIP TYPE 06 CARDS. EACH REGION LOWER BOUNDARY COORDINATE MUST BE .GE.O. AND .LT. THE CORRESPONDING UPPER BOUNDARY COORDINATE FOR THAT REGION

Fatal Error -10400. THE LOWEST MESH POINT FOR CYLINDERS MUST FALL AT 0.

Fatal Error -10500. REGION BOUNDARIES FOR SEQUENTIAL REGIONS MUST AGREE TO AT LEAST 1.E-4

Subroutine TYPE14

Fatal Error -10100. CARD PRESUMED TO BE TYPE 14 OF DATA SET A.NIP NOT FOUND

Fatal Error -10200. COMPOSITION LABELS MUST BE NON-BLANK

Fatal Error -10300. THE FIRST MATERIAL LABEL ON A TYPE 14 CARD OF DATA SET A.NIP WAS BLANK

Fatal Error -10400. MATERIALS IN HETEROGENEOUS REGIONS MUST CORRESPOND TO LIBRARY MATERIALS

CSI010 (AREA 4) Contd.

Subroutine TYPE14 (contd.)

Fatal Error -10500. MATERIALS MAKING UP A GIVEN COMPOSITION MUST BE DISTINCT

Fatal Error -10600. CARD PRESUMED TO BE TYPE 14 OF DATA SET A.NIP NOT FOUND

Fatal Error -10700. COMPOSITION LABELS MUST BE NON-BLANK

Fatal Error -10800. THE FIRST MATERIAL LABEL ON A TYPE 14 CARD OF DATA SET A.NIP WAS BLANK

Subroutine TYPE15

Fatal Error -10100. CARD PRESUMED TO BE TYPE 15 OF DATA SET A.NIP NOT FOUND

Fatal Error -10200. COMPOSITION LABELS MUST BE NON-BLANK

- Fatal Error -10300. THE FIRST REGION LABEL ON A TYPE 15 CARD OF DATA SET A.NIP MUST BE NON-BLANK
- Fatal Error -10400. A COMPOSITION REFERRED TO ON A DATA SET A.NIP TYPE 15 CARD MUST CORRESPOND TO SOME COMPOSITION DEFINED ON A TYPE 14 CARD
- Fatal Error -10500. A REGION REFERRED TO ON A DATA SET A.NIP TYPE 15 CARD MUST CORRESPOND TO SOME REGION DEFINED ON THE TYPE 06 CARDS
- Fatal Error -10600. ONLY ONE COMPOSITION MAY BE ASSIGNED TO A GIVEN REGION

Fatal Error -10700. CARD PRESUMED TO BE TYPE 15 OF DATA SET A.NIP NOT FOUND

Fatal Error -10800. COMPOSITION LABELS MUST BE NON-BLANK

Fatal Error -10900. A COMPOSITION REFERRED TO ON A DATA SET A.NIP TYPE 15 CARD MUST CORRESPOND TO SOME COMPOSITION DEFINED ON A TYPE 14 CARD

Fatal Error -11000. THE FIRST REGION LABEL ON A TYPE 15 CARD OF DATA SET A.NIP MUST BE NON-BLANK

Fatal Error -11100. A REGION REFERRED TO ON A DATA SET A.NIP TYPE 15 CARD MUST CORRESPOND TO SOME REGION DEFINED ON THE TYPE 06 CARDS

Fatal Error -11200. EVERY REGION ON AN A.NIP TYPE 06 CARD MUST ALSO BE PRESENT ON A TYPE 15 CARD

-175-

CSC004 (AREA 5)

Subroutine CSC004 (MAIN)

Fatal Error -10100. PROBLEM MATERIAL NOT IN THE LIBRARY

Fatal Error -10200. PROBLEM MATERIAL IS NOT IN THE LIBRARY

Subroutine MATRIX

Fatal Error -10100. DETERMINANT IS SINGULAR

CSC005 (AREA 6)

Subroutine ADMSTR

Fatal Error -10100. PROBLEM MATERIAL IS NOT IN THE LIBRARY

Subroutine CSCOO5 (MAIN)

Fatal Error -10100. PROBLEM MATERIAL NOT IN THE LIBRARY

CSC006 (AREA 6.5)

Subroutine CSC006 (MAIN)

Fatal Error -10100. PROBLEM MATERIAL IS NOT IN THE LIBRARY

CSC008 (AREA 7)

Subroutine CSC008 (MAIN)

Fatal Error -999 ERROR IN BPOINTER ALLOCATION

Subroutine SETCHI

Fatal Error -1THE FISSION DISTRIBUTION SPECIFIED FOR THE PROBLEM
IS NOT ON THE LIBRARY

Fatal Error -2 ALL FISSION DISTRIBUTIONS WERE NOT FOUND

Fatal Error -3 ALL FISSION DISTRIBUTIONS WERE NOT FOUND

Subroutine SETSCT

Fatal Error -1

ALL PROBLEM MATERIALS COULD NOT BE FOUND

-176-

-177-

TABLE V. Contd.

CSC008 (AREA 7) Contd.

Subroutine SETSCT (contd.)

- Error 2 NONE OF THE MATERIALS ON THE OLD DATASET OLDSGS ARE USED IN THIS PROBLEM. ALL MICROSCOPIC CROSS SECTIONS MUST BE CALCULATED
- Error 3 THE OLD DATASET OLDSGS HAS A DIFFERENT ENERGY STRUCTURE THAN THE PRESENT PROBLEM (DIFFERENT NUMBER OF GROUPS OR DIFFERENT STARTING ENERGY). MICROSCOPIC CROSS SECTIONS WILL BE RECALCULATED
- Error 4 FIRST ORDER SCATTERING MATRICES ARE REQUIRED BY PROBLEM BUT ARE NOT AVAILABLE ON OLD DATASET OLDSGS. ALL MICROSCOPIC CROSS SECTIONS MUST BE RECALCULATED
- Fatal Error -999 ERROR IN BPOINTER ALLOCATION

Subroutine SIGMAC

Fatal Error -1 ALL PROBLEM MATERIALS COULD NOT BE FOUND ON LIBRARY FILE

CSC009 (AREA 8)

Subroutine BGFLE1

Fatal Error -170 ALL PROBLEM MATERIALS COULD NOT BE FOUND

Subroutine BGFLE2

- Fatal Error -200 ALL PROBLEM MATERIALS COULD NOT BE FOUND ON FILE MCC2F5
- Fatal Error -320 NO CORRESPONDENCE CAN BE FOUND BETWEEN PROBLEM MATERIAL AND INELASTIC MATERIAL
- Fatal Error -500 ALL PROBLEM MATERIALS TO BE EDITED COULD NOT BE FOUND

Subroutine BSQITR

Error 1001 BUCKLING ITERATION NOT FEASIBLE-ZERO BUCKLING IS ASSUMED

Subroutine DRIVER

Error 515

INHOMOGENEOUS SOURCE CALCULATION IS MEANINGFUL ONLY FOR SUBCRITICAL CONFIGURATION

CSC009 (AREA 8) Contd.

Subroutine FILE1

Fatal Error -1000	ALL PROBLEM MATERIALS COULD NOT BE FOUND
Subroutine FILE6	
Fatal Error -9000	ALL PROBLEM MATERIALS COULD NOT BE FOUND
Fatal Error -9010	ALL PROBLEM MATERIALS WITH INELASTIC AND/OR (N,2N) DATA COULD NOT BE FOUND
Fatal Error -9020	ALL PROBLEM MATERIALS WITH INELASTIC AND/OR (N,2N) DATA COULD NOT BE FOUND
Subroutine INIT	
Fatal Error -1000	ALL FISSION SPECTRA MUST BE VECTORS
Error 1001	PROBLEM CHI VECTORS ARE NOT PROPERLY NORMALIZED
Subroutine ISOCHI	
Fatal Error -500	FISSION SPECTRUM FOR PROBLEM MATERIAL M COULD NOT BE FOUND
Subroutine CSC009	(MAIN)
Fatal Error -999	AN ERROR HAS OCCURRED IN ALLOCATING VARIABLY DIMENSIONED ARRAYS
Subroutine REORDR	
Fatal Error -1000.	ALL UNRESOLVED RESONANCES WERE NOT PROCESSED
Subroutine RESCAT	
Fatal Error -300	ALL RESOLVED RESONANCE MATERIALS COULD NOT BE FOUND
Subroutine RESCS	
Error 1001	THERE ARE UNRESOLVED MATERIALS IN PROBLEM BUT UNRESOLVED ULTRA-FINE-GROUP CROSS SECTIONS ARE NOT AVAILABLE FROM FILE UNREG

Subroutine SETBG

Error 200

FIXED SOURCE CANNOT BE WRITTEN TO FILE SRATES

Fatal Error -999

ERROR IN BPOINTER ALLOCATIONS

-179-

TABLE V. Contd.

CSC010 (AREA 9)

Subroutine SETIN

Fatal Error -999 AN ERROR HAS OCCURRED IN ALLOCATING VARIABLY DIMENSIONED ARRAYS

Subroutine BGPN

Error 280 INHOMOGENEOUS PROBLEM IS POSSIBLE ONLY IF EIGENVALUE IS LESS THAN UNITY

Subroutine BSQTR

Error 1001 BUCKLING ITERATION NOT FEASIBLE-ZERO BUCKLING IS ASSUMED

Subroutine HOMOG

Error 1 AT LEAST TWO ISOTOPES ON FILE ISOTXS HAVE THE SAME NAME. ONLY THE CROSS SECTIONS OF THE FIRST ARE PROCESSED

Fatal Error -2 SCATTERING TYPE IS NOT DEFINED

Error 3 NO UPSCATTER IS PERMITTED. UPSCATTER MATRIX ELEMENTS WILL NOT BE PROCESSED

Fatal Error -4 ALL PROBLEM ISOTOPES COULD NOT BE FOUND ON FILE ISOTXS

Subroutine CSCO10 (MAIN)

Fatal Error -1SET FISSION VECTOR MUST BE PRESENT ON FILE ISOTXSIF NO EXTERNAL SOURCE IS PROVIDED

Fatal Error -999 ERROR IN BPOINTER ALLOCATION

CSC011 (RABANL)

Subroutine CSC011 (MAIN)

Fatal Error -10100. PROBLEM MATERIAL NOT IN THE LIBRARY

Fatal Error -10200. FOIL MATERIAL NOT IN THE LIBRARY

Subroutine MANAGR

Fatal Error -10100. INSUFFICIENT STORAGE PROVIDED FOR AREA 10 EXECUTION

Subroutine MATRIX

Fatal Error -10100.

DETERMINANT IS SINGULAR

CSC011 (RABANL) Contd.

Subroutine MERGER

Fatal	Error	-10100.	THE PROBLEM CONTAINS NO RESOLVED RESONANCES
Fatal	Error	-10200.	THE DATA SET SRATES IS NOT CONSISTENT WITH THE AREA 10 UPPER ENERGY OF THE CURRENT PROBLEM
121	F	10200	THE DATA CEN CDATES IS NOT CONSIGNENT LITTL THE

Fatal Error -10300. THE DATA SET SRATES IS NOT CONSISTENT WITH THE NUMBER OF ULTRA FINE GROUPS IN THE CURRENT PROBLEM

Subroutine RATES

Fatal Error -10100. A NEGATIVE COLLISION RATE HAS BEEN ENCOUNTERED

Subroutine SOARCE

Fatal Error -10100. A NEGATIVE SOURCE HAS BEEN COMPUTED

CSE009 (ISOTXS Editor)

Subroutine CSE009 (MAIN)

Error 10000 CARD IS OF WRONG TYPE. SHOULD BE CARD OF TYPE 01

Subroutine XSEDIT

- Fatal Error -10000 CARD IS OF WRONG TYPE. SHOULD BE CARD OF TYPE 02
- Fatal Error -20000 REQUESTED ISOTOPE NOT FOUND IN LIBRARY

CSE007 (Generate XS.ISO from ISOTXS)

Subroutine CTD

- Fatal Error -998 SET YOU HAVE ASKED TO ADD TO IS INCOMPATIBLE IN ENERGY STRUCTURE OR GROUPS
- Fatal Error -999 ERROR IN BPOINTER ALLOCATION

٩

Subroutine CTS

- Fatal Error -998 SET YOU HAVE ASKED TO ADD TO IS INCOMPATIBLE IN ENERGY STRUCTURE OR GROUPS
- Fatal Error -999 ERROR IN BPOINTER ALLOCATION

Subroutine PRINXD

Error 500

FILE ISOTXS CONTAINS UNDEFINED SCATTERING TYPE

CSE007 (Generate XS.ISO from ISOTXS) Contd.

Subroutine PRINXS

Error 500 FILE ISOTXS CONTAINS UNDEFINED SCATTERING TYPE

CSE012 (XS.ISO Editor)

Subroutine CSE012

- Fatal Error -10000. DATA SET XS.ISO NOT PROVIDED
- Fatal Error -10100. TYPE 01 CARD NOT FOUND
- Fatal Error -10200. TYPE 02 CARD NOT FOUND

Subroutine SORTNM

- Fatal Error -10000. CARD TYPE 01 NOT FOUND
- Fatal Error -10100. CARD TYPE 01 NOT FOUND
- Fatal Error -10200. ISOTOPE LABELS MUST BE NON-BLANK
- Fatal Error -10300. EACH ISOTOPE SPECIFIED ON AN ACSE12 CARD MUST CORRESPOND TO SOME ISOTOPE IN THE XS.ISO DATA SET

REED/RITE

- Fatal Error -900 LOGICAL UNIT NUMBER IS OUT OF RANGE
- Fatal Error -910 MODE INDEX OUT OF RANGE
- Fatal Error -920 RECORD NUMBER OUT OF RANGE
- Fatal Error -930 NUMBER OF WORDS IS NEGATIVE
- Fatal Error -940 MISSING DD CARD FOR FILE LUN
- Fatal Error -950 SIO ACCESS ERROR
- Error 960 WRITING A RECORD WITH NWDS=0
- Fatal Error -999 ATTEMPTING TO READ PAST END OF FILE

VII. PROGRAMMING INFORMATION

The MC^2-2 code system was developed within the ARC System modular environment⁽¹⁾ on IBM hardware with an OS operating system. Great care was taken in the programming to limit the degree of system dependence. A number of local conventions were adopted to permit ease of code conversion. These local conventions supplemented the procedures adopted by the CCCC⁽²⁾. Two standalone versions of MC^2-2' were created based on the ARC System modular program; (i) an IBM version for use on any IBM 360 or 370 operating system with at least 600K bytes of storage and (ii) a CDC version for use on CDC 7600 hardware with 50K words of SCM, 60K words of directly addressable LCM, and SECMENTATION LOADER capability as offered with SCOPE 3.4. In this chapter some of the details of the MC^2-2 code are given along with a description of the differences between the IBM and CDC versions of the code. A guide for the implementation of both standalone versions of the code is provided.

A. Program Structure

The standalone versions of the MC^2-2 code were adapted from the ARC System modular programs by creating a primary overlay for each of the MC^2-2 modules (c.f. Chapter VI). Figure 16 shows the overlay control cards for the IBM version and Figure 17 shows the directives required for a segmented load of the CDC version of the code. The segmentation loader available with the SCOPE 3.4 operating system was used by MC^2-2 to avoid the need to insert special OVERLAY and PROGRAM statements and to permit the passing of arguments across overlays.

The programming language used in MC^2-2 is almost entirely FORTRAN. The few machine language subprograms used are discussed in Section D below. There is a one-to-one correspondence between the FORTRAN source code of the IBM and CDC versions. Changes required to account for word length differences, large core memory use, or ENTRY point differences, for example, are flagged by CDC* and/or CIBM comment cards. Testing of the FORTRAN code was performed using the IBM FORTRAN H, OPT=2, compiler whereas the CDC code was compiled using the FTN 4.5, OPT=1 compiler. Extensive use is made of comment cards throughout the FORTRAN source code in an attempt to make the code as self-documenting as possible.

B. ARC System Routines

The user specified BCD (card) input to MC^2-2 follows the ARC System conventions ⁽¹⁾ as discussed in Chapter VI. The FORTRAN routines SCAN and STUFF read and process the input as discussed in Reference 1.

-182-

ENTRY MAIN INSERT MAIN INSERT SNIFF, SEEK, REED, TIMER, ERROR INSERT POINTR, PUTPNT, BULK, FREE, WIPOUT, GETPNT, IGET, IPT2, PUTM INSERT IPTERR, ILAST, REDEF, REDEFM, PURGE, STATUS, PRTI1, PRTI1E INSERT PRTI2, PRTI2E, PRTR1, PRTR1E, PRTR2, PRTR2E INSERT ALLOCS, LOCF, TIME, CLOCK#, DATE, ABEND, TRACER, SECOND, JOBID INSERT PRNT1E, PRNT1A, PRNT1I INSERT MIXER, SIGMAX, QUICK1, CSLAB1, SIGESC INSERT OL, FXP1, GRAPH INSERT MATRIX INSERT LENGTH, OUTPUT, NUMBER, ARRAY, SPECS, EDITS, REAIMW, INTEGL INSERT TIMING, OPTUNR, IOPUT, STFARC, LOCATE, TABLES, PTERR, LCMS IZ, BFLAGS OVERLAY ALPHA INSERT CODE INSERT SCAN INSERT STUFF, STUFF1 OVERLAY ALPHA INSERT CSI010, SPACER, PRNT1D, PRNTAE, DISPOS, GOWEST INSERT UNITS, CR DCNT, LBS PEC OVERLAY BETA INSERT TESTBG, CARDO5, CARDO6, CARDO7, CARDO9 OVERLAY BETA INSERT RDANIP, TYPE06, TYPE14, TYPE15, EDGEOM, GEOM OVERLAY BETA INSERT STRTCH, HETERO, CARD10, CARD11, CARD12, SXLSXR, CSDMGI, CARD14 INSERT CARD15, CARD16, CARD17, CARD18, FOILS, CARD21, CARD22, CARD08, CARD23 OVERLAY BETA INSERT WRITER, EDTPRB INSERT REC1 OVERLAY ALPHA INSERT CSC004, SETUPU, UNRINT, DRCTOR, EDITUN INSERT ESMESH, PACKER INSERT WZERO,QUICKJ,INTERP INSERT RATION, MULTIP, FILE3, BOUNDY, DRCUNR OVERLAY ALPHA INSERT CSC005, ADMSTR, SETUPR, SIFTER, RATNL INSERT REORDR, STUFIT, NSIGO INSERT JINT, OVRLAP, WINNER, JINTGL, TRIPLE, LORENT, EDITLZ INSERT BOUNDE, PHILE1, FILE4 INSERT TRIPIN **OVERLAY ALPHA** INSERT CSC006, EDITSP, EDITAT, EDITRS, POTNTL, WRISIT, INTRAC INSERT MAININ OVERLAY ALPHA INSERT CSC008 **INSERT NSCR7** OVERLAY BETA

INSERT SIGAVG, SIGMAC, SETCHI \

OVERLAY BETA INSERT SETSCT, SETCSD, CALCEN, FNAVG, ELSCAT, XTRN, MODPAR OVERLAY BETA INSERT EDIT1, EDIT2 OVERLAY ALPHA INSERT CSC009, DRIVER, SETBG, TABINT, PROBIN INSERT NSCR8, RESINF, LUN8, INDATA, THDATA, ARGUE OVERLAY BETA INSERT ATNSRC, REORDR, EGRID OVERLAY BETA INSERT SETIN, FILE1, FILE6, MATCH, INIT OVEBLAY BETA INSERT INSCAT, MGSPEC, CSDSPC, FISSOR, BSQITR, PARAB OVERLAY BETA INSERT EDTUFG, ORIGIN OVERLAY BETA INSERT BGSPEC, INSBG, DOIO OVERLAY BETA INSERT BGFLE1, FILFID, RESCS, RESCAT OVERLAY BETA INSERT BGFLE2, PRINXS, CVD, BGSCAT, ISOCHI OVERLAY BETA INSERT SRATE OVERLAY ALPHA INSERT CSC010, HOMOG, BGPN, BSQTR, CRAMER, BGSORS INSERT NSCR9, LUN9 OVERLAY ALPHA INSERT CSE009, XSEDIT INSERT CONT, INPUT, NUMBRE, ISOCNT OVERLAY ALPHA INSERT CSC011, E2E3E4, FREEUP OVERLAY BETA INSERT MERGER, SPOOL, FLIPIT, SYM OVERLAY BETA INSERT SIFTIT OVERLAY GAMMA INSERT STUFER, REARNG OVERLAY GAMMA INSERT RESXSC, RATES, SOARCE, MANAGR, EZ3, YZ3, YZ4, XTRAP, PFUNC INSERT GAUSS, EXTRNL, RTS3S4 OVERLAY GAMMA INSERT PREEDT, EDTICS, PREFOL, EDTFOL, DRIVED INSERT BIGLUN OVERLAY GAMMA INSERT PREEDI, EDTICI, PREFOI, EDTFOI, DRIVEI

-184-

ROOT TREE ALLOC1- (CODE, N4, INTERP, EDITLZ, PRTECS, M7, M8, BGPN, XSEDT , T, M 10) M4 TREE GOWEST-(CA-D09, EDGEOM, CARD23, EDTPRB) M7 TREE CSC008-(SETCHI, MODPAR, EDIT2) M 8 TREE PROBIN- (EGRID, INIT, PARAB, EDTUFG, DOIO, RESCAT, ISOCHI, SR .ATE) M 10 TREE FREEUP-(SYM, M10A21) M10A21 TREE SIFTIT-(REARNG, PFUNC, DRIVED, DRIVEI) ALLOC 1 INCLUDE MCC2, PRNT1E, PRNT1A, ERF, CVD, SNIFF, SEEK, REED, DRED, TIMER ,, WIPOUT, GETPNT, IGET, IPT2, PUTM, IPTERR, ILAST, REDEFM, PURGE, PRTI1, PRTI1E, PR ,TI2,PRTI2E,PRTR1,PRTR1E,PRTR2,PRTR2E,PRNT1I,ERROR,ABEND,POINTR,PUTPNT,B , ULK, FREE, FREE1, MENGET, BUOPEN, ZEROIO, MATRIX, STATUS, QUICK1, FXP1, QL, MIXER, ,GRAPH,SIGMAX,CSLAB1,SIGESC INCLUDE SCAN, STUFF, STUFF1 CODE INCLUDE CSI010, SPACER, PRNT1D, PRNTAE, DISPOS GOWEST CARD09 INCLUDE TESTBG, CARDO5, CARDO6, CARDO7 INCLUDE RDANIP, TYPE06, TYPE14, TYPE15 EDGEOM CARD23 INCLUDE STRTCH, HETERO, CARD10, CARD11, CARD12, SXLSXR, CSDMGI, CARD , 14, CARD15, CARD16, CARD17, CARD18, FOILS, CARD21, CARD22, CARD08 INCLUDE WPITER EDTPRB INTERP INCLUDE CSC004, SETUPU, DRCTOR, EDITUN, UNRINT, ESMESH, PACKER, MZER , O, QUICKJ EDITLZ INCLUDE CSC005, ADMSIR, SETUPR, SIFTER, RATNL, REORDR, STUFIT, NSIGO ,,SIGMAX,JINT, OV RLAP, WINNER, JINTGL, TRIPLE, LORENT PRTECS INCLUDE CSC006, EDITSP, EDITAT, EDITRS, POTNTL, WRISIT, INTRAC SETCHI INCLUDE SIGMAC, SIGAVG MODPAR INCLUDE SETSCT, SETCSD, CALCFN, FNAVG, ELSCAT, XTRN EDIT2 INCLUDE EDIT1 INCLUDE CSC009, DRIVER, SETBG, TABINT PROBIN EGRID INCLUDE ATNSRC, REORDR INIT INCLUDE SETIN, FILE1, FILE6, MATCH PARAB INCLUDE INSCAT, MGSPEC, CSDSPC, FISSOR, BSOITR DOIO INCLUDE BGSPEC, INSBG RESCAT INCLUDE BGFLE1, FILEID, RESCS ISOCHI INCLUDE PRINXS, BGFLE2, BGSCAT INCLUDE CSC010, HOMOG, BSQTR, CRAMER, BGSORS BGPN XSEDIT INCLUDE CSE009 INCLUDE CSC011, E2E3E4 FREEUP INCLUDE MERGER, SPOOL, FLIPIT SYM REARNG INCLUDE STUFER PFUNC INCLUDE RESXSC, RATES, SOARCE, MANAGR, EZ3, YZ3, YZ4, XTRAP DRIVED INCLUDE PREEDT, EDTICS, PREFOL, EDTFOL DRIVEI INCLUDE PREEDI, EDTICI, PREFOI, EDTFOI GLOBAL LENGTH, OUTPUT, NUMBER, ARRAY, SPECS, EDITS, INTEGL, TIMING, , OPTUNR, UNITS, LBS PEC, CRDCNT, ARRAY2, CRALOC, STFARC, BFLAGS, LCMSIZ, PTERR, TAB , LES, LOCATE, INITIO, NSCR7, ARGUE, THDATA, INDATA, LUN8, RESINF, NSCR9, IOPUT, REA

, IMW, NSCR10, SAVER, FINDER, LGUNIT, WPNTRS, GEOMBC

END MCC2

The code MC²-2 uses the dynamic storage capability, BPOINTER, described in Reference 1, to manage all variable dimension array allocations. A one-to-one correspondence between the IBM and CDC versions of this subprogram package has been retained except for the machine dependent routines ALLOC and LOCF which are discussed in Section D below. The bulk storage capability of the IBM version of BPOINTER has been modified to manage arrays in large core memory of the CDC 7600. Appendix E describes some of the characteristics of the BPOINTER subprogram package.

The ARC System routine SNIFF is used throughout the code to assign logical unit numbers to named data files. Both the IBM and CDC standalone versions of the code use a modification of the routine SNIFF which calls the CCCC subroutine SEEK described in Section C below.

C. CCCC Standard Subroutines

The Committee on Computer Code Coordination has specified a number of standard routines $^{(2)}$ to be used in RRD funded code development. In principle the routines may be installation dependent. The MC²-2 code uses the standard routines REED, RITE, TIMER and SEEK as specified in Reference 2. While it is possible to substitute installation dependent code, simple versions of these routines are provided with both the IBM and CDC standalone versions of MC²-2.

1. SEEK

The subroutine SEEK is used by MC^2-2 to return the logical unit number associated with a named data file. With only two exceptions, all files referenced by MC^2-2 are assigned unit numbers through calls to subroutine SEEK. The two exceptions are the BCD files input (5) and printed output (6). While it should be possible to use installation dependent versions of SEEK, it is obvious that any changes in unit allocation must also be reflected in the Job Control Language required for the IBM version of MC^2-2 (Fig. 11) or the program card of the CDC version. A single initialization call to SEEK is made from the main program driver of MC^2-2 . In Table VI information is provided about the various files referenced by MC^2-2 . Detailed formats for the interface files MCC2F1 - MCC2F8, ISOTXS and XSISO, card input files A.MCC2, A.NIP, A.STPO15, ACSO09 and ACSE12, and some of the other important files are given in the Appendices B, C and D. It should be noted from Table VI that each named file in the IBM version of the program is assigned a unique logical unit number. Restrictions on the number of unit definitions permitted by CDC 7600 software made it necessary to use the same unit number for more than one file. This equivalencing of files was specified to retain full program generality. The file assignment is specified in the initialization call to SEEK from the main program driver. In spite of the large number of files required, even the largest problem requires that a maximum of seven files be open at any particular time in the calculation.

In Table VI there is a column indicating the modules (overlays) which reference a particular file for reading and/or writing. Many of these references are conditional and depend upon the particular problem specifications. Extensive use is made of the various SEEK options to determine whether a particular file is available for reading and/or writing.

Name	Logical Unit Number IBM (CDC)	Contents	Modules Referencing File*
Input	5 (5)	BCD Input-Not Referenced by SEEK	CSI010 (R)
Output	6 (6)	Printed Output-Not Referenced by SEEK	All Modules (W)
ARC	9 (56)	Processed BCD Input-Not Referenced by SEEK	SCAN (W) STUFF (R)
A.MCC2	11 (11)	MC ² -2 General BCD Input	STUFF (W) CSIO10 (R)
A.NIP	12 (12)	Geometry and composition BCD Input	STUFF (W) CSIO1O (R)
A.STP015	13 (13)	Path BCD Input	STUFF (W) DRIVER (R) CSIO10 (R)
ATNUAT	14 (14) _	Unresolved Attenuation Factors	CSC004 (W) CSC006 (R,W) CSC009 (R)
BC	15 (15)	Boundary Conditions	CSI010 (R,W) CSC004 (R) CSC005 (R) CSC006 (R) CSC011 (R)
BGRES	16 (16)	Broad Group Resonance Cross Sections	CSC009 (R,W)

-187-

Name	Logical Unit Number IBM (CDC)	Contents	Modules Referencing File*
CEOM1	- 17 (17)	Geometry Data	CS1010: (R.W)
GEOMI		Geometry Data	CSC004 (R)
			CSC005 (R)
			CSC006 (R)
e		-	CSC011 (R)
GRPORD	18 (18)	Group Ordered Inelastic Data	CSC009 (R,W)
ISOTXS	19 (19)	Interface Broad Group Cross	CSC009 (W)
		Sections	CSC010 (R)
			CSE009 (R)
•			CSE007 (R)
LORENZ	20 (20)	Lorenzian Distribution	CSC005 (R,W)
MACTOT	21 (21)	Ultra-Fine-Group Macroscopic	CSC004 (R,W)
		Total Cross Section	CSC005 (R,W)
			CSC006 (R,W)
			CSCO11 (R,W)
MCC2F1	22 (22)	Administrative Data	CS1010 (R)
			CSC004 (R)
		· •	CSC005 (R)
			CSC006 (R)
			CSC008 (R)
			CSC009 (R)
		· · ·	CSC011 (R)
MCC2F2	23 (23)	Tabulated Data	CSC004 (R)
	· · · ·		CSC005 (R)
			CSC006 (R)
			CSC009 (R)
			CSC011 (R)
MCC2F3	24 (24)	Unresolved Resonance Data	CSI010 (R)
		с	CSC004 (R)
	·		CSC006 (R)

-188-

)

Name	Logical Unit Number IBM (CDC)	Contents	Modules Referencing File*
MCC2F4	25 (25)	Resolved Resonance Data	CSI010 (R)
			CSC005 (R)
		<u>^</u>	CSC006 (R)
			CSC011 (R)
MCC2F5	26 (26)	Ultra-Fine-Group Non-	CSC004 (R)
		Resonance Cross Sections	CSC005 (R)
			CSC006 (R)
			CSC008 (R)
			CSC009 (R)
	•		CSC011 (R)
MCC2F6	27 (27)	Inelastic and (n,2n) Data	CSC009 (R)
MCC2F7	28 (28)	Fission Spectrum Data	CSI010 (R)
			CSC008 (R)
MCC2F8	29 (29)	Elastic Scattering Distribution Data	CSC008 (R)
MICTOT	30 (30)	Ultra-Fine-Group Microscopic	CSC004 (R,W)
		Total Cross Section	CSC005 (R,W)
			CSC006 (R,W)
OLDSGS	31 (31)	Ultra-Fine-Group Elastic Matrices for Restart	CSC008 (R)
OPTICL	32 (32)	Optical Distance Data	CSC004 (R.W)
		•	CSC005 (R.W)
			CSC006 (R)
ወ፤ ለጥ፤ ጥቃቃ	22 (22)	Dumme File for District Outer	
A POILIAG	33 (33)	Dummy File for Flotting Output	
			C2COOA (M)

-189-

Name	Logical Unit Number IBM (CDC)	Contents	Modules Referencing File*
PRBCHI	34 (34)	Problem Dependent Fission Spectrum Data	CSC008 (W) CSC009 (R,W)
PRBSPC	35 (35)	Problem Specifications	CSI010 (R,W) CSC004 (R) CSC005 (R)
			CSC008 (R) CSC009 (R) CSC010 (R) CSC011 (R) CSE009 (R)
RESINT	36 (38)	Resolved Resonance Integrals	CSE007 (R) CSC005 (W) CSC006 (R,W)
SCR001	37 (39)	Scratch	CSC009 (R) CSC005 (R,W) CSC008 (R,W) CSC009 (R,W)
SCR002	38 (40)	Scratch	CSCO11 (R,W) CSCO05 (R,W) CSCO08 (R,W) CSCO09 (R,W) CSCO11 (R W)
SCR003	39 (41)	Scratch	CSC004 (R,W) CSC005 (R,W) CSC006 (R,W) CSC009 (R,W)
SCR004	40 (42)	Scratch	CSCO11 (R,W) CSCO09 (R,W) CSCO11 (R,W)

-190-

4

Name	Logical Unit Number IBM (CDC)	Contents	Modules Referencing File*	
SCR005	41 (43)	Scratch	CSCO09 (R,W) CSCO11 (R,W)	
SIGMAP	42 (44)	Ultra-Fine-Group Background Cross Section	CSC006 (R,W)	
SMSIGS	43 (45)	Ultra-Fine-Group Elastic Transfer Matrices	CSC008 (R,W) CSC009 (R)	
SPECTR	44 (46)	Ultra-Fine-Group Spectrum	CSCOO9 (W) CSCO10 (R)	
SPECXS	45 (47)	Ultra-Fine-Group Cross Sections and Moderating Parameters	CSC008 (R,W) CSC009 (R)	
SRATES	46 (48)	Ultra-Fine-Group Microscopic Scattering Collision Density and Sources	CSC009 (W) CSC011 (R)	
UNREG	47 (49)	Unresolved Resonance Ultra- Fine-Group Cross Sections	CSC008 (W) CSC009 (R) CSC011 (R)	
UNRES	48 (50)	Unresolved Cross Sections at E* Points	CSC004 (R,W) CSC008 (R)	
XSISO	49 (51)	File 1 of ARC System Broad Group File	CSI010 (R) CSE007 (R,W) CSE012 (R)	
XSISO5	50 (52)	File 2 of ARC System Broad Group File	CSE007 (R,W) CSE012 (R)	
IRESCS	51 (53)	Integral Transport Derived Broad Group Cross Sections	CSCO11 (R,W)	

Name	Logical Unit Number IBM (CDC)	Contents	Modules Referencing File*	
BIGXS1	52 (54)	Ultra-Fine-Group Macroscopic Scattering Data, Foil Data in Integral Transport Module	CSC008 (R,W) CSC009 (R) CSC011 (R,W)	
	- ·	For IBM 25 Files are defined BIGXS1 - BIGXSP with unit Numbers 52-76		
•	4 •	For CDC 14 Files are defined BIGXS1 - BIGXSE with unit Numbers 54, 55, 11, 12, 13, 20, 21, 30, 32, 33, 44, 51, 52, 53		
RANDOM**	77 (-)	Random Access Data File Not Used In Standalone Code Releases	CSC009 (R,W)	
RESDAT	78 (37)	Selected Resonance Parameters	CSC005 (W) CSC009 (R)	
ACS009	79 (36)	BCD Input for ISOTXS Editor	STUFF (W) CSE009 (R)	
ACSE12**	80 (-)	BCD Input for XS.ISO Editor	STUFF (W) CSE012 (R)	

....

ς.

*R Read W Write

** Not Referenced in Standalone Version of MC^2-2

-192-

2. REED/RITE

The standard routines REED and RITE are used to perform the non-formatted (binary) I/O operations for MC^2-2 . The ARC System modular version of REED/RITE makes use of three access methods, standard FORTRAN. asynchronous FORTRAN and a special random access I/O package $SIO^{(40)}$. The IBM and CDC standalone versions of MC^2-2 contain a far simpler version of REED/RITE which uses standard FORTRAN I/O. The CCCC specifications of REED and RITE did not account for the possibility of multilevel data transfers as required by the CDC version of MC^2-2 to permit the transfer of data between large core memory and disk files without use of a buffer array in small core memory. In order to permit such transfers the routines DRED and DRIT were used in the CDC version of the code. The argument lists to DRED/DRIT are precisely the same as the calls to REED/RITE but the array address into (from) which data are transferred is assumed to be in large core memory. This convention differs from a recent proposal to the CCCC where a pointer rather than an address was recommended. Changes to conform with the recommended versions of DRED/DRIT would be quite simple because of the limited number of calls to these routines.

3. TIMER

The standard subroutine TIMER is used by MC^2-2 to provide timing and problem identification information. These data are not essential to the execution of the MC^2-2 program. As a consequence, the CDC version of TIMER is in large part a dummy program. It calls only the standard CDC system code SECOND to provide CPU timing data. The IBM version of TIMER makes use of a number of assembler language routines to provide time and edit information. These assembler routines are described in Section D below.

D. Machine Language Routines

Machine language routines are used in MC^2-2 to provide capability which does not exist in the FORTRAN language. A brief description of these sub-programs is provided in this section.

1. ALLOC

The subroutine POINTR of the dynamic allocation subprogram package BPOINTER assigns storage dynamically at run time based on user input specifications. The routine ALLOC is called to assign and free this storage. The IBM version of the code ALLOC makes use of the OS macros GETMAIN and FREEMAIN to perform this function. The CDC version of ALLOC is a FORTRAN function which calls the COMPASS routine MEMGET to perform the allocation.

2. LOCF

The function subprogram LOCF is used by subroutine POINTR to obtain the address of the dynamic storage container and the common block /ARRAY/. This function is a standard routine of CDC FORTRAN. The IBM version of the function is coded in assembler language. Since the CDC version of LOCF does not obtain the address of LCM variables, the CDC version of POINTR assumes that the LCM common block /ARRAY2/ is located at the start of LCM (address 1). The subroutine GOWEST is used to left-justify Hollerith variables. The IBM version of the routine is written in assembler language. The CDC version of the routine is written in FORTRAN and uses the standard CDC FORTRAN version of the function SHIFT.

4. FILEID

The subroutine FILEID is used in subroutine BGFLE1 of module CSC009 to obtain the dataset name associated with the file ISOTXS. It is an assembler language routine in the IBM version of the code and a dummy FORTRAN routine in the CDC code.

5. CVD

The subroutine CVD is used in subroutine BGFLE2 of module CSC009 to convert a variable from integer to Hollerith format. The IBM version of CVD is written in assembler language and the CDC version of the routine uses the FORTRAN ENCODE capability.

6. TIME

TIME is used by the IBM version of subroutine TIMER to return the wall clock time in the form HH.MM.SS. TIME is not referenced in the CDC version of the code.

7. CLOCK

CLOCK is used by the IBM version of subroutine TIMER to return the wall clock time in units of .01 seconds. CLOCK is not referenced in the CDC version of the code.

8. DATE

The function subprogram DATE is called by the IBM version of subroutine TIMER to return the current date in the form MM/DD/YY. DATE is not referenced by the CDC version of the code.

9. SECOND

The subroutine SECOND is called by TIMER to return the elapsed CPU time in seconds. An IBM assembler language version of SECOND which uses the STIMER and TTIMER macro-instructions is provided. The CDC version of the code requires the standard system version of the subroutine SECOND.

10. ABEND

The subroutine ABEND is called by subroutine ERROR to force an abnormal termination of the job with a DUMP. The IBM version of ABEND is an assembler language routine which invokes the macro-instruction ABEND. An abnormal termination code USER 16 is given and a dump is written to the file SYSUDUMP or SYSABEND. The CDC version of ABEND is a FORTRAN routine which calls the FORTRAN routine EXIT.

11. JOBID

JOBID is used by the IBM version of subroutine TIMER to return the job name supplied on the user's job card. JOBID is not referenced by the CDC version of the code.

12. FXP

FXP is a fast exponential function. The algorithm is described in Appendix A. The IBM version of FXP is programmed in assembler language. An equivalent FORTRAN routine is provided with the CDC version of the code.

13. DISPOS

The subroutine DISPOS is used by the module CSI010 to determine whether the file SRATES is given a permanent or temporary disposition. The IBM version of DISPOS is programmed in assembler language. The CDC version of DISPOS is a dummy FORTRAN routine which returns a flag to the calling routine indicating a temporary disposition for the file.

14. TRACER

TRACER is used by the IBM version of REED/RITE to provide traceback information in case of an error in REED/RITE. The IBM version of TRACER is programmed in assembler language. TRACER is not referenced in the CDC version of the code.

E. Program Implementation

Both the IBM and CDC versions of the program MC^2-2 are available from the Argonne Code Center. Each version of the code requires a program tape and a library tape. A description of the tapes along with a step-by-step implementation strategy for each version of the program is described below.

1. IBM Code Implementation

The IBM program tape is written as described in Table VII. The source code for MC^2-2 is contained on Files 1 and 2 of the program tape. A brief description of each of the subprograms of File 1 is given in Table VIII. The assembler language subprograms on File 2 of the tape were discussed in Section D above. Creation of object code for input to the linkage editor is the first step required for implementation of the MC^2-2 program. It is recommended that this step be performed in the following manner: (i) preallocate an object module dataset; (ii) compile File 1 Fortran code; (iii) assemble File 2 assembler code. The preallocation may be performed by submitting the job shown in Fig. 18. The user may clearly specify any DSN, VOL or UNIT information consistent with local naming conventions. In all of the examples which follow, ANL conventions will be used but the essential nature of each step is intended to be system independent. The compilation of the File 1 subprograms should be performed using the highest level of optimization available. Routinely this means use of the FORTRAN H compiler with OPT=2 specified. It should be noted that FORTRAN compilers are often system dependent in the sense of release identification and system generation (SYSGEN) options. The Level 21.7

compiler is currently in use at ANL but other releases of IBM compilers have been used without problem. The SIZE parameter on the standard ANL Fortran H compiler was set at 215K at SYSGEN time. Several of the MC²-2 subroutines are too large to compile with this specification. A second Fortran H compiler is available at ANL which was specified with SIZE=450K at SYSGEN time. This compiler is used at ANL to compile the subroutines CSI010, DRCTOR, ELSCAT, CSC009, SETBG, CSC011, RATES and SIFTIT since these routines will not compile (ADCON TABLE EXCEEDED) with the smaller compiler. The compilation of File 1 code may be performed in one job step as shown in Fig. 19 or by breaking up the code into a number of smaller blocks. The latter procedure is recommended as it permits one to work with smaller blocks of source code. In order to break up the file into such manageable blocks, one can use the IBM utility IEBGENER to create a partitioned data set (PDS) and then compile each member of the PDS separately into the PDS created by the preallocation step of Fig. 18. If this mode of operation is used, a recommended breakup of the source language is by module as indicated in Table VIII. Whether one uses a PDS or the tape as input, it is recommended that the code be compiled and link edited (with NCAL specified) as shown in Fig. 19. The final step in the processing of the source language tape is the assembly of the assembler language routines on File 2 of the tape. This is done in a straightforward manner as illustrated The File 2 routines should not be broken into blocks although in Fig. 20. all of File 2 (944 source cards) may be copied as a single member of the source language PDS if desired.

If the user does not wish to work directly with source code as recommended above, an object library PDS is available on File 3 of the program tape. One may bypass the steps above and use the IBM utility IEHMOVE to load the object library to a disk pack as shown in Fig. 21. The blocksize of the object module was set to 6K so that it may be moved to either a 2314 or 3330 disk pack. This is not the recommended procedure unless there are particular problems involved in operating with the two files of source code.

A sample problem input deck is provided on File 4 of the program tape. Prior to executing this sample problem three further steps are required: (i) preparation of MC^2-2 binary libraries; (ii) modification of sample problem linkage editor instructions to conform with the object library preparation completed above; (iii) modification of the JCL procedure provided with sample problem.

A library tape is provided along with the MC^2-2 program tape. This second tape contains eight binary sequential files in the format of files MCC2F1 - MCC2F8 described in Appendix C. These files must be copied to a direct access device (e.g. disk pack) as the next step in program implementation. A sample of the job control required to copy these files is given in Fig. 22. The space requirements for these files (on a 3330 disk pack) are given in Table VII. The data available on these library files were processed from the ENDF/B-IV data files by the code ETOE-II. A summary of the library specifications is given in Table IX. The Argonne Code Center also has available four BCD tapes which contain this same eight file library in BCD format along with a Fortran program, MC^2-2 LIBGEN, which reads the BCD files and writes the eight binary files. This library generation program is described in Appendix F.

Having copied the eight library files to a direct access device, the only remaining steps in implementation involve modification of the sample problem linkage edit and JCL procedure to conform with local conventions. The linkage edit step of the sample problem deck builds an executable load module from the object library segments created by compilation and assembly of Files 1 and 2 of the library tape respectively (or alternatively copy of File 3). The user must modify the sample problem deck to reference this object library on the JCL card with the DDNAME MYLIB. Following the linkage edit SYSIN card the user must INCLUDE all members of the object library. The INCLUDE cards provided with the sample problem deck reference the members available on the PDS of File 3 of the library tape. It is obvious that the user must also modify the procedure name FTHEP and step names to conform with local conventions. If desired the load module created in this step may be given a permanent disposition so that the edit step may be bypassed on future problem executions thus saving a considerable amount (\sim 2 minutes) of I/O time.

The JCL procedure (c.f. Fig. 11) ARCSP015, which follows the linkage edit step of the sample problem input must be modified to conform with local conventions. Symbolic parameters have been provided to ease the burden of this modification. For example the parameter UNITSCR should be changed in the PROC statement to reference the standard system direct access scratch unit, FULLBLK should be set to 6136 if 3330 disk packs are not available, etc. In addition the parameters MCC2F1 through MCC2F8 should be modified to conform with the data set names assigned to the eight MC^2-2 library files copied from the library tape. The parameter PRELIB should be set to the data set name (permanent or temporary) assigned to the MC^2-2 load module.

Table VII

IBM Code Center Tape Description

1) Program Tape (Non-Labelled, 9 trk, 800 bpi)

<u>File</u>	Description	RFCFM	LRECL	BLKSIZE
1	Fortran Source (EBCDIC)	FB	80	3200
2	Assembler Source(EBCDIC)	FB	80	3200
3	Load Module (Binary)	FB	80	800
4	Sample Problem (EBCDIC) Input	FB	80	3200
,				

2) Library Tape (Non-Labelled, 9 trk, 1600 bp1)

	- J = I = V	-	•		
File	<u>Space</u> (units of Tracks)	3330	RECFM	LRECL	BLKSIZE
1	1		VBS	X	6447
2	9		VBS	х	6447
3	. 6	•	VBS	x	6447
4	11		VBS	Х	6447
5	223		VBS	Х	6447
6	289		VBS	x	6447
7	1	·	VBS	Х	6447
8	334		VBS	Х .	6447

-198-
Figure 18

Preallocation of Object Library

```
// EXEC PGM=IEFBR14
//OBJECT DD DSN=MCC2.OBJLIB,DISP=(NEW,CATLG),UNIT=SHRT3330,
// SPACE=(CYL,(5,3,1)),VOL=SER=,DCB=BLKSIZE=6144
/*
```

Figure 19

Compilation of Fortran Subprograms

```
// EXEC FTHCEP,OPTIONS='OPT=2',COMPILE=FORTH450,
// EDTOPTS='LET,LIST,MAP,DCBS,NCAL'
//FTH.SYSIN DD UNIT=TAPE9TRK,DISP=OLD,LABEL=(1,NL),
// VOL=(,RETAIN,SER=nnnnn),
// DCB=(RECFM=FB,LRECL=80,BLKSIZE=3200,DEN=2)
//EDT.SYSLMOD DD DISP=OLD,DCB=BLKSIZE=6144,
// DSN=MCC2.OBJLIB(anyname_1)
```

/*

Figure 20

Assembly of File 2 CSECTS

```
// EXEC ASMCEP,EDTOPTS='LIST,MAP,DCBS'
//ASM.SYSIN DD UNIT=TAPE9TRK,DISP=OLD,LABEL=(2,NL),
// VOL=(,RETAIN,SER=nnnnnn),
// DCB=(RECFM=FB,LRECL=80,BLKSIZE=3200,DEN=2)
//EDT.SYSLMOD DD DISP=OLD,DCB=BLKSIZE=6144,
// DSN=MCC2.OBJLIB(anyname<sub>2</sub>)
```

/*

Figure 21

Copy Object Module from File 3 of Program Tape to Disk

// EXEC PGM=IEHMOVE //SYSPRINT DD SYSOUT=A //SYSUT1 DD DISP=SHR,UNIT=3330,VOL=SER=SCR001 //OBJECT DD DISP=OLD,UNIT=3330,VOL=SER=PACKxx //TAPE DD DISP=(OLD,PASS),UNIT=TAPE9TRK,VOL=(,RETAIN,SER=nnnnn), // DCB=(RECFM=FB,LRECL=80,BLKSIZE=800,DEN=2), // LABEL=(3,NL),DSN=FILE3 //SYSIN DD * COPY FROM=2400=(nnnnn,3),TO=3330=PACKxx,FROMDD=TAPE, DSNAME=MCC2.OBJLIB

/*

```
//LIBCOPY PROC TAPE=119361, TUNIT=TAPE1600, FILE=1, NAME=,
                 PRIMARY=1, SPCE=TRK, UN=3330, DSP=KEEP
11
// EXEC PGM=IEBGENER
//SYSPRINT DD SYSOUT=A
//SYSUT1 DD DISP=OLD, DC B= (RECFM=VBS, LRECL=X, BLKSIZE=6447),
11
                 UNIT=STUNIT, LABEL= (SFILE, NL), VOL= (, RETAIN, SER=STAPE)
//SYSUT2 DD DSN=&NAME, UNIT=&UN, DISP=(NEW, &DSP), DCB=*.SYSUT1,
                 SPACE=(SSPCE, (SPRIMARY, 1))
11
//SYSIN DD DUMMY
// PEND
// EXEC LIBCOPY, NAME=MCC2F1
// EXEC LIBCOPY, NAME=MCC2F2, FILE=2, PRIMARY=9
// EXEC LISCOPY, NAME=MCC2F3, FILE=3, PRIMARY=6
// EXEC LIBCOPY, NAME=MCC2F4, FILE=4, PRIMARY=11
// EXEC LIBCOPY, NAME=MCC2F5, FILE=5, PRIMARY=12, SPCE=CYL
// EXEC LIBCOPY, NAME= MCC2F6, FILE=6, PRIMARY=16, SPCE=CYL
// EXEC LIBCOPY, NAME=MCC2F7, FILE=7
// EXEC LIBCOPY, NAME=MCC2F8, FILE=8, PRIMARY=18, SPCE=CYL
```

Driver and System Subprograms

MAIN (000010 - 002110)*

Main program driver for MC^2-2 . Initializes SEEK, calls system routines to read and spool BCD input and calls all modules.

SEEK (002120 - 004290)

CCCC routine returns logical unit numbers associated with named data file.

REED/RITE (004300 - 006500)

CCCC routine performs all binary I/O data transfer operations.

TIMER (006510 - 007830)

CCCC routine performs all timing and system dependent editing (date, ID, etc.) operations.

ERROR (007840 - 008450)

Prints error numbers and calls ABEND for job termination in case of FATAL error.

SNIFF (008460 - 008700)

Dummy ARC System interface routine calls SEEK to obtain unit information.

BPOINTER $(008710^{-} - 021230)$

Dynamic allocation subprogram package - c.f. Appendix

CODE (021240 - 022650)

Reads a BCD card image and inspects it; called by SCAN and STUFF modules.

SCAN (022660 - 024880)

Reads entire BCD input stream from logical unit number 5 and spools the data to file ARC. Calls SEEK to initialize files in BLOCK=OLD.

*Numbers in brackets are sequence numbers of routine on MC^2-2 program tape.

STUFF (024890 - 026270)

Locates next block of data from file ARC for processing.

STUFF1 (026280 - 029800)

Reads and process next block of BCD data from file ARC.

PRNT1A (029810 - 031020)

Edits one-dimensional Hollerith arrays.

PRNT1E (031030 - 032250)

1

Edits one-dimensional floating point single precision arrays.

PRNT11 (032260 - 033470)

Edits one-dimensional integer arrays.

MIXER (033480 - 034950)

Homogenizes smooth u.f.g. total cross sections for homogeneous mixture and each heterogeneous region.

SIGMAX (034960 - 036560)

Calculates optical thickness to left and right of each slab region.

QUICK1/QUICKW (036570 - 038070)

Calculates the real and imaginary parts of the complex W function.

MATRIX (038080 - 039830)

Matrix inversion routine.

CSLAB1/CSLAB (039840 - 040460)

Calculates Dancoff factor for slab region.

SIGESC (040470 - 040800)

Calculates escape cross section for cylindrical two-region pin cells.

QL (040810 - 043580)

Calculates ratio of Legendre functions of the second kind, Q_{N+1}/Q_N - c.f. Appendix A.

GRAPH (043590 - 043650)

Main program driver for MC^2-2 input processor.

CARDO6 (060390 - 062630)

Reads type 06 cards of file A.MCC2 and orders materials consistent with library.

CARD15 (062640 - 066460)

Reads and processes type 15 cards of file A.MCC2.

CARD22 (066470 - 068170)

Reads type 22 cards of file A.MCC2 and loads arrays.

CARD09 (068180 - 071100)

Reads and processes type 09 cards of file A.MCC2.

CARD14 (071110 - 072970)

Reads and processes type 14 cards of file A.MCC2.

CARD18 (072980 - 074070)

Reads and processes type 18 cards of file A.MCC2.

CARD10 (074080 - 075330)

Reads and processes type 10 cards of file A.MCC2. TYPE06 (075340 - 078580)

Reads and processes type 06 cards of file A.NIP. TYPE14 (078590 - 081610)

Reads and processes type 14 cards of file A.NIP.

TYPE15 (081620 - 086030)

Reads and processes type 15 cards of file A.NIP.

CARD16 (086040 - 087860)

Reads and processes type 16 cards of file A.MCC2.

CARD21 (087870 - 089690)

Reads and processes type 21 cards of file A.MCC2.

SPACER (089700 - 090260)

Reads past cards in a BCD file.

CARD23 (090270 - 091960)

Reads and processes type 23 cards of file A.MCC2.

CARD11 (091970 - 093130)

Reads and processes type 11 cards of file A.MCC2.

CARDO8 (093140 - 097070)

Reads and processes type 08 cards of file A.MCC2.

HETERO (097080 - 098250)

Generates homogeneous atom densities from A.NIP data.

CARD17 (098260 - 102370)

Reads and processes type 17 cards of file A.MCC2.

CSDMGI (102380 - 103130)

Determines interface energy between multigroup and continuous slowing down solution algorithms.

PRNTAE (103140 - 103860)

Edits Hollerith and single precision floating point arrays.

TESTBG (103870 - 105610)

Tests broad group energy structure for compatibility with ultra-fine-group energies.

CARD12 (105620 - 112160)

Reads and processes type 12 cards of file A.MCC2.

CARD07/SETGRP (112170 - 117750)

Reads and processes type 07 cards of file A.MCC2; processes prestored broad-group structures.

CARDO5 (117760 - 122280)

Reads and processes type 05 cards of file A.MCC2.

RDANIP (122290 - 127690)

Controls the reading of type 06, 14, 15 cards of file A.NIP.

```
EDGEOM (127700 - 129190)
```

Edits binary file GEOM1

FOILS (129200-129640)

Loads foil atom density array FOILDN

STRTCH (129650 - 131600)

Assures that all materials in heterogeneous problem appear in homogeneous mixture.

PRNT1D (131610 - 132830)

Edits one-dimensional double precision floating point array.

WRITER (132840 - 135310)

Writes the binary file PRBSPC.

SXLSXR (135320 - 137340)

Calculates arrays SXL, SXR used to obtain slab optical thickness.

EDTPRB (137350 - 145270)

Edits file PRBSPC.

Main driver for MC^2-2 unresolved resonance calculation.

-205-

UNRINT/UNRUNT (152920 - 165770)

Calculates the single level unresolved resonance integral including interference scattering and self overlap.

PACKER (165780 - 167120)

Reads unresolved resonance data from library file MCC2F3 and processes data.

INTERP (167130 - 168290)

Linearly interpolates input array onto a second array on a different energy grid.

SETUPU (168300 - 168760)

Identifies mixture materials with unresolved data.

QUICKJ (168770 - 171960)

Calculates the isolated J integral with or without interference scattering using a single level representation.

DRCTOR (171970 - 180300)

Allocates, loads arrays and directs calculation of unresolved resonance cross sections.

EDITUN (180310 - 182320)

Edits file UNRES.

ESMESH (182330 - 183710)

Sets up correspondence between input and collapsed energy mesh.

WZERO (183720 - 184170)

Calculates real part of the function W(o,x) using a rational approximation.

Main driver for MC^2-2 resolved resonance calculation.

ADMSTR (187710 - 191750)

Reads MC^2-2 library data and controls calculation of resolved resonance integrals.

STUFIT (191760 - 193860)

Reads resolved data from file MCC2F4 and loads arrays in compressed form.

JINT (193870 - 204290)

Calculates isolated resonance integrals including interference scattering using either single level or multi-level formálisms.

EDITLZ (204300 - 205880)

Edits file LORENZ.

NSIGO (205890 - 206950)

Calculates No for homogeneous and heterogeneous mixtures.

LORENT/LORANT (206960 - 209070)

Calculates ultra-fine-group integrals of Lorentzian shape resonance integrals.

WINNER (209080 - 210390)

Determines which neighboring resolved resonances contribute to overlap integral.

OVRLAP/OVRLOP (210400 - 216960)

Calculates overlap integrals for each resolved resonance.

RATNL (216970 - 217510)

Calculates real part of W(o,x) using a rational approximation.

SETUPR (217520 - 217990)

Identifies mixures with resolved resonance data.

TRIPLE (218000 - 219450)

Calculates infinite integrals of products of ψ and χ .

JINTGL (219460 - 222330)

- Calculates isolated J integral including interference scattering for the single level representation.

SIFTER (222340 - 228020)

Selects materials from resolved data file MCC2F4, controls reordering of resonances and selects overlap candidates.

REORDR (228030 - 231250)

Arranges resolved resonances in order of decreasing energy and deletes those that fall outside range of problem.

CSC006 (231260 - 239090)

Main driver for resolved-unresolved resonance interaction module.

INTRAC (239100 - 242260)

Reads files ATNUAT and RESINT and calculates interaction factors.

POTNTL (242270 - 245890)

Calculates resonance background cross sections and writes file SIGMAP.

EDITAT (245900 - 248660)

Edits file ATNUAT.

EDITRS (248670 - 250540)

Edits file RESINT.

WRISIT (250550 - 251560)

Calculates resolved overlap factors to be associated with each grid point.

EDITSP (251570 - 252880)

Edits file SIGMAP.

Main driver for macroscopic data calculations.

SIGMAC (259510 - 265370)

Calculates macroscopic total, elastic and $\nu\Sigma_{\mbox{f}}$ ultra-fine-group cross sections.

SETCHI (265380 - 269330)

Calculates fission spectra distributions and writes file PRBCHI.

SETSCT (269340 - 278160)

Controls calculation of elastic matrices, transport cross sections and moderating parameters.

ELSCAT/ELSCT1 (278170 - 290780)

Calculates P_0 and P_1 ultra-fine-group elastic transfer matrices. CALCFN (290790 - 292630)

Calculates (2n+1)f_n(u) at all hyper-fine-group boundaries.

FNAVG (292640 - 294260)

Calculates group averaged values of elastic expansion coefficients.

SIGAVG (294270 - 296400)

Calculates group averaged cross sections from data pairs using interpolation laws.

SETCSD (296410 - 298790)

Prepares coefficients for calculation of moderating parameters.

MODPAR (298800 - 305270)

Calculates continuous slowing down moderating parameters.

XTRN (305280 - 308120)

Calculates ultra-fine-group extended transport cross sections.

EDIT1 (308130 - 309270)

Edits elastic transfer matrices.

EDIT2 (309280 - 311930)

Edits macroscopic ultra-fine-group vectors.

CSC009 (311940 - 322980)

Main program driver for ultra-fine-group spectrum and broadgroup cross section calculation.

FILE1 (322990 - 324560)

Loads data from record 5 of file MCC2F1 to determine storage requirements for inelastic data.

FILE6 (324570 - 331140)

Reads the file MCC2F6 inelastic and (n,2n) data and writes the - file GRPORD.

ATNSRC (331150 - 332070)

Reads file ATNUAT to determine number of spin states and resonances.

EGRID (332080 - 333080)

Sets up energies at which unresolved attenuation factors act.

REORDR (333090 - 336350)

Reads unresolved resonance integrals from file ATNUAT and orders them by energy.

MATCH (336360 - 339070)

Reads resolved resonance integrals from file RESINT and sets up problem material - resonance material correspondence.

INIT (339080 - 340870)

Sets initial fission source distribution.

SETIN (340880 - 347570)

Sets up storage for inelastic and (n, 2n) data.

DRIVER (347580 - 354540)

Controls ultra-fine-group spectrum calculation.

MGSPEC/MGSPC1 (354550 - 358240)

Solves multigroup spectrum equations,

CSDSPC/CSDSP1 (358250 - 363430)

Solves continuous slowing down spectrum equations.

FISSOR/FISOR1 (363440 - 366990)

Calculates ultra-fine-group fission source and eigenvalue.

BSQITR (367000 - 368940)

Tests on convergence of buckling search and sets next buckling guess.

PARAB (368950 - 369350)

Computes coefficients for parabolic fit to three points.

INSCAT/INSCT1 (369360 - 376090)

Calculates inelastic and (n,2n) ultra-fine-group scattering source.

PROBIN (376100 - 378370)

Calculates discrete inelastic scattering source taking account of energy-angle correlation.

TABINT (378380 - 381170)

Calculates inelastic (n,2n) scattering probability from tabulated function.

EDTUFG (381180 - 381600)

Edits ultra-fine values of array input.

SETBG (381610 - 391130)-

Controls calculation of broad-group cross sections.

BGSPEC (391140 - 394860)

Calculates broad-group fluxes and currents; sets correspondence between broad- and ultra-fine-groups.

INSBG (394870 - 402660)

Calculates inelastic and (n,2n) broad group scattering source. RESCS (402670 - 408030)

Calculates homogeneous broad group resonance capture, fission and total cross sections.

RESCAT (408040 - 412060)

Calculates resolved resonance broad group elastic transfer matrices.

BGFLE1 (412070 - 415090)

Sets data and writes first three records of file ISOTXS.

BGFLE? (415100 - 421560)

Writes isotope dependent data to file ISOTXS.

ISOCHI (421570 - 422650)

Calculates broad-group chi vectors by isotope.

PRINXS/PRNXS1 (422660 - 429660)

Calculates broad-group non-resonance cross section vectors.

BGSCAT (429670 - 434230)

Calculates non-resolved resonance broad-group elastic scattering transfer matrices.

SRATE (434240 - 435890)

Writes the file SRATES for use by RABANL.

DOIO/DOIO1/DOIO2/DOIO3 (435900 - 438780)

Prepares data for file SRATES.

ORIGIN (438790 - 438840)

Dummy routine to replace SIMPLOTTER subroutine.

Main driver for broad-group fundamental mode spectrum calculation.

HOMOG (444410 - 448100)

Prepares macroscopic broad group data.

BGPN (448110 - 451640)

Solves broad-group $P_{_{\rm N}}$ equations.

BSQTR (451650 - 453570)

Controls buckling search calculation.

CRAMER (453580 - 453980)

Calculates coefficients for parabolic fit to three sets data points.

BGSORS (453990 - 454540)

Calculates inhomogeneous broad-group source by summing ultra-fine-group sources.

Main driver for editor of broad-group cross section file ISOTXS.

XSEDIT (458220 - 463840)

Main driver for integral transport theory resolved resonance calculation.

E2E3E4/E2/E3/E4 (481320 - 483230)

This function obtains the exponential integrals E2(X), E3(X), and E4(X) by interpolation on tabular values or asymptotic expansions as appropriate.

FREEUP (483240 - 483760)

Closes data sets MCC2F1, MCC2F4, and SRATES.

MERGER (483770 - 489790)

Forms the union of all unique problem and foil materials and computes arrays.

SPOOL (489800 - 495000)

Reads in the elastic, fission, capture, and total ultra fine group cross sections from scratch data sets.

FLIPIT (495010 - 495630)

Converts a slab cell with reflective boundary conditions to the equivalent cell with periodic boundaries.

SYM (495640 496320)

Determines symmetrical mesh interval pairs for slab geometry.

SIFTIT (496330 - 507750)

Selects materials from data set MCC2F4, calls STUFER to load arrays, and calls MANAGR to proceed with the integral theory resolved resonance calculation.

STUFER (507760 - 510000)

Reads data set MCC2F4, loads arrays, multiplies SIGO by the isotope abundances, and obtains the square roots of the resolved resonance energies.

REARNG (510010 - 511040)

Arranges resonances in order of decreasing energy.

EZ3/EZ4 (511050 - 511840)

This function obtains S3(Z,H) defined as the sum of E3(Z+KH) for K between zero and infinity. EZ3 is used for optically thin unit cells.

YZ3 (511850 - 512940)

Obtains the difference in the sums of E3(Y+KH) and E3(Y+DEL+KH) for K equal zero to infinity, S3(Y,H) - S3(Y+DEL,H).

YZ4 (512950 - 513790)

Obtains the difference in the sums of E4(Y+KH) and E4(Y+DEL+KH) for K equal zero to infinity, S4(Y,H) - S4(Y+DEL,H).

XTRAP/XTRAP4 (513800 - 514510)

Obtains S3(Z3,H) defined as the sum of E3(Z3+KH) for K between zero and infinity using gaussian quadrature.

PFUNC (514520 - 515070)

Obtains the slab collision escape probability.

SOARCE/SORCE1/SORCEF (515080 - 523730)

Computes the scattering source into the current hyperfine-group.

MANAGR (523740 - 539000)

Acts as the driver to complete the integral transport theory calculation.

RESXSC/RESTES/XSECT (539010 - 544420)

Determines the resolved resonances to be included for the calculation of the hyper-fine-groups within each intermediate group, and computes the resolved resonance related cross sections for each hyper-fine-group.

RATES/RATESF (544430 - 559260)

Obtains the collision rates for the current hyper-fine-group and for each mesh interval of each region in the cell.

PREEDT (559270 - 561730)

Reads the data set IRESCS and writes out the rearranged data onto scratch data sets, one for each region, in preparation for editing the integral transport resonance cross sections.

EDTICS (561740 - 567650)

Reads scratch data sets, loads arrays, and edits the resolved resonance cross sections.

PREFOL (567660 - 569850)

Reads data set SCR005 and writes out the rearranged data onto scratch data sets, one for each foil, in preparation for editing the resonance foil cross section.

EDTFOL (569860 - 572050)

Reads scratch data sets, load arrays, and edits the resolved resonance cross sections for foil materials.

DRIVED (572060 - 576610)

Acts as the driver for the editing functions of the integral transport theory calculation.

PREEDI (576620 - 577880)

Reads the data set IRESCS and loads arrays for the case when arrays can be core contained in preparation for editing the integral transport theory resonance cross sections.

EDTICI (577890 - 583220)

Edits the resolved resonance transport theory resonance cross sections when arrays can be core contained.

PREFOI (583230 - 584500)

Reads scratch data set and loads arrays for the case when arrays can be core contained in preparation for editing foil cross sections.

EDTFOI (584510 - 585960)

Edits resolved resonance cross sections for foil material when arrays can be core contained.

DRIVEI (585970 - 589070)

Acts as the driver for the editing functions of the integral transport theory calculation when arrays can be core contained.

TABLE IX. MC²-2 ENDF/B-IV Library Files

Maximum Energy- 1.419 x 107 eVLethargy Width- 1/120Number of groups- 2082Number of Materials- 104

.

t

MC^2-2 ID	ENDF/B Mat Number	Comments
TA1824	1127	U, R
W-1824	1128	U, R
W-1834	1129 —	U, R
W-1844	1130	U, R
W-1864	1131	U, R
U-238B	1262	U, R, F, ENDF/B-IV Mat 1262 plus back- ground corrections
U-238M	1262	U, R, F, ENDF/B-IV Mat 1262 with MLBW resolved resonance flag
PU2404	1265	U, R, F
TA1814	1285	U, R
н-3 4	1169	
CA 4	1195	
HE 4	1270	
DY1644	1031	U, R
LU1754	1032	U, R
LU1764	1033	U, R
H-2 4	1120	· · ·
HE3 4	1146	
XE1244 '	1170	R
XE1264	1171	R
XE1284	、 1172	R
XE1294	1173	R
XE1304	1174	R
XE1314	1175	R
XE1324	1176	R
XE1344	1177	R

TABLE IX. MC²-2 ENDF/B-IV Library Files (Cont'd.)

MC^2-2 ID	ENDF/B Mat Number	Comments
XE1364	1178	
KR78 4	1181	•
KR80 4	1182	
KR82 4	1183	
KR83 4	1184	
KR84 4	1185	
KR86 4	1186	
ν 4	1196	
AL27 4	1193	
SI 4	1194	
MG 4	1280	
NI 4	1190	
CR 4	1191	
FE 4	1192	
SM1494	1027	U, R
GD 4	1030	R
RE1854	1083	U, R
RE1874	1084	U, R
RH1034	1125	U, R
TC99 4	1137	
AG1074	1138	R
AG1094	1139	R
CS1334	1141	R
CL 4	1149	
К 4	1150	
NA23 4	1156	1
B-11 4	1160	-
MN 55 4	1197	. ·
U-2384	1262	U, R, F
NP2374	1263	U, R, F
TI 4	1286	
MO 4	1287	· .
C059 4	1199	
EU1514	1290	U, R

-218-

MC^2-2 ID	ENDF/B Mat Number	Comments	
EU1534	1291	U, R	
EU1524	1292	U, R	
EU1544	1293	U, R	
CU 4	1295		
NB93 4	1189	. •	
F 4	1277		
CD 4	1281	· · · ·	
CD1134	1282	•	
AU1974	1283	U, R	
ZIRC 4	1284		
U-2354	1261	U, R, F	
PU2394	1264	U, R, F	
PU2414	1266	U, R, F	
PA2334	1297	U, R, F	
C-12 4	1274		
N-14 4	1275	,	
0-16 4	1276		
PB 4	1288		
U-2344	1043	U, R, F	
PU2384	1050	U, R, F	
AM2414	1056	U, R, F	
AM2434	1057	U, R, F	
PU2424	1161	U, R, F	
CM2444	1162	U, R, F	
U-2464	1163	U, R, F '	
LI-6 4	1271		
LI-7 4	1272		
B-10 4	1273		
XE1354	1294		
TH2324	1296	U, R, F	
U3FP13	1042		
U5FP13	1045		

MC^2-2 ID	ENDF/B Mat Number	Comments
P9FP13	1052	·····
U3FP23	1066	
U3FP 33	1067	
U5FP23	1068	
U5FP33	1069	
P9FP23	1070	
P9FP33	1071	
BE-9 3	1154	• •
CS1353	1,229	·
SM1503	1244	
PM148M	1254	
U-2334	1260	R, F, ENDF/B-III Fission Spectrum
HYDRGN	1269	

TABLE IX. MC²-2 ENDF/B-IV Library Files (Cont'd.)

U This material has unresolved resonance data

R This material has resolved resonance data

F This material has fission spectrum data

2. CDC Code Implementation

The implementation of the MC^2-2 code has been performed at two CDC computer laboratories - Lawrence Berkeley Laboratory and Brookhaven National Laboratory. Each laboratory has one CDC 7600 computer with 65,536 (60-bit) words of small core memory and 512,000 (60-bit) words of large core memory. Also each laboratory has two CDC 6600 computers with 65,536 (60-bit) words of central memory. MC^2-2 was implemented only on the CDC 7600 computers. The examples in this section display the control cards suitable for each installation. The essential nature of each step is intended to be system independent. Each CDC installation has local conventions but Brookhaven National Laboratory adheres almost entirely to standard CDC conventions. The user will find all the necessary CDC control card information and FORTRAN usage in the following four manuals.

- FORTRAN Extended 4 Reference Manual (Publication No. 60305600, Rev. G)
- SCOPE 2.1 Reference Manual (Publication No. 60342600, Rev. H)
- 3. Loader Reference Manual (Publication No. 60344200, Rev. G)
- Update Reference Manual (Publication No. 60342500, Rev. E)

The CDC program tape is written as described in Table X. The source code for MC^2-2 is contained on Files 1 and 2 of the program tape. A brief description of each of the FORTRAN subprograms of File 1 is given in Table VIII, and the subprograms and sequence numbers in this table are the same for both IBM and CDC codes. The source code differs in that all cards between two cards with CDC* in columns 1-4 have a C in column 1 on the IBM tape and a blank in column 1 on the CDC tape. The opposite change is true of cards between two cards with CIBM in columns 1-4. The 14 subprograms on File 2 of the tape are discussed in Table XI. The first 13 subprograms are written in FORTRAN, and the last subprogram, MEMGET, is a COMPASS routine. Creation of object code for input to the segmentation loader is the first step required for implementation of the MC^2-2 program. The compilation of File 1 subprograms and the first 13 subprograms of File 2 should be performed using the highest level of optimization available that will give the correct object code. Currently, this means use of the FORTRAN extended compiler with OPT=1 specified. The FORTRAN extended compiler is necessary because of the presence of LEVEL 2 statements. At the present time there are errors in code generated with the OPT=2 FORTRAN extended compiler. Also the CDC FORTRAN compiler release of 4.4 or later was used. Earlier releases generated incorrect object code for all available optimizations. Figure 23 displays the control cards necessary for reading the two files of the source program tape, compilation of the FORTRAN code, assembly of the one COMPASS routine and writing of the object code onto a magnetic tape. Depending on the installation, it may be necessary to isolate the one COMPASS routine in File 2 before assembly.

This is currently a requirement at the Berkeley site because of interface problems with their FORTRAN extended compiler. Therefore the CDC UPDATE routine was used to separate the FORTRAN and COMPASS source code. Generally for CDC computers, source code can be mixed FORTRAN and COMPASS subprograms. The FORTRAN compiler will recognize COMPASS routines and invoke the COMPASS assembler and then control will be returned to the FORTRAN compiler.

Figure 24 displays the control cards which read the generated object code from the magnetic tape. The segmented loader is then invoked and it builds the segments from the object code in accordance with the directives issued to the segmented loader. These directives are provided on File 3 of the program tape. The INCLUDE directives must be specified in a manner which is dependent on the CDC 7600 computer operating system. For an operating system less (greater) than SCOPE 2.0, a subroutine specified in the verb field must (must not) be specified in the specification field. For example, if subroutines A, B, and C are to appear in the same segment, then the directive statement under a SCOPE operating system equal or greater than 2.0 will be expressed as

A INCLUDE B,C

For a 7600 computer operating system less than 2.0 or the Berkeley 'BKY' operating system, the equivalent directive statement must be stated as

A INCLUDE B,C,A

The segmented loader resolves I/O routines and mathematical subroutines with the system library called FTN4LIB (Berkeley) or FORTRAN (Brookhaven). The executable code now resides on the file designated as MCCA. This file is written onto a magnetic tape.

Two library tapes are provided along with the MC^2-2 program tape. These library tapes contain eight binary sequential files with a total of 4559 records and 2,767,155 words in the format of the files MCC2F1-MCC2F8 described in Appendix C . These library tapes were written at the Berkeley Laboratory on 7 track unlabeled tapes at a density of 800 BPI and x-mode binary. Appendix F describes the process used for generating the MC^2-2 library files on a single 1600 BPI tape from these two 7 track tapes. These files must be copied to a direct access device (e.g. disk pack) as the next step in program implementation. Since no private disk space was available, the library was copied from tape to disk for each problem execution as illustrated in Fig. 25. The data set names assigned to the eight binary MC^2-2 library files are TAPE22-TAPE29 as displayed in Fig. 25. The data available on these library files were processed from the ENDF/B-IV data files by the code ETOE-2. A summary of the library specifications is given in Table IX. The Argonne Code Center also has available four BCD tapes which contain this same eight file library in BCD format along with a FORTRAN program, MC^2-2 LIBGEN, which reads the BCD files and writes the eight binary files. Appendix F illustrates this process. The binary library tapes generated at Berkeley Laboratory were made from the FORTRAN program MC^2-2 LIBGEN.

The control cards used to execute an MC^2-2 problem are given in Fig. 25. The executable code generated by the segmented loader and the eight binary library files are copied from tape to a disk pack. Note that the eight

binary library files are copied from tape to disk for every problem execution. For the Berkeley example the TAPE control card and its appropriate directives require 34 charge units (one charge unit costs the user about 7 cents) to transfer the binary library files from tape to disk pack and assign a data set name to each file. Brookhaven does not have a comparable control card. The control cards used to transfer the library files from tape to disk pack in the Brookhaven example are very inefficient. Staging the data and using the COPYBF control cards to assign a data set name to each of the eight files requires twice the disk space since two copies of the library exist on disk packs. When this method was used at Berkeley for comparison purposes the STAGE command required 8 charge units to transfer all the binary data from tape to disk and the COPYBF control cards which assign data set names to each file require an additional 267 charge units. The user should be aware of the BPOINTER container array size allocated in SCM and LCM and the appropriate field length specified on the CDC control card preceding MC^2-2 problem execution. The size of the dynamic storage capability, BPOINTER, described in Reference 1 and briefly in Appendix E, used to manage all variable dimension array allocations, is provided for on card type 02 of data set A.MCC2. For the CDC version of the code, a container array is allocated in SCM (up to 22,000 decimal words available in small core plus another 10,000 decimal words if no small core buffers exist) as well as a container array in LCM (up to 131,072 decimal words in directly addressable large core memory). Almost all of the large BPOINTER arrays in the CDC version of the MC^2-2 code are placed in the container array in LCM. At this time, no known limitations due to core size are placed upon executions of the CDC version of this code.

Two BCD sample problem input decks are provided on File 4 of the program tape. The first problem displays the execution of a homogeneous consistent P₁ problem. The type Ol card of data set A.STPO15 is set to execute Areas 4, 5, 6, 6.5, 7, 8, and 9 and to edit the broad group cross sections. The second sample problem shows the execution of a homogeneous resolved resonance integral transport calculation. The type Ol card of data set A.STPO15 is set to execute Areas 4 and 10. Data set UNREG, the unresolved resonance ultra-fine-group cross sections, (called TAPE 49) was saved from the first problem. This data set is input to the second problem by specifying DATASET=UNREG under the data set initialization BLOCK=OLD.

File Number	Contents	Number of Card Images
1	MC ² -2 Subprograms See Table VIII	58907
2	MC ² -2 Subprograms for CDC Code See Table XI	654
3	Directives Required for Segmented Loader See Fig. 17	48
4	Two Sample MC ² -2 Problem Input Decks	150

TABLE X. CDC Code Center Tape Description

TABLE XI. Function of MC^2-2 Subprograms for CDC Code

ALLOC1/ALLOC2+ (000010 - 000760)*

Assign storage for the dynamic allocation subprogram package BPOINTER.

FREE1/FREE2 (000770 - 001010)

Dummy FORTRAN routine that releases the storage of the dynamic storage container array that was allocated by the subroutine ALLOC1/ALLOC2.

ABEND/ABSTOP[†] (001020 - 001080)

Called by subroutine ERROR to force an abnormal termination of the job with a DUMP.

FXP1/FXP† (001090 - 001580)

Fast exponential function.

GOWEST[†] (001590 - 001840)

Left-justify Hollerith variables.

DISPOS† (001850 - 001950)

Used by module CSI010 to determine whether the file SRATES is given a permanent or temporary disposition. CDC code always, sets a temporary disposition to the file.

FILEID[†] (001960 - 002110)

A dummy routine that is used in subroutine BGFLE1 of module CSC009 to obtain the dataset name associated with the file ISOTXS.

CVD⁺ (002120 - 002240)

Used in subroutine BGFLE2 of module CSC009 to convert a variable from integer to Hollerith format.

DRED/DRIT (002250 - 004040)

FORTRAN routine that performs the non-formatted (binary) I/O operations between LCM and disk files without use of a buffer array in small core memory.

ZEROIO (004050 - 004150)

FORTRAN routine that initializes the common block /INITIO/ which is common to subroutines REED and DRED.

TABLE XI. Function of MC^2-2 Subprograms for CDC Code (Contd.)

BUOPEN/BUCLOSE (004160 - 004200)

A dummy FORTRAN routine that is intended to open/close dataset buffers.

ERF (004210 - 005130)

A FORTRAN written function that evaluates the error function.

PRTECS (005140 - 006320)

A FORTRAN subroutine that edits a one-dimensional floating point single precision array located in LCM.

MEMGET/MEMGET1 (006330 - 006540)

A COMPASS routine that performs the allocation for ALLOC1/ALLOC2.

Numbers in brackets are sequence numbers of routine on File 2 of CDC program tape.

[†]This is a FORTRAN routine for the CDC code. See Section D above for a further description of this routine.

BERKELEY

(JOBCARD) SCF(R=BCDIN, RL=80, BF=40, D8, nnnnn) COPYBF(BCDIN, TAPE1, 1) COPYBF(BCDIN, TAPE2, 1) REWIND(BCDIN, TAPE1, TAPE2) RFL,100000,100000. FTN4, I=TAPE1, OPT=1, L=0, B=FILE1. REWIND(TAPE1, FILE1) UPDATE(N) UPDATE(F,P=NEWPL) REWIND(COMP) FTN4, I=COMP, OPT=1, B=FILE2A. REWIND (COMP, NEWPL, FILE 2A) UPDATE(F,P=NEWPL) REWIND(COMP) COMPASS, I=COMP, B=FILE2B. REWIND(COMP, FILE2B) COPY (FILE1, 1FXF, FILE2A, 1FXF, FILE2B, 1F, PG) REWIND(PG) STAGE(PG, D9, NT, P3, W, nnnnn, 1F) 7/8/9 *DECK MC *READ TAPE2 7/8/9 *COPY MC, MC.2, MC.633, COMP 7/8/9 *COPY MC, MC.634, MC.655, COMP 7/8/9 6/7/8/9

BROOKHAVEN

(JOBCARD) ACCOUNT (Name, NNNN) STAGE (BCDIN, HY, VSN=Nnnnn) STAGE(PG, POST, PE, E, VSN=Knnn) FILE(BCDIN,RT=F,FL=80,RB=40,BT=K,CM=YES) COPYBF(BCDIN, TAPE1, 1) COPYBF(BCDIN, TAPE2, 1) REWIND(TAPE1, TAPE2) RETURN (BCDIN) FTN(1=TAPE1,OPT=1,L=0,B=FILE1,SYSEDIT,PL=70000) REWIND(TAPE1, FILE1) FTN(1=TAPE2,OPT=1,B=FILE2,SYSEDIT,PL=70000) REWIND(TAPE2,FILE2) COPYBF(FILE1,PG,1) COPYBF(FILE2,PG,1) REWIND(FILE1, FILE2, PG) 6/7/8/9

-227

Fig. 23. Compilation and Assembly of CDC Source Code to Generate Object Code

BERKELEY

BROOKHAVEN

(JOBCARD)
STAGE,PG,D9,NTP3,R,nnnnn.
RFL,100000,1000.
SEGLINK(F=PG,P=FTN4LIB,B=MCCA,LO=BEX)
REWIND(MCCA)
STAGE,MCCA,D9,NT,P3,W,nnnnn.
7/8/9
 (Insert directives required for
 the segmented loader. See Fig. 17)
7/8/9

6/7/8/9

(JOBCARD) ACCOUNT (Name, NNNN) STAGE(FLE1,PE,E,VSN=Knnn) STAGE(MCCB,POST,PE,E,VSN=Knnn) COPYBF(FLE1,PG1,1) COPYBF(FLE1, PG2, 1) REWIND(PG1,PG2) RETURN(FLE1) RFL(140000,L=1) SEGLOAD(B=MCCA) LOAD(PG1) LOAD(PG2) NOGO. REWIND(MCCA) COPY (MCCA, MCCB) REWIND(MCCB) 7/8/9 (Insert directives required for the segmented loader. See Fig. 17) 7/8/9

6/7/8/9

Fig. 24. Build the Segments from Object Code Using the Segmentation Loader

BERKELEY

BROOKHAVEN

-229-

(JOBCARD)	(JOBCARD)
DISKHOG,12000.	ACCOUNT (Name, NNNN)
TAPE, TAPE22, D9, X, NT, R, nnnnn.	STAGE(LIB,PE,E,VSN=Knnn)
STAGE, MCCB, D9, NT, P3, R, nnnnn.	STAGE (MCCA, PE, E, VSN=Knnn)
RFL,170000,400000.	COPYBF(LIB, TAPE22, 1)
MCCB (PL=70000)	COPYBF(LIB, TAPE23, 1)
7/8/9	COPYBF(LIB, TAPE24, 1)
*READ	COPYBF(LIB, TAPE25, 1)
*FILE,TAPE23	COPYBF(LIB, TAPE26, 1)
*READ	COPYBF(LIB, TAPE27, 1)
*FILE,TAPE24	COPYBF(LIB, TAPE28, 1)
*READ	COPYBF(LIB, TAPE29, 1)
*FILE, TAPE25	REWIND(TAPE22, TAPE23, TAPE24, TAPE25, TAPE26, TAPE27)
*READ /	REWIND(TAPE28, TAPE29)
*FILE,TAPE26	RETURN(LIB)
*READ	COPY (MCCA, MCCB)
*FILE,TAPE27	REWIND (MCCB)
*READ	RETURN (MCCA)
*FILE,TAPE28	RFL(160000,L=400)
*READ	MCCB.
*FILE, TAPE29	7/8/9
*READ	(BCD MC ² -2 problem input deck)
7/8/9	7/8/9
(BCD MC ² -2 problem input deck)	6/7/8/9
7/8/9	
6/7/8/9	

Fig. 25. Execution of MC²-2 Problem

APPENDIX A

SPECIAL FUNCTION EVALUATIONS

I. Legendre Functions of the Second Kind

In the B_{N} approximation the extended transport cross section depends on a ratio of Legendre functions of the second kind,

$$R \equiv \frac{Q_{N+1}(z)}{Q_{N}(z)}$$

where

$$1 \le N \le 9$$
$$z \equiv \frac{i\Sigma}{B}$$

and the buckling B can be either real or imaginary. The calculation of the ratio R depends upon both N and z. In Table XII the computational option is identified with the values of N and z. In the expressions below defining the options B is used to represent the real or imaginary part of $\sqrt{B^2}$.

1) Option 1 (Analytic)

$$R_{1} = \frac{i}{2} \frac{\left[\left(1 + \frac{3\Sigma_{t}^{2}}{B^{2}}\right) \tan^{-1}\frac{B}{\Sigma_{t}} - \frac{3\Sigma_{t}}{B} \right]}{\frac{\Sigma_{t}}{B} \tan^{-1}\frac{B}{\Sigma_{t}} - 1}$$

2) Option 2 (Analytic)

$$R_{2} = \frac{\left[\frac{3\Sigma_{t}^{2}}{B^{2}} - 1\right] \ell_{n} \left|\frac{1 + B/\Sigma_{t}}{1 - B/\Sigma_{t}}\right| - 1}{\frac{\Sigma_{t}}{2B} \ell_{n} \left|\frac{1 + B/\Sigma_{t}}{1 - B/\Sigma}\right| - 1}$$

3) Option 3 (Hypergeometric Series)

 $\rm Q_N$ and $\rm Q_{N+1}$ are calculated using an eleven term hypergeometric series expansion, $^{\rm (30)}$

t

^{3Σ}t

$$Q_N(x) = B(N, 0, x) F(\frac{1}{2}, \frac{1}{2}; N + \frac{3}{2}; - t)$$

 $x = \frac{\Sigma_t}{B}, t = \frac{x - (x^2 - 1)^{\frac{1}{2}}}{2(x^2 - 1)^{\frac{1}{2}}}$

$$B(N, 0, x) = \frac{\Gamma(\frac{1}{2}) \Gamma(N + 1)}{\Gamma(N + \frac{3}{2}) \sqrt{2} (x^{2} - 1)^{\frac{1}{4}} (x + (x^{2} - 1)^{\frac{1}{2}})^{N + \frac{1}{2}}}$$

$$F(\frac{1}{2}, \frac{1}{2}; N + \frac{3}{2}; -t) = \sum_{k=0}^{\infty} \frac{\Gamma(\frac{1}{2} + k) \Gamma(\frac{1}{2} + k) \Gamma(N + \frac{3}{2})}{k! \Gamma(\frac{1}{2}) \Gamma(\frac{1}{2}) \Gamma(N + \frac{3}{2} + k)} (-t)^{k}$$

with the coefficients of the eleven term summation precalculated. The ratio, R_3 , is then calculated from $Q_{N+1}(x)/Q_N(x)$.

4) Option 4 (Backward Recursion)

The ratio R_4 is calculated using Gautschi's backward recursion relation⁽³¹⁾. Setting

$$v = N + 9$$

$$S = -\frac{B^2}{|B^2|}$$

$$r_0 = 0$$

the recursive relationship

$$r_{n} = \frac{\frac{(v - n + 1)}{(v - n + 2)}}{\frac{\Sigma_{t}}{B} \left[1 + \frac{v - n + 1}{v - n + 2}\right] + S r_{n-1}} \qquad n = 1, 2, ..., 9$$

is used to define r_q . The ratio R is then given by

	- ir ₉	$B^2 > 0$
$R_4 = $	rġ	$B^2 < 0$

5) Option 5 (Forward Recursion)

The ratio R_5 is calculated using the standard forward recursion relation. Setting

$$\mathbf{r}_{o} = \begin{cases} -\tan^{-1}\frac{B}{\Sigma_{t}} & B^{2} > 0\\ \frac{1}{2} \ln \left|\frac{1+\Sigma_{t}/B}{1-\Sigma_{t}/B}\right| & B^{2} < 0 \end{cases}$$
$$\mathbf{r}_{1} = \frac{\Sigma_{t}}{B} S \mathbf{r}_{o} - 1$$
$$S = -\frac{B^{2}}{|B^{2}|}$$

the recursive relationship

$$r_{n+1} = \frac{\Sigma_{t}}{B} S^{n+1} r_{n} - r_{n-1} + \frac{\Sigma_{t}}{B} S^{n+1} r_{n}$$
$$- \left(\frac{\Sigma_{t}}{B} S^{n+1} r_{n} - r_{n-1}\right) / (n+1) \qquad n = 1, 2, ..., N$$

is used and R is defined

$$R_{5} = \begin{cases} \frac{r_{N+1}}{r_{N}} & B^{2} < 0\\ i \frac{r_{N+1}}{r_{N}} & B^{2} < 0 \end{cases}$$

6) Option 6 (Relationship with P_N)

Use is made of the relationship between Legendre functions of the first kind $P_n(z)$ and the second kind $Q_N(z)$ to calculate the ratio R_6 . Since the forward recursion relation is stable for all P_n , the Legendre functions of the first kind are calculated directly from it,

$$P_{0}(z) = 1$$

$$|P_{1}(z)| = \frac{\Sigma_{t}}{B}$$

$$|P_{n}(z)| = (-)^{n+1} \frac{\Sigma_{t}}{B} |P_{n-1}(z)| - |P_{n-2}(z)| + (-)^{n+1} \frac{\Sigma_{t}}{B} |P_{n-1}(z)|$$

$$- \left[(-)^{n+1} \frac{\Sigma_{t}}{B} |P_{n-1}(z)| - |P_{n-2}(z)| \right] / (n - 1)$$

$$n = 2 \cdot 3 \quad \text{we } N + 1$$

From the $P_n(z)$, the non-singular part of $Q_N(z)$ and $Q_{N+1}(z)$ is calculated,

$$(-)^{\left[\frac{N}{2}+1\right]} W_{N} = \frac{2N+1}{N+1} P_{N} - \frac{2N-3}{3N} P_{N-2} + \frac{2N-7}{5(N-1)} P_{N-4} + \dots$$

$$(-)^{\left[\frac{N+1}{2}+2\right]} W_{N-1} = \frac{2N-1}{N} P_{N-1} + \frac{2N-5}{3(N-1)} P_{N-3} + \dots$$

The functions \boldsymbol{Q}_{N} and \boldsymbol{Q}_{N+1} are then calculated as,

-233-

$$Q_{N+1} = (-)^{\left[\frac{N+1}{2} + 2\right]} \left(\frac{\pi}{2} - \tan^{-1}\frac{\Sigma_{t}}{B}\right) \left|P_{N+1}\left(\frac{i\Sigma_{t}}{B}\right)\right| + W_{N}$$
$$Q_{N} = (-)^{\left[\frac{N}{2} + 2\right]} \left(\frac{\pi}{2} - \tan^{-1}\frac{\Sigma_{t}}{B}\right) \left|P_{N}\left(\frac{i\Sigma_{t}}{B}\right)\right| + W_{N-1}$$

-234-

and

 $R_6 = \frac{Q_{N+1}}{Q_N}$

.
-235-

TABLE 2	X]
---------	-----

BLE XII Legendre Function Evaluation

1	I	B ²	$ \Sigma_t/B $	1. T	Option
	1	· > 0	all :	1	Analytic
	1	< 0	all	2	Analytic
· >	1	< 0	(1.0, 1.15	3	Hypergeometric series
>	1	· < 0	1.15	4	Backward recursion
>	1	> 0	> 1.0	4	Backward recursion
>	1	< 0	<u><</u> 1.0	5	Forward recursion
<	4	> 0	<u><</u> 1.0	5	Forward recursion
_>	1	> 0	(0.0, 0.45]	5	Forward recursion
<u>></u>	4	> 0	(0.45, 1.0]	6	Relationship with P_{N}

.

.

II. A Fast Exponential Subroutine - FXP

Approximately half of the central processor time in a heterogeneous slab calculation in the RABANL module of MC^2-2 is devoted to the calculation of exponentials, $e^{-|\mathbf{x}|}$. In the ultra-fine-group calculation, about 25% of the CPU time for a problem with many isotopes is spent in the calculation of exponentials to treat the inelastic and (n, 2n) evaporation laws. The standard IBM Fortran exponential function is quite general with regard to permissable arguments and quite accurate. On the other hand it is relatively slow ($_{0}$ 9 x 10⁻⁶ sec. on a 360/195). Large gains in speed at the expense of accuracy and core storage are obtained by use of an assembler language routine FXP. For arguments in the range $0 \le x \le 18$, the program FXP uses a linear interpolation scheme

$$e^{-\mathbf{x}} \stackrel{\circ}{\leftarrow} \mathbf{y}_{i} - \mathbf{m}_{i}\mathbf{x} \qquad 0 \leq i \leq 1022$$

where

$$y_{i} = y_{i}^{*} + i\Delta m_{i} \qquad 0 \le i \le 1022$$

$$m_{i} = \frac{y_{i}^{*} - y_{i+1}^{*}}{\Delta} \qquad 0 \le i \le 1022$$

$$y_{i}^{*} = e^{-\Delta} y_{i-1}^{*} \qquad 0 \le i \le 1022$$

$$y_{o}^{*} = \frac{2}{\Delta} \frac{(1 - e^{-\Delta})}{(1 + e^{-\Delta})}$$

$$A' = 18.0/1022$$

(22) This approximation was shown by Olson to be sufficiently accurate for calculations of collision probabilities in the RABID code . The programming of the assembler language routine FXP was optimized to a 195 by A. Hinds of ANL. Computation times less than half that of the Fortran exponential functions are typical of FXP on an IBM 370/195. A Fortran equivalent of FXP is available with the CDC code. The Fortran routine is approximately 20% faster than the CDC library exponential routine.

III. Calculation of Doppler-Broadened Line Shape Functions

In the computation of Doppler-broadened resonance cross sections, use is made of the symmetric and anti-symmetric line shape functions. These functions, usually denoted as ψ and χ , are defined in terms of the real and imaginary parts of the error function for complex arguments as

$$\psi(a,b) = \frac{a\sqrt{\pi}}{2} \text{ Re } W(\frac{ab}{2}, \frac{a}{2})$$
 (A.1)

$$\chi(a,b) = a\sqrt{\pi} \text{ Im } W(\frac{ab}{2}, \frac{a}{2})$$
 (A.2)

where

$$W(z) = W(x,y) = \exp(-z^2) \operatorname{erfc}(-iz)$$
(A.3)

and z = x + iy.

ReW(x,y) and ImW(x,y) were precalculated using the methods described in Ref. 32 and stored in coarse and fine mesh tables as indicated below. In the fine mesh tables, y ranges between -0.02 and 0.5 with the mesh -0.02(0.02)0.5 while in the coarse mesh tables y ranges between 0.4 and 3.0 with the mesh 0.4(0.1)3.0. In both tables, x ranges between -0.1 and 3.9 with the mesh -0.1(0.1)3.9.

If $|x| \le 3.9$ and $y \le 3.0$, the ReW and ImW are obtained using the six-point bivariate interpolation formula (25.2.67) of Ref. 33 in either the fine or coarse mesh tables as appropriate to the value of y.

If $|\mathbf{x}| > 3.9$ or $\mathbf{y} > 3.0$ but $|\mathbf{x}| \le 6.0$ and $\mathbf{y} \le 6.0$, W(z) is approximated by ⁽³⁴⁾

$$W(z) = iz \sum_{i=1}^{3} \frac{a_i}{z^2 - b_i}$$

where

 $a_1 = 0.4613135$ $b_1 = 0.1901635$ $a_2 = 0.09999216$ $b_2 = 1.7844927$ $a_3 = 0.002883894$ $b_3 = 5.5253437$

Setting z = x + iy, Eq. A.4 yields

(A.4)

$$ReW = \sum_{i=1}^{3} \frac{a_i \left[-y \left(x^2 - y^2 - b_i \right) + 2 x^2 y \right]}{\left(x^2 - y^2 - b_i \right)^2 + 4 x^2 y^2}$$
(A.5)

$$ImW = \sum_{i=1}^{3} \frac{a_i \left[x \left(x^2 - y^2 - b_i \right) + 2 x y^2 \right]}{\left(x^2 - y^2 - b_i \right)^2 + 4 x^2 y^2}$$
 (A.6)

If $|\mathbf{x}| > 6.0$ or $\mathbf{y} > 6.0$ but $|\mathbf{x}| \le 100.0$ and $\mathbf{y} \le 100.0$, W(z) is approximated by

$$W(s) = is \sum_{i=1}^{2} \frac{c_i}{z^2 - d_i}$$
(A.7)

where

$$c_1 = 0.5124242$$

$$d_1 = 0.2752551,$$

$$c_2 = 0.05176536$$

$$d_2 = 2.724745$$

Thus

$$ReW = \sum_{i=1}^{2} \frac{c_i \left[-y \left(x^2 - y^2 - d_i \right) + 2 x^2 y \right]}{\left(x^2 - y^2 - d_i \right)^2 + 4 x^2 y^2}$$
(A.8)

$$ImW = \sum_{i=1}^{2} \frac{c_i \left[x \left(x^2 - y^2 - d_i \right) + 2 x y^2 \right]}{\left(x^2 - y^2 - d_i \right)^2 + 4 x^2 y^2}$$
 (A.9)

If |x| > 100.0 or y > 100.0, the code uses an asymptotic approximation. The integral representation of W(z) is given by formula 7.1.4 of Ref. 33 as

$$W(z) = \frac{i}{\pi} \int_{-\infty}^{\infty} \frac{e^{-t^2} dt}{z - t}$$
(A.10)

where the w and t are the weights and abscissae for the Hermite quadrature. In particular, for very large x or y, we set N = 2 and ignore t relative to z so that

$$W(z) \quad \stackrel{\sim}{\sim} \frac{2i}{\pi} \frac{w}{z} \tag{A.12}$$

where w = $\sqrt{\pi}/2$. Setting z = x + iy, we have finally

ReW =
$$\frac{y}{\sqrt{\pi} (x^2 + y^2)}$$
 (A.13)

$$ImW = \frac{x}{\sqrt{\pi} (x^2 + y^2)}$$

For the special case of x = 0, Eq. A.3 becomes

$$W(o,y) = e^{y^2} \operatorname{erfc}(y).$$
 (A.15)

If y < 2.0, Eq.A.15 is evaluated using the rational approximation (7.1.26) of Ref. 33. If $2.0 \le y \le 6.0$, Eq.A.15 is evaluated using Eq.A.5 with x set equal to zero. If $6.0 \le y \le 100.0$, Eq.A.15 is evaluated using Eq.A.8 with x set equal to zero. Finally, for y > 100.0, Eq.A.15 is evaluated using Eq.A.13 with x set equal to zero.

۹<u>ن</u> ۱

(A.14)

IV. Numerical Methods Used in Calculating the Isolated Resonance Integral

The three terms of the factored isolated resonance integral $J(\beta, \theta, 0, 0)$, $I(\beta, \theta, a)$, and $M(\beta, \theta, a)$ given respectively in Eqs.III.46-III.48 are evaluated using fixed point Gauss-Jacobi quadrature or asymptotic algorithms depending upon the value of β . The code calculates the quantity

$$T1 = \frac{\beta + \psi(\theta, 0)}{\psi(\theta, 0)} . \qquad (A.16)$$

If T1 > 4.5, asymptotic expressions are evaluated as described later. Otherwise, the code evaluates the integrals as follows. (25)

The variable of integration is changed from \mathbf{x} to \mathbf{u} using the transformation

$$x = K \frac{u}{\sqrt{1 - u^2}}$$
 (A.17)

so that in general

$$\int_{0}^{\infty} f(x) dx = \frac{1}{K} \int_{0}^{1} \frac{du}{\sqrt{1 - u^{2}}} \cdot \frac{f(x(u))}{1 - u^{2}}$$
$$= \frac{\pi/N}{K} \left\{ \frac{1}{2} f(0) + \frac{(N-1)/2}{\sum_{i=2}^{L} \frac{f(x(u_{i}))}{1 - u_{i}^{2}}} \right\} + R_{N} . \quad (A.18)$$

The Gauss-Jacobi quadrature points u, are given by

$$u_i = \cos \frac{(2i - 1) \pi}{2N}$$
 (A.19)

the quadrature weights are constant and equal to $\pi/N,$ and the remainder is given

$$R_{N} = \frac{\pi}{(2N)! 2^{2N-1}} f^{(2N)}(\xi), \quad 0 < \xi < 1.$$
 (A.20)

The value selected for N (either 9 or 13) and the value for K is determined (25) as shown in Fig. 26 based upon the values of θ and β .

Now when $T1 \ge 4.5$, Hwang has shown ⁽²⁵⁾ that one may write

$$J(\beta,\theta, 0, 0) = T \int_{0}^{\infty} dx \frac{\psi(\theta, \mathbf{x})}{\beta + S \psi(\theta, \mathbf{x})}$$
$$\approx T \left\{ \frac{\frac{\pi}{2}}{\beta + \rho} + S^{2} \frac{\int_{0}^{\infty} \psi^{3}(\theta, \mathbf{x}) dx}{(\beta + \rho)^{3}} - \frac{\frac{\pi}{2} \rho^{2}}{(\beta + \rho)^{3}} \right\}$$
(A.21)

-240-



Fig. 26. Gauss-Jacobi Quadrature Selection

$$I(\beta,\theta, a) = Ta^{2} \int_{0}^{\infty} dx \frac{\psi}{\beta + S\psi} \cdot \frac{\chi^{2}}{(\beta + S\psi)^{2} - a^{2}\chi^{2}}$$

$$\approx \frac{4a^{2}}{3} T \frac{\int_{0}^{\infty} \psi^{3} dx}{(\beta + \rho)^{3}} \qquad (A.22)$$

$$M(\beta,\theta, a) = a \int_0^\infty dx \frac{\chi^2}{(\beta + s\psi)^2 - a^2\chi^2}$$

 $\frac{2\pi a(\beta + 3\rho)\rho}{S(\beta + \rho)^{3}} - \frac{8aS}{3(\beta + \rho)^{3}} \int_{0}^{\infty} \psi^{3} dx .$ (A.23)

In Eqs.A.21-A.23,

$$\rho = \frac{S}{2} \psi(\sqrt{2} \theta, 0) \tag{A.24}$$

and $\psi(\sqrt{2} \ \theta, 0)$ is evaluated using Eq.A.15.

The third order integral which appears in Eqs. A.21-A-23 is evaluated using a power series expansion, or if $\theta > 2.5$, an asymptotic expansion.

For
$$\theta \leq 2.5$$

$$\int_{0}^{\infty} \psi^{3}(\theta, \mathbf{x}) \, d\mathbf{x} \approx \frac{\pi}{2} \left\{ \frac{\rho}{S} \left[3 \, \psi(\sqrt{2/3} \, \theta, \, 0) - \sqrt{3\pi/2} \, \theta \, e^{\frac{\theta^{2}}{6}} \right] + \frac{3}{2} \, \theta^{2} \, S_{1} \right\} \quad (A.25)$$

where

$$S_{1} = \frac{1}{4} + \sum_{n=1}^{6} A_{n} \left[\frac{1}{4} + \sum_{m=1}^{n} B_{m} \theta^{2m} \right]$$
(A.26)

and

$$A_{n} = \frac{n!}{1 \cdot 3 \cdot 5 \cdot \cdots (2n + 1)} \frac{1}{2^{n}}$$

$$B_{m} = \frac{1}{(\frac{3}{2})^{m} \cdot 4 \cdot m!} \cdot (A.27)$$

For $\theta > 2.5$

$$\int_{0}^{\infty} \psi^{3}(\theta, \mathbf{x}) \, d\mathbf{x} \approx \frac{\pi}{8} \left\{ \frac{3}{2} \left[\frac{2}{3} \, \theta^{2} \, e^{\frac{2}{3} \theta^{2}} \mathbf{E}_{1}(\frac{2}{3} \, \theta^{2}) - \mathbf{h}_{1} + \mathbf{h}_{2} - \mathbf{h}_{3} + \mathbf{h}_{4} \right] \right\} \quad (A.28)$$

where

4

 E_1 is the exponential integral defined in Eq.A.37 and

$$h_{1} = 1 - \frac{2}{3} \theta^{2} e^{\frac{2}{3}\theta^{2}} E_{1}(\frac{2}{3} \theta^{2})$$

$$h_{2} = \frac{1}{\frac{2}{3} \theta^{2}} - h_{1}$$

$$h_{3} = \frac{5}{\theta^{4}} - \frac{10}{9} h_{2}$$

$$h_{4} = \frac{105}{4\theta^{6}} - \frac{7}{6} h_{3}$$

(A.29)

For x < 10, $E_1(x)$ is evaluated using the rational expression 5.1.54 of Ref. 33 and for $x \ge 10$, the rational expression 5.1.55 of Ref. 33.

For the case of β \geq 10^{10} , the code uses the first term in Eq.A.21 for J and sets I and M equal to 0.

V. Quadratures for Statistical Integration

In the calculation of unresolved resonance cross sections, integrals of the form

$$\langle f(x,y) \rangle = \int_{0}^{\infty} P_{\mu}(x) dx \int_{0}^{\infty} P_{\nu}(y) f(x,y) dy$$
 (A.30)

are required where P is the Chi-squared distribution of order μ given by

$$P_{\mu}(x) dx = \frac{\mu}{2} \frac{1}{\Gamma\left(\frac{\mu}{2}\right)} \left(\frac{\mu x}{2}\right)^{\frac{\mu}{2}} \exp\left(-\frac{\mu x}{2}\right).$$
(A.31)

Eq.A.30 is evaluated using the method described by Hwang⁽³⁵⁾ as

$$\langle f(x,y) \rangle = \sum_{j=1}^{10} \sum_{k=1}^{10} A_j A_k f(X_j, Y_k)$$
 (A.32)

where the A and X have been calculated for $\mu = 1,2,3$, and 4. The ten point quadrature is used for both neutron and fission width distributions.

For odd
$$\mu$$

 $X_j = 2Z_j^2/\mu$ (A.33)

$$A_{j} = 2W_{j}^{S} Z_{j}^{\mu-1} / \Gamma(\mu/2)$$
 (A.34)

while for even μ

$$X_{j} = (1-S_{j})/(1+S_{j})$$
 (A.35)

$$A_{j} = \mu W_{j}^{L} \left(\frac{\mu}{2} X_{j}\right)^{\mu/2-1} \exp\left(-\frac{\mu}{2} X_{j}\right) / [\Gamma(\mu/2)(1+S_{j})^{2}].$$
 (A.36)

In Eqs.A.33 and A.34, Z_j and W_j^S are respectively the ordinates and weights of the half-range Gauss-Hermite quadrature derived by Steen, et al. ⁽³⁶⁾ while in Eqs.A.35 and A.36, S_j and W_j^L are respectively the usual Gauss-Legendre ordinates and weights.

Table XIII lists A_j and X_j for
$$v = 1, 2, 3$$
, and 4 and $j = 1, 2, ..., 10$.

	One Degree	of Freedom	Two Degrees	of Freedom
	Weight	Abscissa	Weight	Abscissa
1	1.1120413E-01	3.0013465E-03	3.3773418E-02	1.3219203E-02
2	2.3546798E-01	7.8592886E-02	7.9932171E-02	7.2349624E-02
3	2.8440987E-01	4.3282415E-01	1.2835937E-01	1.9089473E-01
4	2.2419127E-01	1.3345267E+00	1.7652616E-01	3.9528842E-01
5	1.0967668E-01	3.0481846E+00	2.1347043E-01	7.4083443E-01
6	3.0493789E-02	5.8263198E+00	2.1154965E-01	1.3498293E+00
7	4.2930874E-03	9.9452656E+00	1.3365186E-01	2.5297983E+00
8	2.5827047E-04	1.5782128E+01	2.2630659E-02	5.2384894E+00
9	4.9031965E-06	2.3996824E+01	1.6313638E-05	1.3821772E+01
10	1.4079206E-08	3.6216208E+01	0.0	7.5647525E+01

Table	XIII.	Ten	Point Quadra	ature	Weights	and	Abscissae
		for	Statistical	Integ	gration		

Weight	Abscissa
3.3376214E-04	1.0004488E-03
1 050(100- 00	0 (107(007 00

1

Three Degrees of Freedom

2	1.8506108E-02	2.6197629E-02
3	1.2309946E-01	1.4427472E-01
4	2.9918923E-01	4.4484223E-01
5	3.3431475E-01	1.0160615E+00
6	1.7766657E-01	1.9421066E+00
7	4.2695894E-02	3.3150885E+00
8	4.0760575E-03	5.2607092E+00
9	1.1766115E-04	7.9989414E+00
10	5.0989546E-07	1.2072069E+01

Four Degree	es of Freedom
Weight	Abscissa
1.7623788E-03	1.3219203E-02
2.1517749E-02	7.2349624E-02
8.0979849E-02	1.9089473E-01
1.8797998E-01	3.9528842E-01
3.0156335E-01	7.4083443E-01
2.9616091E-01	1.3498293E+00
1.0775649E-01	2.5297983E+00
2.5171914E-03	5.2384894E+00
8.9630388E-10	1.3821772E+01

0.0

7.5647525E+01

-245-

VI. Numerical Methods Used in Calculating Slab Collision Rates

a. Exponential Integrals

The exponential integral $E_n(x)$ is defined as

$$E_{n}(x) = \int_{1}^{\infty} \frac{e^{-xt}}{t^{n}} dt \qquad (A.37)$$

The integrals $E_2(x)$, $E_3(x)$, and $E_4(x)$ are obtained using the following algorithms.

 $E_2(x)$ is evaluated using the series expansion (5.1.11) of the Handbook of Mathematical Functions with Formulas, Graphs, and Mathematical Tables ⁽³³⁾ for $E_1(s)$ along with the recursion relationship

$$E_{n+1}(x) = \frac{1}{n} [e^{-x} - x E_n(x)]$$
 (A.38)

to yield

$$E_2(x) \sim 1 - x (1 - \gamma - \ln x) - \frac{x^2}{1 \cdot 2!} + \frac{x^3}{2 \cdot 3!} - \frac{x^4}{3 \cdot 4!}$$
 (A.39)

where Euler's constant γ has the value $\gamma = 0.5772156649 \ldots$

For x < 0.6, $E_3(x)$ as in the case of $E_2(x)$ uses (5.1.11) of Ref.33 with successive application of Eq.A.38 to yield

$$E_{3}(x) \sqrt[\infty]{0.5} - x + \frac{x^{2}}{2} (1.5 - \gamma - \ln x) + \frac{x^{3}}{3!} - \frac{x^{4}}{2 \cdot 4!} + \frac{x^{5}}{3 \cdot 5!} - \frac{x^{6}}{4 \cdot 6!}$$
(A.40)

For $0.6 \le x \le 6.4$, $E_3(x)$ is obtained by linear interpolation in a table of E_3 with tabular entries 0(0.01)2(0.02)4(0.08)6.4. The table values are calculated using the polynomial approximation (5.1.53) of Ref. 33 if $0 \le x \le 1$ and the rational approximation (5.1.56) of Ref. 33 if x > 1 to obtain $E_1(x)$, and successive application of Eq.A.38.

For 6.4 < x \leq 80, E₃(x) is evaluated using the first three terms of the series representation of E_n(x) for large n (5.1.52) of Ref. 33 so that

$$E_3(x) \sim \frac{e^{-x}}{x+3} \left[1 + \frac{3}{(x+3)^2} + \frac{3(3-2x)}{(x+3)^4} \right]$$
 (A.41)

Beyond x = 80, E_3 is assumed to be zero.

For $0 \le x \le 6.4$, $E_4(x)$ is obtained by linear interpolation in a Table of E_4 with table entries the same as for the E_3 table mentioned above. The table values were obtained as in the case of E_3 with one more application of the recursion relation Eq.A.38.

For 6.4 < x \leq 80, E $_4$ is evaluated as in the case of E $_3$ using (5.1.52) of Ref. 33 so that

$$E_{4}(x) \sim \frac{e^{-x}}{x+4} \left[1 + \frac{4}{(x+4)^{2}} + \frac{4(4-2x)}{(x+4)^{4}} \right]$$
(A.42)

Beyond x = 80, E_{4} is assumed to be zero.

b. Slab Collision Escape Probability

The probability of escape without a collision from a slab with optical thickness x is given by

$$P = \frac{0.5 - E_3(x)}{x}$$
 (A.43)

If $x \ge 0.6$, P is evaluated with $E_3(x)$ being obtained as described earlier using table interpolation or using Eq.A.41 as appropriate to the range of x.

For x < 0.6, use of Eq.A.40 in Eq.A.43 yields

$$P \sim 1 - \frac{x}{2} (1.5 - \gamma - \ln x) - \frac{x^2}{3!} + \frac{x^3}{2 \cdot 4!} - \frac{x^4}{3 \cdot 5!} + \frac{x^5}{4 \cdot 6!} - \frac{x^6}{5 \cdot 7!}$$
 (A.44)

c. Infinite Sums of Exponential Integrals

In the case of a unit cell of optical thickness h, the collision rate calculation involves evaluation of infinite sums of exponential integrals. Numerical procedures have been developed $^{(6)}$ for evaluating the functions

$$S_{n}(z,h) = \sum_{k=0}^{\infty} E_{n}(z+kh) = \int_{1}^{\infty} \frac{e^{-zt}}{(1-e^{-ht})t^{n}} dt$$
 (A.45)

which make use of fast and accurate Gaussian quadrature formulas.

The code evaluates $S_3(z,h)$ using

$$S_{3}(z,h) \sim \sum_{i=1}^{M} \left[\frac{w_{i,3} \exp(t_{i,3})}{1 - \exp(-ht_{i,3})} \right] \exp(-zt_{i,3}) + C(z,h)$$
 (A.46)

where

$$C(z,h) = \{0.004126 + 0.00628 \exp(-19.8h)\} \exp(-18.2z), z < 0.3$$

= 0, z > 0.3 (A.47)

-247-

and the abscissae $t_{i,3}$ and weights $w_{i,3}$ are precalculated as described in Ref. 37 data and stored as data in the code.

The number of terms M used in the sum in Eq.A.46 lay between 1 and 4 depending upon the value of z. If

$$z \leq 3.16$$
, M is determined by truncating to an integer
M = 0.633 (7.89 - z). (A.48)

For 3.16 < z < 13.7, M is determined from

$$M = 1.0 + 6.2/z.$$
 (A.49)

For z > 13.7, $S_{3}(z,H)$ is assumed to be zero.

 $S_{\lambda}(z,h)$ is evaluated as

$$S_{4}(z,h) \approx \sum_{i=1}^{N} \left[\frac{w_{i,4} \exp(t_{i,4})}{1 - \exp(-ht_{i,4})} \right] \exp(-zt_{i,4})$$
 (A.50)

with the abscissae and weights again precalculated and stored.

N lay between 1 and 4 depending upon z. If $z \leq 3.64$, M is determined by truncating to an integer

$$N = 0.549 (9.10 - z) . \tag{A.51}$$

For 3.64 < N \leq 15.6, M is determined from

$$N = 1.0 + 6.81/z.$$
 (A.52)

For z > 15.6, $S_{L}(z,H)$ is assumed to be zero.

In addition to the infinite sums $S_n(z,h)$, the code requires differences of these sums

$$D_n(y,\Delta,h) = S_n(y, h) - S_n(y + \Delta, h)$$
 (A.53)

The $D_n(y,\Delta,h)$ are evaluated using various algorithms depending upon the cell optical thickness and the incremental length Δ . In particular, the method of Gaussian quadrature for the evaluation of the S described above is least accurate for either z or h small.

For $\Delta > 0.02$ and $h \le 0.2$, if $z > 1.147h^{1.4}$, the S₃(z,h) of Eq.A.53 are evaluated using the Euler-Maclaurin series

$$S_{3}(z,h) \sim \frac{1}{12} \left\{ e^{-z} \left[\frac{4 - 2z}{h} + 3 - \frac{h^{3}}{60z} \left\langle 1 - \frac{h^{2}}{21z^{2}} \left(1 + z + \frac{z^{2}}{2} \right) \right\rangle \right] + E_{2}(z) \left[h - 3z + \frac{2z^{2}}{h} \right] \right\}.$$
 (A.54)

If
$$z \le 1.147h^{1.4}$$
 but $z + h > 1.147h^{1.4}$,
 $S_3(z,h) = E_3(z) + S_3(z + h, h)$ (A.55)

where $S_3(z + h, h)$ is evaluated using Eq.A.54.

If
$$z + h \le 1.147h^{1.4}$$
 but $z + 2h > 1.147h^{1.4}$,
 $S_3(z,h) = E_3(z) + E_3(z + h) + S_3(z + 2h, h)$ (A.56)

where again $S_3(z + 2h, h)$ uses Eq.A.54. In Eqs.A.54-A.56, the $E_2(z)$ and $E_3(z)$ are obtained using the methods described above for obtaining the exponential integrals.

If $\Delta > 0.02$ and h > 0.2, the S₃(z, h) of Eqs.A.53 are evaluated using Gaussian quadrature as described earlier in the discussion relating to Eq.A.46 with the number of terms used in the quadrature determined using Eqs.A.48 and A.49.

If Δ is \leq 0.02, a Taylor series expansion is used to obtain

$$D_{3}(y,\Delta,h) = S_{3}(y,h) - S_{3}(y+\Delta,h) = S_{3}(z-\frac{\Delta}{2},h) - S_{3}(z+\frac{\Delta}{2},h)$$

$$\sim_{0}^{2} \Delta \cdot \left[S_{2}(y+\Delta/2,h) + \frac{\Delta^{2}}{24} S_{0}(y+\Delta/2,h) + \frac{\Delta^{4}}{1920} S_{-2}(y+\Delta/2,h) \right].$$

(A, 57)

For y + $\Delta/2 < 0.6$, only the S₂ and the S₀ are retained and the first few terms of the series in Eq.A.57 are evaluated by directly summing the E₂ and E₀ functions as

$$S_{1} = \Delta \cdot \left\{ E_{2}(y + \Delta/2) + E_{2}(y + \Delta/2 + h) + E_{2}(y + \Delta/2 + 2h) + \dots + \frac{\Delta^{2}}{24} \left[E_{0}(y + \Delta/2) + E_{0}(y + \Delta/2 + h) + E_{2}(y + \Delta/2 + 2h) + \dots \right] \right\}$$
(A.58)

where the direct summation in Eq.A.58 is continued as long as $y + \Delta/2 + kh < 0.6$, and $E_{o}(z) = \exp(-z)/z$.

When $y + \Delta/2 + kh \ge 0.6$, the remainder of the series of Eq.A.57 is obtained as follows. S₂ is obtained using the Gaussian quadrature algorithm

$$S_{2}(y + \Delta/2 + kh, h) = \sum_{i=1}^{4} \left[\frac{w_{i,2} \exp(t_{i,2})}{1 - \exp(-ht_{i,2})} \right] \exp \left\{ -(y + \Delta/2 + kh)t_{i,2} \right\}$$

(A.59)

using precalculated and stored weights and abscissae. Then, using the fact that

$$\frac{\partial E_n(z)}{\partial z} = -E_{n-1}(z)$$
(A.60)

we may rewrite Eq. A.57 as

$$S_2 = \Delta \cdot S_2(y + \Delta/2 + kh, h) \left\{ 1 + \frac{\Delta^2}{24} t_{1,2}^2 + \frac{\Delta^4}{1920} t_{1,2}^4 \right\}$$
 (A.61)

١

with Eq. 23 used to obtain S_2 . Thus finally

$$D_3(y, \Delta, h) = S_1 + S_2$$
 (A.62)

The $D_4(y, \Delta, h)$ are similarly evaluated using various algorithms depending upon the cell optical thickness h and the incremental length Δ .

For $\Delta > 0.02$ and $h \le 0.2$, if $z > 1.818h^{2.5}$, the S₄(z,h) of Eq.A.53 are evaluated using the Euler-Maclaurin series

$$S_{4}(z,h) \sim \frac{1}{12} \left\{ e^{-z} \left[\frac{3-z}{h} + 2 + \frac{h^{3}}{60z^{2}} (1-z) \right] + E_{3}(z) \left[\frac{z^{2}}{h} - h - 2z - \frac{h^{3}}{30z^{2}} \right] \right\} . (A.63)$$

If $z \le 1.818h^{2.5}$ but $z + h > 1.818h^{2.5}$,

$$S_4(z,h) = E_4(z) + S_4(z+h, h)$$
 (A.64)

where $S_4(z + h, h)$ is evaluated using Eq.A.63. In Eqs.A.63 and A.64, the $E_3(z)$ and $E_4(z)$ are obtained using the methods described earlier for obtaining the exponential integrals.

If $\Delta > 0.02$ and h > 0.2, the S₄(z,h) of Eq.A.53 are evaluated using Gaussian quadrature as described earlier in the discussion relating to Eq.A.50 with the number of terms used in the quadrature determined using Eqs.A.51 and A.52.

If Δ is \leq 0.02, a Taylor series expansion is used as in the case of $D_3(y,\Delta,h)$ to obtain

$$D_{4}(y,\Delta,h) = S_{4}(y,h) - S_{4}(y+\Delta,h)$$

$$\approx \Delta \cdot S_{3}(y+\Delta/2,h) \left\{ 1 + \frac{\Delta^{2}}{24} t_{1,3}^{2} \right\}$$
(A.65)

with S_3 being obtained using the Gaussian quadrature of Eq.A.46.

VII. Cylindrical Transmission Probabilities

As given in Ref. 5, for a cylindrical region i having outer radius r_i , inner radius r_{i-1} , and macroscopic cross section Σ_i , the transmission probabilities from inner surface to outer surface T_i^{OI} and from outer surface to outer surface T_i^{OO} are given by

$$T_{i}^{OI} = \frac{4}{\pi} \int_{0}^{\pi/2} d\phi \cos \phi K_{i3} \left\{ \Sigma_{i} f_{i}(\phi) \right\}$$
(A.66)

$$\mathbf{T}_{i}^{00} = \frac{4}{\pi} \int_{0}^{\pi/2} d\phi \cos \phi \left[\mathbf{K}_{i3} \left\{ \Sigma_{i} \mathbf{g}_{i}(\phi) \right\} - \frac{\mathbf{r}_{i-1}}{\mathbf{r}_{i}} \mathbf{K}_{i3} \left\{ \Sigma_{i} \mathbf{h}_{i}(\phi) \right\} \right]$$
(A.67)

where

$$K_{i_3}(\xi) = \int_0^{\pi/2} e^{-\xi \csc \theta} \sin^2 \theta \, d\theta \qquad (A.68)$$

$$f_{i}(\phi) = -r_{i-1} \cos \phi + \left\{ r_{i}^{2} - r_{i-1}^{2} \sin^{2} \phi \right\}^{\frac{1}{2}}$$
(A.69)

$$g_{i}(\phi) = 2r_{i} \cos \phi \qquad (A.70)$$

$$h_{i}(\phi) = 2 \left\{ r_{i}^{2} - r_{i-1}^{2} \sin^{2} \phi \right\}^{\frac{1}{2}}$$
 (A.71)

The T^{OI} and T^{OO} are tabulated as a function of x and z where

$$x_{i} = \frac{r_{i-1}}{r_{i}} \tag{A.72}$$

$$z_{i} = \Sigma_{i} (r_{i} - r_{i-1})$$
 (A.73)

with x ranging between 0.0 and 1.0 and z between 0.0 and 8.0. For both T^{OI} and T^{OO} , the mesh for z is 0(0.01)0.4(0.02)1.0(0.04)2.6(0.06)5.0(0.10)8.0. For T^{OI} , the x mesh increment is 0.04, and for T^{OO} , the x mesh increment is 0.02.

The K used in Eqs. A.66 and A.67 were obtained from an extended precision i_3 (real*16) function KIN which computes K (x) for $1 \le N \le 10$ and $x \ge 0$ using the following algorithms.

For x > 39.0, KIN returns the value 0.0 for all N. For $0.0 < x \le 39.0$, K and K are evaluated using the rational expressions given by i_2 i_3 I. Gargantine and T. Pomentale⁽³⁸⁾. For $0.0 < x \leq 2.0$, $K_i(x) = K_o(x)$ is obtained from the polynomial approximations (9.8.5) and (9.8.1) of the Handbook of Mathematical Functions with Formulas, Graphs, and Mathematical Tables⁽³³⁾ and for x > 2.0, K_i is obtained from the polynomial approximation (9.8.6) of Ref.33, o

K_i is obtained for 0.0 < x < 7.0 in terms of K_i and K_i as $K_{i_1}(x) = K_{i_{-1}}(x) - \frac{1}{x} K_{i_2}(x)$ (A.74)

where K is obtained as indicated above and K $(x) = K_1(x)$ is obtained using the polynomial approximations (9.8.7) and (9.8.3) of Ref.33 when $0.0 < x \leq 2.0$, and using the polynomial approximation (9.8.8) of Ref.33 when x > 2.0. For x > 7.0, K is obtained from the polynomial approximation (11.1.18) of Ref.33.

For n > 3, the recursion relationship

$$n K_{i_{n+1}}(x) = (n - 1) K_{i_{n-1}}(x) + x \left[K_{i_{n-2}}(x) - K_{i_{n}}(x) \right]$$
(A.75)

is used.

For x = 0.0, KIN uses stored precalculated data to return

$$K_{i_n}(0) = \frac{(\pi)^{\frac{1}{2}}}{2} \frac{\Gamma\left[\frac{n}{2}\right]}{\Gamma\left[\frac{n}{2} + \frac{1}{2}\right]}$$
(A.76)

where

$$\Gamma(\frac{1}{2}) = (\pi)^{\frac{1}{2}}$$

$$\Gamma(\frac{3}{2}) = \frac{1}{2} (\pi)^{\frac{1}{2}}$$

$$\Gamma(n + \frac{1}{2}) = \frac{1 \cdot 3 \cdot 5 \cdot 7 \cdot \cdots (2n - 1)}{2^{n}} (\pi)^{\frac{1}{2}}$$

$$\Gamma(n + 1) = n!$$

$$(-\frac{1}{2})! = (\pi)^{\frac{1}{2}}$$

$$(\frac{1}{2})! = \frac{1}{2} (\pi)^{\frac{1}{2}}$$

The extended precision (real*16) numerical integration of Eqs.A.66 and A.67 was accomplished using an adaptive Simpson's rule quadrature with convergence criteria adjusted for different ranges of x and z to achieve at least six significant figure agreement with very small convergence criterion results.

For a given x and z, T^{OI} and T^{OO} are obtained using the six point bivariate interpolation formula (25.2.67) of Ref.33 within the body of the table and the four point formula (25.2.66) of Ref.33 in the first x or z intervals.

Outside the range of the table T^{OI} is set to zero and an approximate polynomial fit similar to that used in the RABBLE code ⁽⁵⁾ is used for T^{OO} .

APPENDIX B

MC²-2 BCD INPUT FILES

C *** *	* * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *
С		-
C.		PREPARED 3/8//6 AT ANL -
C F	λ Μ.	
CF-	A. H.	TRAI BOD INDUT FOR MC**2-II AND SDY CALCULATIONS -
C	01.0	-
CN		THIS IS A USER SUPPLIED BCD DATA SET
CN`		THE LIST FOR EACH RECORD IS GIVEN IN TERMS -
CN		OF THE BCD FORMAT OF THAT DATA CARD
CN		COLUMNS 1-2 CONTAIN THE CARD TYPE NUMBER
CN		BLANK FIELDS PRODUCE THE DEFAULT OFTIONS.
C		-
C****	* * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *
C		
CR	PRO	BLEM TITLE (TYPE 01) -
C		
CL	FORMAT	(I2,4X,11A6) -
С		-
CD	COLUMNS	CONTENTSIMPLICATIONS, IF ANY -
CD	=======	
CD	1-2	- 01
CD	7-72	
CD C	$I = I \mathbf{Z}_{i}$	ANT ALFHANOMENIC CHARACIENS.
CN		ANY NUMBER OF TYPE O1 CARDS MAY BE USED
C		-
C		
_		
C	CON	
CR C:	CON	
CT.	FORMAT	(12.4X.316)
C		
CD	COLUMNS	CONTENTS IMPLICATIONS, IF ANY -
CD	======	
CD	1-2	- 02
CD		`
CD	7-12 ,	SIZE OF MAIN CORE CONTAINER ARRAY IN 8 BYTE WORDS, -
CD		TYPICALLY 30000 (DEFAULT=30000)
CD		FOR CDC USERS TYPICALLY 20000 (DEFAULT=20000)
CD	12 10	
	13-15	
		FOR COC HEERS TVDICALLY 100000 (DEFAULT-100000) -
CD	,	MAXIMUM ALLONABLE=131072.
CD		

.

CD	19-24	POINTR DEBUGGING EDIT. TYPICALLY 0
CD		0 NO DEBUCCING PRINTOUT (DEENULT) -
CD		
CD		2DEBUGGING TRACE PRINTOUT
CD		3FULL DEBUGGING PRINTOUT
C		
CN		
		NO TIPE OZ CARD WILL GIVE THE FOLLOWING DEFROET
CN		VALUES FOR IBM USERS.
CN		MAIN CORE CONTAINER=30000
CN		BULK CORE CONTAINER=0 -
CN		PRINT OPTION FIAGEO -
CN		
		DECIDE DARAGES OF THE TIPE OF CARD, THE
CN		
CN		BBGION=700K -
CN		-
CN		NO TYPE 02 CARD WILL GIVE THE FOLLOWING DEFAULT -
CN		VALUES FOR CDC USERS.
CN		MAIN CORE CONTAINER=20000
CN		BULK CORE CONTAINER-100000
CN		
		PRINT OPTION FLAG-0
CN		FOR THE DEFAULT VALUES ON THE TYPE OZ CARD, THE -
CN		REDEFINE FIELD LENGTH CONTROL CARD SHOULD SPECIFY
CN		RFL (160000, L=400) -
С		
C		
		·
C		
C	C D Y	
CR	Gen	TARAL PROBLEM SPECIFICATIONS (TIPE 03)
C		s
CL	FORMAT	(T2,4X,4I6,A6,6I6) -
С		=
CD	COLUMNS	CONTENTS IMPLICATIONS, IF ANY -
CD	_ = = = = = = = =	
CD	1-2	<u> </u>
CD		
CD CD	7 10	
	7-12	- JNDAMENIAL HODE TIPE.
CD		-
CD		2B1. –
CD		3CONSISTENT P1. –
CD		4CONSISTENT B1
CD		_
CD	13-19	
	10-10	ONDER OF EXTENDED TRANSFORT REPROACHAILON (DETROBLE).
	10 0 "	
CD	19-24	CONTINUOUS SLOWING DOWN MODERATING PARAMETER OPTION
CD		9IMPROVED GREULING GOERTZEL (DEFAULT)
CD		1GREULING GOERTZEL
CD		-
D	25-30	MAXIMUM NUMBER OF HYPERFINE GROUPS PER ULTRAFINE GROUP -
- ח'		USED IN EVALUATION OF ELASTIC SCATTERING MATRICES -
· • • •		tent to be accorded to the second of the sec

•

CD (DEFAULT=1). CD 31-36 BROAD GROUP STRUCTURE TYPE. CD ANL9...ANL 9 GROUP STRUCTURE. CD ANL11...ANL 11 GROUP STRUCTURE. CD ANL27...ANL 27 GROUP STRUCTURE. CD ANL28...ANL 28 GROUP STRUCTURE. CD BOND26...BONDARENKO 26 GROUP STRUCTURE. CD FFTF30...FFTF 30 GROUP STRUCTURE. CD HANS 16... HANSEN-ROACH 16 GROUP STRUCTURE. CD > SDX156...SDX 156 GROUP STRUCTURE. CD USS212...UNIVERSAL SUPERGROUP 212 GROUP STRUCTURE. CD -USS226...UNIVERSAL SUPERGROUP 226 GROUP STRUCTURE. CD MARD9...WARD 9 GROUP STRUCTURE. CD CD FUEL GEOMETRY. 37-42 CD CD 0...HOMOGENEOUS. 1...SLAB. CD 2...CYLINDER. CD CD RESOLVED RESONANCE OVERLAP OPTION. CD 43-48 O...INCLUDE OVERLAP CALCULATION FOR THE NUMBER OF CD NEIGHBORING RESONANCES DESIGNATED IN COLS. 55-60 CD -(DEFAULT). CD 1...OMIT OVERLAP CALCULATION. CD CD 49-54 UNRESOLVED RESONANCE SELF-OVERLAP OPTION. CD CD 0...INCLUDE SELF-OVERLAP CALCULATION (DEFAULT). 1...OMIT SELF-OVERLAP CALCULATION. CD CD CD 55-60 NUMBER OF RESOLVED RESONANCES TO BE TESTED FOR OVERLAP -ON EACH SIDE OF EACH RESOLVED RESONANCE (DEEAULT=4). _ CD THIS FIELD IS IGNORED IF COLS. 43-48 ARE 1. CD CD 61-66 RESERVED FOR FUTURE USE CD CD 67-72 INELASTIC AND (N,2N) ULTRA-FINE-GROUP TREATMENT. CD 0... APPROXIMATE TREATMENT (DEFAULT). CD CD 1... RIGOROUS ENERGY-ANGLE COPRELATION ACCOUNTED FOR. С IF CARD TYPE 03 IS NOT SUPPLIED, ALL DEFAULT OPTIONS -CN WILL BE UTILIZED. CN IF COLS. 37-42 ARE NON-ZERO, CARD TYPES 04, 06, 14, AND-CN 15 OF DATA SET A.NIP MUST BE SUPPLIED TO SPECIFY THE -CNHETEROGENEOUS PROBLEM GEOMETRY AND COMPOSITIONS. CN IF COLS. 7-12 ARE 1 OR 2, COLS. 13-18 MUST BE 1. CN COLS. 31-36 ARE IGNORED IF ANY TYPE 05 OR 07 CAPDS CN -ARE SUPPLIED OR IF THE DATA SET XS.ISO IS DECLARED TO CN CN BE OLD IN THE EXECUTION DECK. С

-256-

GENERAL PROBLEM CONSTANTS (TYPE 04) CR С CL $FORMAT - - - - (12, 10 \times 5 \times 12.5)$ С CD COLUMNS CONTENTS...IMPLICATIONS, IF ANY _____ CD ======= 1-2 CD 04 CD CONTINUOUS SLOWING DOWN INTEGRATION FACTOR THETA CD 13-24 CD (DEFAULT=0.5). CD CD 25-36 MASS OF MATERIAL USED IN DETERMINATION OF THE DATA CD MANAGEMENT STRATEGY FOR THE SPECTRUM CALCULATION. COLS 25-36 SHOULD BE LESS THAN OR EQUAL TO THE MASS OF -CD CD THE LIGHTEST MATERIAL IN THE PROBLEM (EXCLUDING H). IF COLS. 25-36 ARE BLANK, THE MANAGEMENT STRATEGY IS CD SELECTED BY THE BUILT IN ALGORITHM. IF COLS. 25-36 CD -CONTAIN A VALUE LARGER THAN THE LIGHTEST MASS MATERIAL CD -IN THE PROBLEM, THE LATTER WILL BE USED INSTEAD. CD CD CD 37-48 BOUNDARY ENERGY IN EV. BETWEEN THE MULTIGROUP FLUX SOLUTION AND THE CONTINUOUS SLOWING DOWN FLUX CD CD SOLUTION. THE BOUNDARY ENERGY MUST BE GREATER THAN. THE ENERGY OF THE HIGHEST RESOLVED RESONANCE IN THE CD CD PROBLEM. THE DEFAULT VALUES WILL SATISFY THIS CÐ REQUIREMENT. CD 49-60 CONSTANT A1 USED IN THE EQUIVALENCE PRINCIPLE. CD CD 61-72 CONSTANT A2 USED IN THE EQUIVALENCE PRINCIPLE. CD С CN IF COLS. 49-60 ARE BLANK, COLS. 49-60 AND COLS. 61-72 ARE SET EQUAL TO 1.35 IF COLS 37-42 ON. CΝ CARD TYPE 03 CONTAIN A 2 AND THEY ARE SET EOUAL TO CN 1.09 IF COLS. 37-42 ON CARD TYPE 03 CONTAIN A 1. CΝ IF COLS. 49-60 ARE NON-BLANK AND COLS. 61-72 ARE CN BLANK, COLS. 61-72 ARE SET EQUAL TO COLS. 49-60. CN IF NO CARD TYPE 04 IS SUPPLIED, ALL DEFAULT OPTIONS CN WILL BE UTILIZED. CN С _____ _____ BROAD GROUP ENERGIES (TYPE 05) R

-257-

.

.

C----

.

•

CL	FORMAT	(I2, 10X, 3 (I6, E12.5)) -
C D	COLUMNS	CONTENTSIMPLICATIONS, IF ANY -
CD CD	1-2	05
CD	13-18	BROAD GROUP NUMBER
CD	19-30	UPPER ENERGY OF GROUP (EV)
CD CD	31-36	BROAD GROUP NUMBER
CD CD	37-48	UPPER ENERGY OF GROUP (EV).
CD CD	49-54	BROAD GROUP NUMBER
C D C	55-66	UPPER ENERGY OF GROUP (EV)
CN		THE BROAD GROUP STRUCTURE MAY BE SPECIFIED USING TYPE -
CN		05 CARDS OR TYPE 07 CARDS OR A COMBINATION OF EACH
CN '		ALTERNATIVELY, ONE OF THE PRESTORED GROUP STRUCTURES -
CN		SPECIFIED IN COLS. 31-36 ON THE TYPE 03 CARD OR THE -
CN		BUILT IN DEFAULT DESCRIBED BELOW MAY BE USED
CN		GROUP 1 IS THE GROUP OF HIGHEST ENERGY. THE THERMAL -
CN		GROUP HAVING THE LARGEST GROUP NUMBER HAS AS UPPER -
CN		ENERGY THE INTERFACE BETWEEN THE THERMAL GROUP AND -
CN		EPI-THERMAL GROUPS. AS MANY TYPE 05 CARDS ARE USED AS -
CN		ARE NECESSARY TO SPECIFY THE ENERGIES. IF COLS. 31-36 -
CN		OR 49-54 ARE BLANK, THE REST OF THE DATA ON THE TYPE 05-
CN		CARDS ARE IGNORED
CN		IF ONLY THE BROAD GROUP NUMBER ONE UPPER ENERGY IS -
CN	•	SPECIFIED AND IF NO TYPE 07 CARDS ARE SUPPLIED, A FIXED-
CN`		0.5 LETHARGY WIDTH BROAD GROUP STRUCTURE IS CONSTRUCTED-
CN		EXTENDING DOWN FROM THE SPECIFIED ENERGY TO THE LOWEST -
CN		ENERGY IN THE LIBRARY. IF TYPE 07 CARDS ARE SUPPLIED, -
CN		THE TYPE 07 DATA ARE USED WITH THE SPECIFIED UPPER -
CN		ENERGY TO CONSTRUCT THE BROAD GROUP STRUCTURE
CN		IF MORE THAN ONE BROAD GROUP ENERGY BOUNDARY IS -
CN		SPECIFIED ON THE TYPE 05 CARDS, ANY TYPE 07 CARDS -
CN		SUPPLIED WILL BE IGNORED
CN		IF COLS. 31-36 ON CARD TYPE 03 ARE BLANK AND -
CN		IF NO TYPE 05 CARDS ARE SUPPLIED AND NO TYPE 07 CARDS -
CN		ARE SUPPLIED, A FIXED 0.5 LETHARGY WIDTH BROAD GROUP -
CN		STRUCTURE IS CONSTRUCTED COVERING THE ENERGY RANGE -
CN		OF THE LIBRARY
С		-
C		
	4	

-258--

.

۰.

CR	HOM	OGENEOUS COMPOSITION SPECIFICATIONS (TYPE 06) -
CL	FOPMAT	(12,10X,2A6,3E12.5,16) -
CD	COLUMNS	CONTENTSIMPLICATIONS, IF ANY -
CD	1-2	06
CD	13-18	NUCLIDE IDENTIFICATION LABEL ON LIBRARY.
	19-24	PROBLEM MATERIAL LABEL. THIS LABEL CAN BE ANY ALIAS NAME. IF COLS. 19-24 ARE BLANK, THE MATERIAL NAMED IN COLS. 13-18 WILL BE USED.
	25-36	MATERIAL ATOMIC CONCENTRATION USED TO COMPUTE - HOMOGENEOUS CROSS SECTIONS FOR USE IN THE - SPECTRUM CALCULATION (ATOMS/CC*1.E-24)
	37-48	MATERIAL TEMPERATURE IN DEGREES K (DEFAULT=300.)
	49-50	EPSHET, USED TO DETERMINE NEAREST NEIGHBORING PLATE - CONTAINING MATERIAL NAMED IN COLUMNS 13-18.IF A PLATE K- CONTAINS MATERIAL I, THEN NEIGHBORING PLATES ARE - SEARCHED TO FIND THE NEAREST PLATE M WHICH SATISFIES - THE CONDITION N(I,M)*DX(M).GF.EPSHET*N(I,K)*DX(K), - WHERE N(T,M) IS THE ATOM DENSITY OF MATERIAL T IN - PLATE M AND DX(M) IS THE THICKNESS OF PLATE M. (DEFAULT VALUE=0.1).
CD CD CD CD CD CD CD C	61-66	OUTPUT CROSS SECTION DATA SET FLAG. IF COLS. 61-66 ARE - BLANK, THE MATERIAL NAMED IN COLS. 13-18 WILL BE ADDED - TO THE OUTPUT CROSS SECTION DATA SET. IF COLS. 61-66 - ARE NON-BLANK, THE BROAD GROUP CROSS SECTIONS FOR THE - MATERIAL NAMED WILL NOT BE CALCULTED AND HENCE NOT BE - ADDED TO THE OUTPUT DATA SET
CN CN CN CN CN CN CN CN CN CN CN	、	ONE TYPE 06 CARD MUST BE GIVEN FOR FACH MATERIAI - PRESENT IN THE MACROSCOPIC MIXTURE OF THE SPECTRUM - COMPOSITION IF COLS. 37-42 ON CARD TYPE 03 ARE ZERO - OR IF CARD TYPE 03 IS NOT SUPPLIED. IF NO TYPE 06 CARD - IS GIVEN, THE MACROSCOPIC MIXTURE IS DEPIVED FROM - THE DATA ON THE A.NIP CARD TYPES 06, 14, AND 15. IN - THIS CASE, ALL MATERIALS WILL BE ASSUMED TO HAVE A - TEMPERATURE OF 300 DEGREES K AND LIBRARY NUCLIDE - IDENTIFICATION LABELS WILL BE USED FOR THE PROBLEM - MATERIAL LABELS. IF COLS. 25-36 ARE BLANK OR ZERO FOR - ANY MATERIAL THE ATOMIC DENSITY FOR THAT MATERIAL - IS DERIVED FROM THE A.NIP DATA FOP THE REGIONS - ALTHOUGH MATERIAL LABELS AND TEMPERATURES FILL BE - SET BY THE TYPE 06 DATA. ANY ATOMIC DENSITY WHICH IS -

-259-

CN		GIVEN AS LESS THAN 1. E-20 WILL BE SET TO 1. E-20.	-
С			-
C			-
C			
C====	5 PT-		_
C	1/6 1	THERMAL DROAD GROOF EDIMARGI WIDING (THE OV)	-
CL	FORMAT	(12,10X,2(E12.5,16,16))	_
С			-
CD	ÇOLUMNS	CONTENTSIMPLICATIONS, IF ANY	-
CD	======		-
CD	1-2	· · · · · · · · · · · · · · · · · · ·	-
CD	10.04		-
CD	- 13-24	LETHARGY WIDTH.	-
CD	25 20	THIMTHI DDOND CDOUD NUMDED	-
CD	25-50	INALIAL BROAD GROUP NUMBER.	_
CD	31-36	FINAL BROAD GROUP NUMBER.	_
CD	31 30	A A A A A A A A A A A A A A A A A A A	-
CD	37-48	LETHARGY WIDTH.	-
CD			-
CD	49-54	INITIAL BROAD GROUP NUMBER.	-
CD		0	~
CD	55-60	FINAL BROAD GROUP NUMBER.	~
С			-
CN		SEE NOTES FOR THE TYPE US CARDS.	-
CN		DATA SHOULD BE SUPPLIED FOR UNLY THE EPI-THERMAN.	-
		FOR THE THERMAL CROUD	_
CN		AS MANY TYPE OF CAEDS AS NECESSARY MAY BE USED.	_
CN		AS A MAXIMUM. THERE MAY BE AS MANY BROAD EPT-THERMAL	-
CN		GROUPS AS THERE ARE ULTRA FINE GROUPS IN THE LIBRARY.	-
CN		THE LETHARGY WIDTH SPECIFIED IN COLS. 13-24 IS USED	-
CN		FOR BROAD GROUPS SPECIFIED IN COLS. 25-30 THROUGH	-
CN		31-36. THE LETHARGY WIDTH SPECIFIED IN COLS. 37-48 IS	
CN	• •	USED FOR BROAD GROUPS SPECIFIED IN COLS. 49-54 THROUGH	-
CN		55-60. IF COLS. 25-30 OR 49-54 ARE BLANK, THE LETHARGY	-
CN		WIDTH SPECIFIED IN COLS. 13-24 OF 37-48 IS USED FOR ALL'	_
		TEBARY FREDCY STORED DOWN TO THE THE DULLON OF THE T	_
CN		THE TYPE O7 CARDS ARE IGNORED. IF NO TYPE 05 CARDS ARE	-
CN		SUPPLIED. THE BROAD GROUPS BEGIN AT THE TOP OF THE	
CN		LIBRARY. IF AN UPPER ENERGY IS SPECIFIED ON A	
CN		TYPE 05 CARD, THE BROAD GROUPS WILL EXTEND DOWN FROM	-
CN		THAT ENERGY. IF COLS. 31-36 AND/OR COLS. 55-60 ARE	-
CN		BLANK, THE VALUE SPECIFIED IN COLS. 13-24 AND/OR 37-48	-
CN		IS USED FOR THE BROAD GROUPS SPECIFIED IN COLS. 25-30	-
CN		AND/OR COLS. 49-54. LETHARGY WIDTHS SPECIFIED IN COLS.	
CN		13-24 AND 37-48 ARE ADJUSTED TO CORRESPOND TO A	

-260-

.

C N C		MULTIPLE OF THE LIBRARY GROUF LETHARGY WIDTH.	-
C			-
C CR	EXI	TEPNAL SOURCE SPECIFICATIONS (TYPE 08)	-
C CL C	FORMAT	(I2,10X,2(E12.5,2I6)) OR (I2,10X,E12.5,A6)	-
CD	COLUMNS	CONTENTSIMPLICATIONS, IF ANY	-
CD	1-2	08	_
` CD	13-24	MAGNITUDE OF EXTERNAL SOURCE.	-
CD CD CD	25-30	HIGHER ENERGY ULTRAFINE GROUP NUMBER TO WHICH SOURCE IN COLS. 13-24 APPLIES.	- -
CD CD CD	31-36	LOWER ENERGY ULTRAFINE GROUP NUMBER TO WHICH SOURCE IN COLS. 13-24 APPLIES.	-
CD CD CD	37-48	MAGNITUDE OF EXTERNAL SOURCE.	-
CD CD CD	49-54	HIGHER ENERGY ULTRAFINE GROUP NUMBER TO WHICH SOURCE IN COLS. 37-48 APPLIES.	-
CD CD CD	55-60	LOWER ENERGY ULTRAFINE GROUP NUMBER TO WHICH SOURCE IN COLS. 37-48 APPLIES.	-
CN CN CN CN CN CN CN		IF COLS. 13-24 OF THE FIRST TYPE 08 CARD ARE BLANK OR ZERO THEN COLS. 25-30 OF THIS CARD MUST CONTAIN THE FISSION SPECTRUM NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY (A6 FORMAT) SPECIFYING THE EXTERNAL SOURCE. THIS OPTION SHOULD BE USED WHEN A FISSION SPECTRUM SOURCE IS DESIRED FOR A PROBLEM WHICH CONTAINS NO FISSIONABLE NUCLIDES. ALL OTHER TYPE 08 CARDS ARE NEGLECTED IF COLS. 13-24 OF THE FIRST TYPE 09 CARD	
CN CN CN CN		IF COLS. 37-48 OR 49-54 ON ANY TYPE 08 CARD ARE BLANK OR ZERO, OR IF COLS. 13-24 OR 25-30 ON ANY TYPE 08 CARD- AFTER THE FIRST ARE BLANK OR ZERO, THE BEST OF THE TYPE 08 DATA ARE IGNORED.	-
CN CN CN CN CN		IF COLS. 25-30 ARE BLANK ON THE FIRST TYPE 08 CARD ENCOUNTERED, THE SOURCE SPECIFIED IN COLS. 13-24 IS ASSIGNED TO ALL ULTRAFINE GROUPS AND THE REST OF THE CARD TYPE 08 DATA IS IGNOPED. IF COLS. 31-36 AND/OR 55-60 ARE BLANK, THE SOURCE VALUE GIVEN IN COLS. 13-24 AND/OF 37 40 ADDITION TO THE COURS	-
N N		GIVEN IN COLS. 25-30 AND/OR 49-54. ANY GROUP OF	•

-261-

CN CN CN C		COVERED BY THIS DATA WILL BE ASSIGNED THE EXTERNAL SOURCE VALUE O. AS MANY TYPE O8 CARDS AS NECESSARY MAY USED SPECIFY THE EXTERNAL SOUPCE DATA.	-
C			-
C	вчс	KLING SPECIFICATIONS (TYPE 09)	-
C	2,0		-
CL	FORMAT	(12, 10X, 3E12. 5, 2I6, E12. 5)	
	COLUMNS	CONTENTSIMPLICATIONS, IF ANY	-
CD CD	1-2	09	-
CD CD	13-24	INITIAL BUCKLING GUESS.	-
CD CD	25-36	SECOND BUCKLING GUESS.	-
CD CD CD	37-48	CONVERGENCE CRITERION, EPS, FOR BUCKLING ITERATION TO KEFF=1. ADSOLUTE VALUE OF (KEFF-1.).LE.EPS.	-
CD CD CD	49-54	HIGHER ENERGY ULTRAFINE GROUP NUMBER TO WHICH VALUES OF BUCKLING APPLY.	-
CD CD CD	55-60	LOWER ENERGY ULTRAFINE GROUP NUMBER TO WHICH VALUES OF BUCKLING APPLY.	-
C D C D C	61-72	EXTENDED TRANSPORT APPROXIMATION BUCKLING, KAPPA SQUARED.	-
CN CN CN CN CN CN CN CN CN CN CN CN CN C		AS MANY TIPE OF CARDS MAY BE USED AS NECESSANT TO SPECIFY THE BUCKLING DATA. IF COLS. 49-54 ARE BLANK, THF BUCKLING DATA WILL BE ASSUMED TO BE GROUP INDEPENDENT. IF COLS. 55-60 ARE BLANK, "HE DATA IN COLS. 13-24 WILL APPLY TO THE ULTRAFINE GROUP GIVEN IN COLS. 49-54. IF COLS. 61-72 ARE BLANK, THE BUCKLING GIVEN IN COLS. 13-24 ON THE FIRST TYPE 09 CARD ENCOUNTFRED WILL BE USED FOR KAPPA SQUARED. NOTE THAT THE SECOND BUCKLING GUESS IN COLS. 25-36 AND THE CONVERGENCE CRITERION IN COLS. 37-48 ARE PERTINENT ONLY FOR ENERGY INDEPENDENT BUCKLING PROBLEMS WHICH WILL ITERATE ON BUCKLING TO KEFF=1. IF COLS. 37-48 ARE BLANK OR ZERO, THE SECOND BUCKLING GUESS IS IGNORED AND NO BUCKLING ITERATION IS PERFORMED. COLS. 25-48 WILL	
CN CN CN		BE IGNORED IF COLS. 49-54 ARE NON-BLANK. ANY GROUP NOT COVERED BY THIS DATA WILL BE ASSIGNED THE BUCKLING VALUE O.	-

-262-

٢

,

CN CN C		COLS. 61-72 ARE IGNORED AFTER READING THE FIRST - TYPE 09 CARD
C		
C CR	 TH3	ERMAL GROUP CROSS SECTION DATA (TYPE 10) -
C CL	FORMAT	(I2,4X,A6,5E12.5)
CD	COLUMNS	CONTENTSIMPLICATIONS, IF ANY -
CD ·	1-2	10
CD CD CD	7-12	NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY - (COLS. 13-18 ON TYPE 06 CARDS)
CD CD	13-24	- MICROSCOPIC THERMAL GROUP CAPTURE CROSS SECTION (BARNS)-
CD CD	25-36	- MTCROSCOPIC THERMAL GROUP FISSION CROSS SECTION (BARNS) -
	37-48	NUMBER OF NEUTRONS EMITTED PER FISSION IN THE - THERMAL GROUP
CD CD	49-50	MICROSCOPIC THERMAL GROUP TOTAL CROSS SECTION (BARNS).
CD CD CD	6 1- 72	THERMAL GROUP AVERAGE RECIPROCAL VELOCITY (SEC/CM) - (DEFAULT=1./2.2E+5).
CN CN CN CN CN		AS MANY TYPE 10 CARDS ARE USED AS NECESSARY TO - SPECIFY THE THERMAL GROUP VALUES. ANY MATERIAL NOT - SPECIFIED ON A TYPE 10 CARD VILL BE ASSIGNED THERMAL - GROUP CROSS SECTIONS EQUAL TO THE LAST EPI-THERMAL - GROUP VALUES -
C N C N C N	· ·	THE FIRST POSITIVE VALUE ENCOUNTERED IN COLS. 61-72 - OF ANY TYPE 10 CARD WILL BE USED FOR THE AVERAGE - THERMAL GROUP RECIPROCAL VELOCITY THE THERMAL GROUP -
C N C		VELOCITY IS NOT MATERIAL DEPENDENT
C		
C CE	<u>А</u> ЛХ	ILIARY THERMAL GROUP CROSS SECTION DATA (TYPE 11) -
C CL C	FORMAT	(I2,4X,A6,5E12.5) -
ם: ם: ם:	COLUMNS	CONTENTSIMPLICATIONS, IF ANY -

-263-

CD CD	1-2	11 -
CD CD	7-12	NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY - (COLS. 13-18 ON TYPE 06 CARDS)
CD	13-24	MICROSCOPIC THERMAL GROUP N-ALPHA CROSS SECTION (BARNS) -
CD CD	25-36	MTCROSCOPIC THERMAL GROUP N-P CROSS SECTION (BARNS)
CD CD	37-48	MICROSCOPIC THERMAL GROUP N-D CROSS SECTION (BARNS)
C D C D	49-60	MICROSCOPIC THERMAL GROUP N-H3 CROSS SECTION (BARNS)
C D C	61-72	MICROSCOPIC THERMAL GROUP N-HE3 CROSS SECTION (BARNS)
CN		AS MANY TYPE 11 CARDS ARE USED AS NECESSARY TO -
ĊN		SPECIFY THE AUXILIARY THERMAL GROUP DATA. ANY MATERIAL -
CN		NOT SPECIFIED ON A TYPE 11 CARD WILL BE ASSIGNED -
CN		THERMAL GROUP CROSS SECTIONS EQUAL TO THE LAST -
CN		EPT-THERMAL GROUP VALUES.
C		-
C		
	,	
C		
CR	RESC	DNANCE HETEROGENEITY SPECIFICATIONS (TYPE 12) -
C		- (T2 (17) (5 (2) (5))
CL C	rurrage	(12,4X, A0, 5 (2A0))
CD	COLUMNS	CONTENTSIMPLICATIONS, IF ANY -
CD	======	
CD	1-2	12
CD	7-12	
CD	, = 12	COIS 13-18 ON TYPE OF CARDS OR TE NO TYPE OF CARDS -
CD		ARE SUPPLIED, COLS. $19-24$, $37-42$, AND $55-60$ ON DATA -
CD		SET A.NTP TYPE 14 CARDS).
CD		-
CD	13-18	REGION LABEL.
CD		•
CD .	19-24	REGION LABEL.
CD.	,	-
CD ,	25-30	REGION LABEL.
CD	31-36	REGION LABEL.
C D C D	37-42	REGION LABEL
CD		-
CD	43-48	REGION LABEL.

-264-

١

- CD 49-54 REGION LABEL.
- CD 55-60 REGION LABEL.

CD

CD

CD

Ć CN

CN

CN

CN

CN

CN

CN

CN

CN CN

CN

CN CN

CN

CN

CN

CN

CN

CN CN

CN

CN

CN CN

CN

CN

CN

CN

CN

CN

CN CN

CN

CN

CNCN

CN

CN

CN

CN

'N

L'N

- CD61-65 REGION LABEL.
- CD 67-72 REGION LABEL.

7-12

13-18

M1

R2

MATERIALS NOT SPECIFIED ON THE TYPE 12 CARDS WILL RECEIVE A HOMOGENEOUS RESONANCE TREATMENT. FOR CYLINDRICAL GEOMETRY (NGEOM=2) ONLY COLS. 1-2 AND 7-12 ARE PERTINENT. MATERIALS NAMED IN COLS. 7-12 WILL RECEIVE A RESONANCE HETEROGENEITY TREATMENT IN THE CENTRAL PIN REGION OF THE CYLINDRICAL CELL IF THEY ARE PRESENT IN THE PIN REGION AND IF THEY ARE RESONANCE MATERIALS. FOR SLAB GEOMETRY (NGEOM=1) ONE OR MORE TYPE 12 CARDS ARE SUPPLIED FOR EACH MATERIAL NAMED IN COLS. 7-12 WHICH IS TO RECEIVE A RESONANCE HETEROGENEITY TREATMENT. IF MORE THAN ONE CARD IS NEEDED FOR A GIVEN MATERIAL. THE MATERIAL LABEL IS REPEATED ON EACH CARD. THE REGION LABELS APPEAR IN PAIRS STARTING WITH THE PAIR IN COLS. 13-18 AND 19-24. IF COLS. 13-18 ARE BLANK, THE REST OF THE DATA ON THE CARD IS IGNORED AND THE MATERIAL NAMED IN COLS. 7-12 WILL BE TREATED HETEROGENEOUSLY IN ALL REGIONS IN WHICH IT APPEARS. REGIONS NAMED IN COLS. 13-18, 25-30, 37-42, 49-54, AND 61-66 INDICATE REGIONS IN WHICH THE MATERIAL IS NOT TO BE TREATED HETEROGENEOUSLY. IF ANY OF THESE COLUMNS ARE BLANK, THE REST OF THE DATA ON THE CARD IS IGNORED. IF COLUMNS 19-24, 31-36, 43-48, 55-60, OR 67-72 ARE NON-BLANK, THEN THE MATERIAL WILL RECEIVE THE SAME TREATMENT IN THE REGION NAMED IN COLS. 13-18 AS IN THE REGION NAMED IN COLS. 19-24, THE -SAME IN REGION 25-30 AS IN THE REGION 31-36, ETC. NO REGION NAMED IN COLS. 19-24, 31-36, 43-48, 55-60, OR 67-72 MAY BE NAMED IN COLS. 13-18, 25-30, 36-42 46-54, OR 61-66. AS AN EXAMPLE, LET MATERIAL M1 APPEAR IN REGIONS R1, R2, R3, R4, AND R5. IF M1 IS TO BE TREATED HOMOGENEOUSLY IN REGIONS R2 AND R5 AND IF IT IS TO RECEIVE THE SAME HETEROGENEITY TREATMENT IN REGION R1 AS IN R3, THEN CARD TYPE 12 WOULD CONTAIN THE FOLLOWING: COLUMNS CONTENTS _____ -----1-2 12

-

9

CN		10 0 ¹¹ (RIANK)	_
CN	-	(BLANC)	_
CN		25-30 R5	-
CN		31-36 (BLANK)	-
CN		37-42 B1	_
CN			_
CN			-
CN		49-54 (BLANK)	-
CN		SINCE REGION R4 IS NOT MENTIONED, MATERIAL M1 WILL	-
CN		BE TREATED HETEROGENEOUSLY IN REGION 64. SINCE COLS.	-
CN		19-54 APE BLANK THE REST OF THE DATA IF ANY ON THE	_
CN		GID DE TOWNER	
CN		CARD ARE IGNORED.	-
CN		THE RESONANCE REGION MODULES WIIL NOT GENERATE	-
CN		RESONANCE CROSS SECTIONS FOR MATERIAL M1 IN REGION R1	-
CN	-	BUT WILL GENERATE RESONANCE CROSS SECTIONS FOR	-
CN		MAMEDIAL MI IN DECION DO	_
		HATERIAL WI IN REGION RS.	
CN		NOTE THAT ANY MATERIAL IN A GIVEN REGION WHICH IS	-
CN		EITHER TO RECEIVE A HOMOGENEOUS TREATMENT OR IS TO	-
CN		RECEIVE THE SAME HETEROGENEOUS TREATMENT AS IN SOME	-
CN		OTHER REGION WILL NOT HAVE ANY RESONANCE OVERLAP	_
CN		THEINERGE ON ANY OWIER DECONANCE MARENTICS IN MUR	_
CN		INFLUENCE ON ANI OTHER RESONANCE MATERIALS IN THE	_
CN		REGION IN QUESTION.	-
С			-
C			-
•			
C			-
CR	INT	EGRAL TRANSPORT SPECIFICATIONS (TYPE 14)	-
С			-
ČI	FORMAT-	(T2 10 Y 2F12 5 4T6)	_
CL C	FORMAL	(12,10,72512.5,410)	
C			-
CD	COLUMNS	CONTENTSIMPLICATIONS, IF ANY	-
CD	======		-
CD	1-2	14	-
CD			_
CD	12 24	ενέρον να υμτού τνατορίτ πολνέροσα αυτόρν ποτλαμένα	_
CD	13-24	ENERGY AT WHICH INTEGRAL TRANSPORT THEORY TREATMENT	
CD	,	BEGINS. (DEFAULT=300.0 EV).	-
CD .			-
CD	25-36	CRITERION USED FOR TESTING FOR INCLUSION OF RESONANCES	-
CD .		FOR RYDER FINE GROUDS JITHIN AN INTERMEDIATE GROUP	_
			_
CD		(DEFAULT=0.05). A 0.0 MOST BE SPECIFIED IN COLS.	-
CD		25-36 IF ALL RESONANCES ARE TO BE INCLUDED.	-
CD			-
CD	37-42	NUMBER OF ULTRA FINE GROUPS PER INTERMEDIATE GROUP	-
	51 42	(DEFNIL - 2)	_
		(DEFRODI-Z) ·	-
CD			-
CD	43-48	NUMBER OF HYPER FINE GROUPS PER DOPPLER WIDTH	-
CD		(DEFAULT=4).	-
CD			-
	40 F#	TRANSVERSE LEAVICE ADDIAN	_
CD	49-54	TRANSVERSE LEAKAGE OPTION.	-
CD CD	49-54	TRANSVERSE LEAKAGE OPTION. OOMIT TRANSVERSE LEAKAGE CORRECTION (DEFAULT).	-

-266-

	•	
CD CD		BUCKLING SPECIFIED IN COLS. 61-72 ON THE TYPE 09
CD		CARD.
CD	55-60	INGROUP SCATTERING OPTION.
CD		0INCLUDE INGROUP SCATTERING (DEFAULT).
C D C		1OMIT INGROUP SCATTERING.
CN		IF INVOKED BY THE PATH DRIVER, RESOLVED RESONANCE BRO
CN		GROUP CROSS SECTIONS WILL BE RECOMPUTED USING
CN		INTEGRAL TRANSPORT THEORY ALGORITHMS FOR ALL BROAD
		GROUPS WHOSE LOWER ENERGIES ARE .LT. THE VALUE
		DEVERTIED. THE INTERMEDIATE GROUPS ARE USED TO DEVERMENTE WHICH DESONANCES ADE TO BE INCLUDED FOR THE
CN		HYPER FINE GROUPS CONTAINED IN THAT INTERMEDIATE GROU
CN		USING THE CRITERION SPECIFIED IN COLS. 25-36.
CN		THE CODE WILL ADJUST THE HYPER FINE GROUP WIDTH TO BE
CN		AN INTEGRAL SUB-MULTIPLE OF THE ULTRA FINE GROUP WIDT
CN		WITH A MAXIMUM HYPER FINE GROUP WIDTH OF 0.001.
CN		IF COLS. 43-48 ARE NEGATIVE, A CONSTANT HYPER FINE
CN		GROUP WIDTH IS USED FOR ALL BROAD GROUPS BASED ON THE
CN		ABSOLUTE VALUE OF THE QUANTITY GIVEN IN COLS. 43-48.
CN		IF THE PATH INVOKES AREA 10, THE PROBLEM MIXTURE MUST
CN		CONTAIN MATERIALS HAVING RESOLVED RESONANCES.
C C	v	
C		
CR C	FO	IL SPECIFICATIONS (TYPE 15)
CL C	FORMAT	(I2,4X,A6,2(A6,E12.5),2E12.5)
CD	COLUMNS	CONTENTS IMPLICATIONS, IF ANY
CD	======	=======================================
CD	1-2	
CD		15
CD		15
CD	7-12	15 FOIL IDENTIFICATION LABEL.
ິມ	7-12	15 FOIL IDENTIFICATION LABEL.
(1)	7-12 13-18	15 FOIL IDENTIFICATION LABEL. NUCLIDE IDENTIFICATION IN THE LIBRARY.
CD CD	7-12 13-18 19-30	15 FOIL IDENTIFICATION LABEL. NUCLIDE IDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24).
CD CD CD	7-12 13-18 19-30	15 FOIL IDENTIFICATION LABEL. NUCLIDE IDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24).
CD CD CD CD	7-12 13-18 19-30 - 31-36	15 FOIL IDENTIFICATION LABEL. NUCLIDE IDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24). NUCLIDE IDENTIFICATION IN THE LIBRARY.
CD CD CD CD CD	7-12 13-18 19-30 - 31-36	15 FOIL IDENTIFICATION LABEL. NUCLIDE IDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24). NUCLIDE IDENTIFICATION IN THE LIBRARY.
CD CD CD CD CD CD	7-12 13-18 19-30 - 31-36 37-48	15 FOIL IDENTIFICATION LABEL. NUCLIDE IDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24). NUCLIDE IDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24).
CD CD CD CD CD CD CD	7-12 13-18 19-30 - 31-36 37-48	15 FOIL IDENTIFICATION LABEL. NUCLIDE IDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24). NUCLIDE IDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24).
	7-12 13-18 19-30 - 31-36 37-48 49-60	15 FOIL IDENTIFICATION LABEL. NUCLIDE IDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24). NUCLIDE IDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24). FOIL THICKNESS IN CM.
	7-12 13-18 19-30 - 31-36 37-48 49-60	15 FOIL IDENTIFICATION LABEL. NUCLIDE IDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24). NUCLIDE TDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24). FOIL THICKNESS IN CM.
	7-12 13-18 19-30 - 31-36 37-48 49-60 61-72	15 FOIL IDENTIFICATION LABEL. NUCLIDE IDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24). NUCLIDE IDENTIFICATION IN THE LIBRARY. NUCLIDE ATOMIC DENSITY (ATOMS/CC*1.E-24). FOIL THICKNESS IN CM. FOIL TEMPERATURE IN DEGREES K. (DEFAULT=300.0).

-267-

C CN CN CN CN CN CN CN CN CN CN CN CN CN		AS MANY TYPE 15 CARDS MAY BE USED AS NECESSARY TO SPECIFY THE FOILS. IF MORE THAN ONE TYPE 15 CARD IS NEEDED TO SPECIFY A PARTICULAR FOIL, THE FOIL LABEL MUST BE REPEATED IN COLS. 7-12 ON SUBSEQUENT CARDS. COLS. 49-72 ARE PERTINENT ONLY FOR THE FIRST TYPE 15 CARD FOR A GIVEN FOIL. ALL FOILS SPECIFIED ON TYPE 15 CARDS WILL BE EDITED AT EACH MESH INTERVAL BOUNDARY AS SPECIFIED ON THE DATA SET A.NIP TYPE 06 CARDS FOR HETEROGENEOUS PROBLEMS, OR FOR THE HOMOGENEOUS SPECTRUM- FOR HOMOGENEOUS PROBLEMS WHEN RESOLVED RESONANCE CROSS SECTIONS ARE RECOMPUTED USING INTEGRAL TRANSPORT THEORY. IF THE PATH INVOKES AREA 10, THE PROBLEM MIXTURE MUST CONTAIN MATERIALS HAVING RESOLVED RESONANCES.
С		
C P	LT2.	
CL	FORMAT	(I2,4X,11A6)
CD	COLUMNS	CONTENTSIMPLICATIONS, IF ANY
CD	=======	
CD	1-2	16 -
CD		-
CD CD	7-12	FISSION SPECTRUM NUCLIDE IDENTIFICATION LABEL ON - THE LIBRARY
CD CD	13-18	FISSIONABLE NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY-
CD CD	19-24	FISSIONABLE NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY-
CD CD	25-30	FISSIONABLE NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY-
CD CD	31-36	FISSIONABLE NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY-
CD CD	37-42	FISSIONABLE NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY-
CD CD	43-48	FISSIONABLE NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY-
CD CD	49-54	FISSIONABLE NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY-
CD CD	55-60	FISSIONABLE NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY-
CD CD	6 1- 66	FISSIONABLE NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY-
CD	67-72	FTSSIONABLE NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY

.

С			-
CN		IF NO TYPE 16 CARDS ARE GIVEN, THE FISSION SPECTRUM	-
CN		WILL BE OBTAINED FROM A MIXTURE OF ALL SPECTRA	
CN		REPRESENTED BY THE MATERIALS PRESENT IN THE SPECTRUM	_
CN		COMPOSITION. THE FISSION SPECTRUM SPECIFIED IN	_
CN		COLS. 7-12 NILL BE USED FOR THE FISSIONABLE NUCLIDES	-
CN		SPECIFIED IN COLS. 13-18 19-24 FTC IF COLS 13-18 OF	_
CN		ANY TYDE 16 CARD ARE BIANK THE REF. OF THAT TYDE 16	_
		CARD TO TOWARD AND THE REST OF THAT TIPE TO	_
CN		CARD 13 IGNORED AND THE FISSION SPECIFUS SECTION SPECIFUS IN $-2-12$ WILL BE HEED FOR ANY FIGURENTER MULTIPLE	-
CN		NOR SPECIFIED ON OFFED THE 14 CADDS ONLY ONE THESE	-
CN		- NOI SPECIFIED ON OIDER FIPE TO CARDS. ONLI OND FIPE TO - - CADD-DIME & DIANK PIPED IN COLC: 12-10 IC DEDMINITED	-
CN .		TE NO SUCH CARD TO CLUBN ANY FIGHTONADID	-
CN		IF NO SUCH CARD IS GIVEN ANY FISSIONABLE	-
CN		NUCLIDE NOI SPECIFIED UN A TIPE TO CARD WILL BE	-
		ASSIGNED ITS OWN FISSION SPECIRUM. AS MANY TYPE 16	-
CŅ		CARDS AS NECESSARY MAY BE USED. THE FISSION SPECTRUM	-
CN		IDENTIFICATION LABEL SHOULD BE REPEATED ON	-
CN		ADDITIONAL CARDS. COLS. 13-18, 19-24, ETC. SHOULD	-
CN		CONTAIN LABELS CORRESPONDING TO LABELS IN COLS. 13-18	-
CN		OF THE TYPE 06 CARDS.	-
С			-
C CR	 UNR	ESOLVED RESONANCE FIXED ENERGY MESH (TYPE 17)	-
C			_
CL	FORMAT	(12,10X,2(E12,5,16,16))	-
С			_
CD	COLUMNS	CONTENTS IMPLICATIONS, IF ANY	_
CD	======		-
CD	1-2	17 -	-
CD			-
CD	12-24	LETHARGY WIDTH.	
CD			_
CD	25-30	INITIAL FIXED MESH POINT NUMBER.	_
CD			
CD	31-36	FINAL FIXED MESH POINT NUMBER.	_
CD			-
CD	37-48	LETHARGY WIDTH.	-
CD			-
CD	49-54	INITIAL FIXED MESH POINT NUMBER.	-
CD			_
CD	55-60	FINAL FIXED MESH POINT NUMBER.	_
С			_
CN		A MAXIMUM OF 500 FIXED ENERGY MESH POINTS MAY BE USED.	-
CN		AS MANY TYPE 17 CARDS AS NECESSARY MAY BE USED. IF NO	-
CN		TYPE 17 CARD IS SUPPLIED. THE BUILT IN FIXED ENERGY	-
1		MESH STRUCTURE IS USED. THE LETHARGY WIDTH SPECIFIED	-
J		IN COLS. 12-24 IS USED FOR POINTS SPECIFIED IN COLS.	-

CN CN CN CN CN CN CN CN CN CN CN CN C C C 		25-30 THROUGH 31-36. THE LETHARGY WIDTH SPECIFIED IN COLS. 37-48 IS USED FOR POINTS SPECIFIED IN COLS 49-54 THROUGH 55-60. IF COLS. 25-30 OF 49-54 ARE BLANK, - THE LETHARGY WIDTH SPECIFIED IN COLS. 12-24 OF 37-48 IS- USED FOR ALL REMAINING POINTS AND THE REST OF THE DATA ON THE TYPE 17 CARDS ARE IGNORED. THE FIXED ENERGY MESH COVERS THE ENTIRE ENERGY RANGE AS SPECIFIED ON THE TYPE 05 CARDS IF COLS. 31-36 AND/OR COLS. 55-60 ARE BLANK, THE VALUE SPECIFIED IN COLS.' 12-24 AND/OR COLS. 37-48 IS USED FOR THE MESH POINT SPECIFIED IN COLS. 25-30 AND/OR- COLS. 49-54.
C C R	номс	GENEOUS INFINITELY DILUTE SPECIFICATIONS (TYPE 18) -
C CL	FORMAT	(I2,4X,11A6) -
C CD /	COLUMNS	CONTENTSIMPLICATIONS, IF ANY -
CD CD	1-2,	18
CD CD CD	7 - 12	NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY - (COLS. 13-18 ON TYPE 06 CARDS)
CD CD CD	13-18	NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY - (COLS. 13-18 ON TYPE 06 CARDS)
C D C D C D	19-24	NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY - (COLS. 13-18 ON TYPE 06 CARDS)
CD CD CD	25-30	NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY - (COLS. 13-18 ON TYPE 06 CARDS)
CD CD CD	31-36	NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY - (COLS. 13-18 ON TYPE 06 CARDS)
CD CD CD	37-42	NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY - (COLS. 13-18 ON TYPE 06 CARDS)
CD CD CD	43-48	NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY - (COLS. 13-18 ON TYPE 06 CARDS)
CD CD CD	49-54	NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY - (COLS. 13-18 ON TYPE 06 CARDS)
CD	55-60	NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY

-270-
CD CD CD CD CD CD CD CD CN CN CN CN CN CN	61-66	<pre>(COLS. 13-18 ON TYPE 06 CARDS). NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY (COLS. 13-18 ON TYPE 06 CARDS). NUCLIDE IDENTIFICATION LABEL ON THE LIBRARY (COLS. 13-18 ON TYPE 06 CARDS). MATERIALS NAMED ON THE TYPE 18 CARDS WILL BE ASSUMED TO BE INFINITELY DILUTE FOR THE HOMOGENEOUS RESOLVED RESONANCE CALCULATION. MATERIALS WHICH ARE ASSUMED TO BE INFINITELY DILUTE WILL NOT BE INVOLVED IN THE OVERLAP CALCULATION FOR ANY OTHER MATERIAL. THEIR RESONANCE INTEGRALS WILL BE SET TO THE INFINITELY DILUTE LIMIT OF PI/(2.*BETA). SEE CARD TYPE 12 FOR THE HETEROGENEOUS SPECIFICATIONS.</pre>
С		-
C		
C CR	RES	ONANCE EDIT OPTIONS (TYPE 19)
C		· -
CL	FORMAT	(I2,4X,6I6) -
CD	COLUMNS	CONTENTS IMPLICATIONS, IF ANY -
CD	======	
CD	1-2	19 -
CD CD	7-12	DATA SET UNRES EDIT FLAG
CD CD	13-18	DATA SET ATNUAT EDIT FLAG
CD CD	19-24	DATA SET RESINT EDIT FLAG
CD CD	25-30	DATA SET SIGMAP EDIT FLAG.
CD CD	31-36	DATA SET LORENZ EDIT FLAG
CD CD CD CD CD CD CD CD CD CD CD	37-42	 BROAD GROUP RESONANCE CROSS SECTION FLAG. OINCLUDE RESONANCE CROSS SECTIONS IN THE BROAD GROUP- CROSS SECTIONS (DEFAULT). 1OMIT ALL RESONANCE CROSS SECTIONS FROM THE BROAD - GROUP CROSS SECTIONS. 2OMIT RESONANCE CAPTURE AND FISSION CROSS SECTIONS - FROM THE BROAD GROUP CROSS SECTIONS. EACH DATA SET WHOSE EDIT FLAG IS NON-ZERO WILL BE
N		EDITED. OTHERWISE IT WILL NOT BE EDITED

e**

-271-

C----CR GENERAL PROBLEM EDIT OPTIONS (TYPE 20) С CL FORMAT - - - - (I2, 4X, 6I6)С COLUMNS CONTENTS... IMPLICATIONS, IF ANY CD ======= CD 1-2 20 CD CD ULTRA-FINE-GROUP MACROSCOPIC FISSION, TOTAL, TRANSPORT 7-12 CD CD AND SCATTERING CROSS SECTIONS. CD O...DO NOT EDIT DATA (DEFAULT). 1...EDIT DATA. CD CD 13-18 ULTRA-FINE-GROUP MACROSCOPIC MODERATING PARAMETERS. CD 0...DO NOT EDIT DATA (DEFAULT). CD CD 1...EDIT DATA. CD 19-24 ULTRA-FINE-GROUP MICROSCOPIC PL SCATTERING CROSS CD CD SECTIONS, PO AND P1 ELASTIC TRANSFEP MATRICES. CD 0...DO NOT EDIT DATA (DEFAULT). N...EDIT DATA AT ULTRA-FINE-GROUPS 1,N+1,2*N+1,... CD CD 25 - 30ULTRA-FINE-GROUP SPECTRUM. CD O...EDIT DATA (DEFAULT). CD 1...DO NOT EDIT DATA. CD CD BROAD GROUP RESONANCE CROSS SECTION EDIT FLAG. 31-36 CD 0...DO NOT EDIT DATA (DEFAULT). CD 1...EDIT DATA. CD CD BROAD GROUP REACTION RATE EDIT FLAG. 37-42 CD O...EDIT DATA. CD 1...DO NOT EDIT DATA. CD С COLS. 19-24 SHOULD ROUTINELY BE SET AT ZERO (DEFAULT) CN AS THE EDIT OF TRANSFER MATRICES PRODUCES A GREAT CN DEAL OF PAPER OUTPUT. CN С C-C----CR COMPOSITION TEMPERATURES (TYPE 21) С CL FORMAT - - - - + (I2, 10X, 3(A6, E12.5))С

-272-

1

CD CD	COLUMNS	CONTENTSIMPLICATIONS, IF ANY	-
CD CD	1-2	21	_
CD CD	13-18	COMPOSITION LABEL.	-
CD CD	19-30	TEMPERATURE IN DEGREES K (DEFAULT=300.0).	-
CD CD	31-36	COMPOSITION LABEL.	-
C D C D	37-48	TEMPERATURE IN DEGREES K (DEFAULT=300.0).	_
CD CD	49-54	COMPOSITION LABEL.	-
CD C	55-66	TEMPERATURE IN DEGREES K (DEFAULT=300.0).	-
CN CN CN CN CN CN CN CN CN CN CN CN CN C		AS MANY TYPE 21 CARDS AS NECESSARY MAY BE USED TO SPECIFY THE COMPOSITION TEMPERATURES. COMPOSITION LABELS IN COLS. 13-18, 31-36, AND 49-54 MUST CORRESPOND TO LABELS GIVEN IN COLS. 13-18 ON DATA SET A.NIP TYPE 14 CARDS. THE COMPOSITION TEMPERATURES ARE USED ONLY FOR THE INTEGRAL TRANSPORT THEORY CALCULATIONS. IN THE CASE OF HOMOGENEOUS PROBLEMS FOR WHICH COLS. 37-42 ON THE TYPE 03 CARD ARE ZERO, THE TEMPERATURE USED FOR AN INTEGRAL TRANSPORT THEORY CALCULATION WILL BE THE VALUE SPECIFIED IN COLS 19-30 ON THE FIRST TYPE 21 CARD PROVIDED. IF NO TYPE 21 CARDS ARE GIVEN, THE TEMPERATURE FOR A HOMOGENEOUS PROBLEM WILL BE THAT SPECIFIED ON THE TYPE 06 CARD FOR THE FIRST MATERIAL AFTER THE INPUT IS REORDERED TO CORRESPOND TO THE ORDER OF MATERIALS IN THE LIBRARY. IF THE PATH INVOKES AREA 10, THE PROBLEM MIXTURE MUST CONTAIN MATERIALS HAVING RESOLVED RESONANCES.	
C			-
CK C CL	FORMAT	-(12,10X,A6,16,2E12.5)	-
C C D	COLUMNS	CONTENTSIMPLICATIONS, IF ANY	-
CD CD CD	1- 2	22	-
CD CD	13-18	NUCLIDE IDENTIFICATION LABEL ON LIBRARY.	-
~ ~	19-24	ISOTOPE CLASSIFICATION.	-

-273-

.

CD CD CD CD CD CD CD CD CD CD		0UNDEFINED (DEFAULT)1FISSILE2FERTILE3OTHER ACTINIDE4FISSION PRODUCT5STRUCTURAL6COOLANT7CONTROL
CD CD	25-36	TOTAL THERMAL ENERGY YIELD/FISSION (MEV/FISSION)
C D	37-48	TOTAL THERMAL ENERGY YIELD/CAPTURE (MEV/CAPTURE)
CN CN C	•	IF COLS. 25-36 OR 37-48 ARE BLANK, THE CORRESPONDING - DATA IS OBTAINED FROM DATA SET MCC2F1
(
C CR C	FISS	ION SPECTRA TEMPERATURES (TYPE 23)
C	r Ogmar = = = =	- (12,4x,5(A0,E(2.5))
CD CD CD	COLUMNS ====== 1-2	CONTENTSIMPLICATIONS, IF ANY
CD	7-12	NUCLIDE IDENTIFICATION LABEL ON LIBRARY
CD	13-24	FISSION SPECTRUM TEMPERATURE, E.V.
CD	25-30	NUCLIDE IDENTIFICATION LABEL ON LIBRARY
CD CD	31-42	FISSION SPECTRUM TEMPERATURE, E.V.
CD	43-48	NUCLIDE IDENTIFICATION LABEL ON LIBRAPY
CD	49-60	FISSION SPECTRUM TEMPERATURE, F.V
CN CN CN CN CN CN CN CN CN		NOTE THAT THE TYPE 23 DATA ARE PERTINENT ONLY FOR FISSIONABLE ISOTOPES THE TEMPERATURES ON THE TYPE 23 CARDS ARE USED TO OVERRIDE THE CORRESPONDING TEMPERATURES OF PROBLEM - MATERIALS ON THE LIBRARY FOR USE IN GENERATING - FISSION SPECTRA. IF AN EXTERNAL SOURCE IS SPECIFIED - SUCH THAT ON THE TYPE 08 CARD COLS. 13-24 ARE BLANK - AND COLS. 25-30 CONTAIN THE FISSION SPECTRUM NUCLIDE - IDENTIFICATION LABEL ON THE LIBRARY, THEN THAT LABEL - AND THE CORRESPONDING LABEL IN COLS. 7-12 ON THE TYPE

1990 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -

-274-

CN	23 CARD NEED NOT CORRESPOND TO ONE OF THE PROBLEM	-
CN	MATERIALS. ANY LIBRARY MATERIAL NOT REFERENCED	-
CN	ON THE TYPE 23 CARDS WILL USE THE LIBRARY SPECIFIED	-
CN	FISSION SPECTRUM TEMPERATURE.	-
CN	IF COLS. 7-12, 25-30, OR 43-48 ARE BLANK ON ANY TYPE	-
CN	23 CARD, THE REST OF ANY SUPPLIED TYPE 23 DATA WILL BE	-
CN	IGNORED.	-
С		-
C		

CEOF

-275-

APPENDIX B. MC²-2 BCD Input Files. A.NIP

C****	*****	* * * * * * * * * * * * * * * * * * * *
C C	. •	
c		FREERRED OVISVIS EI ERE
CF	A . N	IP -
CE	BCD	INPUT FOR HETEROGENEOUS MC**2-II AND SDX CALCULATIONS -
C N		THIS IS A USER-SUPPLIED BCD DATA SET. IT IS -
CN		AN ABBREVIATED VERSION OF THE DATA SET -
CN		A.NIP
		THE LIST FOR BACH RECORD IS GIVEN IN IERAS OF -
CN		COLUMNS 1 2 CONTAIN THE CARD TYPE NUMBER.
C		-
C * * * * * :	* * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *
C		
CR	EXT	ERNAL BOUNDARY CONDITIONS (TYPE 04)
CL	FORMAT	(I2,10X,216) -
С		-
CD	COLUMNS	CONTENTSIMPLICATIONS, IF ANY -
CD	1-2	
CD		-
CD ·	13-18	BOUNDARY CONDITION AT LOWER "X" BOUNDARY OF CELL
CD	19-24	BOUNDARY CONDITION AT UPPER "X" BOUNDARY OF CELL
CD		· · · · · · · · · · · · · · · · · · ·
CD		10REFLECTIVE.
		12WHITE.
CD		· · · · · · · · · · · · · · · · · · ·
CN		THE LEFT BOUNDARY CONDITION MUST BE REFLECTIVE AND -
		THE RIGHT BOUNDARY CONDITION MUST BE WHITE FOR -
CN		CONDITIONS MUST BE THE SAME AND MAY BE ONLY REFLECTIVE -
CN		OR PERIODIC FOR SLAB GEOMETRIES
С		-
ι = = = = .		
C		
CR	ድር (ጥህ	TON DOUNDARY COORDINATES AND MESH STRUCTURE -
· C	(-
CL	FORMAT	(12,4X,A6,2E12.5,I6) -
C C D	COLUMNS	CONTENTS IMPLICATIONS IF ANY -
		Southerstellion for a nut

-276-

•

- CD ======= CD 1-2 06 CD 7-12 CD **EEGION LABEL (REPEATED ON ADDITIONAL TYPE 06 CARDS).** CD CD 13-24 "X"-DIRECTION LOWER-BOUNDARY COOPDINATE. CD CD 25-36 "X"-DIRECTION UPPER-BOUNDARY COORDINATE. CD CD 37-42 NUMBER OF INTERVALS IN "X" DIRECTION. CD CN "X" REPRESENTS X OR R. CN CN REGIONS MAY BE DEFINED USING THE OVERLAY PROCEDURE. CΝ WITH THE LATEST REGION ASSIGNMENT OVERLAYING THE CN PREVIOUS CONFIGURATION, OR USING THE USUSAL PRECEDURE, CN WITH EACH REGION'S BOUNDARIES GIVEN EXPLICITLY. CN REGION LABELS MUST BE NON-BLANK. ONLY THE "X"-DIRECTION UPPER BOUNDARIES NEED CN CN BE GIVEN FOR REGIONS AFTER THE FIRST. С CR COMPOSITION SPECIFICATIONS (TYPE 14) С CL FORMAT - - - - (I2, 10X, A6, 3(A6, E12.5))С CONTENTS...IMPLICATIONS, IF ANY CD COLUMNS CD ====== 14 CD 1-2 CD COMPOSITION LABEL (REPEATED ON ADDITIONAL TYPE 14 CD 13-18 ĊĎ CARDS). CD · 19-24 TSOTOPE LABEL. CD CD · ISOTOPE ATOM DENSITY. 25-36 CD CD CD 37-42 ISOTOPE LABEL. CD **TSOTOPE ATOM DENSITY.** 43-54 CD CD CD 55-60 ISOTOPE LABEL. CD 61-72 **TSOTOPE ATOM DENSITY.** CD CD COMPOSITION LABELS MUST BE NON-BLANK. ΓN ISOTOPE LABELS IN COLS. 19-24, 37-42, AND 55-60 MUST Ν

-277-

.

CN		CORRESPOND TO NUCLIDE IDENTIFICATION LABELS ON THE -
CN		LIBRARY IF THEY DO NOT CORRESPOND TO ANY OF THE -
CN		MATERIALS NAMED ON THE DATA SET A.MCC2 TYPE 06 CARDS
CN		IN OTHER CASES, THEY MUST CORFESPOND TO SOME PROBLEM -
CN		MATERIAL LABEL IN COLS. 19-24 ON THE DATA SET A.MCC2 -
CN		TYPE 06 CARDS
С		. –
C		
C		
CR	COME	CONTION AND REGION ASSIGNMENTS (TYPE 15)
C		······································
CL	FORMAT	(I2,4X,11A6) -
C		
CD	COLUMNS	CONTENTSIMPLICATIONS, IF ANY -
CD	======	***************************************
CD	1-2	15 -
CD		-
CD	7-12	COMPOSITION LABEL (REPEATED ON ADDITIONAL TYPE 15 -
CD		CAPDS).
CD		•
CD	13-18	REGION LABEL OF REGION CONTAINING SPECIFIED -
CD		COMPOSITION
CD		-
CD	19-24	REGION LABEL OF REGION CONTAINING SPECIFIED -
CD		COMPOSITION.
CD	25 20	PROTON TARREL ON DECTON CONMITNING SDECTETED
CĐ CĐ	25-30	REGION LABEL OF REGION CONTAINING SPECIFIED
		COMPOSITION.
CD	31-36	REGION LABEL OF REGION CONTAINING SPECIFIED -
CD .	51 50	COMPOSITION.
CD		-
CD	37-42	REGION LABEL OF REGION CONTAINING SPECIFIED -
CD		COMPOSITION
CD		-
CD	43-48	REGION LABEL OF REGION CONTAINING SPECIFIED
CD		COMPOSITION
CD		-
CD	49-54	REGION LABEL OF REGION CONTAINING SPECIFIED -
CD		COMPOSITION
CD		-
CD	55-60	REGION LABEL OF REGION CONTAINING SPECIFIED -
CD		COMPOSITION
CD		
CD	61-66	REGION LABEL OF REGION CONTAINING SPECIFIED -
CD		COMPOSITION.
CD	7 7 7 1	
CD	0/-/2	KEPTON LAREF OF REGION CONTAINING SELCTERED

I

-278-

.

CD C	COMPOSITION
C	
CEOF	

-279-

.

APPENDIX B. MC²-2 BCD Input Files. A.STP015

C****	* * * * * * * * * * *	************************************
C		-
C		-
CF	A.51	
CE	GENE	GRAL BCD INPUT FOR MC**2-II PATH DRIVER -
		THIS IS A USER SUPPLIED BCD DATA SET
CN		THE LIST FOR EACH RECORD IS GIVEN IN TERMS -
CN		OF THE BCD FORMAT OF THAT DATA CARD
CN		COLUMNS 1-2 CONTAIN THE CARD TYPE NUMBER
С М Ĉ		BLANK FIELDS FRODUCE THE DEFROET OF TONS.
C****	* * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *
C	 PATH	1 OPTIONS (TYPE 01)
C		-
CL	FORMAT	(I2,4X,9I6) -
CD	COLUMNS	CONTENTSIMPLICATIONS, IF ANY -
CD	====== 1-2	01
CD		- · · · · · · · · - · · · · · · · · · ·
CD	7-12	MC**2-II INPUT PROCESSOR CALCULATION -
CD ·		O INVOKE INPUT PROCESSOR (DEFAULT).
CD		1DO NOT INVOKE INPUT PROCESSOR
CD		
CD	13-18	UNRESOLVED RESONANCE CALCULATION -
CD		0DO UNRESOLVED RESONANCE CALCULATION (DEFAULT).
CD		1DO NOT DO UNRESOLVED RESONANCE CALCULATION
CD		-
CD	19-24	RESOLVED RESONANCE CALCULATION -
CD		0DO RESOLVED RESONANCE CALCULATION (DEFAULT).
CD		1DO NOT DO RESOLVED RESONANCE CALCULATION
CD		-
CD	25-30	MACROSCOPIC CROSS SECTION AND MODERATING PARAMETER -
CD CD		APEN 7 (CSCOOR)
CD	•	0DO CALCULATION OF MACROSCOPIC CROSS SECTIONS -
CD		AND CONTINUOUS SLOWING DOWN MODERATING PARAMETERS -
CD		(DEFAULT).
CD		TDO NOT CALCULATE MODERATING PARAMETERS AND -
		CHCROSCOFIC CROSS SECTIONS.
CD	31-36	UFG SPECTRUM AND BROAD GROUP CROSS SECTIONS -
•		

CD		AREA 8 (CSC009).
CD		0DO ULTRA-FINE-GROUP CALCULATION AND BROAD GROUP -
СЪ		CROSS SECTIONS (DEFAULT)
CD		1DO NOT DO ULTRA-FINE-GROUP CALCULATION AND BROAD -
CD		GROUP CROSS SECTIONS
CD		-
CD	37-42	BROAD GROUP SPECTRUM CALCULATION -
CD		AREA 9 (CSC010)
CD		0DO BROAD GROUP SPECTRUM CALCULATION (DEFAULT)
CD	. ,	1DO NOT DO BROAD GROUP SPECTRUM CALCULATION
CD		-
CD	43-48	BROAD GROUP CROSS SECTION EDITS (CSE009)
CD		0EDIT BROAD GROUP CROSS SECTION FILE ISOTXS -
CD		(DEFAULT). –
CD		1DO NOT EDIT BROAD GROUP CROSS SECTION FILE ISOTXS
CD		-
CD	49-54	ISOTXS TO XS.ISO CONVERSION (CSE007).
CD		0CREATE EQUIVALENT OF DATA SET ISOTXS IN THE DOUBLE -
CD		PRECISION DATA SET XS.ISO (DEFAULT)
CD		1CREATE EQUIVALENT OF DATA SET ISOTXS IN A SINGLE -
CD		PRECISION VERSION OF DATA SET XS.ISO
CD		-1DO NOT CREATE DATA SET XS.ISO
CD		(OPTION NOT AVAILABLE IN ARGONNE CODE CENTER VERSION -
CD		OF MC**2-II.) -
CD		-
CD	55-60	HYPER FINE GROUP INTEGRAL TRANSPORT CALCULATION -
CD		AREA 10 (CSC011).
CD		0DU NOT DU HIPER FINE GROUP INTEGRAL TRANSPORT -
		- CALCULATION (DEFAULT).
		1DU HIPER FINE GROUP ININGRAL TRANSPORT CALCULATION
CD	61-66	
CD	01-00	$\frac{1}{2} = \frac{1}{2} = \frac{1}$
		1 EXECUTE CSEU12 (XS-150 ED110K) -
		COPTION NOT AVAILABLE IN ARCONNE CODE CENTER VERSION -
CD		OF MC**2-II)
c		
CN		TE COLS. 13-18 OR COLS. 19-24 ARE 1. AREA 6.5 WILL BE -
CN		EXECUTED TO OBTAIN THE RESOLVED-UNRESOLVED OVERIAP -
CN		CALCHLATION.
C		-
Č	. 	
C		
CR	UN	RESOLVED RESONANCE OPTION (TYPE 02)
C	₽<u>∩</u>₽МХ ₩	
CL	r U A B A 1	
D	COLUMNS	CONTENTSIMPLICATIONS, IF ANY -

-

CD	======		-
CD	1-2	02	-
CD		· · · · · · · · · · · · · · · · · · ·	-
CD	7-12	ARFA 10 UNRESOLVED RESONANCE OPTION (CSC011).	-
CD		0EXCLUDE UNRESOLVED RESONANCE CROSS SECTIONS FROM	-
CD		THE HYPER FINE GROUP INTEGRAL TRANSPORT	-
CD		CALCULATION (DEFAULT).	-
CD		1INCLUDE UNRESOLVED RESONANCE CROSS SECTIONS IN THE -	-
CD		HYPER FINE GROUP INTEGRAL TRANSPORT CALCULATION.	-
С			-
C			•

CEOF

-282-

APPENDIX B. MC²-2 BCD Input Files. ACSE12

C****	****	* * * * * * * * * * * * * * * * * * * *
С		-
C		PREPARED 12/09/74 AT ANL -
C F	AC 51	F 12
CE	BCD	INPUT FOR XS.ISO EDITOR CSE012 -
C	201	-
CN		THIS IS A USER-SUPPLIED BCD DATA SET
CN		THE LIST FOR EACH RECORD IS GIVEN IN TERMS OF -
CN		THE BCD FORMAT OF THE DATA CARD
CN		COLUMNS 1-2 CONTAIN THE CARD TIPE NUMBER.
C****	****	
0		
C		
CR	EDI	T SPECIFICATIONS (TYPE 01)
CI.	FORMAT	$= -(T_2, 4X, A_6, T_6)$
C	TORNET	(12,48,80,10)
CD	COLUMNS	CONTENTSIMPLICATIONS, IF ANY -
CD	======	
CD	1-2	
CD	7 10	
	7-12	ISOTOPE LABEL.
CD	13-18	EDIT FLAG.
CD		-
CN		IF COLS. 7-12 ON THE FIRST TYPE 01 CARD ARE BLANK, THE -
CN		ENTIRE XS.ISO DATA SET WILL BE EDITED. IF COLS. 13-18 -
CN		ARE 0, ALL CROSS SECTION TYPES WILL BE EDITED. IF -
CN		COLS. 13-18 ARE 1, ONLY THE PRINCIPAL CROSS SECTIONS -
		SCATTERING ARRAYS WILL BE FORTED TO ARE 2, UNLI 105
CN		3. ONLY FILE 1 OF DATA SET XS.ISO WILL BE EDITED
CN		IF COLS. 7-12 ARE BLANK ON THE FIRST TYPE 01 CARD BUT -
CN		ADDITIONAL TYPE 01 CARDS ARE SUPPLIED, THE ISOTOPE -
CN		NAMED IN COLS. 7-12 OF EACH ADDITIONAL CARD WILL BE -
CN		EDITED ACCORDING TO THE FLAG SET IN COLS. 13-18 OF THAT-
CN		BE EDITED ACCORDING TO THE VALUE IN COLS 13-18 ON THE -
		FIRST TYPE 01 CARD (WHICH HAS COLS. 7-12 BLANK). IF -
CN		COLS. 13-18 ON AN ADDITIONAL TYPE 01 CARD ARE -1, THE -
CN		ISOTOPE NAMED IN COLS. 7-12 OF THAT CARD WILL NOT BE -
CN		EDITED. IF THE FIRST TYPE 01 CARD IS NON-BLANK IN -
CN		COLS. 7-12, ONLY THE ISOTOPES NAMED ON THE TYPE 01 -
CN		CARDS WILL BE EDITED, EACH ACCORDING TO THE FLAG IN -
CN		LULS. 13-18 ON THE SAME CARD. IF DATA SET ACSET2 IS -
		SECTION TYPES ARE EDITED FOR EACH ISOTOPE.

-

ŗ

CN CN CN C		IF NO TYPE 01 CARDS ARE SUPPLIED, THE EDIT WILL - CORRESPOND TO HAVING PUT IN A SINGLE BLANK TYPE 01 - CARD
C		
CR	B PC	DINTER EDIT OPIION (TYPE 02) -
CL	FORMAT	(12,4X,16)
CD	COLUMNS	CONTENTS IMPLICATIONS, IF ANY -
CD CD CD	====== 1-2	02
CD CD CD CD	7-12	BPOINTER EDIT FLAG0NO BPOINTER DEBUGGING PRINTS1DEBUGGING DUMP PRINTOUT2DEBUGGING TRACE PRINTOUT
CD C C		3FULL DEBUGGING PRINTOUT.

CEOF

-284-

APPENDIX B. MC²-2 BCD Input Files. ACS009

		· ·	
C****	*****	* * * * * * * * *	*****
С			-
С	• · ·		PREPARED 2/27/75 AT ANL -
Ċ			-
CF	ACS) () 9	-
C F	הסצ	רא מות אר	R EDITING ISOTIS DATA SET (CSE009) -
c			-
CN			THIS IS & HERR SHEDITED BOD DATA SET
CN			THE IS A COLL SCITCID DED DATA SD .
CN			
			COLUMNS 1-2 CONDITIN THE CARD THAT AND THE NUMPER
CN			COLUMNS 1-2 CONTAIN THE CARD TIPE NUMBER
CN			ELANK FIELDS PRODUCE THE DEFAULT OPTIONS
С			
C * * * * *	*****	* * * * * * * *	************
C			
CR	COME	PUTER CON	TAINER ARRAY (TYPE 01) -
С			· -
CC	OPTI	CONAL CAR	D TYPE -
С			-
CL	FORMAT	(I2,4X,	316) -
С			-
CD	COLUMNS		CONTENTSIMPLICATIONS, IF ANY -
CD	=======	========	
CD	1-2	01	•
CD			-
CD	7-12	SIZE OF	MAIN STORAGE ARRAY IN REAL*8 WORDS -
CD		(DEFAULT	= 30000).
CD		, .	-
CD	13-18	STZE OF	BULK CORE STORAGE ON REAL*8 MORDS (DEFAULT=0)
CD	15 10		
CD	19-2/1	ם קיזאדהם	FRUGGING EDIT
CD	12 24		
CD		ע טאיייט וופקת 1	
CD CD		טמשע •••ו ממשת	
CD C		5FULL	, DEBUGGING PAINIOUI.
C av			
CN		THIS CAR	D TIPE IS USED UNLY AFTER THE BPOINTER -
CN		CONTAINE	R ARRAY CAN NOT BE ALLOCATED BI: -
CN		(I) ATTE	MPTING TO CALCULATE THE SIZE OF THE CONTAINER -
CN		ARRA	AY FROM THE FILE CONTROL RECORD OF THE ISOTES -
CN		DATA	SET, OR -
CN		(2) OBTA	INING THE SIZE OF THE CONTAINER ARRAY SPECIFIED-
CN		IN D	ATA SET PRBSPC (IF IT EXISTS) WHICH CONTAINS -
CN		THE	SIZE OF THE CONTAINER ARRAY SPECIFIED BY THE -
CN		DATA	SET A.MCC2.
С			-
C			

-285-

. ISO	OTOPES TO BE EDITED (TYPE 02)
0.53	TIONAL CARD TYPE
	(T) UV 10361
FORMAT ==	(12,4%,1240)
COLUMNS	CONTENTS IMPLICATIONS, IF ANY
======	=======================================
1-2	02
7-12.	TSOTOPE TO BE EDITED.
2	
13-18	ISOTOPE TO BE EDITED.
19-24	ISOTOPE TO BE EDITED.
~~ ~~ [.]	
25-30	ISOTOPE TO BE EDITED.
31-36	TSOTOPE TO BE EDITED
51-50	IDDIOL TO DE EDITOR.
37-42	ISOTOPE TO BE EDITED.
43-49	ISOTOPE TO BE EDITED.
49-54	ISOTOPE TO BE EDITED.
00-00	IDUIURE TO DE EDITED.
61-66	TSOTOPE TO BE EDITED.
67-72	ISOTOPE TO BE EDITED.
73-80	ISOTOPE TO BE EDITED.
	LISUTUPE NAME MUST BE LEFT-JUSTIFIED IN FURNAT FIST
	ΑΕ ΜΊΝΑ ΜΑΡΔ ΟΟ ΟΊΟΡΟ ΤΟ ΝΆΟΔΟΟΥΡΑ ΜΙΑ ΘΩ ΠΟΡΡ ΜΤΙΟ ΤΠΟΡΑΝΕΝ ΟΤΨΝΚΟ ΕΚΕΡΓΚΑΦΑ ΜΙΑ ΘΔ ΠΟΡΡ
	AS MANI TIPE VZ CARDO AS NECESSARI MAI DE USEV.
	NU TIPE UZ CARD WILL RESULT IN ALL THE ISOTOPES 1"
	THE ISOTXS DATA SET TO BE EDITED.
1	A TYPE OZ CARU WITH NU ISOTOPE NAMES WILL RESULT J
	ONLY ISOTOPE INDEPENDENT DATA TO BE EDITED E.G., A
	EDIT OF THE FIRST THREE OR FOUR RECORDS OF DATA SE
	LSUTXS.

CEOF

APPENDIX C

MC²-2 BINARY INTERFACE FILES

C ****	*****	* * * * * * * * * * * * *	****	****	**~
C					-
Ċ			PREPARED 03/03/75		-
Ċ			,,,		_
CF		ISOTXS			-
CE		MICROSCOPIC	GROUP NEUTRON CROSS SEC	TIONS	-
C					-
CN			THIS FILE PROVIDES A BA	SIC BROAD GROUP	-
CN			LIBRARY, ORDERED BY ISO	TOPE	-
С			·		-
C * * * * *	*****	****	*****	*****	** *
CD	אוזד יי		2 FOR TRM MACHINES 1 O	ТНЕРИТСЕ	
CD	10121		z tok ibi inclikis, i o	INDAWISE	
C					
CS		FILE STRUCTU	UR E		-
C 5		PECOPD TV	ក្រ	DRESENT TE	_
C3			. F Li 		_
C3 CS		 FTIF TDFN		ATWAVS	_
C 2		FILE LODA	POT	AT WAYS	-
CS CS			NOE	ATWAYS	-
C5			ሰ ጥ እ	TCHIST GT 1	_
C3 CS	*****	***** /BEDEN	T FOR ALL TSOTOPES	ichi51.61.1	-
C5	*	ממשמת איייייי המתקיד הייייי		λτωλγς	-
CS CS	*		ATA TRALE	ALWAIS .	-
CS	*		CROSS SECTIONS	ATUAVS	-
CS	*	TSOTOPF C	HT DATA		-
C5 C5	* ***	***** (REDEA	T TO NSCMAX SCATTERING	BLOCKS	_
CS	* *	**** (REPEN	T FROM 1 TO NSBLOK	biocks)	-
C5	* *	* SCYLLER * SCYLLER	IC SUB-BLOCK	IORD (N) GT O	_
CS	* * * * * *	*****	9 22P PFOCK	* LOKD (N) • 31• 0	_
C S					-
C					
C					
C B		FILE IDENTIE	ICATION (TYPE 1)		-
С			o		-
CL	HNAME,	$(HUS \in (I), I=1)$,2),IVERS		-
С					-
CW	1+3*MU	ΓĨ			-
C	 · –				-
CD	HNAME		HOLLERITH FILE NAME - I	SOTXS - (A6)	-
CD	HUSE		HOLLERITH USER IDENTIFI	CATION (A6)	-
CD	IVERS		FILE VERSION NUMBER		-
C					-
C					

CR FILE CONTROL (TYPE 2) С CL NGROUP, NISO, MAXUP, MAXDN, MAXORD, ICHIST, NSCMAX, NSBLOK С CW 8 С CD NGPOUP NUMBER OF ENERGY GROUPS IN SET CD NUMBER OF ISOTOPES IN SET NISO CD MAXUP MAXIMUM NUMBER OF UPSCATTER GROUPS MAXIMUM NUMBER OF DOWNSCATTER GROUPS CD MAXDN MAXIMUM NUMBER OF SCATTERING ORDERS CD MAXORD CD SET FISSION SPECTRUM FLAG ICHIST CDICHIST.EO.O, NO SET FISSION SPECTRUM CD ICHIST.EQ.1, SET VECTOR CD ICHIST.GT.1, SET MATRIX MAXIMUM NUMBER OF BLOCKS OF SCATTERING DATA CD NSCMAX BLOCKING CONTROL FOR SCATTER MATRICES. THE CD NSBLOK CD SCATTERING DATA ARE BLOCKED INTO NSBLOK RECORDS PER SCATTERING BLOCK CD С C-------FILE DATA (TYPE 3) CP С (HSETID(I), I=1, 12), (HISONM(I), I=1, NISO), CL CL 1 (CHI (J), J=1, NGROUP), (VEL (J), J=1, NGROUP),CL2(EMAX(J), J=1, NGROUP), EMIN, (LOCA(I), I=1, NISO)С (12+NISO) *MULT+1+NISO+ (2+ICHIST* (2/(ICHIST+1))) *NGROUP CW С HOLLERITH IDENTIFICATION OF SET (A6) CDHSETID HOLLERITH ISOTOPE LABEL FOR ISOTOPE I (A6) CD HISONM (I) SET FISSION SPECTRUM (PRESENT IF ICHIST.EQ.1) CD CHI (J) MEAN NEUTRON VELOCITY IN GROUP J (CM/SEC) CD VEL (J) MAXIMUM ENERGY BOUND OF GROUP J (EV) CD EMAX (J) MINIMUM ENERGY BOUND OF SET (FV) CD EMIN NUMBER OF RECORDS TO BE SKIPPED TO READ DATA CD LOCA(I) CD FOR ISOTOPE I. LOCA (1) = 0С _____ C----SET CHI DATA (TYPE 4). CR С

-289-

cc	PRESENT IF	ICHIST.GT.1	-
CL	((CHI(K,J),K=1,ICH	HIST), J=1, NGROUP), (ISSPEC(I), I=1, NGROUP)	-
C W	NGROUP* (ICHIST+1)	-	-
CD CD	CHI (K,J)	FRACTION OF NEUTRONS EMITTED IN GROUP J AS A - RESULT OF FISSION IN ANY GROUP USING SPECTRUM K-	-
CD CD CD CD	ISSPEC	ISSPEC (I) = K IMPLIES THAT SPECTRUM K IS USED TO CALCULATE EMISSION SPECTRUM FROM FISSION IN CROUP I	-
C			•
С		PROLAND GROUP INDEPENDENT DATA (TYPE 5)	-
C N			-
CL	HABSID, IDENT, MAT, A	AMASS, EFISS, ECAPT, TEMP, SIGPOT, ADENS, KBR, ICHI,	-
CL	1IFIS, IALF, INP, IN21	N, IND, INT, LTOT, LTRN, ISTRPD,	-
CL	2 (IDSCT (N), N=1, NSCN	(LORD(N), N=1, NSCMAX),	-
CL	3((JBAND(J,N), J=1, N))	NGROUP), N=1, NSCMAX),	-
CL	4((IJJ(J,N), J=1, NG))	ROUP), N=1, NSCMAX)	-
С		•	-
CW	(2*NGROUP+2) *NSCM1	AX+17+MULT*3	-
С) _	· · · · ·	-
CD	HABSID	HOLLERITH ABSOLUTE ISCTOPE LABEL - SAME FOR	-
CD		ALL VERSIONS OF THE SAME ISOTOPE IN SET (A6) -	-
CD	ITENT	IDENTIFIER OF LIBRARY FROM WHICH BASIC DATA	-
CD		CAME (E.G. ENDF/B) (A6)	-
CD	MAT	ISOTOPE IDENTIFICATION (E.G. ENDE/B MAT NO.)	-
CD		(A6)	-
CD	AMASS	GRAM ATOMIC WEIGHT	-
CD	EFISS	TOTAL THERMAL ENERGY YIELD/FISSION (4.SEC/FISS)	-
CD	ECAPT	TOTAL THERMAL ENERGY YIELD/CAPTURE (W.SEC/CAPT) -	-
CD	TEMP	ISOTOPE TEMPERATURE (DEGREES KELVIN)	~
CD	SIGPOT	AVERAGE EFFECTIVE POTENTIAL SCATTERING IN	-
CD	•	RESONANCE RANGE (BARNS/ATOM)	÷.
CD	ADENS	DENSITY OF ISOTOPE IN MIXTURE IN WHICH ISOTOPE -	-
CD		CROSS SECTIONS WERE GENERATED (A/BARN.CH)	
CD	KBR	ISOTOPE CLASSIFICATION	_
CD		KBE=0, UNDEFINED	
CD		=1, FISSILE	_
CD		= 2 OTHER ACTIVE	_
CD		= 5, OTHER ACTINIDE -// RISCION DRODUCE	_
CD		-4, FISSION PRODUCT	
CD CD		-D, SIRUCTURE -6 COOLNIT	_
CD ·		-0, COULANI -7 CONTROL	
CD	TOUT	TRADA BIRCIAN CORCUPIEM DIAC	
CD	TCHT	TOTOLE LTODION SECTION LTVG	

-290-

.

C-D		ICHI.EO.O, USE SET CHI -
ĊD		ICHI.EO.1. ISOTOPE CHI VECTOR -
CD		TCHT.GT.1. ISOTOPE CHT MATRIX -
CD	TETS	(N.F) CROSS SECTION FLAG -
CD	2120	TETS=0. NO FISSION DATA IN PRINCIPAL CROSS -
CD		SECTION RECORD -
CD		=1. FISSION DATA PRESENT IN PRINCIPAL -
CD		CROSS SECTION RECORD -
	таты	(N AIPHA) CROSS SECTION FLAG
CD	14111	SIME OPTIONS AS TELS
CD	тир	(N D) CROSS SECTION FING
CD		SAME ODTIONS AS TELS
CD	тиом	AND CROSS SECTION FINC
CD		CAME ODDIONS AS TELS
CD	TND	AME OPIIONS AS IFIS
	IND	(N, D) CROSS SECTION FLAG
CD		SAME OPTIONS AS IFIS -
CD	\mathbf{T} .N.L.	(N,T) CROSS SECTION FLAG
CD		SAME OPTIONS AS IFIS -
CD	LTOT	NUMBER OF MOMENTS OF TOTAL CROSS SECTION -
CD		PROVIDED IN PRINCIPAL CROSS SECTIONS RECORD -
CD	LTRN	NUMBER OF MOMENTS OF TRANSPORT CROSS SECTION -
CD		PROVIDED IN PRINCIPAL CROSS SECTIONS RECORD -
CD	ISTRPD	NUMBER OF COORDINATE DIRECTIONS FOR WHICH -
CD		COORDINATE DEPENDENT TRANSPORT CROSS SECTIONS -
CD		ARE GIVEN. IF ISTRPD=0, NO COORDINATE DEPENDENT-
CD		TRANSPORT CROSS SECTIONS ARE GIVEN -
CD	IDSCT(N)	SCATTERING MATRIX TYPE IDENTIFICATION FOR -
CD		SCATTERING BLOCK N. SIGNIFICANT ONLY IF -
CD	· ·	LORD (N).GT.0 -
CD		IDSCT (N) =000 + NN, TOTAL SCATTERING -
CD		=100 + NN, ELASTIC SCATTERING -
CD		=200 + NN, INELASTIC SCATTERING -
CD		= 300 + NN, (N, 2N) SCATTERING PER -
CD		EMITTED NEUTRON, -
CD		WHERE NN IS THE LEGENDRE EXPANSION INDEX OF -
CD		THE FIRST MATRIX IN BLOCK N -
CD CD	LORD (N)	NUMBER OF SCATTERING ORDERS IN BLOCK N. IF -
CD		LORD (N) =0. THIS BLOCK IS NOT PRESENT FOR THIS -
CD ·		TSOTOPE. IF NN TS THE VALUE TAKEN FROM -
CD		TDSCT(N) THEN THE MATRICES IN THIS BLOCK -
CD		HAVE LEGENDRE FYDANSTON INDICES OF NN NN+1 -
CD		NN+2 NN+IOPD(N) -1 -
	TO ND (T N)	
CD CD	ованы (о,и)	DIOCK N -
	T T T (T))	DACEMENTAL CRAMERENTIA CRACE CRAMERIA
	100 (J, N)	TH COMMUTATING DAMA FOR CROUP A COMMERTIC
CD .		IN SUATTERING DATA FOR GROUP J, SUAFERRING -
CD.		BLOCK N, COUNTED FROM THE FIRST WORD OF -
CD		GROUP J DATA -
C		· · ·
[] = = = = =		

.

-291-

C----CR PRINCIPAL CROSS SECTIONS (TYPE 6) С CL((STRPL(J,L), J=1, NGROUP), L=1, LTRN),CL 1 ((STOTPL(J,L), J=1, NGROUP), L=1, LTOT), (SNGAM(J), J=1, NGROUP), CL 2 (SFIS(J), J=1, NGROUP), (SNUTOT(J), J=1, NGROUP), CL 3 (CHIISO (J), J=1, NGROUP), (SNALF (J), J=1, NGROUP), 4 (SNP(J), J=1, NGROUP), (SN2N(J), J=1, NGROUP),CL CL 5(SND(J), J=1, NGROUP), (SNT(J), J=1, NGROUP),CL 6 ((STRPD(J, I), J=1, NGROUP), I=1, ISTRPD) -С CW (1+LTRN+LTOT+IALF+INP+IN2N+IND+INT+ISTRPD+2*IFIS+ CW ICHI*(2/(ICHI+1))) *NGROUP С CD STRPL PL WEIGHTED TRANSPORT CROSS SECTION THE FIRST ELEMENT OF ARRAY STRPL IS THE CD -CURRENT (P1) WEIGHTED TRANSPORT CROSS SECTION CD PL WEIGHTED TOTAL CROSS SECTION CD STOTPL CD THE FIRST ELEMENT OF ARRAY STOTPL IS THE CD FLUX (PO) WEIGHTED TOTAL CROSS SECTION CD. SNGAM (N,GAMMA) (PRESENT IF IFIS.GT.0) CD SFIS (N,F)TOTAL NEUTRON YIELD/ (PRESENT IF IFIS.GT.0) CD -SNUTOT CD FISSION -CHISO ISOTOPE CHI (PRESENT IF ICHI.EO.1) CD (PRESENT IF IALF.GT.O) CD SNALF (N, ALPHA) (PRESENT IF INP.GT.O) CD SNP (N, P)(PRESENT IF IN2N.GT.0) CD SN2N (N, 2N)(PRESENT IF IND.GT.O) CD SND (N, D)(PRESENT IF INT.GT.0) CD SNT (N,T) COORDINATE DIRECTION (PRESENT IF ISTRPD.GT.O) CD STRPD CD I TRANSPORT CROSS CD SECTION C C----C______ CR ISOTOPE CHI DATA (TYPE 7) С PRESENT IF ICHI.GT.1 CC С ((CHIISO(K,J),K=1,ICHI),J=1,NGROUP),(ISOPEC(I),I=1,NGROUP) CL С CW NGROUP* (ICHI+1) С CD CHITSO **ISOTOPE FISSION SPECTRUM** CD ISOPEC FISSION SPECTRUM USED FOR A GIVEN SOURCE GROUP

-292-

С _____ C----CR SCATTERING SUB-BLOCK (TYPE 8) С CC PRESENT IF LORD (N) .GT.0 С CL((SCAT(K,L), K=1, KMAX), L=1, LORDN)Ç KMAX=SUM OVER J OF JBAND (J) WITHIN THE J-GROUP RANGE OF THIS CC CC SUB-BLOCK. IF M IS THE INDEX OF THE SUB-BLOCK, THE J-GROUP CC RANGE CONTAINED WITHIN THIS SUB-BLOCK IS CC JL = (M-1) * ((NGROUP-1)/NSBLOK+1) + 1 TO JU = M * ((NGROUP-1)/NSBLOK+1)CC LORDN=LORD (N) С CW KMAX*LORDN С CD SCATTERING MATRIX OF SCATTERING ORDER L, FOR SCAT (K, L) - REACTION TYPE IDENTIFIED BY IDSCT (N) FOR THIS CD BLOCK. JBAND (J) VALUES FOR SCATTERING INTO CD GROUP J ARE STORED AT LOCATIONS K=SUM FROM 1 CD TO (J-1) OF JBAND (I) PLUS 1 TO K+JBAND(J)-1. CD THE SUM IS ZERO WHEN J=1. J-TO-J SCATTER IS CD THE IJJ(J)-TH ENTRY IN THE RANGE JBAND(J). CD VALUES ARE STORED IN THE ORDER (J+MAXUP), CD (J + MAXUP - 1),..., (J + 1), J, (J - 1), ..., (J - MAXDN), CD . **:-**WHERE MAXUP=IJJ(J)-1 AND MAXDN=JBAND(J)-IJJ(J) -CD С

CEOF

-293-

APPENDIX C. MC²-2 Binary Interface Files. MCC2F1

C****	* * * * * * *	* ** * * * * * * * * * * * * * * * * * *
С		
С		PREPARED 2/11/75 AT ANL -
С		-
CF		1CC2F1 -
CE		- DMINISTRATIVE -
С		· · · · · · · · · · · · · · · · · · ·
C * * * *	*****	< * * * * * * * * * * * * * * * * * * *
CD	MMNO	
	PIDA 1	NUMBER OF MATERIALS IN THE LIBRARY WITH
CD		INELASTIC AND/OR (N, 2N) DATA (C.F. DATA SET
CD		MCC2F6)
.CD	MULT	2 FOR IBM MACHINES, 1 OTHERWISE
CD	NMAT	NUMBER OF MATERIALS IN LIBRARY
CD	NMAX	MAXIMUM NUMBER OF GROUPS OF INELASTIC OR
CD		(N, 2N) DATA FOR ANY MATERIAL IN THE LIBRARY
CD		(.EQ.MAX (NINEL,N2NTH) WHERE NINEL AND
CD		N2NTH ARE THE THRESHHOLD GROUP NUMBERS FOR
CD		INELASTIC AND (N.2N) SCATTERING RESPECTIVELY)
CD	NRESMT	NUMBER OF MATERIALS IN LIBRARY WITH RESOLVED
CD		RESONANCE PARAMETERS (C.F. DATA SET MCC2F4)
CD	NUNRMT	NUMBER OF MATERIALS IN LIBRARY WITH
CD		UNRESOLVED RESONANCE DARAMETERS
		(C = D) $(C = D)$ $(C =$
CD		(C.T. DRIR SEI MCCZIS)
C		
CR		SPECIFICATIONS (TYPE 1) -
С		-
CC		LWAYS PRESENT -
С		-
CL	NMAT, N	ROUP, NRESMT, NUNRMT, MSORS, NPASS, NPL, IPTMAX, ETOP, DELTAU, -
CL	1MANY1,	IMAT, NMAX -
С		-
CW	13	-
C		-
съ	NGROUP	NUMBER OF ENERGY GROUPS IN LIBRARY -
CD	MSORS	NUMBER OF FISSION SPECTRA SPECIFIED IN ITERARY -
CD	12010	(C = DATA SET MCC2E7) =
CD	NDASS	NUMBER OF IBLOCKS! OF IECENDEE DATA IN ITERADY -
CD	NEMOJ	ICE DATA SET MCCOPOLE DAIA IN LIDRARI -
	NDT	(C, r, DATE SET HCC2ro) =
	NFL	
CD		TRANSPORT APPROXIMATION -
CD		(C.F. DATA SET MCC2F8) -
CD	IPTMAX	MAXIMUM NUMBER OF LEGENDRE COMPONENTS -
CD		PROVIDED IN THE LIBRARY -
CD		(C.F. DATASET MCC2F8) -
CD	ETOP	HIGHEST ENERGY POINT IN LIBRARY, I.E. ENERGY -
CD		AT TOP OF FIRST ENERGY GROUP

CD CD CD CD C C	DELTAU MANY 1	GROUP LETHARGY WIDTH FOR ALL ENERGY GROUPS IN LIBRARY NUMBER OF ENERGY LEVELS FOR EACH 'BLOCK' OF DATA	
CR CC	MATERIAL N	AMES (TYPE 2)	
C CL C C	(NAME(I), I=1, NMA	T)	
C CD C C		DOUBLE PRECISION (R*8) MATERIAL IDENTIFICATION	-
C CR C CC	MATERIAL I Always pre	DENTIFICATION (TYPE 3)	
C CL CL C	(A(I), I=1, NMAT), 1 (EFISS(I), I=1, NM	(IZ(I),I=1,NMAT), (MAT(I),I=1,NMAT), AT), (ECAPT(I),I=1,NMAT)	
CW CD CD CD CD CD CD CD CD CD CD C	5*NMAT A IZ MAT EFISS ECAPT	MATERIAL MASS/NEUTRON MASS ATOMIC NUMBER OF MATERIAL ENDF/B MATERIAL IDENTIFICATION NUMBER MEV/FISSION FOR EACH MATERIAL MEV/CAPTURE FOR EACH MATERIAL WHERE CAPTURE REFERS TO NON-FISSION ABSORPTION	
C CR C CC CC CL CL	RESONANCE PRESENT IF (NRES(I),I=1,NRE 1(EMAXR(I),T=1,NR	CONTROL INFORMATION (TYPE 4) NRESMT.GT.O OR IF NUNRMT.GT.O SMT), (NREC(I), I= 1, NRESMT), ESMT), (EMAXU(I), I= 1, NUNRMT),	

-295-

С		· · ·	-
CW C	3*NRESMT+2*NUNRMT	· · · · · · · · · · · · · · · · · · ·	-
CD	NBES	NUMBER OF RESOLVED RESONANCES FOR EACH	
CD		RESOLVED RESONANCE MATERIAL. NRES INCLUDES ALL	-
CD		RESONANCES FOR ALL ISOTOPES OF A MULTI-ISOTOPE	-
CD		MATERIAL	-
CD	NRምሮ	NUMBER OF RECORDS FOR EACH RESOLVED RESONANCE	-
CD .	NI 3C	MATERIAL	-
CD CD	тмаур	ENERGY OF THE HIGHEST RESOLVED RESONANCE	-
CD	DURAD	FOR FACH MATERIAL	_
CD	EMAYI	ENERGY OF THE HIGHEST POINT AT WHICH	_
CD	DINGAG	UNRESOLVED RESONANCE CALCULATIONS ARE	-
CD		PERFORMED. FOR EACH MATERIAL	_
CD	EMINI	ENERGY OF THE LOWEST POINT AT WHICH UNRESOLVED	-
CD		RESONANCE CALCULATIONS ARE PERFORMED.	-
CD		FOR EACH MATERIAL	-
c	·		-
CN		THE INFORMATION IN THIS RECORD REFERS TO THE	-
CN		STRUCTURE OF DATA SETS MCC2F3 AND MCC2F4	-
С			-
C			
		· · ·	
C			
CR	INELASTIC A	ND N2N DISTRIBUTIONS CONTROL INFORMATION	
CR	(TYPE 5)		
Ċ			-
СС	PRESENT IF	MMAT.GT.O	-
С			-
CL	(ANAME(1), I=1, MMA)	Γ), (NINEL(I), I=1, MMAT), (N2NTH(I), I=1, MMAT),	-
CL	1 (NLEVLS (1), 1=1, MM)	AT), $(N2NLEV(1), 1 = 1, MMAT$), $(MAXI(1), 1 = 1, MMAT)$,	-
CL	2(MAX2(1), 1=1, MMAT)), (MAX3(1), 1=1, MMAT), (MAX4(1), 1=1, MMAT), (MAX3(1), 1=1, MMAT),	-
CL	3(NSINK1(1), 1=1, MM)	AT), (NSINK2(1), 1= 1, MMAT),	-
CL	4 (NUMREC (1), $1 = 1$, NM.	AX), MAXREC	-
C		· · · · · · · · · · · · · · · · · · ·	-
C.₩	· IU*MMAT+MULT*MMAT	T IT NUAX	_
CD	а NI а M 17		_
CD CD	ANADG NTNET	THREE PARCISION (RTO) HAIMAIAL LOBNITICATION	_
CD	NIWSL	CAMPROND GROUP NUMBER FOR INDERSING CAMPROND (-O IR NO INRIASTIC SCAMPRONC)	_
	N ⁽¹⁾ N (11) L	TURESHOLD CROUD NUMBER FOR (N 2N)	_
	NZNIH	$\frac{1}{2} \left(\frac{1}{2} \right) = \frac{1}{2} \left(\frac{1}{2} \right) = \frac{1}$	_
	NITUS	NUMPER OF DISCREET NO (N, 2N) SCRIEDING	_
CD CD	C T A S T N	NOUDER OF PICKEIE INELASIIC SCALLEFING	_
	NONTEW	ОИТСЕЧИТСЯ ИХ СИС ИХ ЭРССТИС СООЧЕНИИ ОМТСЕЧИТСЯ ИХ СИС ИХ ЭРССТИС СООЧЕНИИ	_
CD		TRURIS ROD RACH MAMPETAT	-
	M A V 1	МУАТМИМ ЛИМВЕЕ ОЕ ТИЕЈУСИТС ЕЛУБОБУАТОМ Релего год руси цитегтур	_
	EAAT	CDEGED BUD EVCH WYWEDINI Revision Nondry of Intervisio Pakeoustics	
CD CD	May 2	ΑΥΣΤΜΠΑ ΝΠΑΒΣΟ ΟΥ ΦΑΡΠΙΑΦΣΟ ΤΝΣΙΑΟΦΤΟ ΜΑΥΤΜΠΑ ΝΠΑΒΣΟ ΟΥ ΦΑΡΠΙΑΦΣΟ ΤΝΣΙΑΟΦΤΟ	
CD	11 A A Z	UNYING MOUDER OF INDORIED IMPROVIEC	

۱

0

CD CD CD CD CD	M A X 3 M A X 4	DISTRIBUTIONS FOR EACH MATERIAL - MAX 2. EQ. ZERO OP ONE - MAXIMUM NUMBER OF (N, 2N) EVAPORATION - SPECTRA FOR EACH MATEPIAL - MAXIMUM NUMBER OF TABULATED (N, 2N) -
CD CD CD CD CD	NSTNK1	DISTRIBUTIONS FOR EACH MATERIAL - MAX4.EQ. ZERO OR ONE - NUMBER OF ENERGIES PROVIDED IN TABULATED - INELASTIC DISTRIBUTIONS FOR EACH MATERIAL - IF MAX2.EQ.0 THEN NSINK1.EQ.0
CD CD CD	NSINK2	NUMBER OF ENERGIES PPOVIDED IN TABULATED - (N,2N) DISTRIBUTIONS FOR EACH MATERIAI - IF MAX4.EQ.O THEN NSINK2.EQ.O -
CD CD	MAXREC	MAXIMUM RECORD LENGTH (WORDS) IN FILE MCC2F6 - FOR ANY RECORD TYPE -
CD CD CD	NUMREC	DATA FOR EACH GROUP. EQUAL TO 2 OR 3 - FOR EACH GROUP -
CN CN C		THE INFORMATION IN THIS PECORD REFERS TO - THE STRUCTURE OF DATA SET MCC2F6 -
C		
C CP	HARD SPHERE	POTENTIAL SCATTERING CROSS SECTIONS (TYPE 6)
cc c	ALMAYS PRESE	ENT -
C C L	(SIGP(I), T=1, NMAT)	
C M C	NMAT	- -
CD C	SIGP	HARD SPHERE POTENTIAL SCATTERING CROSS SECTIONS -
CN CN CN C		THE SIGP ARE THE ENERGY INDEPENDNT VALUES - OBTAINED FROM THE NUCLEAR RADJUS USING - 4*PI*(RADIUS) **2 -
CEOF		

-297-

APPENDIX C. MC²-2 Binary Interface Files. MCC2F2

C****	******
C	- התקהות היה 10/15/72 את אאנ
c	PREPARED JZ/15//5 AI ANE
CF	MCC2F2
CE C	FUNCTION TABLE
C * * * * *	******
C	
CR	REAL V COARSE (TYPE 1)
cc	ALWAYS PRESENT
CL	((NRC(I,J),I=1,41),J=1,27)
C W	1107
CD	WRC REAL PART OF W(X,Y) TABULATED AT INCREMENTS OF -
C D C	0.1 FOR -0.1.LE.X.LE. 3.9 AND 0.4.LE.Y.LE.3.0 -
C	
C	
CR	IMAGINARY W COARSE (TYPE 2)
cc cc	ALWÁYS PRESENT
CL	(("IC(I,J),I=1,41),J=1,27)
ĊW	1107 -
C CD	WTC TMAGINARY PART OF W(X,Y) TABULATED AT -
CD	INCREMENTS OF 0.1 FOR -0.1.LE.X.LE.3.9 AND -
CD	0.4.LE.Y.LE.3.0
C	
C	
CR C	REAL W FINE (TYPE 3)
cc	ALVAYS PRESENT
C CL	$((\mathbb{W} \mathbb{P} \mathbb{P} (\mathbb{I}, \mathbb{J}), \mathbb{I} = 1, 41), \mathbb{J} = 1, 27)$
c	
CW C	1107 -

-298-

CD CD CD C	WRF	REAL PART OF W(X,Y) TABULATED AT INCREMENTS - OF 0.1 FOR -0.1.LE.X.LE.3.9 AND AT INCREMENTS - OF 0.02 FOR -0.02.LE.Y.LE.0.5 -
Ŭ		
C CR		IMAGINARY W FINE (TYPE 4)
cc		ALVAYS PRESENT
C CL	((¹ 1F	(I,J),I=1,41),J=1,27) -
C CW	1107	-
C CD CD CD C	A I P	IMAGINARY PART OF W (X,Y) TABULATED AT INCREMENTS OF 0.1 FOR -0.1.LE.X.LE.3.9 AND - AT INCREMENTS OF 0.02 FOR -0.02.LE.Y.LE.0.5 -
C		
C CR	· · ·	EXPONENTIAL INTEGRAL E3 (TYPE 5) -
cc		AL TAYS PRESENT -
CL	(E3(I)	, I=1,1001)
CF	1001	· • •
CD CD C	E.3	EXPONENTIAL INTEGRAL, E3(X), TABULATED AT - INCREMENTS OF 0.01 FOR 0.0.LE.X.LE.10.0 -
C	,,	
C CR		EXPONENTIAL FUNCTION (TYPE 6)
C CC		ALWAYS PRESENT
CL	(EX 201	N(I), I=1, K1)
C W	2*NGR0	
СС СС С		K1=2*NGROUP-1 WHERE NGROUP IS THE NUMBER OF LIBRARY - ENERGY GROUPS AS SPECIFIED IN THE ADMINISTRATIVE FILE - MCC2F1 -

-299-

C CD CD CD CD CD CD CD	EXPON	EXPON(I) = (1.+(1./C) ** (EXP(-(1./C) **(NGROUP-I C=EXP(-DELTAU) AND DEL GROUP LETHARGY WIDTH A ADMINISTRATIVE FILE MC	NGROUP-I))* -)) WHERE - TAU IS THE LIBRARY - S SPECIFIED IN THE - C2F1 -
C		·	
	ν.	· · ·	
C====== CR CR	FIRST FLIGHT (TYPE 7)	TRANSMISSION PROBABIL	ITIES - -
cc c	ALWAYS PRESI	ENT	-
CL CL	((T1(I,J),I=1,181)	,J=1,26),((T2(I,J),I=1	, 181), J=1, 51) -
C W ···	13937		 -
CD CD CD	T1(T,J)	TRANSMISSION PROBABILI SURFACE FOR ANNULAR RE Z AND J-TH VALUE OF X,	TY, INNER TO OUTER - GION, FOR I-TH VALUE OF - WHERE THE X INCREMENT -
CD CD CD CD CD	T2(T,J)	IS 0.04 TRANSMISSION PROBABILI SURFACE FOR ANNULAR RE Z AND J-TH VALUE OF X, IS 0.02	TY, OUTER TO OUTER - GION, FOR I-TH VALUE OF - WHERE THE X INCREMENT - -
CN CN CN CN CN		Z=TOTAL CROSS SECIION RADIUS), AND X=(INNER RADIUS)/(OUTE X TAKES ON VALUES BETW	* (OUTER RADIUS-INNER - R RADIUS) EEN 0.0 AND 1.0 -
C N	,	RANGE OF Z Z	INCREMENT -
CN CN CN CN CN CN CN	· · ·	0.0 TO 0.4 0.4 TO 1.0 1.0 TO 2.6 2.6 TO 5.0 5.0 TO 8.0	0.01 0.02 0.04 0.06 0.10
C	·		
C CR	EXPONENTIAL	INTEGRALS AND FUNCTION	S (TYPE 8) -
C CC [.] C	ALWAYS PRES	ENT	-

÷

-300-

CL	(EXPY(I), I=1, 10)	23), $(EXPM(I), I=1, 1022)$, $(E3(I), I=1, 331)$,
CL	1(E4(I), I=1, 331)	
CW	2707	
CD CD	E3	EXPONENTIAL INTEGRAL, E3(X), TABULATED AT $0(0,01)2(0,02)4(0,08)6.4$
C D C D	E 4	EXPONENTIAL INTEGRAL, E4(X), TABULATED AT $0(0,01)2(0,02)4(0,08)6.4$.
CD CD	ЕХРҮ	ORDINATES Y(I) FOR $EXP(-X) = Y(I) - M(I) * X$ TABULATED FOR 0.0.LE.X.LE.18.0 WITH TABULAR
CD	•	INTERVAL 18.0/1022.
CD CD CD	EXPM	SLOPES M(I) FOR EXP($-X$) = Y(I) - M(I) *X TABULATED FOR 0.0.LE.X.LE.18.0 WITH TABULAR INTERVAL 18.0/1022.
C		

CEOF

-301-

,

APPENDIX C. MC²-2 Binary Interface Files. MCC2F3

C****	*****	* * * * * * * * * * * * * * *	* * * * * * * * * *	* * * * * * * * *	*****	*****	****	* >
С								-
С		~	PREPARED	2/11/75	AT ANL			-
С								-
CF		MCC2F3						-
CF		UNPESOLVED	PESONANCE	<u>ከአጥአ</u>	•			-
0		O W V D D O L V D D	NESONENCE	DAIA				_
C				1 ماد ماد ماد باد ماد ماد ماد داد باد	الد ماد عاد داد ماد ماد ماد د	او ماد ماد بار بار بار بار بار بار بار بار بار	بله مله بله بله ماه بله بله بله ماه مله ماه با	ست ست ست .
C * * * * *	******	****	*******	* * * * * * * * * *	* * * * * * * * *	* * * * * * * * * * * * *	* * * * * * * * * * * * * * *	***
							,	
CD	JST		NUMBER O	F CHANNE	L SPIN S	STATES ASSO	DCIATED WITH	ſ
cn			A DARTIC	ITAR ANG		ENTIM STAT	PE TN	
CD.		• •	11 E 11 (1 ± 0		አክሮም ፖለተ	21131 A 11 3 2 1 1		
C.17			ONS ROOTA		ANCE CAI	CULATION		
CD	LST		NUMBER O	F ANGULA	R MOMENT	CUM STATES	CONSIDERED	
CD			FOR A GI	VEN ISOT	OPE IN U	JNRESOLVED	CALCULATION	
CD	MULT		2 FOR IB	M MACHIN	ES, 1 07	CHERWISE		
СД	NISO		NUMBER O	F ISOTOP	ES IN MA	ATERIAL		
CD	NPT	•	NUMBER O	F ENERGT	ES AT WH	ITCH UNRESC	DLVED	
CD ·	19 A. 1 1			TON TO D	0NF			
CD			CALCULAI	TON TO D				
CD	NUNKMI		NUMBER O	F MATERI	ALS WITH	1 UNRESULV	SD RESUNANCE	
CD			PARAMETE	RS AS SP	ECIFIED	IN THE AD	MINISTRATIVE	;
CD			FILE MCC	2F 1				
		`						
<u> </u>								
(_
CS		FLLE STRUC	TURE					
CS				•				-164
CS		RECORD TYPI	3			PRESENT	IF	-
CS		==================		=========	======	=======	==============	== -
CS		UNRESOLVED	RESONANCE	MATERIA	L NAMES	ALWAYS		-
ĈŜ		MATERTAL SI	PECTETCATT	ONS		ALWAYS		
C.5	* * * * * *		ROD ALL M	AMPDIATC	иттц			-
65		THE LADE DAL	TOR ALL N	WIDUTWD2	N T T D D C V			
CS	*	UNRESUL	VED RESON	ANCE PAR	AMETERS			-
CS	*	ISOTOPE COL	NTROL ·			ALWAYS		7
CS	* ***	*** (REPEAT	FOR ALL I	SOTOPES)	•			-
CS	* *	UNRESOLVED	SPIN STAT	E. AND		ALWAYS		-
C S	* *	FNFRCY DATE	A	_,				-
CC -				אספעם היי	NCE	λτリλΥς		-
05	* *	STATISTICAT	- UNREQUEV.	ED RESON	ANCE	ALWAIS		
CS	* *	TNFORMATIO	N	• • • •				-
CS	*****	* * *						-
С								-
C			~ ~					
C								
0								
CR		ОИКЕЗОТАЕД	RESUNANCE	MATERIA	L NAMES	(IIRE 1)		
C								-
CC	·	ALWAYS PRES	SENT					-
с.								-
ĊŢ.	INAME	(T) . T=1 . NIIN	<u> የ </u>					-
<u>بر</u> ب	/ HETA	(····· • • •					

-302-

C C प्र	M [] [. T * N [] N R M T	-				
C		· · -				
CD C	NAME	DOUBLE PRECISION (R*8) MATERIAL IDENTIFICATION -				
(
_	·	· · ·				
C CR	MATERIAL	SPECIFICATIONS (TYPE 2) -				
с сс	ALVAYS P	RESENT -				
С		-				
C L C I C	(NISO(I), I=1, NUNRMT), (IFI(I), I=1, NUNRMT), 1(ISK(I), I=1, NUNRMT), LSTMAX, JSTMAX, NPTMAX -					
CW C	3*NUNRMT+3					
CD	IFI	FISSILE ISOTOPE INDEX -				
CD	ı	IFI=0 FOR NON-FISSILE MATERIAL				
CD	ISK	NUMBER OF LOGICAL RECORDS OF UNRESOLVED -				
CD	LSTMAX	MAXIMUM VALUE OF LST OVER ALL MATERIALS IN FILE-				
CD	JSTMAX	MAXIMUM VALUE OF JST OVER ALL MATERIALS IN FILE-				
C D C	NPTMAX	MAXIMUM VALUE OF NPT OVER ALL MATERIALS IN FILE-				
C		•				
C	TROPORT					
C	ISOTOPE (CONTROL DATA (TYPE 3) -				
cc C	ALWAYS P	RESENT				
CL CL	(ABUN(I), I=1,NI 1(LST(I), T=1,NIS	ISO), (A (I), I=1, NISO), (RPF (T), I=1, NISO), - 50), (NPT (I), I=1, NISO), (RPS (I), I=1, NISO) -				
C W	6*1150	-				
CD	ABUN	ABUNDANCE OF EACH ISOTOPE -				
CD	A	ISOTOPIC MASS/NEUTRON MASS -				
CD	RDF	RPF=K#R/SQRT(E), K= WAVE NUMBER, - R=CHANNEL RADIUS_E=ENERGY, FOR USE IN -				
CD		OBTAINING P AND D WAVE PENETRATION FACTOR -				
CD	RPS	RPS=K*R/SQRT(E), K= WAVE NUMBER,				
CD		R=EFFECTIVE SCATTERING RADIUS, E=ENEPGY. FOR USE -				
c c		TH ODIAINING IND SPEP AND D WAVE PRASE SULFES -				

-303-

APPENDIX C. $MC^{2}-2$ Binary Interface Files. MCC2F3 (Contd.)

C----UNRESOLVED SPIN STATE AND ENERGY DATA (TYPE 4) CR С CC ALWAYS PRESENT С CL (ES(I), T=1, NPT), (DEL(I), I=1, NPT), (JST(I), I=1, LST)С CV 2*NPT+LST С ĊĐ ĒŜ ENERGIES AT WHICH UNRESOLVED CALCULATION IS TO BE DONE, IN EV. AND ORDER OF DECREASING CD ENERGY, ES(I).GT.ES(I+1) CD DEL. DOPPLER WIDTH/(SQUARE ROOT OF THE TEMPERATURE) CD С _____ _____ C---------C------STATISTICAL UNRESOLVED RESONANCE INFORMATION (TYPE 5) CR С CC ALWAYS PRESENT С (((GA(I,J,L),I=1,NPT),J=1,JST),L=1,LST),CL 1(((D(T,J,L), I=1, NPT), J=1, JST), L=1, LST),CL 2(((GF(I,J,L),I=1,NPT),J=1,JST),L=1,LST), CL 3(((GNO(I,J,L),I=1,NPT),J=1,JST),L=1,LST),CL 4((G(J,L), J=1, JST), L=1, LST), ((NDFF(J,L), J=1, JST), L=1, LST),CL CL 5((NDEN(J,L), J=1, JST), L=1, LST)С CIJ (4*NPT+3) * (JST (1) +JST (2) +...+JST (LST)) С CD GA AVERAGE RADIATION WIDTH, EV., FOR EACH ENERGY ES CD D AVERAGE SPACING, EV., FOR EACH ENERGY ES CD GF AVERAGE FISSION WIDTH, EV., FCR EACH ENERGY ES CD AVERAGE REDUCED NEUTRON WIDTH, EV., CD GNO CD FOR EACH ENERGY ES NUMBER OF DEGREES OF FREEDOM IN FISSION WIDTH NDFF CD DISTRIBUTION (1, 2, 3 OR 4)CD STATISTICAL FACTOR CD G NDFN NUMBER OF DEGREES OF FREEDOM IN NEUTRON WIDTH CD DISTRIBUTION (1 OR 2). NUMBER OF ENTRANCE CD CHANNELS FOR NEUTRONS IN AN L, J SEQUENCE CD С C-

CEOF

-304-

APPENDIX C. MC²-2 Binary Interface Files. MCC2F4

****'	*****	* * * * * * * * * * * * *	****	****	*****	*****	***-
C							-
C		,	PREPAREI	5 1/13/76	AT ANL		-
		Maabu					-
CF		MCCZF4					-
C E		RESULVED RE	SUNANCE L	DATA			-
C * * * * *	******	****	* * * * * * * * * *	****	* * * * * * * * * *	***	-
C + + + + + +	• • • • • • • •	* * * * * * * * * * * * * * *	° ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	• • • • • • • • • • • •	• • • • • • • • • • • •	****	****
CD	GTT.GT	FLGTC	RESPECTI	VELY, TH	E ASYMMET	RTCAL ADLER-ADLER	
CD		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	TOTAL. I	FISSION.	AND CAPTU	RE CROSS SECTION	
CD			PARAMETE	ERS SUPPL	TED BY EN	DF/B	
CD	GRT, GR	FGRC	RESPECTI	VELY, TH	E SYMMETP	ICAL ADLER-ADLER	
CD	•	•	TOTAL. H	ISSION.	AND CAPTU	RE CROSS SECTION	
CD			PARAMETE	ERS SUPPL	IED BY EN	DF/B	
CD	IVR		RESOLVED	RESONAN	CE PARAME	TER INDEX	
CD			IWR=1 M	ATERIAL	HAS RESOL	VED RESONANCE	
CD			S	SINGLE-LE	VEL BREIT	-WIGNER PARAMETERS	
CD			IWR=2 M	ATERIAL	HAS MULTI	-LEVEL ADLER-ADLER	
CD			F	RESOLVED	RESONANCE	PARAMETERS	
CD			IWR=3 M	ATERIAL	HAS MULTI	-LEVEL BREIT-WIGNER	
CD			F	RESOLVED	RESONANCE	PARAMETERS	
CD	MULT		2 FOR IE	BM MACHINI	ES, 1 OTH	ERWISE	
CD	NTSO		NUMBER C	DF ISOTOPE	ES IN MAT	ERIAL	
CD	NRESMT	1	NUMBER C	OF MATERI	ALS WITH	RESOLVED RESONANCE	
CD			PARAMETE	ERS AS SPI	ECIFIED I	N THE ADMINISTRATIVE	2
CD			FILE MCC	2F1			
CD	NRGYS		NUMBER C	OF RESOLVI	ED RESONA	NCES FOR AN ISOTOPE	
CD	PHI		S WAVE P	PHASE SHII	FT EQUAL	TO K*R. THE NEUTRON	
CD		·	WAVE NUM	IBER K=2.	196771E-3	* (A/ (A+1)) *SQUARE	
CD			ROOT OF	RESONANC	E ENERGY,	A IS THE RATIO OF	_
CD			THE MASS	OF THE I	PARTICULA	R ISOTOPE TO THAT OF	
CD			THE NEUT	RON, AND	R IS THE	EFFECTIVE SCATTERIN	IG
CD			RADIUS.	AANDRA	ARE SUPPL	TED BY ENDENE	
C							
cs		FILE STRUCT	JRE				-
CS							-
CS		RECORD TYPE				PRESENT IF	-
CS		==============			******		= = -
CS		RESOLVED RES	SUNANCE M	ATERIAL 1	NAMES	ALWAYS	-
CS	نه به به به ماد ماد	MATERIAL SPI	SCIFICATI	UNS		ALWAYS	-
CS	*****	** (REPEAT]	FOR ALL M	ATERIALS	WITH		-
CS	ж ж	RESOLVEI	D RESONAN	ICE PARAMI	STERS)	A T 11 A V C	-
CS CC	* *	ISUTOPE CONT	KOL	COMODEC		ALWAIS	-
	~ ~ ~ ~ ~ ~ ~ ~	TT (REPEAT I	OK ALL I	SUTUPES)	n a	THD CP 1	-
US	* *	WESOFAED KES	SUNANCE E	NERGY DAT	L' P	TAK • (2D • 1	-

•

* * RESOLVED RESONANCE PEAK CROSS

,

25

•

.

-

٠

IWR.GE.1 .

-305-

~ c	*	*	SECTION			
(.) CC	*	*	MODIFIED RESONANCE NATURAL TO	TWR.GE.1		
	т •	т #	NODIED WIDTH			
	≁ ≁	т •¥	THERE ALDIN THEREFORE CATTRETIC FACTOR	TWR.GE.1		
	т т	*	THE PREME SCATTERING TROTOR	TWR.EO.1		
	Ŧ *	*	NEGRON LINE WIDTHS			
	т 	т 	DECONNER DIDING THE WIDTH	TWR.EC.1		
	* 	<u>т</u>	RESONANCE RESERVE LINE WIDTH	TWR, EQ. 1		
CS	*	*	RESUMANCE FISSION LINE RIDIN			
CS	*	*	S-MATRIX TOTAL LINE WIDIN GUMMENDER DARAMENTE ROD CADMURE			
CS	∓	*	SYMMETRIC PARAMETER FOR CAPIORE	IND. EV. 2		
CS ~~	Ť	*	YEACTION			
CS	*	-7	SYMMETRIC PARAMETER FOR FISSION	INN.LQ.S		
CS	*	*	REACTION			
CS	*	*	SYMMETRIC PARAMETER FOR TOTAL	IWR. LQ. 2		
CS	*	*	REACTION			
CS	*	*	ADLER-ADLER FACTOR FOR J CAPTURE	IWF.EQ.2		
CS	*	*	INTEGRAL	TUD DO 3		
CS	*	*	ADLER-ADLER FACTOR FOR J FISSION	IWR.EQ.2		
CS	*.	×	INTEGRAL			
CS	*	*	BREIT-WIGNER MULTI-LEVEL TOTAL LINE	IWR.CQ.J		
CS	*	*	WIDTH			
CS	*	*	BREIT-WIGNER MULTI-LEVEL SYMMETRIC	IWE . EQ. 5		
CS	*	*	PARAMETER FOR CAPTURE REACTION			
CS	*	*	BRUIT-VIGNER MULTI-LEVEL SYMMETRIC	IWR.EQ.5		
CS	*	*	PARAMETER FOR FISSION REACTION			
CS	*	*	BREIT-WIGNER MULTI-LEVEL SYMMETRIC	$\mathbf{T} \mathbf{W} \mathbf{K} \bullet \mathbf{F} \mathbf{Q} \bullet \mathbf{S}$		
CS	*	*	PARAMETER FOR TOTAL REACTION			
CS	*	*	BREIT-WIGNER MULTI-LEVEL RELATIVE	1 % R. 5Q. 5		
CS	*	*	ASYMMETRIC CONTRIBUTION TO CAPTURE			
CS	*	*	REACTION	THE DO 3		
CS	*	*	BREIT-WIGNER MULTI-LEVEL RELATIVE	IWR.EQ.5		
CS	*	*	ASYMMETRIC CONTRIBUTION TO FISSION			
CS	*	*	PEACTION			
CS	* * *	***	< * *			
С						
C						
CR			PESOLVED RESONANCE MATERIAL NAMES (TYP	?E 1)		
CC CC			ALVAYS PRESENT			
CI	(N A	ME ((I), I = 1, NRESMT)			
CW C	MULT*NRESMT					
10 0	NAM	1 E	DOUBLE PRECISION (R*8) MAY	TERIAL IDENTIFICATION		
C						

.
<u> </u>	•					
C		MATERIAL SPECIFICATIONS (TYPE 2)				
C CC		ALWAYS PRESENT				
C CL CL	(NISO 1 (EU (I	<pre>(I),I=1,NRESMT),(IWR(I),I=1,NRESMT),(EL(I),I=1,NRESMT), ,I=1,NRESMT)</pre>				
C N	4 * N R E	5 M T				
CD	ΕL	LOWEST ENERGY FOR WHICH RESONANCE PARAMETERS				
CD CD CD C	EIJ	APPLY FOR ANY ISOTOPE OF EACH MATERIAL HIGHEST ENERGY FOR WHICH RESONANCE PARAMETERS - APPLY FOR ANY ISOTOPE OF EACH MATERIAL -				
C						
C CE		ISOTOPE CONTROL DATA (TYPE 3)				
cc		ALVAYS PRESENT				
CL	(АВИМ	JUN(I), I=1, NISO), (NRGYS(I), I=1, NISO)				
C W	2*NTS	D				
CD CD C	ABUN	ABUNDANCE OF EACH ISOTOPE				
C						
C CR		RESOLVED RESONANCE ENERGY DATA (TYPE 4)				
cc cc		PRESENT IF IWR.GE.1				
CL	(FN (J)	, J=1, NRGYS)				
CW CW	NRGYS					
CD CD C	E N	RESONANCE ENERGY EN(J).GT.EN(J+1)				
~						
R		RESOLVED RESONANCE PEAK CROSS SECTION (TYPE 5) -				

-307-

•		•
	PRESENT IF	IWR.GE.1
(SIGO	(J), J = 1, NRGY	YS)
NRGYS	•	
SIGO		CROSS SECTION AT RESONANCE
		FOR IWR.EQ.1, SIGO IS 4*PI*G*LAMBDA-BAR SQUA *((GAMT-GAMGAM-GAF)/GAMT)*((A+1)/A) SQUARED. (SEE RECORD TYPES 8,9, AND 10)
		FOR IWR.EQ.2, SIGO IS 4*PI*LAMBDA-BAR SQUARD ((A+1)/A) SQUARED*ABS(GT)/(2*GAMS). (SEE RECORD TYPES 11 AND 14)
		FOR IWR.EQ.3, SIGO IS THE VALUE COMPUTED FOR IWR.EQ.1 MULTIPLIED BY GT. (SEE RECORD TYPE 20)
	RATIO OF RI	ESONANCE NATURAL TO DOPPLER WIDTH (TYPE 6)
	RATIO OF RI PRESENT IF	ESONANCE NATURAL TO DOPPLER WIDTH (TYPE 6) IWR.GE.1
(TH E T.	RATIO OF R PRESENT IF AP(J),J=1,N	ESONANCE NATURAL TO DOPPLER WIDTH (TYPE 6) IWR.GE.1 RGYS)
(THET. NRGYS	RATIO OF RI PRESENT IF AP(J),J=1,NI	ESONANCE NATURAL TO DOPPLER WIDTH (TYPE 6) IWR.GE.1 RGYS)
(THET NRGYS THETA	RATIO OF RI PRESENT IF AP(J),J=1,NP	ESONANCE NATURAL TO DOPPLER WIDTH (TYPE 6) IWR.GE.1 RGYS) RATIO OF NATURAL WIDTH TO DOPPLER WIDTH * SQRT (TEMPERATURE) IF IWR.EQ.1 OR IWR.EQ.3. RATIO OF S-MATRIX TOTAL LINE WIDTH TO DOPPLY WIDTH*SQRT (TEMPERATURE) IF IWR.EQ.2
(THET NRGYS THETA	RATIO OF R PRESENT IF AP(J),J=1,N P	ESONANCE NATURAL TO DOPPLER WIDTH (TYPE 6) IWR.GE.1 RGYS) RATIO OF NATURAL WIDTH TO DOPPLER WIDTH * SQRT (TEMPERATURE) IF IWK.EQ.1 OR IWR.EQ.3. RATIO OF S-MATRIX TOTAL LINE WIDTH TO DOPPL WIDTH*SQRT (TEMPERATURE) IF IWK.EQ.2
(THET NRGYS THETA	RATIO OF RI PRESENT IF AP(J),J=1,N P INTERFEREN PRESENT IF	ESONANCE NATURAL TO DOPPLER WIDTH (TYPE 6) IWR.GE.1 RGYS) RATIO OF NATURAL WIDTH TO DOPPLER WIDTH * SQRT (TEMPERATURE) IF IWE.EQ.1 OR IWR.EQ.3. RATIO OF S-MATRIX TOTAL LINE WIDTH TO DOPPLY WIDTH*SQRT (TEMPERATURE) IF IWE.EQ.2 CE SCATTERING FACTOR (TYPE 7) IWR.GE.1
(THET) NRGYS THETA	RATIO OF REPRESENT IF AP(J),J=1,NE P INTERFERENC PRESENT IF (J),J=1,NRG	ESONANCE NATURAL TO DOPPLER WIDTH (TYPE 6) IWR.GE.1 RGYS) RATIO OF NATURAL WIDTH TO DOPPLER WIDTH * SQRT (TEMPERATURE) IF IWE.EQ.1 OR IWR.EQ.3. RATIO OF S-MATRIX TOTAL LINE WIDTH TO DOPPLE WIDTH*SQRT (TEMPERATURE) IF IWE.EQ.2 CE SCATTERING FACTOR (TYPE 7) IWR.GE.1 YS)

-308-

CD CD	AFAC	FACTOR TO MULTIPLY CHI TO OBTAIN INTERFERENCE - SCATTERING -
CN CN CN CN CN CN		FOR IWR.EQ.1, AFAC IS THE SQUARE ROOT OF (G*((GAMT-GAMGAM-GAF)/GAMT)* ATOM POTENTIAL SCATTERING CROSS SECTION /SIGO). G, THE STATISTICAL FACTOR, IS (2J+1)/(4I+2) WHERE J IS THE SPIN OF THE COMPOUND NUCLEUS RESONANCE AND I IS THE TARGET NUCLEUS SPIN.
CN CN CN		AFAC IS SET TO 0 FOR P AND D WAVE RESONANCES -
CN CN		FOR IWR.EQ.2, AFAC IS -0.5*(GIT*COS(2*PHI) GRT*SIN(2*PHI))/(GRT*COS(2*PHI)+GIT*SIN(2*PHI))
CN CN CN CN CN C C C		FOR IWR.EQ.3, AFAC IS THE VALUE COMPUTED FOR IWR.EQ.1 PLUS THE ASYMMETRIC LEVEL-LEVEL INTERFERENCE CONTRIBUTION, ALL DIVIDED BY GT. (SEE RECORD TYPE 20) AFAC IS SET TO 0 FOR P AND D WAVE RESONANCES
C		
C R C	RESOLVED RE	SONANCE TOTAL AND NEUTRON LINE WIDTHS (TYPE 8) -
CC C	PRESENT IF	IWR.EQ.1 -
CL C	(GAMT(J), J=1, NRGY)	S), (GAMN (J), J=1, NRGYS) -
CW C	2*NRGYS	-
CD CD C	GAMT GAMN	RESOLVED RESONANCE TOTAL LINF WIDTH - RESOLVED RESONANCE NEUTRON LÎNE WIDTH -
C		· · · · · · · · · · · · · · · · · · ·
C CR	RESONANCE R	ADIATION LINE WIDTH (TYPE 9)
cc c	PRESENT IF	IWR.EQ.1 -
CL C	(GAMGAM(J), J=1, NR)	
CW C	NRGYS	-
Ω	самсам	

-309-

_ _ _ _ _ _ _ _ _ _ _ _ _ _ _ _ _ _ _ C-CR RESONANCE FISSION LINE WIDTH (TYPE 10) С СС PRESENT IF IWR.EQ.1 С CL (GAF(J), J=1, NRGYS)С СЫ NRGYS C RESOLVED RESONANCE FISSION LINE WIDTH CD GAF С C _____ С-S-MATRIX TOTAL LINE WIDTH (TYPE 11) CR С СС PRESENT IF IWR.EO.2 С (GAMS(J), J=1, NRGYS)CL С C VNRGYS С GAMS S-MATRIX TOTAL LINE WIDTH FOR ADLER-ADLER CD CD FORMULATION (.EQ.2.0*NU) С _____ C - -CF SYMMETRIC PARAMETER FOR CAPTURE REACTION (TYPE 12) С CC PRESENT IF INR.EQ.2 С CL (GC(J), J=1, NRGYS)С CK NRGYS С SYMMETRIC PARAMETER FOR CAPTURE REACTION CD GC С CN GC = (GRC*COS(2*PHI)+GIC*SIN(2*PHI)) * SQUARE ROOT OF THE RESONANCE ENERGY CN С

-310-

CR		SYMMETRIC PARAMETER FOR FISSION REACTION (TYPE 13) -
C CC		PRESENT IF IWR.EQ.2
	(GF (J	- , J=1, NRGYS)
C CW	NRGYS	- - -
CD	GF	SYMMETRIC PARAMETER FOR FISSION REACTION -
CN CN C		GF = (GRF*COS(2*PHI)+GIF*SIN(2*PHI)) * SOUARE - ROOT OF THE RESONANCE ENERGY -
C		
C CR		SYMMETRIC PARAMETER FOR TOTAL REACTION (TYPE 14) -
C CC		PRESENT IF IWR.EQ.2
CL	(GT (J)	- , J= 1 , NRGYS) -
C W	NRGYS	-
C D	GT	SYMMETRIC PARAMETER FOR TOTAL REACTION -
CN CN C		GT = (GRT*COS(2*PHI)+GIT*SIN(2*PHI)) * SQUARE - ROOT OF THE RESONANCE ENERGY -
C		· · · · · · · · · · · · · · · · · · ·
CR		ADLER-ADLER FACTOR FOR J CAPTURE INTEGRAL (TYPE 15) -
cc C		PRESENT IF IWR.EQ.2
CL	(BC (J)	, J=1, NRGYS)
С И С	NRGYS	· -
	BC	-0.5*(GIC*COS(2*PHI)-GRC*SIN(2*PHI)) * SQUARE - ROOT OF THE RESONANCE ENERGY/GC, WHERE GC IS - DEFINED IN RECORD TYPE 12 -

-311-

APPENDIX C. MC²-2 Binary Interface Files. MCC2F4 (Contd.) CR ADLER-ADLER FACTOR FOR J FISSION INTEGRAL (TYPE 16) С СС PRESENT IF IWR.EQ.2 С CL (BF(J), J=1, NRGYS)С CW NRGYS С CD -0.5*(GIF*COS(2*PHI) - GRF*SIN(2*PHI)) * SQUARE BF · ROOT OF THE RESONANCE ENEPGY/GF, WHERE GF IS CD CD . DEFINED IN RECORD TYPE 13 С CNBF IS SET TO ZERO FOR UNFISSIONABLE CN MULTI-LEVEL MATERIALS С C -_____ C-BREIT-WIGNER MULTI-LEVEL TOTAL LINE WIDTH (TYPE 17) CP С CC PRESENT IF IWR.EO.3 С CL (GAMS(J), J=1, NRGYS)С CW NRGYS С CD GAMS BREIT-WIGNER MULTI-LEVEL TOTAL LINE WIDTH С C-_____ С-BREIT-VIGNER MULTI-LEVEL SYMMETRIC PARAMETER FOR CR CR CAPTURE REACTION (TYPE 18) С PRESENT IF IWR.EQ.3 CC С CL (GC (J), J=1, NRGYS)С CW NRGYS C_{-} CD BREIT-WIGNER MULTI-LEVEL SYMMETRIC PARAMETER GC CD FOR CAPTURE REACTION, THE RATIO OF GAMGAM TO CD GAMS. (SEE RECORD TYPE 9 AND 17) С C-

APPENDIX C. MC²-2 Binary Interface Files. MCC2F4 (Contd.) CR BREIT-WIGNER MULTI-LEVEL SYMMETRIC PARAMETER FOR CR FISSION REACTION (TYPE 19) С CC PRESENT IF IWR.EQ.3 С CL (GF(J), J=1, NRGYS)С CW NRGYS С CD GF BREIT-WIGNER MULTI-LEVEL SYMMETRIC PARAMETER FOR FISSION REACTION, THE RATIO OF GAF TO GAMS. -СD (SEE RECORD TYPE 10 AND 17) CD С _____ C --_____ CR BREIT-NIGNER MULTI-LEVEL SYMMETRIC PARAMETER FOR CR TOTAL REACTION (TYPE 20) С СС PRESENT IF IWR.EQ.3 С (GT(J), J=1, NRGYS)CL С CW NEGYS С GΤ BREIT-WIGNER MULTI-LEVEL SYMMETRIC PARAMETER CD CD FOR TOTAL REACTION, 1 PLUS THE SYMMETRIC LEVEL-LEVEL INTERFERENCE CONTRIBUTION CD С C-_____ C---BREIT-WIGNER MULTI-LEVEL RELATIVE ASYMMETRIC CONTRIBUTION CR CR TO CAPTURE REACTION (TYPE 21) С PRESENT IF IWR.EQ.3 CC C (BC(J), J=1, NRGYS)CL С NRGYS CW С BREIT-WIGNER MULTI-LEVEL RELATIVE ASYMMETRIC BC CD CONTRIBUTION TO CAPTURE REACTION CD С NOTE THAT BC IS IDENTICALLY 0.0 FOR BREIT-CN WIGNER MULTI-LEVEL FORMALISM CN

-313-

1

_____ C-BREIT-WIGNER MULTI-LEVEL RELATIVE ASYMMETRIC CONTRIBUTION CR TO FISSION REACTION (TYPE 22) CR С PRESENT IF INR. EQ. 3 CC С CL (BF(J), J=1, NRGYS)С NEGYS CW С BREIT-WIGNER MULTI-LEVEL RELATIVE ASYMMETRIC BF CD CD CONTRIBUTION TO FISSION REACTION С NOTE THAT BF IS IDENTICALLY 0.0 FOR BREIT-CN WIGNER MULTI-LEVEL FORMALISM CN С C -

CEOF

C**** C	*****	* * ** * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * *
с с		PREPARED 2/11/75 AT ANL	-
CF		MCC2F5	-
CE		SMOOTH (TABULATED) NON-RESONANT DATA	-
C			ست. ماه ماه ماه باه باه ماه ماه ماه باه ماه ماه ماه ماه باه ماه باه ماه باه ماه ماه باه ماه باه ماه باه باه .
C****	*****	* * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *
CD	MULT	2 FOR IBM MACHINES, 1 OTHERWI	SE
CD	NREAC	NREAC=NUMBER OF REACTION TYPE	S FOR A
CD ·		GIVEN MATERIAL	1
·			
C		FILE STRUCTURE	
CS			-
CS		RECORD TYPE PRF	SENT IF -
CS	*****		
CS CS	*****	MATERIAL DA ALL HALLALS)	AYS -
CS	*	SPECIFICATIONS ALW	AYS -
CS	* ***	*** (REPEAT FOR NREAC REACTION TYPES)	-
CS	* *	CROSS SECTIONS NRE	AC.GT.0 -
CS	*****	* * *	-
С			-
C			
C	·	MATTRIAL NAME (TYDE 1)	
C A		HRISAIRD WRID (ITED I)	_
cc		ALWAYS PRESENT	-
C	•		-
CF	NAME	·	-
С			-
CW	MULT*1	1	· -
C			-
CD	NAME	DOUBLE PRECISION (R*O) MAIRFI	AL IDENTIFICATION -
C			
C			
C			
CR		SPECIFICATIONS (TYPE 2)	. –
C		· , ,	-
cc		ALVAYS PRESENT	-
C	NDDAG	ע שטשע גנים זגע כיניע כעע מא איי אנים א דמע	-
ىد	NESAC,	, NEL, NE, NGAM, NE, ND, NGS, NGES, NALEDA, NTOT, AU	, R + , R 4 , R 3 -

C			-
CW	14		-
C	•		-
CD	NEL	FLAG FOR ELASTIC CROSS SECTIONS	-
CD	NF	FLAG FOR FISSION CROSS SECTIONS	7
CD	NGAM	FLAG FOR (N,GAMMA) CROSS SECTIONS	-
CD	NP .	FLAG FOR (N,P) CROSS SECTIONS	-
CD 🕚	ND	FLAG FOR (N, D) CROSS SECTIONS	-
CD	NH3	FLAG FOR (N,H3) CROSS SECTIONS	-
CD	NHEJ	FLAC FOR (N, HE3) CROSS SECTIONS	-
CD		FLAG FOR (N,ALPHA) CROSS SECTIONS	-
CD CD	NTOT	FLAG FOR TOTAL CROSS SECTIONS	_
CD		FOR FACE OF THE ABOVE PEACTION TYDES THE	_
		FINCE ARE TREATE NO CROSS SECTION DATA ARE	_
CD		GIVEN AND EQUAL TO THE HIGHEST ENERGY GROUP	-
CD		NUMBER (LOWEST ENERGY) FOR WHICH CROSS SECTION	-
CD		DATA ARE GIVEN	-
CD			-
CD	A0,A1,A2,A3	COEFFICIENTS FOR CALCULATION OF NU(E), THE	-
CD		NUMBER OF NEUTRONS PER FISSION. IF THE	-
CD		ISOTOPE IS NOT FISSIONABLE, THEN THE	-
CD		COEFFICIENTS WILL BE SET TO ZERO	-
С			-
С	CROSS SECTI		
C	CROSS SDOIL		_
cc	PRESENT IF	NREAC.GT.O	-
CL.	$(XSTG(J) \rightarrow J=1 - NLAS)$	ሞ)	_
c		-,	-
CW	NGROUP		
С			-
	•		-
CD	XSIG	CROSS SECTION ASSOCIATED WITH ULTRAFINE	- - -
CD CD	XSIG	CROSS SECTION ASSOCIATED WITH ULTRAFINE GROUP J FOR A GIVEN REACTION TYPE	
CD CD CD	XSIG NLAST	CROSS SECTION ASSOCIATED WITH ULTRAFINE GROUP J FOR A GIVEN REACTION TYPE FLAG FOR REACTION TYPE IN QUESTION. (THAT	
CD CD CD CD	XSIG NLAST	CROSS SECTION ASSOCIATED WITH ULTRAFINE GROUP J FOR A GIVEN REACTION TYPE FLAG FOR REACTION TYPE IN QUESTION. (THAT IS, NEL,NF,ETC.)	
CD CD CD CD CD	XSIG NLAST NGROUP	CROSS SECTION ASSOCIATED WITH ULTRAFINE GROUP J FOR A GIVEN REACTION TYPE FLAG FOR REACTION TYPE IN QUESTION. (THAT IS, NEL,NF,ETC.) NUMBER OF ENERGY GROUPS IN LIBRARY	
CD CD CD CD CD CD CD	XSIG NLAST NGROUP	CROSS SECTION ASSOCIATED WITH ULTRAFINE GROUP J FOR A GIVEN REACTION TYPE FLAG FOR REACTION TYPE IN QUESTION. (THAT IS, NEL,NF,ETC.) NUMBER OF ENERGY GROUPS IN LIBRARY	
CD CD CD CD CD CD C CN	XSIG NLAST NGROUP	CROSS SECTION ASSOCIATED WITH ULTRAFINE GROUP J FOR A GIVEN REACTION TYPE FLAG FOR REACTION TYPE IN QUESTION. (THAT IS, NEL,NF,ETC.) NUMBER OF ENERGY GROUPS IN LIBRARY THE CROSS SECTION RECORDS FOR THE DIFFERENT DEACTION TYPES WHICH ARE DEESENT WILL BE IN THE	
CD CD CD CD CD CD CN CN	XSIG NLAST NGROUP	CROSS SECTION ASSOCIATED WITH ULTRAFINE GROUP J FOR A GIVEN REACTION TYPE FLAG FOR REACTION TYPE IN QUESTION. (THAT IS, NEL,NF,ETC.) NUMBER OF ENERGY GROUPS IN LIBRARY THE CROSS SECTION RECORDS FOR THE DIFFERENT REACTION TYPES WHICH ARE PRESENT WILL BE IN THE FOLLOWING ORDER: FLASTIC FISSION (N GAMMA)	
CD CD CD CD CD CD CN CN CN CN	XSIG NLAST NGROUP	CROSS SECTION ASSOCIATED WITH ULTRAFINE GROUP J FOR A GIVEN REACTION TYPE FLAG FOR REACTION TYPE IN QUESTION. (THAT IS, NEL,NF,ETC.) NUMBER OF ENERGY GROUPS IN LIBRARY THE CROSS SECTION RECORDS FOR THE DIFFERENT REACTION TYPES WHICH ARE PRESENT WILL BE IN THE FOLLOWING ORDER: ELASTIC, FISSION, (N, GAMMA), (N, P), (N, D), (N, H3), (N, HE3), (N, AIPHA),	
CD CD CD CD CD CD CN CN CN CN CN	XSIG NLAST NGROUP	CROSS SECTION ASSOCIATED WITH ULTRAFINE GROUP J FOR A GIVEN REACTION TYPE FLAG FOR REACTION TYPE IN QUESTION. (THAT IS, NEL,NF,ETC.) NUMBER OF ENERGY GROUPS IN LIBRARY THE CROSS SECTION RECORDS FOR THE DIFFERENT REACTION TYPES WHICH ARE PRESENT WILL BE IN THE FOLLOWING ORDER: ELASTIC,FISSION, (N,GAMMA), (N,P), (N,D), (N,H3), (N,HE3), (N,AIPHA), AND TOTAL.	
CD CD CD CD CD CD CN CN CN CN CN CN	XSIG NLAST NGROUP	CROSS SECTION ASSOCIATED WITH ULTRAFINE GROUP J FOR A GIVEN REACTION TYPE FLAG FOR REACTION TYPE IN QUESTION. (THAT IS, NEL,NF,ETC.) NUMBER OF ENERGY GROUPS IN LIBRARY THE CROSS SECTION RECORDS FOR THE DIFFERENT REACTION TYPES WHICH ARE PRESENT WILL BE IN THE FOLLOWING ORDER: ELASTIC,FISSION, (N,GAMMA), (N,P), (N,D), (N,H3), (N,HE3), (N,ALPHA), AND TOTAL. THE TOTAL CROSS SECTION FOR ULTRA-	
CD CD CD CD CD CN CN CN CN CN CN CN	XSIG NLAST NGROUP	CROSS SECTION ASSOCIATED WITH ULTRAFINE GROUP J FOR A GIVEN REACTION TYPE FLAG FOR REACTION TYPE IN QUESTION. (THAT IS, NEL,NF,ETC.) NUMBER OF ENERGY GROUPS IN LIBRARY THE CROSS SECTION RECORDS FOR THE DIFFERENT REACTION TYPES WHICH ARE PRESENT WILL BE IN THE FOLLOWING ORDER: ELASTIC,FISSION, (N,GAMMA), (N,P), (N,D), (N,H3), (N,HE3), (N,AIPHA), AND TOTAL. THE TOTAL CROSS SECTION FOR ULTRA- FINE GROUP J =ELASTIC SCATTERING+INELASTIC	

.

-316-

CN	(N,P) + (N,D) + (N,H3) + (N,HE3) + (N,ALPHA). THE	-
CN	ELASTIC SCATTERING=SMOOTH (TABULATED) ELASTIC	-
CN	SCATTERING+UNRESOLVED RESONANCE SCATTERING+	-
CN	RESOLVED RESONANCE SCATTERING. THE ELASTIC	-
CN	SCATTERING DOES NOT INCLUDE THE UNRESOLVED	_
CN	RESONANCE SCATTERING AND RESOLVED RESONANCE	-
CN	SCATTERING CONTRIBUTIONS FOR MATERIALS WITH	-
CN	'HEAVY' MASSES. A 'HEAVY' MASS MATERIAL IS	-
CN	DESIGNATED BY THE USER. ELASTIC SCATTERING	-
CN	EXPLICITLY INCLUDES THE POTENTIAL SCATTERING	-
CN ·	CROSS SECTION	-
CN		-
CN	ONE CROSS SECTION RECORD IS PRESENT FOR EACH	-
CN	NON-ZERO FLAG	-
С		-
C		

CEOF

.

.

C * * * * * :	****	* * * * * * * * * * * * * * * * * * * *
С	·	
С		PREPARED 10/27/75 AT ANL -
С		· •
CF	MCC2F6	-
CE	INELASTIC A	ND N2N DISTRIBUTIONS -
C ·	· .	-
C * * * * *	* * * * * * * * * * * * * * * * * * * *	*****
,		
CD ·	MMAT .	NUMBER OF MATERIALS IN THE LIBRAPY WITH
CD		INELASTIC AND/OR (N,2N) DATA AS SPECIFIED IN
CD		THE ADMINISTRATIVE FILE MCC2F1
CD	MMAT1	NUMBER OF MATERIALS THAT HAVE INELASTIC AND/OR
CD		(N,2N) DATA FOR THE ENERGY GROUP BEING
CD		PROCESSED. IF THE VALUE OF NINEL OR N2NTH
CD		FOR A MATERIAL IS LESS THAN THE ENERGY GROUP
CDÍ		NUMBER BEING PROCESSED, THEN NO DATA ARE
CD		PRESENT FOR THIS MATERIAL. THE CONTROL
CD		INFORMATION NINEL AND N2NTH ARE SPECIFIED IN
CD		RECORD TYPE 5 OF MCC2F1
CD	NEND 1	NUMBER OF LAST SINK ENERGY POINT (EIN) FOR
CD		WHICH THE TABULATED INELASTIC PPOBABILITY
CD		TS NON-ZERO FOR THE SOURCE GROUP IN OUESTION
CD	N FN D 2	NUMBER OF LAST SINK ENERGY POINT (EN2N) FOR
		WHICH THE TABILATED (N.2N) PROBABILITY
CD		TS NON-ZERO FOR THE SOURCE GROUP IN OUESTION
CD	ΝΤΝΕΥΡ	NUMBER OF INFLASTIC FUNDORATION SPECTRA
CD	EN LE CA ELS VIET	FOR ENERGY GROUD IN OUFSTION STRETCH
CD	ΝΤΝΟΛΡ	NUMBER OF TABULATED INFLASTIC DISTRIBUTIONS
CD	N.C.WI HD	ROADER OF TROUBRIED INCLERISTIC DESTRIBUTIONS
CD		LOU PUPULAT ANOLE IN COPILION
CD	NTUC	ΝΙΝΙΚΟ-ΕΥ-ΔΕΚΟ ΟΚ ΟΝΕ ΝΙΝΡΕΊ ΟΓ ΠΙΚΟΙΡΊΓ ΙΝΕΙΛΟΎΤΟ ΚΟΛΠΠΈΡΙΝΟ ΙΕΎΓΙς
	NLV S	NUMBER OF DISCRETE INCLASTIC SCRITTERING LEVELS
CD CD		FUR EACH MATERIAL FUR THE ENERGY GROUP IN
CD		QUESTION
CD-	N ĽI A X	MAXIMUM NUMBER OF GROUPS OF INELASILO OR
CD		(N, ZN) DATA FOR ANY MATERIAL IN THE LIBRARY
CD		(.EQ.MAX(NINEL,NZNTH) AS SPECIFIED IN THE
CD		ADMINISTRATIVE FILE MCC2F4)
CD	NSINKI	NUMBER OF ENERGIES PROVIDED IN TABULATED
CD		INELASTIC DISTRIBUTIONS FOR EACH MATERIAL
CD ·	_	AS SPECIFIED IN THE ADMINISTRATIVE FILE MCC2F1
CD	NSINK2	NUMBER OF ENERGIES PROVIDED IN TABULATED
CD	· .	(N, 2N) DISTRIBUTIONS FOR EACH MATERIAL
CD		AS SPECIFIED IN THE ADMINISTRATIVE FILE MCC2F1
CD	NSTRT1	NUMBER OF FIRST SINK ENERGY POINT (EIN) FOR
CD		WHICH THE TABULATED INELASTIC PROBABILITY
CD		IS NON-ZERO FOR THE SOURCE GROUP IN QUESTION
CD	NSTRT2	NUMBER OF FIRST SINK ENERGY POINT (EN 2N) FOR
CD		WHICH THE TABULATED (N, 2N) PROBABILITY

ς.

.

CD		IS NON-ZERO FOR THE SOURCE GROUP IN QUESTION
CD	NZINEV P	NUMBER OF (N.ZN) EVAPORATION SPECTRA
CD		FOR ENERGY GROUP IN QUESTION
CD	N2NLV	NUMBER OF DISCRETE (N,2N) SCATTERING LEVELS
CD		FOR EACH MATERIAL FOR THE ENERGY GROUP IN
CD		QUESTION
CD	N 2N T A B	NUMBER OF TABULATED (N,2N) DISTRIBUTIONS
CD		FOR ENERGY GROUP IN QUESTION
CD		N2NTAB.EQ.ZERO OR ONE

C------CS FILE STRUCTURE CS -CS RECORD TYPE PRESENT IF -CS 👘 =============================== CS GROUP INDEPENDENT INELASTIC MMAT.GT.O CS AND (N,2N) DATA ******* (REPEAT FOR NMAX GROUPS) CS CONTROL INFORMATION AND CROSS CS * NMAX.GT.O CS SECTION DATA * CS * SECONDARY DISTRIBUTION DATA (NLVS.GT.O) OP _ CS (N2NLV.GT.O) OR * CS ׿. (NINEVP.GT.O) OR CS * (N2NEVP.GT.O) CS * FOR ANY MATERIAL CS TABULATED INELASTIC AND (N,2N) * (NINTAB.GT. 0) OR CS * PROBABILITIES (N2NTAB.GT. 0) CS * FOR ANY MATERIAL CS *** * * * * * С _____ C----

C	
CR	GROUP INDEPENDENT INELASTIC AND (N,2N) DATA (TYPE 1) -
cc	PRESENT IF MMAT.GT.O -
С	
CL	((EGAM(I,J), I=1, NLSJ), (U(I,J), I=1, MAX1J), -
CL	1 (GAMN2N(I, J), I=1, N2NLJ), (EIN(I, J), I=1, NSNK1J), -
CL	2(EN2N(I,J), I=1, NSNK2J), J=1, MMAT) -
С	· –
CW	SUMJ -
С	· · · · · · · · · · · · · · · · · · ·
CD	EGAM GAMMA RAY ENERGY ASSOCIATED WITH THE L'TH -
CD	RESOLVED SCATTERING LEVEL. IF Q IS THE -
CD	ASSOCIATED Q-VALUE OF THE REACTION, THEN -
CD	EGAM = -Q (EGAM.GT.0) -
CD	EGAM IS ORDERED SUCH THAT EGAM(1).LE.EGAM(2)

-319-

CD	U	DEFINES INELASTIC UPPER ENERGY LIMIT FOR THE -
CD		SECONDARY NEUTRON ENERGY E' SUCH THAT -
CD		O.LE.E".LE.E-U (GIVEN IN THE LABORATORY) WHERE -
CD		E IS THE INCOMING NEUTRON ENERGY -
CD	GAMN2N	GAMMA RAY ENERGY ASSOCIATED WITH THE L'TH -
CD		(N.2N) RESOLVED SCATTERING LEVEL. IF O IS THE -
CD		ASSOCIATED O-VALUE OF THE REACTION, THEN -
CD	,	$GAMN2N = -O (GAMN2N, GT, 0) \qquad -$
CD	•	GAMN2N IS ORDERED SUCH THAT
CD		CAMN2N(1) IF CAMN2N(2) IF CAMN2N(3) -
	FTN	STAK ENERGIES FOR WHICH TABULATED INFLASTIC -
CD .	۲) ـل در	SINK BRENGIES FOR WHICH INDURIND INDURSTIC
	ចលាហ	
	E IN Z IN	COMMENTAL DEORADITITATE ADE CIVEN
	CUMT	CUM OF ANGLANDY LANDNI LANGNY LANGNY DA OUFD
CD	2020	SUM OF (NESJ+MAXIJ+NZNEJ+NSNKIJ+NSNKZJ) OVER -
CO		ALL MMAT MATERIALS -
CD	NLSJ	= NLEVLS(J). FOR CURRENT MATERIAL J
CD	MAX1J	= MAXT(J) FOR CURRENT MATERIAL J -
CD	N 2 N L J	= N2NLEV(J) FOR CURRENT MATERIAL J -
CD	NSNK 1J	= NSINK1(J) FOR CURRENT MATERIAL J -
CD	NSNK2J	= NSINK2(J) FOR CURRENT MATERIAL J -
С		· –
CN		THE INELASTIC AND (N, 2N) CONTROL INFORMATION -
CN		NLEVLS, MAX1, N2NLEV, NSINK1, AND NSINK2 ARE -
CN		SPECIFIED IN RECORD TYPE 5 OF MCC2F1 -
CN		
CN		DATA ARE PRESENT ONLY FOR MATERIALS THAT HAVE -
CN		EITHER INELASTIC OR (N, 2N) DATA
С		-
C		
~		
C=	CONTR	ROL INFORMATION AND CROSS SECTION DATA (TYPE 2) -
C	CONTI	
čc	I P H A G	NT TE NMAX.GT.O -
c c	1 111 3 4	-
CI	(STGTN (T) S	
	ע (ד) אבטיבט (ב) אר (ד) אבטייבאר	$\frac{1}{1} \frac{1}{1} \frac{1}$
CL		$\frac{1}{1} = \frac{1}{1} = \frac{1}$
C L		(UV(1), 1 - i, HIR 1)
	СПМТ	-
Cw	SUMJ	-
C	0.7.0.7.V	
CD	SIGIN	GROUP TOTAL INELASTIC SCATTERING CROSS SECTION -
CD	SIGN2N	GROUP (N, 2N) SCATTERING CROSS SECTION -
CD	KT1	INTERPOLATION LAW FOR CALCULATING GROUP TO -
CD		GROUP TABULATED INELASTIC SCATTERING -
CD		PROBABILITIES FROM THE GROUP TO ENERGY (PINTAB) -
CD		PROBABILITIES GIVEN IN RECORD TYPE 4
CD	:	KT=0 CONSTANT PROBABILITY

-320-

-1

CD CD CD CD CD CD CD CD CD CD CD CD CD C	KT=1LN(ENERGY)VS.LN(PROBABILITY)KT=2LN(ENERGY)VS.PROBABILITYKT=3ENERGYVS.PROBABILITYKT=4ENERGYVS.LN(PROBABILITY)INTERPOLATIONLAWFORCALCULATINGGROUPTABULATED(N,2N)SCATTERINGPROBABILITIESFROMTHEGROUPTOPROBABILITIESGIVENINRECORDTYPEPROBABILITIESGIVENINRECORDTYPEKT=0CONSTANTPROBABILITY-KT=1LN(ENERGY)VS.LN(PROBABILITY)KT=2LN(ENERGY)VS.PROBABILITYKT=3ENERGYVS.PROBABILITYKT=4ENERGYVS.LN(PROBABILITY)	
CD SUM.T CD C	SUM OVER I FOR ALL 14 ARRAYS FOR ALL MMAT1 MATERIALS	-
CN CN CN CN CN CN CN CN	CROSS SECTION DATA AND CONTROL INFORMATION DATA ARE PRESENT FOR A MATERIAL ONLY IF THE MATERIAL HAS THESE DATA PRESENT AS SPECIFIED IN RECORD TYPE 5 OF MCC2F1. ALSO THESE DATA ARE PRESENT ONLY IF THE GROUP BEING PROCESSED IS ABOVE THE THRESHOLD ENERGY OF THE REACTION BEING CONSIDERED	
CN CN CN	SIGIN IS PRESENT FOR MATERIALS THAT HAVE - INELASTIC DATA FOR THE ENERGY GROUP BEING - PROCESSED. NINEVP IS PRESENT FOR ALL MATERIALS -	•
CN CN CN CN	THAT HAVE MAX1.GT.O. NLVS IS PRESENT FOR ALL MATERIALS THAT HAVE NLEVLS.GT.O. IN ADDITION SIGIN, NINEVP, AND NLVS ARE PRESENT ONLY FOR GROUPS.LE.NINEL	-
CN CN CN CN CN CN CN	SIGN2N IS PRESENT FOR MATERIALS THAT HAVE (N,2N) DATA FOR THE ENERGY GROUP BEING PROCESSED. N2NEVP IS PRESENT FOR ALL MATERIALS THAT HAVE MAX3.GT.O. N2NLV IS PRESENT FOR ALL MATERIALS THAT HAVE N2NLEV.GT.O. IN ADDITION SIGN2N, N2NEVP, AND N2NLV ARE PRESENT ONLY FOR GROUPS.LE.N2NTH	•
CN CN CN CN CN	NINTAB, NSTRT1, NEND1, AND KT1 ARE PRESENT FOR ALL MATERIALS THAT HAVE MAX2.GT.O. DATA PRESENT FOR ALL GROUPS.LE.NINEL	
CN CN CN CN CN	FOR ALL MATERIALS THAT HAVE MAX4.GT.O. DATA PRESENT FOR ALL GROUPS.LE.N2NTH NINEL, N2NTH, NLEVLS, N2NLEV, MAX1, MAX2, MAX3, AND MAX4 ARE PRESENT IN RECORD 5	

۰.

C N C		OF MCC 2F1 -
C	*	
C CR	SECONDARY D	ISTRIBUTION DATA (TYPF 3) -
C		
сс сс	PRESENT IL	(NLVS.GI.0) OR (N2NLV.GI.0) OR - (N2NLV.GI.0)
CC		FOR ANY MATERIAL -
ູ້		
CL	((SIGLEV(I,J),I=1)	,NLVSJ), (AVGMU(İ,J),I=1,NLVSJ), -
CL	1 (SIGN(I,J), I=1, N2)	NLVJ), (TSTAT (I, J), I=1, NINVPJ), -
CI,	2(PIN(I,J), I=1, NIN)	VPJ), (TN2N(I,J), I=1, N2NVPJ), -
CL	3(PN2N(T,J), T=1, N2)	NVPJ, $J=1$, $MMAT1$) -
C 7	T M T	· · · · · · · · · · · · · · · · · · ·
C «	3090	
CD	SIGLEV	GROUP CROSS SECTION ASSOCIATED WITH THE -
CD		L'TH RESOLVED INELASTIC SCATTERING LEVEL -
CD	SIGN	GROUP CROSS SECTION ASSOCIATED WITH THE -
CD		L'TH RESOLVED (N, 2N) SCATTERING LEVEL -
CD	AVGMU	AVERAGE COSINE OF THE SCATTERING ANGLE IN -
CD		THE CENTER OF MASS SYSTEM FOR THE GROUP IN -
(;)) C D	<u>መድመእጥ</u>	QUESTION FOR DISCRETE INFLASTIC SCATTERING -
	10141	TEMPERATURE -
CD	PIN	FRACTIONAL PROBABILITY THAT THE LITH INFLASTIC -
CD	//	EVAPORATION LAW CAN BE USED FOR THE GROUP IN -
CD	· ·	QUESTION -
CD	TN2N	AVERAGE GROUP TEMPERATURE ASSOCIATED WITH A -
CD		STATISTICAL CALCULATION OF (N, 2N) SCATTERING -
CD		FOR THE I'TH DISTRIBUTION -
CD CD	PNZN	FRACTIONAL PROBABILITY THAT THE IVTH $(N, 2^{M})$ -
	•	- OURSTION LAW CAN BE USED FOR THE GROUP IN -
CD	SUMJ	SUM OF $(2*(NLVSJ+NINVPJ+N2NVPJ)+N2NLVJ) OVER -$
CD		ALL MMAT MATERIALS -
CD	NLVSJ	= NLVS (J) FOR CURRENT MATERIAL J -
CD	N2NLVJ	= N2NLV(J) FOR CURRENT MATERIAL J -
CD	NINVPJ	= NINEVP(J) FOR CURRENT MATERIAL J -
CD	N2NV PJ	= N2NEVP(J) FOR CURRENT MATERIAL J -
C		-
		THE INELASTIC AND (N/2N) CONTROL INFORMATION -
		TN RECORD TYPE 2 OF MCC2F6 -
CN		-
CN		IF FOR A GIVEN MATERIAL J, THE VALUE OF NLVSJ
CN		N2NLVJ, NINVPJ, OR N2NVPJ ARE ZERO, THEN THE

TABULATED II PRESENT IF	NELASTIC AND (N,2N) PROBABILITIES (TYPE 4) - (NINTAB.GT.O) OR (N2NTAB.GT.O) - FOR ANY MATERIAL -
(PINTAB(I,J),I=1	
К1+К2) *ММАТ1	
INTAB 1 NNTAB	TABULATED PROBABILITY OF A NEUTRON BEING INELASTICALLY SCATTERED FROM THE GROUP IN QUESTION TO ALL POSSIFLE SINK ENERGIES (EIN) PINTAB (1) = PROBABILITY OF SCATTERING FROM GROUP IN QUESTION TO ENERGY EIN (NSTRT1) PINTAB (K1) = PROBABILITY OF SCATTERING FROM GROUP- IN QUESTION INTO ENERGY EIN (NEND1) PINTAB IS CALCULATED AS THE SUM OVER ALL (ENDF/R) GIVEN TABULATED INELASTIC DISTRIBUTIONS OF THE PARTIAL ENERGY DISTRIBUTION*FRACTIONAL PROBABILITY NEND1(J)-NSTRT1(J)+1 TABULATED PROBABILITY OF A NEUTRON BEING (N, 2N) SCATTERED FROM THE GROUP IN QUESTION INTO ALL POSSIBLE SINK ENERGIES (EN2N)- PNNTAB(1) = PROBABILITY OF SCATTERING FROM GROUP IN QUESTION TO ENERGY EN2N (NSTRT2) PNNTAB(K2) = PROBABILITY OF SCATTERING FROM GROUP- IN QUESTION INTO ENERGY EN2N (NEND2) PNNTAB IS CALCULATED AS THE SUM OVER ALL (ENDF/B) GIVEN TABULATED (N, 2N)
2	(LNDI) B) GIVEN TREBURATED (N,2N)DISTRIBUTIONS OF THE PARTIAL ENERGYDISTRIBUTION*FRACTIONAL PROBABILITYNEND2(J) -NSTRT2(J) +1PINTAB IS PRESENT ONLY IF NINTAB.GT.0PNNTAB IS PRESENT ONLY IF N2NTAB.GT.0NEND1, NSTRT1, NEND2, NSTRT2, NINTAB, ANDN2NTAB ARE SPECIFIED IN RECORD TYPE 2OF MCC2F6
1	TABULATED I PRESENT IF (PINTAB(I,J),I=1 K1+K2) *MMAT1 INTAB

EOF

-323-

C****	*****	* * * * * * * * * * * * *	*****	* * * * * * * * * * * * * * * * * * * *
C			· .	-
C			PREPARED 2/11/75 AT ANL	-
Ċ				· -
CF		MCC2F7		· -
CE		FISSION SPEC	CTRA DATA	-
С		а. А.		-
C****	*****	****	* * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *
CD	ICHI		FLAG INDICATING WHETHER PA	ARAMETERS IN
CD			GENERALIZED FISSION SPECTR	KUM ARE ENERGY
СÐ			(GROUP) DEPENDENT OR NOT	·
CD			ICHI=1 PARAMETERS NOT ENH	ERGY DEPENDENT
CD			ICHI=NGROUP PARAMETERS AN	RE ENERGY DEPENDENT
CD			AND NGROUP,	THE NUMBER OF GROUPS
CD.			IN THE LIBRAN	RY IS SPECIFIED IN THE
CD			ADMINISTRATIV	/E FILE MCC2F1
CD	MSORS		NUMBER OF FISSION SPECTRA	IN LIBRARY
CD			AS SPECIFIED IN THE ADMIN	ISTRATIVE
CD		· · ·	FILE MCC2F1	
CD · .	MULT		2 FOR IBM MACHINES, 1 OTHI	ERWISE
			1	
C				
CS		FILE STRUCTU	IRE	-
CS			í.	
CS		RECORD TYPE		PRESENT 1F
CS				
CS		MATERIAL NAC		
CS	الدمالد مالد مالد بالد	SPECIFICATIO		ALWAIS
CS	*****	*** (REPEAT (ISURS TIMES)	
CS	*	FISSION SPEC		ALWAIS
CS	*****	κ π π		
C				
C				
C				
CP			4 FS (ΦVDF 1)	
CF.		OFIGNIER NET		-
C CC		AT TAVS DREST	ר אית	-
c c		REARIS ENDI		-
CT	(NAME)	(T) $T=1$ MSORS	3)	-
C	(Nenna)	(x), $x = (x)$	51	-
CW	MIIT THAN	ISORS		
С." С		1001.0		-
CD	NAMF		DOUBLE PRECISION (R*8) MA	PERTAL IDENTIFICATION -
	NEALS		ASSOCIATED WITH THE GIVEN	FISSION SPECTRUM -
CD			PARAMETERS	
00				

,

-

с [.]	•			; ·							-
(
CR C		SPECIFICAT	IONS	(TYPE	2)						
CC C		ALVAYS PRE	SENT								-
CL C	(ICHI	(I), I=1, MSO	RS)	1 -					•		-
C W C	MSORS			8							-
C											
C											
CR C		FISSION SP	ECTRA	А (ТҮР	E 3)						- -
CC C		ALWAYS PRE	SENT			•					-
CL C	(BETA	(I), I = 1, ICH	I),(A	LPHA (I),I=	1, ICHI	.) , (TAU	(I),I	=1, IC HI	.)	-
CW C	3*JCH3	[-
CD C	BETA,	ALPHA,TAU	PAF	AMETE	RS IN	GENER	RALIZED	FISS	ION SPE	CTRUM	-
(• • • • • • •				·; -

CEOF

C**	* *	*****	****	****	*****
C C C				PREPARED 2/11/75 AT ANL	-
CF			MCC2P8	ת ת	-
C					-
C**	* *	*****	****	* * * * * * * * * * * * * * * * * * * *	****
				· ·	
CD CD CD		IL		NUMBER OF BLOCKS OF DATA NO DATA ARE PROVIDED FOR WHICH SCATTERING IS ISOTR	FOR A GIVEN MATEPIAL ENERGIES BELON OPIC
ĊD CD		IPT		NUMBER OF LEGENDRE COEFFI EACH BLOCK OF DATA BY MAT	CIENTS PROVIDED FOR FRIAL
CD CD		IR		NUMBER OF INTERPOLATION R DATA RECORD	EGIONS FOR LEGENDRE
CD CD CD		MULT NPL		2 FOR IBM MACHINES, 1 OTH HIGHEST ORDER PERMITTED F APPROXIMATION AS SPECIFIE	ERWISE OR EXTENDED TRANSPORT D IN THE
́СD С				ADMINISTRATIVE FILE MCC2F	1
CS CS			FILE STRUCT	JRE	- -
CS			RECORD TYPE	•	PRESENT IF -
CS					
CS CS		*****	*** (REPEAT	POR ALL MATERIALS)	
cs		*	MATERIAL ID	ENTIFICATION	ALWAYS -
СS		*	T MATRIX AND	D INTERPOLATION DATA	ALWAYS -
CS CS		* ***	*** (PEPEAT] LEGENDRE CO]	IL TIMES) EFFICIENTS	
CS C		*****	* * *	•	-
c			``	· · · · · · · · · · · · · · · · · · ·	
CP CR			SPECIFICATI	ONS (TYPE 1)	-
cc cc			ALVAYS PRES	ENT	-
CL CL	·	(IGTH 1 ((IPT)	(I), I=1, NMAT) (I,J), I=1, NPA	, (IR (I) , I= 1, NMAT) , (IL (I) , ASS) , J= 1, NMAT)	I=1,NMAT), -
ट C ग C	Υ.	<u>NMAT</u> *	(3+NPASS)		
CD		LGTH	,	LENGTH (IN WORDS) OF	

-326-

CD CD CD CD CD CD CD C	NMAT NPASS	T MATRI NUMBER IN THE NUMBER AS SPEC MCC2F1	AND INTERPOLATION DATA RECORD OF MATERIALS IN LIBRARY AS SPECIFIED ADMINISTRATIVE FILE MCC2F1 OF 'BLOCKS' OF LEGENDRE DATA LFIED IN THE ADMINISTRATIVE FILE -
C			
CŘ - C		MATERIAL IDENTIFICA:	TION (TYPE 2) -
сс с		ALWAYS PRESENT	-
CL C	NAME		-
C W C	MULT*1		· · · ·
CD C	NAME	DOUBLE	PRECISION (R*8) MATERIAL IDENTIFICATION -
C			
C CR		T MATRIX AND INTERPO	DLATION DATA (TYPE 3) -
C CC		ALWAYS PRESENT	-
CL CL	(KT(I) 1((TLJ)	, I = 1, IR), (NG(I), I = 1, L.J., L=1, NPL), J=1, MA	IR),
CL CL	2 ((TLJ1 3 ((TLJ2	(L,J), L=1,2), J=1, MAX (L,J), L=1,2), J=1,6)	(IPT) + 1), ((FACK(I,J),I=1,6),J=1,6), -
CL C	4 ((FAC (I,J),I=1,4),J=1,MAX	(IPT) +1) -
C W C W C	2*TR+№ 2*TP+№	ULT* (54+MAX (IPT) * (N) ULT* (50+MAX (IPT) * (N)	PL+6)+NPL) FOR DELTAU.GE.Q (SEE BELOW) - PL+2)+NPL) FOR DELTAU.LT.Q (SEE BELOW) -
CD CD CD	КТ	INTERPO GROUP VI F1,F2,	LATION LAW USED TO OBTAIN HYPERFINE - LUES OF THE LEGENDRE COEFFICIENTS, - ., FN, IN THE INTERPOLATION REGION I -
CD CD CD CD	•	KT=0 $KT=1$ $KT=2$ $KT=3$ E	N E VS. LN FN-I E VS. FN-VS. FN-
CD CD CD	NG	KT=4 E LOWEST H FOR WHIC	VS. LN FN ENERGY GROUP NUMBER (HIGHEST ENERGY) - CH INTERPOLATION LAW APPLIES -
CD CD	TLJ TLJ1	STANDARI STANDARI	D ZERO ORDER T MATRIX ELEMENTS-O FIRST ORDER T MATRIX ELEMENTS-

1

CD CD CD CD CD CD CD CD CD CD CD CD CD C	TLJ2 FACK FAC	STANDARD SECOND ORDER T MATRIX ELEMENTS MASS DEPENDENT CONSTANTS USED IN CALCULATION OF CONTINUOUS SLOWING DOWN MODERATING PARAMETERS 'INCOMPLETE' T MATRIX ELEMENTS THE ARRAY FAC IS PRESENT ONLY FOR THOSE ELEMENTS THAT HAVE DELTAU.GE.Q WHERE Q=LOG(((A+1)/(A-1))**2)/3. DELTAU AND A ARE THE GROUP LETHARGY WIDTH AND MATERIAL MASS/NEUTRON MASS RESPECTIVELY. BOTH ARE SPECIFIED IN THE ADMINISTRATIVE FILE MCC2F1 ALL REAL ARRAYS, TLJ,TLJ1,TLJ2,FACK, AND FAC ARE WRITTEN IN DOUBLE PRECISION (REAL*8)	
C			
CR	LEGENDRE CO)EFFICIENTS (TYPE 4)	_
C			_
	PRESENT IF		-
CL	(F1(J), F2(J),,	FN(J), $J=1$, MANY 1)	-
c			-
CW	ΜΑΝΥ1*ΙΡΤ		-
C			-
CD	F1,F2,,FN	FIRST, SECOND,, N'TH. ORDER EXPANSION	-
CD		COEFFICIENT AT THE ENERGY LEVEL SPECIFIED	-
CD		BY THE INDEX J. FOR EXAMPLE, ON THE THIRD	-
CD		PASS, THE THIRD BLOCK OF DATA, F1(1) WOULD	-
CD		BE THE FIRST ORDER COEFFICIENT FOR ENERGY	-
CD		LEVEL 2*MANY1-1	-
CD		THE NUMBER OF COEFFICIENTS (N) FOR A GIVEN	-
CD		BLOCK, OF DATA IS GIVEN BY THE ARRAY LPT	-
CD		IN THE TABLE OF CONTENTS. NOTE THAT	-
CD		FU(J) = 1.0 AND 1S THEREFORE NOT TABULATED. THIS	-
CD	•	RECORD IS NOT PROVIDED FOR THOSE ENERGY BLOCKS	-
CD		CORRESPONDING TO ENERGIES BELOW WHICH	-
CD		SCATTERING IS ISOTROPIC IN THE CENTER OF	-
CD	_ •	MASS SYSTEM.	• —
CD	MANY 1	NUMBER OF ENERGY LEVELS FOR EACH 'BLOCK' OF	-
CD	•	DATA (EXCEPT THE LAST) AS SPECIFIED IN THE	-
CD		ADMINISTRATIVE FILE MCC2F1	. –
CD		RECOMMENDED MAXIMUM VALUE FOR MANY1 WOULD BE	-
CÐ		MANY 1= 126	-
С			
C			

CEOF

.

C***	******	* * * * * * * * * * * * * * * *	******
C			-
C C			PREPARED 3/11//6 AT ANL -
CF		XS.ISO	
СE		MICROSCOPIC	GROUP CROSS SECTIONS, FILE 1 -
С			-
CN			THIS IS FILE 1 OF A TWO-FILE DATA SET
C			CONTAINING MICROSCOPIC GROUP CROSS SECTIONS.
C***	*****	* * * * * * * * * * * * *	******
CD	NGROUE)	NUMBER OF BROAD ENERGY GROUPS IN SET.
CD	NISO		NUMBER OF ISOTOPES IN SET.
CD	MULT		2 FOR IBM MACHINES, 1 OTHERWISE.
C		FTLE STZE (7	
C			-
СС		ALWAYS PRESI	ENT -
C	NCDOUR		
CL	NGROUP	, NºSO, MAXUP,	MAXDN, MAXORD, IPR EC
CV	6		· · · ·
С			-
CD	MAXUP		MAXIMUM NUMBER OF GROUPS OF UPSCATTER IN THE -
CD	ΜΛΥΓΝ		SET
CD	EAAD N		SET
CD	MAXORD)	(MAXIMUM OF LEL, LIN, OR LN2N FOR THE SET) -1 -
CD			(SEE ISOTOPE HEADING RECORD OF DATA SET XS.ISO, -
CD	TODOG		FILE 2)
CD	TBKEC		V FOR DOUBLE PRECISION VERSION OF DATA SET
CD			DATA SET
č			-
C			
C			
CR		TSOTOPE NAME	
С			-
CC		ALWAYS PRESE	
C	(TCONA	M (T) T+1 NTC	= -
CL CL	A אוטכבן אאמאד 1 (דאמאד	□(⊥),⊥=I,N±3 !(T),T=1.NTSC	$(1)_{1} (1)_{1} = 1, NISO(1)_{1} (1)_{1} = 1, NISO(1)_{1} = 1, NISO(1)_{$
C	(Trogen		
C	NISO1=	NISO+1	-

1

•

.

.

,

С NTSO+1+2*MULT*NISO+2*NISO*(2-IPREC) CV С LOCAL NAME OF I-TH ISOTOPE (REAL*8). CD ISONAM(I) NUMBER OF RECORDS IN FILE 2 OF XS.ISO TO BE CD LOCA (I) SKIPPED TO READ DATA FOR ISOTOPE I. LOCA(1)=0. CD TEMPERATURE OF I-TH ISOTOPE. CD TEMP(I) I-TH ISOTOPE NAME AS GIVEN IN ENDF/B FILES. CD TNAME GRAM ATOMIC WEIGHTS OF THE ISOTOPES. CD AMASS С _____ C----CR GROUP STRUCTURE (TYPE 3) С ALWAYS PRESENT CC C ICHI, (E(I), I=1, NGP1), (U(I), I=1, NGP1), (VEL(I), I=1, NGROUP) CL C CC : NGP1=NGROUP+1 С C₩ 1+(3*NGROUP+2)*(2-IPREC) С FISSION SPECTRUM FLAG FOR SET. CD ICHI ICHI=0, NO SET CHI. CD =1, SET CHI VECTOR. CD -=NGROUP, SET CHI MATRIX. CD ENERGY BOUNDARTES OF GROUPS. E(1) IS СÐ E(I) THE MAXIMUM ENERGY. CD LETHARGY BOUNDARIES OF GROUPS. U(1) =0. CD U(T) NEUTRON SPEED FOR GROUP I, CD VEL(I) SPEED 1./(1./V). CD С _____ C----SET FISSION SPECTRUM (TYPE 4) CR С CC PRESENT IF ICHI.NE.0 С ((CHT(I,J), I=1, ICHI), J=1, NGROUP)CL С ICHI*NGPOUP* (2-IPREC) CW С PROMPT FISSION FRACTION INTO GROUP J CD CHI(J,J) FROM GROUP I. CD С

CEOF

-330-

C****	****	****	*****
C			-
С		PREPARED 3/11/76 AT ANL	. –
С	•		• • • •
CF	XSISO2		-
CE	MICROSCOPIC	GROUP CROSS SECTIONS, FIL	E 2 –
С		•	_
CN		THIS IS FILE 2 OF A TWO-F	TLE DATA SET -
CN		CONTAINING MICROSCOPIC GRO	OUP CROSS SECTIONS
c.			-
C****	****	****	****
C++++		• • • • • • • • • • • • • • • • • • •	* * * * * * * * * * * * * * * * * * * *
c			
CS '	FILE STRUCT	URE	-
CS			· _
CS	RECORD T	YPE	PRESENT IF -
CS	=======================================		
CS	*****	AT FOR ALL ISOTOPES)	· · · ·
C S	* TSOTOPE	NAME	AT MAYS -
CS	* TSOTOPE	HENDING	
C 5	* 150102E * TSOTOPE	NERDING RICCION CDROMBHM	
CS CC		NE FOR ALL CROUPS	
CS 66		AT FOR ALL GROUPS)	-
CS	* * PRINCIPA	L CRUSS SECTIONS	ALWAYS -
CS	* * * * * * * * * * *		-
CS	*		-
CS	* ******** (REPE	AT FOR EACH SCATTERING ORD	ER –
CS	* * L=1,	LMAX, WHERE LMAX IS THE	
CS	* * LARG	EST OF LIN, LEL, AND LN2N.	-
CS	* * SEE	ISOTOPE HEADING RECORD.)	· -
CS	* * ****** (REPE	AT FOR ALL GROUPS)	· –
CS	* * * TNDEX FO	R SCATTERING GROUP	ALVAYS -
CS	* * * INELASTI	C SCATTERING	LIN.GE.L -
CS	* * * ELASTIC	SCATTERING	LEL.GE.L -
CS	* * * (N.2N) S	CATTERING	LN2N.GE.L -
CS	*** ***		
c			-
C			
• ·		·	
CD	LELDN	NUMBER OF ELASTIC DOWNSCAT	ITER GROUPS.
CD	LELUP	NUMBER OF ELASTIC UPSCATTI	ER GROUPS.
CD	LINDN	NUMBER OF INELASTIC DOWNSO	CATTER GROUPS.
CD	LINUP	NUMBER OF INELASTIC UPSCAT	ITER GROUPS.
CD	LN2NDN	NUMBER OF (N. 2N) DOWNSCAT	TER GROUPS.
CD	LN2NUP	NUMBER OF (N.2N) UPSCATTER	R GROUPS.
CD	NGROUP	NUMBER OF FNERGY GROUPS IN	N THE SET.
CD		2 FOR TRM MACHINES 1 OTH	RWISF
		A FOR DOUBLE DEFORTON VE	28TAN AF DATA SPT
ים גר מיזי	トロン ラク	AC ICO 1 EOD CINCIE DEBCI	NOTON ALDGION OF MAD
JU		VO*TOON I LOU DINGTE BUECT	FOTON ARUPTON OU THE

-331-

CD	\$	DATA SET.	
C CR	ISOTOPE	NAME (TYPE 1)	
C CC	ALVAYS PI	RESENT	-
C CL	TSONAM		_
C			-
Ç	MULT		—
CD C	ISONAM	ISOTOPE NAME (REAL*8).	-
C			
C			
CR	ISOTOPE	HEADING RECORD (TYPE 2)	-
cc	ALWAYS P	RESENT	-
CL	ICHI, LIN, LEL, L	N2N, EFISS, ECAPT	-
C CW	4+2*(2-IPREC)		-
C CD CD CD CD	ICHI	ISOTOPE FISSION SPECTRUM FLAG. ICHI=-1, ISOTOPE USES PROMPT FISSION SPECTRUM FOR SET. =0, ISOTOPE IS NOT FISSIONABLE.	
CD CD CD CD		<pre>=1, ISOTOPE USES OWN PROMPT FISSION SPECTRUM WHICH IS NOT INCIDENT-ENERGY- DEPENDENT (VECTOR). =NGROUP, ISOTOPE USES OWN PROMPT FISSION</pre>	- - -
CD	· · ·	SPECTRUM WHICH IS INCIDENT-ENERGY- DEPENDENT (MATRIX).	-
CD CD CD CD	LIN.	MAXIMUM ORDER OF INELASTIC SCATTERING. LIN=0, NO SCATTERING. = 1, ISOTROPIC SCATTERING. = 2, LINEAR ANISOTROPIC SCATTERING.	- - -
CD CD CD CD CD	LEL	=N, ORDER N-1 ANISOTROPIC SCATTERING. MAXIMUM ORDER OF ELASTIC SCATTERING. LEL=0, NO SCATTERING. = 1, ISOTROPIC SCATTERING. = 2, LINEAR ANISOTROPIC SCATTERING.	-
CD CD CD CD	l n2 n	<pre>MAXIMUM ORDER OF (N,2N) SCATTERING. MAXIMUM ORDER OF (N,2N) SCATTERING. LN2N=0, NO SCATTERING. =1, ISOTROPIC SCATTERING. =2, LINEAR ANISOTROPIC SCATTERING.</pre>	-

-332-

CD =N, ORDER N-1 ANISOTROPIC SCATTERING. WATT-SECONDS/FISSION. EFISS CD WATT-SECONDS/CAPTURE WHERE CAPTURE REFERS TO CD ECAPT CD NON-FISSION ABSORPTION. C _____ C-----CR ISOTOPE FISSION SPECTRUM (TYPE 3) С СС PRESENT IF ICHI.GT.O (SEE ISOTOPE HEADING RECORD) С CL ((CHI(I,J),I=1,ICHI),J=1,NGROUP) , С CW ICHI*NGROUP*(2-IPREC) С CD FISSION SPECTRUM INCIDENT IN GROUP J, BORN CHI(L,J) CD IN GROUP I. С -----C----C-----CR PRINCIPAL CROSS SECTIONS (TYPE 4) С CC ALVAYS PRESENT С CL J, STR, SCAP, SNALF, SNP, ANISO, SFIS, FISNU С CW 1+7*(2-IPREC) IF ICHI.NE.0 CW 1+5* (2-IPREC) IF ICHI.EO.0 С CD Ĵ GROUP INDEX. CD STR TRANSPORT CROSS SECTION. CD SCAP RADIATIVE CAPTURE (N, GAMMA) CROSS SECTION. (N, ALPHA) CROSS SECTION. CD SNALF CD SNP (N,P) CROSS SECTION. ISOTROPIC ELASTIC SELF-SCATTERING CONSISTENT CD ANISO CD WITH ANISOTROPIC TOTAL CROSS SECTION. ANISO, WHEN ADDED TO THE SUM OF ALL EVENTS CD CD WHICH REMOVE A NEUTRON FROM THE GROUP. YIELDS THE TOTAL GROUP CROSS SECTION. CD FISSION CROSS SECTION, PRESENT IF ICHI.NE.O. CD SFIS CD FISNU NUMBER OF FISSION NEUTRONS PER FISSION TIMES CD FISSION CROSS SECTION, PRESENT IF ICHI.NE.O. С

-333-

APPENDIX C. MC²-2 Binary Interface Files. XS.ISO (Contd.)

ه . العام عبد ۲

INDEX FOR SCATTERING GROUP (TYPE 5) CR С CC PRESENT IF LIN+LEL+LN2N.NE.O CC (SEE ISOTOPE HEADING RECORD) C LINUP, LINDN, LELUP, LELDN, LN2NUP, LN2NDN CL С CW 6 С C----۰. C----CR INFLASTIC SCATTERING (TYPE 6) Ċ PRESENT IF LIN.GE.CURRENT SCATTERING ORDER CC (SEE ISOTOPE HEADING RECORD). CC С SINL (J+LINUP), SINL (J+LINUP-1),..., SINL (J),..., SINL (J-LINDN) CL С CW (LTNUP+LINDN+1) * (2-IPREC)С INELASTIC SCATTERING, GROUP K TO GROUP J. CD SINL(K) C C-------------ELASTIC SCATTERING (TYPE 7) CF С CC PRESENT IF LEL.GE.CURRENT SCATTERING ORDER (SEE ISOTOPE HEADING RECORD) CC С SELT (J+LFLUP), SELT (J+LELUP-1), ..., SELT (J), ..., SELT (J-LELDN) CL С (LELUP+LELDN+1) * (2-IPREC) CW С ELASTIC SCATTERING, GROUP K TO GROUP J. CD SELT(K) C C - - - -(N,2N) SCATTERING (TYPE 8) CR С PRESENT IF LN2N.GE.CURRENT SCATTERING ORDER CC CC (SEE ISOTOPE HEADING RECORD) С SN2N (J+LN2NUP), SN2N (J+LN2NUP-1),..., SN2N (J), ..., SN2N (J-LN2NDN)CL

-334-

Ċ		-						· -
CŴ	(LN2NUP+LN2NDN	+1) * (2-IPR	EC)					-
С								-
CD	S N 2 N (K)	(N, 2N)	SCATTERING,	GROUP	к то	GROUP	J.	-
С	-	•						-
C								

CEOF

-335-

MC²-2 BINARY FILES

.

,

.

,

_ * * * * * *	*****	* * * * * * * * * * * * * * * * * * * *	****
C C			-
C C		PREPARED 3705775 AT ANL	
ĊF		ATNUAT	-
CE		UNRESOLVED RESONANCE INTEGRALS	· · · -
С			-
CN		THIS DATA SET IS WRITTEN	BY MC**2-II AREA 5 -
CN		(CSC004) IF UNRESOLVED MA	TERIALS ARE PRESENT -
C****	*****	* * * * * * * * * * * * * * * * * * * *	
0			
CD	JFI	FISSILE MATERIAL INDEX	
CD		IFI=0 FOR NON-FISSILE MAT	ERIAL
·CD	`	IFI=1 FOR FISSILE MATERIA	L
CD	JL	NUMBER OF SPIN STATES FOR	CURRENT ISOTOPE
CD	MULT	2 FOR IBM MACHINES, 1 OTH	ERWISE
CD	NESF	NUMBER OF FIXED ENERGY ME	SH POINTS
CD	NREG	NUMBER OF REGIONS	
CD		NREG=1 FOR HOMOGENEOUS PR	EOBLEMS (NGEOM=0)
CD		NREG=2 FOR PIN CELLS (NGE	OM=2) AND 1F
CD		MAXHIM.GI.U NDEC-1 4 NUMPED OF CLAR D	ECTÓNE BOD CIND
CD		, NREGET + NUMBER OF SLAD R	AND TE MAYUMM OM O
CD CD		PRODLEMS (NGEOM-I) (MAYHTM IS THE MAYTMIM NII	MARD IF MARGING(.V MARD OF MATERIALS IN
CD		ANY HETEROGENEOUS PECTON	NDER OF HRIERCRES IN
CD	NUMPPO	NUMBER OF UNRESOLVED RESO	INANCE MATERIALS
CD	NUNRES	NUNRES=0 IF UNRESOLVED RESO	SONANCE CROSS SECTIONS
CD	NO MILIC	ARE GENERATED, NUNRES=1 T	F UNRESOLVED RESONANCE
CD		INTEGRALS ARE GENERATED	
C			
CS		FILE STRUCTURE	-
CS			
CS CS		RECORD TIPE	PRESENT IF
C3		MATERIAL NAMES	ALWAYS -
CS		FILED UNRESOLVED RESONANCE	ALWAYS -
CS		ENERGY MESH	
CS	*****	***** (REPEAT FOR NUMBES MATERIALS)	-
cs	*	MATERIAL SPECIFICATIONS	ALWAYS -
CS	* ***	***** (REPEAT FOR NISO ISOTOPES)	-
CS	* *	SPIN STATE DATA	ALWAYS -
CS	* *	LEVEL SPACING	ALWAYS -
CS	* *	***** (REPEAT FOR 1+NO. OF REGIONS	-
CS	* *	* IN WHICH MATERIAL IS TREATED	
CS	* *	* HETEROGENEOUSLY)	-

-337-

.

cs	* * * CAPTURE	J INTEGRAL	ALWAYS				
CS	* * * TOTAL J	INTEGRAL					
ĊS	* * * ¥ ¥15510N	J INTEGRAL	IFI.EQ.I				
CS CS C	**************************************	LUX CORRECTION FACTOR	ALWAYS				
C	SPECIFICAT	IONS (TYPE 1)					
C							
CC C	CC ALWAYS PRESENT C						
CL C	NUMRES, NESF, NREG	, JLMAX, MAXISO					
C W C	5		· · ·				
CD CD CD CD CD CD CD	JI.MAX	LSTMAX*JSTMAX WHERE LST VALUE OF THE NUMBER OF STATES OVER ALL MATERIA JSTMAX IS THE MAXIMUM V CHANNEL SPIN STATES ASS PARTICULAR ANGULAR MOME	MAX IS THE MAXIMUM ANGULAR MOMENTUM ALS IN THE FILE AND VALUE OF THE NUMBER OF SOCIATED WITH A ENTUM STATE OVER ALL				
CD CD C	MAXISO	MAXIMUM NUMBER OF ISOTO	OPES IN THE MIXTURE				
C C'R	MATERIAL N	AMES (TYPE 2)	_ 				
c cc	ALWAYS PRI	CS ENT	· · · ·				
C CL C	(UNRMAT (I), $I=1, N$	UMRES)					
CW C	MULT*NUMRES						
CD CD CD C	UNRMAT	DOUBLE PRECISION (R*8) MATERIAL NAMES	UNRESOLVED RESONANCE				
C							
C CP	FIXED UNRI	SOLVED RESONANCE ENERGY	MESH (TYPF 3)				
cc c	ALWAYS PRI	SENT					

-

-338-

C C	(ESF(I), I=1, NESF)						
C W C	NESF	- -					
CD CD C C	ESF	FIXED ENERGY MESH POINTS GIVEN IN ORDER OF DECREASING ENERGY					
C CR		MATERIAL SPECIFICATIONS (TYPE 4)					
CC	ALWAYS PRESENT						
CL	NISO, J						
C W	2						
CD C	NISO	NUMBER OF ISOTOPES -					
C		· · · · · · · · · · · · · · · · · · ·					
C CR		SPIN STATE DATA (TYPE 5)					
cc c		ALWAYS PRESENT					
CL	JL						
CW C	1						
C	• 						
C		LEVEL SPACING (TYPE 6)					
cc cc		ALWAYS PRESENT					
CL	((D(I),	-J), T=1,NESF), J=1,JL) -					
ČV C	NESF*J						
CD. C	D ·	AVERAGE LEVEL SPACING -					
C							

-339-

,

CR CAPTURE J INTEGRAL (TYPE 7) С CC ALWAYS PRESENT FOR THE HOMOGENEOUS MIXTURE. CC ALSO PRESENT FOR EACH HETEROGENEOUS REGION IN WHICH CC CURRENT MATERIAL IS TREATED HETEROGENEOUSLY С CL((UCJ(I,J), I=1, NESF), J=1, JL)С CW NESF*JL С UCJ CD UNRESOLVED RESONANCE CAPTURE INTEGRAL С C-C----CR TOTAL J INTEGRAL (TYPE 8) С СС ALWAYS PRESENT FOR THE HOMOGENEOUS MIXTURE. CC ALSO PRESENT FOR EACH HETEROGENEOUS REGION IN WHICH CURRENT MATERIAL IS TREATED HETEROGENEOUSLY CC С CL ((UTJ(I,J), I=1, NESF), J=1, JL)С CW NESF*JL С ÇD UTJ UNRESOLVED RESONANCE TOTAL INTEGRAL С C---FISSION J INTEGRAL (TYPE 9) CF С CC IF IFI.EQ.1, ALWAYS PRESENT FOR THE HOMOGENEOUS MIXTURE. ALSO PRESENT FOR EACH HETEROGENEOUS REGION IN WHICH CC CC CURRENT MATERIAL IS TREATED HETEROGENEOUSLY С CL ((JFJ(I,J), I=1, NESF), J=1, JL)С CW NESF*JL С CD UFJ UNRESOLVED RESONANCE FISSION INTEGRAL С C-----CP TOTAL FLUX CORRECTION FACTOR (TYPE 10)

-340-

C CC C	ALWAYS	PRESENT	- -		
CL C	(TOTFCF(I,J), I=1, NESF), J=1, NREG)				
C M C	NESF*NREG		-		
C D C	TOTFCF	TOTAL FLUX CORRECTION FACTOR	-		
C					

· CEOF

-341-

APPENDIX D. MC²-2 Binary Files. BC

C***	****	* * * * * * * * * * * *	****	****	*****	****	*****	*****	****	****
C^{++++}	• • • • • • •	· · · · · · · · · · · · ·	*****							-
c			וגמפסת		6/13/7	5 እጥ	λNT			_
C			PREPAR		0/15/7	JAL	ANL			_
CE		DC								-
		BC	CONDENTO		POTRIC	ATON	C			-
CE		BOUNDARY	CONDITION	UNDITION SPECIFICATIONS						_
C av								CTON OF	mu p	_
CN			THIS		N ABBR	EVIAT	ED VER	5100 0r		-
CN ·			DATA	DATA SET BC. IT IS USED FOR MC##2-11				-		
CN			CALCUI	ATI	UNS.					
(و مالد مال مال مالد مالد مال	، ماد باد باد باد باد ماد باد ماد باد با	. باد باد باد باد باد باد باد باد باد با	ىدىدىد	. مد مد بد مد بد بد ب	. بد بد بد بد	******	* * * * * * * *	****	
C****	* * * * * *	* * * * * * * * * * *	* * * * * * * * * * *	* * * *	****	* * * * *	* * * * * *	* * * * * * * * *	****	* * * * * * * *
		,	r							
c	s *									
CP		SPECIFIC		 / D	1					
C		SPECIFICI	MITONS (1)	ננים	1)					_
CT	NACT	NBCB								-
C	NDC 11									
CW	2									-
с "	2									-
CD	NBCI		BOUND	λ Ŕ V	CONDIT	ד א די	VDE NI	MRER FOR	. ग्रेस्ट्रेग २	-
	NDC L			N R V	OF THE	CELL	(SEE	BELOW).		-
CD	NRCP		BOUND	A DV	CONDIT			MBER FOR	RTCHT	-
CD	NDUM			N D V	OF THE	CEIL	ILD NON	BFIOW)		-
CD			DOUNDI	111	OF THE					
CD			ית עיס			ית	SCRTDT	TON		· _
CD										
CD			0	ъъ	TET POTT	VF				-
CD			0	DE	TODICI I	V Ľ				_
			10	- F 11 11 U						_
C D			10	WI	TID					-
(C										
(
CEOF	.'	\$								
CLOL										

-342-
APPENDIX D. MC^2-2 Binary Files. BIGXS(I) *********** PREPARED 3/10/75 AT ANL BIGXS1 THROUGH BIGXSP MACROSCOPIC ELASTIC SCATTERING THESE FILES ARE WRITTEN BY MC**2-II AREA 7 (CSC008) NDOJN NUMBER OF SOURCE GROUPS OF DATA

CD	NDO AN	NUMBER OF SOURCE CROOLS OF BUILT
CD	· · · ·	REPRESENTED IN EACH CROSS SECTION RECORD
CD	NDOWN1	NUMBER OF GROUPS OF DOWNSCATTER CHARACTERISTIC
CD		OF THE PROBLEM ISOTOPE OF LIGHTEST MASS

_____ C--SPECIFICATIONS (TYPE 1) CR С ۱. CC ALWAYS PRESENT FOR FILE BIGXS1 NEVER PRESENT FOR FILES BIGXS2 THROUGH BIGCSP CC С CL NDOWN, NDOWN1, NUMSPC С CW 3 С NUMBER OF BIGXS (I.E. BIGXS1 THROUGH BIGXSP) CD NUMSPC FILES REQUIRED FOR PROBLEM. THE MAX. NUMBER OF CD FILES ALLOWED IS 25. THE VALUE OF NUMSPC CD CD DEPENDS UPON THE NUMBER OF ULTRA-FINE-GEOUPS IN THE MULTIGROUP PORTION OF THE SPECTRUM CD CALCULATION AND THE MASS OF THE MATERIAL CD USED TO DETERMINE THE DATA MANAGEMENT STRATEGY .-CD THE VALUE OF NUMSPC IS SET BY THE CODE AT RUN CD CD TIME C C--

_____ CR CROSS SECTION (TYPE 2) С ALWAYS PRESENT CC С CL ((SIGS(I), I=1, NORD1), (SIGO(I), I=1, NDN1), (SIG1(I), I=1, NDN1), CL 1J=1, NDOWN) С

Ċ С

С

CF

CE C٠

CN

CN С

CD

C₩ C	NDOWN* (NOR	D1+ISP*NDN1)	-
CD	ISP=1 FOR	INCONSISTENT SPECTRUM OPTIONS (ISPOPT.LE.2)	-
CD	ISP=2 FOR	CONSISTENT SPECTRUM OPTIONS (ISPOPT.GE.3)	-
CD	NORD1	ORDER OF EXTENDED TRANSPORT APPROXIMATION+1	
CD	NDN1 = NDOWN	1+1	-
CD			-
CD	SIGS	LEGENDRE MOMENTS OF MACROSCOPIC SCATTERING	-
CD	•	CROSS SECTION THE THE J'TH GROUP.	-
CD		I=1 CORRESPONDS TO THE NORDER COMPONENT	-
CD		I=NORDER CORRESPONDS TO THE P1 COMPONENT	-
CD		ITNORD1 CORRESPONDS TO THE PO COMPONENT	-
ĊŮ	SIGO	MACROSCOPIC PO SCATTERING MATRIX ORDERED AS	-
CD		J TO J+I-1 SO THAT I=1 IS THE IN-GROUP TERM	-
CD	SIG1	MACROSCOPIC P1 SCATTERING MATRIX ORDERED AS	-
CD		J TO J+I-1 SO THAT I=1 IS THE IN-GROUP TERM	-
С			-
CN		THE J'TH GROUP IN THE ABOVE LISTS CORRESPONDS	-
CN		TO A PARTICULAR ULTRA-FINE-GROUP IN THE	-
CN		PROBLEM. THAT IS, FOR BIGXS1 THE J'TH	-
CN		GROUP CORRESPONDS TO THE J'TH ULTRA-FINE GROUP.	-
CN		FOR BIGXS2 THE J'TH GEOUP CORRESPONDS TO THE	-
CN		ULTRA-FINE-GROUP IN THE PROBLEM OF J+NDOWN,	-
CN		AND SO ON THROUGH BIGXSP.	-
CN		THE ARRAY SIG1 IS PRESENT ONLY FOR ISP=2	-
С			-
C			

CEOF

APPENDIX D. MC²-2 Binary Files. GEOM1

****5	* * * * * * *	* ** * * * * * * *	**************************************
С			-
С			PREPARED 6/13/75 AT ANL -
С			-
CF		GEOM1	-
CE		GEOMETRY	DATA -
C			-
CN			THIS IS AN ABBREVIATED VERSION OF THE -
CN	÷		DATA SET GEOM. IT IS USED FOR MC**2-II -
CN			CALCULATIONS
CN			ALL NON-INTEGER QUANTITIES IN THIS FILE ARE -
CN			REAL*8
С			-
C * * * *:	* * * * * * *	****	** * * * * * * * * * * * * * * * * * * *
an	w 11 7 m		A RAR TRA MACUINES 1 AMUERNISE
CD	MULT		2 FOR IBM MACHINES, I DINERWISE.
CD	NCMP		NUMBER OF COMPOSITIONS.
CD	NINTS		NUMBER OF MECH INTERVALD.
CD	NPTI		NUMBER OF MESH INTERVAL LINES.
CD	NREG		NUMBER OF REGIONS.
C			
CB		SPECIFICA	FIONS (TYPE 1) -
С			-
CL	NDJM,N	GEOM, NPTI	, NDUM, NINTI, NDUM, NREG, NCMP, NDUM, NDUM, NDUM, -
CL	1NDUM,N	DUM, NDUM	·
С			
CW	14		、
С			-
CD	NDIM		NUMBER OF DIMENSIONS=1
CD	NGEOM		GEOMETRY
CD			NGEOM=1, ONE-DIMENSIONAL SLAB
C D '			=2, ONE-DIMENSIONAL CYLINDER
CD	NDUM		DUMMY VARIABLE SET TO 0
С			- -
C			
			•
C			
CR		MESH INTE	RVAL BUUNDARIES (TIPE 2)
C			-
CL	(XMESE	(1), 1=1, N	
C			-
CW	MULT*N	I P.T.T	· · · · · · · · · · · · · · · · · · ·
C			-
CD	XMESH		MESH BOUNDARIES (REAL*8).
C		-	• •
1]====			

. .

- --- -

C--REGION-INTERVAL CORRESPONDENCE (TYPE 3) CR С CL(MP(I), I=1, NINTI)С CW NINTI С REGION INDEX. IF MR(I) = N, THEN REG(N) IS CD MR THE REGION LABEL IN INTERVAL J CD (SEE LABELS RECORD BELOW) . CD Ĉ. C-C---COMPOSITION-INTERVAL CORRESPONDENCE (TYPE 4) CR C · CL (MC(I), I=1, NINTI)С CW NINTT С COMPOSITION INDEX. IF MC(I) = N, THEN CNAME(N) CD MC IS THE COMPOSITION LABEL IN INTERVAL I CD (SEE LABELS RECORD BELOW). CD С C-_____ C--COMPOSITION-REGION CORRESPONDENCE (TYPE 5) CR С CL (NC(I), I=1, NREG)С CW NREG С COMPOSITION INDEX. IF NC(I) = N, THEN CNAME(N) CD · NC IS THE COMPOSITION LABEL IN REGION I CD (SEE LABELS RECORD BELOW) . CD С C -_____ C---LABELS (TYPE 6) CP С (REG(I), T=1, NREG), (CNAME(I), I=1, NCMP)CL С C٣ MULT* (NREG+NCMP)

-346-

C CD CD C	R E G C N A M E	REGION LABELS (REAL*8). COMPOSITION LABELS (REAL*8).
C		
CR C	REGION VOL	JMES (TYPE 7)
CL C	(VOL (I), I= 1, NREG))
CW C	MULT*NREG	-
CD CD CD C	AOT	VOLUME OF REGION REG(I) (REAL*8) (SEE LABELS RECORD ABOVE). VOL EQUALS THE VOLUME PEP UNIT HEIGHT IN THE TRANSVERSE DIRECTION.
	. ''	

CEOF

APPENDIX D. MC²-2 Binary Files. IRESCS

C****	******	* * * * * * * * * * * * * * *	< * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *
C				-
C			PREPARED 4/26/76 AT ANL	-
C				
C F		TRECC		-
CF			NEDORT RESONANCE CROSS SI	CTTONS -
C B		ININGWED INC	INSPORT RESONANCE CROSS SI	-
CN			PROND CROUD PREONINCE CN	
			GROAD GROOP RESONANCE CAL	TORE, FISSION,
C N			SCATTERING AND IRANSPER C	$\frac{10}{10} \frac{10}{10} 10$
CN			CALCULATED IN MC**2-II AN	$\frac{2}{2} = \frac{1}{2} \left(\frac{1}{2} \left(\frac{1}{2} \right) - \frac{1}{2} \right)$
CN			INTEGRAL TRANSPORT THEORY	-
С				
C * * * * *	******	****	* * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * *
			· .	
CD	MULT		2 FOR IBM MACHINES, 1 OT	HERWISE
CD	NBROAI) .	NUMBER OF BROAD GROUPS IN	N THE RABANL (CSC011)
C D ·			ENERGY RANGE	
CD	NCNTM		NUMBER OF RESOLVED RESON	NCE MATERIALS IN THE
CD		· •	PROBLEM MIXTURE EXCLUDING	G ANY UNIQUE FOIL
CD			MATERIALS	•
CD	NGEOM		GEOMETRY TYPE	
CD			NGEOM=0 FOR HOMOGENEOUS	PROBLEM
CD			NGEOM= 1 FOR SLAB GEOMETRY	
			NGEOM=2 FOR CYLINDRICAL (GEOMETRY
	NREG		NUMBER OF REGIONS IN THE	CELL, NREG=1 FOR
CD	11110		HOMOGENEOUS PROBLEMS	
C				
C S		FTLE STRUCT	IRE	_
C 5		LIDD SINCER		-
C5		σταιση πνοτ		DRESENT TE
C5 C5				
		SDRCIFICATI	NC ,	ATWAVS DRESENT -
		OPECIFICATION		ALWAIS PRESENT -
65		MATERIAL NAP		
CS	ata ata ata ata ata a	REGION NAMES		NGLOM.GI.U
CS	, ** ****	*** (REPEAT D	BROAD TIMES)	
CS	*	DOWNSCATTERS	<u> </u>	ALWAYS PRESENT -
CS	*	RESONANCE CH	ROSS SECTIONS	ALWAYS PRESENT -
CS	*	TRANSFER CRO	DSS SECTIONS	ALWAYS PRESENT -
CS	*****	* * *		-
CS		CELL CROSS S	SECTIONS	NGEOM.GT.0 -
CS		GROUPS SCAT	T ER ED	NGEOM.GT.O -
CS		CELL TRANSFI	ER CROSS SECTIONS	NGEOM.GT.0 -
С				-
C				

c-

-348-

CR	SPECIFICAT	IONS (TYPE 1)			· -
cc	ALWAYS PRE	SENT			-
C CL	NCNTM, NREG, NBROA	D,NGEOM	•		-
C CW	4				-
С С					-
C	MATERTAL N	AMES (TYPE 2)			
C	ALTING DDE				-
C	AL AYS PRE	SENT		•	_ , _ ,
CL ,C	RESNAM(M), M=1, NC	'N T M		• · ·	-
C W C	MULT*NCNTM				-
CD CD CD	RESNAM	<pre> REAL*8 NAMES AN INTEGRAL T: (CSC011) </pre>	OF RESONANCE (RANSPORT TREAT	MATERIALS HAVING IMENT BY AREA 10	; –) – –
C C					-
-					
C	PECTON NAM	FS (TVDF 3)			
C	ABOLON MAN				-
c	PRESENT IF	NGEON.GI.U	. •		-
CL C	RLABEL (N), N=1, NR	EG			-
CW C	MULT*NREG				-
C D C	RLABEL	REAL*8 NAMES	OF REGIONS		-
C					
C					
CR	DOWN SCATTE	RS (TYPE 4)		· · ·	-
C CC	ALWAYS PRE	SENT			-
C CL	NBGMAX (M), M= 1, NC	NTM			-
C CW	NCNTM			•	
~					_

-349-

APPENDIX D. MC^2-2 Binary Files. IRESCS (Contd.)

CD CD C C	N B G M A X	MAXIMUM NUMBER OF BROAD GROUPS DOWN-SCATTERED FOR EACH MATERIAL AND CURRENT BROAD GROUP	
C CP	RESONANCE	CROSS SECTIONS (TYPE 5)	
cc c	ALWAYS PR	ESENT	-
CL CL	((RESCAP(M,K),M 1 J=1,NREG),((R	= 1, NCNTM), K=1, NREG), ((RESFIS(M,K), M= 1, NC NTM), ESSCT(M,K), M= 1, NCNTM), K= 1, NREG)	
CW	3*NCNTM*NREG		-
CD CD	RESCAP (M, K)	RESONANCE CAPTURE CROSS SECTION FOR MATERIAL M AND REGION K FOR THE CURRENT BROAD GROUP	-
C D C D	RESFIS(M,K)	RESONANCE FISSION CROSS SECTION FOR MATERIAL M AND REGION K FOR THE CURRENT BROAD GROUP	-
CD CD CD	RESSCT(M,K)	RESONANCE SCATTERING CROSS SECTIONS FOR MATERIAL M AND REGION K FOR THE CURRENT BROAD GROUP	-
с с			-
		,	
C CR	TRANSFER	CROSS SECTIONS (TYPE 6)	-
CC CC	ALVAYS PR	ESENT	-
CL C	(((TRNSFR (M,K,J	, M = 1, NCNTM), K = 1, NREG), J = 1, NMAX)	-
CW C	NCN TM*NREG*NMAX	ι (-
CD CD CD	TRNSFR (M,K,J)	RESONANCE MATERIAL M TRANSFER CROSS SECTION FOR REGION K FROM THE CURRENT BROAD GROUP, SAY	-
CD CD CD C	N MA X	LARGEST OF THE NEGMAX FOR THE CURRENT BROAD GROUP FOR ANY OF THE NCNTM MATERIALS	-
с			
C CR C	CELL CROS	S SECTIONS (TYPE 7)	- - -
cc c	PRESENT I	F NGEOM.GT.O	-

-350-

CL ((CELCAP(M,K), M=1, NCNTM), K=1, NBROAD), ((CELFIS(M,K), M=1, NCNTM)),CL 1K=1, NBROAD), ((CELSCT (M,K), M=1, NCNTM), K=1, NBROAD) С CW '3*NCNTM*NBROAD С CELL AVERAGED RESOLVED RESONANCE CAPTURE CD CELCAP (M,K) CD CROSS SECTION FOR MATERIAL M AND BROAD GROUP K -CD CELFIS(M,K) CELL AVERAGED RESOLVED RESONANCE FISSION CROSS SECTION FOR MATERIAL M AND BROAD GROUP K -CD CELSCT(M,K) CD CELL AVERAGED RESOLVED RESONANCE SCATTERING CD CROSS SECTION FOR MATERIAL M AND BROAD GROUP K -С _____ C--GROUPS SCATTERED (TYPE 8) CR С PRESENT IF NGEOM.GT.0 CC С CL ((NGRPDN(M,K), M=1, NCNTM), K=1, NBROAD)С CW NCNTM*NBROAD С CD NGRPDN (M,K) NUMBER OF BROAD GROUPS DOWNSCATTERED BY CD MATERIAL M'IN BROAD GROUP K С _____ CR CELL TRANSFER CROSS SECTIONS (TYPE 9) С PRESENT IF NGEOM.GT.0 CC С CL ((CELTFP(M,K,J),M=1,NCNTM),K=1,NBROAD),J=1,MOST)С CW NCNTM*NBROAD*MOST С CD CELL AVERAGED TRANSFER CROSS SECTION FOR CELTFR (M,K,J) MATERIAL M FROM BROAD GROUP K TO BROAD GROUP CD CD K+J LARGEST OF THE NBGMAX FOR ANY BROAD GROUP CD MOST С C-

CEOF

APPENDIX D. MC²-2 Binary Files. LORENZ

C****	* * * * * * * * * * * * * * * * * * * *	****	*****
С			
С		PREPARED 3/05/75 AT ANL	
С		· ·	· _
CF	LORENZ		۰ ـ
CE	LORENTZIAN	SHAPE RESONANCE INTEGRALS	-
С			-
CN	•	THIS INTERFACE DATA SET IS	5 WRITTEN -
CN	1	BY MC**2-II AREA 6 (CSC00	5) –
CN	-	THIS DATA SET IS PRESENT (DNLY IF MAXNOL.GT.O -
C			
C****	* * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *	****
		¢	
CD	NOLINT	NUMBER OF ULTRA FINE GROUP	PS HAVING
CD		LORENTZIAN SHAPE RESONANCI	E INTEGRALS FOR
CD		EACH RESONACE SO TREATED	
CD	NOLRES	NUMBER OF RESONANCES GIVEN	N A LORENTZIAN SHAPE
CD		TREATMENT FOR EACH REGION	
CD	NREG1	1 FOR HOMOGENEOUS PROBLEMS	5, 2 FOR CYLINDRICAL
CD		PROBLEMS, 1+THE NUMBER OF	SLAB REGIONS FOR
CD		SLAB GEOMETRY PROBLEMS	
a			
C			
CS CC	FILE STRUCT	URE	-
			-
CS CS	RECORD TYPE		PRESENT IF -

	SPECIFICATI		ALWAYS -
CS	******** (REPEAT	NREGI TIMES)	-
CS CS	* RESONANCE S	PECIFICATIONS	ALWAYS -
CS CS	* RESUNANCE N	UMBERS	NOLRES.GT.0 -
CS	* **** (REPEAT	NOLRES TIMES)	-
CS	* * ULTRA FINE	GROUP NUMBERS	NOLRES.GT.0 -
CS	* * LORENTZIAN	RESONANCE INTEGRALS	NOLRES.GT.0 -
CS	* * * * * * * *	· · · ·	· -
C			· -
C			
C			
CP	SDECTETCAUT		
C	SPECIFICATI	UND (TIPE 1)	
C C C	ATUAVE DDDC	ត សាល	-
	ATARIS LKP2		-
CI	MAVNOT MAYODD MAY	יו מי	
С Г	HAANUL, HAAGKE, MAX	3rn	-
	2		
C ^w	J		-
C			-

-352-

CD CD CD CD C C	MAXNOI. MAXGRP MAXGPH	MAXIMUM VALUE OF NOLRES OVER ALL REGIONS - MAXIMUM VALUE OF NOLINT OVER ALL REGIONS - MAXIMUM VALUE OF NOLINT OVER THE HOMOGENEOUS - MIXTURE (REGION 1) -
		· · ·
C CR	RESONANCE S	PECIFICATIONS (TYPE 2) -
C CC	ALWAYS PRES	ENT -
C	NOLRES, N1MIN, N2MA	x -
C CW	3	
C CD	NIMIN	HIGHEST ENERGY ULTRA FINE GROUP HAVING
CD CD CD C	N2MAX	LORENTZIAN SHAPE RESONANCE INTEGRALS - LOWEST ENERGY ULTRA FINE GROUP HAVING - LORENTZIAN SHAPE RESONANCE INTEGRALS -
C		·
C CR	RESONANCE N	UMBERS (TYPE 3)
cc	PRESENT ONL	Y IF NOLRES.GT.0
CL	LRESNO(I), I=1, NOL	RES -
CW C	NOLRES	
CD CD C	LRÈSNO	RESOLVED RESONANCE NUMBERS FOR RESONANCES - HAVING A LORENTZIAN SHAPE TREATMENT -
C	ULTRA FINE	GROUP NUMBERS (TYPE 4)
C CC	PRESENT ONL	Y IF NOLRES.GT.0
CL	NOLINT, N1, N2	-
CW C	3	
CD	N 1	HIGHEST ENERGY ULTRA FINE GROUP HAVING -

-353-

•

.

.

ò

C D C D C D C	N 2	LORENTZIAN SHAPE RESONANCE INTEGRALS - LOWEST ENERGY ULTRA FINE GROUP HAVING - LORENTZIAN SHAPE RESONANCE INTEGRALS -
C		
C		
CR	LORENTZIAN	RESONANCE INTEGRALS (TYPE 5)
Ç.		-
cc	PRESENT ONL	Y IF NOLRES.GT.O -
С		
CL	CJINTL(I), I=1, NOL:	- ENT -
С		·
CW	NOLINT	
С		» • • •
C D C	CJINTL	LORENTZIAN SHAPE RESONANCE INTEGRALS -
C		

CEOF

APPENDIX D. MC²-2 Binary Files. MACTOT

C****	* * * * * *	* * * * * * * * * * *	****	* * * * * * * * * * * * * * * * * * * *
С				-
С	•	I	PREPARED 3/06/75 AT ANL	-
С				-
CE		MACTOT		-
CE		MACROSCOPIC	TOTAL CROSS SECTIONS	· -
С		د	•	· -
CN			THIS DATA SET IS WRITTEN B	Y MC**2-II AREA 5 -
CN			(CSC004), AREA 6 (CSC005)	AND/OR AREA 6.5 -
CN			(CSC006)	- ·
С				-
C****	*****	* ** * * * * * * * * *	**************************************	******
CD	NDEC		NUMBER OF PECORDS IN THE D	ልጥል ዓድጥ
	NRSC		NDEC-1 FOR HOMOGENEOUS DRO	RIFMS
			NREC-3 FOR CVIINDRICAL GEO	METRV
			NREC-3 FOR CILINDRICAL GEO	B PECTONS FOR
CD:			SING GEOMETRY	, KEGIONS IOM
CD			STAD GLONEIKI	· · ·
C				
CS		FILE STRUCTU	RE	· _
CS			· ·	
CS		RECORD TYPE	ν.	PRESENT IF -
CS		=============	=======================================	
CS	* * * * * *	** (REPEAT N	IREC TIMES)	· –
CS	*	MACROSCOPIC	TOTAL CROSS SECTION	ALWAYS -
CS	* * * * * *	* *		-
С				· –
C				
C				
CR C		MACROSCOPIC	TOTAL CROSS SECTION (TYPE	
cc		ALWAYS, PRESE	INT	-
CL	(STGMA	T(I), I=1, NGE	(OUP)	-
C C₩	NGROUP			- -
C				-
CD	SIGMAT	I	MACROSCOPIC TOTAL CROSS SE	CTION (1/CM) -
CD	NGROUP)	NUMBER OF ENERGY GROUPS	- · · · · · -
c			. • .	· · · · · · · · · · · · · · · · · · ·
C				
-				•

CEOF

C****	*****	****	* * * * * * * * * * * *	*****	****	*****	****	****
Ċ								-
Č .			PREPARED	3/10/75	ል ጥ	ANT.		-
c		\$		5/10/15				_
C P		MACTOT						-
Cr an		MACIUI			CEC	TONE		_
CE		ULTRA FINE	GROUP TOTA	L CROSS	250	2110112		_
С								-
CN			BOTH CONF	IGURATI	ONS	OF TH.	LS SCRATCH DATA	263 -
CN			MACTOT AR	E WRITT	EN I	BY MC≭:	*2-11 AREA 10	
CN			(CSC011).	THE FI	RST	CONFI	GURATION CONTAIN	
CN		•	ULTRA FIN	E GROUP	MIC	CROSCO	PIC TOTAL CROSS	-
CN			SECTIONS.	THE 3E	CONI) CONF	IGURATION CONTAI	NS -
CN			ULTRA FIN	E GROUP	MAG	ROSCO	PIC TOTAL CROSS	-
CN			SECTIONS	. ,				-
C								
~ ~****	****	***	* * * * * * * * * * * *	******	****	*****	****	****
0		• •						
							· · · · · · · · · · · · · · · · · · ·	·····
CD	т 1		תיון האמדת	RA FTNF	GRO	יזא קוור	AD FOR CHERENT F	PASS
	⊥ ' T 2			A PINF	GROI	ID REAL	D FOR CHRRENT PA	55
CD	12 ND3 C C		NUMBED OF	ים אבי דע סיד 1	01.01 DT0	0 P D S	DRESENT IN THE F	קודי
	NPASS	· 、			יםת נתכח	NDASS.	- 111 11 110 1 10 1110 110 100 1	ONE
			LT NULINA	T DECOD	יע אני ומ מ	NPADO		O N G
CD			ADDITIONA NUITED ND	L RECOR	onn Cnn		ETNE CROUPS	
CD		· . ·	NULTRA-NP	ASSANUF	GRD	ULTRA	FINE GROUPS	
CD	NPRMTC		NUMBER OF	MATERI	ALS	IN TH.	E PROBLEM MIXTUR	(E
CD			WHICH HAV	E TABUL	ATEI) TOTA	L CRUSS SECTIONS	S
CD			IN THĘ AR	EA 10 (CSC)11) E	NERGY RANGE	
CD	NUFGRI)	NUMBER OF	ULTRA	FINI	E GROU	PS READ PER PASS	5.
CD			NUFGRD=I2	-I1+1°	•		ι,	
С	* * *	* * * * * * * * * * * * *	* * * * * * * * * * * *	*****	÷			
С	.*					•		
C====	:===*	FIRST COL	NFIGURATION	*	====	=====	=======================================	=======
С	*	v		*				
C	***	****	* * * * * * * * * * *	*****			•	
				~				
C								
CS		FILE STRUC	TURE					-
CS								-
CS		RECORD TYP	E .				PRESENT IF	-
cš		=================	===============	======	====	====	=======================================	=======
CS	*****	*** (REPELT	NPRMTO TTM	ES)				-
C 9	*	(), Dr. DAT	lithito 11	,			THE CURRENT MAT	FERIAL -
C3 C6	*	ת אסת דו	CROUP TOTA	т			HAS TOTAL CROSS	-
C3 C5		OPTOM STNE	TONS				SECTIONS IN THE	ARFA -
C.5	т Т	CAUSS SECI.	1000					
C5 CC	۰۰ مد							, <u> </u>
CS aa	ېت د د د د د د د د د د د	ملك ماك م					RANGE	, –
US	***	የ ጥ ሳኑ						

;

С C----ULTRA FINE GROUP TOTAL CROSS SECTIONS (TYPE 1) CR С CC ALWAYS PRESENT С CL (SIGTOT(I), I=NITUFG, NGROUP) С CW NGPOUP-NITUFG+1 С SIGTOT(I) ULTRA FINE GROUP MICROSCOPIC TOTAL CROSS CD SECTION FOR ULTRA FINE GROUP I CD HIGHEST ENERGY ULTRA FINE GROUP ON THE AREA 10 -CD · NITUFG (CSC011) ENERGY RANGE CD LOWEST ENERGY ULTRA FINE GROUP IN THE PROBLEM CD NGROUP С CEOF **** С С * SECOND CONFIGURATION C====== * С * С _____ C-CS FILE STRUCTURE CS PRESENT IF CS RECORD TYPE CS CS ******* (REPEAT NPASS TIMES) ULTRA FINE GROUP MACROSCOPIC TOTAL ALWAYS PRESENT CS CS CROSS SECTIONS CS ****** ULTRA FINE GROUP MACROSCOPIC TOTAL NULTRA.GT. CS NUFGRD*NPASS CS CROSS SECTIONS С _____ C----ULTRA FINE GROUP MACROSCOPIC TOTAL CROSS SECTIONS (TYPE 1) -CR С CC ALWAYS PRESENT

.

С	• • •	· · · · · · · · · · · · · · · · · · ·	
CL	((SIGTOT(M,I),M=	=1,NCMPFL),I=I1,I2) -	
С		۰ · · · · -	
CW	NCMPFL*NUFGRD	· · · · · · · · · · · · · · · · · · ·	
Ç		-	
C.D	SIGTOT(M,I)	ULTRA FINE GROUP MACROSCOPIC TOTAL CROSS -	
CD		SECTION FOR COMPOSITION M AND ULTRA FINE -	
CD.	•	GROUP I -	
CD	NCMPFL	NUMBER OF COMPOSITIONS PLUS NUMBER OF FOILS -	
CD		IN THE PROBLEM	
C		-	
CN		NUFGRD ULTRA FINE GROUPS ARE READ FOR EACH PASS-	
CN		WHERE NUFGRD=12-11+1. THERE WILL BE -	
CN		ONE ADDITIONAL RECORD AFTER THE NPASS RECORDS -	
CN		WHICH CONTAINS NULTRA-NPASS*NUFGRD ULTRA FINE -	
CN		GROUPS IF NULTRA IS NOT AN INTEGRAL MULTIPLE OF-	
CN	• . •	NUFGRD. NULTRA IS THE NUMBER OF ULTRA FINE -	
CN		GROUPS IN THE AREA 10 (CSCO11) ENERGY RANGE -	
C ·		- · · · · · -	
C			
		•	

CEOF

APPENDIX D. MC²-2 Binary Files. MICTOT

C****	*****	* * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *	** ** *
c c c		PREPARED 3/06/75 AT	ANL	- - -
CF CE C	MICTOT MICROSCOPIC	TOTAL CROSS SECTIONS	1	-
CN CN CN		THIS DATA SET IS WRI (CSC004), AREA 6 (CS (CSC006)	TTEN BY MC**2-II AREA 5 C005) AND/OR AREA 6.5	-
C * * * *	*****	*****	****	** * * *
CD	NPRMAT	NUMBER OF PROBLEM MA	TERIALS	
c cs	FILE STRUCT	URE		
CS CS	RECORD TYPE	}	PRESENT IF	
CS CS CS C C C	******** (REPEAT * MICROSCOPIC *******	FOR NPRMAT MATERIALS) TOTAL CROSS SECTION	ALWAYS	- - - -
C CR	MICROSCOPIC	TOTAL CROSS SECTION	(TYPE 1)	
C CC C	ALWAYS PRES	ENT		 -
CL .	(SIGMAT(I), I=1, NG	ROUP)		-
CW C	NGROUP			- -,
CD CD C	SIGMAT NGROUP	MICROSCOPIC TOTAL CR NUMBER OF ENERGY GRO	OSS SECTION (BARNS) UPS	- - -
C				

CEOF

APPENDIX D. MC²-2 Binary Files. OLDSGS

C******	*****
С	- · · · · · · · · · · · · · · · · · · ·
с	PREPARED 3/19/75 AT ANL -
с	· · · · · · · · · · · · · · · · · · ·
CF OLDSGS	-
CE MICROSCOPIC	ELASTIC SCATTERING -
Ċ	• •
CN	THIS FILE IS READ BY MC**2-II AREA 7 (CSC008)
CN	THE FORMAT OF THIS DATA SET IS IDENTICAL -
CN	TO THAT OF DATA SET SMSIGS -
с	-
C******	* * * * * * * * * * * * * * * * * * * *

CEOF

1

APPENDIX D. MC^2-2 Binary Files. OPTICL С PREPARED 3/06/75 AT ANL C С CF OPTICL CE SLAB GEOMETRY OPTICAL THICKNESSES С THIS DATA SET IS WRITTEN BY MC**2-II AREA 5 CN (CSC004), AREA 6 (CSC005) AND/OR AREA 6.5 CN (CSC006) AND IS PRESENT IF NGEOM.EQ.1 CN (SLAB GEOMETRY) CN С CD NREG NUMBER OF SLAB REGIONS C-FILE STRUCTURE CS CS PRESENT IF CS RECORD TYPE CS ******* (REPEAT FOR NREG SLAB REGIONS) CS ALWAYS CS * SIGLXL SIGRXR CS ***** С C-_____ C-----CR SIGLXL SIGRXR (TYPE 1) С ALVAYS PRESENT CC С ((SIGLXL(I,K), I=1, MAXHTM), K=1, NGROUP),CL 1((SIGRXP(I,K), I=1, MAXHTM), K=1, NGROUP)CL С CW MAXHTM*NGROUP*2 С OPTICAL THICKNESS TO THE LEFT OF THE CURRENT CD SIGLXL SLAB REGION FOR MATERIAL I AND GROUP K. THE CD CONNECTION BETWEEN I AND PROBLEM MATERIAL CD IS DETERMINED BY THE ARRAY MATHET (SEE CD DATA SET PRBSPC) CD OPTICAL THICKNESS TO THE RIGHT OF THE CURRENT CD SIGRXR SLAB REGION FOR MATERIAL I AND GROUP K. THE CD CONNECTION BETWEEN I AND PROBLEM MATERIAL IS CD DETERMINED BY THE ARRAY MATHET (SEE CD DATA SET PRBSPC) CD

,

CD	MAXHTM	MAXIMUM NUMBER OF MATERIALS IN ANY	-
CD		HETEROGENEOUS REGION	_
CD	NGROUP	NUMBER OF ENERGY GROUPS	_
С			-
CN		SOME REGIONS WILL HAVE SIGLXL AND SIGRXR	-
CN		FILLED OUT WITH ZEROS IF THERF ARE FEWER	_
CN	• .	MATERIALS THAN MAXHTM TREATED HETEROGENEOUSLY	-
CN		IN THAT REGION	_
Ç.			-
C			
	•		

СЕОР

-362-

APPENDIX D. MC²-2 Binary Files. PRBCHL

C****	*****	* * * * * * * * * * * * * * * * * * * *	*****
C			-
C C		PREPARED 3/12//5 AT ANL	-
C C F		PBBCHT	- · · · · -
CE		FISSION SPECTRUM DATA	
С		, I	. –
CN		THIS FILE IS WRITTEN BY MO	C**2-II AREA 7 -
CN		(CSC008)	
C	*****	* * * * * * * * * * * * * * * * * * * *	
Ç * * * * *			
CD	NCHI	FISSION SPECTRUM VECTOR O	R MATRIX FLAG
CD		NCHI=1 FOR VECTOR CHI	
CD		NCHI.GT.1 FOR MATRIX CHI	
CD	NGROUI	P NUMBER OF UFG IN MC**2 LI	SRARY ·
CD ·		NUMBER OF FIGINN SPECTRA NUMBER OF FIGINN SPECTRA	TN PROBLEM
CD	NUMERI	NONBER OF FISSION SPECIAL	IN PRODUM
C			
CS		FILE STRUCTURE	-
CS			
CS		RECORD TYPE	PRESENT IF -
.CS			
		DEARLEN MATERIAL CORRESPONDENCE	
CS CS		FISTON SPECTRUM FLAG	ATWAYS -
CS		FISSION SPECTRUM USE COUNT	ALWAYS -
CS	* * * * * *	*** (REPEAT FOR NUMCHI SPECTRA)	-
CS	*	FISSION SPECTRUM VALUES	NCHI.EQ.1 -
CS	*	FISSION SPECTRUM PARAMETERS	NCHI.GT.1 -
CS	* ***	*** (REPEAT FOR ALL ISOTOPES USING	-
CS	* *	SPECTRUM)	-
CS	* *	FJSSION CROSS SECTIONS	NUMCHI.GT.1 -
CS	*****	< * *	-
С			-
U			
C			
CR		SPECIFICATIONS (TYPE 1)	-
С			, -
CC		ALWAYS PRESENT	· · · · · -
С			-
CL	NUMCHI		-
C	4	·	- -
CW	1		-
L			

-363-

C--PROBLEM MATERIAL CORRESPONDENCE (TYPE 2) CR С CC ALWAYS PRESENT С, (IFIS(I), I=1, NPRMAT)CL С NPRMAT CW 15 С FISSION SPECTRUM IDENTIFICATION ĊD IFIS IFIS=0 PROBLEM MATERIAL IS NOT FISSIONABLE CD IFIS=N PROBLEM MATERIAL USES THE N'TH FISSION CD SPECTRUM CD С _____ C----CR FISSION SPECTRUM FLAG (TYPE, 3) С CC ALVAÝS PRESENT С CL (NCHI(I), I=1, NUMCHI)С CW NUMCHI С -----c c--_____ CR FISSION SPECTRUM USE COUNT (TYPE 4) С ALWAYS PRESENT CC С CL(LCHI(T), I=1, NUMCHI)С C₩ NUMCHI С NUMBER OF PROBLEM MATERIALS USING A GIVEN CD LCHI FISSION SPECTRUM DISTRIBUTION CD С C -_____ C----FISSION SPECTRUM VALUES (TYPE 5) CR C٠

-364-

APPENDIX D. MC²-2 Binary Files. PRBCHI (Contd.) CC PRESENT IF NCHI.EQ.1 С CL(CHI(I), I=1, NGRP)С CW NGRP С CD CHI FRACTION OF FISSION NEUTRONS BORN IN GROUP С C---C--CR FISSION SPECTRUM PARAMETERS (TYPE 6) С CC PRESENT IF NCHI.GT.1 С CL (BETA(I), I=1, NGROUP), (ALPHA(I), I=1, NGROUP), CL1 (TAU (I), I = 1, NGROUP), (ANORM (I), I = 1, NGRP) C CW $3 \times NGROUP + NGRP$. С CD ALPHA FRACTIONAL PROBABILITY OF USING EVAPORATION CD SPECTRUM LAW. 1-ALPHA IS FRACTIONAL CD PROBABILITY OF USING FISSION DISTRIBUTION LAW. CD BETA NUCLEAR TEMPERATURE (EV.) FOR EVAPORATION SPECTRUM LAW CD TAU CD NUCLEAR TEMPERATURE (EV.) FOR FISSION DISTRUBUTION LAW CD CD FISSION SPECTRUM NORMALIZATION FACTOR ANORM С C-. C--FISSION CROSS SECTIONS (TYPE 7) CR С CC PRESENT IF NUMCHI.GT.1 С CL (BSIGF(I), I=1, NGRP)С CW NGRP С CD BSIGF **ISOTOPIC ATOM DENSITY*NUMBER OF NEUTRONS** CD PER FISSION*MICROSCOPIC FISSION CROSS SECTION С ------C-

CEOF

-365-

.

C****	* * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *
С		· · · · · · · · · · · · · · · · · · ·
C		PREPARED 10/29/75 AT ANL -
С		
CF	PRBSPC	-
CE	BINARY VERS	ION OF MC**2-II BCD INPUT DATA -
С		-
С	,	-
C****	* * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *
•		
CD ·	BSQ1	FIXED OR FIRST GUESS FOR BUCKLING
CD	ICHT	ICHI=0, USE LIBRARY FISSION SPECTRA FOR ALL
CD.		MATERIALS IN THE PROBLEM
CD	,	=1, USE FISSION SPECTRUM OF MATERIAL
CD		IDENTIFIED IN FISSION SPECTRUM RECORD
CD		FOR ALL MATERIALS IN THE PROBLEM
CD		=NPRMAT, USE FISSION SPECTRUM OF MATERIAL
CD		GIVEN IN FISSION SPECTRUM RECORD
CD		/ FOR THE PROBLEM MATERIAL I
CD	MAXHTM	MAXIMUM NUMBER OF MATERIALS TREATED
CD		HETEROGENEOUSLY IN ANY REGION. FOR
CD	- -	CYLINDRICAL PROBLEMS, ONLY REGION 1 IS
CD	•	CONSIDERED
CD	MULT	2 FOR IBM MACHINES, 1 OTHERWISE
CD	NBG1	1+THE NUMBER OF EPI-THERMAL BROAD GROUPS
CD	NCMP	NUMBER OF COMPOSITIONS FOR USE IN HETEROGENOUS
CD		REGIONS AS DEFINED ON DATA SET A.NIP TYPE 14
CD		CARDS
CD ·	NESE	NUMBER OF POINTS IN THE FIXED UNRESOLVED
CD		RESONANCE ENERGY MESH
CD	NFLMAT	NUMBER OF MATERIALS USED IN FOILS IN THE
CD		INTEGRAL TRANSPORT THEORY CALCULATION
CD	NFOILS	NUMBER OF FOILS TO BE EDITED IN THE INTEGRAL
CD		TRANSPORT THEORY CALCULATION
CD	NGROUP	NUMBER OF UFG IN THE PROBLEM
CD.	NPRMAT	NUMBER OF MATERIALS IN THE HOMOGENEOUS
CD		MIXTURE OF THE PROBLEM
CD	NRFG	NUMBER OF HETEROGENEOUS REGIONS. NREG=1 FOR
CD		A HOMOGENEOUS PROBLEM (NGEOM=0), NREG=2 FOR
CD		CYLINDRICAL GEOMETRY (NGEOM=2), AND NREG EQUALS
CD		THE NUMBER OF PLATES FOR SLAB GEOMETRY (NGEOM=1)
	·	
C		
CD	SPECIFICATI	ONS (TYPE 1) -
С		-
сс	ALWAYS PRES	ENT -
С		· · · · · · · · · · · · · · · · · · ·
CL	NPRMAT, NGROUP, MGC	SD,NGROP,NORDER,ISPOPT,ICSOPT,ITPANS,NGEOM,

.1

-366-

CL 1NREG, NHFG, ICHI, IBSQ, MAXSIZ, MAXBLK, THETA, CAPSQR, AMIN, MAXHTM, 2MATSLB, NOHET, NESF, NUNRES, NBG1, NOVRLP, NCAND, IPRINT, NCSD, NCMP, CL CL 3A1, A2, ISORS, NITBG, NUFGIG, NHFGDW, NFOILS, NINGRP, NLEAK, NFLMAT, CL 4RESTST, HOMTEM, INELAS, NHYDRO С 43 CW С CD MGCSD UFG NUMBER AT WHICH THE CONTINUOUS SLOWING CD CALCULATION BEGINS CD NGROP LIBRARY UFG NUMBER CORRESPONDING TO THE CD HIGHEST UFG OF THE PROBLEM CDNORDER ORDER OF THE EXTENDED TRANSPORT APPROXIMATION CD ISPOPT SPECTRUM OPTION CD ISPOPT=1, P1 CD =2, B1 CD =3, CONSISTENT P1 CD =4, CONSISTENT B1 CD ICSOPT CONTINUOUS SLOWING DOWN PARAMETER OPTION CD ICSOPT=0, IMPROVED GOERTZEL-GREULING CD =1, ORDINARY GOERTZEL-GREULING CD TRANSPORT APPROXIMATION ITRANS CD ITRANS=0, USE ALL LEGENDRE COMPONENTS CD =1, USE STANDARD TRANSPORT APPROXIMATION CD =2, USE IMPROVED TRANSPORT APPROXIMATION CD NGEOM GEOMETRY OPTION CD NGEOM=0, HOMOGENEOUS CD =1, SLAB GEOMETRY CD =2, CYLINDRICAL GEOMETRY CD NHFG MAXIMUM NUMBER OF HYPER FINE GROUPS PER CD ULTRA FINE GROUP TO BE USED IN CALCULATING CD ELASTIC SCATTERING MATRICES CD IBSQ BUCKLING OPTION CD IBSQ=-1, DO NOT ITERATE ON BUCKLING AND CD BUCKLING IS NOT ENERGY DEPENDENT CD =0, ITERATE ON BUCKLING AND BUCKLING IS CD NOT ENERGY DEPENDENT CD =1, DO NOT ITERATE ON BUCKLING AND CD BUCKLING IS GROUP DEPENDENT CD MAXSTZ NUMBER OF REAL*8 WORDS ASSIGNED TO THE MAIN CD CORE CONTAINER ARRAY CD MAXBLK NUMBER OF REAL*8 WORDS ASSIGNED TO THE BULK CD CORE CONTAINER ARRAY CD THETA CONTINUOUS SLOWING DOWN INTEGRATING FACTOR CD CAPSOR KAPPA SQUARED, THE BUCKLING USED IN CD CALCULATING THE EXTENDED TRANSPORT CROSS CD SECTION AND THE LEAKAGE TERM FOR THE CD RESONANCE INTEGRAL CALCULATIONS CD AMIN MASS OF MATERIAL USED IN DETERMINATION OF THE CD DATA MANAGEMENT STRATEGY FOR THE SPECTRUM CD CALCULATION. AMIN.LE. THE MASS OF THE

-367-

CD		LIGHTEST MATERIAL IN THE PROBLEM (EXCLUDING H) -
CD	MATSLB	MATSLB=NPRMAT IF NGEOM.EQ.1. OTHERWISE MATSLB=1-
CD	NOHET	NOHET=NPRMAT IF NGEOM.GT.O. OTHERWISE NOHET=1 -
CD	NUNRES	RESOLVED RESONANCE OVERLAP OPTION -
CD		NUNRES=0, INCLUDE OVERLAP CALCULATION FOR UP -
CD		TO NCAND NEIGHBORING RESONANCES -
CD		=1. OMIT OVERLAP CALCULATION -
CD	NOVRLP	UNRESOLVED RESONANCE SELF-OVERLAP OPTION -
CD	· · · · · · · · · · · · · · · · · · ·	NOVRLP=0. INCLUDE SELF-OVERLAP CALCULATION -
CD		= 1. OMIT SELF-OVERLAP CALCULATION -
CD	NCAND	NUMBER OF RESOLVED RESONANCES TO BE TESTED -
CD		FOR OVERLAP ON EACH SIDE OF EACH RESOLVED -
CD		RESONANCE. NCAND WILL BE 1 IF NUNRES=1 -
CD	IPRINT	BPOTNTER DEBUGGING PRINT FLAG
CD		IPRINT=0. NO DEBUGGING PRINTOUT -
CD		=1, DEBUGGING DUMP PRINTOUT
CD		= 2 DEBUGGING TRACE PRINTOUT
CD		=3. FULL DEBUGGING PRINTOUT -
CD	NCSD	NUMBER OF ULTRA FINE GROUPS PER CONTINUOUS -
CD	NCO D	SLOWING DOWN GROUP
CD	Δ 1	CONSTANT USED IN THE EQUIVALENCE PRINCIPLE -
CD	Δ2	CONSTANT USED IN THE EQUIVALENCE PRINCIPLE -
CD	TSORS	TE O NO EXTERNAL SOURCES ARE USED. TE 1
CD	TOCHO	FYTERNAL SOURCES ARE USED. II I
CD	NTTRC	HIGHEST ENERGY BROAD GROUD NUMBER FOR WHICH -
CD CD	N L I D G	TNTEGENT TENERGI DRORD GROOF RONDER FOR SHICK
CD		CPOSS SECTIONS MAY BE CALCULATED TE NUMBER -
CD		THE INTEGRAL TRANSPORT THEORY CALCULATION WILL -
CD		NOT BE INVOKED -
	NUEGTO	NUMBER OF ULTRA FINE CROUDS DER INTERMENTATE
	NOTGLG	
CD	NHECDE	NUMBER OF FURE FINE COUDS DEP DODIER WITTEN
CD	N II G U V	
	ΝΤΝΟΡΟ	
CD	NINGRE	NUCCOD-O INCLUDE INCOMID SCATTERING OF -
CD		INTROLE-U, INCLUDE INGROUP SCHILERING IN THE
	•	
CD	NT ፑአሄ	TEANSURE INVICE OPTION
CD	NUDAN	INANSVERSE LEARAGE OFIION
CD CD		NEERR-O, ONTI THE INFROMETAL DERNAGE COFRECTION-
CD		-1 INCLUDE TRANSPORT CALCULATION -
		-1, INCLUDE TRANSVERSE LEARAGE CORRECTION -
CD CD		CRIMERION HERD FOR CRIECTION OF DECONDUCES TO
	RESTST	CRITERION USED FOR SELECTION OF RESONANCES TO -
		DE INCLUDED IN THE INTEGRAL TRANSPORT -
	ПОМФЕМ	- UALCULATION -
	ноалра	TEMPERATURE OF HUMUGENEOUS MIXTURE FOR INTEGRAL-
CD CD		TRANSPORT CALCULATIONS -
	INELAS	INELASTIC AND (N, ZN) ULTRA-FINE-GROUP TREATMENT-
CD CD		INGLASEU, KIGOROUS TREATMENT
CD		≈1, APPROXIMATE TREATMENT

h

.

-368-

CD CD CD CD CD C C C	NHYDRO	HYDROGEN TREATMENT NHYDRO=0, HYDROGEN IS NOT IN THE PROBLEM MIXTURE = 1, HYDROGEN IS PRESENT IN THE PROBLEM MIXTURE
C		
CR	PROBLEM NAM	LS (TIPE 2)
CC	ALWAVS PRES	- ምእጥ
C		-
CL	(PRBNAM(I), I=1, NP)	RMAT), (ALIAS(I), I=1, NPRMAT), -
CL	1 (IEDIT (I), I=1, NPR	MAT), (ISTYPE (I), I=1, NPEMAT),
CL	2 (EFISS(I), I=1, NPR	MAT), (ECAP(I), I=1, NPRMAT) -
С		-
CW	2*MULT*NPRMAT+4*N	PRMAT -
с		· · · · · · · · · · · · · · · · · · ·
CD	PRBNAM	REAL*8 NAMES OF LIBRARY NUCLIDES COPRESPONDING -
CD	7	TO PROBLEM MATERIALS -
CD	ALIAS	REAL*8 ALIAS NAMES OF PROBLEM MATERIALS -
CD	IEDIT(I)	O IF MATERIAL I IS TO BE ADDED TO THE OUTPUT -
CD		CROSS SECTION, DATA SET, 1 IF MATERIAL I IS NOT -
CD		TO BE ADDED TO THE OUTPUT DATA SET -
CD	ISTYPE	MATERIAL TYPES -
CD		ISTYPE=1, FISSILE -
CD		=2, FERTILE -
CD		=3, OTHER ACTIVIDE -
CD		=4, FISSION PRODUCT -
CD		=5, STRUCTURAL =
		=0, COULANT -
	ססדקע	-/, CONIROL -
	ELTOO ECND	ENERGI RELEASE FOR FISSION (MEV/FISSION) -
CD C		ENERGY RELEASE FOR CAFIORE (HEV/CAFIORE)
C		_
с с		
CR C	HOMOGENEOUS	ATOM DENSITIES (TYPE 3)
cc c	ALWAYS PRESI	ENT -
CL C	(ATMDEN(I),I=1,NPI	R MAT)
CW C	NPRMAT	
CD	ATMDEN	ATOMIC DENSITIES OF PROBLEM MATERIALS IN THE -

•

-369-

CD		HOMOGENOUS MIXTURE -
C		
	۲	
C CR	TEMPERATURE	S (TYPE 4) -
c cc	ALWAYS PRES	ENT -
C C L	(TEMP(I),I=1,NPRM	AT), (CHITEM (I), I= 1, NPRMAT) -
C C W	2*NPRMAT	
C CD CD CD CD CD	TEMP CHITEM	TEMPERATURES OF PROBLEM MATERIALS, DEGREES K FISSION SPECTRUM TEMPERATURES IN E.V. FOR THE - PROBLEM MATERIALS. CHITEM(I)=0.0 IF MATERIAL - I WAS NOT SPECIFIED ON THE DATA SET A.MCC2 - TYPE 23 CARDS
C		
C CR	HETEROGENOU	S ATOM DENSITIES (TYPE 5) -
c cc	PRESENT ONL	Y IF MAXHTM.GT.0
C CL	((HETDEN(I,K),I=1	,NPRMAT),K=1,NREG) -
C C W	NPRMAT*NREG	
C CD CD C	HETDEN	ATOMIC DENSITIES OF MATERIALS IN HETEROGENOUS - REGIONS -
C		
C CR	HETEROGENOU	S MATERIAL DESIGNATIONS (TYPE 6) -
c cc	PRESENT ONL	Y IF NGEOM.GT.O -
C I	((MATHET (I,K),I=1	, NPRMAT) , K=1, NREG 1) -
CW CW	NFRMAT*NFEG1	
C CD CD CD	MATHET	MATHET (I,K) IS .GT.O IF MATERIAL I IS - A RESONANCE MATERIAL AND IS TREATED - HETEROGENOUSLY IN REGION K. OTHERWISE MATHET=(

-370-

.

i

.

CD CD CD CD CD CD C C	NREG1	MATHET (I,K) IS INDEXED BY 1 FOR EACH MATERIAL I TO BE TREATED HETEROGENOUSLY IN EACH REGION K. ANY MATERIAL HAVING MATHET=0 WILL NOT BE INVOLVED IN THE CALCUALTION OF RESONANCE OVERLAP FOR ANY OTHER MATERIAL IN THAT REGION 1 IF NGEOM=2, NREG IF NGEOM=1	
C CR		ATOM DENSITIES TIMES LEFT SLAB THICKNESSES (TYPE 7)	-
čc		PRESENT ONLY IF NGEOM.EQ.1 AND MAXHTM.GT.0	-
CL	(((SXI	. (I, J, K), I=1, MAXHTM), J=1, NPRMAT), K=1, NREG)	-
CW CW	MAXHTM	I*NPRMAT*NREG	-
CD CD CD CD CD CD CD CD CD CD CD CD CD	SXL	SUM OF ATOM DENSITY*THICKNESS OF SLAB REGIONS FOR ALL MATERIALS AND REGIONS TO THE LEFT OF EACH SLAB REGION. SXL(I,J,K) IS THE VALUE FOR MATERIAL I IN SLAB REGION K, FOR ALL MATERIALS J IN ALL SLAB REGIONS TO THE LEFT OF REGION K. SXL IS FILLED WITH ZEROS IN REGIONS K IN WHICH THERE ARE FEWER THAN MAXHTM MATERIALS TREATED HETEROGENEOULSY	
C		i	-
CR		ATOM DENSITIES TIMES RIGHT SLAB THICKNESSES (TYPE 8)	-
cc c		PRESENT ONLY IF NGEOM.EQ.1 AND MAXHTM.GT.0	-
CL .	(((SXR	(I,J,K), I=1, MAXHTM), J=1, NPRMAT), K=1, NREG)	- ·
C₩ C	МАХНТМ	+NPRMAT*NREG	-
CD CD CD CD CD CD CD CD CD CD C C	SXR	SUM OF ATOM DENSITY*THICKNESS OF SLAB REGIONS FOR ALL MATERIALS AND REGIONS TO THE FIGHT OF EACH SLAB REGION. SXR(I,J,K) IS THE VALUE FOR MATERIAL I IN SLAB REGION K, FOR ALL MATERIALS J IN ALL SLAB REGIONS TO THE RIGHT OF REGION K. SXR IS FILLED WITH ZEROS IN REGIONS K IN WHICH THERE ARE FEWER THAN MAXHTM MATERIALS TREATED HETEROGENEOULSY	

C-REGION OUTER BOUNDARIES (TYPE 9) CR С CC PRESENT ONLY IF MAXHTM.GT.0 С CL (R(I), I=1, NREG)С CW NREG 0 R(I) IS THE OUTER DIMENSION OF REGION J. THE CD R CD OUTER DIMENSIONS ARE MEASURED RELATIVE TO THE CENTER OF REGION 1 FOR CYLINDERS, AND CD RELATIVE TO THE LEFT BOUNDARY OF REGION 1 FOR CD SLABS .CD С C-----CR FISSION SPECTRUM (TYPE 10) С PRESENT IF ICHI.GE.1 CC С (CHINAM(I), I=1, ICHI)CL С CW MULT*ICHI С RÉAL*8 LIBRARY NUCLIDE IDENTIFICATION LABEL CD CHINAM (I) WHOSE FISSION SPECTRUM WILL BE USED FOR CD PROBLEM MATERIAL I. IF ONLY ONE VALUE IS GIVEN, CD THE FISSION SPECTRUM OF THE GIVEN NUCLIDE CD WILL BE USED FOR ALL FISSIONABLE MATERIALS IN CD THE PROBLEM CD C _____ C----FIXED UNRESOLVED RESONANCE ENERGY MESH (TYPE 11) CR С CC ALWAYS PRESENT С (ESF(I), I=1, NESF) $\mathbf{C}\mathbf{L}$ C C₩ NESF С FIXED ENERGY MESH FOR UNRESOLVED RESONANCE CD ESF CALCULATIONS. (ES(I).GT.ES(I+1)) CD

-372-

С C----C--HETEROGENOUS MATERIAL EDITS (TYPE 12) CP С PRESENT ONLY IF MAXHTM.GT.O CC С CL ((LOCHET(I,K), I=1, NPRMAT), K=1, NREG)С C₩ NPRMAT*NREG С LOCHET (I,K) LOCHET (I, K) = 0, IF MATERIAL I IS NOT A RESONANCE-CD MATERIAL OR IS NOT PRESENT IN CD REGION K CD =-1, IF MATERIAL I IN REGION K USES -CD THE HOMOGENEOUS CROSS SECTIONS -CD FOR MATERIAL I CD =L, IF MATERIAL I IN REGION K USES CD THE HETEROGENOUS CROSS SECTIONS CD FOR MATERIAL I IN REGION L CD С IF LOCHET (I, K) . NE.K THEN MATHET (I, K) . EQ. 0 CNNOTE ALSO THAT FOR NGEOM=2, LOCHET(I,2) IS CN EQUAL TO 0 OR -1. MATHET IS NOT DEFINED FOR CN K=2 IN THIS CASE CN С _____ C-C-----_____ FIXED BUCKLING GUESS (TYPE 13) CR С CC. PRESENT ONLY IF IBSQ=-1 С BSQ1 CL С CW 1 С C-----VARIABLE BUCKLING GUESSES (TYPE 14) CR С PRESENT ONLY IF IBSO=0 CC С CL BSQ1, BSQ2, EPS С

-373-

CW 3 С SECOND GUESS FOR BUCKLING CD BS02 CONVERGENCE CRITERION FOR BUCKLING ITERATION CD EPS С C -_____ C----GROUP DEPENDENT BUCKLINGS (TYPE 15) CR С PRESENT ONLY IF IBSU=1 CC С CL (BSQ(I), I=1, NGROUP)С CW NGROUP С BUCKLINGS FOR ULTRA FINE GROUPS CD BSO С COMPOSITION DENSITIES (TYPE 16) CR С PRESENT ONLY IF NCMP.GT.O CC С (COMDEN(I,K), I=1, NPRMAT), K=1, NCMP), (COMTEM(K), K=1, NCMP)CL С (NPRMAT+1) *NCMP C₩ С ATOM DENSITY OF MATERIAL I IN COMPOSITION K CD COMDEN (I,K) COMPOSITION TEMPERATURES IN DEGREES K CD COMTEM(K) С COMTEM IS USED ONLY BY THE INTEGRAL TRANSPORT CN THEORY CALCULATION. ALL MATERIALS IN A REGION CNCONTAINING A PARTICULAR COMPOSITION ARE CN ASSIGNED THE TEMPERATURE OF THAT COMPOSITION. CN NOTE THAT NOT ALL DEFINED COMPOSITIONS NEED CNBE ASSIGNED TO REGIONS IN A GIVEN PROBLEM. CN THE COMPOSITION-REGION ASSIGNMENTS ARE GIVEN CN IN DATA SET GEOM CN С C-C----_____ EXTERNAL SOURCES (TYPE 17) CR С PRESENT ONLY IF ISORS.GT.0 CC

-374-

.

C CL C CW	(SOURCE(I),I=	
C CD C	SOURCE	EXTERNAL SOURCES FOR EACH ULTRA FINE GROUP -
C		
CR	BROAD ~G	ROUP ENERGIES (TYPE 18)
cc	ALWAYS	PRESENT
CL	(EBG(I), J=1, N)	BG1) -
C W	NBG1	
CD CD CD C	EBG(I)	UPPER ENERGY OF BROAD GROUP I. EBG(NBG1) IS THE- ENERGY AT THE INTERFACE BETWEEN THE THERMAL GROUP AND LOWEST EPI-THERMAL BROAD GROUP -
C		
C R C	THERMAL	GROUP CROSS SECTIONS (TYPE 19) -
CC C	ALWAYS	PRESENT -
CL CL CL CL CL CL	(SIGCAP (I), I= 1 (GNU (I), I=1, N 2 (SIGNA (I), I=1 3 (SIGND (I), I=1 4 (SIGHE3 (I), I=	1, NPRMAT), (SIGFIS(I), I=1, NPRMAT), PRMAT), (SIGTOT(I), I=1, NPRMAT), , NPRMAT), (SIGNP(I), I=1, NPRMAT), , NPRMAT), (SIGNH3(I), I=1, NPRMAT), 1, NPRMAT), RECVEL)
CM C	9*NPRMAT+1	- - -
CD CD CD CD CD CD CD CD CD	SIGCAP SIGFIS GNU SIGTOT SIGNA SIGNP SIGND SIGNH3 SIGNH3	THERMAL GROUP CAPTURE CROSS SECTIONS-THERMAL GROUP FISSION CROSS SECTIONS-THERMAL GROUP NUMBER OF NEUTRONS PER FISSION-THERMAL GROUP TOTAL CROSS SECTIONS-THERMAL GROUP N-ALPHA CROSS SECTIONS-THERMAL GROUP N-P CROSS SECTIONS-THERMAL GROUP N-D CROSS SECTIONS-THERMAL GROUP N-H3 CROSS SECTIONS-THERMAL GROUP N-H3 CROSS SECTIONS-THERMAL GROUP N-H3 CROSS SECTIONS-THERMAL GROUP N-H3 CROSS SECTIONS-THERMAL GROUP N-H43 CROSS SECTIONS-THERMAL GROUP N-H44 CROSS SECTIONS

· . .

-

.

-375-

CN CN CN C		ANY QUANTITY HAVING THE VALUE -1.0E-20 WILL BE - ASSIGNED THE CORRESPONDING LAST EPI-THERMAL - GROUP VALUE FOR THE THERMAL GROUP -
	•	
C CR	HOMOGENEOUS	S INFITITELY DILUTE SPECIFICATIONS (TYPE 20) -
cc	ALMAYS PRES	ENT -
C CL	(INFINT(I), I=1, NE	PRMAT)
C CW	NPRMAT	
C CD CD CD CD CD CD CD CD CD C	INFINT	IF 0, MATERIAL I IS NOT INFITELY DILUTE. IF 1, MATERIAL I IS ASSUMED TO BE INFINITELY DILUTE FOR THE HOMOGENEOUS RESOLVED RESONANCE CALCULATION. MATERIALS HAVING INFINT=1 ARE NOT INVOLVED IN THE OVERLAP CALCULATION OF ANY OTHER RESONANCES. THEIR RESONANCE INTEGRALS WILL BE SET TO THE INFINITELY DILUTE LIMIT OF PI/(2.*BETA)
C	 FDTT OPTO	
C C CC	ALWAYS PRES	SENT -
C CL CL	KUNRES, KATNUA, KRI 1KSPEC, KBGRES, KBGF	ESIN, KSIGMA, KLORNZ, KUFGCS, KUFGMP, KUFGPL, KRESED, - RR
CW CW	12	
C D C D	KUNRES	IF 0, DO NOT EDIT DATA SET UNRES. OTHERWISE - EDIT THE DATA SET -
C D C D	KATNUA	IF 0, DO NOT EDIT DATA SET ATNUAT. OTHERWISE - EDIT THE DATA SET -
C D C D	KRFSIN	IF 0, DO NOT EDIT DATA SET RESINT. OTHERWISE - EDIT THE DATA SET -
C D C D	KSIGMA	IF 0, DO NOT EDIT DATA SET SIGMAP. OTHERWISE - EDIT THE DATA SET -
CD		
CD	KLORNZ	EDIT THE DATA SET

*.**

-376-

CD		SECTIONS. OTHERWISE DO NOT EDIT THESE DATA -
CD	KUEGHP	
CD		HODEFRIING FRARMETERS. CIMERWISE BO NOT
		TE A DO NOM EDIM NIMDA EINE CDOUD MICDOCCODIC -
0	KUTGPL	IF U, DU NUI EDII ULIRA FINE GROUP MICROSCOPIC -
CD		PL SCATTERING CROSS SECTIONS, PO AND PI -
CD		ELASTIC TRANSFER MATRICES. OTHERWISE EDIT -
CD		THESE DATA -
CD	KRESED	IF O, RESONANCE CROSS SECTIONS ARE ADDED TO THE-
CD .		BROAD GROUP OUTPUT DATA SET. IF 1, RESONANCE -
CD		CROSS SECTIONS ARE OMITTED FROM THE BROAD GROUP-
CD		CROSS SECTIONS. IF 2, RESONANCE CAPTURE AND -
CD		FISSION CROSS SECTIONS ARE OMITTED FROM THE -
CD		BROAD GROUP CROSS SECTIONS -
CD	KSPEC	IF 0, EDIT ULTRA-FINE-GROUP FLUX, CURRENT AND -
CD		SLOWING DOWN DENSITIES. OTHERWISE DO NOT EDIT -
CD		THESE DATA -
CD	KBGRES	IF 0, DC NOT EDIT BROAD GROUP RESONANCE CROSS -
CD		SECTIONS. OTHERWISE EDIT THESE DATA
CD	KBGRR	IF O, EDIT BROAD GROUP REACTION RATES
CD		OTHERWISE DO NOT EDIT THESE DATA
С		-
C		

C		
CR	FOIL SPECIE	FICATIONS (TYPE 22) -
С		-
CC	PRESENT IF	NFOILS.GT.0 -
С		-
CL	(FOILNM(K), K=1, NH	FOILS), (FOILMT (I), I=1, NFLMAT), -
CL	1 (FOILDX (K), K=1, N)	FOILS), ((FOILDN (I, K), I=1, NFLMAT), K=1, NFOILS), -
CL	2 (FOILTM (K), K=1, NH	FOILS) -
С		- · · · · · · · · · · · · · · · · · · ·
Cu	(NFLMAT+2) *NFOILS	S+MULT* (NFOILS+NFLMAT) -
С	•	-
CD	FOILNM	REAL*8 LABELS OF FOILS -
CD	FOILMT	REAL*8 LABELS OF LIBRARY NUCLIDES USED IN FOILS-
CD	FOILDX	FOIL THICKNESSES IN CM -
CD	FOILDN (I,K)	ATOMIC DENSITY OF MATERIAL I IN FOIL K -
CD		(ATOMS/CC*1.E-24) -
CD	FOILTM	FOIL TEMPERATURES IN DEGREES K -
с		-
C		
-		·
C		
CR	NON-CELL AV	VERAGED CROSS SECTION SPECIFICATIONS (TYPE 23) -
С		-
CC	PRESENT IF	NGEOM.GT. 0 -

-377-

C CL	((ALEDIT(N,I), N=1, NREG2), I=1, NPRMAT)			
C CW C	NREG2*NPRMAT*M	ULT		
C D C D	ALEDIT (N,I)	REAL*8 ALIAS NAME FOR PROBLEM MATERIAL I TO BE EDITED FOR REGION N		
C D C	NREG2	NREG+1		
C===				

CEOF

2
APPENDIX D. MC²-2 Binary Files. RESDAT

C****	*****	* * * * * * * * * * * * * * * * * * * *
C C		PREPARED 3/07/75 AT ANL -
CF CE		RESDAT - RESOLVED RESONANCE DATA -
C CN CN C		THIS DATA SET IS WRITTEN BY MC**2-II AREA 6 - (CSC005) -
C****	*****	* * * * * * * * * * * * * * * * * * * *
CD CD CD	NRES	TOTAL NUMBER OF RESOLVED RESONANCES IN THE PROBLEM. THE RESONANCES ARE ORDERED ACCORDING TO DECREASING ENERGY
C CR C		RESONANCE ENERGY - TOTAL WIDTH RATIOS (TYPE 1) -
cc		ALWAYS PRESENT
CL	(X(I),	I=1, NRES)
CW C	NRES	
CD CD CD CD C C C	X .	2.0*RESONANCE ENERGY/GAMMA, WHERE GAMMA IS THE - TOTAL LINE WIDTH FOR SINGLE LEVEL RESONANCES - OR THE S-MATRIX TOTAL LINE WIDTH FOR - ADLER-ADLER MULTILEVEL RESONANCES -
C CR		TOTAL TO NEUTRON LINE WIDTH RATIOS (TYPE 2)
C CC		ALWAYS PRESENT
CL	(GAMTN	(I), I=1, NRES)
C C W	NRES	
C CD CD CD CD	GAMTN	IF A EQUALS ZERO, WHERE A IS THE FACTOR WHICH - MULTIPLIES CHI TO OBTAIN THE INTERFERENCE - SCATTERING (SEE RECORD TYPE 4), THEN GAMTN - EQUALS ZERO -
CD CD		FOR SINGLE LEVEL RESONANCES, -

-379-

CD GAMTN = GAMT/GAMN, WHERE GAMT AND GAMN AND CD	RE - WIDTHS -
	WIDTHS -
CD RESPECTIVELY THE TOTAL AND NEUTRON LINE	
CD	-
CD FOR ADLER-ADLER MULTILEVEL RESONANCES,	-
CD $GAMTN = -0.5*(HT-HC-HF)/(A*(GT-GC-GF)),$	THERE -
CD HT, HC, AND HF ARE ASYMMETRIC ADLER-ADLER	g –
CD PARAMETERS FOR TOTAL, CAPTURE, AND FISSIC	JN
CD AND GT, GC, AND GF ARE THE CORRESPONDING	-
CD SYMMETRIC PARAMETERS	-
C	-
C	

C									
CR	•	RESOLVED	RESONANCE P	EAK CROSS	SECTION	Т (ТҮРЕ)	3)		-
С									-
СС		ALWAYS PH	RESENT						-
С									-
CL	(SIGO(I),I=1,NE	RES)						-
С		·							-
Cü	NRES								-
С									-
CD	SIGO		RESOLVED	RESONANCE	CROSS	SECTION	ΑT	RESONANCE	-
C ·									-
C					,-				

C			
C R		INTERFERENCE SCATTERING FACTOR (TYPE 4)	-
cc c		ALWAYS PRESENT	-
CL C	(A(I),	, I = 1, NRES)	-
C ฬ C	NRES		-
CD CD C	A 	FACTOR TO MUTLIPLY CHI TO OBTAIN INTERFERENCE SCATTERING	-
CD CD C	A 	SCATTERING	-

CEOF

-380-

APPENDIX D. MC²-2 Binary Files. RESINT

C***** C	****	* * * * * * * * * * * * * *	******	* * * * * * * * * * * * * * * * * * * *	**************************************
C C			PREPARI	ED 3/05/75 AT ANL	-
CF		RESINT			· . –
CE C		RESONANCE IN	T ER FAC I	E DATA SET	-
CN			THIS IN	NTERFACE DATA SET IN	I WRITTEN -
CN			BY MC**	*2-II AREA 6 (CSC005	
C ·					
C * * * * *	*****	* * * * * * * * * * * * * *	*****	* * * * * * * * * * * * * * * * * * * *	< * * * * * * * * * * * * * * * * * * *
CD	NREG		NUMBER	OF REGIONS	
CD			NREG = 1	FOR HOMOGENEOUS PRO	DBLEMS
CD			NREG=2	FOR CYLINDRICAL PIN	I FROBLEMS AND IF
CD	•			MAXHTM.GT.O	
CD			NREG=1	+ THE NUMBER OF SLA	B REGIONS FOR
CD				SLAB PROBLEMS AND I	F MAXHTM.GT.O
CD	NRES	,	NUMBER	OF RESOLVED RESONAN	ICES
C				·	
CS CS		FILE STRUCTU	RE		-
CS	~	RECOPD TYPE			PRESENT IF -
CS		=======================================	=======		
CS		SPECIFICATIO	NS		ALWAYS -
CS		RESOLVED RES	ONANCE	GROUPS	ALVAYS -
CS		RESOLVED RES	ONANCE	MATERIALS	ALWAYS -
CS		RESOLVED RES	ONANCE	ENERGIES	ALWAYS -
CS	* * * * * *	** (REPEAT N	REG TIM	1ES)	-
CS	*	RESOLVED CAP	TURE IN	ITEGRALS	ALWAYS -
CS	*	RESOLVED FIS	SION IN	ITEGRALS	ALWAYS -
CS	*	TOTAL RESONA	NCE INT	TEGRALS	ALWAYS -
CS	*****	* *			-
CS	*****	** (REPEAT N	REG TIM	1ES)	-
CS	*	UNRESOLVED O	VERLAP	CORRECTED	UNRESOLVED DATA -
CS	*	CAPTURE INTE	GRALS		IS PRESENT -
CS	*	UNPESOLVED O	VERLAP	CORRECTED	UNRESOLVED DATA -
CS	*	FISSION INTE	GRALS		IS PRESENT -
CS	*	UNRESOLVED O	VERLAP	CORRECTED	UNRESOLVED DATA -
CS	*	TOTAL INTEGR	ALS		IS PRESENT -
CS	*****	·**		· · · ·	-
C					-
C					
C					
CR		SPECIFICATIO	NS (TYP	PE 1)	

-381-

C CC	
c	ALTAIS FRESENT
ÇL C	NRES, NREG, NRESMT
C W	3
C D C D	NRESMT NUMBER OF RESOLVED RESONANCE PROBLEM MATERIALS -
C	
C· CR	RESOLVED RESONANCE GROUPS (TYPE 2)
cc	ALWAYS PRESENT
CL	(NGPRES(I), I=1, NRES)
C भ C	NRES
CD C	NGPRES NUMBER OF RESOLVED RESONANCE GROUPS
C	
۰,	
C CR	RESOLVED RESONANCE MATERIALS (TYPE 3)
cc c	ALWAYS PRESENT
CL C	(MATRES(I), I=1, NRES)
CV C	NRES -
CD CD C	MATRES(I) PROBLEM MATERIAL NUMBER CORRESPONDING TO - RESONANCE I -
C	
C CR	RESOLVED RESONANCE ENERGIES (TYPE 4) -
CC	- ALWAYS PRESENT -
CL	(EN (I), I=1, NRES)
CW C	NRES

-382-

CD C C C	E N	RESOLVED RESONANCE ENERGIES
C		
CR		RESOLVED CAPTURE INTEGRALS (TYPE 5) -
		ALWAYS PRESENT FOR HOMOGENEOUS MIXTURE. ALSO PRESENT - FOR EACH HETEROGENEOUS REGION -
CL	(ÇJ (I)	, I = 1, NRES) -
C CW	NRES	
C CD C	ÇJ	RESOLVED CAPTURE INTEGRALS -
C		
C		RESOLVED FISSION INTEGRALS (TYPE 6)
		ALWAYS PRESENT FOR HOMOGENEOUS MIXTURE. ALSO PRESENT - FOR EACH HETEROGENEOUS REGION -
CL	(FJ (I)	, I = 1, NRES) -
C CW C	NRES	-
CD C	FJ	RESOLVED FISSION INTEGRALS -
C		
C CR C		TOTAL RESONANCE INTEGRALS (TYPE 7) -
cc cc	•	ALWAYS PRESENT FOR HOMOGENEOUS MIXTURE. ALSO PRESENT - FOR EACH HETEROGENEOUS REGION -
C CL	(TJ (T)	. I=1, NRES)
C CW	NRES	
C C D C C	TJ [:]	TOTAL RESONANCE INTEGRALS

-383-

C-----UNRESOLVED OVERLAP CORRECTED CAPTURE INTEGRALS (TYPE 8) CR С CC PRESENT ONLY IF PROBLEM CONTAINS UNRESOLVED DATA С CL (CJFL(I), I=1, NRES)С CW NRES С CJFL RESOLVED CAPTURE INTEGRALS MULTIPLIED BY ĊĎ APPROPRIATE UNRESOLVED OVERLAP FACTORS CD С $C - \cdot$ _____ C----UNRESOLVED OVERLAP CORRECTED FISSION INTEGRALS (TYPE 9) CR С PRESENT ONLY IF PROBLEM CONTAINS UNRESOLVED DATA CC С (FJFL(I), I=1, NRES)CLС CW NRES С RESOLVED FISSION INTEGRALS MULTIPLIED BY CD FJFL APPROPRIATE UNRESOLVED OVERLAP FACTORS CD C C------_____ CR UNRESOLVED OVERLAP CORRECTED TOTAL INTEGRALS (TYPE 10) C CC PRESENT ONLY IF PROBLEM CONTAINS UNRESOLVED DATA С (TJFL(I), I=1, NRES)CL С C₩ NRES С TOTAL RESONANCE INTEGRALS MULTIPLIED BY CD JFL APPROPRIATE UNRESOLVED OVERLAF FACTORS CD С C----

CEOF

-384-

APPENDIX D. MC²-2 Binary Files. SCR001

~			e and a star ato	مال مان حال مان
C****	*****	* * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *
С				-
С			PREPARED 3/05/75 AT ANL	-
С			,	-
CF		SCR001	,	· · · · · · · · · · · · · · · · · · ·
CE		CSC005 SCRA	TCH DATA SET SCR001	-
C C		000,00 00m		-
CN				עם איז היה דיסה.
CN	,		MOTTO TT ADDA C (CCCOOC)	WATTIEN DI
CN			MC + 2 - 11 AREA, O (CSCUUS)	-
С				-
C****	*****	****	* * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *
			۶ .	
CD	NOLINT	I	NUMBER OF ULTRA FINE GROU	PS HAVING
CD			LORENTZIAN SHAPE RESONANC	E INTEGRALS FOR
CD			EACH RESONACE SO TREATED	
	NOTRES		NUMBER OF RESONANCES CIVE	N & LORENTZIAN SHAPE
CD	100 A 65			NOIDES IS NOTEEN
CD CD			IN DIEN COR CODOO	• NOTVER TO MUTITEN
CD			IN DATA SET SCRUUZ	
_				
C				
CS		FILE STRUCT	TURE	·
CS				·
CS		RECORD TYPE	3	PRESENT TF -
CS		=======================================	=======================================	=======================================
CS	*****	** (REPEAT	NOLRES TIMES)	-
20	*	ULTRA FINE	GROUP NUMBERS	ALWAYS -
C.5	*	TODDAR TIND	DECONANCE INTECOME	ALWAIS -
05	ىلەر مادرىلەر بادرىلەر بادرىلەر	TOWPN ATEN	RESUNANCE INTEGRADS	ELKEIS -
CS	****	* *		-
С				-
C				
-				
C			$\frac{1}{2}$	
CR		OLTRA FINE	GROUP NUMBERS (TYPE I)	-
С	٠			
CC		ALWAYS PRES	SENT	-
С				-
CL	NOLINT	.N1.N2		-
C C				-
CN	Э	,		· · · · · · · · · · · · · · · · · · ·
	J			_
C an				
CD	NJ		HIGHEST ENERGY ULTRA FINE	GROUP HAVING -
CD			LORENTZIAN SHAPE RESONANC	E INTEGRALS -
CD	N2		LOWEST ENERGY ULTRA FINE (GROUP HAVING -
CD			LORENTZIAN SHAPE RESONANC	E INTEGRALS -
Ċ				· · · · · · · · · · · · · · · · · · ·
C				
-				

.

-385-

C--LORENTZIAN RESONANCE INTEGRALS (TYPE 2) CR С СС ALWAYS PRESENT С CL CJINTL(I), I=1, NOLINT C CΨ NOLINT С CD CJINTL LORENTZIAN SHAPE RESONANCE INTEGRALS С , C--

CEOF

-386-

C****	*****	< * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * *	******	****	** ** **	******	*****	*****	* *
C C			PREPARED	3/07/7	5 AT	ANL				-
		CCD001		,						_
CF CE		ULTRA FINE (GROUPSCA	TTERING	CROS	SS SEC	TIONS		,	-
CN CN CN			BOTH COND SCROO1 AD (CSCO11)	FIGURAT RE WRIT	IONS TEN E	OF THI BY MC*	IS SCRATO *2-II ARI	CH DAT Ea 10	A SET	-
C		، بېله بېله بېله بېله بېله بېله بېله بېله		*****	****	***	*****	*****	****	**
C****	* * * * * * *	• ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	• • • • • • • • • • • • • • •	• • • • • • • • •	****	• • • • • • •	• • • • • • • • • • • •	•••••	<u>.</u>	
CD CD	I1 . I2		FIRST ULT	FRA FIN RA FINE	E GRO	DUF RE. JP REAL	AD FOR CU	JRRENT RRENT	PASS PASS	
CD	NPASS		NUMBER OF	TYPE	1 REC	CORDS 1	PRESENT 1 THERE V	IN THE	FILE.	
CD			ADDITION	AL RECO	RDRE	EAD FO	R THE REN	1A IN IN	G	
CD /			NULTRA-NI	PASS*NU	FGRD	ULTRA	FINE GRO	OUPS		
CD	NPRMSC		NUMBER OI	F MATER	IALS	IN TH	E PROBLEN	MIXT	URE	
CD			WHICH HAV	E TABU	LATE	SCAT	FERING CE	ROSS S	ECTIONS	
CD			IN THE AN	REA 10	(CSCC)11) EI	NERGY RAN	IGE		
CD	NUFGRD)	NUMBER OI	F ULTRA	FINE	GROU	PS READ I	PER PA	SS.	
CD			NUFGRD = I	2-I1+1						
			•							
~	.118		ب مان مان مان مان مان مان مان م	ىلە ملە بەلە ماد بەلە بالە مە	- 					
С	***	* * * * * * * * * * * * * * * *	* * * * * * * * * *	• • • • • • • • •	*				•	
C	*			T	* *					
C=====	* ===	FIRST CON	FIGURATION	N	*					
C	~ ***	****	****	*****	*					
C										
CS CS	·	FILE STRUCT	JRE				λ,			-
CS		RECORD TYPE		,			PRESENT	IF		-
CS		=======================================	=======================================	=======	=====	===	========	=====	========	= -
CS	****	** (REPEAT I	NPRMSC TI	MES)						-
CS	* ·						THE CURE	RENT M	ATERIAL	-
CS	*	ULTRA FINE (GROUP SCAT	TTERING			HAS SCAT	CTER IN	G CROSS	
CS	*	CPOSS SECTIO	DNS				SECTIONS	5 IN T	HE AREA	-
CS	*						10 (CSC()11) E	NERGY	-
CS	*		:				RANGE			-
CS	*****	***								-
C										
C										

-387-

_____ C----ULTRA FINE GROUP SCATTERING CROSS SECTIONS (TYPE 1) CR С CC ALWAYS PRESENT С (SIGSCT(I), I=NITUFG.NGROUP) CL С CW NGROUP-NITUFG+1 С ULTRA FINE GROUP MICROSCOPIC SCATTERING CROSS CD SIGSCT(I) CD SECTION FOR ULTRA FINE GROUP I ĊĎ NITUFG HIGHEST ENERGY ULTRA FINE GROUP ON THE AREA 10 -(CSC011) ENERGY RANGE CD CD NGROUP LOWEST ENERGY ULTRA FINE GROUP IN THE PROBLEM С _____ C----CEOF **** С С C======= * SECOND CONFIGURATION С ****** С C-------CS. FILE STRUCTURE CS CS RECORD TYPE PRESENT IF ______ CS ******** (REPEAT NPASS TIMES) CS ULTRA FINE GROUP SCATTERING CROSS ALWAYS PRESENT CS * CS SECTIONS * ****** CS ULTRA FINE GROUP SCATTERING CROSS CS NULTRA.GT. CS NUFGRD*NPASS SECTIONS С _____ C-----CR ULTRA FINE GROUP SCATTERING CROSS SECTIONS (TYPE 1) С ALVAYS PRESENT CC С CL ((SIGSCT(M,I), M=1, NPRMAT), I=I1, I2) С CW NPRMAT*NUFGRD

-388-

С			-
CD	SIGSCT (M,I)	ULTRA FINE GROUP SCATTERING CROSS SECTION FOR	-
CD		MATERIAL M AND ULTRA FINE GROUP I	-
CD	NPFMAT	NUMBER OF MATERIALS IN THE PROBLEM	-
С			-
CN		NUFGRD ULTRA FINE GROUPS ARE READ FOR EACH	. –
CN		PASS WHERE NUFGRD=12-11+1. THERE WILL BE	-
CN		ONE ADDITIONAL RECORD AFTER THE NPASS RECORDS	-
CN		WHICH CONTAINS NULTRA-NPASS*NUFGRD ULTRA FINE	-
CN		GROUPS IF NULTRA IS NOT AN INTEGRAL MULTIPLE OF	-
CN		NUFGRD. NULTRA IS THE NUMBER OF ULTRA FINE	-
CN		GROUPS IN THE AREA 10 (CSCO11) ENERGY RANGE	-
С			-
C			

CEOF

-389-

APPENDIX D. MC^2-2 Binary Files. SCR002 С C. PREPARED 3/05/75 AT ANL С CF SCR002 CE CSC005 SCRATCH DATA SET SCR002 С CN THIS SCRATCH DATA SET IS WRITTEN BY CN MC**2-II AREA 6 (CSC005) С CD NOLRES NUMBER OF RESONANCES GIVEN A LORENTZIAN SHAPE 1 FOR HOMOGENEOUS PROBLEMS, 2 FOR CYLINDRICAL CD NREG1 Ç₽ PROBLEMS, 1+THE NUMBER OF SLAB REGIONS FOR SLAB GEOMETRY PROBLEMS CD C--______ CS FILE STRUCTURE CS CS PRESENT IF RECORD TYPE CS CS ******* (REPEAT NREG1 TIMES) CS RESONANCE SPECIFICATIONS ALWAYS * CS RESONANCE NUMBERS ALWAYS * CS ***** С C -C---CR **RESONANCE SPECIFICATIONS (TYPE 1)** С CC ALWAYS PRESENT С CLNOLRES, N1MIN, N2MAX С CW 3 . C CD N 1MIN HIGHEST ENERGY ULTRA FINE GROUP HAVING CD LORENTZIAN SHAPE RESONANCE INTEGRALS CD N2MAX LOWEST ENERGY ULTRA FINE GROUP HAVING CD LORENTZIAN SHAPE RESONANCE INTEGRALS С

-390-

CR	RESONANCE NUMBERS (TYPE 2)	-
cc c	ALWAYS PRESENT	¢
CL C	LRESNO(I), I=1, NOLRES	-
CW C	NOLRES	-
CD CD C	LRESNO RESOLVED RESONANCE HAVING A LORENTZIAN	NUMBERS FOR RESONANCES - N SHAPE TREATMENT - -

CEOF

· . . .

•

-391-

C*****	*****	* * * * * * * * * * * * * * * * *	****	****	********
C C	- · · ·	PREPARED 3/10/	75 AT ANL		-
CF	SCR002	(,		-
CE	ULTRA FINE	GROUP FISSION C	ROSS SECTIO	N S	· –
CN CN		BOTH CONFIGURA SCR002 ARE WRI	TIONS OF TH TTEN BY MC*	IS SCRATCH D *2-II AREA 1	ATA SET -
CN C	-	(CSC011)	• •	X	-
C****	* * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *	****	****	** * * * * * * * * *
	н. — — — — — — — — — — — — — — — — — — —	. /			
CD	I1	FIRST ULTRA FI	NE GROUP RE	AD FOR CURRE	NT PASS
CD	I2	LAST ULTRA FIN	E GROUP REA	D FOR CURREN	T PASS
CD	NPASS	NUMBER OF TYPE	I RECORDS	PRESENT IN T	HE FILE.
CD		ADDITIONAL REC	OF GED FRID FO	R THERE WILL R THE PEMAIN	
CD		NULTRA-NPASS*N	UFGRD ULTRA	FINE GROUPS	ING
CD	NPRMFS	NUMBER OF MATE	RIALS IN TH	E PROBLEM MI	XTURE
CD		WHICH HAVE TAB	ULATED FISS	ION CROSS SE	CTIONS
CD		IN THE AREA 10	(CSC011) E	NERGY RANGE	
CD	NUFGRD	NUMBER OF ULTR	A FINE GROU	PS READ PER	PASS.
CD		NUFGRD=12-11+1			
				: 1	
.C	****	* * * * * * * * * * * * * * * * * * * *	**		
C	*		*		
C====	====* FIRST CO	NFIGURATION	*===========		=============
С	*.		*	•	
С	****	* * * * * * * * * * * * * * * * * * *	* *		
		· .			
C					
CS	FTLE STRUC	 TURE			
CS					-
CS	RECORD TYP	Е		PRESENT IF	-
CS	==========		=======	==============	
CS	******* (REPEAT	NPRMFS TIMES)			-
CS .	*			THE CURRENT	MATERIAL -
CS	* ULTRA FINE	GROUP FISSION		HAS FISSION	CROSS -
CS	* CROSS SECT	LONS		SECTIONS IN	THE AREA -
CS CS	≁ . ↓			TO (CSCOTT)	ENERGY -
CS CS	** <u>*</u> *****			RANGE	- -
C S	"""" " " " " " " " "			•	-
C					
~			t		

-392-

C-----CR ULTRA FINE GROUP FISSION CROSS SECTIONS (TYPE 1) С CC ALWAYS PRESENT С CL (SIGFIS(I), I=NITUFG, NGROUP) C CW NGROUP-NITUFG+1 С CD SIGFIS(I) ULTRA FINE GROUP MICROSCOPIC FISSION CROSS CD SECTION FOR ULTRA FINE GROUP I HIGHEST ENERGY ULTRA FINE GROUP ON THE AREA 10 CD NITUFG CD (CSC011) ENERGY RANGE CD NGROUP LOWEST ENERGY ULTRA FINE GROUP IN THE PROBLEM C C-CEOF С С SECOND CONFIGURATION C=====* С **** С C-FILE STRUCTURE CS CS PRESENT IF CS RECORD TYPE CS _________________________ ******* (REPEAT NPASS TIMES) CS ULTRA FINE GROUP FISSION CROSS ALWAYS PRESENT CS SECTIONS CS ****** CS ULTRA FINE GROUP FISSION CROSS CS NULTRA.GT. NUFGRD*NPASS CS SECTIONS С C-------C----ULTRA FINE GROUP FISSION CROSS SECTIONS (TYPE 1) CR С ALWAYS PRESENT CC С ((SIGFIS(M,I), M=1, NPRMAT), I=I1, I2)CL С CV NPRMAT*NUFGRD

С		- · · · · · · · · · · · · · · · · · · ·
CD	SIGFIS(M,I)	ULTRA FINE GROUP FISSION CROSS SECTION FOR -
CD		MATERIAL M AND ULTRA FINE GROUP I -
CÐ	NPRMAT	NUMBER OF MATERIALS IN THE PROBLEM -
С		· · · · · · · · · · · · · · · · · · ·
CN		NUFGRD ULTRA FINE GROUPS ARE READ FOR EACH PASS-
CN		WHERE NUFGRD=12-11+1. THERE WILL BE -
CN		ONE ADDITIONAL RECORD AFTER THE NPASS RECORDS -
CN	:	WHICH CONTAINS NULTRA-NPASS*NUFGRD ULTRA FINE -
CN		GROUPS IF NULTRA IS NOT AN INTEGRAL MULTIPLE OF-
CN		NUFGRD. NULTRA IS THE NUMBER OF ULTRA FINE -
CN		GROUPS IN THE AREA 10 (CSCUII) ENERGY RANGE -
С		• • •
C		

CEOF

-394-

APPENDIX D. MC²-2 Binary Files. SCR003

C****	*****	* * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *	* * *
C C		ראסגרססס אין	r.	
c		PREPARED SYOSY /S AT AN.	L	-
CF	9	SCR003	. ^	-
CE	(CSC004 SCRATCH DATA SET SCR003		-
C				-
CN		THIS SCRATCH DATA SET . MC**2-II AREA 5 (CSCOO)	IS WRITTEN BI	-
		MATERIALS ARE PRESENT	4) IT UNRESOLVED	-
C		······································		-
C****	******	< * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *	** *
			· · · ·	
CD	TFT	FISSILE MATERIAL INDEX	·	
CD		IFI=0 FOR NON-FISSILE	MATERIAL	
CD		IFI=1 FOR FISSILE MATE	RIAL	
CD	JL	NUMBER OF SPIN STATES	FOR CURRENT ISOTOPE	
CD	NESF	NUMBER OF FIXED ENERGY	MESH POINTS	
CĎ	NISO	NUMBER OF ISOTOPES		
CD	NREG	NUMBER OF REGIONS		• •
CD		NREG=1 FOR HOMOGENEOUS	PROBLEMS	
CD		NREG=2 FOR CYLINDRICAL	GEOMETRY	
CD		NREG=1 + THE NUMBER OF	SLAB REGIONS FOR	
CD		SLAB GEOMETRY	• •	
C = = = =				
C				
CS	F	TILE STRUCTURE		
CS CS	ㅋ	ILE STRUCTURE		
CS CS CS	न	TILE STRUCTURE	PRESENT IF	 - -
CS CS CS CS	Ţ	TILE STRUCTURE RECORD TYPE	PRESENT IF	 - - - ===
CS CS CS CS CS	******	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF	 - - - - - -
CS CS CS CS CS CS	******* *	TILE STRUCTURE RECORD TYPE ***** (REPEAT FOR NPRMAT MATERIALS) MATERIAL SPECIFICATIONS	PRESENT IF = ==================================	
CS CS CS CS CS CS CS	****** * * ****	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF	
CS CS CS CS CS CS CS CS	****** * * **** * *	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF = ==================================	
CS CS CS CS CS CS CS CS CS	****** * * **** * * * *	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF = ==================================	
CS CS CS CS CS CS CS CS CS CS CS	****** * * **** * * * * * *	TILE STRUCTURE RECORD TYPE ***** (REPEAT FOR NPRMAT MATERIALS) MATERIAL SPECIFICATIONS ***** (REPEAT FOR NISO ISOTOPES) SPIN STATE DATA LEVEL SPACING ***** (REPEAT FOR 1+NO. OF REGIONS	PRESENT IF = = = = = = = = = = = = = = = = = = =	
CS CS CS CS CS CS CS CS CS CS CS	******* * * **** * * * * * * * * * * *	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF = ==================================	
CS CS CS CS CS CS CS CS CS CS CS CS CS C	****** * * **** * * * * * * * * * * * *	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF = ==================================	
CS CS CS CS CS CS CS CS CS CS CS CS CS C	***** * * * * * * * * * * * * * * * *	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF ALWAYS ALWAYS ALWAYS ALWAYS ALWAYS	
CS CS CS CS CS CS CS CS CS CS CS CS CS C	****** * * * * * * * * * * * * * * * *	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF ALWAYS ALWAYS ALWAYS ALWAYS ALWAYS ALWAYS ALWAYS ALWAYS	
CS CS CS CS CS CS CS CS CS CS CS CS CS C	****** * * * * * * * * * * * * * * * *	TILE STRUCTURE RECORD TYPE ***** (REPEAT FOR NPRMAT MATERIALS) MATERIAL SPECIFICATIONS ***** (REPEAT FOR NISO ISOTOPES) SPIN STATE DATA LEVEL SPACING ***** (REPEAT FOR 1+NO. OF REGIONS IN WHICH MATERIAL IS TREATED HETEROGENEOUSLY) CAPTURE J INTEGRAL TOTAL J INTEGRAL FISSION J INTEGRAL	PRESENT IF ALWAYS ALWAYS ALWAYS ALWAYS ALWAYS ALWAYS IFI.EQ.1	
	****** * * * * * * * * * * * *	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF = ==================================	
CS CS CS CS CS CS CS CS CS CS CS CS CS C	****** * * * * * * * * * * * * * * * *	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF ALWAYS ALWAYS ALWAYS ALWAYS ALWAYS IFI.EQ.1	
CS CS CS CS CS CS CS CS CS CS CS CS CS C	****** * **** * * *** * * * * * *	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF ALWAYS ALWAYS ALWAYS ALWAYS ALWAYS IFI.EQ.1	
CS CS CS CS CS CS CS CS CS CS CS CS CS C	****** * **** * * * * * * * * * * * * *	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF ALWAYS ALWAYS ALWAYS ALWAYS ALWAYS IFI.EQ.1	
CS CS CS CS CS CS CS CS CS CS CS CS CS C	****** * **** * * * * * * * * * * * * *	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF = ===================================	
CS CS CS CS CS CS CS CS CS CS CS CS CS C	****** * **** * * * * *	TILE STRUCTURE RECORD TYPE ====================================	PRESENT IF ALWAYS ALWAYS ALWAYS ALWAYS ALWAYS IFI.EQ.1	

-395-

•

с		-					
сс		PRESENT IF NUNRES.EQ.1 -					
C		- · · · ·					
CL	NISU,III						
CW	2						
C	L	-					
C							
C							
CB		SPIN STATES (TYPE 2) -					
C		-					
CC		PRESENT IF NUNRES.EQ.1 -					
C	**						
C L	JL	-					
CW	1	_					
C		· · · · · · · -					
C							
		· · ·					
C							
CR		LEVEL SPACING (TYPE 3).					
C							
сс		PRESENT IF NUNRES.EQ.1 -					
C CT		$T_{1} = 1 \text{ NESE} = 1 \text{ II} =$					
C	([0 (1						
CW	NESF*J	1L –					
C		-					
CD	D	AVERAGE LEVEL SPACING -					
C							
0	·						
C							
CR		CAPTURE J INTEGRAL (TIPE 4)					
cc		ALMAYS PRESENT FOR HOMOGENEOUS MIXTURE.					
cc		ALSO PRESENT FOR EACH HETEROGENEOUS REGION IN WHICH -					
CC		CURRENT MATERIAL IS TREATED HETEROGENEOUSLY -					
C							
CL C	((CD E8	$DE(T^{*}O) = T = (T^{*}OE) = (T^{*}OE)$					
C₩	NESF*C	JL –					
С							
CD	CJESF	UNRESOLVED RESONANCE CAPTURE INTEGRAL -					
C							
L							

C-_____ CR TOTAL J INTEGRAL (TYPE 5) С CC ALWAYS PRESENT FOR HOMOGENEOUS MIXTURE. CC ALSO PRESENT FOR EACH HETEROGENEOUS REGION IN WHICH CC CURRENT MATERIAL IS TREATED HETEROGENEOUSLY С CL ((TJESF(I,J),I=1,NESF),J=1,JL)С CW NESF*JL С CD TJESF UNRESOLVED RESONANCE TOTAL INTEGRAL С C---------C--CR FISSION J INTEGRAL (TYPE 6) С CC IF IFI.EQ.1, ALWAYS PRESENT FOR HOMOGENEOUS MIXTURE. ALSO PRESENT FOR EACH HETEROGENEOUS REGION IN WHICH CC CURRENT MATERIAL IS TREATED HETEROGENEOUSLY CC С CL ((FJESF(I,J), I=1, NESF), J=1, JL)С СУ NESF*JL С CD FJESF UNRESOLVED RESONANCE FISSION INTEGRAL С C-

CEOF

-397-

		APPENDIX D.	MC ² −2 B	inary Files.	SCR003 (Con	ntd.)	
C****	****	* * * * * * * * * * * * *	****	****	****	* * * * * * * * * * * * * * * * * * * *	****
С						. •	-
C		•	PREPAR	ED 3/05/75	AT ANL		-
C							-
CF CF		SC2003	תגם ארי	A SET SCROO	3		-
C E		CDCUUD DCIA	ICH DAL	A SEI SCROO	5		-
CN		•	THIS S	CRATCH DATA	SET IS WY	RITTEN	. –
CN			BY MC*	*2-II AREA	6 (CSC005)		-
С			· · · · · · ·				• • •
C * * * *	****	****	*****	*****	* * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *	****
	ر						
CD	NREG		NUMBER	OF REGIONS			·
CD		1	NREG-1	FOR HOMOGE	NEOUS PROI	BLEMS	
CD	, ·	•	NREG=2	FOR CYLIND	RICAL GEOR	IETRY	
CD			NREG = 1	+ THE NUMB	ER OF SLAP	B REGIONS	
CD				FOR SLAB G	EOMETRY		
c							
CS		FILE STRUCT	URE .				-
CS							-
CS		PECORD TYPE			I	PRESENT IF	-
CS	ماد ماد باد باد باد			=======================================	====== , =	.======	
CS CS	*****	TEAK CROSS	SECTION	MES)	1	LUAVS DRESENT	-
CS	*****	**	5 661 100			EWRIG ANDODNI	· _
C						```	-
C		·				&	
C							
CR		PEAK CROSS	SECTION	(TYPE 1)	, í		, —
С							-
СС		ALWAYS PRES	ENT				. –
C	(ma = a0						· _
CL	(NSIGO	(1), I=1, MAX	RES)				-
	MAYRES						-
C "	(15 × 17 P)						-
CD	MAXŔES		MAXIMU	M NUMBER OF	RESOLVED	RESONANCES	-
CD ·	NSIGO		HOMOGE	NIZED MACRO	SCOPIC SIG	GMA ZERO,	-
CD			PEAK R	ESONANCE CR	OSS SECTIO	DN S	-
С							-
CN			NSIGO	IS FILLED W	TTH ZEROS	AS NECESSARY	-
C							
C	 -						

CEOF

,

C****	*****	* * * * * * * * * * * * *	* * * * * * * * * * * * * * *	****	****	*****
С						-
С		•	PREPARED 3/10	/75 AT ANL		-
С						
CF		SCR003		ŧ		-
CE		ULTRA FINE	GROUP CAPTURE	CROSS SECTION	SNC	-
С	,			-		-
CN			BOTH CONFIGUR	ATIONS OF T	HJS SCRATCH D	ATA SET -
CN		•.	SCR003 ARE WR	ITTEN BY MC	**2-II AREA 1	0 -
CN			(CSC011)			-
С						-
C****	* * * * * * *	* * * * * * * * * * * * *	· * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * *	* * * * * * * * * * * * * * * *	****
a b	+ 4					
CD	7.1		FIRST ULTRA F.	INE GROUP R.	EAD FOR CURRE	NT PASS
CD	1Z		LAST ULTRA FI	NE GROUP REA	AD FOR CURREN	T PASS
	NPASS		NUMBER OF TIP.	E I RECORDS	PRESENT IN T	HE FILE.
CD CD			ADDITIONNI DE	NUFGKD*NPAS:	D HUE DEWATN	BE UNE
CD			NUITEDAL REV	JURU REAU FO	JE IDS ESMAIN	TNG
CD	NDDMCD	,	NUMBED OF MAN	NULGAD ULIEN FDINIS IN MI	A LINE GROOPS MATEORG AL	עתווסדי
CD	MENHOE	ī	WHICH HAVE TAL	BULATED CAD	THE FRODUCES SE	CTIONS
CD			TN THE AREA 1/	$\frac{1}{1} \frac{1}{1} \frac{1}$	ENTROV DANCE	CITONS
	NHEGED)	NUMBER OF UTT	RA FINE GROU	IDS READ DER	DNSS
CD	NO. 0 00		$\frac{NUFGRD = T2 - T1 + T}{2}$	1		ENDJ.
00				•		
С	* * *	* * * * * * * * * * * * *	* * * * * * * * * * * * * * * * *	* * *		
С	*			*		
C====	====*	FIRST CON	FIGURATION	*========	================	=================
С	*			*		
С	***	****	****	* * *		
					•	
~	,		·			
C						
CS		FILE STRUCT	URE			-
CS					SSBGBWM TR	-
CS		RECORD TYPE			PRESENT IF	
	ىبە بىد بىد بىد					
	******	TTT (REFERT	wrancy Times)		קטר כווססבאת	ד א ד מים היא א
CS CS	*		מפחוום מאמיתיםים		TUP CORVENT	CROSS -
C3 CS	*	CRUCE LINE	ONG CAPIDAL		SECTIONS IN	- גבליעים
C3 C3	*	CUCCO ORCLE	0110			ENERGY -
C 5	*				RANGE	- 10 <i>22</i> 01 -
CS	*****	**			1.110.00	-
c						-
-						

.

-399-

		· ·		e . 19		· · · ·		
C CR		ULTRA FINE	GROUP	CAPTURE	CROSS'S	ECTIONS (TYPE 1)		
C CC		ALWAYS PRE	SENT					
C CL	(SIGCAP(I),I=NITUFG,NGROUP) NGROUP-NITUFG+1							
CN								
CD.	SIGCAE	(I)	ULTR	A FINE G	ROUP MIC	ROSCOPIC CAPTURE CROSS		
CD CD	NITUFO	;	HIGH	HIGHEST ENERGY ULTRA FINE GROUP ON THE AREA 10				
CD C	NGROUI		LOWE	ST ENERG	Y ULTRA	FINE GROUP IN THE PROBLEM		
C								
CEOF			:					
c	k *k *k	* * * * * * * * * * * *	* * * * * *	*****	****			
C	*				*			
C====	:===:*	SECOND C	ONFIGU	RATION	*====	=======================================		
С	*	، ملك علم	ىلەر بار بار بار بار	به باد باد باد باد باد باد	* * * *			
	4.4.2							
C								
CS		FILE STRUC	TURE		•			
CS CS		RECORD TYP	E			PRESENT IF		
CS CS	* * * * * *	======================================	NPASS	TIMES)				
CS CS	* *	ULTRA FINE SECTIONS	GROUP	CAPTURE	CROSS	ALWAYS PRESENT		
CS CS	* * * * * *	*** ULTRA FINE	GROUP	CAPTURE	CROSS	NULTRA.GT.		
CS		SECTIONS	. ,			NUFGRD*NPASS		
с с								
C								
CR		ULTRA FINE	GRUUP	CAPIONE	, CR033 3			
CC C		ALWAYS PRE	SENT	1				
CL	((SIG	CAP(M,I),M=	1,NPRM	AT),I=I1	, I2)			
CW	NPRMA	T*NUFGRD						
						· · · · · · · · · · · · · · · · · · ·		

.

-400-

С		· · · · · · · · · · · · · · · · · · ·	-
CD	SIGCAP (M, I)	ULTRA FINE GROUP CAPTURE CROSS SECTION FOR	-
CD		MATERIAL M AND ULTRA FINE GROUP I	-
CD	NPRMAT	NUMBER OF MATERIALS IN THE PROBLEM	•
С		-	-
CN		NUFGRD ULTRA FINE GROUPS ARE READ FOR EACH PASS-	-
CN		WHERE NUFGRD=12-11+1. THERE WILL BE -	-
CN		ONE ADDITIONAL RECORD AFTER THE NPASS RECORDS -	-
CN		WHICH CONTAINS NULTRA-NPASS*NUFGRD ULTRA FINE -	-
CN	•	GROUPS IF NULTRA IS NOT AN INTEGRAL MULTIPLE OF-	-
CN		NUFGRD. NULTRA IS THE NUMBER OF ULTRA FINE -	-
CN		GROUPS IN THE AREA 10 (CSCO11) ENERGY RANGE -	-
С			-
C		· - •	-

CEOF

-401-

′ C****	****	*****	****
Ċ			-
C		PREPARED 10/29/75 AT ANL	-
Ċ		, ,	-
CF	SCB003		-
CE	CSC006 SC	RATCH DATA SET	-
C C			-
CN		THIS SCRATCH DATA SET IS	WRTTTEN BY MC**2-TT
CN		BOTH CONFIGURATIONS OF TH	TS SCRATCH DATA SET -
CN .		SCROOS ARE WRITTEN BY MC	(*2) TT AREA 6 5 -
		CRCOAG	·2 II AKGA U-J
(, N) ()	•	(636000)	_
C****	* * * * * * * * * * * * * * * * * * * *	****	***
(++++	क क क क क क क क क क क क क क क क क क क		
CD ·	τοτ	FISSIF MATERIAL TAREY	
	T E 1 ·	TISSILE MAIDNIAL INDEX	ידפדאד
		TET-4 FOR RUN-FISSIDE HAI	T .
	77	TITEL LON LISSIFE HAIPUN	L CUPPENT TROTODE
CD		NUMBER OF SPIN STALES FOR	CURRENT ISOTOPE
CD	NESF	NUMBER OF FIXED ENERGY ME	SH POINTS
CD	NREG	NUMBER OF REGIONS	
CD		NREG=1 FOR HOMOGENEOUS PE	COBLEMS
CD		NREG=2 FOR CYLINDRICAL GE	COMETRY
CD		NREG=1 + THE NUMBER OF SI	AB REGIONS FOR
CĎ		SLAB GEONETRY	
CD	NRES	NUMBER OF RESOLVED RESONA	NCES
CD	NUMRES	NUMBER OF UNRESOLVED RESC	NANCE MATERIALS
С	*****	*****	
C	*	*	
C====	====* FIRST CON	FIGURATION *======	=======================================
С	*	*	
С	******	****	
	,		
C			
CS	FILE STRUCT	URE	-
CS			. -
CS	RECORD TYPE		PRESENT IF -
CS	===========	=======================================	
CS	********** (REPE	AT FOR NUMRES MATERIALS)	-
CS	* MATERIAL	SPECIFICATIONS	ALWAYS -
CS	* ******* (REPE	AT FOR NISO ISOTOPES)	- · · · -
CS	* * SPIN STA	TE DATA	ALWAYS -
cs	* * LEVEL SP	ACING	ALWAYS -
CS	* * ***** (REPE	AT FOR 1+NO. OF REGIONS	-
CS	* * * IN W	HICH MATERIAL IS TREATED	-
CS	* * * HETE	ROGENEOUSLY)	-
ĊS	* * * CAPTURE	J INTEGRAL	ALWAYS -
cs	* * * TOTAL I	INTEGRAL	ALWAYS
~~			

-402-

CS * * * FISSION J INTEGRAL IFI.EQ.1 CS С 1 C-----CR MATERIAL SPECIFICATIONS (TYPE 1) С CC ALWAYS PRESENT С CL NISO,IFI С 2 СЯ С CD NISO NUMBER OF ISOTOPES С C------SPIN STATE DATA (TYPE 2) CR С ALVAYS PRESENT CC С CL JL С CW 1 С ----------C---------CR LEVEL SPACING (TYPE 3) С CC ALWAYS PRESENT С CL ((D(I,J), I=1, NESF), J=1, JL)С CW NESF*JL С AVERAGE LEVEL SPACING CD D С C----C--CAPTURE J INTEGRAL (TYPE 4) CR С CC ALMAYS PRESENT FOR THE HOMOGENEOUS MIXTURE.

-403-

APPENDIX D. MC²-2 Binary Files. SCR003 (Contd.)

cc cc c		ALSO PRESENT FOR EACH HETEROGENEOUS REGION IN WHICH - CURRENT MATERIAL IS TREATED HETEROGENEOUSLY -					
CL	((UCJ(I,J), I=1, NESF), J=1, JL)						
CW C	NESF*J						
CD CD C C	UCJ	UNRESOLVED RESONANCE CAPTURE INTEGRAL TIMES - THE TOTAL REOLVED OVERLAP FACTOR -					
C CR		TOTAL J INTEGRAL (TYPE 5)					
	١	ALWAYS PRESENT FOR THE HOMOGENEOUS MIXTURE ALSO PRESENT FOR EACH HETEROGENEOUS REGION IN WHICH - CURRENT MATERIAL IS TREATED HETEROGENEOUSLY -					
CL	(UTJ)	(I,J),I=1,NESF),J=1,JL) -					
CW CW	NESF*(
CD CD C	UTJ	UNRESOLVED RESONANCE TOTAL INTEGRAL TIMES - THE TOTAL RESOLVED OVERLAP FACTOR -					
C	• •• •• •• •• •• ••						
C		FISSION J INTEGRAL (TYPE 6)					
		IF IFI.EQ.1, ALWAYS PRESENT FOR THE HOMOGENEOUS MIXTURE ALSO PRESENT FOR EACH HETEROGENEOUS REGION IN WHICH - CURRENT MATERIAL IS TREATED HETEROGENEOUSLY -					
CL	((IJFJ	(I,J), I=1,NESF), J=1,JL) -					
CW C	NESF*	JL					
CD CD C	UFJ	UNRESOLVED RESONANCE FISSION INTEGRAL TIMES - THE TOTAL REOLVED OVERLAP FACTOR -					
CEOF							

7

С

-404-

4

с	*			*			
C====	==== *	SECOND	CONFIGURATION	*======================================			
ĉ	*	0.1100110		*			
c	***	* * * * * * * * * * * *	****	· · · · · · · · · · · · · · · · · · ·			
C		• • • • • • • • • • • •	• • • • • • • • • • • • • • • • • • • •	Ť			
C							
ČS.		FTLE STRU	ICTURE				
C 5			OI OND				
C2		שת תפססק					
C3 CC							
	ىلە بىلەر بىلەر بىلەر بىلەر						
03	*****	TT (REPEA	T NREG TIMES)				
CS	*	RESOLVED	CAPTURE INTEGRALS	ALWAYS -			
CS	*	RESOLVED	FISSION INTEGRALS	ALWAYS -			
CS	*	TOTAL RES	SONANCE INTEGRALS	ALWAYS -			
CS	*****	* *		-			
С				-			
C							
				×** '			
C							
CR		RESOLVED	CAPTURE INTEGRAL	(TYPE 1) -			
С		-					
CC		ALWAYS PRESENT FOR HOMOGENEOUS MIXTURE. ALSO PRESENT -					
CC		FOR EACH HETEROGENEOUS REGION					
С				-			
CL	(CJ(I)	,I=1,NRES	5)	-			
С				-			
CW	NRES			-			
C				· -			
CD	CJ		RESOLVED CAPTUR	E INTEGRALS TIMES THE -			
CD	00		APPROPRIATE UNE	ESOLVED OVERLAP FACTORS -			
с С							
C							
C							
C							
CR		RESOLVED	FISSION INTEGRAL	(TYPE 2) -			
С	•						
CC 22		ALWAYS PR	ESENT FOR HOMOGENE	COUS MIXTURE. ALSO PRESENT -			
ČČ		FOR EACH	HETEROGENEOUS REGI				
c		100 2000					
CI	(51/1)	T-1 NPPS		-			
с <u>г</u>	(10 (L)	1 - 1 N V 03	· /	-			
C II	NDEC			_			
	NK 25			-			
C		•					
CD	ťЈ		KESOLVED FISSIC	IN INTEGRALS TIMES THE -			
CD			APPROPRIATE UNE	RESOLVED OVERLAP FACTORS -			
С							
C							

`

-405-

.

C-TOTAL RESONANCE INTEGRAL (TYPE 3) CR С ALWAYS PRESENT FOR HOMOGENEOUS MIXTURE. ALSO PRESENT CC FOR EACH HETEROGENEOUS REGION CC С CL (TJ(I), I=1, NRES)С C₩ NRES Ĉ TOTAL RESONANCE INTEGRAL TIMES THE CD · TJAPPROPRIATE OVERLAP FACTORS CD С C---

CEOF

-406-

APPENDIX D. MC²-2 Binary Files. SCR004

.

]****	*****	* * * * * * * * * * * * *	****	****	****	* *
С			,			-
С			PREPARED 3/	/10/75 AT ANL		-
С						-
CF		SCR004				-
CE		ULTRA FINE (GROUP FIXED	SOURCES		-
С		۱				-
CN			SCRATCH DAT	TA SET WRITTEN	BY MC**2-II AREA 10	-
CN			(CSC011)	•		-
С						-
C****	******	*********	*****	< * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *	*
CD CD CD CD CD	I1 I2 NPASS		FIRST ULTRA LAST ULTRA NUMBER OF T IF NULTRA. ADDITIONAL	A FINE GROUP R FINE GROUP RE TYPE 1 RECORDS GT.NUFGRD*NPAS RECORD READ F	EAD FOR CURRENT PASS AD FOR CURRENT PASS PRESENT IN THE FILE. S, THERE WILL BE ONE OR THE REMAINING	
CD	NURCOR		NULTRA-NPAS	STNUFGRD ULTR.	A FINE GROUPS	
	NUTGRD)	NUMBER OF UNITOR NUTCH	ILIRA FINE GRU	UPS READ PER PASS.	
CD			NUrGRD+12-1		,	
C						
cs		FILE STRUCT	JRE			
CS						-
CS		RECORD TYPE			PRESENJ IF	-
CS ·		=======================================		================	**=====================================	:
CS	*****	*** (REPEAT 1	NPASS TIMES)			-
CS	*	ULTRA FINE (GROUP FIXED	SOURCES	ALWAYS PRESENT	-
CS	*****	* *				-
CS		ULȚRA FINE (GROUP FIXED	SOURCES	NULTRA.GT.	-
CS					NUFGRD*NPASS	-
C			·	,		-
C						• -
C					1	_
CR		ULTRA FINE (BROOP LIYED	SOURCES (IIPE	1)	_
		AT WAVE DEFEN	ጉእነጥ			_
		ALVAIS PRESI	2101			_
C CT	//CETV	(M T) M-1 N	ד (תאמס	T 2)	,	_
C T	((or cv		ENGRIJ,I-II,	12)		_
CW	Иремал	*NUFGBD				_
C						-
CD	SFIX (M	.I)	ÚLTRA FINE	GROUP FIXED S	OURCES FOR	-
CD		• • •	MATERIAL M	AND ULTRA FINI	E GROUP I	-
С			•			-
CN			NUFGRD ULTR	A FINE GROUPS	ARE READ FOR EACH PASS	. –

CN CN CN CN CN CN C	WHERE NUFGRD=12-11+1. THERE WILL BE ONE ADDITIONAL RECORD AFTER THE NPASS RECORDS WHICH CONTAINS NULTRA-NPASS*NUFGRD ULTRA FINE GROUPS IF NULTRA IS NOT AN INTEGRAL MULTIPLE OF NUFGRD. NULTRA IS THE NUMBER OF ULTRA FINE GROUPS IN THE AREA 10 (CSCO11) ENERGY RANGE	
C		-

CEOF

-408-

APPENDIX D. MC²-2 Binary Files. SCR005

1

C****	*****	****	* * * * * * * *	*****	****	****	*****	*****	*****	*****
C C C				PREPARED	3/10/75	A T	ANL			- - -
CF		SCR00	0.5						,	s 🕳
CE		RESON	NANCE FO	IL CROSS	SECTION	S				-
С										-
С										. –
C****	*****	* * * * * *	* * * * * * * *	****	****	****	*****	*****	*****	*****
CD	NBROAD)		NUMBER O	F BROAD	GROU	JPS IN	THE	AREA 10	(CSC011)
CD				ENERGY R	ANGE					
CD	NCNTF			NUMBER O	F RESONA	NCE	FOIL	MATER	IALS	
CD	NFOILS	5		NUMBER O	F FOILS					
CD CD	NINTI			NUMBER O. FOR HOMO	F MESH I GENEOUS	NTEE PROE	RVALS	IN TH	E CELL.	NINTI = 1
C CS		FILE	STRUCTU	IRE						
CS										-
CS		PECOR	RD TYPE					PRESI	ENT IF	-
CS	·	=====			==========	=====	===	=====	========	
CS	*****	*** (}	REPEAT N	BROAD TI	MES)	•		N T 17 N 1		-
CS	*	FOIL	CAPTURE	CROSS S	ECTIONS			ALWAS		-
CS	*	FOIL	FISSION	CROSS S	ECTIONS	NC		ALWAI	15	-
CS	*	FOIL	SCATTER	ING CROS	S SECTIO	INS		ALWA	15	- -
CS C	* * * * * *	* * *								· · · ·
C										
C		·						· ´ ·	`	
CR C C		FOIL	CAPTURE	CROSS S	ECTIONS	(T Y I	?E 1)			
CL . C	((FOI	LCAP(M,K,L), M=1, NCNTF), K=1, NINTI), L=1, NFOILS) -								
CW C	NCNTF*	NINTI	I*NFOILS		·					-
CD CD CD C	FOLCAE	Р(М,К,	,L)	RESONANC M IN FOI INTERVAL	E ÇAPTUR L L AT T I	E CF	ROSS S RIGHT	ECTION	N FOR M. OF MESH	ATERIAL - - - -
C										
CR		FOIL	FISSION	CROSS S	ECTIONS	(TY4	PE 2)			-

-409-

	· .								
C CC CL C C C C C C D C D C D	ALVAYS PRESENT								
	(((FOLFIS(M,K,L), M=1, NCNTF), K=1, NINTI), L=1, NFOILS) -								
	NCNTF*NINTI*NFOILS								
	FOLFIS (M,K,L)	RESONANCE FISSION CROSS SECTION FOR MATERIAL - M IN FOIL L AT THE RIGHT EDGE OF MESH - INTERVAL I -							
С С	· · · · · · · · · · · · · · · · · · ·	~ ~ ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~							
	·								
C CR	FOIL SCATTERING CROSS SECTIONS (TYPE 3)								
CC .	ALWAYS PRESENT								
CLCL									
C W	NCNTF*NINTI*NFOILS -								
C CD CD CD	FOLSCT (M, K, L)	RESONANCE SCATTERING CROSS SECTION FOR MATERIAL- M IN FOIL L AT THE RIGHT EDGE OF MESH INTERVAL I							
C									
CEOF									
••••									
	· · · · ·								

-410-

APPENDIX D. MC²-2 Binary Files. SIGMAP

C ****	* * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *	*****
С	•		-
C C	Ň	PREPARED 3/06/15 AT ANL	· · · · · ·
CF	SIGMAP		_
CE	BACKGROUND	SMOOTH SCATTERING CROSS SEC	CTIONS -
С			-
CN		THIS DATA SET IS WRITTEN I	BY MC**2-II -
CN		AREA 6.5 (CSC006) AND PRES	SENT IF DATA SET -
		ATNUAT HAS BEEN WRITTEN.	JR IF DATA SEL –
C		ALGOAL MAD DDDA WALLUM	. –
C****	* * * * * * * * * * * * * * * * * * * *	****	*****
CD	MAXHTM	MAXIMUM NUMBER OF MATERIAL	LS IN ANY
CD		HETEROGENEOUS REGION	
CD	MGCUT	-UFG ABOVE THE HIGHEST RESO	DNANCE ENERGY IN THE
CD	·	PROBLEM (RESOLVED OR UNRES	SOLVED)
CD	NGROUD	MULTIGROUP TO CONTINUOUS S	SLOWING DOWN
	NGROUP	NUMBER OF PROBLEM MATERIAL	.s
CD	NREG1	REGION INDEX	
CD		NREG1=1 FOR CYLINDERS	
CD		NREG1=NUMBER OF SLAB REGIO	ONS FOR SLABS
		· · · · · · · · · · · · · · · · · · ·	
C			
CS	FILE STRUCT	URE	-
CS	DECODD TYDE	•	
CS CS	ECORD TIPE		
CS	SPECIFICATI	DNS	ALWAYS -
CS	HOMOGENEOUS	SMOOTH SCATTERING	ALWAYS -
CS	CROSS SECTI	ONS	· –
CS	****** (REPEAT	FOR NREG1 HETEROGENEOUS	· _
CS	* KEGIONS		-
CS CS	* * (FEPEAL)	COR NERMAI MAIERIALS	-
CS	* * TREATED	HETEROGENEOUSLY IN THE	
CS	* * CURRENT	REGION. FOR CYLINDERS,	-
CS	*. * ONLY 1	RECORD IS PRESENT SINCE	-
CD	* * ALL MAT	ERIALS HAVE THE SAME	-
CD	* * ESCAPE	CROSS SECTION)	
CS CS		JS SHOUTH SCATTERING	HEATIN.GI.U -
CS	**************************************	0.00	· · · · · · · · · · · · · · · · · · ·
C			. –
C			

.

-411-

APPENDIX D. MC²-2 Binary Files. SIGMAP

C-----CR SPECIFICATIONS (TYPE 1) С CC ALVAYS PRESENT С NGROUP, MGCUT, NPRMAT, NGEOM, NREG1, MAXHTM CL С CW 6 С GEOMETRY TYPE CD NGEOM NCEOM-O FOR HOMOGENEOUS PROBLEMS CD NGEOM=1 FOR SLAB GEOMETRY CD NGEOM=2 FOR CYLINDRICAL GEOMETRY CD С C--C----HOMOGENEOUS SMOOTH SCATTERING CROSS SECTIONS (TYPE 2) CR С CC ALWAYS PRESENT С (SIGMAP(I), I=MGCUT, NGROUP) CL С NGROUP-MGCUT+1 CV С HOMOGENIZED SMOOTH UFG CROSS SECTIONS FOR CD SIGMAP CD THE HOMOGENEOUS MIXTURE С C---_____ C-----HETEROGENEOUS SMOOTH SCATTERING CROSS SECTIONS (TYPE 3) CR С CC / PRESENT IF MAXHTM.GT.O. С (SIGPP(I), I=MGCUT, NGROUP) CL С CW NGROUP-MGCUT+1 С HOMOGENIZED SMOOTH UFG CROSS SECTIONS PLUS CD SIGP? THE ESCAPE CROSS SECTION FOR THE REGION CD IN QUESTION CD С CEOF

-412-

APPENDIX D. MC²-2 Binary Files. SMSIGS

C****	*****	* * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * * *
С				-
C		, • · ·	PREPARED 3/10/75 AT ANL	-
C .				-
C P	· .	CMCTCC	· · · · · · · ·	
Cr		505165		-
CE		MICROSCOPIC	ELASTIC SCATTERING	-
·C			х _х	· -
CN			THIS FILE IS WRITTEN, BY N	IC**2-TT AREA 7 -
CN			(CSC008)	
				_
				-
C * * * *	* * * * * * *	****	* * * * * * * * * * * * * * * * * * * *	******
			·	
CD	TSDODT	7	SDECTRUM ODTION	
	10:051	•		
CD			ISPOPT=1 FOR P1	
CD			ISPOPT=2 FOR B1	
CD			ISPOPT=3 FOR CONSISTENT F	21
CD		,	TSPOPT=4 FOR CONSTSTENT	31
CD		•	2 FOR TRM MACUTARS 1 OF	
C D	TOLT		2 FOR IDE MACHINES, I OTH	I D R W 1 O B
CD	NGRP		NUMBER OF GROUPS IN PROBI	, EM
CD	NORDER	k	ORDER OF EXTENDED TRANSPO	ORT APPROXIMATION
CD	NPRMAT	1	NUMBER OF PROBLEM MATERIA	LS
-				
C				
CS		FILE STRUCT	JRE	-
CS				· · · · ·
CS		RECORD TVDE		DRESENT TE -
CC			·-·	
CS				
CS		SPECIFICATIO	DNS	ALWAYS -
CS		MATERIAL NAM	IES	ALWAYS -
CS	*****	** (REPEAT F	FOR NERMAT MATERIALS)	· –
C S	* ***	** (DEDENT 1	TOP NCPD CPOUDS)	_
03		TIACETC COM	TOR NORP GROUPS)	
CS	* *	SLASTIC SCAT	TERING DATA	ALWAYS -
CS	*****	***	·	• –
С				, –
C				
-				
				· · ·
_				
C				
CR		SPECIFICATIO	NS (TYPE 1)	-
C			• • •	-
čc		ATUAVO DEPCE	יאסי	_
		HLARIS FRESE	5 N 1	-
C			· .	-
CL	NPRMAT	,NGRP,ISPOPI	, NORDER, NGROP, ITRANS	-
С			-	-
CW	6			-
	0			· · · · ·
C				
CD.	NGROP		LIBRARY UFG NUMBER CORRES	PONDING TO THE -
CD			HIGHEST UFG IN THE PROBLE	M -

-413-

CD CD CD CD C C C	ITRANS	TRANSPORT APPROXIMATION ITRANS=0 USE ALL LEGENDRE COMPONENTS ITRANS=1 USE STANDARD TRANSPORT APPROXIMATION ITRANS=2 USE IMPROVED TRANSPORT APPROXIMATION			
C					
CR	MATERIAL NA	MES (TYPE 2)	-		
cc	ALWAYS PRESENT				
CL	(PRBNAM (I), I=1, NPRMAT)				
CW ,	MULT*NPRMAT		-		
C CD C	PRBNAM	PROBLEM MATERIAL NAMES	-		
C			-		
C CR	FLASTIC SCA	TTERING DATA (TYPE 3)	-		
cc	ALWAYS PRESENT				
CL CL C	(SIGS(1), I=1,NORD 1(SIG1(1),I=1,NDN1)	1), SIGUNR, BETA, (SIGO (I), $I = 1$, NDN 1),	-		
CW C	NORD1+ISP*NDN1+2		-		
CD CD CD CD CD	NORD 1=NORDER+1 ISP=1 IF ISPOPT.L. ISP=2 IF ISPOPT.G NDN1=NUMBER OF GRO	E.2 T.2 OUPS OF DOWNSCATTER PLUS 1			
	SIGS	LEGENDRE MOMENTS OF NONRESONANCE SCATTERING CROSS SECTION I=1 CORRESPONDS TO THE NORDER COMPONENT I=NORDER CORRESPONDS TO THE P1 COMPONENT I=NORD1 CORRESPONDS TO THE P0 COMPONENT			
CD CD CD CD CD	SIGUNR BETA SIGO	UNRESOLVED RESONANCE SCATTERING CROSS SECTION TRANSPORT CORRECTION FACTOR PO SCATTERING MATRIX ORDERED AS J TO J+I-1 SO THAT I=1 IS THE IN-GROUP TERM			
CD CD CN	, 	J TO J+I-1 SO THAT I=1 IS THE IN-GROUP TERM THE ARRAY SIG1 IS PRESENT ONLY FOR ISP=2	-		
APPENDIX D. MC²-2 Binary Files. SMSIGS (Contd.)

-415-

		•
CN	THE MATRIX ELEMENTS SIGO AND SIG1 ARE	-
CN	NORMALIZED TO THE TRANSPORT CORRECTED SMOOTH	-
CN	PLUS UNRESOLVED SCATTERING CROSS SECTION,	' -
CN	THAT IS ((SIGS (NORD1) + SIGUNR) * (1 BETA))	` -
С		-
C		
CEOF		

APPENDIX D. MC²-2 Binary Files. SPECTR

C****	****	* * * * * * * * * * * * * * * * * * * *	*****
Ċ	1		-
č		DDEDADED 5/13/76 AT ANT	-
C		PREPARED J/15/10 AI AND	
С		,	-
CF	SPECTR		-
CE	ULTRA F	INE GROUP SPECTRUM	-
Ċ		· · · · · · · · · · · · · · · · · · ·	-
CN		MUTC PTIP TC UPTAVEN BV N	
CN		INTO FILE TO WATTIEN DI P	ITTZ-II ARDA O
CN		(CSC009)	-
С			· •
C****	******	*****	*****
		· · · · ·	
~~	* ~ ~ ~ ~ .		TH ADDITONS
CD	15P	FOR INCONSISTENT SPECTS	COM OPTIONS
CD		2 FOR CONSISTENT SPECTRUM	OPTIONS
CD	NGRP	NUMBER OF UFG IN PROBLEM	
	· ·	· · · · · · · · · · · · · · · · · · ·	
		,	·
<u> </u>			
(
CS	FILE ST	RUCTURE	-
CS)	-
CS	RECORD 3	TYPE	PRESENT IF -
C S			
C3	CDECIEL		AT WAYC -
CS	SPECIFIC	CATIONS	ALWAIS
CS	FLUX		ALWAYS -
CS	CURRENT		ISP.EQ.2 -
С			-
Č			
C			
C			
CR	SPECIFI	CATIONS (TYPE 1)	**
۲.			-
	ATTAVC	ה ה ה ה ה את	-
	ALWAID .	PKESEN1	
С			. –
CL ·	RHO, BSQ, EMQX,	DELTAU, NGRP, MGCSD, NCSD	-
С			-
ĊW	7		-
C "	•		• · · _
			-
CD	KHO	ULTRA FINE GROUP EIGENVAI	.UE -
CD	BSQ	B**2 FROM UFG CALCULATION	
CD ·	EMAX	HIGHEST ENERGY IN PROBLEM	1 -
CD CD	DEL TAIL	UFG LETHARGY WIDTH	-
	MCCSD		NHOUS SLOWING DOWN -
	rige 5 D	OFG NUMBER AI WAICH CONTI	THOOD STOATUR DOMM
CD		CALCULATION BEGINS	-
CD	NCSD	NUMBER OF UFG IN CONTINUC	US SLOWING DOWN -
CD		CALCULATION	-
Ċ		-	-
U			

.

.

APPENDIX D. MC²-2 Binary Files. SPECTR (Contd.)

~			
CR		FLUX (TYPE 2)	-
C			
сс		ALWAYS PRESENT	· · -
С			· –
CL C	(PHI (I), I=1, NGRP)	-
CW C	NGRP		-
CD C	PHI	ULTRA FINE GROUP FLUX	-
c			
CR CR		CURRENT (TYPE 3)	-
cc c		PRESENT IF ISP.EQ.2	-
CL C	(CURN)	T (I), I=1, NGRP)	- -
CW C	NGRP	· ·	-
CD C	CURNT	ULTRA FINE GROUP CURRENT	- -
C			

CEOF

-417-

APPENDIX D. MC²-2 Binary Files. SPECXS

C****	* * * * * * * * * * * * * * * * * * * *	* ***************	**
C · ·	· ·	PREPARED 3/12/75 AT ANL	-
C CF	SPECXS	e e e e e e e e e e e e e e e e e e e	-
CE C	MACROSCOPI	C CROSS SECTIONS AND MODERATING PARAMETERS	-
C N CN	r	THIS FILE IS WRITTEN BY MC**2-II AREA 7 (CSC008)	-
С			-
C***	****	* * * * * * * * * * * * * * * * * * * *	* *
CD CD CD CD	ISP	SPECTRUM OPTION ISP=1 FOR INCONSISTENT OPTION (ISPOPT.LE.2) ISP=2 FOR CONSISTENT SPECTRUM OPTION (ISPOPT.GE.3)	
C D C D	NCSD	NUMBER OF ENERGY POINTS IN CONTINUOUS SLOWING DOWN REGION (=NGRP-MGCSD+2)	
CD	NGRP	NUMBER OF UFG IN PROBLEM	
C			
CR C	NUSIGMA FIS	SSION (TYPE 1),	-
CC	ALWAYS PRES	5 E NT	-
CL	(BNSIGF (I), I=1, NG	GRP)	-
CW C	NGRP		-
CD CD C	BNSIGF	MACROSCOPIC UFG FISSION CROSS SECTION* AVERAGE NUMBER OF NEUTRONS PER FISSION	-
.C		· · · · · · · · · · · · · · · · ·	
C	NU COEFFIC	LENTS (TYPE 2)	
C CC	ALWAYS PRES	SENT	-
CL	(AO(I), A1(I), A2(I	I), A3(I), I=1, NPRMAT)	-
CW	4 * N P R M A T		-
CD CD CD CD	NPRMAT A0, A1, A2, A 3	NUMBER OF PROBLEM MATERIALS MATERIAL DEPENDENT COEFFICIENTS USED IN FIT OF AVERAGE NUMBER OF NEUTRONS PER FISSION VERSUS ENERGY	-

-418-

APPENDIX D. MC²-2 Binary Files. SPECXS (Contd.)

С C--C--TOTAL CROSS SECTION (TYPE 3) CR С CC ALVAYS PRESENT С CL (BSIGT(I), I=1, NGRP)С C₩ NGRP С MACROSCOPIC UFG TOTAL CROSS SECTION CD BSIGT С C-C---CR SCATTERING CROSS SECTION (TYPE 4) С CC ALVAYS PRESENT Ċ. CL (BSTGS(I), I=1, NGRP)С CW NGRP С CD BSIGS MACROSCOPIC UFG PO SCATTERING CROSS SECTION С C----EXTENDED TRANSPORT CROSS SECTION (TYPE 5) CR С CC, ALVAYS PRESENT С CL ((AL(I,K), I=1, NGRP), K=1, NORDER)С NGRP*NORDER CW С ĊD NORDER ORDER OF EXTENDED TRANSPORT APPROXIMATION MACROSCOPIC UFG EXTENDED TRANSPORT CROSS CD AL CD SECTION С]______ P1 SCATTERING (TYPE 6) CR

-419-

APPENDIX D. MC²-2 Binary Files. SPECXS (Contd.)

1

5

С	<i>,</i>						
CC C	,	ALWAYS PRESENT					
CL	(SIGTRN(I), I=1, NGRP)						
CW	NGRP	· · · · · · · · · · · · · · · · · · ·					
C CD C	SIGTRN	MACROSCOPIC UFG P1 SCATTERING CROSS SECTION -					
C							
CR		MODERATING PARAMETER ZETA (TYPE 7)					
cc		ALWAYS PRESENT					
C CL	((ZETA(I,J),T=1,NCSD),J=1,ISP)						
CW CW	ISP*NC	CSD -					
CD CD CD CD C	ZETA	CONTINUOUS SLOWING DOWN MODERATING PARAMETER - ZETA AT EACH ENERGY POINT IN CSD REGION FOR - PO AND P1 SCATTERING -					
C	<i>;</i>)					
CR		MODERATING PARAMETER EPS (TYPE 8)					
cc		ALWAYS PRESENT					
CL	((EPS)	(I,J),I=1,NCSD),J=1,ISP) -					
C C W	JSP*NC	CSD -					
C CD CD CD C C	EPS	CONTINUOUS SLOWING DOWN MODERATING PARAMETER - 1./GAMMA AT EACH ENERGY POINT IN CSD REGION FOR- PO AND P1 SCATTERING -					
ĆEOF							

-420-

.

APPENDIX D. MC²-2 Binary Files. SRATES

C * * * * *	*****	* * * * * * * * * * * * *	*****	******	*****	*****	***
С					ć	3	-
Ċ			PREPARED	3/06/75	AT ANT.		-
c ·			1 1 2 1 1 2 2	5/00/15			-
C D		CDAMBC					_
CF		SRATES					-
CE		SCATTERING	RATES AND	SOURCES			-
С							-
CN			SCATTERIN	G RATES	AND SOUR	CES FOR USE IN	-
CN			AREA 10 (CSC011)	INTEGRAL	TRANSPORT THEORY	-
CN			CALCULATI	ONS			_
C			CALCODALL	010			_
C	اد ماد عاد ماد ماد ما	ا الله مله عله عله عله عله عله عله عله عله عله ع	بله	د بد بد بار بد بد بد بد		* * * * * * * * * * * * * * * * * * * *	
(****	* * * * * * * *	* * * * * * * * * * * * * * * * * * * *	• • • • • • • • • • • • • • • • • • • •	***	• • • • • • • • • • •	• • • • • • • • • • • • • • • • • • • 	• ተ ተ ተ
CD	MAXUFO		LOWEST EN	ERGY UL?	TRA FINE (GROUP FOR WHICH	
CD		•	SCATTERIN	G RATES	ARE PROV	TDED. ULTRA FINE	
CD	•		CPOUD MAY	UPC TS "	גסתוו קעק	FINE GROUP	
			GROUP MAA				
CD			IMMEDIATE	TI VROAF	S THE TOP	OF THE HIGHEST	
CD			ENERGY BR	OAD GROU	JP IN THE	AREA 10 (CSC011)	
CD)	ENERGY RA	NGE			
CD	MINUFO	5	HIGHEST E	NERGY UI	TRA FINE	GROUP NUMBER FOR	
CÐ		,	WHICH SCA	TTERING	RATES AR	E PROVIDED	
CD	NDASS		NUMBER OF	TYPE 2	RECORDS	PRESENT IN THE FILE	'
CD		.	NUMBER OF		NECONDO 1	F DROBTEM	
	NPRMAT	· ·	NUMBER OF	MAIERIA		E PROBLEM	. ~
CD	NUFGRI)	NUMBER OF	ULTRA E	TNE GROUI	PS <u>READ</u> FOR EACH LAS	55
CD			OF RECORD) 2. NUF	GRD = I2 - I	1+1 (SEE RECORD 2)	
							*
C							
CS		FILE STRUCT	JRE		•		-
CS				•			-
CS		RECORD TYPE				PRESENT IF	-
CS.		=======================================	=========================	=======================================			==-
C5 C5		SDECTETCART	ONC				
C 5	الدماد ماد ماد باد ماد	SECTIONIT					-
CS	* * * * * *		UNS JDACC MINE	C)		ALWAYS	-
CS		*** (REPEAT	NPASS TIME	S)		ALWAYS	- -
	*	*** (REPEAT) Sources	NPASS TIME	(S)		ALWAYS NPASS.GT.0	_ _ _
CS	* * * * * * * *	*** (REPEAT) Sources ***	UNS NPASS TIME	S)		ALWAYS NPASS.GT.0	- - -
CS CS	* * * * * * * *	*** (REPEAT) SOURCES *** SCATTERING]	ONS NPASS TIME RATES	S)		ALWAYS NPASS.GT.O ALWAYS PRESENT	-
CS CS CS	* * * * * * *	*** (REPEAT) SOURCES *** SCATTERING HYDROGEN SC	ONS NPASS TIME RATES ATTERING R	ATES	•	ALWAYS NPASS.GT.O ALWAYS PRESENT PRESENT IF NHYDRO=1	
CS CS CS	* * * * * * * *	*** (REPEAT) SOURCES *** SCATTERING : HYDROGEN SC	NPASS TIME RATES ATTERING R	S) Ates	•	ALWAYS NPASS.GT.O ALWAYS PRESENT PRESENT IF NHYDRO=1 (SEE RECORD 1 OF DA	
CS CS CS CS	* * * * * * *	*** (REPEAT) SOURCES *** SCATTERING) HYDROGEN SC	NPASS TIME RATES ATTERING R	S) Ates	•	ALWAYS NPASS.GT.O ALWAYS PRESENT PRESENT IF NHYDRO=1 (SEE RECORD 1 OF DA	- - - - - - - - - - - - - - - - - - -
CS CS CS CS	* * * * * * * *	*** (REPEAT) SOURCES *** SCATTERING) HYDROGEN SC	NPASS TIME RATES ATTERING R	S) ATES		ALWAYS NPASS.GT.O ALWAYS PRESENT PRESENT IF NHYDRO=1 (SEE RECORD 1 OF DA SET PRBSPC)	- - - - - - - - - - - - - - - - - - -
CS CS CS CS CS C	* * * * * * * *	*** (REPEAT) SOURCES *** SCATTERING) HYDROGEN SC	ONS NPASS TIME RATES ATTERING R	S) ATES		ALWAYS NPASS.GT.O ALWAYS PRESENT PRESENT IF NHYDRO=1 (SEE RECORD 1 OF DA SET PRBSPC)	
cs cs cs cs cs c c c c c	* * * * * * * *	*** (REPEAT) SOURCES *** SCATTERING HYDROGEN SC	NPASS TIME RATES ATTERING R	S) ATES		ALWAYS NPASS.GT.O ALWAYS PRESENT PRESENT IF NHYDRO=1 (SEE RECORD 1 OF DA SET PRBSPC)	
cs cs cs cs cs c C	* * * * * * * *	*** (REPEAT) SOURCES *** SCATTERING : HYDROGEN SC	NPASS TIME RATES ATTERING R	S) ATES		ALWAYS NPASS.GT.0 ALWAYS PRESENT PRESENT IF NHYDRO=1 (SEE RECORD 1 OF DA SET PRBSPC)	
cs cs cs cs c c c c c c	* * * * * * *	*** (REPEAT) SOURCES *** SCATTERING HYDROGEN SC	NPASS TIME RATES ATTERING F	S) ATES		ALWAYS NPASS.GT.O ALWAYS PRESENT PRESENT IF NHYDRO=1 (SEE RECORD 1 OF DA SET PRBSPC)	
cs cs cs cs c c c c c c c c c c c c c c	* * * * * * *	*** (REPEAT) SOURCES *** SCATTERING HYDROGEN SC	NPASS TIME RATES ATTERING R	2S) ATES		ALWAYS NPASS.GT.O ALWAYS PRESENT PRESENT IF NHYDRO=1 (SEE RECORD 1 OF DA SET PRBSPC)	
CS CS CS CS C C CR	* * * * * * * *	*** (REPEAT) SOURCES *** SCATTERING HYDROGEN SC	NPASS TIME RATES ATTERING R 	2S) ATES 		ALWAYS NPASS.GT.O ALWAYS PRESENT PRESENT IF NHYDRO=1 (SEE RECORD 1 OF DA SET PRBSPC)	
CS CS CS CS C C CR C	* * * * * * *	*** (REPEAT) SOURCES *** SCATTERING HYDROGEN SC	NPASS TIME RATES ATTERING R	2S) ATES 1)		ALWAYS NPASS.GT.O ALWAYS PRESENT PRESENT IF NHYDRO=1 (SEE RECORD 1 OF DA SET PRBSPC)	
CS CS CS CS C C CR C CL	* ******	*** (REPEAT) SOURCES *** SCATTERING HYDROGEN SC SPECIFICATION , MAXUFG, NUFO	NPASS TIME RATES ATTERING R 	2S) ATES 1)		ALWAYS NPASS.GT.0 ALWAYS PRESENT PRESENT IF NHYDRO=1 (SEE RECORD 1 OF DA SET PRBSPC)	
CS CS CS CS C C C C C C C C C C C C C C	* ******	*** (REPEAT) SOURCES *** SCATTERING HYDROGEN SC SPECIFICATION , MAXUFG, NUFO	NPASS TIME RATES ATTERING R 	2S) ATES 		ALWAYS NPASS.GT. 0 ALWAYS PRESENT PRESENT IF NHYDRO=1 (SEE RECORD 1 OF DA SET PRBSPC)	

-421-

APPENDIX D. MC²-2 Binary Files. SRATES (Contd.)

CT 4 С C-_____ C-----SOURCES (TYPE 2) CR С CC PRESENT IF NPASS.GT.0 С ĊL ((SFIX(M,I),M=1,NPRMAT),I=I1,I2) (: CW NUFGED*NPRMAT С CD SFIX (M,T) MICROSCOPIC SOURCE INTO ULTRA FINE GROUP I CD DUE TO FISSION, INELASTIC SCATTERING, AND N, 2N SCATTERING IN MATERIAL M. CD CD I RANGES OVER THE ULTRA FINE GROUPS IN THE CD AREA 10 (CSC011) ENERGY RANGE, MAXUFG+1 CD THROUGH NGROUP, WHERE NGROUP IS THE CD LOWEST ENERGY ULTRA FINE GROUP IN THE PROBLEM. CD NUFGRD ULTRA FINE GROUPS ARE INCLUDED FOR EACH CD PASS. FIRST ULTRA FINE GROUP READ FOR CURRENT PASS CD I1 CD I2 LAST ULTRA FINE GROUP READ FOR CURRENT PASS С CR SCATTERING RATES (TYPE 3) С CC ALWAYS PRESENT С CL ((SS(I,M), I=1, MINMAX), M=1, NPRO)С CW NPRO*MINMAX С SS CD PRODUCT OF MICROSCOPIC SCATTERING CROSS SECTION TIMES ÜLTRA FINE GROUP FLUX DIVIDED BY CD CD THE ULTRA FINE GROUP LETHARGY WIDTH FOR THE CD FINE GROUPS MINUFG THROUGH MAXUFG FOR EACH CD MATERIAL CD MINMAX MAXUFG-MINUFG+1 NPRMAT IF HYDROGEN IS NOT PRESENT IN THE CD NPRO CD PROBLEM MIXTURE. NPRMAT-1 IF HYDROGEN IS IN CD PROBLEM MIXTURE С C-

-422-

APPENDIX D. MC²-2 Binary Files. SRATES (Contd.)

С-_____ HYDROGEN SCATTERING RATES (TYPE 4) CR С PRESENT IF NHYDRO=1 (SEE RECORD 1 OF DATA SET PRBSPC) CC С CL (SHYDRO(I), I=1, MAXUFG)С ′C₹ MAXUFG С CD SHYDRO PRODUCT OF HYDROGEN MICROSCOPIC SCATTERING CROSS SECTION TIMES ULTRA FINE GROUP FLUX CD DIVIDED BY THE ULTRA FINE GROUP LETHARGY WIDTH CD -FOR ALL ULTRA FINE GROUPS ABOVE THE AREA 10 CD (CSC011) ENERGY RANGE CD C. C--

CEOF

-423-

C***	****	* * * * * * * * * * * * * * * * * * * *	**:
С			
c c		DREDARED 3/10/75 AT AND	
		PREPARED SYTUY IS AT ANE	
	HND D C	,	
CF	UNREG		
CE	UNRESOLV	VED UFG CROSS SECTIONS	
С			
CN		THIS FILE IS WRITTEN BY MC**2-II AREA 7	
CN		(CSC008)	
с		· ·	
C * * *	*****	* * * * * * * * * * * * * * * * * * * *	**
CD	I.F.I.	FISSION FLAG	
CD		IFI=0 MATERIAL IS NOT FISSIONABLE	
CD		IFI=1 MATERIAL IS FISSIONABLE	
CD	IMAX1	LOWEST ENERGY UFG FOR WHICH MATEPIAL HAS	
CD		NON-ZERO UNRESOLVED CROSS SECTION	
CD	IMTN1	HIGHEST ENERGY UFG FOR WHICH MATERIAL HAS	
C D		NON-ZERO UNRESOLVED CROSS SECTION	
	M ፲፲፻ ጥ	2 FOR TRM MACHINES 1 OTHERETSE	
	NDEC	Z FOR IDE HACHINES, V OINERWISE	
	NACG	NUMBER OF REGIONS	
		$\mathbf{N} \mathbf{D} \mathbf{E} \mathbf{C} = 1 \mathbf{E} \mathbf{O} \mathbf{D} \mathbf{U} \mathbf{O} \mathbf{K} \mathbf{O} \mathbf{C} \mathbf{E} \mathbf{N} \mathbf{E} \mathbf{O} \mathbf{U} \mathbf{C} \mathbf{D} \mathbf{D} \mathbf{O} \mathbf{D} \mathbf{I} \mathbf{E} \mathbf{M} \mathbf{C} \mathbf{I} \mathbf{N} \mathbf{C} \mathbf{E} \mathbf{O} \mathbf{M} = \mathbf{O} \mathbf{V}$	
CD		NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0)	
CD CD		NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF	
CD CD CD		NREG=1 FOR HOMOGENEOUS PROBLEMS (NGFOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.0	
CD CD CD CD		NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.O NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB	_
CD CD CD CD CD		NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.O NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT.	0
CD CD CD CD CD CD		NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.O NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS	0 IN
CD CD CD CD CD CD CD		NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.O NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION)	0 IN
CD CD CD CD CD CD CD CD CD	N U M R E S	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.O NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS	0 IN
	NUMRES	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.0 NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS	0 IN
	NUMRES FILE STR	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.O NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS RUCTURE	0 IN
CD CD CD CD CD CD CD CD CD CD CD CD CD C	NUMRES FILE STI PECORD 1	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.0 NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS RUCTURE TYPE PRESENT IF	0 IN
CD CD CD CD CD CD CD CD CD CD CD CD CD C	NUMRES FILE STI PECORD I =======	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.0 NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS RUCTURE PRESENT IF PRESENT IF	0 IN
CD CD CD CD CD CD CD CD CD CD CD CD CD C	NUMRES FILE STI PECORD I SPECIFIC	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.0 NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS RUCTURE TYPE PRESENT IF CATIONS ALWAYS	0 IN
CD CD CD CD CD CD CD CD CD CD CD CD CD C	NUMRES FILE STE PECORD I ====== SPECIFIC UNRESOLV	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.0 NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS RUCTURE TYPE PRESENT IF CATIONS ALWAYS VED MATERIAL NAMES ALWAYS	0 IN ====
CD CD CD CD CD CD CD CD CD CD CD CD CD C	NUMRES FILE STE PECORD T ======= SPECIFIC UNRESOLV ******* (REPE	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.0 NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS RUCTURE TYPE PRESENT IF CATIONS ALWAYS VED MATERIAL NAMES ALWAYS EAT FOR NUMRES MATERIALS) ALWAYS	0 IN ====
CD CD CD CD CD CD CD CD CD CD CD CD CD C	NUMRES FILE STR PECORD T ======= SPECIFIC UNRESOLV ******* (REPR * MATERIAL	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.0 NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS RUCTURE TYPE PRESENT IF CATIONS ALWAYS VED MATERIAL NAMES ALWAYS EAT FOR NUMRES MATERIALS) ALWAYS L SPECIFICATIONS ALWAYS	0 IN ====
CD CD CD CD CD CD CD CD CD CD CD CD CD C	NUMRES FILE STR PECORD T ======= SPECIFIC UNRESOLV ******** (REPR * MATERIAL * ***** (REPR	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.0 NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS RUCTURE PRESENT IF CATIONS ALWAYS VED MATERIAL NAMES ALWAYS EAT FOR NUMRES MATERIALS) ALWAYS L SPECIFICATIONS ALWAYS EAT FOR 1+NO. OF REGIONS ALWAYS	0 IN ====
CD CD CD CD CD CD CD CD CD CD CD CD CD C	NUMRES FILE STR PECORD T ====== SPECIFIC UNRESOLV ******* (REPR * MATERIAN * ***** (REPR * N N V	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.0 NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS RUCTURE TYPE PRESENT IF ====================================	0 IN ====
	NUMRES FILE STR PECORD T ====== SPECIFIC UNRESOLV ******* (REPR * MATERIAN * ***** (REPR * MATERIAN * ** IN *	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=O) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.O NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS RUCTURE TYPE PRESENT IF TYPE PRESENT IF CATIONS ALWAYS VED MATERIAL NAMES ALWAYS EAT FOR NUMRES MATERIALS) L SPECIFICATIONS ALWAYS EAT FOR 1+NO. OF REGIONS WHICH MATERIAL IS TREATED EROGENEOUSLY)	0 IN ===
	NUMRES FILE STR PECORD T ====== SPECIFIC UNRESOLV ******* (REPR * MATERIAR * ***** (REPR * MATERIAR * ** HETR * * HETR * * HETR	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.0 NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS RUCTURE PRESENT IF CATIONS ALWAYS VED MATERIAL NAMES ALWAYS EAT FOR NUMRES MATERIALS) ALWAYS L SPECIFICATIONS ALWAYS EAT FOR 1+NO. OF REGIONS ALWAYS EAT FOR 1+NO. OF REGIONS ALWAYS EAT FOR 1+NO. OF REGIONS ALWAYS CROSS SECTION IMIN1.GT.0	0 IN ====
	NUMRES FILE STR PECORD T ====== SPECIFIC UNRESOLV ******** (REPR * MATERIAR * ***** (REPR * MATERIAR * ** HETR * * HETR * * HETR * * FISSION	NREG=1 FOR HOMOGENEOUS PROBLEMS (NGEOM=0) NREG=2 FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.0 NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALS RUCTURE PRESENT IF CATIONS ALWAYS VED MATERIAL NAMES ALWAYS VET FOR NUMRES MATERIALS) ALWAYS L SPECIFICATIONS ALWAYS EAT FOR 1+NO. OF REGIONS ALWAYS EAT FOR 1+NO. OF REGIONS ALWAYS EAT FOR 1+NO. OF REGIONS MIN1.GT.0 CROSS SECTION IMIN1.GT.0 CROSS SECTION IMIN1.GT.0	0 IN ====
	NUMRES FILE STR PECORD T ====== SPECIFIC UNRESOLV ******* (REPR * MATERIAL * ***** (REPR * MATERIAL * ** HETR * * HETR * * HETR * * FISSION * * TOTAL CE	NREG=1FOR HOMOGENEOUS PROBLEMS (NGEOM=0)NREG=2FOR PIN CELLS (NGEOM=2) AND IF MAXHTM.GT.0NREG=1 + NUMBER OF SLAB REGIONS FOR SLAB PROBLEMS (NGEOM=1) AND IF MAXHTM.GT. (MAXHTM IS THE MAXIMUM NUMBER OF MATERIALS ANY HETEROGENEOUS REGION) NUMBER OF UNRESOLVED RESONANCE MATERIALSRUCTUREPRESENT IF PRESENT IF ETHER RATERIAL NAMES EAT FOR NUMRES MATERIALS) L SPECIFICATIONS L SPECIFICATIONS ALWAYS EAT FOR 1+NO. OF REGIONS MHICH MATERIAL IS TREATED EROGENEOUSLY) CROSS SECTION CROSS SECTIONIMIN1.GT.0 IMIN1.GT.0, IFI.G'	0 IN ====
	NUMRES FILE STI PECORD T ====== SPECIFIC UNRESOLV ******** (REPI * MATERIAI * ***** (REPI * MATERIAI * ** HETI * * HETI * * CAPTURE * * FISSION * * TOTAI CE	NREG=1FORHOMOGENEOUSPROBLEMS (NGEOM=0)NREG=2FORPIN CELLS (NGEOM=2)AND IFMAXHTM.GT.0NREG=1+ NUMBER OF SLAB REGIONS FOR SLABPROBLEMS (NGEOM=1)AND IF MAXHTM.GT.(MAXHTM IS THE MAXIMUM NUMBER OF MATERIALSANY HETEROGENEOUS REGION)NUMBER OF UNRESOLVED RESONANCE MATERIALSRUCTURETYPEPRESENT IFCATIONSALWAYSVED MATERIAL NAMESALWAYSEAT FOR NUMRES MATERIALS)ALWAYSL SPECIFICATIONSALWAYSEAT FOR 1+NO. OF REGIONSALWAYSEAT FOR 1+NO. OF REGIONSMHICH MATERIAL IS TREATEDEROGENEOUSLY)CROSS SECTIONIMIN1.GT.0CROSS SECTIONIMIN1.GT.0, IFI.G'ROSS SECTIONIMIN1.GT.0	0 IN ====

.

.

-424-

.

APPENDIX D. MC²-2 Binary Files. UNREG (Contd.)

U.

C	SPECIFICATIONS (TYPE 1)
cc	AL"AYS PRESENT
C CL	NUM RES, NREG
C C₩ C	2
C	
~	
C CR	UNRESOLVED MATERIAL NAMES (TYPE 2)
cc c	ALWAYS PRESENT
C L C	(UNRMAT (I), I=1, NUMRES)
CW C	MULT*NUMRES -
C D C	UNRMAT NAME OF UNRESOLVED MATERIAL (DOUBLE PRECISION) -
C	
C÷ CR	MATERIAL SPECIFICATIONS (TYPE 3) -
c cc	ALWAYS PRESENT
C CL	- IFI, IMIN1, IMAX1
C CW	3
C	-
C	
C CR	CAPTURE CROSS SECTION (TYPE 4)
	PRESENT FOR HOMOGENEOUS MIXTURE IF IMIN1.GT.O ALSO PRESENT FOR EACH HETEROGENEOUS REGION IN WHICH CURRENT- MATERIAL IS TREATED HETEROGENEOUSLY -
CL CL	(SIGC(I), I=IMIN1, IMAX1)
CW	IMAX1-IMIN1+1 -

.

APPENDIX D. MC^2-2 Binary Files. UNREG (Contd.)

۱

C CD C		SIGC		UNRESOLVED	RESONANCE	CAPTURE (CROSS SECTI	
C	ţ							,)
C R C R			FISSION CROS	S SECTION	(TYPE 5)			-
		۵	PRESENT FOR ALSO PRESENT MATERIAL IS	HOMOGENEOUS FOR EACH I TREATED HES	5 MIXTURE HETEROGENE CEROGENEOU	IF IMIN1. OUS REGION SLY AND IN	GT.O, IFI.O N IN WHICH FI.GT.O	GT.0 CURRENT- -
CL C		(SIGF	(I),I=IMIN1,I	MAX1)	s			-
CW C		IMAX1-	IMIN1+1					-
CD C		SIGF	.*	UNRESOLVED	RESONANCE	FISSION (CROSS SECTI	- IONS
C			·					
CP CP	.,		TOTAL CROSS	SECTION (T)	(PE 6)			-
	•.		PRESENT FOR ALSO PRESENT MATERIAL IS	HOMOGENEOUS FOR EACH F TREATED HES	5 MIXTURE HETEROGENE FEROGENEOU	IF IMIN1.0 OUS REGIO SLY	GT.O. N IN WHICH	CURRENT-
CL		(SIGT)	(I),I=IMIN1,I	MAX 1)				-
C W	-	IMAX1-	IMIN 1+1					· _
CD C C		SIGT		UNRESOLVED	RESONANCE	TOTAL CRO	DSS SECTION	IS - -
			,					

CEOF

-426-

APPENDIX D. MC^2-2 Binary Files. UNRES

C****	****	* * * * * * * * * * * * * * *	******	*****	****	****
С						-
C C		PREPARED	3/05/15 A1	CANL	· · ·	-
CF	UNRES					-
CE	UNRESOL	VED RESONANCE	CROSS SECT	TIONS		-
С						-
CN		THIS DATA	SET IS WE	RITTEN B	Y MC**2-II	AREA 5 -
CN C		(CSC004)	IF UNRESUL	LVED MAT	ERIALS ARE	PRESENT -
C****	****	, **** ********	*****	*****	* * * * * * * * * * * *	****
-		``	·	~		
CD	MULT	2 FOR IBM	MACHINES,	1 OTHE	RWISE	
CD	NISO	NUMBER OF	ISOTOPES	FOR EAC	H MATERIAL	
CD	NPTS	NUMBER OF	ENERGY (1	ESTAR) P	OINTS FOR	
CD	NDEC	EACH ISOT	OPE			χ.
CD	NREG	NORBER OF	R HOMOGENI	COUS PRO	BLEMS (NGEC	M = 0
CD	•	NREG=2 FO	R PIN CELI	S (NGEO	M=2) AND IF	
CD		MA	XHTM.GT.O	•	, .	•
CD		\bullet NREG=1 +	NUMBER OF	SLAB PE	GIONS FOR S	LAB
CD		PR	OBLEMS (NO	G = 0 M = 1)	AND IF MAXH	TM.GT.O
CD	• •	(MAXHTM 1	S THE MAXE OCENEOUS I	LMUM NUM	BER OF MATE	SRIALS IN
CD	NUMRES	NUMBER OF	UNRESOLVI	LU RESON	ANCE MATERI	ALS
02						
					•	
C						
CS CS	FILE ST	RUCTURE				-
CS	RECO	RD TYPE	· '		PRESENT IF	-
CS	====:	=======================================	=========	====	.============	-============
CS	SPEC	IFICATIONS			ALWAYS	-
CS	MATE	RIAL NAMES	DEC		ALWAYS	-
CS	*****	REPEAT FOR NUM	RES			-
CS CS	* MATE	RTAL SPECIFICA	TTON	. `	ALWAYS	-
CS	* ENER	GY POINT INDEX	1100		ALWAYS	· _
CS ·	* ******* (REPEAT NISO TI	MES)	λ.		-
CS	* * ENER	GY SPECIFICATI	ONS		ALWAYS	-
CS	* * ***** (REPEAT FOR 1+N	O. OF REGI	IONS		-
CS	* * *	IN WHICH MATER	IAL IS TRI	SATED	1	-
CS CS	ት ቶ ቶ ቋ ቋ ድ ሮ እዑጦ	NETEROGENEOUSL	IJ TONS		ATWAYS	· –
CS CS	* * * FTSS	TON CROSS SECT	IONS		IFI.EO.1	· _
cs	* * * TOTA	L CROSS SECTIO	NS		ALWAYS	-
CS	*****	•	~			-
С						
(

-427-

APPENDIX D. MC²-2 Binary Files. UNRES (Contd.)

CR		ENERGY POINT INDEX (TYPE 4)
C		-
cc		ALWAYS PRESENT
С		-
CL	(NPTS (I), I=1, NISO) -
CW	NISO	$\sim 10^{-10}$ m s $\sim 10^{-10}$
C "	И.1.90	- -
C		
		,
~		
CR .	*****	FNERGY SDECTERTCAPTIONS (TYDE 4)
C		-
čc		ALWAYS PRESENT
С		- · · · · · · · · · · · · · · · · · · ·
CL	(ESTAR	(I), I=1, NPTS)
C C M	NDTC	-
C w	NP.15	
CD	ESTAR	ENERGY POINTS -
С		-
C		
C		
CR		CAPTURE CROSS SECTIONS (TYPE 6)
CC	•	ALWAYS PRESENT FOR HOMOGENEOUS MIXTURE, ALSO PRESENT -
cc		FOR EACH HETEROGENEOUS REGION IN WHICH CURRENT MATERIAL
CC		IS TREATED HETEROGENEOUSLY -
С		-
CL	(SIGCA	P(I), I=1, NPTS
C W	אסידק	
C	11	· · · · · · · · · · · · · · · · · · ·
CD	SIGCAP	UNRESOLVED RESONANCE CAPTURE CROSS SECTIONS -
С		-
C		
C		
CR		FISSION CROSS SECTIONS (TYPE 7)
C		-
СС		ALWAYS PRESENT FOR HOMOGENEOUS MIXTURE IF IFI.EQ.1 -
CC		ALSO PRESENT FOR EACH HETEROGENEOUS REGIONS IF IFI.EQ.1 -
CC		AND CURRENT MATERIAL IS TREATED HETEROGENEOUSLY IN REGION -
CI	(STOFT	S (T) T=1 NPTS)
C	lararr	
		• *

4

-428-

APPENDIX D. MC²-2 Binary Files. UNRES (Contd.)

5

€_

~			-
C CR	SPECIFIC	ATIONS (TYPE 1)	
C			_
сс	ALWAYS P	RESENT	-
с			-
CL	NUMRES, NREG, NP	TMAX, MAXISO	-
С			-
CW	4		
С	>	1	-
ĊD	NPTMAX	MAXIMUM NUMBER OF POINTS FOR ANY MATERIAL	-
CD		IN THE LIBRARY	-
CD	MAXISO	MAXIMUM NUMBER OF ISOTOPES IN THE MIXTURE	-
C			-
C			
C			
C=	MATERIAL		-
C			-
čc	ALVAYS PI	RESENT	-
c			-
ČL.	(IINRMAT(T), T=1)	NUMBES)	-
C			-
ĊW	MULT*NUMRES		-
C			-
CD	UNRMAT	DOUBLE PRECISION (R*8) UNRESOLVED RESONANCE	-
CD		MATERIAL NAMES	-
C ·			-
C			
	1	·	
C			
CR	MATERIAL	SPECIFICATION (TYPE 3)	-
С			+
CC	ALWAYS PI	RESENT	-
C		:	-
CL	N150,1F1		
C	2	١	-
CW	2		_
C C	TDT	ПТССТІЛ МАЛЛЛІТАТ ТАПЛУ	-
CD		TET-O ROD NON RECEILE MAMERIAL	-
CD	1	LFI=U FUK NUN→FISSILE MATEKIAL	
CD		ILT=1 FOR FISSIDE MATERIAL	-
C i	`	·	
C			
~	_		
C			~ ~

-429-

APPENDIX D. MC²-2 Binary Files. UNRES (Contd.)

CW NPTS С UNRESOLVED RESONANCE FISSION CROSS SECTIONS CD SIGFIS С Ċ-C-----CR TOTAL CROSS SECTIONS (TYPE 8) · C ALWAYS PRESENT FOR HOMOGENEOUS MIXTURE, ALSO PRESENT СС FOR EACH HETERUGENEOUS REGION IN WHICH CURRENT MATERIAL CC IS TREATED HETEROGENEOUSLY ÜÜ С (SIGTOT(I), I=1, NPTS)CL С CW NPTS С UNRESOLVED RESONANCE TOTAL CROSS SECTIONS CD SIGTOT С ._____ С

CEOF

-430-

APPENDIX E

BPOINTER, A DYNAMIC STORAGE ALLOCATION PROGRAM

,

APPENDIX E

BPOINTER, A Dynamic Storage Allocation Program

1. Description of Subprogram Package

BPOINTER is a FORTRAN subprogram package which was developed to alleviate bookkeeping chores associated with the use of dynamic storage allocation techniques.

Programs which use the BPOINTER capability tend to be structured in subroutine form. A control routine is used to define one or two large blocks of storage (called the container arrays), and make the appropriate calls to BPOINTER to control the allocation of storage within these block(s). Calls to calculational subroutines transmit pointers corresponding to appropriate array locations through the calling sequences. All BPOINTER capabilities are accessed through an appropriate call to an entry point, subroutine, or function subprogram. The following capabilities are available in the BPOINTER system:

- (a) Storage of data in and retrieval of data from the container array, via user defined variable arrays.
 - (b) Purge of variable arrays stored in the container array.
 - (c) Automatic "cleanup" of the container array when more storage is required.
 - (d) Re-definition of array sizes without loss of data already stored in the array.
 - (e) Array dump of selected integer, floating point or BCD arrays in a prescribed format.
 - (f) Trace dumps of BPOINTER activities.
 - (g) Status reports of the BPOINTER tables,

Detailed program documentation including flow charts, common block information and subprogram descriptions is available in Reference 1. This Appendix is intended to provide a brief description of how the program package operates. The major differences between the IBM and CDC standalone versions of the program package are also noted.

The short example listed in Fig. 27 is intended to illustrate the structure of a program using the BPOINTER package. This example demonstrates the manner in which a container is allocated, pointers defined and used, and the container released.

Brief descriptions of all the BPOINTER entry points, subroutines and functions are given in Table XIV.

All dynamically allocated arrays are addressed relative to the common block /ARRAY/ which contains a single array element, BLK(1). In the IBM version of the code the element must be declared as DOUBLE PRECISION. A second common block /ARRAY2/ is used in the CDC version of BPOINTER to address arrays allocated to a large core memory container. This common block also contains a single array element BLKECS(1) which must be declared a LEVEL 2 variable. The equivalent of the large core memory container on IBM equipment is a second container which may be given a HIARCHY 1 location but is addressed in precisely the same manner as the first (SCM) container. The one word assigned to the container by the source language program provides a reference address. At execution time machine language routines (ALLOC1, ALLOC2 on IBM, MEMGET1, MEMGET2 on CDC) are used to obtain the addresses of core which are available to the program for the allocation of data arrays. These blocks of core are allocated in the following manner.

(a) IBM allocation

The standard IBM macro instructions GETMAIN and FREEMAIN are used to allocate and free consecutive words of core which are available to the program. The designations subpool 1 and 2 are assigned to the bulk (LCM) and fast (SCM) containers respectively. Since allocations are performed in units of 256 (eight byte) words, it is most efficient to request blocks of core in such multiplies;

(b) CDC allocation

The COMPASS routine MEMGET uses the standard CDC macro instruction MEMORY to determine the jobs SCM and LCM field length. The core available as a container for the BPOINTER SCM arrays is determined by subtracting the address of blank common from the SCM field length. Thus a program using the CDC version of BPOINTER should not use blank common. Blank common is used for this purpose because of the CDC loader convention which places blank common after all other program sections in core. Although blank common is used in this manner to determine the available core for a container, arrays in the container are addressed relative to the common block /ARRAY/ as noted above, The BPOINTER program package accounts for the offsets between the address of /ARRAY/ in core and the address of the container. It should also be noted that the conventions used by BPOINTER are such that the first word of the container is not in general set to the first word of blank common. This is important since the loader convention noted above is not adhered to by all CDC Systems. The user is therefore responsible for providing enough SCM memory to accommodate the program, any SCM buffers, and the BPOINTER container as there is currently no effective check to make sure that data stored in the BPOINTER container do not overlap code. It is assumed by the CDC version of BPOINTER that the LEVEL 2 common block /ARRAY2/ is addressed as the first word of LCM and the entire LCM field length is assumed to be available to BPOINTER for its LCM container.

The letters M and B are used as neumonics within BPOINTER to designate routines which operate on the SCM and LCM containers respectively. Thus PUTM allocates an array in the SCM container while PUTB allocates an array which must be referenced on CDC equipment as a LEVEL 2 array. On IBM equipment without HIARCHY support (e.g. 370/195) the two containers are equivalent. The distinctions noted above between the two dynamic containers are important on CDC equipment where the containers are addressed quite differently and on IBM equipment with HIARCHY support where access to the BULK core container (HIARCHY 1, subpool 1) is significantly slower than access to the MAIN core container (HIARCHY 0, subpool 2).

С BPOINTER EXAMPLE С DEFINE CONTAINER COMMON BLOCK С REAL*8 BLK, FLUX, POWER COMMON/ARRAY/BLK(1) DIMENSION BLK4(1) EQUIVALENCE (BLK(1), BLK4(1)) DATA FLUX/6HFLUX /, POWER/6HPOWER /, MAXSIZ/10000/ DATA 14/4/, 18/8/, 10/0/, NG/27/ С С ALLOCATE CONTAINER WITH MAXSIZ WORDS OF SCM AND NO LCM С CALL BULK (IO) CALL POINTR (BLK, MAXSIZ, IO) С С ALLOCATE SPACE FOR ARRAYS POWER, FLUX AND CURRENT C CALL PUTM (POWER, 18, NG, IPOWR) CALL PUTM (FLUX, I4, 2*NG, IFLUX) С DETERMINE POINTER FOR SUB-ARRAY CURRENT WHICH FOLLOWS THE С С NG SINGLE PRECISION WORDS FOR THE ARRAY FLUX С ICURNT=IPT2 (IFLUX, NG, IO) С Ç CHECK ON BPOINTER ERROR С IF (IPTERR (DUM).GT.0) PRINT 500 500 FORMAT(1H0,14HBPOINTER ERROR) С С CALL SUBROUTINE INIT TO USE THESE ARRAYS С CALL INIT (BLK (IFLUX), BLK (IPOWR), BLK4 (ICURNT), NG) С С FREE CONTAINER AND RETURN С CALL FREE RETURN END SUBROUTINE INIT (PHI, POWER, CURENT, NG) С С USE BPOINTER ARRAYS JUST AS ANY OTHER VARIABLES С REAL*8 POWER DIMENSION PHI(1), POWER(1), CURENT(1) DO 10 I=1, NGPHI(I) = 1.0POWER (I) = 3.1E+06CURENT(I) = .333**10 CONTINUE** RETURN END

Fig. 27. BPOINTER Example

TABLE XIV. BPOINTER Subprogram Descriptions

POINTR (008710 - 010270)*	Initializes tables of dynamic allocation program package and calls ALLOC1 and ALLOC2 to allocate con- tainer(s) for variably dimensioned arrays.
PUTPNT/PUTBLK (010280 - 010590)	Dummy routine calls PUTM to allocate array storage.
BULK (010600 - 010720)	Sets number of words of BULK(LCM) core to be allocated,
FREE (010730 - 011020)	Calls FREEL and FREE2 to release con- tainers allocated by calls from sub- routine POINTR.
WIPOUT/CLEAR (011030 - 011780)	Deletes a named array from BPOINTER tables; zeroes all locations assigned to a named array.
GETPNT/GETN/DUMP (011790 - 013310)	Returns pointer for a named array; return index in BPOINTER tables of a named array; controls printing of a named array.
IGET (013320 - 013730)	Returns pointer for a named array,
IPT2 (013740 - 013870)	Returns pointer to a sub-array relative to a single precision word length container.
PUTM/PUTB (013880 - 015460)	Enters named arrays into fast and bulk(LCM) containers respectively.
IPTERR/NNAMSF (015470 - 015740)	Returns number of BPOINTER errors; returns number of named arrays in BPOINTER tables.
ILAST/ILASTB (015750 - 015940)	Returns word number of first available word in SCM/LCM container.
REDEF (015950 - 016130)	Dummy routine calls REDEFM to redefine size and/or location of named array.
REDEFM/REDEFB (016140 - 017990)	Redefine the size and/or location of named array within BPOINTER tables and containers.

* Numbers in brackets are card sequence numbers of routine on MC^2-2 program tape.

TABLE XIV. BPOINTER Subprogram Descriptions (Contd.)

PURGE/PURG	GEB	(018000 - 019270)	Sift storage in SCM/LCM containers to eliminate unused blocks created by WIPOUT calls.
STATUS	(0192	280 - 019660)	Edits status of BPOINTER tables.
PRTI1	(0196	570 - 019790)	Prints half word integer array from SCM container.
PRTI1E (019800 - 020000)		300 - 020000)	Prints half word integer array from LCM container.
PRTI2	(0200	010 - 020090)	Prints full word integer array from SCM container.
PRTI2E	(020]	00 - 020270)	Prints full word integer array from LCM container.
PRTR1/PRT4	A1	(020280 - 020450)	Prints full word real array from SCM container.
PRTR1E/PR	TA1E	(020460 - 020720)	Prints full word real array from LCM container.
PRTR2/PRT	A2	(020730 - 020930)	Prints double word read array from SCM container.
PRTR2E/PRTA2E (020940 - 021230)		(020940 - 021230)	Prints double word real array from LCM container.

ć

t

APPENDIX F

MC²-2 LIBRARY GENERATION

)

APPENDIX F.

MC²-2 Library Generation

The MC^2-2 library contains eight files, MCC2F1-MCC2F8, that were processed from the ENDF/B-IV data files by the code ETOE-II. Appendix C contains a description of these files. The user has the option of obtaining the binary MC^2-2 library or generating the MC^2-2 library from BCD card images on magnetic tapes. The library data are briefly summarized in Table IX.

The binary library for IBM users is contained on one 9 track, nonlabeled, LRECL=X, RECFM=VBS, and BLKSIZE=6447 magnetic tape with a recording density of 1600 BPI. The eight files are written in sequential order MCC2F1-MCC2F8. A sample of the job control cards required to copy these files from tape to a direct access device (e.g. disk pack) is given in Fig. 22. The binary library for CDC users is contained on two 7 track, non-labeled, and x-mode binary tapes written with a recording density of 800 BPI. The first binary tape contains the files MCC2F1-MCC2F4, MCC2F7, and MCC2F8. The second tape contains the files MCC2F5 and MCC2F6. These two binary tapes were generated at BERKELEY. Figure 28 displays the control cards necessary to read the two 7 track x-mode binary tapes and write a 9 track 1600 BPI tape. The BERKELEY example displays the generation of a non-labeled, phase encoded, and x-mode binary-odd parity tape. The BROOKHAVEN example displays the generation of a labeled, SCOPE standard format tape. The 9 track tape which is generated in each case is used directly in the execution of the MC^2-2 problem as displayed in Fig. 25.

The user may generate the MC^2-2 library from three BCD data tapes which contain the same eight file library data in BCD format along with a BCD tape which contains a Fortran program, MC^2-2 LIBGEN, which reads the three BCD data tapes and writes the eight binary files. All four tapes are 7 track, unlabeled, 40 card images per physical block, and written at 800 BPI. Figure 29 displays the CDC control cards used to read the three BCD data tapes as well as the BCD program tape, MC^2-2 LIBGEN. The control cards also show the compilation and execution of this program which generates the eight binary MC^2-2 library files. The data tapes which are physically labeled MC2BCDLIB1, MC2BCDLIB2, and MC2BCDLIB3 are assigned the data set names TAPE11, TAPE12, and TAPE13 respectively in the program execution. The eight binary library files are written on one data set named TAPE21, and these data are written onto a 9 track 1600 BPI tape. This binary library tape is used directly in the execution of the MC^2-2 problem as shown in Fig. 25.

,

BERKELEY

BROOKHAVEN

(JOBCARD)

STAGE(IN1,D8,P3,MT,R,nnnnn,6F) STAGE(IN2,D8,P3,MT,R,nnnnn,2F) COPYBF(IN1,BINLIB,4) COPYBF(IN2,BINLIB,2) COPYBF(IN1,BINLIB,2) REWIND(IN1,IN2,BINLIB) STAGE(BINLIB,D9,P3,NT,W,nnnnn,8F) 6/7/8/9 (JOBCARD) ACCOUNT(Name,NNNN) STAGE(IN1,HY,VSN=Nnnnnn) STAGE(IN2,HY,VSN=Nnnnnn) STAGE(BINLIB,POST,E,PE,VSN=Knnn) FILE(IN1,RT=X,BT=C,MBL=5120) FILE(IN2,RT=X,BT=C,MBL=5120) COPYBF(IN1,BINLIB,4) COPYBF(IN1,BINLIB,4) COPYFB(IN1,BINLIB,2) REWIND(IN1,IN2,BINLIB) 6/7/8/9

-440

Fig. 28. Generation of MC²-2 Library from Two X-Mode Binary Library Tapes that were Generated at Berkeley

BERKELEY

(JOBCARD)

SCF(R=TAPE11,RL=80,BF=40,D8,nnnnn) SCF(R=TAPE12,RL=80,BF=40,D8,nnnnn) SCF(R=TAPE13,RL=80,BF=40,D8,nnnnn) SCF(R=LIBGN,RL=80,BF=40,D8,nnnnn) FTN(I=LIBGN,OPT=1) LINK,X. STAGE(TAPE21,D9,P3,NT,W,nnnnn,8F) 6/7/8/9 BROOKHAVEN

(JOBCARD) ACCOUNT (Name, NNNN) STAGE(TAPEA1,HY,VSN=Nnnnnn) STAGE(TAPEA2,HY,VSN=Nnnnn) STAGE (TAPEA3, HY, VSN=Nnnnn) STAGE(LIBGN1,HY,VSN=Nnnnn) STAGE(TAPE21,POST,E,PE,VSN=Knnn) FILE (TAPEA1, RT=F, FL=80, RB=40, BT=K, CM=YES) FILE(TAPEA2, RT=F, FL=80, RB=40, BT=K, CM=YES) FILE(TAPEA3, RT=F, FL=80, RB=40, BT=K, CM=YES) FILE(LIBGN1,RT=F,FL=80,RB=40,BT=K,CM=YES) COPYBF(TAPEA1, TAPE11, 1) COPYBF(TAPEA2, TAPE12, 1) COPYBF(TAPEA3, TAPE13, 1) COPYBF(LIBGN1,LIBGN,1) REWIND(TAPE11, TAPE12, TAPE13, LIBGN) RETURN (TAPEA1, TAPEA2, TAPEA3, LIBGN1) FTN(I=LIBGN,OPT=1) FEWIND(LGO) LOAD(LGO) EXECUTE. REWIND(TAPE21) 6/7/8/9

Fig. 29. Generation of MC^2-2 Library from Three BCD Data Tapes Using the Fortran Program MC^2-2 LIBGEN

441-

ACKNOWLEDGMENTS

Many people made significant contributions to the development of the program system MC^2-2 so that it would be impossible to acknowledge them all. However, the work of W. M. Stacey, Jr., in neutron slowing down theory and R. N. Hwang in resonance theory were instrumental in the theoretical modelling for the program. The assistance provided by A. Olson in the development of the RABANL module and of E. Pennington in leading the authors through the vagaries of ENDF/B were most appreciated. Special thanks are also due to the users of MC^2-2 . R. Prael, D. Wade, M. Lineberry, and C. Beck are among those users whose special efforts in specification and verification of the code are largely responsible for any of its good features. The work of Pat Jones, Linda Juergens, and Lyn Young in wading through the many drafts of this report and making sense of it all is most appreciated.

.

-443-

REFERENCES

- L. C. Just, H. Henryson, II, A. S. Kennedy, S. D. Sparck, B. J. Toppel, and P. M. Walker, The System Aspects and Interface Data Sets of the Argonne Reactor Computation (ARC) System, Argonne National Laboratory, ANL-7711, (April, 1971).
- 2. Standard Interface Files and Procedures for Reactor Physics Codes, Version III, Los Alamos Scientific Laboratory, LA-5486-MS (Feb. 1974).
- 3. M. K. Drake, ed., Data Formats and Procedures for the ENDF Neutron Cross Section Library, Brookhaven National Laboratory, BNL 50274 (Oct. 1970).
- 4. C. G. Stenberg, ETOE-2, A Program for Data Conversion of ENDF/B to MC^2-2/SDX , Argonne National Laboratory, to be published.
- 5. P. H. Kier and A. A. Robba, *RABBLE*, A Program for Computation of Resonance Absorption in Multiregion Reactor Cells, Argonne National Laboratory, ANL-7326 (April 1967).
- 6. A. P. Olson, RABID: An Integral Transport-theory Code for Neutron Slowing Down in Slab Cells, Argonne National Laboratory, ANL-7645 (April 1970).
- H. Henryson, II, B. J. Toppel, and C. G. Stenberg, ETOE-2/MC²-2/SDX, Multigroup Nautron Cross Section Processing, Seminar on Codes for Nuclear Data Processing, Ispra 5-7 (December 1973).
- H. Hurwitz, Jr., and P. F. Zweifel, Slowing Down of Neutrons by Hydrogenous Moderators, Journal of Applied Physics, <u>26</u>, 8, 923 (1955).
- 9. W. M. Stacey, Jr., The Effect of Anisotropic Scattering Upon the Elastic Moderation of Fast Neutrons, Nucl. Sci. Eng., 44, 194 (1971).
- B. J. Toppel, A. L. Rago, and D. M. O'Shea, MC², A Code to Calculate Multigroup Cross Sections, Argonne National Laboratory, ANL-7318 (June 1967).
- R. L. Hellens, R. W. Long, and B. H. Mount, Multigroup Fourier Transform Calculation Description of MUFT-III Code, WAPD-TM-4, Bettis (July 1956).
- 12. W. M. Stacey, Jr., Continuous Slowing Down Theory Applied to Fast-Reactor Assemblies, Nucl. Sci. Eng., 41, 381 (1970).
- W. M. Stacey, Jr., Advances in Neutron Continuous Slowing Down Theory, Proc. Conf. New Methods in Reactor Physics, Kiamesha Lake, New York, USAEC-CONF-720901, Book 1, 143 (Sept. 1972).
- 14. G. Goertzel and E. Greuling, An Approximate Method for Treating Neutron Slowing Down, Nucl. Sci. Eng. 7, 69 (1960).

•.

- 15. R. V. Meghreblian and D. K. Holmes, *Reactor Analysis*, McGraw-Hill Book Company, Inc. New York (1960).
- 16. H. Amster, Heavy Moderator Approximations in Neutron Transport Theory, Journal of Applied Physics, <u>29</u>, 4, 623 (1958).
- 17. P. F. Zweifel and H. Hurwitz, Jr., Transformation of Scattering Cross Sections, Journal of Applied Physics, 25, 10, 1241 (1954).
- 18. R. F. Berland, CHAD Code to Handle Angular Data, Atomics International, NAA-SR-11231 (Dec. 1965).
- 19. N. Corngold, Slowing Down of Neutrons in Infinite Homogeneous Media, Proc. Phys. Suc. LXK, II-Λ, 793 (1957).
- H. Henryson, II, C. G. Stenberg and B. J. Toppel, Calculation of Elastic Scattering Matrices, Applied Physics Division Annual Report July 1, 1969 to June 30, 1970, Argonne National Laboratory, ANL-7710 (Jan. 1971).
- 21. H. Henryson, II, Multigroup Elastic Scattering Cross Sections for Heavy Elements, Nucl. Sci. Eng., <u>43</u>, 235 (1971).
- 22. A. Olson, A Fast Exponential Subroutine for Calculating Collision Probabilities on the IBM/360, Applied Physics Division Annual Report July 1, 1969 to June 30, 1970, Argonne National Laboratory, ANL-7710 (Jan. 1971).
- 23. M. Segev, Group Transfer Matrices and Their Relation to Basic Cross Section Data, Nucl. Sci. Eng., <u>45</u>, 269 (1971).
- 24. D. E. Neal, G. K. Leaf and A. S. Kennedy, The ARC System One-Dimensional Diffusion Theory Capability, DARC1D, Argonne National Laboratory, ANL-7715, 357 (May 1971).
- 25. R. N. Hwang, Efficient Methods for the Treatment of Resonance Integrals, Nucl. Sci. Eng., <u>52</u>, 157 (1973).
- 26. L. Dresner, Resonance Absorption in Nuclear Reactors, Pergamon Press, New York (1960).
- 27. R. N. Hwang, Application of Statistical Theory and Multilevel Formalism to Doppler Effect Analysis-II, Nucl. Sci. Eng., 36, 82 (1969).
- 28. R. N. Hwang, Doppler Effect Calculations with Interference Corrections, Nucl. Sci. Eng., <u>21</u>, 523 (1965).
- 29. Herbert Goldstein, Fundamental Aspects of Reactor Shielding, Addison-Wesley, 355-357 (1959).
- 30. A. N. Lowman, Project Director, Tables of Associated Legendre Functions, Columbia University Press, New York xix (1945).

- 31. W. Gautschi, Computational Aspects of Three-Term Recurrence Relations, Siam Review, 9, 1, 24 (1967).
- 32. D. M. O'Shea and H. C. Thacher, Computation of Resonance Line Shape Functions, Trans. Am. Nucl. Soc., <u>6</u>, 36 (1963).
- 33. Milton Abramowitz and Irene A. Stegun, Editors, ibid, p. 328.
- 34. Milton Abramowitz and Irene A. Stegun, Editors, Handbook of Mathematical Functions with Formulas, Graphs, and Mathematical Tables, U. S. Dept. of Commerce, National Bureau of Standards, Applied Mathematics Series 65, 328 (June 1964).
- 35. R. N. Hwang and H. Henryson, II, Critical Examination of Low-Order Quadratures for Statistical Integrations, Trans. Am. Nucl. Soc., 22, 712 (1975).
- 36. N. M. Steen, G. D. Byrne and E. M. Gelbard, Gaussian Quadratures for the Integrals

$$\int_{0}^{\infty} \exp(-x^{2}) f(x) dx \quad and \quad \int_{0}^{b} \exp(-x^{2}) f(x) dx$$

Math. Comp., 23, 107 (1969).

37. A. P. Olson, Gaussian Quadratures for $\int_{1}^{\infty} \exp(-x)f(x)dx/x$ and

 $\int_{1}^{\infty} g(x) dx/_{x} m, Math. Comput., <u>23</u>, 106, 447 (1969).$

- 38. I. Gargantini and T. Pomentale, Rational Chebyshev Approximations to the Bessel Function Integrals K_i (x), Communications of the ACM, <u>7</u>, 727 (1964).
- 39. W. M. Stacey, Jr., The Effect of Wide Scattering Resonances on Neutron Multigroup Cross Sections, Nucl. Sci. Eng., <u>47</u>, 29 (1972).
- C. C. Klimasauskas, D. R. Ferguson and H. Henryson, II, SIO - An Access Method for Direct Access Device Scratch File Management, Argonne National Laboratory, to be published.
- 41. E. L. Wachspress, Iterative Solution of Elliptic Systems, Prentice-Hall, Englewood Cliffs, 26 (1966).